Haines Borough Borough Assembly Meeting #452 AGENDA

September 1	3 <u>,</u> 2022	-6:30pm Location: In Assembly Chambers and on ZOOM
Douglas Olerud Mayor		Haines Borough ZOOM information (You must download Zoom first) Webinar ID: 847 9571 9137
Tyler Huling Assembly Member		Passcode: 134797
Gabe Thomas Assembly Member	1. 2.	CALL TO ORDER/PLEDGE TO THE FLAG/ROLL CALL APPROVAL OF AGENDA & CONSENT AGENDA
Cheryl D. Stickler Assembly Member		[The following Consent Agenda items are indicated by an <u>asterisk</u> (*) and will be enacted by the motion to approve the agenda. There will be no separate discussion of these items unless an assembly member or other person so requests, in which
Debra Schnabel Assembly Member		event the asterisk will be removed and that item will be considered by the assembly on the regular agenda.
Paul Rogers Assembly Member		Consent Agenda: 3 – Approve Minutes from 8-24-22 Regular Assembly Meeting
Jerry L. Lapp Interim Assembly	*3.	APPROVAL OF MINUTES- 8-24-22 Regular Assembly Meeting Minutes
Member	4.	PUBLIC COMMENTS – [For any topics <u>not</u> scheduled for public hearing. Individual comments are limited to 3 minutes] <i>Note: during this section of the agenda, the assembly will listen and take notes. No official action will be taken at this time.</i>
Annette Kreitzer	5.	ASSEMBLY COMMENTS
Alekka Fullerton	6.	MAYOR'S REPORT AND COMMENTS
Borough Clerk	7.	PUBLIC HEARINGS
		An Ordinance of the Haines Borough Assembly allowing a Temporary Exemption from Property Tax Levy for Highland's Estates, Inc. Hilltop Subdivision.
		No motion is necessary at this meeting since this Ordinance is already scheduled for its second public hearing on September 27, 2022. Any desired amendment is appropriate at this time.
		 B. <u>Ordinance 22-08-626</u> – First Hearing An Ordinance of the Haines Borough amending Haines Borough Code Title 12 to Clarify Inspection Costs.
		No motion is necessary at this meeting since this Ordinance is already scheduled for its second public hearing on September 27, 2022. Any desired amendment is appropriate at this time.
		C. <u>Ordinance 22-08-627</u> – First Hearing An Ordinance of the Haines Borough determining whether recently foreclosed property deeded to the borough shall be retained for a public purpose or sold.
		No motion is necessary at this meeting since this Ordinance is already scheduled for its second public hearing on September 27, 2022. Any desired amendment is appropriate at this time.
	8. 3	STAFF/FACILITY REPORTS AND DIRECTION TO OFFICERS A. Borough Manager Report- To be provided at meeting. B. CFO Report – Comparison of General Sales Tax Revenue FY18-FY22
	9.	COMMITTEE/COMMISSION/BOARD REPORTS & APPROVED MINUTES – A. Port and Harbor Advisory Board – Minutes from June 23, 2022
	10. L	JNFINISHED BUSINESS – None

11. NEW BUSINESS

A. Resolutions [Individual comments are limited to 3 minutes]

1. <u>Resolution 22-09-994</u>

A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a construction contract with Dawson Construction, Inc for the Haines Boat Harbor E Float Repair for an amount not to exceed \$59,000.00. <u>Motion</u>: Adopt Resolution 22-09-994

2. Resolution 22-09-995

Resolution 22-09-995 A Resolution of the Haines Borough Assembly authorizing the Borough Manager to amend Ioan CW 395261-S with the Alaska Department of Environmental Conservation for Phase II of the Wastewater Influent and Lift Station Pump Upgrade to include Beach Road Lift Station Sewer Main Extension for an amended amount not to exceed \$659,867.

Motion: Adopt Resolution 22-09-995

3. <u>Resolution 22-09-996</u>

A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a contract with Turnagain Marine Construction, for the Progressive Design-Build of the Lutak Dock Replacement Project for an amount not-to-exceed \$310,000.00. <u>Motion</u>: Adopt Resolution 22-09-996

4. <u>Resolution 22-09-997</u>

A Resolution of the Haines Borough Assembly authorizing the Election Supervisor to appoint Election Officials for the regular election to be held October 4, 2022, and establishing the wages. <u>Motion</u>: Adopt Resolution 22-09-997

5. <u>Resolution 22-09-998</u>

A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a change order with Glacier Construction, Inc dba Southeast Road Builders, for additional Young Road repairs for an amount not-to-exceed \$128,173.00. Motion: Adopt Resolution 22-09-998

6. <u>Resolution 22-09-999</u>

A Resolution Of The Haines Borough Assembly joining the Alaska Municipal League in expressing concern in the consideration of a Constitutional Convention, and taking an active role in educating the public and policy makers about its potential impact. <u>Motion</u>: Adopt Resolution 22-09-999

B. Ordinances for Introduction

Note: Five minutes allowed for the assembly to determine if a proposed ordinance has merit and should be advanced to a public hearing.

1. Ordinance 22-09-628

An ordinance of the Haines Borough reimbursing funds withdrawn to pay FY20 and FY21 school bond debt underfunded by the State of Alaska.

This ordinance is being introduced by the Mayor.

11. NEW BUSINESS

B. Ordinances for Introduction

1. Ordinance 22-09-628 - Continued

Motion: Introduce Ordinance 22-09-628 and schedule it for its first public hearing 9/27/22 and a second public hearing for 10/11/22.

2. Ordinance 22-09-629

An Ordinance of the Haines Borough Adopting a Revised Fee Schedule for Harbor Facility Usage.

This ordinance is being introduced by the Mayor and is recommended by the Harbormaster and the Port and Harbor Advisory Committee. Motion: Introduce Ordinance 22-09-629 and schedule it for its first public hearing 9/27/22 and a second public hearing for 10/11/22.

3. Ordinance 22-09-630

An Ordinance of the Haines Borough Assembly amending Haines Borough Code Title 18 Intent of Mud Bay Planning/Zoning District. *This ordinance is recommended by the Government Affairs and Services Committee and is being introduced by Assembly Member Schnabel.* <u>Motion</u>: Introduce Ordinance 22-09-630 and schedule it for its first public hearing 9/27/22 and a second public hearing for 10/11/22.

C. Other New Business

- Memo from the Manager regarding possible compensatory relief for property owners affected by the December 2020 Weather Event. <u>Proposed Motion</u>: Direct the Manager to gauge the interest in BRIC, HMGP or State Land Swap of property owners with qualifying property from the December 2020 Weather Event.
- 2. Recommendation from Government Affairs and Services Committee to fully fund all FY23 Non-Profit Grant Applications, namely:

Haines Friends of Recycling	\$ 7,190
KHNS	\$ 5,000
SAIL	\$ 8,000
Southeast Senior Services	\$15,000

<u>GASC Proposed Motion</u>: All requests (HFR, SAIL, KHNS, Senior Center) be funded and the balance of funds (\$2,810) be available for application as overage for capital projects.

12. CORRESPONDENCE

A. Alaska DNR Notice of Preliminary Decision- Haines Huts

- 13. SET MEETING DATES
- 14. PUBLIC COMMENT
- 15. ANNOUNCEMENTS/ASSEMBLY COMMENTS
- 16. ADJOURNMENT

Haines Borough Borough Assembly Meeting #451 August 24, 2022 MINUTES Draft

1. <u>CALL TO ORDER/PLEDGE TO THE FLAG/ROLL CALL</u>: Mayor **OLERUD** called the meeting to order at 6:30 pm in the Assembly Chambers and led the pledge to the flag.

<u>Present</u>: Mayor OLERUD, Assembly Members Gabe THOMAS, Debra SCHNABEL, Paul ROGERS, Cheryl STICKLER and Tyler HULING. Interim Assembly Member Jerry LAPP was seated at the dais after being appointed and sworn during item 5 below.

<u>Staff Present</u>: Annette KREITZER/Borough Manager, Alekka FULLERTON/Clerk, Steven AUCH/Tourism Director.

<u>Visitors Present</u>: Sharon RESNICK, Sabrina STICKLER, Zach WENTZEL, Margarette JONES, Karen GARCIA, Lucy SILBAUGH/CVN, Ted POPLEY, Julie KORSMEYER, Evelyna VIGNOLA and others.

2. APPROVAL OF AGENDA & CONSENT AGENDA

The following Items were on the published consent agenda indicated by an <u>asterisk</u> (*) 3 – Approve Minutes from 8-09-22 Regular Assembly Meeting

Motion: SCHNABEL moved to "approve the agenda/consent agenda," and the motion carried unanimously.

*3. APPROVAL OF MINUTES - Regular 8-09-22 Assembly Meeting

<u>Note</u>: The Minutes were approved by approval of the consent agenda: "Approval of Regular 8-09-22 Assembly Meeting."

- 4. **PUBLIC COMMENTS**: None
- 5. ASSEMBLY COMMENTS:
 - A. Interim Appointment to Assembly Seat F

<u>Motion</u>: SCHNABEL moved to "Appoint Jerry Lapp to Assembly Seat F to serve until the new Assembly member is seated from the regular October 4, 2022 Municipal Election," and the motion carried unanimously.

B. **Oath of Office** – Mayor Olerud administered the Oath of Office to Jerry Lapp and Interim Assembly Member Lapp took his seat at the dais and participated in the remainder of the meeting.

6. MAYOR'S REPORT AND COMMENTS:

Mayor **OLERUD** reported on the AML Meeting he recently attended; his tour of Sitka with the Sitka Mayor; Meetings with Skagway Officials; Discussion about the Alaska Constitutional Convention; Resignation of Nik LaCroix from the Planning Commission; Jerry Lapp's absence for at least two meetings from the Planning Commission.

7. PUBLIC HEARINGS

A. <u>Ordinance 22-07-623</u> – Second Hearing

An Ordinance of the Haines Borough Placing a Proposition on the October 4, 2022 Haines Borough General Election Ballot to Reallocate Sales Tax Revenue; Increase Townsite Service Area Sales Tax and Increase Transient Lodging Sales Tax.

Mayor **OLERUD** opened and closed the public hearing at 6:41 pm since no member of the public chose to speak on this issue.

Motion: SCHNABEL moved to "Adopt Ordinance 22-07-623," and the motion FAILED unanimously in a roll call vote.

7. **PUBLIC HEARINGS** (Continued)

B. <u>Ordinance 22-07-624</u> – Second Hearing An Ordinance of the Haines Borough, Providing for the Addition or Amendment of Specific Line Items to the FY23 Budget.

Mayor **OLERUD** opened and closed the public hearing at 6:47 pm since no member of the public chose to speak with respect to this ordinance.

Motion: SCHNABEL moved to "Amend Ordinance 22-07-624 to recognize additional Community Assistance funding revenue in the amount of \$103,100 and additional FY21 School Debt Reimbursement in the amount of \$896,298," and the motion carried unanimously.

Motion: SCHNABEL moved to "Adopt Ordinance 22-07-624 as amended," and the motion carried unanimously.

- 8. STAFF/FACILITY REPORTS AND DIRECTION TO OFFICERS
 - A. Borough Manager Report- 8/18/22
 - **B.** Borough Clerk Report on Election 8/18/22
 - C. Planner's Memo re Platting Actions 8/10/22

9. COMMITTEE/COMMISSION/BOARD REPORTS & APPROVED MINUTES

A. Planning Commission – Approved Minutes from July 14, 2022 Meeting and Records of Decision for 35% Design of Jenae's Playground, 95% Design for Letnikof South Parking, 95% Design for Lutak Clean Metal Collection

10. UNFINISHED BUSINESS

A. Redeposit into Permanent Fund of Voter Approved Withdrawl of Funds used to pay part of the FY20 and FY21 School Bond Debt

<u>Motion</u>: STICKLER moved to "Direct the Clerk to draft a Resolution to repay the permanent fund for the deficiency in the School Bond Debt Reimbursement from FY17 and FY20,"

<u>Primary Amendment</u>: **SCHNABEL** moved to amend the word "Resolution" with the word "Ordinance", and the motion carried unanimously.

And the motion carried unanimously.

<u>Motion</u>: **THOMAS** moved to "Direct the Clerk to draft a Resolution to set aside \$448,717.50 anticipated deficiency for the FY24 School Bond Debt Reimbursement and reimbursement of \$324,075 of CIP funds previously used for School Bond Debt Reimbursement," and

Motion to Divide Question: SCHNABEL moved to divide the question and the motion carried unanimously.

Divided question:

"Direct the Clerk to draft a Resolution to set aside \$448,717.50 anticipated deficiency for the FY24 School Bond Debt Reimbursement," and the motion carried unanimously.

"Moved to reimburse \$324,075 of CIP funds previously used for School Bond Debt Reimbursement," and the motion carried unanimously.

11. NEW BUSINESS

A. Resolutions

1. <u>Resolution 22-08-990</u>

A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a change order with Glacier Construction Inc. dba Southeast Road Builders for the Young Road Repairs Contract in and an amount not to exceed \$53,650.

No member of the public chose to speak with respect to this Resolution.

Motion: SCHNABEL moved to "Adopt Resolution 22-08-990," and the motion carried unanimously in a roll call vote.

2. <u>Resolution 22-08-991</u>

A Resolution of the Haines Borough Assembly Authorizing the Borough Manager to Enter into a Contract with Theodore W. Popely, PC to Provide State Lobbying Services for \$48,000.

No member of the public chose to speak with respect to this Resolution.

Motion: THOMAS moved to "Adopt Resolution 22-08-991," and the motion carried unanimously in a roll call vote.

3. <u>Resolution 22-08-992</u>

A Resolution of the Haines Borough Assembly Authorizing the Borough Manager to Enter into a Contract with Windward Strategies to Provide Federal Lobbying Services for \$36,000.

No member of the public chose to speak with respect to this Resolution.

Motion: THOMAS moved to "Adopt Resolution 22-08-992," and the motion carried unanimously in a roll call vote.

4. <u>Resolution 22-08-993</u>

A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a contract with ADCOMM, LLC for Radio and Dispatch Technical Services (Task Order #10) for an amount not to exceed \$ 29,680.00

No member of the public chose to speak with respect to this Resolution.

Motion: THOMAS moved to "Adopt Resolution 22-08-993," and the motion carried unanimously in a roll call vote.

B. Ordinances for Introduction

1. <u>Ordinance 22-08-625</u>

An Ordinance of the Haines Borough Assembly allowing a Temporary Exemption from Property Tax Levy for Highland's Estates, Inc. Hilltop Subdivision.

Motion: **SCHNABEL** moved to "Introduce Ordinance 22-08-625 and schedule it for its first public hearing 9/13/22 and a second public hearing for 9/27/22," and the motion carried unanimously.

2. Ordinance 22-08-626

An Ordinance of the Haines Borough amending Haines Borough Code Title 12 to Clarify Inspection Costs

Motion: **SCHNABEL** moved to "Introduce Ordinance 22-08-625 and schedule it for its first public hearing 9/13/22 and a second public hearing for 9/27/22," and the motion carried unanimously.

11. <u>NEW BUSINESS</u>

- B. Ordinances for Introduction Continued
 - 3. Ordinance 22-08-627

An Ordinance of the Haines Borough determining whether recently foreclosed property deeded to the borough shall be retained for a public purpose or sold.

Motion: **THOMAS** moved to "Introduce Ordinance 22-08-625 and schedule it for its first public hearing 9/13/22 and a second public hearing for 9/27/22," and the motion carried unanimously.

C. Other New Business- None

12. CORRESPONDENCE

- A. Notice of Meeting Change of Date
- B. AMHS Winter Schedule- Public Comments due by August 26.

13. SET MEETING DATES

A. Lutak Dock Tour – Scheduled for Tuesday August 30, 2022 at 5:00pm

14. PUBLIC COMMENT: None

15. ANNOUNCEMENTS/ASSEMBLY COMMENTS

Motion: **SCHNABEL** moved to "Negotiate and prepare for assembly approval a contract for borough paid Solid Waste removal in the Haines Borough for the area south of the southern boundary line of Township 34 south (within the Haines Borough) starting April 1 2023 and continuing on an annual basis," and the motion carried unanimously.

16. ADJOURNMENT – 7:44 pm

ATTEST:

Douglas Olerud, Mayor

Alekka Fullerton, CMC, Borough Clerk



7A

Assembly Agenda Bill

Agenda Bill No.: 22- 1202

Assembly Meeting Date: 9/13/22

Business Item Description:	Attachments:
Subject: Authorizing Temporary Exemption from	1. Ordinance 22-08-625
Property Tax for Improvements for Economic Dev	2. Application 3. HBC 3.70.040
Originator:	0.1120 0.10.010
Clerk	
Originating Department:	
Lands	
Date Submitted: 8/17/22	

Full Title/Motion:

No motion necessary at this time since Ordinance 22-08-625 is already scheduled for its second public hearing September 27, 2022.

Administrative Recommendation:

Fiscal Impact:			
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets
\$0	\$0	\$ 0	\$0

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

s 🗌 No

Summary Statement:

Haines Borough Code Section 3.70.040 Local exemptions and exclusions (from Property Tax Levy) was recently amended to include additional Temporary Exemption for Improvements made for economic development purposes (HBC 3.70.040)(C)(3).

On August 16, 2022, Highland's Estates, Inc. submitted an application per HBC 3.70.040(C)(3) for its Hilltop Subdivision project. The Haines Borough Planning Commission has approved the 100% Preliminary Plat for this project. The Assessor has timely forwarded the completed application with a recommendation for approval. Since the Assessor has recommended approval, the attached ordinance has been prepared.

The assembly must decide to grant the exemption and the length of the exemption (up to 10 years).

Referral:	
Referred to:	Referral Date:
Recommendation:	Meeting Date:

Assembly Action:

Meeting Date(s): 8/24/22

Public Hearing Date(s): 9/13/22 and 9/27/22 Postponed to Date: HAINES BOROUGH, ALASKA ORDINANCE No. 22-08-625



An Ordinance of the Haines Borough Assembly allowing a Temporary Exemption from Property Tax Levy for Highland's Estates, Inc. Hilltop Subdivision.

WHEREAS, Haines Borough Code 3.70.040(C)(3) provides for a temporary (up to 10 years) property tax exemption for improvements made for economic development purposes, provided the organization applies and meets the code standards and requirements, and receives assembly approval; and

WHEREAS, a complete application has been received from Highland's Estates, Inc. for its Hilltop Subdivision parcels, more particularly described as:

Tracts A-3-1 and A-3-2, Subdivision of Martin Cordes Property according to Plat 89-8, Haines Recording District, First Judicial District, State of Alaska, described as C-USS-A2-2716 and C-USS-A3-2716; and

WHEREAS, the borough assessor has reviewed and approved the application; and

WHEREAS, the proposed improvements will provide economic benefit to the borough; and

WHEREAS, the exemption will provide measureable public benefits commensurate with the level of incentive granted;

WHEREAS, the property owner is in compliance with all Alaska municipal and state of Alaska tax obligations; and

WHEREAS, the location of the business is compatible with land use and development plans of the borough; and

WHEREAS, the exemption is necessary to allow adequate time for improvements to be completed; and

WHEREAS, the proposed improvements will provide employment in the borough; and

WHEREAS, an exemption on the property enables a significant capital investment in physical infrastructure that will expand the tax base of the municipality and will generate property tax revenue after the exemption expires,

NOW, THEREFORE, BE IT ENACTED, by the Haines Borough Assembly, that the temporary exemption from property tax levy for the value of improvements made to Highland's Estates, Inc. Hilltop Subdivision is granted commencing January 1, 2023 and continuing year to year while Highland's Estates, Inc still owns the parcels, and terminates December 31, ______ (*10 years maximum*). The parcels will be fully taxable when the parcels are no longer eligible for tax exemption under HBC 3.70.040.

Section 1. <u>Classification</u>. This ordinance is a non-code ordinance.

Section 2. <u>Severability</u>. If any provision of this ordinance or any application thereof to any person or circumstance is held to be invalid, the

Haines Borough Resolution No. 22-08-625 Page 2 of 2

remainder of this ordinance and the application to other persons or circumstances shall not be affected thereby.

Section 3. <u>Effective Date</u>. This ordinance shall become effective on January 1, 2023.

Section 4. <u>Repealer</u>. This ordinance may be repealed by the voters through referendum.

Section 5. <u>Purpose</u>. To provide property tax incentive for improvements made for economic development purposes.

Adopted by a duly constituted quorum of the Haines Borough Assembly this 27th day of September, 2022.

Douglas Olerud, Mayor

ATTEST:

Alekka Fullerton, CMC, Borough Clerk

Date Introduced:8/24/22Date of First Public Hearing:9/13/22Date of Second Public Hearing:9/27/22

APPLICATION FOR LO Temporary Exemption	OCAL EXEMPTIONS & EXCLUS 1 for Improvements (HBC 3.70.040	IONS CC RIECEIWED
H H H H H H H H H H H H H H H H H H H	Haines Borough Office of the Assessor 103 Third Avenue S. P.O. Box 1209 Haines, AK 99827 907-766-6400	AUG 1 6 2022 HAINES BOROUGI CLERK'S OFFICE
This form is to be used to apply for a temporary exe qualifying improvements per HBC 3.70.040(C). The appl t h e exemption. The exemption will begin for the tax yo DEADLINE is January 15, of the applicable year.	emption and exclusion from an increase in icant has the burden of proof to show t ear following completion of the improvem	the assessed value of the property is eligible for ent project. POSTMARK
<u>Return applications to</u> : Haines Borough Administration B available at 907.766.6400.	Building, 103 Third Ave S., or FAX to 907.76	6.2716. Telephone assistance is
Applicant Contact Information: Name Highland's Estates, Inc Address PO BOX 1129 Hainer AK 998	Phone # <u>907-766-2821</u> 821[Email <u>roger & seroad.c</u>	<u>com</u>
Assessment/Tax Year _ 2022		
Parcel # C-USS-A2-2716 C-USS	-A3-2714	
Assessed Value prior to exterior enhancement:	<u>205, 600</u> Provide copy of contractor's hid or work or	ter)
Property Owner Name Highland's Estates	The	JCI.]
Property Description: (type of property, building size, lot si	ize, etc.)	
Property type: Residential 🔯 Commercial 🗆		
Building Size: N:/A		
Lot Size: 5 acres		
(equest:		
O HBC 3.70.040(C)1 – Maintenance, repair or renovat Was the principle purpose of the improvemen YesNoIf no explain:	tion to exterior of existing structure or nate at to increase space of occupancy or nonres	ural features of land idential use in the structure?
HBC 3.70.040(C)2 – Improvements made for the de three or more lots which adhere to the long plat re Have you submitted a platting action applicati If so, has the Planning Commission approved the	evelopment of a subdivision of a single para quirements. on to the Planner? YesNo e plat? YesNoIf no explain:	cel of property into
HBC 3.70.040(C) – Improvements made for econom Have you submitted a platting action applicati If so, has the Planning Commission approved the	nic development purposes. on to the Planper? Yes NoNo plat? Yes NoIf no explain:	

×.

Firefox

.

APPLICATION FOR LOCAL EXEMPTIONS & EXCLUSIONS Temporary Exemption for Improvements (HBC 3.70.040 C)

Required additional attachments (for economic development exemption):

Map or proof the entire property is within the Haines Borough boundary;

0

Documents describing how the investment will create employment in the Borough, expand the tax base of the Borough, and how it will generate property tax revenue after the exemption expires;



Proof of Planning Commission approval

I hereby certify that the information provided above is true and correct to the best of my knowledge. I acknowledge that the parcels will be taxable when the parcels are no longer eligible for tax exemption under HBC 3.70.040(C).

Date: August 15,2022

Jeanard

		Office L	Jse Only Below This Line			
Fees Paid \$ MA	Permit(s) _	MA	Date Paid	_ Receipt No	NIA	ck#_N/A cash N/A

All required attachments and fees received ---> (Must be checked before hearing can be conducted or permit issued.)
Application Approved Denied

Reason for Denial/Comments/Conditions:

Ilsen

Assessor

Clerk:

8-17-2022 Date signed

Date Referred to Clev

Date Received

Date Scheduled for Assembly

Planner: Received final plat

Date Received

Dated Scheduled for PC

Date of Final Approval by Planning Commission Approved Mylar Received Recorded

Starting Date of exemption: _____ Ending Date of exemption: _____

Haines Development Inc. PO Box 1129 Haines, AK 99827

RE: Hill Top Subdivision Contractor Bid

.

Please see the following bid associated with the construction and development of the Hill Top Subdivision Project per the plans given.

Engineering/Surveying/Permitting/Inspection Services:	\$ 250,000
Clearing and Grubbing:	\$ 300,000
Excavation: Rock and Common:	\$ 325,000
Roadway Embankment:	\$ 250,000
Subbase:	\$ 150,000
Base Course:	\$ 200,000
Water system:	\$ 300,000
Waste Water System:	\$ 350.000
Electrical:	\$_450.000

Total Project Cost

\$2, 575,000.00





Highland's Estates, Inc. PO Box 1129 Haines, AK 99827

RE: Hill Top Subdivision Property Tax Exemption Application

To whom it concerns,

Highland's Estates has received 100% preliminary plat approval from the planning commission on August 11, 2022 for the Hill Top Subdivision Project and is now submitting an application for a 10 year property tax exemption. At this time we have our approval letter dated July 15th.

Without the 10 year time frame for tax exemption, this project is not feasible due to the overall size and cost of construction as well as the increased property value and additional taxation.

Highland's Estates will be creating employment through the construction and until completion of the project with local hire, by Haines Development, being high priority.

The 65 individual lots that will be created and available upon completion of project will not only directly increase the property tax revenue but will also allow for population growth and expansion of the overall Haines Borough tax base.

Thank you for your consideration.



HAINES BOROUGH, ALASKA P.O. BOX 1209 HAINES, AK 99827 (907) 766-6400 FAX (907) 766-2716

July 15, 2022

Highland's Estates, Inc. PO Box 1129 Haines, Alaska 99827

Re: Hilltop Subdivision Long Plat 22-37 Preliminary Plat Approval

Thank you for attending the 7-14-2022 Planning Commission Meeting, as you are aware, the Planning Commission (PC) voted to approve the Hilltop Subdivision Preliminary Plat with the 6-9-2022 Planning Commission requirement list as clarified in Staff Memorandum dated 7-12-2022 and additional items required at this 7-14-2022 meeting. In review, here are the required items from the 6-9-2022 and 7-14-2022 lists:

1) Develop subdivision roads to the following categories according to HBC

Road nameDevelopment categoryHillside LoopCategory IBartlett BlvdCategory IMartin AvenueCategory IIHilltop WayCategory ICordes DriveCategory IAllie RoadCategory I

HBC 18.100.075 Streets. The design and construction of streets, roads, and sidewalks in subdivisions shall be governed by the provisions of HBC 12.08.030 through 12.08.190

2) Snow Storage areas - The PC accepted the Port Chilkoot Company MOU for snow storage.

 Port Chilkoot Company Permission – The PC accepted the Port Chilkoot Company MOU dated 6/10/2022 as permission.

4) Drainage code- The PC accepted the proHNS LLC Drainage analysis report dated 6-24-2022.

5) Access to Lots 17-22 - The PC accepted the 20' private driveway easements.

6) Lots 17-22 Utilities - The PC and HB Code require water and sewer service to every lot.

7) Topographic contour sheet required - The PC accepted Sheet 4 of 20 as topographic map.

8) Final Plat signatures – The PC approved updated plan sheet 2 of 2. Port Chilkoot Company should also have an owner signature block as they are providing 30° road dedications.

 QC/QA inspections – Haines Borough will hire an inspector for road construction and provide a schedule of inspections prior to final plat approval.

10) No Construction prior to final plat approval – Clarify to read No Road Construction until final plat approval.

And the additional items from the 7-14 2022 PC meeting:

1) Regarding drainage plan – Increase ditch depths from 2.5 feet to 3 feet.

2) Update Plat note #5 on Sheet 1 of 2, to read water and sewer utilities to serve every lot. Add that the Haines Borough will retain no responsibility for these lines.

3) Develop Bartlett Boulevard from Major Road to Rivers Street as a Category I road.

This preliminary plat approval now allows you to perform the following work items:

- a) Completing surveying and monumentation.
- b) Complying with plat conditions required by the commission as conditions of approval, including but not limited to physical improvements to the property such as land clearing, installation of drainage and identification of rights of way and easements.
- c) Preparing a reproducible mylar plat as approved by the commission.

In addition, Site Development permit 21-36 is still in effect with the allowance of clear, grub, grade, fill and minor pre-leveling.

No road construction is allowed under either permit until final plat approval.

Let me know when you ready to submit the final plat for approval, my next meeting agenda item deadlines are:

August 11, 2022 Meeting – August 1, 2022 September 8, 2022 Meeting – August 29, 2022

Thank you and please let me know if you have any questions,

ong ve

Haines Borough Planner dlong *u* haines.ak.us 907 766 6412

The same of a plane with the state with the state of the

3.70.040 Local exemptions and exclusions.

A. The following property is exempt from general taxation...

...

C. *Temporary Exemption for Improvements.* The increase in assessed value of improvements to real property shall be temporarily exempt from taxation if an increase in assessed value is directly attributable to either:

...

3. Improvements made for economic development purposes.

a. The exemption for economic development purposes must provide economic benefit to the borough; and

(1) The exemption will provide measurable public benefits commensurate with the level of incentive granted; and

(2) The property owner is in compliance with all Alaska municipal and state of Alaska tax obligations; and

(3) The location of the trade, industry or business is compatible with land use and development plans of the borough; and

(4) The exemption is necessary to allow adequate time for improvements to be completed.

b. For purposes of this section, "economic development property" means real or personal property, including developed property conveyed under 43 U.S.C. 1601 through 1629e, Alaska Native Claims Settlement Act, to which one or more of the following apply:

(1) Has not previously been taxed as real or personal property by the borough.

(2) Is used in a trade or business in a way that:

(a) Creates employment in the borough; or

(b) Generates sales outside of the borough of goods or services produced in the borough; or

(c) Materially reduces the importation of goods or services from outside the borough.

(3) An exemption on the property enables a significant capital investment in physical infrastructure that:

- (a) Expands the tax base of the municipality; and
- (b) Will generate property tax revenue after the exemption expires.

(4) Has not been used in the same trade or business in another municipality for at least six months before the application for deferral or exemption is filed. This subsection does not apply if the property was used in the same trade or business in an area that has been annexed to the municipality within six months before the application for deferral or exemption is filed. This subsection does not apply to inventories.

c. The exemption for improvements made for economic development purposes is limited to a maximum of 10 years.

d. *Application*. A completed borough-approved application must be submitted by the property owner prior to submission of a final plat. Applications submitted after construction begins shall be rejected. Applications for an exemption under this section must be addressed to the assessor. The assessor must forward a completed application to the clerk with a recommendation for approval or denial within 14 days of receipt. If the assessor recommends approval, the clerk shall prepare an ordinance for placement on the next agenda. In addition to any information required by the assessor, the application shall contain:

(1) A map or other proof that the entire property is within the Haines Borough boundary;

(2) Documents describing how the investment will create employment in the borough and expand the tax base of the borough; and how it will generate property tax revenue after the exemption expires;

(3) A review by the borough planning commission;

(4) An acknowledgement by the applicant that the parcels will be taxable when the parcels are no longer eligible for tax exemption under this chapter.

e. This exemption may be repealed by the voters through referendum.



7B

Assembly Agenda Bill

Agenda Bill No.: 22-1207

Assembly Meeting Date:09/13/2022

Business Item Description:	Attachments:
Subject: Clarifying Road Construction Inspection Costs	1. Ordinance 22-08-626 2. Planning Commission Recommendation
Originator:	
Administration	
Originating Department:	
Lands	
Date Submitted: 7/28/22	

Full Title/Motion:

No motion necessary since Ordinance 22-08-626 is already scheduled for its second public hearing 9/27/22.

Administrative Recommendation:

This resolution is recommended by the Planning Commission.

Fiscal Impact:			
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets
\$	\$ See below	\$ 0	Reduced maintenance costs

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Consistent: Yes

No

Summary Statement:

This Code change clarifies financial responsibility for road inspection costs.

Referral:

Referred to: Recommendation: Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 08/24/2022

Public Hearing Date(s): 9/13 and 9/27/22 Postponed to Date:

An Ordinance of the Haines Borough amending Haines Borough Code Title 12 to Clarify Inspection Costs

DRAFT

BE IT ENACTED BY THE HAINES BOROUGH ASSEMBLY:

Section 1. <u>Classification</u>. This ordinance is of a general and permanent nature and the adopted amendment shall become a part of the Haines Borough Code.

Section 2. <u>Severability</u>. If any provision of this ordinance or any application thereof to any person or circumstance is held to be invalid, the remainder of this ordinance and the application to other persons or circumstances shall not be affected thereby.

Section 3. <u>Effective Date</u>. This ordinance is effective upon adoption.

Section 4. <u>Amend Section 12.08.070.</u> Section 12.08.070 of the Haines Borough Code is hereby amended as follows:

NOTE: **Bolded**/<u>UNDERLINED</u> ITEMS ARE TO BE ADDED STRIKETHROUGH ITEMS ARE DELETED

12.08.070 Road construction standards – Introduction

A. *Enforcement.* The borough shall inspect road construction to ensure adherence to an approved plat and borough standards. <u>The developer shall bear all the costs of such</u> <u>road construction inspection.</u> Inspection may include test holes; engineering analysis of road geometry, drainage, and general adequacy for anticipated traffic. Construction not adhering to an approved design or construction standards shall be brought into compliance by the developer.

•••

Section 5. <u>Amend Section 12.08.130.</u> Section 12.08.130 of the Haines Borough Code is hereby amended as follows:

12.08.130 Road construction standards – Typical section materials.

Sub-base shall contain no muck, frozen materials, roots, sod or other deleterious matter. The Haines Borough will conduct an inspection prior to grading-C or D1 application of the sub-base on Category I and II roads. The Category III road's sub-base will adhere to specifications called out in the project specific design described in HMC <u>12.08.140(B)</u>. <u>The</u> <u>developer shall bear all costs for inspection of road construction.</u>

ADOPTED BY A DULY CONSTITUTED QUORUM OF THE HAINES BOROUGH ASSEMBLY THIS _____ DAY OF ______, 2022.

ATTEST:

Douglas Olerud, Mayor

Alekka Fullerton, CMC, Borough Clerk



- DATE: August 11, 2022
- TO: Borough Assembly
- FROM: <u>Haines Borough Planning Commission</u>

PLANNING COMMISSION ACTION:

<u>Motion</u> - J Mitman moved to recommend the Assembly adopt <u>Ordinance 22-07-624 - an Ordinance of the Haines Borough</u> <u>amending Haines Borough Code title 12 to Clarify Inspection</u> <u>Costs</u>

And the motion carried unanimously.

RATIONALE:

The recent Hill Top Subdivision long plat application and review has revealed that requirements for a developer to bear all costs of road construction inspection is lacking in current code. This will clarify developer expenses that will be required.

PLANNING COMMISSION REQUEST:

Amend H.B.C. 12.08.070 Road Construction Standards to include "The developer shall bear all the costs of such road construction inspection", and H.B.C 12.08.130 Road Construction Standards to include "The developer shall bear all costs for inspection of road construction" as written in Draft Ordinance 22-07-624

sham SUBMITTED BY (signature)

Diana Lapham Planning Commission Chair



7*C*

Assembly Agenda Bill

Agenda Bill No.: 22-1208

Assembly Meeting Date:09/13/2022

Business Item Description:	Attachments:
Subject: Classification of Foreclosed Property for Sale	 Ordinance 22-08-627 Planning Commission Recommendation Judgment and Clerk's Deed
Originator: Administration	4CIA Letter
Originating Department: Clerk's Office	
Date Submitted: 8/5/22	

Full Title/Motion:

No motion necessary since Ordinance 22-08-627 is already scheduled for its second hearing 9/27/22.

Administrative Recommendation:

This resolution is recommended by the Planning Commission.

Fiscal Impact:							
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets				
\$	\$ See below	\$ 0	Reduced maintenance costs				

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

s I	
	s L

es □No

Summary Statement:

This is a continuation of the foreclosure procedures. The subject property has now been confirmed to the Haines Borough. The Planning Commission has indicated to the Assembly that the property should be sold rather than being retained for municipal purposes. Now, through this ordinance, the Assembly may classify the property for sale.

The Manager has received a letter from CIA indicating that they are interested in an negotiated sale to acquire the property to renovate it and return it to the local housing inventory for low to moderate income families. The next step is for the planning commission to review CIA's proposal per HBC 14.20.100.

For now, the Assembly only needs to classify the property for sale.

Referral:

Referred to: Recommendation: Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 08/24/2022

Public Hearing Date(s): 9/13/22 and 9/27/22 Postponed to Date: HAINES BOROUGH, ALASKA ORDINANCE No. 22-08-627



An Ordinance of the Haines Borough determining whether recently foreclosed property deeded to the borough shall be retained for a public purpose or sold.

WHEREAS, delinquent property tax liens resulted in foreclosure of the following parcel within the Haines Borough (First Judicial District, State of Alaska):

Lot 13, Block A of the DASCHU ISTHMUS SUBDIVISION, located according to Plat No. 81-3, Haines Recording District, First Judicial District, Alaska.

WHEREAS, on May 17, 2021, the Court of the State of Alaska issued a judgment finalizing the foreclosure and conveying all rights, titles, and interest in the real property to the Haines Borough; and

WHEREAS, on August 5, 2022, the Clerk's Deed transferring ownership to the Haines Borough was received from the court; and

WHEREAS, the parcel is now Haines Borough property, and HBC 14.20.040 provides that borough land may be classified for sale by the assembly with the advice of the planning commission to discuss any such classification and designation before making any recommendations to the assembly; and

WHEREAS, following discussion of the parcel during a public meeting on August 11, 2022, the planning commission decided to recommend the parcel be sold; and

WHEREAS, HBC 3.74.220 requires the borough assembly determine by ordinance whether foreclosed property deeded to the borough shall be retained for a public purpose; and

WHEREAS, foreclosed property conveyed to the borough and not required for a public purpose may be sold provided the borough assembly, by ordinance, determines that a public need for the property does not exist,

NOW, **THEREFORE BE IT RESOLVED** the Haines Borough Assembly determines the aforementioned parcel is not required for a public purpose and may be sold according to HBC 14.20.

Section 1. <u>Classification</u>. This ordinance is a non-code ordinance.

Section 2. <u>Severability</u>. If any provision of this ordinance or any application thereof to any person or circumstance is held to be invalid, the remainder of this ordinance and the application to other persons or circumstances shall not be affected thereby.

Section 3. <u>Effective Date</u>. This ordinance shall become effective immediately upon adoption.

Section 4. <u>Purpose</u>. To determine whether foreclosed properties deeded to the borough shall be retained for a public purpose.

Adopted by a duly-constituted quorum of the Haines Borough Assembly on this 24th day of August, 2022.

Attest:

Douglas Olerud, Mayor

Alekka Fullerton, CMC, Borough Clerk



- DATE: August 11, 2022
- TO: Borough Assembly
- FROM: The Haines Planning Commission

PLANNING COMMISSION DECISION:

<u>R</u> Goldberg moved to classify Tax Parcel C-DAS-0A-1300 also known as 14 Deishu Drive, for sale and confirm the current use as residential, and the motion passed unanimously

The Planning Commission indicated a preference to return the property to Chilkoot Indian Association or other native ownership.

RATIONALE:

Foreclosed home Charlie Jimmie, Sr.

The Haines Borough will be receiving a Clerk's deed for ownership as a result of a judgement and decree of foreclosure dated 5/17/2021. The property has a one-story home (1056 SF), all utilities and is in need of repairs due to deferred maintenance items. It was determined there is no reason to retain this property under Borough ownership.

SUBMITTED BY (signature)

Diana Lapham Planning Commission Chair



HAINES RECORDING DISTRICT

IN THE SUPERIOR COURT FOR THE STATE OF ALASKA

FIRST JUDICIAL DISTRICT AT HAINES

)

)

)

In the Matter of 2015 through 2016 Delinquent Real Property Taxes Owed to the Haines Borough, Alaska

Case No. 1JU-20-00572 CI

THIS COVER SHEET HAS BEEN ADDED TO THE DOCUMENT TO PROVIDE SPACE FOR THE RECORDING DATA. THIS COVER SHEET APPEARS AS THE FIRST PAGE OF THE DOCUMENT IN THE OFFICIAL PUBLIC RECORD.

DO NOT DETACH

After Recording Return To:

Samuel C. Severin CHANDLER, FALCONER, MUNSON & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alaska 99501

	STATE OF ALL OF THE COMPANY
	Samuel C. Severin AK Bar No. 0606035 CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP 911 W. 8 th Avenue, Suite 302 Anchorage, AK 99501 (907) 272-8401
	Attorneys for Haines Borough
	IN THE SUPERIOR COURT FOR THE STATE OF ALASKA
	FIRST JUDICIAL DISTRICT AT JUNEAU
	In the Matter of 2015 through 2016) Delinquent Real Property Taxes) Owed to the Haines Borough, Alaska.) Case No. 1JU-20-00572 CI
	JUDGMENT AND DECREE OF FORECLOSURE OF REAL PROPERTY TAX LIENS
	IT IS HEREBY ORDERED AND ADJUDGED that the Haines Borough, Alaska, a
	municipal corporation, shall have judgment against certain parcels of real property for the
	amount of delinquent real property taxes, penalty, interest, and costs for the tax years 2015 and
8	2016. A description of the parcels of real property, together with respective delinquent taxes,
: 302 01.r 74-369	penalty, interest, and apportioned costs, is attached as Exhibit A to the Findings of Fact and
r, Suite ka 995 907) 2	Conclusions of Law, a copy of which is appended hereto and incorporated herein by reference.
Avenue e, Alas Fax (Costs of \$4,490.00 are apportioned to the properties accordingly.
V. 8th. chorag	It is further ADJUDGED that the said parcels of real property are conveyed to the
911 V Am	Haines Borough in consideration of the foregoing delinquent amounts, subject to the delinquent
5	taxpayers' statutory redemption rights.
	JUDGMENT AND DECREE OF FORECLOSURE OF REAL PROPERTY TAX LIENS ITMO 2015-2016 Real Property Taxes Owed Haines Borough, 1JU-20-00572 Cl Page 1 of 2
	Page 2 of 11 2021-000212-0

1. · · · ĺ ľ 7 day of May DATED this 2021. HON. PHILIP PALLENBERG CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alaska 99501 (907) 272-8401 Fax (907) 274-3698 I certify that this is a full, true and correct copy of an original document on file in the Alaska Trial Courts at Juneau. Witness my hand and CERTIFICATION Copies Distributed Date 5/3/2) To Samuel County the seal of this court: 5/13/21 | Jule W. Linne - ma Leetful copy Date Megistrate/Clerk. and and By Su and the second 2° 1 JUDGMENT AND DECREE OF FORECLOSURE OF REAL PROPERTY TAX LIENS ITMO 2015-2016 Real Property Taxes Owed Haines Borough, 1JU-20-00572 Cl Page 2 of 2 Page 3 of 11 2021-000212-0 Page 3 of 11 (\mathbf{A}) ,⊕-5 Ċ -X 14

	Samuel C. Severin Alaska Bar'No. 0606035 CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alaska 99501 (907) 272-8401 sseverin@bcfaklaw.com
	Attorneys for the Haines Borough
	IN THE SUPERIOR COURT FOR THE STATE OF ALASKA
	FIRST JUDICIAL DISTRICT AT JUNEAU
	In the Matter of 2015 through 2016) Delinquent Real Property Taxes) Owed to the Haines Borough, Alaska.) Case No. 1JU-20-00572 CI
	FINDINGS OF FACT AND CONCLUSIONS OF LAW
	The Haines Borough has filed its Petition for Judgment and Decree of Foreclosure of
LLP	Tax Liens for the years 2013 and 2014, and has moved for summary judgment in this matter.
OLA,	Based upon the Affidavit of the Borough Clerk, a review of the file, and applicable statutory
ACCI	authority, the Court makes the following findings:
k, & C 1, & C 10 274-36	1. The Haines Borough ("the Borough") is a municipal corporation duly organized
NSON MSON Me, Suit (907)	and existing under and by the laws of the State of Alaska, and is situated within the First
2, MU 1 Aven 85 Als	Judicial District, State of Alaska.
ONEI W. 8th 72-840	2. In both of the years 2015 and 2016, the Borough Assessor conducted an
7 FALC 911 911	assessment for all the real property within the corporate boundaries of the Borough known to
Que Ler,	be subject to taxation for school and municipal purposes. On the assessment roll, the Borough
IC	Clerk entered (1) the names and addresses of all persons owning property subject to taxation
24 X	FINDINGS OF FACT AND CONCLUSIONS OF LAW Page 1 of 7 ITMO 2015-2016 Delinquent Real Property Taxes Owed to the Haines Borough, 1JU-20-00572CI
Be	
la	
Ŋ	
	Page 4 of 11 🕢 🕟

and assessment, (2) a description of all taxable property, and (3) the assessed value of the said property.

3. Each year the Borough Clerk caused to be mailed by first-class mail to every person named in the assessment roll a notice showing the assessed value of the property, the time when the Borough Assembly acting as a Board of Equalization would sit for purposes of equalizing the assessment, and the dates when the taxes were payable, delinquent, and subject to interest.

4. The Board of Equalization then conducted hearings at which time any party appealing the Assessor's valuations of the property was given an opportunity to be heard. After considering all of the evidence presented to it, the Board rendered its decision.

5. Records were kept of the persons who appeared before the Board to protest the assessments. The Borough Clerk and the Borough Assessor made the corrections and additions as directed by the Board and completed and certified the annual assessment roll each year, based on the values as of January 1, both years 2015 and 2016.

6. Each year the Borough Assessor and Borough Clerk then delivered to the Borough Assembly a statement of the total assessed valuation of all real property within the Borough. The Borough Assembly fixed a tax levy for each taxable year, to begin January 1 and end December 31 of each tax year, at the mill rate identified below upon each dollar of taxable property within the Borough. The mill rates for each taxable year for each subdivision within the Borough that is relevant to this action were established as follows:

FINDINGS OF FACT AND CONCLUSIONS OF LAW Page 2 of 7 ITMO 2015-2016 Delinquent Real Property Taxes Owed to the Haines Borough, 1JU-20-00572CI

(►)

Page 5 of 11

PO1

∙€

Page 5 of 11 2021-000212-0

C

1

Ċ

X

	l (Č	1
		<u>MILL RATES</u> 2015 2016	
	Townsite Fire District #1(Outside the Townsite) Fire District #3 Dalton Trail RMSA Dalton Trail RMSA (No fire service) Dalton Trail and Eagle Vista RMSA Dalton Trail and Chilkat Lake RMSA Riverview RMSA Letnikof RMSA Borough	10.17 10.47 7.82 8.12 7.72 8.15 8.45 8.76 7.65 7.86 11.13 10.97 7.95 8.09 9.55 9.06 8.97 9.31 6.92 7.25	
CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alasta 99501 (907) 272-8401 Fax (907) 274-3698	The Borough Clerk prepared and mailed tax state property listed on the tax rolls of the Borough. 7. The Haines Borough Code of Ord that the taxpayer may pay the real property tax in December 1 of each year if the total tax is more to due, then the entire tax becomes delinquent and p paid when due, then the second half is payable of installment. The tax is delinquent after that date ten percent (10%) of the tax due is added to all of unpaid taxes from the due date until paid in full a 8. The Borough Clerk, in accordance up a roll of all property subject to foreclosure. The last known owner of the tax-delinquent properties together with the year or years for which the tax. FINDINGS OF FACT AND CONCLUSIONS OF LAW <i>ITMO 2015-2016 Delinquent Real Property Taxes Owed i</i>	ements to the persons listed as owners of linances ("HBC") in place at the time provident in two installments on by September 1 and by than \$200.00. If the first half is not paid who penalties and interest accrue. If the first half on or before the due date for the second and penalties and interest accrue. A penalty of the delinquent taxes. Interest accrues on the at the rate of twelve percent (12%). with AS 29.4.330 and HBC 3.74.070, make the Foreclosure List provides the name of the ess and the descriptions of each property, es were delinquent. The Foreclosure List als to the Hainer Borough, 1JU-20-00572C1 Page 3 of 7 to the Hainer Borough, 1JU-20-00572C1	des y en Tis of he e so
	ll	2021 - 000212 - 0	1
	Page 6 of 11		
		► S C X	
0 🖌 Inbox - dlambert@ 💄 1 Remi	nder(s) 💄 🔂 🛴 🤇	🚫 Caselle Connect® 🔄 106-2	021-000212-0 💮 55

states the amounts of delinquent tax for each year, the penalty and interest accruing thereon, and that costs of foreclosure would also be due in order to redeem the property. The Borough Clerk has certified the Foreclosure List (Exhibit B) as a true and correct record of the remaining delinquent taxes, interest, penalties, and costs owed to the Borough for both years 2015 and 2016.

9. The Borough Clerk caused to be published in *The Chilkat Valley News*, a newspaper of general circulation within the Borough, a notice under the hand of the Borough Clerk setting forth the Foreclosure List for real property for both years 2015 and 2016. The notice provided that the Foreclosure List was open for public inspection at the Clerk's office and that it would be presented to the Superior Court for judgment and decree of foreclosure. The notice was published on March 12 and March 19, 2020. Publication was then suspended, owing to the COVID-19 pandemic. Publication resumed in February of 2021 and was published February 4, February 11, February 18 and February 25, 2021.

 Since filing this suit, the five of the seven owners on the original Foreclosure List have paid all the delinquent amounts due. In total, two properties remain on the Foreclosure List.

The Borough has incurred \$4,490.00 in costs to prosecute this foreclosure.
 These costs were reasonably incurred and include the cost of publication, the filing fee, mailing costs, and estimated attorney fees of \$3500.00.

These costs of foreclosures are apportioned to each property on the Foreclosure
 List in proportion to the real property tax lien amount of the respective property as follows:

FINDINGS OF FACT AND CONCLUSIONS OF LAW Page 4 of 7 ITMO 2015-2016 Delinquent Real Property Taxes Owed to the Haines Borough, 1JU-20-00572C1

∢)(►

Page 7 of 11

PO1

Ð₹



CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alaska 99501 (907) 272-8401 Fax (907) 274-3698

Tax ID No.	Legal Description	Apportioned Cost			
B-EXN-12-0200	Lot 2A, Blk 2, Excursion Inlet N 2009-20	lorth, Plat \$1490.68			
C-DAS-0A-1300	Lot 13, Blk A, Daschu Isthmus	\$2301.13			
The Court ha	wing made the preceding Findings of	Fact, now makes the following:			
	CONCLUSIONS OF L	AW			
1. This	Court has jurisdiction over the subject	t matter of this cause and over the			
properties stated in t	he Foreclosure List.				
2. All p	roperty now appearing on the Foreclo	sure List for the Borough for both of			
the years 2015 and 2	016 has been assessed and taxed acco	ording to law.			
 Proper notice of these proceedings has been given. 					
The assessment roll has been prepared by the Borough in the proper and correct					
manner as according to law.					
5. The a	ssessments of the properties which ap	opear on the assessment rolls are			
correct and have bee	n made according to law.				
6. Notic	es of valuation and assessment of pro-	perty have been made as required by			
law.					
7. The I	Board of Equalization has duly and pro	operly equalized all taxes levied			
against the property	which appears in the Foreclosure List	of the Haines Borough for both of			
the years 2015 and 2	016.				
8. The I	8. The Borough Assembly fixed a tax levy for each taxable year, to begin January				
1 and end December 31 of each tax year, at the mill rate identified below upon each dollar of					
FINDINGS OF FACT AND CONCLUSIONS OF LAW Page 5 of 7 ITMO 2015-2016 Delinquent Real Property Taxes Owed to the Haines Borough, 1JU-20-00572C1					
	a de la composition de la composit	Page 8 of 11 2021-000212-0			
 P					

alle Conn

A

c.

+0

į

Ċ

h

ninder(c)

1 ٤

-

CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP

ahert@

·

.

÷-•

105-2021-000212-0

taxable property within the Borough. The mill rates for each taxable year for each subdivision

within the Borough that is relevant to this action were established as follows:

	MILL RATES		
	2013	2014	
Townsite	10.17	10.47	
Fire District #I (Outside the Townsite)	7.82	8.12	
Fire District #3	7.72	8.15	
Dalton Trail RMSA	8.45	8.76	
Dalton Trail RMSA (No fire service)	7.65	7.86	
Dalton Trail and Eagle Vista RMSA	11.13	10.97	
Dalton Trail and Chilkat Lake RMSA	7.95	8.09	
Riverview RMSA	9.55	9.06	
Letnikof RMSA	8.97	9.31	
Borough	6.92	7.25	

 The Borough Clerk has mailed tax statements to the persons listed as owner(s) of the property now on the Foreclosure List as required by law.

10. Publication and notice of these proceedings have been made in accordance with

the requirements of law.

 All taxes charged against property appearing on the Foreclosure List of the Haines Borough for both of the years 2015 and 2016 are delinquent and penalty, interest, and

costs have accrued upon such taxes as shown more fully on said Foreclosure List and on the

Affidavit of the Borough Clerk, Alekka Fullerton

That real property taxes chargeable to the property now upon the Foreclosure

List for both of the years 2015 and 2016 have been regularly and duly assessed and levied according to law.

FINDINGS OF FACT AND CONCLUSIONS OF LAW Page 6 of 7 ITMO 2015-2016 Delinquent Real Property Taxes Owed to the Haines Borough, 1JU-20-00572C1



х

3

C

Page 9 of 11

١<u>Ū</u>

Ð٠

∢)(►

CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alaska 99501 (907) 272-8401 Fax (907) 274-3698

. .

13. The Borough should have judgment for all delinquent taxes, penalty, interest, and costs appearing to be due on the several parcels of real property described on the Foreclosure List and the Affidavit of Alekka Fullerton. The now subsisting liens on said properties for general taxes and assessments shall be foreclosed and a decree of foreclosure shall be entered in accordance herewith.

DATED this 7K day of May 2021. HON. PHILIP PALLENBERG SUPERIOR COURT JUDGE

CERTIFICATION Copies Distributed 5/1/2

- WYC

Page 7 of 7

A MARKA

Sit By.

Page 10 of 11 2021-000212-0

4

C

Date.

CHANDLER, FALCONER, MUNSON, & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alaska 99501 (907) 272-8401 Fax (907) 274-3698

I certify that this is a full, true, and correct copy of an original document on file in the Alaska Trial Courts at

Witness my hand and the seal of this coult.

3

1.

strate/Clerk

ITMO 2015-2016 Delinquent Real Property Taxes Owed to the Haines Borough, 1JU-20-00572Cl

4

1

.

FINDINGS OF FACT AND CONCLUSIONS OF LAW

Page 10 of 11

1-4

PO1

Junéau.

ۍ

5



HAINES BOROUGH 2015-2016 REAL PROPERTY FORECLOSURE LIST

OWNERS NAME	TAX ID NO	LEGAL DESCRIPTION	2015 TAX	2016 TAX	*P&I 3/31/2021	**FORE CLOSURE COSTS	Mailing cost 4 cert @6.90	Tetal due
JIMMIE, SR., CHARLES	C-DAS-0A-1300	LOT 13, BLK A DASCHU ISTHMUS SUB	545.28	863.78	1,144.72	2,301.13	27.60	4582.51
O'DONNELL, TIMOTHY	B-EXN-12-0200	LOT 2A, BLK 2, EXCURSION INLET NORTH, PLAT 2009-20	375 25	493.00	685.72	1,490.68	27.60	3,072 25

* PENALTY & INTEREST Plus ten percent (10%) at twelve percent (12%) per annum from date due until date paid.

**COSTS including but not limited to court, publication, & attorney fees.

For more information call the Borough office at (907) 766-6401

This list is available for public inspection at the office of the Borough Clerk at 103 3ⁿⁿ Avenue, Haines AK. A petition for JUDGMENT and DECREE OF FORECLOSURE has been filed with the SUPERIOR COURT in Juneau, Alaska on March 2020.

Exhibit B - Page 1 of 1

 $\overline{\mathbf{O}}$

·Ď·

2

4

0

×

Page 11

ç,

=

V

Page 11 of 11 2021-000212-0
Samuel C. Severin AK Bar No. 0606035 CHANDLER, FALCONER, MUNSON & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, AK 99501 (907) 272-8401

Attorneys for Haines Borough

IN THE SUPERIOR COURT FOR THE STATE OF ALASKA

FIRST JUDICIAL DISTRICT AT JUNEAU

In the Matter of 2015 through 2016 Delinquent Real Property Taxes Owed to the Haines Borough, Alaska.

Case No. 1JU-20-00572 CI

CLERK'S DEED

Upon this Court's Judgment and Decree of Foreclosure of Real Property Tax Liens

entered in case Case No. 1JU-20-00572 CI,

IT IS HEREBY ORDERED that all rights, title, and interest of the former owner of

the below-listed property is hereby conveyed to the Haines Borough, Alaska, PO Box 1209,

Haines, Alaska, 99827:

Lot 13, Block A of the DASCHU ISTHMUS SUBDIVISION, located according to Plat No. 81-3, Haines Recording District, First Judicial District, Alaska.

DATED this 5th day of ______ 2022.

CLERK'S DEED Page 1 of 1

CERTIFICATION Copies Distributed Date_ 8/5/22 To Samuel Seven mail (ortified con



CHANDLER, FALCONER, MUNSON & CACCIOLA, LLP 911 W. 8th Avenue, Suite 302 Anchorage, Alaska 99501 (907) 272-8401 Fax (907) 274-3698

lged 71932 SWE

I certify that this is a full, true and correct copy of an original document on file in the Alaska Trial Courts at

Juneau. Witness my hand and the seal of this court:

Ele W. wours 8/5/99 Date Magistrate/Clerk



August 18, 2022

Ms. Annette Kreitzer Borough Manager Haines Borough PO Box 1209 Haines, Alaska 99827

Dear Ms. Kreizter,

The Chilkoot Indian Association (CIA) is interested in a negotiated sale of the Charles Jimmie home. The CIA proposes to rebuild the house and return it to the local housing inventory for low to moderate income families.

Sincerely,

arrit Brouillette 3

Harriet Brouillette Tribal Administrator

Haines Borough General Sales Tax Revenue by Fiscal Year



FY18 - FY22 (FY22 unaudited)

	FY18	% FY19		% FY20		%		FY21	%		FY22	%		
	Actual	Change		Actual	Change		Actual	Change		Actual	Change		Actual	Change
01 - Areawide General	\$ 581,666	8%	\$	645,613	11%	\$	590,832	-8%	\$	534,441	-10%	\$	670,741	26%
02 - Townsite Service Area	783,511	9%		875,047	12%		799,027	-9%		742,890	-7%		913,454	23%
20 - Medical Service Area	290,833	8%		322,807	11%		294,879	-9%		253,457	-14%		320,520	26%
23 - Economic Development	581,667	8%		645,613	11%		591,007	-8%		530,812	-10%		670,741	26%
50 - Capital Improv. Projects	872,498	8%		968,420	11%		886,510	-8%		796,218	-10%		1,006,111	26%
	\$ 3,110,175	8%	\$	3,457,500	11%	\$	3,162,255	-9%	\$	2,857,818	-10%	\$	3,581,568	25%

MINUTES APPROVED Port and Harbor Advisory Committee

Meeting Date: June 23, 2022

Date of Approval: Aug 25, 2022

1. <u>Call to Order</u>: A meeting of the Haines Borough Ports and Harbors Advisory Committee took place at 10:30 am, June 23, 2022, Diana Lapham as Chair.

2. Roll Call:

Members in Attendance: Lapham, Hughes, Upton, Turner, Gray, & Eckhardt **Members Not in Attendance**:

Others in Attendance: Shawn Bell/Harbormaster, Alekka Fullerton/Clerk, Annette Kreitzer/Manager, Douglas Olerud/Mayor

- 3. <u>Approval of Agenda:</u> Hughes moved, Gray seconded, none opposed.
- Approval of Minutes: Hughes moved, Upton seconded, to approve May 26th meeting minutes. None opposed.

5. <u>Public Comment:</u>

6. Harbormaster's Report:

- A. PC Dock Gangway Lift Bulk of work compete, not operational yet
- B. Harbor Gangway Float Replacement Starting to receive submittals
- **C.** Lutak Dock Rebuild Project Update R&M responded for Owners Advisor, RFP for project contractor advertised and closes 30June.
- D. Aging Report

7. Unfinished Business: None

8. New Business:

A. Harbor Fee Schedule and Port Tariff Review: Motion by Hughes, second by Lapham: Increase the annual moorage rate by \$1 each year for the next 3 years. Vote: Hughes in favor; Turner, Gray, Eckhardt, Upton, and Lapham against. Motion failed

Motion by Upton, second by Eckhardt: Recommend to the Assembly to increase annual moorage rate by 3% each year and increase the transient moorage rate by 5% each year for the next 3 years.

Vote: Turner, Gray, Eckhardt, Upton, and Lapham in favor; Hughes against. Motion passed

B. Letnikof South Parking Lot 65% Design: 65% design approved with recommendations by the Planning Commission. Motion for approval by PHAC?

9. <u>Public Comments:</u>

10. <u>Committee Comments:</u>

11. <u>Set Meeting Dates:</u>

A. Next PHAC Meeting July 28, 2022 at 10:30 a.m.

12. <u>Adjournment:</u>



11A1

Assembly Agenda Bill

Agenda Bill No.: 22-1209

Assembly Meeting Date: 09/13/2022

Business Item Description:	Attachments:
Subject: Authorize the contract with Dawson Construct	1. Resolution 22-09-994
for the Haines Harbor E Float Repair (\$59,000.00)	2. Bid documents from Dawson Construction and Duwamash
Originator:	
Contracts and Grants Administrator	
Originating Department:	
Public Facilities	
Date Submitted: 8/25/2022	

Full Title/Motion:

Motion: Adopt Resolution 22-09-994.

Administrative Recommendation:

This resolution is recommended by the Director of Public Facilities and the WWTP Superintendent

Fiscal Impact:									
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets						
\$ 59,000	\$ See below	\$ 0	Reduced maintenance costs						

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Objective 2B, Pages 56-57

Consistent: Yes

□No

Summary Statement:

In December 2020 the Haines Borough experienced a major winter weather event that resulted in flooding, debris flows, and landslides throughout the borough. The Haines Harbor E Float sustained damage and needs repairs to return it to tis pre-disaster design, function and capacity related to the disaster damage. The Haines Borough posted the Haines Boat Harbor E Float Repair Request for Proposals on July 6, 2022 and received two responsive bids, one from Duwamash Services for \$262,900.00 and Dawson Construction for \$59,000. The Haines Borough has an active grant DR 4585 AK, with the Federal Emergency Management Agency (FEMA) to reimburse the costs for repairs to Borough infrastructure and roads affected by the storm and the funds are sufficient to cover the costs of repair. Recommend approval of contract with Dawson Construction for an amount not to exceed \$59,000 for the repairs of the Haines Harbor E Float.

Referral:

Referred to: Recommendation: Referral Date:

Μ

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/22

Public Hearing Date(s): Postponed to Date:

HAINES BOROUGH, ALASKA RESOLUTION No. 22-09-994



A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a construction contract with Dawson Construction, Inc for the Haines Boat Harbor E Float Repair for an amount not to exceed \$59,000.00.

WHEREAS, In December 2020 the Haines Borough experienced a major winter weather event that resulted in flooding, debris flows, and landslides throughout the borough; and

WHEREAS, the Haines Harbor E Float sustained damage and needs repairs to return it to its pre-disaster design, function and capacity related to the disaster damage; and

WHEREAS, the Haines Borough posted the Haines Boat Harbor E Float Repair Request for Proposals on July 6, 2022 and received two responsive bids, one from Duwamash Services for \$262,900.00 and Dawson Construction for \$59,000; and

WHEREAS, the Haines Borough has an active grant DR 4585 AK, with the Federal Emergency Management Agency (FEMA) to reimburse the costs for repairs to Borough infrastructure and roads affected by the storm and the funds are sufficient to cover the costs of repair; and

NOW, THEREFORE, BE IT RESOLVED that the Haines Borough Assembly authorizes the Borough Manager to execute a construction contract with Dawson Construction for the Haines Boat Harbor E Float Repair in an amount not to exceed \$59,000.00.

Adopted by a duly-constituted quorum of the Haines Borough Assembly on this 13th day of September 2022.

Attest:

Douglas Olerud, Mayor

Alekka Fullerton, CMC, Borough Clerk

Haines Borough, AK

Solicitation Number

HB 22-14

Deadline

07/06/2022 03:00 PM AKDT

Advertised

06/14/2022 03:30 PM AKDT

Revised

06/27/2022 04:22 PM AKDT

Description

Haines Boat Harbor E Float Repair (Rebid) December 2020 Storm Event 4585DR-AK PW 00032 (#185993)

Bidders

Business Name	Address	Phone
Dawson Construction	PO Box 30920 Bellingham, WA 98228	(360) 756-1000
Duwamish Services	6627 Ellis Ave S Seattle, WA 98108	(206) 930-0459
2 Bidders		

Item	Quantity	Dawson Co Price	onstruction Extension	Quantity	Duwam Price	ish Services Extension
SECTION 00310			Extension		1100	Extension
SECTION 00310		DULE				
1.1 Mobilization/ Demobilization Lump Sum	1.00	\$8,500.00	\$8,500.00	1.00	\$181,500.00	\$181,500.00
2.1 Submittals including BMP Polution Control Plan, Lump Sum	1.00	\$7,500.00	\$7,500.00	1.00	\$1,650.00	\$1,650.00
3.1 Float Demoltion and Waste Disposal including Di Lump Sum	1.00	\$21,500.00	\$21,500.00	1.00	\$24,200.00	\$24,200.00
4.1 Furnish and Install Treated Timber including Ma Lump Sum	1.00	\$13,000.00	\$13,000.00	1.00	\$36,300.00	\$36,300.00
5.1 Furnish and Install Billets including Labor and Each	2.00	\$2,500.00	\$5,000.00	2.00	\$8,800.00	\$17,600.00
6.1 Final Completion and Close-Out including As- Bui Lump Sum	1.00	\$3,500.00	\$3,500.00	1.00	\$1,650.00	\$1,650.00
6 Items	Totals:		\$59,000.00			\$262,900.00



11A2

Assembly Agenda Bill

Agenda Bill No.: 22-1210

Assembly Meeting Date: 9/13/2022

Business Item Description:	Attachments:
Subject: Authorize to Amend Application CW 395261	1. Resolution 22-09-995
Clean Water Fund Loan Wastewater Influent Lift Pump	2. Loan Agreement CW 395261-S 3. Alaska Clean Water Fund- SEX 2022 Project Priority
Originator: Contracts and Grants Administrator	List 4. Cleanwater IUP list.
Originating Department: Public Facilities	
Date Submitted: 8/23/2022	

Full Title/Motion:

Motion: Adopt Resolution 22-09-995.

Administrative Recommendation:

This resolution is recommended by the Director of Public Facilities.

Fiscal Impact:			
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets
\$ 80,000	\$ See below	\$ 0	Reduced maintenance costs

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Objective 2B, Pages 56-57

Consistent: Yes

□No

Summary Statement:

The Alaska Department of Environmental Conservation lists structurally impaired lift station wet wells as a public health threat. In January 2020 the Assembly approved a loan through the Alaska DEC State Revolving Fund to replace the lift stations at Beach Road and Skyline Road. In the course of the work on Beach Road, they found the existing sewer main was not where our original mapping had believed. Instead the sewer main is over 20 feet below the current system and out of alignment requiring the need for an additional approximately 500 LF of new main to bring it back into alignment. Due to the change in the Scope of Work, the Haines Borough applied to DEC for an amendment to the loan CW 395261-S for an additional \$80K to correct the issue. This project qualifies for a \$289,934 Disadvantaged Community Assistance. Recommend approve the amendment to CW 395261-S.

Referral:

Referred to: Recommendation: Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/2022

Public Hearing Date(s): Postponed to Date:

HAINES BOROUGH, ALASKA RESOLUTION No. 22-09-995

Draft A Resolution of the Haines Borough Assembly authorizing the

Borough Manager to amend loan CW 395261-S with the Alaska Department of Environmental Conservation for Phase II of the Wastewater Influent and Lift Station Pump Upgrade to include Beach Road Lift Station Sewer Main Extension for an amended amount not to exceed \$659,867.

WHEREAS, the Wastewater Influent and Lift Station Pump Upgrade project includes design and construction of improvements for wastewater lift stations at Beach Road and Skyline Drive and these pumps have repeatedly been unable to keep up with flows during heavy rain events; and

WHEREAS, in 2020, the Haines Borough received a loan for up to \$579,867 to replace the pumps at Beach Road and Skyline Lift Stations with a \$289,934 subsidy in the form of principal forgiveness; and

WHEREAS, in the course of the work on Beach Road, they found the existing sewer main was not where our original mapping indicated; the sewer main is over 20 feet below the current system and out of alignment requiring the need for an additional approximately 500 LF of new main to bring it back into alignment; and

WHEREAS, the Haines Borough applied to Alaska DEC SRF for an amendment to the current loan for an additional \$80,000 to correct the issue; and

WHEREAS the project was approved by Alaska DEC in the SRF Program SFY 23 Intended Use Plan to amend loan CW 395261-S for a total loan of up to \$659,867 that would be repaid over no more than a 20-year term, with an estimated 1.5 percent finance rate; and

WHEREAS, the loan continues to carry a \$289,934 subsidy in the form of principal forgiveness for Disadvantaged Community Assistance; and

NOW, **THEREFORE**, **BE IT RESOLVED** that the Haines Borough Assembly authorizes the Borough Manager to apply for an amendment to CW 395261-S from the State of Alaska, Department of Environmental Conservation for a loan from the Alaska Clean Water Fund for the project entitled Wastewater Influent and Lift Station Pump Upgrade in an amended amount of \$659,867.

Adopted by a duly-constituted quorum of the Haines Borough Assembly on this 13th day of September 2022.

Attest:

Douglas Olerud, Mayor

Alekka Fullerton, CMC, Borough Clerk

Department of Environmental Conservation





DIVISION OF WATER State Revolving Fund Program

> P.O. Box 111800 Juneau, Alaska 99811-1800 Main: 907.465.6594 Fax: 907.465.5177

April 15, 2020

Ms. Debra Schnabel Borough Manager Haines Borough P.O. Box 1209 Haines, Alaska 99827

Re: Loan Agreement No. 395261 for Wastewater Influent and Lift Station Upgrade (ACWF No. 395261)

Dear Ms. Schnabel:

Enclosed for signature is the loan agreement in the amount of \$579,867 for the Wastewater Influent and Lift Station Upgrade (Alaska Clean Water Fund No. 395261). The finance charge rate for this loan is 1.5 percent.

Please return a copy of the fully signed loan agreement to DECDASGRANTLOANADMIN@alaska.gov or mail a hard copy to the address identified below. The signed original agreement should be retained for your records.

Alaska Department of Environmental Conservation Division of Administrative Services Attn: Grant Administrative Section 555 Cordova Street, 4th Floor Anchorage, Alaska 99501

This loan is not effective, and no disbursements will be made, until the Department has received a copy of the fully signed agreement. If you have any questions regarding the loan agreement you may contact Carrie Bohan, Program Manager, at 465-5143 or Beth Verrelli, Project Engineer, at 269-7603.

Sincerely

Randy Bates Director

Enclosure: ACWF No. 395261 Wastewater Influent and Lift Station Upgrade Loan Agreement

Clean Water State Revolving Fund Loan Agreement No. 395261

Between

State of Alaska Department of Environmental Conservation Division of Water State Revolving Fund Program

And

Haines Borough

Wastewater Influent and Lift Station Upgrade						
Loan Amount	\$579,867					
Subsidy Amount	\$289,934					
Repayment Amount	\$289,933					
Finance Rate	1.5%					
Term	20 years					

Table of Contents

Article	. Loan Terms	. 3
1.01	Borrower Information	. 3
1.02	Project Description	.3
1.03	Loan Amount	.3
1.04	Principal Forgiveness	.3
1.05	Loan Term	. 3
1.06	Finance Charge	. 3
Article	2. General Terms and Conditions	.4
2.01	Accounting Practices	. 4
2.02	Timely Use of Funds	. 4
2.03	Disbursement of Funds	. 4
2.04	Principal Forgiveness	. 5
2.05	Loan Repayment	. 5
2.06	Late Payment Fee	. 6
2.07	Loan Default	.7
2.08	Notification	.7
2.09	Insurance	. 7
2.10	Environmental Review	. 8
2.11	Archaeological and Historical Preservation	.9
2.12	Cost and Effectiveness Analysis	. 9
2.13	Fiscal Sustainability Plan	. 9
2.14	Site Access	10
2.15	Construction	10
2.16	Compliance with Laws, Regulations, Etc.	11
2.17	Project Completion	14
2.18	Amendments and Modifications	15
2.19	Disputes	15
2.20	Termination	16
2.21	Indemnification	16
Article 3	. Definitions	17
Article 4	. Signatures	18
Exhibit	"A" – List of Federal Laws and Authorities	20

Article 1. Loan Terms

This loan agreement ("Agreement") is made and entered into as of the date of final signature by the Alaska Department of Environmental Conservation ("Department") and the Haines Borough ("Borrower") as identified in Article 5, and continues in full force and effect until the final day of the Agreement Period. This Agreement is made subject to, and conditional upon, the availability of funds.

1.01 Borrower Information

Borrower Name:	Haines Borough
Borrower Mailing Address:	P.O. Box 1209, Haines, Alaska, 99827
Name of Authorized Signatory:	Debra Schnabel
Resolution Number:	20-01-844

1.02 Project Description

The Borrower shall use this loan for treatment plant influent and 1-mile Lift Station Upgrades, and design and construction of improvements for wastewater lift stations at Beach Road and Skyline Drive.

1.03 Loan Amount

\$579,867

1.04 Principal Forgiveness

\$289,934

1.05 Loan Term

20 years

1.06 Finance Charge

Interest Rate: 1.0% Fee: 0.5% Total Finance Charge: 1.5%

Article 2. General Terms and Conditions

The Borrower shall comply with all applicable federal, state and local laws, requirements, and ordinances for the planning, design, construction, implementation, and administration of the Project and this Agreement, including but not limited to those identified in the General Terms and Conditions and Attachments.

2.01 Accounting Practices

The Borrower shall separately account for all monies received from the Alaska Clean Water Fund and shall maintain project accounts in accordance with generally accepted governmental accounting principles. The Department shall have the right to audit Borrower's records related to the Project.

2.02 Timely Use of Funds

- a. Concurrent with the execution and delivery of this Agreement, or as soon thereafter as practicable, the Borrower shall take all steps necessary to complete the Project in a timely manner in accordance with all applicable loan conditions.
- b. The Department will, in its discretion, revoke this Agreement if the Borrower has not initiated the Project within one year after signing the Agreement.
- c. If no Disbursement Request is made within the two year period, the Department may take action to recall the loan.

2.03 Disbursement of Funds

Subject to the terms and conditions of this Agreement, the eligible project costs less other funding sources will be disbursed by the Department upon receipt and approval of Disbursement Requests and Progress Status Updates.

The Borrower shall submit Disbursement Requests and Progress Status Updates to the Department via the Division of Water's Online Application System (OASys). OASys may be accessed at the following link: https://dec.alaska.gov/water/oasys.aspx.

- a. Disbursement Requests including Progress Status Updates must be submitted to the Department on a quarterly basis, within 30 days following the end of each quarter.
- b. Should the Borrower fail to submit the quarterly Disbursement Requests and Progress Status Updates as required, the Department will not process subsequent Disbursement Requests until all outstanding quarterly report(s) are received.
- c. Departmental approvals required by this Agreement will not be unreasonably withheld.
- d. The Department will disburse funds only as necessary to complete the Project. Any funds remaining after completion of the Project will remain in the Alaska Clean Water Fund.

- e. Borrower shall provide the Department with written evidence of materials and labor furnished to and performed upon the Project and such receipts of the payment of the same, releases, satisfactions and other signed statements and forms as the Department may reasonably require.
- f. The Department may at any time review and audit requests for disbursement and make adjustments for, among other things, ineligible expenditures, mathematical errors, items not built or bought, unacceptable work and other discrepancies.

2.04 Principal Forgiveness

As part of this Agreement, the Department has offered the Borrower \$289,934 of subsidy, in the form of principal forgiveness.

- a. Subsidy will be applied to each disbursement at 50% until all available subsidy has been applied.
- b. All subsidy shall be spent within one year of the date of this Agreement. Any subsidy not used within one year of the date of this Agreement may be withdrawn by the Department.

2.05 Loan Repayment

- a. The Borrower shall repay the principal amount and the finance charges on all cash disbursements made to the Borrower according to the repayment schedule. The repayment schedule will be prepared by the Department and confirmed by the Borrower, and will provide that:
 - i. The Borrower shall pay a finance charge of 1.0% on each disbursement. The finance charge is comprised of a fee of 0.5% on the total loan disbursed, plus the interest rate of 1.5%. Accrual of the finance charge will begin one year after the date of the first disbursement to the Borrower.
 - ii. The Borrower shall pay back the loan amount within 20 years from initiation of repayment. Repayment of the loan will be made with either equal annual principal payments plus the finance charge or equal annual total payments including the finance charge. Other repayment methods may be negotiated with the Department. Payments shall be applied first to any costs or charges incurred by the Department, outstanding interest, and, finally, to principal.
 - iii. The first repayment is due one year following substantial completion and Initiation of Operation of the Project.
- b. The Borrower hereby grants to the Department a security interest in and irrevocably pledges its Net Operating Revenues of the Borrower's water system to secure payment of and to pay the amounts due under this Loan Agreement. The Net Operating Revenues so pledged and hereafter received by the Borrower shall immediately be subject to the lien of such pledge without physical delivery or further act, and the lien of

the pledge shall be superior to all other claims and liens whatsoever, to the fullest extent permitted by law. The Borrower represents and warrants that the pledge of its Net Operating Revenues hereby made by the Borrower complies with, and shall be valid and binding from the date of this Agreement. The Borrower covenants with the Department and any assignee of this Agreement that, except as otherwise expressly provided herein, the Borrower shall not issue any other obligations which have a pledge or lien on its Net Operating Revenues superior to or on a parity with the pledge herein granted without the written permission of the Department. This Loan is a parity obligation with all other State Revolving Fund (SRF) loans between the Department and the Borrower.

- c. The Borrower represents and warrants to the Department that the Borrower has not pledged revenues for the repayment of this Loan that have been previously pledged or encumbered, unless specifically set forth in the Borrower's Approved Application. The Net Operating Revenues pledged in this Loan Agreement for repayment of this Loan and each separate source of revenue are specifically identified and described in the Borrower's Approved Application.
- d. If the Borrower's Net Operating Revenues are insufficient to meet any loan payment to the Department when due, the Borrower shall pay the deficiency in its loan payment from any legally available funds of the Borrower. Repayment of the Loan shall not be a direct and general obligation of the Borrower.
- e. If the Project Facility is damaged or destroyed prior to completion of the Agreement Period, the Borrower is liable to the Department for all amounts due under this Agreement.

2.06 Late Payment Fee

The Borrower shall be subject to a late charge for any repayment that is delinquent by more than 30 days, in accordance with the following conditions.

- a. If the Borrower is in good standing with the Department and has no late payments on any loans within the last five years:
 - i. And a payment is more than two months late a 1% charge will be applied against the outstanding amount due;
 - ii. And a payment is more than three months late a 3% charge will be applied against the outstanding amount due;
 - iii. And a payment is more than four months late a 5% charge will be applied against the outstanding amount due.
- b. If the Borrower has had late loan payments in the last five years.
 - i. And a payment is more than one month late a 1% charge will be applied against the outstanding amount due;

- ii. And a payment is more than two months late a 3% charge will be applied against the outstanding amount due;
- iii. And a payment is more than three months late a 5% charge will be applied against the outstanding amount due.
- c. Additionally, interest on the unpaid balance will continue to accrue at the interest rate established in Section 1.05 and must be paid in addition to the late charge. Payments in arrears when the 5% late charge is assessed will be referred to the Alaska Department of Law for collection.

2.07 Loan Default

The Borrower shall be in default, if a loan repayment has not been made within 90 days of the due date, as determined by the repayment schedule prepared by the Department and provided to the Borrower.

The provisions of AS 37.15.575 relating to state aid interception apply to the loan made under this Agreement.

2.08 Notification

Any disbursement or repayment made by the Department or Borrower under this Agreement shall be delivered by electronic transfer or by registered or certified mail.

- a. In order to submit repayments electronically, the Borrower must submit a request to the following email: <u>dec.adec.userfees@alaska.gov</u> to initiate the process and complete required forms.
- b. Any repayment addressed to the Department will be sent to:

Alaska Department of Environmental Conservation Division of Administrative Services Financial Services PO Box 11800 Juneau, AK 99811-1800

c. Any disbursement addressed to the Borrower will be sent to:

Debra Schnabel Borough Manager P.O. Box 1209 Haines, AK 99827

2.09 Insurance

a. If applicable, until the Project is completed by the Borrower, the Borrower (or at the option of the Borrower, the contractor) shall maintain insurance for the loss of the

Project Facility for the benefit of the Department, the Borrower and the prime contractor, and all subcontractors, as their interests in the Project may appear. The Borrower shall insure the Facility against loss or damage in an amount at least equal to the Allowable Project Cost specified in Article 4(e).

- b. If applicable, an insurance policy issued pursuant to Section 2.17 must be written or endorsed to make losses payable to the Department and the Borrower as their interests may appear. The interests of the Department are limited to the unpaid principal balance of the loan and any finance charge and penalties accrued as of the date such loan may be paid in full as a result of any insurance payoff, following destruction or damage to the Project Facility.
- c. In the event the Borrower fails to maintain the full insurance coverage required by this Agreement, the Department may take out the required policies of insurance and pay the premiums. All amounts so advanced by the Department will become an additional obligation of the Borrower to the Department.
- d. The Borrower shall require its contractors and subcontractors to maintain workers compensation, commercial general liability, property damage, and vehicle liability insurance. Until the Project is complete, the Borrower (or at the option of the Borrower, the contractor) shall maintain insurance for the loss of the facility for the benefit of the Department, the Borrower, the prime contractor, and all subcontractors, as their interests in the Project may appear.

2.10 Environmental Review

Prior to initiating the Project, the Borrower shall consult with the Department to determine the required level of environmental review. The Department will notify the Borrower of the type of environmental documentation that will be required, if any.

The Borrower shall complete an environmental review in accordance with the State Environmental Review Process (SERP), and in compliance with state and federal environmental laws prior to any ground disturbing or construction activities conducted as part of this Project. Disbursement Requests for costs related to construction activities will not be accepted until the SERP review has been completed. Any ground disturbing or construction activities that occur prior to the notification to the Borrower, by the Department, that Department's decision has been finalized are ineligible for reimbursement.

Any mitigation measures identified through the environmental review shall be fully implemented by the Borrower.

An environmental determination is valid for five years. Any activities occurring more than five years following the original environmental determination must undergo an additional review.

2.11 Archaeological and Historical Preservation

If historical or cultural artifacts are discovered during the Project, the Borrower shall immediately stop construction and implement reasonable measures to protect the discovery site from further disturbance; take reasonable steps to ensure confidentiality of the discovery site, restrict access to the site; and notify the concerned tribe's cultural staff or committee, the Department, and the State of Alaska's Historical Preservation Officer. If human remains are uncovered, the Borrower shall immediately report the presence and location of the remains to law enforcement, the concerned tribe's cultural staff or committee, and the Department.

2.12 Cost and Effectiveness Analysis

Under the Federal Water Pollution Control Act section 602(b)(13), the Borrower shall certify that they have conducted studies and evaluations for determining the cost and effectiveness of the Project. The cost and effectiveness analysis at minimum requires:

- a. the study and evaluation of the cost and effectiveness of the processes, materials, techniques, and technologies for carrying out the proposed Project or activity for which assistance is sought under this title; and
- b. the selection, to the maximum extent practicable, of a Project or activity that maximizes the potential for efficient water use, reuse, recapture, and conservation, and energy conservation, taking into account: the cost of constructing the Project or activity; the cost of operating and maintaining the Project or activity over the life of the Project or activity; and, the cost of replacing the Project or activity.
- c. Certification shall be provided to the Department by the Borrower before proceeding with final design or construction. The Borrower must use the certification form supplied by the Department to ensure compliance with this requirement.

2.13 Fiscal Sustainability Plan

- a. The Borrower shall, under amendments to the Federal Water Pollution Control Act, under Section 603(d)(1)(E), have a Fiscal Sustainability Plan (FSP) that covers the funded Project and closely associated components in place by the time of submission of the final disbursement request.
- b. A FSP is a living document that is regularly reviewed, revised, expanded, and implemented as an integral part of the operation and management of the system. The plan, at a minimum, shall include the following:
 - i. An inventory of critical assets that are a part of the treatment works;
 - ii. An evaluation of the condition and performance of inventoried assets or asset groupings;

- iii. A certification that the Borrower has evaluated and will be implementing water and energy conservation efforts as part of the plan; and
- iv. A plan for maintaining, repairing, and, as necessary, replacing the treatment works and a plan for funding such activities.
- c. The Borrower shall certify, on a Department supplied form, that they have developed and are implementing a FSP. Department specific FSP development criteria may either be obtained through the contact information given under Article 2.09 of this Agreement, or through the Department's web site at the following web address: <u>http://dec.alaska.gov/water/technical-assistance-and-financing/state-revolvingfund/guidance-and-forms</u>

2.14 Site Access

The Department has the right, at all reasonable times, to enter the Project Site, for the purpose of inspecting the Project and Project Facility.

2.15 Construction

- a. With the exception of land easements, all real estate and personal property constituting the Project Site and the Project must belong to the Borrower.
- b. The Borrower shall not begin construction until the Project has received an Approval to Construct (ATC), if one is required. If an ATC is not required by the Department's Engineering Support and Plan Review (ESPR) Program, the Borrower shall provide a statement from ESPR to that effect.
 - i. In its approvals, the Department may specify changes or conditions to the plans and specifications.
 - ii. The Department must approve any subsequent changes to, or deviations from, approved plans.
- c. If an ATC as described in (b) above is not required, the Borrower shall not begin construction until the plans and specifications have been reviewed and approved by the assigned State Revolving Fund Program engineer.
- d. Any construction contract estimated to equal or exceed \$100,000 shall be awarded through a competitive bidding process and any construction contract estimated to be less than \$100,000 may be negotiated if the Department approves the solicitation and negotiation procedures.
- e. All construction contracts and contractors' estimate forms shall be prepared so that materials and equipment may be readily itemized as to eligible project costs and non-eligible costs.

- f. Any change in a construction contract that will alter the contract specifications, time, price, or will substantially modify the proposed treatment process shall be submitted to the Department for approval.
- g. When applicable, the Borrower shall require each construction contractor to furnish a performance and payment bond in an amount at least equal to 100 percent of the contract price.
- h. Construction of the Project shall conform to applicable federal, state, and local laws, ordinances, and regulations.
- i. The Borrower shall proceed expeditiously and complete the Project in accordance with the Approved Application, project schedule, surveys, plans, profiles, cross-sections, specifications, and amendments.

2.16 Compliance with Laws, Regulations, Etc.

The Borrower shall comply with, and require its contractors and subcontractors to comply with, all applicable federal and state laws, rules, guidelines, regulations, and requirements to include, but not limited to, the following:

- a. The "List of Federal Laws and Authorities (Federal 'Cross-Cutting' Authorities)" as identified in Exhibit "A" and made a part hereof.
- b. Lobbying

No portion of the loan amount may be used for lobbying or propaganda purposes as prohibited by 18 U.S.C. Section 1913 or Section 607(a) of Public Law 96-74.

c. Davis-Bacon Act

For construction, alteration, and repair of treatment works, the Borrower shall ensure that contract wages paid are the higher of the State or Federal wage rate on a classification by classification basis for the construction of the Project. Both prevailing wage rates established for the locality by the Alaska Department of Labor under AS 36.05.010, and Federal standards in accordance with 40 U.S.C. Subtitle II Part A Subchapter IV (commonly referred to as the "Davis Bacon Act") apply. Laborers and mechanics employed by contractors and subcontractors shall be paid wages not less often than once a week and at rates not less than those prevailing on projects of a character similar in the locality as determined by the Secretary of Labor.

The Borrower shall obtain the wage determination for the area in which the project is located prior to issuing requests for bids, proposals, quotes or other methods for soliciting contracts (solicitation) less than 10 days before posting. Wages are locked-in at bid opening if awarded within 90 days. Wages must be updated if contract award is more than 90 days after bid opening. Once a Davis-Bacon wage rate has been locked, it stays in effect for the duration of the project. These wage determinations shall be incorporated into solicitations and any

subsequent contracts. In addition, the wage determination and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor or subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

- The Borrower shall ensure that the required EPA contract language regarding Davis-Bacon Wages is in all contracts and sub-contracts in excess of \$2,000.
 Borrower shall ensure no contracts are awarded to contractors excluded from federal contracts.
- iii. The Borrower shall periodically interview 10% of the work force entitled to Davis-Bacon prevailing wages to verify that contractors or subcontractors are paying the appropriate wage rates. Borrowers shall immediately conduct interviews in response to an alleged violation of the prevailing wage requirements. As provided in 29 CFR 5.6(a)(5), all interviews must be conducted in confidence. The Borrower must use Standard Form 1445 or equivalent documentation to memorialize the interviews.
- iv. The Borrower shall periodically conduct spot checks of a representative sample of weekly payroll data to verify that contractors or subcontractors are paying the appropriate wage rates. In addition, during the examinations, the Borrower shall verify evidence of fringe benefit plans and payments thereunder by contractors and subcontractors who claim credit for fringe benefit contributions. The Borrower shall maintain records sufficient to document compliance with the Davis-Bacon Act, and make such records available for review upon request.
- v. In addition, the Borrower shall consult with the Department on any required contract or bid documents to ensure that appropriate federal "Davis Bacon Act" language and material is included in the documentation.
- d. Title I Employment of the Americans with Disabilities Act of 1990

When applicable, the Borrower shall comply with Title I-Employment of the Americans with Disabilities Act of 1990 (P.L. 101-336) and in accordance with Title I of that Act, shall not discriminate against a qualified individual with a disability because of the disability of such individual in regard to job application procedures, the hiring, advancement, or discharge of employees, employee compensation, job training, and other terms, conditions, and privileges of employment.

e. Title II-Public Services of the Americans with Disabilities Act of 1990

When applicable, the Borrower shall comply with Title II-Public Services of the Americans with Disabilities Act of 1990 (P.L. 101-336) and in accordance with Title II of the Act, no qualified individual with a disability shall, by reason of such disability, be excluded from participation in or be denied the benefits of the services, programs, or activities of a public entity, or be subjected to discrimination by any such entity.

f. Title II, Part 35, Section 35.151 of the Americans with Disabilities Act "New Construction and Alterations"

When applicable, the Borrower shall comply with Title II, Part 35, Section 35.151 of the Americans with Disabilities Act "New Construction and Alterations."

- Design and construction: Each facility or part of a facility constructed by, on behalf of, or for the use of a public entity shall be designed and constructed in such manner that the facility or part of the facility is readily accessible to and usable by individuals with disabilities, if the construction was commenced after January 26, 1992.
- ii. Alteration: Each facility or part of a facility altered by, on behalf, of or for the use of a public entity in a manner that affects or could affect the usability of the facility or part of the facility shall, to the maximum extent feasible, be altered in such manner that the altered portion of the facility is readily accessible to and usable by individuals with disabilities, if the alteration was commenced after January 26, 1992.
- iii. Accessibility standards: Design, construction or alteration of facilities in conformance with the Uniform Federal Accessibility Standards (UFAS) (Appendix A to 41 CRF part 101-19.6) or with the Americans with Disabilities Act Accessibility Guidelines for Buildings and Facilities (ADAAG) (Appendix A to 28 CFR Part 36) shall be deemed to comply with the requirements of this section with respect to those facilities, except that the elevator exemption contained at section 4.1.3(5) and section 4.1.5(1)(j) of ADAAG shall not apply.
- g. Title III, Part 36, Section 36.401 of the Americans with Disabilities Act "New Construction"

When applicable, the Borrower shall comply with Title III, Part 36, Section 36.401 of the Americans with Disabilities Act "New Construction." Except as provided in paragraph (b) and (c) of the Act, discrimination for purposes of this part includes a failure to design and construct facilities for first occupancy after January 26, 1993, that are readily accessible to and usable by individuals with disabilities.

h. Title III, Part 36, Section 36.402 of the Americans with Disabilities Act "Alterations"

When applicable, the Borrower shall comply with Title III, Part 36, Section 36.402 of the Americans with Disabilities Act "Alterations."

i. General: Any alteration to a place of public accommodation or a commercial facility, after January 26, 1992, shall be made so as to ensure that, to the maximum extent feasible, the altered portions of the facility are readily accessible to and usable by individuals with disabilities, including individuals who use wheelchairs.

- ii. Alteration: An alteration is a change to a place of public accommodation or a commercial facility that affects or could affect the usability of the building or facility or any part thereof.
- i. 2 CFR Part 180, Responsibilities of Participants Regarding Transactions

The Borrower shall fully comply with Subpart C of 2 CFR Part 180, entitled "Responsibilities of Participants Regarding Transactions." The Borrower is responsible for ensuring that any lower tier covered transaction, as described in Subpart B of 2 CFR Part 180, entitled "Covered Transactions," includes a term or condition requiring compliance with Subpart C. The Borrower is responsible for further requiring the inclusion of a similar term or condition in any subsequent lower tier covered transactions. The Borrower acknowledges that failing to disclose the information required under 2 CFR Part 180 may result in the delay or negation of this assistance Agreement, or pursuance of legal remedies, including suspension and debarment.

j. American Iron and Steel

Per Section 608 of the Clean Water Act, none of the funds made available to the Borrower shall be used for a project for the construction, alteration, maintenance, or repair of a treatment works unless all of the iron and steel products used in the project are produced in the United States.

The Borrower may request a waiver to this requirements if:

- i. It is inconsistent with the public interest;
- ii. Iron and steel products that are not produced in the United States in sufficient and reasonably available quantities and of a satisfactory quality; or
- iii. Inclusion of iron and steel products produced in the United States that will increase the cost of the overall Project by more than 25 percent.

Waiver requests shall be submitted to the Department by the Borrower; the Department will then forward the request to the Environmental Protection Agency (EPA) for consideration. EPA will make a copy of the request, and information available to the Administrator concerning the request, available to the public on an EPA website for at least fifteen days for informal public input prior to making a finding.

2.17 **Project Completion**

- a. Upon completion of the Project, the Borrower shall provide a statement to the Department of the project final costs by category of expenditure, including but not limited to costs for administration, design, construction engineering, construction and equipment.
- b. Prior to initiation of operations, the Borrower shall submit to the Department the following:

- i. Criteria for project performance.
- ii. An adopted sewer use ordinance and a user charge.
- c. Upon substantial completion of the Project, the Borrower shall initiate operation of the Project Facility and immediately notify the Department in writing of Initiation of Operation. If construction is complete except for minor items, and the facility is operating, but the Borrower has not sent a notice of Initiation of Operation, the Department will, in its discretion, assign an Initiation of Operations date.
- d. For any Project that requires an Approval to Construct from the Department, the Borrower must obtain a Final Approval to Operate.
- e. Upon Initiation of Operation, the Borrower shall certify that the Project Facility is and will be operated by sufficiently qualified operating personnel certified at the system classification level by the State of Alaska.
- f. Within one year of the initiation of operation, the Borrower shall:
 - i. Be subject to a final inspection by the Department;
 - ii. Submit to the Department a manual for operations and maintenance of the Project Facility for Department approval;
 - iii. Submit to the Department a Project Performance Certification certifying that the facility is or is not performing up to design standards.
- g. Nothing contained in this Agreement shall be construed as an obligation or pledge of the Borrower to appropriate or expend general funds and general revenues of the Borrower to operate or maintain the Project Facility.

2.18 Amendments and Modifications

This Agreement may only be modified or amended in writing and executed by the authorized representatives of all parties to this Agreement.

2.19 Disputes

The Borrower shall raise any concerns or issues it may have regarding the Project with the Department promptly and prior to project completion. If those concerns or issues are not satisfactorily resolved, the Borrower shall promptly give written notice to the Department with a detailed description of the continuing concerns or issues. Jurisdiction and venue for any legal dispute shall be in the Superior Court for the State of Alaska, Third Judicial District at Anchorage, and in no other court or location. In the event of a legal dispute, both parties knowingly and voluntarily waive their right to trial by jury (including any advisory jury) and elect to have the dispute tried only to a judge. In the event of litigation, the prevailing party shall be entitled to an award of its reasonable, actual attorney's fees and costs of litigation. Each party has had an opportunity to review this Agreement with legal counsel of its

choosing (or waived such opportunity), therefore this Agreement shall not be interpreted in favor of either party. This Agreement shall be governed by the laws of the State of Alaska.

2.20 Termination

The Department may cancel all or any part of this Agreement if:

- a. Any representation or other statement made by the Borrower to the Department in connection with its application for a loan from the Alaska Clean Water Fund is incorrect or incomplete in any material respect;
- b. The Borrower has violated commitments made in the Approved Application and supporting documents, has not adhered to the regulations of the Alaska Clean Water Fund (18 AAC 76), has violated any of the terms of this Agreement; or
- c. The financial position of the Borrower has, in the opinion of the Department, suffered a materially adverse change.

2.21 Indemnification

The Borrower shall defend with counsel of the Department's choosing, indemnify, and hold harmless the Department and the State of Alaska, and their agents, servants, contractors, and employees, from and against any and all claims, demands, causes of action, actions, and liabilities arising out of, or in any way connected with this funding or the Project for which the funding is made, howsoever caused, except to the extent that such claims, demands, causes of action, actions or liabilities are the proximate result of the sole negligence or willful misconduct of employees or agents of the Department or the State of Alaska.

Article 3. Definitions

Except where the context clearly indicates otherwise, terms used in this Agreement will have the meaning ascribed to them in this section.

- a. "Approved Application" means the application submitted to the Department on February 13, 2020, together with all attachments and supporting documentation, as approved by the Department.
- b. "Finance Charge" means 1.5 percent per annum.
- c. "Agreement Period" means the time period commencing on the date this Agreement is signed by the Department's Finance Officer and terminating on the date the Borrower repays the loan in full.
- d. "Subsidy" means principal forgiveness awarded under this Agreement.
- e. "Eligible Project Costs" include the following costs disbursed from the Alaska Clean Water Fund, estimated to not exceed \$579,867 for demolition, construction, engineering, machinery, furnishings, equipment, surveys, plans, estimates, specifications, necessary insurance, financial and environmental investigations, laboratory testing, resident engineering and inspection fees, force account, legal expenses, and any other necessary miscellaneous expenditures, minus the amount of any grant applicable to foregoing costs.
- f. "Initiation of Operation" means the date of which the project or project facility begins operating for the purposes for which it was planned, designed, or built.
- g. "Iron and Steel Products" means the following products are primarily of iron and steel: lined or unlined pipes and fittings, manhole covers and other municipal castings, hydrants, tanks, flanges, pipe clamps, and restraints, vales, structural steel, reinforced precast concrete, construction materials.
- h. "Net Operating Revenues" means revenues of Borrower's system after payment of operation and maintenance costs of the system.
- i. "Project" means the activities described in Article 1.02.
- j. "Project Facility" means wastewater treatment plant, collection system, or related facilities in which the Project activities are occurring.
- k. "Project Site" means the location at which the Project activities are occurring.

Article 4. Signatures

This Agreement is binding upon the parties specified below, and to any person, office, or board succeeding either of the parties. This Agreement may not be assigned by the Borrower without written consent of the Department.

Nothing in this Agreement, whether or not accepted, may be deemed to constitute a contractual obligation on the part of the Department until the Agreement is signed by all parties.

Alaska Department of Environmental Conservation

Randy Bates, Director Division of Water

> ACKNOWLEDGEMENT STATE OF ALASKA First Judicial District



The foregoing instrument was acknowledged before me this 15th day of April , 2020

Notary Public, State of Alaska My commission expires: with office

Alaska Department of Environmental Conservation

By:

By:

Elizabeth Pederson, Finance Officer Division of Administrative Services

> ACKNOWLEDGEMENT STATE OF ALASKA First Judicial District

The foregoing instrument was acknowledged before me this 15 day of 1940	pril, 2020
---	------------

Notary Public, State of Alaska My commission expires: <u>with office</u>

Haines Borough

By:

Debra Schnabel Borough Manager

ACKNOWLEDGEMENT STATE OF ALASKA First Judicial District

The foregoing instrument was acknowledged before me this _____ day of _____, 2020

Notary Public, State of Alaska My commission expires: _____

Exhibit "A" – List of Federal Laws and Authorities (FEDERAL CROSS-CUTTING AUTHORITIES)

Environmental and Cultural Authorities:

- Archeological and Historic Preservation Act of 1974, Public Law 93-291
- Archeological Resources Protection Act, Public Law 96-95 as amended
- Bald and Golden Eagle Protection Act, 16 U.S.C. 668-668c
- Clean Air Act, Public Law 95-95, as amended
- Clean Water Act, Public Law 92-50, as amended
- Coastal Barriers Resources Act, Public Law 97-348
- Coastal Zone Management Act of 1972, Public Law 92-583, as amended
- Consultation and Coordination with Indian Tribal Governments, Executive Order 13175
- Endangered Species Act, Public Law 93-2015 as amended
- Environmental Justice, Executive Order 12898
- Essential Fish Habitat Consultation Process under the Magnuson-Stevens Fishery Conservation and Management Act, Public Law 94-265, as amended
- Farmland Protection Policy Act, Public Law 97-98
- Fish and Wildlife Coordination Act, Public Law 85-624, as amended
- Floodplain Management, Executive Order 11988, as amended by Executive Order 12148
- Marine Mammal Protection Act, 16 U.S.C. 1361
- Migratory Bird Treaty Act, 16 U.S.C. 703
- National Historic Preservation Act of 1966, Public Law 89-665
- Native American Graves Protection and Repatriation Act, Public Law 101-601
- Protection and Enhancement of the Cultural Environment, Executive Order 11593
- Protection of Wetlands, Executive Order 11990, as amended by Executive Order 12608
- Rivers and Harbors Act, 33 U.S.C. 403
- Safe Drinking Water Act, Public Law 93-523, as amended
- Wild and Scenic Rivers Act, Public Law 90-542

Social Policy Authorities:

- Age Discrimination Act of 1975, Public Law 94-135
- Title VI of the Civil Rights Act of 1964, Public Law 88-352
- Section 13 of the Federal Water Pollution Control Act Amendments of 1972, Public Law 92-500 (the Clean Water Act)
- Section 504 of the Rehabilitation Act of 1973, Public Law 93-112
- Equal Employment Opportunity, Executive Order 11246
- Disadvantage Business Enterprise Provisions
 - Promoting the Use of Small, Minority, and Women-owned Businesses, Executive Orders 11625, 12138, and 12432
 - Section 129 of the Small Business Administration Reauthorization and Amendment Act of 1988, Public Law 100-590
 - Department of Veterans Affairs and Housing and Urban Development, and Independent Agencies appropriations Act of 1993, Public Law 102-389

Economic Authorities:

- Procurement Prohibitions Under Section 306 of the Clean Air Act and Section 508 of the Clean Water Act, included Executive Order 11738, Administration of the Clean Water Act with Respect to Federal Contracts, Grants, or Loans
- Demonstration Cities and Metropolitan Development Act of 1996, Public Law 89-754 as amended

Miscellaneous Authority:

- Debarment and Suspension, Executive Order 12549
- Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, Public Law 91-646 as amended
- Preservation of Open Competition and Government Neutrality Towards Government contractors' Labor Relations on Federally Funded Constructed Projects, Executive Order 13202, as amended by Executive Order 13208
- Prohibition Against Sex Discrimination Under the Federal Water Pollution Control Act, Section 13 of Public Law 92-500
- 40 CFR Part 34, New Restrictions on Lobbying

Alaska Clean Water Fund - State Fiscal Year 2023 (SFY23) Project Priority List - 1st and 2nd Quarter Updates

Note: Available funding for SFY23 projects is \$94.1 million.

(1) Within Funding Limits column indicates that the project is within the current fundable limit of the Alaska Drinking Water Fund. Large projects (over \$5 million) may be phased based on projected funding needs during the next year.

Loan applications may be submitted for any project within the funding limits that is ready to proceed.

(2) Principal forgiveness is subject to change depending on the readiness of projects to proceed.

(a) Independent of the second s

Rank Score	Within Funding Limit	APDES Permit Number	Clean Water Needs Category	Applicant	Project Name and Description	Requested Loan Amount	Estimated Principal Forgiveness (SFY22 and previous years) (2)	Estimated Principal Forgiveness (SFY23) (2)	Disadvantaged Community	Requested Loan Term (years) (3)	Green Project Category & Amount	Sustainability Policy	Estimated Construction Start	Quarter Added to PPL
---------------	----------------------	------------------------	-------------------------------------	-----------	------------------------------	--------------------------	--	---	-------------------------	---	---------------------------------------	--------------------------	------------------------------------	----------------------------

POINT SOURCE PROJECT QUESTIONNAIRES

1	655	x		хіі	Mile 8 Utilities, LLC	Leach Field Design and Construction - Design and construct an aerated leach field to eliminate a point source discharge into an anadromous stream. Rehabilitate pretreatment works to ensure leach field longevity. Replace pre-treatment lagoon liner.	\$525,987			x	20 to 30		Fix It First	5/1/2021	SFY21-Q1
2	650	x	AKG573029	III-B	Bristol Bay Borough	King Salmon Lagoon Upgrade - Upgrade current lagoon system to a ultraviolet (UV) treatment system to ensure discharges are compliant with permit requirements.	\$3,100,000		\$500,000	x	5 to 20		Fix It First	5/30/2023	SFY23-Q1
3	625	x	AKG572028	I	Ketchikan Gateway Borough	Mountain Point Wastewater Treatment Plant Upgrades - Install new vector waste intake at headworks, install new ultraviolet disinfection system, extend influent piping to reduce odors, new flow meters and additional basin instrumentation. These improvements will improve the quality of wastewater discharged to the ocean.	\$2,250,000		\$500,000	x	20 to 30		Fix It First	6/30/2024	SFY23-Q1
4	520	x	AK0021440	III-B	Ketchikan	Water Street Sewer Main Replacement - Replace or rehabilitate existing sewer lines that have been determined to be significant contributors to inflow and infiltration at the Charcoal Point Wastewater Treatment Plan and also contribute to a general decline in water quality in the area.	\$3,900,000	\$500,000		x	20 to 30		Fix It First	7/1/2022	SFY22-Q4
5	405	x	AK0021385	I	Haines Borough	Wastewater Treatment Plant Influent Upgrade - Demolish the existing wet well located within the control building and provide a new exterior wet well and a below-grade valve vault. This project will prevent debris from entering the plant during significant storm events and provide for safer working conditions within the plant.	\$2,115,758			x	20 to 30		Fix It First	6/1/2022	SFY23-Q1
6	310	x	AK0021890	I	Seward	Lowell Point Lagoon Blower Improvements - Remove and replace the main blowers at the Lowell Point wastewater treatment plant with high efficiency blowers.	\$547,500			x	5 to 20	Energy Efficiency TBD	Fix It First	8/5/2022	SFY23-Q1
7	310	x	AK0021890	I	Seward	Lowell Point Lagoon Fine Bubble Aeration - Upgrade 30-year old coarse bubble diffuser with new fine bubble diffuser to increase bacteria efficiency and reduce lagoon odors.	\$637,500			х	5 to 20	Energy Efficiency TBD	Fix It First	5/27/2022	SFY23-Q1
8	325	x	AK0021440	III-B	Ketchikan	Tongass Sewer Force Main Rehabilitation Phase II - Reconstruct a segment of aging force main. An in- situ rehabilitation technique called slip-lining has been proposed. This project would involve slip-lining approximately 1,250 feet or more of force main.	\$1,500,000			x	5 to 20	Energy Efficiency TBD	Fix It First	11/1/2021	SFY22-Q2
9	320	x	AK0022951	I	Juneau	Mendenhall Wastewater Treatment Plant (MWWTP) Influent Piping - Install new piping to bypass the now obsolete screening equipment located one floor above the rest of the treatment plant.	\$994,000				20 to 30	Energy Efficiency \$994,000	Fix It First	1/1/2022	SFY22-Q2
10	315	x	AK0021245	III-B	Homer	Beluga Sewer Lift Station Improvements - Reconfigure and rehabilitate the lift station to reduce corrosion and allow for greater ease of maintenance.	\$2,937,353			x	20 to 30	Energy Efficiency TBD	Fix It First	1/17/2022	SFY22-Q3
11	282 ⁽⁴⁾	x	AK0022551	I III-A III-B	Anchorage AWWU	SFY23 Programmatic Financing (Pro Fi) Loan - The applicant has provided a list of eligible projects including planning, design, engineering, and construction activities for wastewater infrastructure projects that may be financed through the SFY23 Pro Fi loan agreement (see attached Pro Fi project list).	\$22,511,580				20	Energy Efficiency \$2,000,000	Fix It First	5/1/2023	SFY23-Q1
12	280	x	AK0021555	III-B	Kodiak	Aleutian Homes Phase VII Wastewater Main Replacement - Replace 65-year old asbestos concrete wastewater collection system.	\$2,600,000			х	20 to 30		Fix It First	5/11/2021	SFY21-Q3
13	270	x	AKG573004	III-B IV-A	Dillingham	Waterfront Wastewater System Upgrade (Design) - Complete design for the extension and rehabilitation of the existing wastewater collection system in the Dillingham waterfront area.	\$44,125			x	20 to 30		Effective Utility Mgmt	6/1/2021	SFY22-Q1
Note: Available funding for SFY23 projects is \$94.1 million.

(1) Within Funding Limits column indicates that the project is within the current fundable limit of the Alaska Drinking Water Fund. Large projects (over \$5 million) may be phased based on projected funding needs during the next year.

Loan applications may be submitted for any project within the funding limits that is ready to proceed.

(2) Principal forgiveness is subject to change depending on the readiness of projects to proceed.

(a) Independent of greaters and the compared of the compared of the processing of the processing of the compared of the processing of the processing

Rank	Score	Within Funding Limit	APDES Permit Number	Clean Water Needs Category	Applicant	Project Name and Description	Requested Loan Amount	Estimated Principal Forgiveness (SFY22 and previous years) (2)	Estimated Principal Forgiveness (SFY23) (2)	Disadvantaged Community	Requested Loan Term (years) (3)	Green Project Category & Amount	Sustainability Policy	Estimated Construction Start	Quarter Added to PPL
14	270	x	AKG573004	III-B IV-A	Dillingham	Waterfront Wastewater System Upgrade (Construction) - Based on the proposed design plan for the waterfront area, construct improvements including the extension of the wastewater system as well as rehabilitation of the existing collection system.	\$603,550			x	20 to 30		Effective Utility Mgmt	7/1/2021	SFY22-Q1
15	270	x	AK0020010	IV-A	Skagway	Waterfront Sewer Extension - Extend the collection system to waterfront properties for service to existing structures that are currently on septic systems and holding tanks.	\$2,187,500			х	20 to 30		Effective Utility Mgmt	4/1/2022	SFY22-Q1
16	270	x	AK0021245	IV-A	Homer	Mission Road Sewer Trunk Line - Install approximately 5,340 feet of 8-inch HDPE sewer trunk line pipe. This project would provide piped service to four homes located directly adjacent to the main and provide the opportunity to serve many more homes in nearby subdivisions.	\$1,493,506			x	20 to 30		Effective Utility Mgmt	1/17/2022	SFY22-Q2
17	270	x	AK0021245	IV-A	Homer	West Hill Road Sewer Trunk Line - Install approximately 17,320 feet of 8-inch ductile iron pipe and 2,250 linear feet of 3-inch force main. This project would provide piped service to about 75 homes located directly adjacent to the main and provide the opportunity to serve many more homes in nearby subdivisions.	\$2,937,353			x	20 to 30		Effective Utility Mgmt	1/17/2022	SFY22-Q2
18	260	x	AK0021245	IV-A	Homer	Bunnell-Charles Way Sewer Main Extension - Extend the wastewater collection system to provide piped service to 23 developed central business district lots that currently use temporary service connections, holding tanks, or outhouses for sanitary service.	\$631,834			x	20 to 30		Effective Utility Mgmt	8/1/2021	SFY22-Q1
19	190	x		III-B	Kotzebue	Fire Hall Lift Station and Sewer System - Replace sections of existing gravity main with 8-inch insulated pipe, replace the existing Fire Hall Lift Station, construct an additional 8-inch insulated arctic force main to allow for increased capacity in transmission of wastewater to Lagoon Cell 1 from existing lift stations.	\$2,662,000			x	5 to 20		Fix It First	9/1/2022	SFY23-Q2
20	180	x	AK0020036	1	Soldotna	Biosolids Dewatering System - Design and construct dewatering belt press replacement including equipment selection, facility modifications, and installation.	\$1,200,000			x	5 to 20		Fix It First	7/1/2023	SFY23-Q2
21	180	x	AK0020036	I	Soldotna	Refurbish Headworks Building - Update the existing headworks building to include air sensors, screening, dewatering, compacting, and grit removal. The existing equipment has been in place more than 30 years and has exceeded its useful life.	\$850,000			x	5 to 20		Fix It First	1/1/2027	SFY23-Q2
22	125	х	AK0021890		Seward	Lowell Point Lagoon Fence - Replace security fencing around wastewater treatment lagoon.	\$49,094			х	<5 years			5/1/2022	SFY22-Q4
23	115	x	2007- DB0003		Nome	Equipment Response / Storage / Office Facility - Construct a building to support sewer utility, amalgamate ancillary facilities, reduce operating costs, protect equipment, and improve health and safety of the work environment. The facility will also support the drinking water utility. The cost of construction would be split between the Alaska Clean Water Fund and the Alaska Drinking Water Fund.	\$5,025,000			x	20 to 30	Energy Efficiency \$1,000,000	Effective Utility Mgmt	1/17/2022	SFY22-Q2
24	80	x	2003DB0096 1016		Craig	Wastewater Treatment Plant Roof Replacement - Replace leaking roof to protect treatment plant components. Upgrade insulation designed for corrosive environment.	\$400,000			x	5 to 20			8/15/2022	SFY23-Q1
25	55	x	AK0020036	I	Soldotna	Water Treatment - Study and treat groundwater at existing municipal wells to limit concentrations of metals (copper and zinc) from the City's wastewater treatment plant effluent discharges to the Kenai River in accordance with anticipated new permit limits.	\$2,600,000			x	5 to 20			7/1/2025	SFY23-Q2
26	55	x	AK0020036	I	Soldotna	pH Control at Wastewater Treatment Plant - Design and construct modifications to allow continuous monitoring of effluent pH levels.	\$260,000			x	5 to 20			3/1/2023	SFY23-Q2
27	35	х	AK0020010	IV-A	Skagway	Klondike Highway Sanitary Sewer Extension - Extend sanitary sewer to an unserved area.	\$3,948,700			х	20 to 30			4/1/2021	SFY21-Q1

Note: Available funding for SFY23 projects is \$94.1 million.

(1) Within Funding Limits column indicates that the project is within the current fundable limit of the Alaska Drinking Water Fund. Large projects (over \$5 million) may be phased based on projected funding needs during the next year.

Loan applications may be submitted for any project within the funding limits that is ready to proceed.

(2) Principal forgiveness is subject to change depending on the readiness of projects to proceed.

(a) Independent of projects are reviewed and assigned a weighted score based on the total project cost. The overall score for the Pro Fi questionnaire is the sum of weighed scores for all of the Pro Fi projects.

Rank	Score	Within Funding Limit	APDES Permit Number	Clean Water Needs Category	Applicant	Project Name and Description	Requested Loan Amount	Estimated Principal Forgiveness (SFY22 and previous years) (2)	Estimated Principal Forgiveness (SFY23) (2)	Disadvantaged Community	Requested Loan Term (years) (3)	Green Project Category & Amount	Sustainability Policy	Estimated Construction Start	Quarter Added to PPL
28	30	x	9725DB005		Bethel	Refinance USDA RD Loan for Construction of Jetty at Sewage Lagoon - Refinance principal balance of existing loan/grant issued by US Department of Agriculture Rural Development for construction of a jetty and the purchase of two sewage haul trucks.	\$913,000			x	5 to 20			6/22/2022	SFY23-Q2
						POINT SOURCE SUBTOTAL	\$69,425,340	\$500,000	\$1,000,000			\$4,444,000			

NONPOINT SOURCE PROJECT QUESTIONNAIRES

-												
	1 195	x	 VII-F	Cordova	Piling Replacement and Waste Handling - Remove and replace approximately 135 creosote pilings in the South Harbor with steel pilings. Install a marine boat sewage pump station to allow boats to dispose of sewage and gray water.	\$2,000,000		x	20 to 30		1/2/2022	SFY22-Q1
	2 160	x	 VI-B	Homer	Baycrest Storm Drainage - Design and construct a system to capture and convey stormwater away from highly erodible bluffs. The project would include property acquisition as well as storm drain and retention basin construction in conformance with state and federal permitting requirements. Through the conveyance system, concentrated runoff may be used to generate hydroelectricity.	\$1,000,000		x	5 to 20	Environ- mental Innovation TBD	5/1/2022	SFY22-Q4
	3 160	x		Kotzebue	Storm Drain Planning, Design and Construction - Conduct inflow and infiltration study for Lift Station 8. Conduct hydrologic study to identify areas draining toward Lift Station 8 to estimate stormwater flow diversion needs, assess snow storage methods and locations. Construct storm drain with thaw wire. Based on recommendations of snow management planning, implement eligible capital improvements for snow management in catchment area.	\$2,456,000		x	5 to 20		9/1/2022	SFY23-Q2
	4 140	x	 VI-B	Homer	Ben Walters Drainage Stormwater Treatment - Acquire 8.18 acres of private, undeveloped land adjacent to Beluga Lake. Design and construct storm drain and sediment control works in conformance with state and federal permitting requirements.	\$280,190		x	5 to 20	Environ- mental Innovation TBD	5/1/2022	SFY22-Q4
	5 140	x	 VI-B	Homer	Bishop's Beach Stormwater Pollution Control - Design and construct a system to channel untreated stormwater into a green infrastructure feature before discharge to Beluga Slough and Kachemak Bay. Project would include acquisition of 2.49 acres of land and construction of green infrastructure features in conformance with state and federal permitting requirements.	\$290,978		x	5 to 20	Environ- mental Innovation TBD	7/1/2022	SFY22-Q4
	5 140	x		Nome	Tank Farm Relocation - Relocate the existing tank farm to a more stable location. Due permafrost and climate change, the existing tank farm location is subject to differential settling that requires ongoing leveling and maintenance to avoid tank failure. The bulk fuel tank farm supports community electric power generation needs which in turn provides essential support to the community water and sewer system. The tank relocation site is a former US Air Force contaminate site that will require specific site development and construction attributable to the brownfield site. These costs are proposed for financing through the Clean Water Fund as a nonpoint source project.	\$4,500,000		x	5 to 20		5/15/2023	SFY23-Q2
	7 135	x	 VII-J	King Cove	Landfill Cell Capping and Closure - Install a partial closure system as required by state regulations (18 AAC 60.390) to stabilize slopes, minimize infiltration of liquids and soil erosion, and protect against the release of hazardous constituents to the environment at the King Cove Landfill.	\$67,318		x	5 to 20		10/1/2021	SFY22-Q3
	3 115	x	 VI-B	Homer	Beluga Wetland / East Kachemak Drive - This project would involve the acquisition, or conservation easement designation, of 80 acres of wetland in a predominately industrial area to be used as a stormwater retention and treatment area. Design and construct storm drain and outfall in conformance with state and federal permitting requirements.	\$1,000,000		x	5 to 20	Environ- mental Innovation TBD	1/31/2022	SFY22-Q4

Note: Available funding for SFY23 projects is \$94.1 million.

(1) Within Funding Limits column indicates that the project is within the current fundable limit of the Alaska Drinking Water Fund. Large projects (over \$5 million) may be phased based on projected funding needs during the next year.

Loan applications may be submitted for any project within the funding limits that is ready to proceed.

(2) Principal forgiveness is subject to change depending on the readiness of projects to proceed.

(3) Loan terms will be finalized when a loan agreement is offered. The finance rate will be based on a calculation identified in Alaska Administrative Code (18 AAC 76). (4) Individual Pro Fi projects are reviewed and assigned a weighted score based on the total project cost. The overall score for the Pro Fi questionnaire is the sum of weighed scores for all of the Pro Fi projects.

Rank	Score	Within Funding Limit	APDES Permit Number	Clean Water Needs Category	Applicant	Project Name and Description	Requested Loan Amount	Estimated Principal Forgiveness (SFY22 and previous years) (2)	Estimated Principal Forgiveness (SFY23) (2)	Disadvantaged Community	Requested Loan Term (years) (3)	Green Project Category & Amount	Sustainability Policy	Estimated Construction Start	Quarter Added to PPL
9	45	x		VII-J	Fairbanks North Star Borough	Cell 4 Expansion - Design and construct a new lined landfill cell. Costs specifically associated with landfill leachate collection and treatment may be eligible for financing through the SRF Program.	\$7,000,000				5 to 20			3/15/2022	SFY23-Q1
10	0 10	x		VII-J	Matanuska Susitna Borough	Landfill Gas Collection System - Install vertical wells in two closed cells to extract gas that will be burned with a flare. Proper management of the landfill reduces leachate quality issues.	\$2,420,000			x	5 to 20				SFY21-Q2
						NONPOINT SOURCE SUBTOTAL	\$21,014,486	\$0	\$0						

AMENDMENT TO EXISTING LOAN AGREEMENT

	x	AK0021385	IV-A	Haines Borough	Wastewater Influent and Pump Station Upgrade (Loan 395261-S) - Loan amendment to increase existing loan amount by \$80,000 (total loan request \$659,867) and modify the scope of the existing loan agreement to include construction of 500 linear feet of sewer main at the correct and depth an alignment to tie into the original main. The project scope has also been amended to include Supervisory Control and Data Acquisition (SCADA) system and PLC upgrades to monitor and track the system remotely.	\$80,000			x	20		SFY23-Q1
		2007- DB0003	III-B	Nome	Nome Bering Street Sewer Improvements (Loan 627251-SG) - Loan amendment to modify the scope of the existing Bering Street Ioan agreement to include replacement of sewer lines along Seppala Drive. No additional Ioan funds are requested.				x	20		SFY22-Q1
	x	Pending	I	Matanuska Susitna Borough	Matsu Septage and Leachate Treatment Facility (Loan 561041) - Loan amendment to increase existing loan amount by \$1,000,000 and amend project scope as follows: Design a new energy efficient septage and leachate facility to minimize septage and leachate costs and environmental impacts in the Matanuska-Susitna Valley. Additionally, this project will install a landfill leachate treatment facility.	\$1,000,000			x	20	Effective Utility Mgmt	SFY21-Q1
						\$1,080,000	\$0	\$0				

SUSTAINABLE INFRASTRUCTURE PLANNING PROJECT QUESTIONNAIRES

1	65	x	AK0021555	Plan & Assess	Kodiak	Infiltration and Inflow (18.1) Assessment and Reduction - Flow monitoring, flow data analysis and identification of areas with high 18.1 through closed-circuit television inspections and manhole inspections.	\$165,000	\$75,000	x	5	Planning	6/1/2020	SFY21-Q1
2	55	x	AK0021385	Plan & Assess	Haines Borough	Haines Sanitary Sewer Inflow and Infiltration (I&I) Study - Flow monitoring, flow data analysis and identification of areas with high I&I through closed-circuit television inspections and manhole inspections.	\$100,000	\$75,000	x	5	Planning	10/3/2022	SFY23-Q1
3	55	x	9725DB005	Plan & Assess	Bethel	Community-wide Utility System Expansion Preliminary Engineering Report and Environmental Assessment - Complete the planning documents necessary to estimate the cost to construct a wastewater collection system to serve over 2,000 households, commercial, and institutional connections. This planning document will address the construction of both water distribution and wastewater collection system with the cost split between the Alaska Drinking Water and Clean Water Funds.	\$100,450	\$75,000	x	5	Planning	3/22/2021	SFY22-Q1
4	55	x	AKG573004	Plan & Assess	Dillingham	Wastewater Rate Study - Update the 2014 Rate Study to reflect current conditions and future planning considerations. The Rate Study will include both water and wastewater utility rates; the cost of the study will be split evenly between the Alaska Drinking Water and Clean Water Funds.	\$30,000	\$30,000	x	5	Planning	6/1/2021	SFY22-Q1

Note: Available funding for SFY23 projects is \$94.1 million.

(1) Within Funding Limits column indicates that the project is within the current fundable limit of the Alaska Drinking Water Fund. Large projects (over \$5 million) may be phased based on projected funding needs during the next year.

Loan applications may be submitted for any project within the funding limits that is ready to proceed.

(2) Principal forgiveness is subject to change depending on the readiness of projects to proceed.

(a) Independent of projects are reviewed and assigned a weighted score based on the total project cost. The overall score for the Pro Fi questionnaire is the sum of weighed scores for all of the Pro Fi projects.

-	Rank Score	Within Funding Limit	APDES Permit Number	Clean Water Needs Category	Applicant	Project Name and Description	Requested Loan Amount	Estimated Principal Forgiveness (SFY22 and previous years) (2)	Estimated Principal Forgiveness (SFY23) (2)	Disadvantaged Community	Requested Loan Term (years) (3)	Green Project Category & Amount	Sustainability Policy	Estimated Construction Start	Quarter Added to PPL
	5 55	x	AKG573004	Plan & Assess	Dillingham	Wastewater Master Plan - Update the wastewater portion of the 2003 Water and Sewer Master Plan.	\$69,183	\$45,000		x	5		Planning	6/1/2021	SFY22-Q1
				SUSTAINABLE INFRASTRUCTURE PLANNING LOAN SUBTOTAL	\$464,633	\$300,000	\$0								

MICRO LOAN QUESTIONNAIRES

1	475	х	AK2250053	III-B	Unalakleet	Covenant Lift Station Rehabilitation and Septic Pumper Purchase - The purpose of this project is to rehabilitate a 45-year old lift station subject to freezing issues and sewage backups. Rehabilitation will include replacement of the heating and ventilation systems, overhead crane, safety grating, and electrical systems. A new septic pumper truck will also be purchased.	\$488,620	\$342,034		x	20		Fix It First	5/1/2022	SFY20-Q2
2	310	x	AKG380006	III-B	Seldovia	Lift Station Pump Replacement - Purchase and install two new pumps in the Beach and Slough lift stations and purchase one additional pump to serve as backup in case one pump fails.	\$48,125	\$33,688		x	10	Energy Efficiency \$40,000	Fix It First		SFY22-Q1
3	200	x		III-B	Togiak	Lift Station Pump Replacement - Rebuild/replace three lift station pumps including new impellers, bearings and armatures and purchase three backup pumps. Purchase a small backhoe specifically for the purpose of repairing/replacing utility lines. Purchase a new jetter truck for sewer line maintenance.	\$500,000		\$450,000	x	20	Energy Efficiency TBD	Fix It First		SFY23-Q2
						MICRO LOAN SUBTOTAL	\$536,745	\$375,722	\$450,000			\$0			
						TOTAL FUNDING REQUESTED (ALL CATEGORIES)	\$92,521,204	\$875,722	\$1,450,000			\$4,484,000			

EPA Needs	I Clean Water Treatment - Secondary Treatment Plant	III-B Clean Water Treatment - Sewer System Replacement/Rehabilitation	VI-B Green Infrastructure	VII-J Nonpoint Source Resource Activity - Sanitary Landfills
Category Codes	III-A Clean Water Treatment - Infiltration/Inflow Correction	IV-A Clean Water Treatment - New Collector Sewers & Appurtenances	VII-F Nonpoint Source Resource Activity - Marinas	XII Nonpoint Source Resource Activity - Individual/Decentralized Systems

Alaska Clean Water Fund

Programmatic Financing (Pro Fi) Projects

Applicant: Anchorage Water and Wastewater Utility SFY22 Loan Request: \$10,000,000 SFY23 Loan Request: \$22,511,580 Loan Term: 20 years

Y	ear	#	Project Name	Description
SFY22		C-19-03	Downtown Sewer Rehabilitation Phase III (individual projects listed below) Downtown Sewer Phase III, C&D Street Downtown Sewer Phase III, West 8th, N-P Street Downtown Sewer Phase III, D&E Street Downtown Sewer Phase III, M Street Downtown Sewer Phase III, M Street	Rehabilitate sewer main in downtown Anchorage. The sewer mains are located within the streets noted below.
SFY22		C-19-04	AWWTF Scum Pump & Inline Grinder	Make improvements to the scum handling system from the clarifiers to the incinerator. Improvements include piping, pumps, heating, insulation and controls.
SFY22		C-19-05b	King Street Septage Receiving Station	Design and construct upgrades to existing Septage Receiving Station with pretreatment equipment and increase user access. The pretreatment equipment will prevent collection system from having sanitary sewer overflows.
SFY22	SFY23	C-19-05c	King Street Warm Vehicle Storage	Design and construct a storage building to house equipment, necessary to operate and maintain the AWWU water and sewer infrastructure.
SFY22	SFY23	C-19-05e	King Street Main Building Upgrade	Design and construct various improvements to AWWU's King Street O&M Facility Administrative Building. Improvements include expanding and remodeling interior spaces and systems, and enclosing covered areas to increase the capacity, productivity, and efficiency of AWWU's support maintenance group.
SFY22	SFY23	C-19-05f	f King Street Fuel Storage Improvements	Relocate the existing fuel storage and dispensing system. This project will also streamline the traffic pattern within the facility.
SFY22		C-19-09	Pump Station 52 Improvements	Abandon and demolish existing sewage lift station and piping. Design and construct or install new sewage pump station, valve vault, wet well, sanitary sewer manholes, two pumps, check valves, pump controls, electrical upgrades, and standby generator. Temporary sewer bypass system will be used during construction. Existing utilities will be relocated within existing developed easements and right-of-way to accommodate work and provide better access.
SFY22		C-19-10	AWWTF Storage	Add additional warm storage for equipment, materials and sodium hypochlorite at Asplund Wastewater Treatment Facility.
SFY22		C-19-14	AWWTF Raw Sludge Pumps	Replace existing raw sludge pumps at Asplund Wastewater Treatment Facility.
SFY22	SFY23	C-20-25	Pump Station 2 Rehabilitation	Rehabilitate Pump Station 2 in order to reduce the risk of sanitary sewer overflows, emergency repairs. Replace high voltage electrical system, aging and corroding piping, valves, control systems, and various site improvements for Pump Station 2.
SFY22		C-22-01	E 42nd Ave Upgrade - Sewer	To prevent sewer backups associated with bellies and damaged pipe, re-route a section of sewer main to a new alignment in a dedicated sewer easement within MOA right-of-way. The replacement sewer and manholes will be constructed on helical piles.
SFY22	SFY23	C-22-02	Pump Station 12 Force Main Interceptor C - Gravity Junction Rehab	Assess and rehabilitate Pump Station 12, force mains, gravity junction box, and the receiving 48-inch gravity sewer. The culverts that support the force mains for the Campbell Creek crossing will also be assessed and rehabilitated as part of the project.
SFY22	SFY23	C-22-03	Turpin Septage Receiving Station	Assess and rehabilitate the Turpin Septage Receiving Station.
SFY22		C-22-04	W 72nd Ave Trunk Rehabilitation	Rehabilitate a corroded 15-inch corrugated metal sewer main. This project will either line with cured-in-place pipe or directly replace the failing pipe.
SFY22		C-22-05	Wastewater Master Plan	Update the 25-year comprehensive plan for AWWU to maintain and modify sewer service for the Municipality of Anchorage. The plan describes the condition of the AWWU wastewater system, projects future wastewater needs, analyzes system deficiencies, recommends system improvements, and provides a schedule for implementation through the capital improvements program.
	SFY23	C-23-01	D-2-4 Trunk Improvements	Design and construct improvements to the D-2-4 trunk main to improve the ability to access and maintain the line and to enhance capacity to avoid sanitary sewer overflows.



11A3

Assembly Agenda Bill

Agenda Bill No.: 22-1211

Assembly Meeting Date: ____9/13/22

Business I tem Description:	Attachments:
Subject: Authorize contract for Design-Build with	1. Resolution 22-09-996
Turnagain Marine Construction, \$310,000, for Lutak	2. RFP Design-Build Teams
Originator: Contracts and Grants Administrator	Turnagain Marine Construction. 4. Scoring Sheet and Summaries of selection process
Originating Department: Public Facilities	5. Proposed Contract
Date Submitted: 8/17/2022	

Full Title/Motion:

Motion: Adopt Resolution 22-09-996

Administrative Recommendation:

This resolution is recommended by the Director of Public Facilities.

Fiscal Impact:											
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets								
\$ 310,000.00	\$ See below	\$ 0	Reduced maintenance costs								

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

· · · · · · · · · · · · · · · · · · ·	
Objective 2B, Pages 56-57	Consiste

ent: Yes ΠNo

Summary Statement:

The Haines Borough is pursuing a Progressive Design Build delivery method for the Lutak Dock Replacement Project. The Progressive Design Build process includes two contracts, one with the Owner's Advisor, and the second with the chosen Design Build team. The Design Build Team performs the project in two phases; Phase 1 - permitting and design of the project; and Phase 2 for the GMP for construction of the project. The Borough solicited RFP's from interested firms and received three qualified responses from Pacific Pile & Marine; Western Marine, and Turnagain Marine Construction. Turnagain Marine Construction was the highest scorer during the selection process of qualifications + cost. Recommend awarding contract for Design-Build to Turnagain Marine Construction for a Phase 1 contract NTE \$310,000.00.

Referral:

Referred to: Recommendation: **Referral Date:**

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/22

Public Hearing Date(s):

Postponed to Date:



HAINES BOROUGH, ALASKA RESOLUTION 22-09-996

A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a contract with Turnagain Marine Construction, for the Progressive Design-Build of the Lutak Dock Replacement Project for an amount not-to-exceed \$310,000.00.

WHEREAS, the Haines Borough was awarded up to \$20,000,000 from the federal government for a FY21 Rebuild American Infrastructure with Sustainability and Equity (RAISE) grant for the Lutak Dock Design and Development Concept; and

WHEREAS, after exploration of design options, the Haines Borough learned of the Progressive Design-Build (PDB) option which facilitates involvement of the design-build team during the earliest stages of the owner's project development, ensuring they are part of the project team developing design solutions; and

WHEREAS, the Borough Attorney has evaluated the PDB approach and determined that it is consistent with both Haines Borough Code and federal procurement requirements; and

WHEREAS, the PDB option emphasizes collaboration with the designer and the contractor working directly with the owner; and

WHEREAS, while a project design is usually 35 percent complete by the time a design-builder is procured in the traditional two-step design-build process, PDB adds the design-builder to the owner's team (including the public) even earlier in the design phase; and

WHEREAS, when the design is approximately 50-75 percent complete, the designbuilder issues a Guaranteed Maximum Price (GMP) which allows for the identification and mitigation of risk earlier in the design phase resulting in a GMP that reflects the actual cost to construct the project; and

WHEREAS, in the PDB option, there are two phases of the Design-Build agreement, one for Phase I to perform permitting and design of the project, and Phase 2 to agree on the Guaranteed Maximum Price (GMP) for the construction of the dock; the Design-Build team is selected based upon qualifications in addition to cost; and

WHEREAS, the Haines Borough requested qualifications (RFQ) from qualified and licensed firms to perform the work for the Design-Build Team of the Lutak Dock using the PDB option; and

WHEREAS, the borough received three responses from firms with professional qualifications, Pacific Pile and Marine, Western Marine, and Turnagain Marine Construction; and

WHEREAS, the borough then issued a request for proposals (RFP) and set up interviews with each of the three responders to the RFQ; and

WHEREAS, the selection committee scored each of the proposals and Turnagain Marine Construction received the highest scores; and

HAINES BOROUGH, ALASKA RESOLUTION No. 22-09-996 Page 2 of 2

WHEREAS, the Design-Builder funds can be paid from the grant funds provided,

NOW, THEREFORE, BE IT RESOLVED that the Haines Borough Assembly authorizes the Borough Manager to execute a contract with Turnagain Marine Construction for an amount not to exceed \$310,000.00.

Adopted by a duly-constituted quorum of the Haines Borough Assembly on this 24th day of August, 2022.

Douglas Olerud, Mayor

Attest:

Alekka Fullerton, CMC, Borough Clerk



Haines Borough

Request for Proposals ("RFP") Design-Builder

LUTAK DOCK REPLACEMENT

Date of Issue: June 17, 2022

Closing Date and Time: July 15, 2022, 3 pm Alaska Time

Single Point of Contact ("SPC"): Carolann Wooton, Contracts and Grants Administrator

Address: City, State, Zip Phone (voice) E-mail: 103 Third Ave S Haines, AK 99827 907-766-6409 cwooton@haines.ak.us

REQUEST FOR PROPOSALS Haines Borough Lutak Dock Replacement Owner's Advisor

Haines Borough (the "Owner" or "Haines") requests Proposals from the Design-Build Team for the Lutak Dock Replacement Project (the "Project"). Proposers shall submit the DB Proposal and Price Proposal to the Single Point of Contact ("SPC") via Bid Express no later than 3:00 p.m. Alaska Time on the date set forth in the Project Solicitation Schedule.

I. DEFINITIONS

In addition to the definitions set forth in the RFQ and any addenda issued thereto and the definitions set forth in the Design-Build Agreement, the following supplemental definitions shall apply:

"Initial Basis of Design Documents" means the initial requirements set forth by the Owner and that are attached to the Design-Build Agreement.

"Project Goals" means the following:

- 1) **Design and Construct a Dock that Maximizes the Program Requirements within the Limited Budget.** The Design-Build Team will leverage the efficiencies of the progressive design-build process through innovative and lean design and construction techniques that provide an efficient and effective design with the most scope and programming within the Owner's established budget. The design will also optimize the efficiency of operations and reduce long term maintenance.
- 2) **Execute a successful, collaborative Progressive Design-Build (PDB) Process to produce the envisioned project:** The Design-Build team will develop and utilize a collaborative relationship between the Owner, its stakeholders, and the Design-Build Team to exceed the Project Goals within the Owner's budget and schedule and demonstrating exemplary design and project management.
- 3) **Efficient Pricing and Schedule**. The Design-Build Team will provide transparent pricing and scheduling that allows the Owner to track design and construction concurrently as well as fast track design and construction to maximize the Owner's budget within the Project Schedule.
- 4) **Comply with Legal Requirements.** The Design-Build Team will understand and comply with all applicable State and Federal Legal Requirements.
- 5) **Design for Safety**. The Design-Build Team will create a design that enhances the safety of the project. The design and construction process will reduce re-work and interference with operations with a goal of no recordable incidents.

"Projects of Similar Scope and Complexity" mean projects where one or more of the following characteristics are present. Owner determines at its sole discretion whether a project is of similar scope and complexity.

- 1) Projects of a similar size and budget that include design and construction of large dock facilities;
- Projects that utilize an integrated delivery method that require strong coordination and integration of the design and construction professionals and early involvement of the construction professionals during design;
- Projects where the Design-Builder was selected prior to the establishment of the scope, schedule and GMP where the Design-Builder collaborated with the Owner to develop the final scope, schedule GMP;
- 4) Projects with a limited budget where an owner 's goal is to maximize the available scope within the budget.

II. INITIAL BASIS OF DESIGN DOCUMENTS

The Initial Basis of Design Documents are set forth in Attachment B to this RFP. For the purposes of

establishing prices in the Price Proposal, Finalists may rely on the information set forth in the Initial Basis of Design Documents. However, the Design-Builder will be required to validate the information set forth in the Initial Basis of Design Documents as part of Phase 1 of the Project.

III. RFP PROJECT SOLICITATION SCHEDULE

The following is an estimated procurement schedule. Owner reserves the right to modify the schedule at any time.

Date	Activity
Issue RFP	June 17, 2022
Interactive Meetings with Finalists	Week of June 27, 2022
Last Date to Submit Questions and Proposed	July 1, 2022
Changes to Contract	
Last Date to Issue Addenda	July 8, 2022
Proposal Due Date	July 15, 2022
Intent to Award Notice	Week of July 25, 2022

IV. RFP PROCUREMENT PROCESS

To be responsive to the RFP, Finalists will participate in the following elements of the RFP Procurement Process:

A. Interactive Meeting with Finalists

The Owner will conduct a proprietary Interactive Meeting with each Finalist individually prior to the submission of the Proposals. The Interactive Meetings will provide an opportunity for direct interaction between the Finalist and the Evaluation Committee. The intent of the Interactive Meeting is to evaluate how well each Finalist and its Proposed Design-Build Team understood the project and demonstrate their ability to collaborate with the Owner regarding the Project and propose solutions to the Owner to address the Owner's concerns. Finalists should consider this meeting to be the initial project meeting with the Owner and be prepared to interact with the Owner as if they were selected on the Project. Finalists should be prepared to specifically address the Project Goals and the Design-Build Team's plan to exceed the Project Goals. Specifically, Finalists should discuss the following issues:

- 1) The three biggest risks that they foresee on the Project;
- 2) How they will incorporate the input of the Stakeholders into the Project; and
- 3) Possible innovations in the project, including but not limited to innovation in the design, the sequencing and constructability, or the schedule.

Finalists will be evaluated on their ability to explain their experience and knowledge in the delivery method, effectively communicate and collaborate with Owner Staff, and provide achievable and collaborative solutions to address the Owner concerns. Interactive Meetings are anticipated to last for 2 hours.

The Interactive Meetings will be scheduled with the Finalists. Two business days prior to the date of the scheduled Interactive Meeting, Finalists should provide to the Owner an agenda for the Interactive Meeting. The Interactive Meeting will take place on a virtual platform of the Finalists' choice. Finalists shall provide connection information for the Interactive Meeting with the agenda.

Only Key Team Members who are expected to perform substantial work on the Project should attend the Interactive Meeting, with a maximum number of 10 people, Other Key Team Members may be available by or consulted during the Interactive Meeting.

B. Management Proposals

Finalists shall submit Management Proposals pursuant to the documentation requirements set forth below. The Management Proposal should focus on the Project, the Project Goals set forth above, the concerns expressed by the Owner, and the management solutions proposed by the Finalist.

C. Requests for Clarification and Proposed Changes to Contract Documents

By the date set forth in the solicitation schedule, Finalists may submit a request for clarification to the RFP and/or suggest a list of any changes proposed in the insurance requirements, bonding requirements, Design-Build Agreement, or its attachments. With every request for clarification or proposed change, Finalists must include the following information:

- a. The document and section number;
- b. Proposed alternate language;
- c. An explanation for the requested change; and
- d. Any price implication of the requested change.

The Owner, at its sole discretion, may issue addenda with a clarification or reflecting any accepted changes. The Owner reserves the right to reject any and all proposed changes and to accept any proposed change to the Contract Documents via Addendum to the RFP. The Owner also reserves the right to negotiate such provisions with the selected Finalist.

D. Price Proposals

The Finalists will submit their Price Proposals pursuant to the Solicitation Schedule and according to the instructions in Attachment A. Price Proposals will include any addenda issued by the Owner. Finalists should be prepared to include the terms of the Final Price Proposals in the Design-Build Agreement if the Finalist is determined to be the highest scored Finalist by the Owner.

E. Substitution of Team Members.

Consultants, sub-consultants, subcontractors, and individual Key Team Members included by the Design-Builder in either the SOQ or the Management Proposal (collectively "Team Members"), will be used as a basis for selection. Substitution of Team Members at any time during the solicitation process and in the performance of the work will not be allowed without written authorization from the Owner, which shall not be unreasonably withheld. Proposers and Finalists must submit the qualifications information of all proposed substituted Team Members to the Owner. Even with written authorization from the Owner, a change to any submitted Team Member will result in re-evaluation and may result in a change to the evaluation and ranking of the Proposer. If a Finalist proposes to substitute a Team Member, the Finalist must provide notification and the substituted Team Member's qualifications and resume and seek the Owner's authorization as soon as practicable. The Owner will re-evaluate the Finalist with the new information. Resumes must not exceed 1 page.

F. Evaluation

Finalists' Management and Price Proposals will be evaluated pursuant to the criteria and standards set forth below. In assigning points, Owner is not limited to the information in the Management Proposal and reserves the right to consider information from any source, including but not limited to

the Statements of Qualifications previously submitted, the Interactive Meetings, and references.

Response to RFP		
	Overall Management Approach	30 points
	Project Controls, Cost Tracking and GMP Development	20 points
	Design Development	15 points
	Construction Management, Sequencing and Scheduling	15 points
Price		20 points
Components		
Total Points		100 points

V. DOCUMENTATION REQUIREMENTS

A. Submission of Documents

- 1. Owner is requiring electronic submissions for this procurement. Hard copy submittals will not be accepted. Proposals must be submitted to Owner via Bid Express no later than 3 pm on the date set forth in the schedule.
- 2. Proposers shall follow the instructions and provide the submittals as set forth in Bid Express.
- 3. The Management Proposal shall be provided as an electronically searchable PDF with bookmarks for each section of the Management Proposal. File sizes shall be limited to 20MB.
- 4. Proposers are responsible for ensuring timely delivery of submittals. Owner is not responsible for technical difficulties in submitting electronically. Owner reserves the right not to consider late submittals.
- 5. All submissions must be made in compliance with the instructions provided to the prospective proposers. The Owner reserves the right to reject any submissions that are not in compliance with the RFP and/or redact those portions of the submissions that are not in compliance and not evaluate non-compliant sections.
- 6. Price Proposals must be submitted via Bid Express through the applicable upload.
- 7. Submissions must use a minimum of 10 pt type. A "page" shall be defined as (when printed in hard copy) one single-sided piece of 8.5 x 11-inch paper that has words, charts, tables, pictures, or graphics.
- 8. With the exception of the Identification of Projects Table, pages larger than 8 1/2 x 11 inches will <u>not</u> be accepted. Any materials received that do not comply with the required format will be removed from the Proposers proposal prior to being given to the evaluation committee for review.
- 9. The body of the Management Proposal shall be organized in accordance with the Evaluation Criteria set forth in the RFP. The Management Proposal shall be no longer than 25 pages. The only documentation that is not included in the page limit is the following:
 - a. Cover letter;
 - b. Identification of Projects Table; and
 - c. Divider tabs and/or cover pages, provided that they contain no substantive content.

- 10. All materials submitted will become the property of the Owner.
- 11. Owner reserves the right to reject any or all submittals if the Owner deems it to be in its best interests, or to reject any or all proposers who fail to satisfy qualification requirements or fail to meet standards of responsibility, or submission dates and times.
- 12. No compensation will be made by the Owner for submission of Management Proposals.
- 13. All materials submitted will become the property of the Owner.
- 14. Materials submitted by Proposers may be subject to Alaska public records laws.

B. Reservation of Rights

The Owner reserves without limitation and may exercise at its sole discretion, the following rights and conditions with regard to this solicitation process:

- 1. To cancel the solicitation process and reject any and all SOQs and/or proposals;
- 2. To waive any immaterial informality or irregularity;
- 3. To revise the solicitation documents and schedule via an addendum;
- 4. To reject any Proposer that submits an incomplete or inadequate response or is not responsive to the requirements of the RFP;
- 5. To reduce the number of pages in the Proposals to the maximum allowed number of pages;
- 6. To provide clarifications or conduct discussions, at any time, with one or more Proposers;
- To contact references that are not listed in the Proposer's SOQs and/or Proposals and investigate statements on the SOQs and Proposals and/or the qualifications of the Proposer or Finalists and any firms or individuals identified in the SOQ and/or Proposals;
- 8. To consider the claims history of any Proposer or Finalist as part of the evaluation of the Proposer or Finalist;
- 9. To negotiate the final Owner's Project Requirements and/or contract documents with the highest scored Finalist; and
- 10. To take any action affecting the RFQ process, the RFP process, or the Project that is determined to be in the Owner's best interests.

VI. RFP SUBMITTAL INFORMATION

A. Cover Page (Not scored)

The Proposal must include a cover letter that includes the following: (1) name, address, telephone number, and e-mail address for each Proposed Design-Build Team Member that has been added to the Proposed Design-Build Team, including but not limited to Key Team Members, since the submission of the SOQ and (2) any requested changes to the Proposed Design-Build Team. The cover letter shall be a maximum of two (2) pages.

B. Management Proposal Contents and Organization

The Management Proposal may not be longer than twenty (25) pages. Finalists should focus their discussions in the Management Proposal on their approach to the Project

1. Overall Management Approach

The Owner is seeking a Design-Build Team that will exceed Project Goal Number 2 in developing a collaborative Project Team, incorporating the Owner Staff and all Owner sub consultants, including but not limited to the Owner's Representative.

- a. Describe the Finalist's overall management approach to the Project. In responding to this evaluation factor,
- b. Keeping Project Goal Number 2 in mind:
 - i. Explain the Design-Build Team's approach to creating a collaborative environment for the Project and exceed Project Goal number 2. Include an explanation of where the design-build team will be located during the various phases of the Project.
 - ii. Describe the Design-Build Team's approach for outreach to project Stakeholders and incorporating their input into the project.

2. Maximize Design Within Limited Budget

The Owner is seeking a Design-Build Team that will exceed Project Goal Number 1 in developing an efficient and effective design within the Owner's established budget.

- a. Describe the Design-Build Team's overall approach to exceeding Project Goal Number 1.
- b. Describe specific strategies and design ideas for exceeding Project Goal Number 1. Include in the discussion the following topics:
 - i. Ideas for creating spaces that will have flexible use over time;
 - ii. Incorporating Stakeholder input into the design; and
 - ii. Innovative constructability solutions that could reduce the overall budget.
- c. Identify the challenges in developing the design for the Project and explain how the Design-Build Team will address those challenges.
- d. Explain how the Design-Build Team will communicate and collaborate with Owner Staff as well as the various stakeholders and ultimately integrate their input into the design of the Project.

3. Project Controls, Cost Tracking and GMP Development

The Owner is selecting the Design-Builder before the Scope of Work for the Project is finalized. The Owner expects a collaborative process with the Design-Build Team to develop the final project scope and the GMP. The Owner is seeking a Design-Build Team that will exceed Project Goal Number 3 and create transparent pricing that takes advantage of the efficiencies of progressive design-build. Explain the Design-Build Team's strategies to exceed Project Goal Number 3, including but not limited to the following

- a. Describe three strategies for exceeding Project Goal Number 3.
- b. Describe the Design-Builder's processes and tools for monitoring, reporting and managing cost, including but not limited to:
 - i. Design to budget control and reporting processes, including the software that the Design-Builder will use to monitor and communicate the project costs to the Owner.

- ii. Scope, cost, and schedule baseline development and management/change control processes and the participation and interaction among the scheduling and estimating teams, project, design, construction and operations management teams to execute these processes.
- iii. Incorporating input from design-build or other sub-contractors;
- iv. The primary challenges in establishing the GMP; and
- v. The differentiating resources of the Proposed Design-Build Team that will meet the challenges of establishing the GMP.
- b. Phase 1 Level of Effort. Provide a proposed Level of Effort for the Phase 1 set forth in Section 6.6.1 of the Agreement("Phase 1 LOE"). The proposed Phase 1 LOE should provide the following detail:
 - i. Describe in detail the tasks the Design-Build Team intends to perform during the Phase 1 including the number of hours anticipated for each task;
 - ii. Identify the Key Team Members who will be performing the tasks
 - iii. The Phase 1 LOE should encompass the tasks required for the Phase 1 Scope of Work that are set forth in Section 6.6.1 and Exhibit C of the Agreement. Pursuant to the Agreement, the Design-Builder will be bound to the hourly rates proposed and submitted in its Price Proposal. The Phase 1 LOE will be scored as part of the Management Proposal.
- c. Provide examples of deliverables the proposed Design-Build Team will use to communicate the development of the project costs and project schedule to the Owner.

4. Construction Management, Sequencing, and Scheduling

The Owner is seeking a Design-Build Team that will in developing a design and construction schedule that maximizes efficiency and minimizes Contract Time while maintaining a safe workplace and meet the project sustainability requirements. Explain the Design-Build Team's strategies to meeting these goals, including but not limited to the following:

- a. Describe the Design-Build Team's specific plan with respect to using construction means and methods and the progressive design-build approach to achieve efficiencies in scheduling and construction sequencing for the Project. Provide a single page, high level, achievable proposed schedule for the Project that strives to achieve the goal of completing the project as quickly as possible.
- b. Describe the Design-Build Team's approach achieving the performance requirements and optimizing the quality of the project. Include a discussion of a specific approach to quality assurance/quality control, including testing and commissioning of the Project.
- c. Describe the Design-Build Team's approach exceeding Project Goal Number 5 to maximize safety during the Work.
- d. Identify the challenges in the topics noted above and explain how the Design-Build Team will address those challenges.
- d. Provide details regarding the tools used in this process and how those tools will assist the Design-Builder exceeding the Project Goals.

VII. IDENTIFICATION OF PROJECTS (NOT SCORED)

- A. The Finalist must submit an Identification of Projects Table with the required information set forth herein for all projects cited or mentioned in the Management Proposal *that were not listed* in the Identification of Projects Table that was provided with the Finalists' SOQ. The Identification of Projects Table may be submitted on 8.5" x 14" paper and may be no more than two pages in length. The Finalist is responsible for ensuring that contact information contained in their Identification of Projects is correct. The inability to contact a reference may have a detrimental impact on the evaluating qualifications. The Owner reserves the right to contact any person listed in the Identification of Projects or any other person with knowledge regarding any Project in which any Design-Build Team Member or Key Team Member participated.
 - a. Name of project;
 - b. Owner/Customer;
 - c. Location of project (include address);
 - d. Description of the delivery method and integration of design and construction and identify the firm(s) role as a prime consultant, subconsultant, contractor, subcontractor or other;
 - e. Project description and applicability and relevance of the referenced project to the evaluation criteria Project.
 - f. Name of each Key Team Member who is proposed for this contract who played a significant role on the project example, including a description of their project responsibilities and functions;
 - g. The initial contract price, the final contract price, and an explanation for any difference between the two amounts;
 - h. The initial date scheduled for substantial completion, the actual date of completion, and an explanation for any difference between the two dates;
 - i. Number of recordable injuries; and
 - j. Project contact of the owner or customer (current address, e-mail, and phone number) who can verify the characteristics of the submitted project example.
- 2. The identification of projects will not be evaluated separately. Rather, the projects will be evaluated in the context of the criteria in which the project is cited.

VIII. PRICE PROPOSAL CONTENTS

A. Design-Builder's Fee Percentage

Finalists shall submit a Price Proposal that provides the proposed Design-Builder's Fee Percentage that will be included in Section 6.2.1 of the Progressive Design-Build Agreement between Owner and Designer, attached hereto as Attachment 2. For scoring purposes, the Design-Builder's Fee Percentage shall be multiplied by the estimated Cost of the Work for the Project (\$_____). Note that the estimated Cost of the Work is provided for scoring purposes only and does not guarantee a minimum amount of Work.

B. Phase 1 Not to Exceed Amount

1. Provide the Proposed Phase 1 Not to Exceed Amount that will be inserted into Section 6.6.1.1 of the Design-Build Agreement and, if accepted by the Owner after negotiations, shall become binding on the successful Finalist, subject to the terms and conditions of the

Contract Documents.

- a. The Proposed Phase 1 Not to Exceed Amount should include all compensation to the Design-Builder during the Phase 1 set forth in the Agreement as proposed in the Phase 1 Level of Effort described in the Management Proposal.
- b. The Owner reserves the right to reconcile the various proposals received and also reserves the right to seek best and final proposals for the scope and the cost of the Phase 1 Services and the Phase 1 Not to Exceed Amount; however, by submitting the Phase 1 Not to Exceed Amount, the Finalist warrants the following:
 - i. That the Phase 1 Level of Effort described in the Management Proposal is sufficient for the Design Build Team to perform the Work described in Exhibit C of the Agreement and provide the Owner with a Phase 2 Report.
 - ii. That the Phase 1 Not to Exceed Amount set forth in the Management Proposal is sufficient to perform the Work described in the Phase 1 Level of Effort.
- c. The Not to Exceed Amount will not be scored. The Owner reserves the right to negotiate both the Proposed Scope and Not to Exceed Amount with the apparent successful Finalist.
- 2. Provide the Key Team Members Hourly Rates. The Hourly Rates will not be scored. Rather, they will be in Exhibit D to the Agreement.

C. Scoring of Price Proposal

The Design-Builder's Fee Percentage shall be scored as follows:

Price Element	Estimated Cost of the Work of the Project.	Price Proposal
a. Design-Builder's Fee Percentage	X \$20,000,000	\$

The Finalist with the lowest Price Proposal will receive all fifteen points. The remaining Finalists will receive a proportionate share of the fifteen points, based on the proportion that the Price Proposal for their proposals exceeds the lowest Price Proposal. The points will be rounded to the next lowest whole number. No partial points will be awarded By way of example, if the second low Finalist proposes a Price Proposal that is fourteen percent higher than the lowest Price Proposal, the second low Finalist shall receive 17 of the 20 allotted points. Fourteen percent of 20 is 2.8. 20 minus 2.8 equals 17.2. 17 is the next lowest whole number.

IX. LIST OF ATTACHMENTS TO RFP

- A. Price Proposal Form and Instructions
- B. Design-Build Agreement and General Conditions of Contract

Exhibit A Design-Builder's Insurance

Exhibit B-1 Form of Payment Bond

Exhibit B-2 Form of Performance Bond

Exhibit C Phase 1 and 2 Scope of Services

Exhibit D Owner's Program/Initial Basis of Design Documents

Exhibit F-1 Phase 1 Change Order Form Exhibit F-2 Phase 2 Change Order Form Exhibit G Form Phase 2 Amendment

X. REQUESTS FOR CLARIFICATION AND CHANGES

- A. Request for Clarification:
 - 1. Owner will respond to each properly submitted written request for clarification.
 - 2. All questions about the meaning or intent of the RFP Documents must be directed to the Owner through Bid Express.
 - 3. Interpretations or clarifications of the RFP considered necessary by Owner in response to such questions will be issued by Addenda.
 - 4. Questions received less than seven (7) calendar days prior to the Proposal due date may not be answered.
- B. Request for Change:
 - 1. Any Proposer may submit a request for changes to the RFP terms or contract. Owner will respond to each properly submitted written request for change of RFP terms. Where appropriate, Owner will issue revisions or clarifications via addenda posted on the Bid Express.
 - 2. To be considered, requests for changes must include the reason for requested changes supported by factual documentation supporting the requested changes.
 - 3. To be considered, the request must be in writing and received by Owner by July 1, 2022 at 2:00 pm.
 - 4. The request for clarification or changes must be submitted through Bid Express.

Lutak Dock Replacement

Request for Proposals ("RFP") Design-Builder









Pacific Pile & Marine, LP 4753 West 80th Ave Anchorage, AK 99502 Chris Willis P 206.331.3873 chrisw@pacificpile.com

COLE

Management Proposal Technical Submission - July 15, 2022

THIS PAGE INTENTIONALLY LEFT BLANK FOR DUPLEX PRINTING



Pacific Pile & Marine, LP 4753 West 80th Ave Anchorage, AK 99502 T: 907 276-3873 F: 907 278-0306 www.pacificpile.com



COVER LETTER

July 15, 2022

Haines Borough Public Facilities Office PO Box 1209

Subject: Lutak Dock Replacement - RFP Design-Builder Management Proposal

Attention: Carolann Wooton Contracts & Grants Administrator

Ms. Wooton and the Haines Borough Review Team:

Pacific Pile & Marine (PPM) and our design partner, PND Engineers (PND) are thrilled to submit our response for the Haines Borough's Lutak Dock Replacement Project. PPM and PND, referred to herein collectively as the PPM/PND Progressive Design-Build Team (Team), have closely collaborated on numerous projects over the past 25 years and bring a deep understanding of the Progressive Design-Build process that will allow us to execute this project safely and efficiently. Our Team is intimately familiar with the project and site and have successfully delivered critical waterfront infrastructure projects similar in nature throughout Alaska over the past 30 years.

PPM has decades of experience installing and removing sheet pile circular cells and bulkheads and pile-supported piers in remote areas of Alaska. PPM has successfully delivered more than \$300M in alternate delivery projects over the past 10 years as well as delivered more than \$400M of Alaska marine and highway infrastructure projects during that period. PND brings more than 40 years of expertise and experience successfully designing thousands of marine projects, including several projects in Haines.

The Progressive Design-Build delivery model will allow our Team to leverage our extensive experience to provide innovative solutions and collaborate closely with all stakeholders to meet the Project's goals. Our Team understands the importance of producing an effective design within budget while meeting the permit and schedule restraints. We believe in a "Project-First" approach and are committed to open and honest communication. Our Team is committed to designing and constructing a dock that maximizes safety, reduces interference with on-going operations, creates value for all stakeholders involved, and meets or exceeds the Project's goals:

- Design and Construct a Dock that maximizes the Program Requirements within the Limited Budget
- Execute a successful, collaborative Progressive Design-Build process to produce the envisioned project
- Provide efficient pricing and schedule
- Comply with Legal Requirements
- Design and Construct for Safety

Our Pre-Construction Team will be led by Chris Willis supported by Chris Lundfeldt, John Demuth and Dick Somerville. Chris Willis was heavily involved in estimating and pre-construction services for the recently completed \$30M CM/GC Palmer Pier Replacement in Antarctica as well as the \$150M CM/GC Seattle Multimodal Terminal at Colman Dock, an on-going marine heavy civil marine project involving the replacement of the existing structure with a pile-supported, concrete pier. Both projects required extensive front-end pre-construction services including constructability reviews and producing 35%, 65% and 95% open book estimates as well as the final cost proposal.

Chris Lundfelt was directly involved in both the pre-construction and construction phases of the Palmer Station Pier Replacement. With over 30 years of experience constructing similar projects, Chris was instrumental in providing value engineering and constructability



Pacific Pile & Marine, LP 4753 West 80th Ave Anchorage, AK 99502 T: 907 276-3873 F: 907 278-0306 www.pacificpile.com



input at each stage. John DeMuth also brings significant expertise with over 33 years of experience in engineering development, design, and construction management of waterfront facilities in Alaska. Dick Somerville further bolsters John's expertise with over 40 years of civil engineering and project management experience in with an emphasis on planning, permitting, site investigations, design, construction inspection and contract administration in Alaska.

PPM requests two changes to the proposed Team for this project. Aaron Athanas will replace Randy Downing as the Mechanical Design Lead. Aaron has over 20 years of experience in mechanical engineering experience in the Alaska region and extensive knowledge of the arctic environment. Aaron will be responsible for mechanical engineering providing demolition plans, fuel system design, and coordination with civil and electrical designers as necessary. Aaron's resume is included in this proposal.

- Name: Aaron Athanas, P.E.
- Address: Great Northern Engineering 137 E. Artic Ave, Palmer, Alaska 99645
- Telephone: (907) 306-0449
- Email: aathanas@gne-ak.com

Torsten Mayrberger will be added to the Team as the Geotechnical Lead. Torsten has been working in Alaska for more than 35 years, translating to a deep knowledge of the conditions and challenges presented throughout the state. Mr. Mayrberger has 18 years of geotechnical engineering experience involving large, remote, arctic, and marine geotechnical investigations, as well as deep foundation design in non-permafrost and permafrost soils, marine environments, and rock mass structures. Torsten will oversee the project's geotechnical investigations, analysis, and reporting for the design team. Torsten's resume is included in this proposal.

- Name: Torsten Mayrberger
- Address: PND Engineers 1506 W 36th Ave, Anchorage, Alaska 99503
- Telephone: (907) 561-1011
- Email: torsten@pndengineers.com

Our Team has the expertise and experience necessary to exceed the Project's goals and will bring exceptional value to both the design and execution of this work. Utilizing our Team's knowledge of the work and experience with the Progressive Design-Build process, we will be able to readily identify, mitigate, and manage risk at every phase of this project.

Should you have any questions or concerns, please contact the undersigned below at (206) 331-3873.

Respectfully,

Que

Chris Willis 206.331.3873 chrisw@pacificpile.com



Contents

OVERALL MANAGEMENT APPROACH p. 1

	1.a Management Approach to the Project	p. 1
	1.b.i Approach to Creating a Collaborative	
	Environment and Exceed Project Goal #2	p. 2
	1.b.ii Approach for Outreach to Stakeholders	
-	and incorporating their input	p. 2
/	MAXIMIZE DESIGN WITHIN LIMITED	
	BODGET	p. 3
	2.a Approach to Exceeding Project Goal #1	p. 3
	2.b Strategies and Design Ideas for Exceeding	
	Project Goal #1	p. 3
	2.c Challenges in Developing the Design	p. 5
	2.d Communication and Collaboration with	
	Owner Staff and Stakeholders	p. 5
PROJECT CONTROLS, COST TRACKING		
2	AND GMP DEVELOPMENT	p. 5
	3 a Three Strategies for Exceeding Project Goal	
	#3	p. 5
	3.b Processes and Tools for Monitoring, Reporting and	
	Managing Cost	p. 6
	3.c Phase 1 Level of Effort	p. 6
	3.c Phase 1 Level of Effort3.d Examples of Deliverables to Communicate the	p. 6
	3.c Phase 1 Level of Effort3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule	р. 6 р. 8
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, 	p. 6 p. 8
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING 	р. 6 р. 8 р. 8
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING 4.a Means and Methods to Achieve Efficiencies in 	p. 6 p. 8 p. 8
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING 4.a Means and Methods to Achieve Efficiencies in Scheduling and Construction Sequencing 	p. 6 p. 8 p. 8 p. 8
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING 4.a Means and Methods to Achieve Efficiencies in Scheduling and Construction Sequencing 4.b Approach for Achieving the Performance 	p. 6 p. 8 p. 8 p. 8
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING 4.a Means and Methods to Achieve Efficiencies in Scheduling and Construction Sequencing 4.b Approach for Achieving the Performance Requirements and Optimizing for Quality 	p. 6 p. 8 p. 8 p. 8 p. 9
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING 4.a Means and Methods to Achieve Efficiencies in Scheduling and Construction Sequencing 4.b Approach for Achieving the Performance Requirements and Optimizing for Quality 4.c Approach to exceed Project Goal #5 to Maximize Safety During the Work 	p. 6 p. 8 p. 8 p. 8 p. 9
	 3.c Phase 1 Level of Effort 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING 4.a Means and Methods to Achieve Efficiencies in Scheduling and Construction Sequencing 4.b Approach for Achieving the Performance Requirements and Optimizing for Quality 4.c Approach to exceed Project Goal #5 to Maximize Safety During the Work 4.d Identify the Challenges and How the Term will 	p. 6 p. 8 p. 8 p. 8 p. 9 p. 11

will Assist to Exceed the Goals

5

p. 14
p. 18
p. 22
N/A

p. 12

4.e Detail the Tools Used in this Process and How they

THIS PAGE INTENTIONALLY LEFT BLANK FOR DUPLEX PRINTING

1 OVERALL MANAGEMENT APPROACH

1.a Management Approach to the Project

Our Team will work collaboratively with the Haines Borough (Borough) and Project Stakeholders to progress the design and develop the Final Basis of Design (BOD) documents, Project Schedule, and Guaranteed Maximum Price (GMP) within the Project Budget.

To accomplish this, we will proceed as follows:

- Start Up Meeting: Following award, our Team will ask the Borough to set up a Start Up meeting at Site. Despite the capabilities of teleconferencing and other technologies that make successful interfacing remotely possible, it is important for the Project Team (inclusive of the Borough, Stakeholders and PD/B Team) to meet in-person in Haines and discuss the tasks ahead. This is expected to be a two-day meeting. The meeting agenda will include items to discuss the design concepts currently envisaged, the schedule, listen to Borough and Stakeholder input and agree on the timetable for key elements of the Phase 1 Schedule.
- Weekly Meetings: The Project Team will establish a time for a weekly meeting and establish an initial agenda for the recurring weekly meeting. This weekly meeting will be where the Project Team collaborates on the progress to date, any challenges and issues that have been identified, and make decision to progress the work to achieve the Project Milestones agree upon.
- 35%, 65%, 95% Design/Cost Milestone Review Meetings: The Project Team will review the initial Phase 1 design schedule and agree upon the post milestone review meetings. These one to two day meetings are usually conducted at Site and review the design progress (drawings and specification development), the estimated cost of the work to date (D/B cost estimate and risk analysis), and the main challenges and issues that have been identified.
- Working Sessions: During the Phase 1 design and cost development, there will be items that require more input than can usually be accomplished during the weekly meetings. For instance, permitting issues will be crucial to this project and our Team will set up working sessions with the necessary members of the Borough and Project Stakeholders to more fully discuss and troubleshoot these tasks.

 Clearly Defined Roles and Responsibilities: Within the PPM/PND Team, specific responsibilities have been established for the Phase 1 goals.



Chris Willis shall act as the Team's Project Director. The design and construction team will report to him. He will be responsible for directing the Team to produce the Final BOD documents, Project Schedule, and GMP within the Project Budget. He will also manage the estimating team

for producing the milestone estimates and be the contract authority for negotiation and finalizing the GMP.



Dick Somerville shall act as the Team's Engineering Quality Control Manager and is tasked with ensuring the design and construction elements will be done to the highest quality to meet the Project's goals.



John DeMuth shall act as the Design Manger and will manage the production of the drawings and specifications through the initial conceptual phase leading to the Final BOD. The PND structural, geotechnical, and permitting engineers will report to John.



Stewart Willis will be the Project Manager for the Construction Team working with the Superintendent, Chris Lundfelt, to develop the means and methods for constructing the project and developing the Project Schedule with the design, estimating, and scheduling team.

All of the Management team listed above will attend the Start Up meeting, the weekly meetings, the milestone reviews, and participate in working sessions as needed.

 Collaboration: Effective collaboration requires a dedicated, intentional approach and needs to be managed. During the Phase 1 development period, many challenges may be identified that will require problem solving by the entire Project Team. Our Team will be instrumental in identifying these challenges and providing professional, clear, and accurate information throughout the process so that decision-makers have the necessary information to make the best fiscal and operational decisions for the Project. We are confident that our extensive experience on previous progressive design-build projects and the quality of the team we have assembled will enable us to excel at accomplishing this task.

1.b.i Approach to Creating a Collaborative Environment and Exceed Project Goal #2

The delivery of quality and value to the Haines Borough is our Team's guiding principle. To achieve that goal, it is paramount that we develop and utilize a collaborative relationship with the Borough and Stakeholders in an effort to exceed the Project goals.

Our Team has extensive experience working hand-in-hand with communities throughout Alaska and are very aware of the inherent desire for community members to roll-up their sleeves and get involved in the development of a local project. For this project in particular, our Team recognizes the importance of having a safe, functional, cost-effective port facility to accommodate a variety of vessel operations to support consumers and industrial activities throughout the region for decades to come.

The process to accomplish this begins by building on the successful relationship our Team has already established over the past 30-40 years with the community of Haines. We believe early, open and honest communication is essential for overall Project success. We will work diligently to facilitate a teaming approach for identifying potential risks, discussing options to mitigate them, and resolving any hurdles that may arise during the project design and construction. Our Team's preference will be to launch this project by collaborating in an in-person Start Up meeting with the Borough and Stakeholders, complemented with a thorough Site visit. Alternatively, at the Borough's discretion, we can facilitate this initial step as a video conferencing and select alternative means to capture the necessary data points. The objective will be to introduce individual members of our Team, foster an open dialogue about the Project goals and challenges and the Project Team's abilities to meet those goals and challenges. We intend to wrap-up the meeting by discussing immediate and short-term goals in an effort to expedite the development of this essential, long-awaited project.

The RFP specifically requires that our Team validate the "Initial Basis of Design" which primarily consists of the concept drawings included in the RFP documents developed from Stakeholder and community input thus far. Our Team has already spent considerable time reviewing the documents and identifying key elements critical to the success of the project. Our Team will coordinate with the Borough to identify and discuss potential risks, regulatory and funding requirements, existing site information and the potential for obtaining additional site information in an effort to mitigate the risks so we can provide the Borough with a solid, vetted design and accurate pricing for project costs and early procurement of materials.

1.b.ii Approach for Outreach to Stakeholders and Incorporating their Input

As part of this process, we will further engage Stakeholders and the community by establishing a Project-specific website that is regularly updated to serve as an avenue for Stakeholders and the public to be kept informed with regard to meetings, schedule, design, and progress.

We will also coordinate with the Borough to schedule and conduct workshops that will serve to engage the interests of participants and maximize opportunities for discussion through an efficient, interactive format – either in person or with video conferencing. As part of this process, we will collaborate with the Borough to develop detailed questionnaires that solicit the specific information necessary to confirm user group operations, operational constraints and/or requirements during construction and additionally serve to assist in developing the Final BOD documents for a successful project that will function and be constructed as the community expects.

Input from the workshops and responses to the questionnaires will be summarized, and our Team will work with the Borough to review and prioritize the results to maximize the Project goals and determine what will be incorporated into the final design scope of work. This strategic interaction with the community will serve to expedite the design development as necessary to facilitate the permitting process.

The teamwork employed to expedite the design development will function well to engage the Federal Agencies as early as possible through preliminary consultation where informed discussions can be initiated regarding design and construction of the project. This is critical to advance the project as quickly as possible as the agencies will acquire familiarity with the project and the permitting process will be accelerated due to many typical questions and concerns being addressed during the preliminary consultation.

The public process and stakeholder engagement are crucial to the construction of public infrastructure and should simultaneously advance and even improve the project. However, should there be undue opposition from private interests or regulatory agencies, our Team can consider engaging a lobbyist at the direction of the Borough.

Design submittals at 35%, 65%, 95% and 100% will benefit from comprehensive working sessions with the Borough to collectively discuss and make key decisions regarding design, construction, costs, schedule, quality, durability, operability, functionality, and service life.

Personnel for PND's design team are located primarily in Juneau with assistance from the Anchorage office. Personnel for PPM's construction team are located primarily in Seattle with assistance from their Anchorage office.

2 MAXIMIZE DESIGN WITHIN LIMITED BUDGET

2.a Approach to Exceeding Project Goal #1

- Verification of the Existing R&M Project Concept, Phasing and Project Budget: Task one is to review the existing design concept produced by R&M issued in Addendum #2. This review needs to occur with the Project Team at the Start Up meeting. The verification will involve agreeing to the initial project concept, design phasing, and schedule as outlined and then revising the concept, if required, and producing a 35% set of drawings and specifications. Our Team will then produce a 35% cost estimate based upon this information. This 35% stage is to provide the Borough with the information to make decisions necessary for the next stage of the process to progress. The 35% milestone usually produces a "menu" of concepts and associated costs that the Owner and Stakeholders can refine and focus in on the concept that meets their ultimate operational and fiscal goals.
- Productions of 65% Design and Cost Package: From The 35% conceptual package, the Borough will direct our Team to progress design development upon an agreed upon 35% concept. This is a very important stage as permitting usually begins with an agreed-upon 35% design and often relies upon 65% design to provide critical details for final agency agreement. This requires sufficient accuracy at the 65% design that no major changes will occur to upset the receipt of final permits. At this stage, the cost estimate is becoming more accurate and concentrating on the main cost elements of the project. A risk matrix has been developed identifying the major risks and initially specifying ownership (the Borough, our Team, or shared) and jointly discussing and problem solving to reduce and mitigate. The cost estimate is produced open book and the means and methods, project materials, productivities, subcontract costs and phasing will be discussed in detail with the Project Team. At the 65% review meeting, the Project Team will collaborate on solving any operational or design issues, discuss any further innovations to reduce project costs, and make any decision necessary to keep the project within the established Budget. Following the 65% review meeting, the 65% design package will be amended as necessary and the Borough will instruct the Team to proceed to the 95% design development stage.

- Production of the 95% Design and Cost Package: By this stage, the Project design, planning and cost estimation is nearing completion and the Project Team will collaborate on a final full understanding of the Project plan to enable the Team to proceed to the GMP stage of the process. At this milestone, the main design elements are finalized and the risk matrix has been sufficiently discussed in detail with the allocation of risk agreed to. Any project contingencies have been identified and agreed upon and the terms and conditions of the eventual Phase 2 contract have been initially negotiated between the parties. The Borough will have been given accurate cost information and cash flow forecasts and any final challenges or uncertainties are presumed to have been collaboratively solved by the Project Team.
- Production of the Final Basin of Design, Project Schedule and GMP: This is the final stage where the design is developed to 100% Issued for Construction (IFC) and the Team has negotiated a GMP for the work identifying a mutually agreed upon GMP, project contingency, and schedule for the Work.
- Open, Accurate Cost Development: Fully optimizing the opportunities inherent to the Progressive Design-Build process requires open and honest communication combined with competency and experience. Our approach is to provide that level of service from the start. In line with that level of transparency, we have included the full estimate recap within in the Price Proposal submission.

2.b Strategies and Design Ideas for Exceeding Project Goal #1

- a. Cost-benefit analysis for a variety of bulkhead wall types.
 - i. MSE Wall
 - ii. Combi-Wall
 - iii. Open Cell Wall
- b. Cost-effective and efficient Dolphin Design
 - i. Pile Cap Design
 - Minimize bending moment in piles- reduced pile size required.
 - Incorporate batter pile guides for ease of construction.

- ii. Spin-Fin Pile Tips reduce pile driving time and therefore underwater sound impacts.
- iii. Fender System Design
 - High energy capacity
 - Minimal maintenance and ease of maintenance
- c. Cost-effective and Efficient Launch Ramp Design
 - i. Precast Concrete Planks
 - ii. Sleeper Design anchor planks and maintain spacing
- d. Permitting Innovations
 - i. Efficiencies in permitting timeline resulting from early coordination with agencies regarding potential project impacts and proposed mitigation measures.
 - Utilization of a lobbyist if stakeholder engagement and early agency coordination do not fully remove undue roadblocks to project approvals.
- e. Reuse of Materials
 - i. Recovery of existing armor rock and fill materials from the existing dock for beneficial reuse on the project.
- Materials Procurement
 - i. Early identification of permanent materials will allow for the pre-ordering of the raw materials needed for the project, locking in pricing before the design is finalized.
 - ii. Strategic Partnerships with local materials providers.
 - Colaska (Southeast Road Builders)

The collective experience of our Team with designing and constructing marine and waterfront projects in Alaska and our background of successfully working with the Haines community over several decades will provide the Borough with a proven avenue to make this long-awaited project a successful reality.

Specific design strategies and ideas include the following:

 Utilize our design team's extensive experience in evaluating site conditions and performing efficient, effective stability analyses of large waterfront fill areas and conducting the met-ocean analysis required for correctly sizing armor rock. Design will be efficient and effective. Our Team has more waterfront and marine experience in Northern Lynn Canal and specifically Haines than any other team. Our combined knowledge of geologic and wind/wave conditions in Haines will ensure the Borough receives the best design in terms of suitable use of local materials, seismic stability, durability for operations and wind/wave environment, and ease of construction.

- Apply our design team's acquired experience in design of boat launch ramps with efficient, effective ramp design elements. Our design team has worked with ADFG for many years to develop cost-effective boat launch designs that have been constructed throughout the State. Our design team has more experience with boat launch facilities in Alaska than any other firm in Alaska. The Borough will benefit greatly from this experience.
- Employ our design team's unparalleled knowledge and experience in the design of mooring/breasting dolphins and fendering systems. Our team has designed and constructed dolphins for a wide variety of applications including the cruise industry, oil and gas industry, cargo industry and the logging industry. We understand how best to configure the dolphin piles to minimize imposed loads and incorporate prefabricated elements into the dolphin cap that facilitate construction. PND also sees potential for their proprietary Spin-Fin technology to provide an efficient, cost-effective design that reduces the pile lengths required to potentially eliminate the need for rock anchors when shallow bedrock conditions exist. Additionally, a reduction in pile lengths and minimizing pile loads will ultimately reduce the total amount of sound transmission which the Federal agencies look upon favorably when reviewing the IHA permit application.
 - » We have a catalog of fendering designs that have the capacity, durability, and low maintenance characteristics necessary to provide the Borough with an effective, sustainable system for safely berthing vessels at the facility.
- Effectively use our design team's wealth of experience in design and construction of various bulkhead wall types. Our Team will work with the Borough to conduct costbenefit analysis and evaluate the best option. Criteria will include load capacity, material cost and availability, lowest maintenance costs, durability, service life, permitting considerations and ease of construction.

2.c Challenges in Developing the Design

Primarily due to financial limitations, the current concept drawings included in the RFP represent a facility layout that is significantly simplified compared to past concepts and more closely aligns with the Borough's financial parameters. Given these constraints, our Team considers the following as challenges to developing the final design for this project.

- Finalizing the BOD and establishing a scope of work that incorporates all Stakeholder and public input while also meeting the needs of all user groups within the established budget. Addressing this challenge will require the level of experience and leadership our Team processes to closely collaborate with the Borough to evaluate, prioritize, and incorporate input to the maximum extent possible after careful consideration of costs and budget constraints.
- Obtaining current, accurate site information that includes topographic survey, bathymetry survey, geophysical survey, and strategic bore holes within the established design budget. Our Team will address this challenge through development of an efficient, effective field investigations plan designed to maximize information collection efforts (personnel, equipment, work plan, scheduling, etc.) and minimize costs.
- Impacts of global supply, supply chain, inflation, and fuel prices influences the cost of materials and the costs associated with fabrication, delivery, and construction costs for the project. We will address this challenge by maximizing efficiencies in the design of key, costly project elements so that strategic, significant cost savings can be accomplished. For example, there are many options for design of the dolphins. The load capacity, size, and number of piles required to achieve that load requirement, how to address shallow bedrock in a cost-effective manner, the type of fender system and its construction will all play a role in being able to design for maximum cost savings. Our Team's design-build collaboration capabilities enable us to prioritize specific project elements such as dolphin and approach dock piles and advance the preliminary design of those elements to enable the procurement of steel coils that will eventually be used to fabricate the piles. Finally, our Team will leverage long-standing relationships with steel suppliers, fabricators, and local shot rock and armor stone suppliers in order to further minimize costs and yield the best value for the Borough.

2.d Communication and Collaboration with Owner Staff and Stakeholders

As previously noted, Borough and Stakeholder communication and collaboration will in part be conducted through a series of project meetings and workshops held in-person and/or via an agreed upon telecommunication platform such as Microsoft Teams or Zoom. There will also be a public-facing Project information site for interested parties to remain apprised of approved updates. Stakeholder workshops including questionnaires to compile input will be utilized to capture design and operational feedback to be incorporated as agreed by the Project Team.

Effective communication is inextricably linked to proactively setting expectations and identifying approved channels and preferred mediums to ensure a consistent, transparent flow of information. This is further aided by fostering a shared approach to Partnering and Risk Management. Appropriate Project controls will provide a framework to facilitate the level of communication and collaboration needed to exceed the Project goals. Several of these tools will include:

- Risk Register used to identify and develop potential areas of concern or cost items with less certainty to focus on during pre-construction
- HCSS HeavyBid estimating software used to generate detailed estimates and associated cost reports
- Viewpoint Team project management platform for tracking and progressing items as well as document control
- Microsoft SharePoint alternate project management platform
- Microsoft Teams telecommunications platform for collaboration and web conferencing
- Zoom alternate telecommunications platform

Even with project controls in place, good communication requires a certain diligence and early alignment towards relationship building. Our Team has not only the skills and tools but the desire to effectively integrate the Borough and Stakeholders to allow all parties the opportunity to contribute to the success of this Project.

3 PROJECT CONTROLS, COST TRACKING AND GMP DEVELOPMENT

3.a Three Strategies for Exceeding Project Goal #3

 Strategy #1: Our Team will produce our cost estimate using our established estimating procedure which includes the use of HCSS HeavyBid estimating software. The estimate will be based upon a mutually agreed Work Breakdown structure (WBS) that follows the R&M cost estimate WBS so that comparisons between the 35% Verification cost estimate and the R&M cost estimate included in Addendum #2 can be easily reviewed and any differences investigated and discussed.

 Strategy #2: Our Team's cost estimate will be open book and be sufficiently detailed so that cost elements can be shared with the Owner's team and easily understood and reviewed. All subcontractor and material supply quotes received during the cost estimation process will be copied to the Borough. The cost estimate will be given to the Borough in a timely manner and reviewed in detail at each post milestone review meeting.

 Strategy #3: At the 35% Verification Stage, our Team will establish a risk matrix to identify project risk, identify whom is responsible for each risk contractually, quantify the potential cost and discuss how to eliminate or mitigate the risk item. This risk matrix will be updated at each Design and Cost Package Milestone and the result of elimination and mitigation discussed and agreed. This strategy will result in a clear understanding of the risk issues and lead the way for incorporation into the final GMP.

3.b Processes and Tools for Monitoring, Reporting and Managing Cost

Our Team has a long history of working together on designbuild projects. Both PPM and PND boast an experienced, highly qualified, and communicative staff. Our Team will be comprised of estimators, superintendents, project managers, schedulers, and all appropriate design disciplines.

Our Team will coordinate with targeted subcontractors to assist in the design. Daily coordination and targeted collaborative meetings will steer the design towards the most economical and timely design.

PPM estimators will provide daily feedback to the PND design team offering historical costs and supplier rough order of magnitude (ROM) pricing during the design concept phase. This approach will allow our Team to minimize construction costs and ensure the project schedule aligns with the Borough's expectations. Our Team will meet with the Borough weekly to provide updates on design and construction budget as they progress. Additional communication will occur as needed to relay critical information as it becomes available and/or design direction needs to be made. One of our Team's internal measures of success will be to optimize the design-build process for the timely and transparent exchange of critical information.

PPM will utilize HCSS HeavyBid estimating software to estimate this project. This software allows for detailed cost reporting by construction category (i.e. labor, equipment rental, permanent materials, subcontractor costs, etc.). These reports will be made available to the Borough.

PPM will utilize Oracle P6 scheduling software to develop the Project's CPM baseline schedule. This allows PPM to develop a detailed WBS to properly sequence work activities through logical relationships. PPM's scheduler will work alongside the estimating team to provide real-time schedule feedback.

Monthly billing detailing all the hours and costs during that period will be produced monthly. The Not to Exceed (NTE) number will be based upon the Level of Effort defined in this proposal. If the level of effort is increased by the Borough, a change may be requested with the supporting information to be approved by the Borough.

The primary challenges in establishing the GMP include a full understanding of the permitting, geotechnical conditions, safe demolition process, and fast-tracked design process to meet the Borough's schedule - all of which we believe to be achievable through collaboration.

The differentiating resources of our Team include the extensive previous experience PPM and PND have working together as well as PPM, PND, and the Borough having successfully completed several projects together. PPM and PND's team leaders have 90 years of combined experience. Our Team has the equipment and available manpower in Southeast Alaska to begin right away. And our Team is supported by one of the best permitting engineers in the industry.

3.c Phase 1 Level of Effort

The Phase 1 Level of Effort (LOE) is a collaborative effort between PPM, PND, the Borough, and Project Stakeholders. The Phase 1 LOE is comprised of many different tasks beginning with Preliminary Engineering & Conceptual Design.

Preconstruction Design and Estimating

Geotechnical Investigation

- Investigation would provide information for the analysis of global and internal stabilities of the dock, liquefaction potential of foundation soils, pile design and drivability, and depth to bedrock.
- PND proposes to advance nine boreholes, 6 boreholes to a depth of 65 feet and 3 boreholes to a depth of 100 feet, distributed throughout the project area.
- Samples will be sent back to PND's AASHTO/ASTM accredited soil testing lab to verify field observations and characterize engineering properties. Graphical subsurface cross-sections will be provided based on this collected data.
- Depth to bedrock or bedrock profile and updated bathymetry will be characterized by a boat-towed geophysical survey.

35% Design Review and Submittal

- Our Team will perform the proper site investigation to begin design.
- Our Team will review the existing design concepts provided by R&M to verify feasibility of design and schedule.
- Once confirmed, PND will provide 35% design drawings and technical specifications.
- PPM will provide a 35% cost estimate based on drawings provided by PND.
- Our Team may provide multiple design concepts and associated costs at this phase for the Borough and Stakeholders to make decisions as the design progresses.
- Complete 35% submittal package (Drawings, Estimate, Schedule) will be provided to the Borough for review and public comment at the end of this phase.

65% Design Review and Submittal

 PND will provide further detailed drawings to the 65% level and address comments from the Borough based on the 35% submittal. A design concept should be decided upon at this stage.

- PPM will price updated drawings and revise schedule to reflect changes made in this phase.
- Complete 65% submittal package will be provided to the Borough for review and public comment at the end of this phase.

95% Design Review and Submittal

- PND to create 95% design drawings incorporating comments from 65% submittal and main design elements are finalized.
- PPM to price updated 95% drawings and revise schedule.
- Risk allocation is agreed upon at this stage.
- Complete 95% submittal package will be provided with more detailed information on cost and cash flow forecasts.

IFC Drawings GMP Negotiation

- PND will now have enough information to complete and stamp IFC drawings.
- Stamped drawings will be reviewed by PPM for final pricing.
- · Final price and drawings will be presented to the Borough.
- Our Team and the Borough will negotiate a GMP contract to perform the work and enter into the Phase 2 Amendment.

Design Meetings and Workshops

- Our Team will have weekly meetings with the Borough to discuss outstanding items, progress on drawings, and other design and estimating elements that need to be addressed.
- Design workshops are meetings to work through the major, complex work items on the project.

Permitting

Permitting is a key item in Phase 1 to be able to build the project. PND's permitting team is experienced in working with permitting agencies to secure permits for highly complex construction projects that may impact the environment. This phase is important because it will play a factor in means and methods, available working windows, marine mammal monitoring, and use of certain equipment. PND has an industryleading permitting team that will be instrumental during this phase.

- Acquisition of general permits through USACE, ADEC, and ADFG.
- Acquisition of Incidental Harassment Authorization (IHA) major permit item that will assess the impact construction activities will have on marine mammals near the project.
- PND to provide Marine Mammal Monitoring and Mitigation Plan.
- PND to coordinate with NEPA and perform environmental assessments.
- 3.d Examples of Deliverables to Communicate the Development of Project Costs and Schedule

*Refer to Appendix for detailed breakdown of Phase 1 Level of Effort.

The following are examples of the deliverables our Team will use to communicate the development of the project costs and project schedule to the Borough:

Design Phase

- Drawings and Specifications refer to sample Seward Pier Replacement 35% Conceptual Document in Appendix*
- Cost Report refer to sample Seward Pier Replacement 35% Estimate in Appendix*
- Schedule refer to sample Seward Pier Replacement 35% Schedule and Schedule Narrative in Appendix*

*Single pages have been provided as a point of reference for brevity. Full packages can be made available upon request.

Our Team will provide detailed cost reports with design drawings for each stage.

Our Team will provide a detailed summary report of expected and known project work and timing restrictions.

Our Team will provide a detailed summary schedule including narrative and report of key sequencing and means and methods to build the project safely and efficiently.

4 CONSTRUCTION MANAGEMENT, SEQUENCING, AND SCHEDULING

4.a Means and Methods to Achieve Efficiencies in Scheduling and Construction Sequencing

Upon Award of the Contract, our Team will immediately start design collaboration with the Borough. Keeping the Project goals in mind, our Team will optimize the design for cost and schedule to execute the work as efficiently as possible.

PPM will provide timely feedback on construction means and methods and historical and current ROM costs to PND to guide design features. As the design progresses, our Team will contact the permitting agencies to manage the securing of all required environmental permits.

PND has a superior understanding and successful track record of securing USACE and IHA permits and maintaining NEPA compliance. PND's permitting department will work integrally with the Team to relay expected permit conditions and the permitting timeline for the project. This will allow the estimators to accurately forecast construction costs based within the project's schedule duration.

Throughout the design process, elements will be refined in consideration of the execution to ensure timely delivery of the Project. Our Team will keep the Borough abreast as to material cost inflation risk and procurement lead times to minimize potential price increases and construction schedule material delays.

Upon mobilization to site, the Construction Team will immediately perform all baseline surveys and temporary environmental controls. Phase 1 demolition of the existing launch ramp will be followed by the construction of the new launch ramp and uplands area. Following completion of those activities, PPM will move on to Phase 2 of construction. Once PPM has demolished the existing structure all excavation, dredging, and disposal will be completed. With the site cleared, the new bulkhead, fill, rip rap, dolphins, catwalks, and utilities will be installed.

If awarded, PPM will then continue with Phase 3 of construction. The approach dock and additional dolphin and catwalk will be installed. The project will conclude with all close-out activities and demobilization from the site.

Should the Borough and terminal operations allow for it, PPM will condense construction sequencing to optimize the construction schedule by seamlessly transitioning from construction activity to construction activity independent of construction phase.

Throughout the design and construction process, construction means and methods will be used to guide the project schedule. Deconstruction and construction methods will be considered during the pricing and planning of the project.

Safety, environment considerations, and efficiency will be discussed in conjunction with pricing of design elements. For every major area of the Work, work plans will be developed showing sequenced layout drawings detailing the resources to be used. Included in the work plans will be equipment capacity charts, located pick points and centers of gravity, template designs, disposal plans, etc.

4.b Approach for Achieving the Performance Requirements and Optimizing for Quality

- Dick Somerville will act as the Design Quality Control Manager. Dick has extensive knowledge of the project and has worked for the Borough directly on many past projects and understands the quality requirements required.
- PND will perform fabrication inspection of all the major materials before they are shipped to the Project. This will include piles, structural steel, coatings and fender elements.
- PPM will produce submittal and shop drawings for review by the Borough and/or any identified Representative(s) during the course of the Project.
- The Specifications will identify all Quality Assurance and Quality Control (QA/QC) testing required during the execution of the project.
- PND will review all materials submittals and certificates of compliance for the materials on the project and the Field Testing results.
- PND will provide Construction Oversight of the construction process and ensure all construction meets the specifications.
- PPM will contract an independent testing agency to carry out all field testing required by the specifications.

 Commissioning of the Project will be handled by the Project Manager, Stewart Willis, who recently accomplished a large commissioning at the Port of Alaska on a major \$83M project with many complex systems. A commissioning plan will be formulated and submitted detailing the testing, asbuilt information, Q&M manuals, Operational training, and information required for each part of the Project.

Quality control measures to keep the project on budget and on schedule will include a comprehensive set of quality review measures, checks, and counter-checks that will be performed on every deliverable. The process begins at the Start Up meeting to make sure all parties fully understand the project objectives, functional needs, and client expectations as well as potential issues and risks. Applicable codes and standards to which the technical review will be conducted will be verified. Design milestones and dates will be confirmed at this time. Regardless of which technical discipline or sub-consultant is performing the work, all documents produced and delivered to the Borough will undergo internal document checks prior to project milestones. These checks will include:

- Coordination Checks (inter-discipline checks)
- Technical Checks Plans and Specifications
- Constructability Review

Prior to the release of deliverables, QA/QC checklists will be completed to document that the reviews have been accomplished, and responses to comments and outstanding issues have been adequately addressed to the reviewers' satisfaction.

In addition to the Start Up meeting, our Team will hold regularly scheduled coordination meetings throughout the design process to keep the Borough informed of progress and address any questions that may arise during the design process. The design team will also remain engaged throughout construction and will provide shop drawing/material submittal reviews, inspections at key milestones and problem solving as needed during construction.

Our design team's recognition and success in the industry for over 40 years is founded on sound project management and design QC programs. PND has a complete set of quality control guidelines and procedures for design, and if selected for the Lutak Dock project, our Team will submit a QA/QC plan for the Borough's review.

Quality Management Plan

DESIGN CONTROL PURPOSE AND SCOPE

Design control is intended to control project costs, schedule, and quality by ensuring that engineering designs are technically correct, in accordance with pertinent codes and regulations, and are constructible. This purpose is to ensure the following:

- Design specifications, regulatory code requirements, and engineering standards are correctly incorporated and applied to drawings, specifications, procedures, and instructions.
- Appropriate construction standards are specified in the design documents.
- Selection and review of materials and processes that are essential to construction are suitable.
- Design review and checking by appropriate licensed professionals are performed.
- All design documents are reviewed and approved in accordance with established QA/QC policies.
- Issuance and distribution of all design documents are properly controlled.

<u>QUALITY CONTROL AND QUALITY ASSURANCE OF DESIGN</u> <u>General procedures for the QC process for design shall include</u> the following activities.

- Detail checking of all drawings and calculations prior to release of deliverables.
- Principal In Charge (PIC) technical review of drawings and calculations.
- Consideration of constructability, alternatives, and cost benefits.
- Verification that applicable quality levels and standards have been specified for the intended use, materials, and processes specified and appropriate to the application.
- Review of suitability for design methodologies, such as modeling and analysis.

QC checking of calculations, drawings, and other design documents shall be performed by a registered engineer with an appropriate level of expertise and adequate experience to perform the work being checked. The person shall be approved by the PIC. The reviewer may be the PIC or another designated engineer.

REVIEW AND DESIGN DOCUMENT APPROVAL

- Internal Design Review: The intent of internal design review is to establish that the design aspects have been adequately and accurately expressed, that the design is in accordance with applicable codes, standards, and regulations, and to verify the constructability and approach of the design. Design reviews shall be performed by the Design Project Manager, PIC, and independent project engineer.
- Quality Assurance Review Meetings: Design review meetings will be held at project milestones to coordinate between disciplines, reconcile comments, and establish direction for proceeding to the next level. Participants in design review meetings shall include our Team, subconsultants, Borough Representatives, and any necessary Stakeholders.
- Reconciliation of Comments: The Design Project Manager is responsible for the compilation and reconciliation of comments from all reviewers and transfer of the reconciled comments to the design team. Final reconciled comments submitted to the designers shall reference the appropriate drawing or specification, shall be clear and concise, and shall be non-contradictory. The designers shall prepare corrections/clarifications and responses to comments. Any comment that is not to be incorporated in the next phase of design must be approved by the Design Project Manager and PIC.
- Quality Assurance Project Approval: Approval for projects to proceed to construction shall be provided by the PIC. The Design Project Manager is responsible for obtaining appropriate approvals and signatures. All design documents, including drawings and specifications, shall be sealed by one or more registered professionals who are specifically approved by the PND Board of Directors.

QUALITY ASSURANCE/FIELD ENGINEERING SUPPORT

 During Fabrication: PND personnel will provide engineering support and QA inspections during material procurement and fabrication. All drawing changes and/or Request for Information (RFIs) that are generated during the fabrication phase shall be reviewed by the Design Manager to ensure any design modifications are complete, accurate and that RFI's are adequately addressed.

PND will develop a submittal register to track submittals for all project materials. Mill certificates, cut sheets, shop drawings and other submittals necessary to ensure compliance will be reviewed by PND for conformance with Plans and Specifications. No project components will be ordered or fabricated prior to approval of the procurement submittals. PND will periodically visit the fabricators to review workmanship and prepare a report of those visits.

 During Construction: PND personnel will provide engineering support and QA inspections during construction. Qualified and certified inspectors will ensure that construction is conducted in accordance with the design documents, and constant communication with PPM will be maintained to clarify design intent for critical project elements. Pile driving will be monitored to verify design loads are being achieved. Materials testing will be performed by certified PND personnel as needed. Daily construction inspection reports with representative photos will be produced by PND to document construction activities.

All drawing changes and RFIs that are generated during the construction phase shall be reviewed by the Design Manager to ensure any design modifications are complete and accurate and that the RFI's are adequately addressed. A close-out punch list will be generated in collaboration with PPM and the Owner and items identified will be completed prior to final completion and commissioning of the project.

4.c Approach to exceed Project Goal #5 to Maximize Safety During the Work

The safety and well-being of our workforce, partners, and the surrounding community is and always will be our first priority. This begins at the design stage when considering the execution of the work and designing to reduce safety concerns and health risks associated with and challenging execution. Designing for Safety (DfS) follows principles associated with systematic risk management, product life cycles, project controls, information transfer, and incorporating elements of design to enhance safety during the construction phase.

PPM utilizes tools such as HCSS Safety to conduct toolbox meetings, reporting, and capture safety metrics to track leading and lagging indicators. PPM follows a behavioral-based approach to behavior believing there is always an opportunity to improve and a lesson to be learned to apply moving forward. Our attitude of continuous improvement has allowed us to keep safety and quality at the forefront of all design and planning discussions. These discussions will be utilized during constructibility input for design.

Safe design will integrate hazard identification and mitigation paired with risk assessment based on management of the Risk Matrix that will be advanced during Site investigations.

Our approach will be enhanced through ample pre-planning prior to execution of the work as well as continuous 'check and confirm' hold points throughout construction to discuss the plan and verify all parties understand and support the plan and no betterments are identified for further consideration and vetting.

Methods such as these are attributable to the success of similar projects such as:

- \$14.1M Dakota Creek Industries Shipyard Redevelopment [450-ft Open Cell Sheet Pile Bulkhead and 14,800 SF pile-supported pier using alternate delivery contracting including *design-build* elements performed by PND]
- \$10.3M Hebgen Dam Cellular Cofferdam [design-build, emergency deep intake structure in environmentally sensitive headwaters involving construction of a closed cell cofferdam and tensioned rock anchors comprised of 36-in diameter shafts]

\$33.3M Palmer Pier Replacement

[CM/GC pre-construction services and construction contract to remove the existing sheet pile bulkhead with a new pier on Anvers Island, Antarctica]

4.d Identify the Challenges and How the Team will Address those Challenges

The biggest challenges this project faces are: Permitting, Geotechnical Design, and Demolition of the Existing Sheetpile Cells.

Though there are currently unknowns associated with all these challenges, our Team will develop and execute a plan to quickly decipher the necessary answers required to make this project highly successful for the Borough and its Stakeholders.

Our Team will work with the Borough and the permitting agencies so all parties understand the construction scope and the environmental concerns and regulations.
Discussion topics will be:

- Federal Funding Requirements (NEPA); release of funds for design
- Separation of Phases (separate permits or one combined
- Schedule of Permitting and Construction
 - » IHA
 - Procurement of materials; construction window (winter weather)

This will allow for the construction work window to be established and provide environment guidance in developing the final work restrictions, construction means and methods, and budget.

During the design process, our Team will perform the required geotechnical site investigation. This is necessitated by the lack of geotechnical information available.

Upon site investigation, the our Team will gather information to characterize the existing conditions regarding stability, liquefaction, bedrock elevation and profile, etc. This will allow the design to be completed, risks to be minimized, and the project to be properly budgeted.

The biggest construction challenge is in deconstructing the existing closed sheetpile cells. The sheets are old and heavily corroded and it is unlikely they cannot be extracted in single pieces. Our Team will perform thorough site investigations and develop a step-by-step plan to remove the sheets in a safe and efficient way. This understanding will allow our Team to consider all options and select the method that provides the greatest benefit to all project Stakeholders.

As with any project such as this, additional challenges may arise, but our Team is prepared and sufficiently experienced to overcome these challenges through early and ongoing coordination efforts with the Borough.

Early and continued communication, proper planning, and detailed execution will be the keys to delivering this Project successfully.

4.e Detail the Tools Used in this Process and How they will Assist to Exceed the Goals

- To design and construct the dock and maximize the program requirements with the limited budget, our Team will start by drawing upon our vast construction knowledge and experience to quickly vet the various design possibilities and systematically narrow down the options to select the optimum design and reduce long term maintenance. Additionally, PPM will use HCSS HeavyBid estimating software and Oracle P6 scheduling software to allow for quick estimate data entry, refinement, and optimization.
- To execute a successful, collaborative progressive designbuild process to produce the envisioned project, our Team will work with the Borough and its Stakeholders to optimize the Project within the Project constraints. Our Team will host weekly progress meetings. For time-critical design/schedule/cost elements, our Team will contact the Borough to discuss matters via impromptu video conference meetings, telephone calls, and/or emails, as required. Additionally, our Team will be available, as needed, to respond to any inquiries and/or suggestions the Borough may have.
- 3. Our Team will provide transparent pricing via HCSS HeavyBid cost reports and scheduling via Oracle P6 CPM schedules that will allow the Borough to track design and construction concurrently as well as fast-track design and construction to maximize the budget. Open-book estimating will be provided throughout the process and all construction means and methods will be described in sufficient detail. Open, transparent communication will be facilitated timely to address any time and/or schedule critical issues. At each design milestone, our Team will provide the Borough with progressively more complete and detailed cost reports and schedules.
- 4. Our Team will comply with all applicable State and Federal Legal requirements by working with the permitting agencies to incorporate all environmental requirements for this Project. Additionally, the Construction Team will review in their entirety the Project Contract Documents before ever stepping foot on the project site and be in communication with the Borough should any question arise.
- 5. Safety considerations will be at the forefront of this Project's design. Throughout the design process, PPM will be developing high-level work plans for every major work activity. These work plans will become the starting point for the construction team's work plans once the project is

awarded. Each crew will work through every step of the activities to produce and execute safe work practices. These work plans will minimize safety risks and reduce the need for rework. Our Team is committed to achieving a recordable-free and incident-free project.

. .

A 4

- ---

.

APPENDIX PHASE 1 LEVEL OF EFFORT SUPPLEMENT

444

1.4 1.5

a. 64



Pacific Pile & Marine Lutak Dock RFP Phase 1 Level of Effort

		Chief Estimator	Senior Estimator	Estimator	Estimating Manager	Superintendent				
	Hourly Rate	\$ -	\$ 100.00	\$ 55.00	\$ 55.00	\$ 100.00	Labor	Sub	Expenses	
No.	Scope									
1	35% Estimate + Review	112	40	96	40	72	\$ 18,680.00		\$ 7,000.00	
2	65% Estimate + Review	136	56	136	60	56	\$ 21,980.00		\$ 5,600.00	
3	95% Estimate + Review	136	40	136	40	72	\$ 20,880.00		\$ 5,600.00	
4	Weekly Meetings	52	52	52	52	52	\$ 16,120.00			
5	Workshops	48	48		48	48	\$ 12,240.00			
6	GMP Pricing and Negotiations	96	36	40	60	36	\$ 12,700.00		\$ 4,200.00	
	Subtotal Hrs	580	272	460	300	336				Total
	Subtotal \$	\$ -	\$ 27,200.00	\$ 25,300.00	\$ 16,500.00	\$ 33,600.00	\$ 102,600.00	\$ -	\$ 22,400.00	\$ 125,000.00

PND Engineers, Inc.

Lutak Dock Replacement

PND Project 22J022

Engineering Services Fee Proposal - July 15, 2022

P N D

Scope of Services

No.

[Senior Eng. VII	Senior Eng. VI	Senior Eng. V	Senior Eng. III	Senior LS III	Senior LS II	Tech. VI	Tech. V	CAD Design VI		
Billing Rate	\$225.00	\$210.00	\$190.00	\$165.00	\$135.00	\$125.00	\$150.00	\$130.00	\$130.00	Labor	Expenses
			Predesign	Site Invest	igation Serv	vices				the state of the second	
lination	8				36		4			\$7,260	
					20	20				\$5,200	\$9,400
					20	16				\$4,700	
					40	40				\$10,400	
					20	20				\$5,200	
					8	8				\$2,080	

Sub

Total

						B								and the second se
	Task 1 - Topographic and Bathymetric Surveying													
1.1	Admin, Management, Site Research, Meetings & Client Coordination	8				36		4			\$7,260			\$7,260
1.2	Field Prep, Mobe, Demobe & Expenses					20	20				\$5,200	\$9,400		\$14,600
1.3	Horizontal & Vertical Control, Utility Locates and Asbuilts					20	16				\$4,700			\$4,700
1.4	Upland Topographic Survey					40	40				\$10,400			\$10,400
1.5	Bathymetric Survey					20	20				\$5,200			\$5,200
1.6	UAV Survey					8	8				\$2,080			\$2,080
1.7	Data Reduction and Prepare Base Map					16	48			12	\$9,720			\$9,720
1.8	Final QC and Deliverables	4				4	4			4	\$2,460			\$2,460
														\$56,420
	Task 2 - Geotechnical Investigation													
2.1	Management, Site Research, Coordination with Client & Users	16	8	4				8			\$7,240			\$7,240
2.2	Field Prep, Mobe, Demobe & Field Expenses							16			\$2,400	\$7,450	\$38,896	\$48,746
2.3	Field Drilling Investigation - Field Log Boreholes			10				108			\$18,100		\$71,775	\$89,875
2.4	Lab Testing		4	8							\$2,360	\$5,600		\$7,960
2.5	Data Reduction and Final Borehole Logs		4	8				40			\$8,360			\$8,360
2.6	Slope Stability & Pile Foundation Analyses	8	16	24							\$9,720			\$9,720
2.7	Geotechnical Report	8	16	16						8	\$9,240			\$9,240
														\$181,141
	Subtotal Hrs	44	48	70	0	164	156	176	0	24				682
	Subtotal S	\$9,900	\$10,080	\$13,300	\$0	\$22,140	\$19,500	\$26,400	\$0	\$3,120	\$104,440	\$22.450	\$110,671	\$237,561

PND Engineers, Inc. Lutak Dock Replacement



Engineering Services Fee Proposal - July 15, 2022 PND Project 221022

LIND	110	ect	22 022

Scope of Services	Senior Eng. VII	Senior Eng. VI	Senior Eng. V	Senior Eng. III	Senior Eng. II	Env. Scientist I	Tech. VI	Tech. V	CAD Design VI	1.1	6.1	r	T 1
Buing Kate	\$225.00	\$210.00	5190.00	\$165.00	\$155.00	5145.00	\$150.00	\$130.00	\$130.00	Labor	Sub	Expenses	Iotal
1 Prelim Envincering & Concentual Design Confirmation & Undates	40	40	40	icering Desi	Supervices		16		40	\$32,600		\$3,000	\$35.600
Public Involvement	60	40	40				16		40	\$37.100		58,000	\$45.100
3 Demolition Plan & Work Summary	24	40	40						40	\$26,600			\$26,600
4 Site Layout Plans	24	24	60		40				60	\$35,840			\$35,840
5 Earthwork - Grading, Drainage and Surface Course	24	24	80		60				60	\$42,740			\$42,740
6 Fire Suppression	8	8	24						24	\$11,160			\$11,160
7 Armor Rock Shore Protection	8	8	20	40					24	\$17,000			\$17,000
8 Approach Dock - Abutment	16	40	40	8					60	\$28,720			\$28,720
9 Approach Dock - Piles	16	40	60	8					40	\$29,920			\$29,920
10 Approach Dock - Superstructure	16	60	40	24					60	\$35,560			\$35,560
11 Boat Launch Ramp	16	20	80	24					60	\$34,760			\$34,760
12 Bulkhead	24	80	80	40					60	\$51,800			\$51,800
13 Dolphin - Pile Design	16	40	24	8					40	\$23,080			\$23,080
14 Dolphin - Fender System	24	60	40	16				-	60	\$36,040			\$36,040
15 Dolphin - Pile Cap and Attachments	16	60	24	16		1			60	\$31,200			\$31,200
16 Dolphin - Access Gangway, Catwalk and Landings	16	60	24	16					60	\$31,200			\$31,200
17 Dolphin - Safety Ladder & Light Pole Supports	16	24	24	8					40	\$19,720			\$19,720
18 Pin Pile & Pile Sockets	16	40	24	16					40	\$24,400			\$24,400
19 Pile Rock Anchors	16	60	40	16					40	\$31,640			\$31,640
20 Sacrificial Anodes	8	24	16	8					20	\$13,800			\$13,800
21 Civil & Structural Calculations Package	24	120	80	40			16			\$54,800			\$54,800
22 Technical Specifications	24	40	40	40	20		40			\$37,100			\$37,100
23 Design Coordination Meetings w/ Team and Owner	80	60	40	24						\$42,160		\$10,000	\$52,160
24 Prepare 35% Design Review Submittal & Respond to Comments	24	32	32	32			16		24	\$29,000			\$29,000
25 Prepare 65% Design Review Submittal & Respond to Comments	24	32	32	32			16		24	\$29,000			\$29,000
26 Prepare 95% Design Review Submittal & Respond to Comments	24	32	32	32			16		24	\$29,000			\$29,000
27 PND Internal Design QA Audit - Plans, Specs and Calcs.	24	40	40	40			16		24	\$33,520			\$33,520
28 Prepare 100% Stamped Final Design Documents - IFC	24	40	40	40			16		16	\$32,480			\$32,480
29 Mechanical Design and Coordination - Fuel System	16	40	40	8			16		24	\$26,440	\$82,500		\$108,940
30 Electrical Design and Coordination - Power and Lighting	16	40	40	8		-	16		24	\$26,440	\$82,500		\$108,940
Subtotal His	684	1268	1236	544	120	0	200	0	1088				5140
Subtatal S	\$153,900	\$266,280	\$234,840	\$\$9,760	\$18,600	\$0	\$30,000	\$0	\$141,440	\$934,820	\$165,000	\$21,000	\$1,120,820

PND Engineers, Inc.

Lutak Dock Replacement



Engineering Services Fee Proposal - July 15, 2022

PND Project 22J022

		Senior	Senior	Senior	Senior	Senior	Env.	Tech.	Tech.	CAD				
	Scope of Services	Eng. VII	Eng. VI	Eng. V	Eng. III	Eng. II	Scientist I	VI	v	Design VI				
-	Billing Rate	\$225.00	\$210.00	\$190.00	\$165.00	\$155.00	\$145.00	\$150.00	\$130.00	\$130.00	Labor	Sub	Expenses	Total
No.				Environn	iental and Pe	rmitting Ser	vices							
	Task 1 - General (USACE, ADEC, ADFG Fish Habitat)													
1.1	Admin, Management and Client Coordination	4	4								\$1,740			
1.2	Develop USACE Permit Drawings and Material Quantities	2	4			12	20			16	\$8,130			
1.3	Prepare Permit Applications Including Alternatives Analysis	2	4			12	16			12	\$7,030			
1.4	Prepare Biological Assessment	2	2				50			12	\$9,680			
1.5	Submit Permit Applications to Owner / Incorporate Comments	2	2			4	8			4	\$3,170			
1.6	Respond to Agency Comments / Questions	2	2				16				\$3,190			\$32,940
	Task 2 - Incidental Harassment Authorization (IHA)													
2.1	Admin, Management and Client Coordination	8	20								\$6,000			
2.2	Develop IHA Request	6	10				160			20	\$29,250			
2.3	Develop Expanded Biological Assessment for IHA						60				\$8,700			
2.4	Develop Marine Mammal Monitoring and Mitigation Plan (4MP)	6	10				120			20	\$23,450			
2.5	Submit Draft Documents to Owner / Respond to Comments	2	2				20				\$3,770			
2.6	NMFS Coordination / Respond to Comments and Questions	10	10				80				\$15,950			
2.7	Closeout Reporting	2	4				60				\$9,990			\$97,110
	Task 3 - NEPA													
3.1	Admin, Management and Client Coordination	16	24								\$8,640			
3.2	Environmental Assessment /NEPA Documentation	16	24				240			24	\$46,560			
3.3	Submit Draft Documents to Agency/ Respond to Comments	8	16				80			12	\$18,320			
3.4	Agency Coordination / Respond to Questions and Comments	8	16				60				\$13,860			\$87,380
	Subtotal His	96	154	0	0	28	990	0	0	120		0	.0	1388
	Subtatal S	\$21,600	\$32,340	\$0	\$0	\$4,340	\$143,550	50	50	\$15,600	\$217,430			\$217,430

APPENDIX

Examples of Deliverables to Communicate the Development of Project Costs and Schedule



Pacific Pile & Marine 22-045 Seward 35% Steve Grayson

07/12/2022 14:44

Activity Resource	Desc	Pcs	Quantity Unit			Unit Cost	La	Pern bor Materia	n Constr I Matl/Exp	Equip Ment	Sub- Contract	Total	
BID ITEM =	= 32 New Waterline Work				Unit	= LF	Take	off Quan:	185,000) <u>E</u> ngr	Quan:	0.000	
32.07	Patch Trench				Qua	n: 3,700.0) <u>\$</u> F	Hrs/Shft:	10.00 Cal	: 610 WC	: AKUSI	<u>.</u> Нокала **11	
Assumed 185 4 PAV E	Y x 20' Trench Paving Sub	1.00	3,700.00 SF			4.150					15,355	15,355	eviewed
32.08	Parking Lot Pavement Ma	urkings			Qua	à: 1.0).LS	Hrs/Shift;	10.00. Cal	: 610 WC	: AKUSI	a 🔬	
Repaint Par 4TRAFFICPNT	king Stalls Traffic Paint Subcontractor	1.00	1.00 LS			1,500.000					1,500	**Unn	eviewed
32.09	Disposal of Surplus Mater	ial			Qua	n: 356.0) CY	Hrs/Shft;	10.00 Cal	610 WC	: AKUSI	н	
AKOP1	1 Operator Crew	1.00	1 79 MI	1.78	СН	Pro	d:	0.0050 MU	Lab Pcs:	1.00	Eqp Per	**Unro s: 1.00	eviewed
3SAF	*** SAFETY ***	1.00	1.78 MH			13.000			27			27	
3STS	Small Tools and Supplies	1.00	1.78 MH			5.000			9			9	
8EWL724	724 Loader 4.75 cy - O	1.00	1.78 HR			79.000				141		141	
AKOIA	Marine AK- Oper Group Ia	1.00	1.78 MH			45,290	1	57	27	1.4.1		157	
\$\$\$\$.55	0.0050 WIII/C	I	1.76 WH			[0.204]	1	57	57	141		333	
	Totals: 32 -	New Wa	terline Work			-							
\$62,157,29	1.3263 MH/LF		245.38 MH			f 64.132 1	20.9	88	5,153	17.850	18,167	62,157	
335.985	185 LF					ľ	113.	45	27.85	96,48	98.20	335,99	
·												··· <u>-</u>	
<u></u>	<u></u>		x			Total	of Aboy	e Sub-Bidit	ems	· ···	. <u>.</u>	·	
====⇒ Item '	Totals: 30 -	Phase 1	Upland Water	line &	: Hydı	ants	20.0	80 77 757	0.100	17.050	10.077	1 (1 0 5 0	
5141,949.57 141,949.570	245.3800 MH/LS 1 LS		245.38 MH		l	11804.34]	20,9 20,987.	64 73,757.28	9,388 9,387.98	17,850	19,967 19,967.00	141,950 141,949,57	
,													
PARENT ITEM Description = Listing of Sub-B	I = 40 CLIEN Phase 1- Existing Dock Sheet Biditems of Parent Item 40:	T# = 40 Pile & C) Conn		Unit =	= LS	Takeo	off Quan:	1.000	Engr	Quan:	CJW 1.000	
BID ITEM = Description =	41 Purchase Sheet Pile & Attacht	ments			Unit =	= LS	Takeo	off Quan:	1.000	Engr	Quan:	0.000	
41.01	Purchase zz38 Sheet Piles I	TOB SE		6 A)	Quan	1 5,102,040.00	LB	Hrs/Shft;	10.00 Cal:	610 WC	; AKUSL	Ĥ	
80 shhets su 2SP	<pre>upply lenght= 82? *** SHEET PILING ***</pre>	1.00 5,1	02,040.00 LB			0.999		5,099,489				5,099,489	
41.015	Purchae Omega corners				Quan	: 164.00	LF	Irs/Shft:	10,00 Cal:	610 WC	AKUSL	H	an a
2SP	*** SHEET PILING ***	1.00	164.00 LF			70.000		11,480				11,480	
41.02	Burn Liftign Hole in Sheets				Quan	: 366.00	EA)	Irs/Shft:	10,00 Cal:	610 WC	AKUSL	H	

Page 9

1.1.1.1.1.1.1.1.1

Seward Dock	Replacement - 35% Design Schedu	ule	for a second second					(Classic So	hedule La	yout											18-Ma	ay-22 12:17
	And the second s	Dutin	Front Ploof Apr	Lby	an au	Ala filo	Red Nov	En: Jan	Fee	the Ap	the .	An Au	Aq	Seg Cut	New Dec		eb Mar	ka May	1 30	м	Aug Sep	0.1	No. Dec
22-045 Seward	pock Replacement = 35% Design Schedule =	11 11-14-22	Distant B																			- 3	04-34 22645.1 (Besture)
See 50100	1019 Availat Pre-Costadan - Plate 1	0 8-44-32	ap MIPA	manuel Fre-Graduation Ph	un 105-77-31				+														
Alizo	Briterige Starting	4	11-34-22 5	P and and	CT IN Decarde	DIRA																	
Mar MICO	Subst ConstructionServices Coal Proposit (CSOP) Fergilized CSOP Complete	0	0.400 2" 0 0.500 2" 0		4	Satria Constuction Server	Cart Propaga (CCP)																
SIS MILES	AFR: Bassigna	0	22-5ep-37 0				ARC Burglaurosa							1	1 1		1 1						
See Mild?D	TableAconders	0 0-04/2	24454-34 7				Million and the College											Summer	and the second				
THE MILEO	Pranci Paul Completion	0	21-Ou3# D															0140-24.2				- 14	entPeulCongletin
22 045.21 See	nit : Naraper 2022	12 8436-23	869.2 1											28/Sep-21.2	4621 Season 1 Summer	22							
WEITERS	2025-second Operations - Health Abbreach & Operations - Stat 2022-second Operations - Health Abbreach & Operations - Engl	0 BiAte, 27	Tolian JY C						1 1		all's Bearing	Operations -1 Bulle Ticks	THE REPORTS STREET	day 27	a constant, 1000-00 line	an el suemos des							
221H5.2.7 Sens	cen 2 - Materius 2024	U DEADARD	and a state						1									¥ 83-654-34 3	245.2.2 Seenan.2	Summer 2024			
22.045.10 Des	2045 Autor Operation - 20 who full in the instance - Stat	Minute Contraction	2401 3						2 Jan 21 22 06 10	Deag Plane			1 1					2024 Season	nativentare 204	the Till Broak & Oper	initiana -Stat 01404-34"		
1 22.045 HLT 35*	Desga	TO HE AND	thouse 1		08-An-22 2046 101 375 Dea									1									
- D1010	D 3/% Design Scientist APRC Pastery & Acceptance	24 (6-Apr-32 10 20-May-22	B-Am22 C	DELOS	D-5% Deven Suznime	Bearing Agenciance	+	· · · · · · · · · · · · · · · · · · ·							·}								
22/045, 10,2 654	Contaget	2 (8.40.2	25.3622 22		11 0000	22, 22-64(00), 69% Dem																	
- C2210	E-ID's Design Scarsing AFRE Paster & Acceptance	W 50.44.22	3-4-22 25		00	Deaptioned AFR	Senarch Accession																
22-045-10.3 Fm	FDFirst Despt5stratel	72 25-14-22	Diana) S				dama and an and a second	Porezolaritaria	25 Jun 21 22-04 10	1 Peut Design													
A1010	PDPinal Design Submittel - APRC Reason & Automatics	24 (0+Dec-22	25-305-21 Di					Ferrer	rornatinay ta	the ADEC Passes SA	Acceptance	_	-										
PERnau	Seura 4 Regard Pende	15 (B-10-22	18-34-23 6									Secure Addressed	Poon						1				
5 22-045.3 Cont	racts for Subcontractors and Suppliers	W E-Dep-22	2 and 1			-	245-0-22 2249.3 Corah	to Subcreaker and Supplem															
C 1000	Concrete Municipalities - Sup Concrete Municipalities - Sup	10 13.5ep.27 10 13.5ep.27	215ep-22 5 21-5ep-22 17			-	Covereith Agregate - Sup		T													1	
C100	Detword Sub	10 13-5-62-22	345ep.21 17			-	Westerical-Sto																
L C.520	Feters Sp	10 15 op.2 10 13 5ep.2	3-5-q-2 1/ 3-5-q-2 1/				Ferdra Sup																
10 C100	Steel Pas Piling - Sta	10 135452	3-5ep.22 5			-	Shellige Parg-Sup				Solid Camata				1				1	1			
B 22045.4.1 Pirc	antian a storne	123.03	2000 2				1			214 te-21 2	2464 www.muter	area the											
am 5500	Renarg Reconstates Salester	22 25-400-33 10 10 440 23	C-41a-D 25							Anamight coase	an Dublinder	in Parameter Ser							1				
21045.4.7 Cos	and the Second and	H 2444-2	Network B							19/4- 21 21-0	45.42 October Supra	44	-		++								
ana 56000	ShetPing PasPing	5 255q52 5 265q52	07-003-22 5 17-003-32 6				Par Paul	1 1 1												1 1			
an (x252)	Medanca .	2 359-2	2346+2 17				Abduer.	4															
an 19940	Persex:	20 26 Sep 21 21 26 Sep 21	(5-160-2) 17 (5-160-2) 17				Fundan	+++-														++	+
- SAE/20	Carcole McAgregies	20 3-5 - 20	01449-20 17 76(145-72 8)				Curumet	Balagengaber															
- 182011	ARE Relevant Contract PterPiling	10 12 04422	360432 8				-Carlo Milic Fening	ant-control - Per Pary											1				
All 58351	AFFL Fellew and Acceptance Mediance AFFC Fellew and Acceptance Electrical	10 (2516+22 10 (2516+22	244a-21 1/ 374a-21 1/					ARC Poles and Cardina - 5 ARC Poles and Cardina - E	Detroit														+
aa 58625	AURIC Reades and incurplance - Persiens	10 10-140+22	2144-2 17					Addrec Penany and Auropance P	ran							1 1							
Mar (1900)	Perseng Contractor Summer	24 23-44-23	3.Fm-2) 31	1 1				1 4		Arrianting Compaction 1	Sienitia			1									
10 14501	AFE Reasonable uptonse Renaming Conductor Sucretion	10 28462	(SAI#2) 31							AIR: Firsh	and-cue and - Personal	Constantion Section Section Section 1997	raet.		+								
an Pitte	Dog Encloser Steel Parg	18/ 2-04122	24429-273 15	1		1			-		D SENANOS	Par				1 1					1		
PIGIE PIGIE	CCP WeetPlang	180 2540e5-22 183 2640e5-22	24 Apr 23 15 36 day 23 15		1 1		-	1 1	1 1		DazarPing					1 1	1 1		1				
🚥 P100	Funders	181 22 883-22	20440-20 20								Fer												
P100	Entrui	180 2140-22	2449-27 2								- D	end											
E PIND	Listancia	180 20 Nov-32	21.May-31 12		1 1			1	1 1			Parical											5 04 24 23 045 6 Downlean
22-045.6 Con	Station/ Demotrikation	28 Bills-21	DO4A																				10434 20466 Materia
and an other states	The set of the set of the set of the set of the set	31 (0.11)-31	2,402									Si-As	e-23 20 Kin Hill Materian										
	Terflagest: Severs	23-an-21	23023 8									Ter Sar	ages to Servers										
ALL	In and and Official Darges inConvert Insial TESC Measures	1 33021	Ban2) 6									E bate	TESC Stations										
ALL	Deta Bergaray (Mel Alline	N 25-346-21	19-34-21				1						Sap Tanay Sefation									Boast :	abort 6.11 3 Gerganuty San P.a.
- MONING	Sette Couver Buth Plat	10 8-34-23	2-34-23 %									5	APCRONE LAS	Ples				1	me come				
MELIZING	Person Cascole Data Part Pierson Terporacy Tit Pacifica	10 3-Apr.24 5 25-949-24	10-Oct-24 10																			Per Due Re	reany file Far Ste
20EGRIS	Pres Racinitat Terr	2 1014	21-04-34																			-	Hep Darges to Kar
D15,2010	TandagastunSe	8	210634 2															240.00	200561 540	a second	AD4 DA	- P*	na Carges Itom Sale
12 22/045.0.1 Ext	Hard Press All Physicial and Carolik class States States States States States States Stress States Stress States Stress S	3 2-46-21	25-an21 15										grate allowarten Saley						Contra Contra	1			
- EP 1010	Destilius NrWater	16 05-34-23	0 Equinal	1		1		1			T	4	Dension Bar	and the									
- 6P100	Inal SteeParg EarlingPer	4 GAND	250423 5										A second second		Had StreePing Entrol	*				1 1			
an EPISA	LCCP8 Intel Latters	18y 25Ga21 1 25G421	244pr34 4 2740r33 193					1			and the second				a poullaster			T					
- (DP103)	Inal 2000 Class	6 24CH23	31-04-22 5					1				1			Pold 23Appredate				Farm In 27 M	the Star Same			
EP 1000	I sold Heigh orderstic 2024 Frade Ship Season SP Peer Takonakan	2 3.40 JA	BAR A I														+	- Martin day	and the second second			SP Parter	
CCSP NRC	Braul Weak Trents	14 21-240-23	314+3 8												Public C	utiese				to Steel Ping Case			
CCSP100	Dia // Collection Col	00 2010-20 01 204pt-24	Ban N 4									1			1			- Georgeone		Ita BCoherandor			1
COSPIERO COSPIERO	Face PCC Transmittiggent Studied Steel	15 15-Jun-34 23 23-Jun-34	1-34-24 b																4	VacaPOCT	nmentappinaused galandF2	7	
= 0.59108	Inad Facebon ad Bullia	11 20.000	17.342N 34						1									_		- A had	elematican		
		No. 332 4		1.025					-	-													
A	ctual Level of Effort Rem	aining Work	< +	 Milesto 	me				Pa	ge 1 of 2				1	TASK filter: /	All Activities							
A	ctual Work Critic	al Remaining	g Work	Summa	ary																9	Oracle	Corporation

Schedule and Schedule Narrative

See attached.

Narrative:

Preconstruction Activities:

To ensure all contract milestones and work restrictions are met, the following is the critical path required to achieve them. Once the 35% Design is complete and ARRC has accepted it, the design is sent to the permitting agencies to secure the required permits. While the permitting agencies are reviewing and processing the permit application, PPM/PND will be working with ARRC to finalize the 65% and then Final design, getting key subcontractors and suppliers under contract, achieving submittal approvals, and procuring materials. Once the materials have been procured and all the required permits have been secured, PPM will mobilize to the jobsite to begin construction.

<u>Note</u>: To achieve the April 30, 2024, Substantial Completion Milestone, critical permanent materials must be ordered prior to the completion of the Final design.

Construction Activities:

Construction will be performed continuously onsite making allowances to ensure there are no disruptions to cruise ship operations.

As this work begins during the 2023 Cruise Ship Season, PPM will sequence the installation to avoid any disruptions to cruise ship operations.

Existing Pier Activities in Preparation for the 2024 Cruise Ship Season:

This onsite work will commence in late June 2023 and run through late-April 2024. Once the TESC measures are in place and the hydrographic and construction surveying is complete, PPM will perform all necessary demolition to begin the installation of the sheet piling encapsulating the existing pier. Upon completion of the sheet piling installation, PPM will install the LCC fill. In parallel to the completion of the LCC fill, PPM will complete the remaining activities required to complete the upgrades to the existing pier.

OCSP Pier Extension:

The OCSP work is scheduled to begin in early November 2023 and continue until late-October 2024.

Once the existing pier sheet piling and 220-ton bollards are installed, PPM will install the temporary work trestle for the OCSP and begin the installation of the OCSP. Upon completing the installation of approximately two-thirds of the OCSP sheet piling, PPM will commence the bulk fill, vibrocompation, and layer compaction operations. Additionally, once the OCSP sheet piling installation is complete, the precast transition and face beam/bullrail will be installed and the temporary work trestle will be removed. The MEP, concrete pads, and surfacing will be installed to complete the OCSP.

PPM will sequence all work activities to avoid any disruptions to the 2024 Cruise Ship Season.

APPENDIX Identification of Projects Table

14 1 A A A

10.5

. .

a. Name of Project	b. Owner/ Customer	c. Location of Project	d. Description	e. Project Description	f. Name of Key Team Members	g. Contract Price	h. Contract Dates	i. Recordable Injuries	j. Project Contact
Seward Passenger Dock Replacement	Alaska Railroad Corporation	Seward Cruise Ship Terminal Seward, AK 99664	1. Delivery Method: Progressive Design-Build 2. Integration of Design and Construction: PPM was selected with PND to advance from concept through 35%, 65%, and IFC 3. Role: Prime Contractor (PD-B Contractor)	Develop a design, and permit and construct a new 120 x 1200 passenger dock expected to be constructed as a retained fill structure in the ARRC Reserve Port.	Chris Willis - Lead Estimator Chris Lundfelt - Superintendent	Initial Contract Price - \$1,801,409 (design services) Final Contract Price - In Progress (estimated construction cost above \$56M)	Contract Date for Completion - 04/2025 Substantial Completion - 04/2024 Actual Date of Completion - In Progress	None (0)	Elizabeth Greer 327 W Ship Creek Ave Anchorage, AK 99501 greere@akrr.com (907) 261-6750
Dakota Creek Industries Shipyard Redevelopment	Dakota Creek Industries	Dakota Creek Industries Commercial Ave, Anacortes, WA 98221	 Delivery Method: Progressive Design-Build Integration of Design and Construction: PPM was selected on a cost-plus basis with design- build elements performed by PND to develop the design and bring to final construction Role: Prime Contractor (D-B Contractor) 	Major redevelopment and expansion involving demolition; dredging; a 450-LF OPEN CELL SHEET PILE Bulkhead; and a 370 LF long, 14,800 SF pile-supported pier. 36,000 CY of contaminated sediments and 105,000 CY of non-contaminated sediments were dredged.	N/A	Initial Contract Price - \$13,003,029 Final Contract Price - \$14,103,505	Contract Date for Completion - 01/2010 Actual Date of Completion - 01/2010	None (0)	Mike Nelson 820 4th St. Anacortes, WA 98221 mike@ dakotacreek.com (360) 293-9575
Hebgen Dam Cellular Cofferdam	NorthWestern Energy	Madison River West Yellowstone, MT 59758	 Delivery Method: Design-Build Integration of Design and Construction: PPM worked with the designer of record to develop plans and successfully complete construction Role: Prime Contractor (D-B Contractor) 	Closed cell cofferdam with tensioned rock anchors. 36-in diameter shafts installed 20-ft into bedrock. 1,350 CY tremie concrete poured in water depths up to 60-ft. Circular cells spaced 58-ft in diameter with 70-ft centers using sheet pile between 80 and 85-ft.	Chris Willis - Lead Estimator / Project Manager	Initial Contract Price - \$6,785,261 Final Contract Price - \$10,257,047	Contract Date for Completion - 12/2009 Actual Date of Completion - 09/2010 Owner-initiated changes extending the work into a second season	None (0)	BJ Cope 40 E. Broadway St. Butte, MT 59701 bj.cope@ northwestern. com (406) 581-6355
Palmer Station Pier Replacement	National Science Foundation / Leidos, Inc Pacifie ile&Ma	Palmer Station Anvers, Island	1. Delivery Method: CM/GC [pre- construction] - Lump Sum with Provisional Sums [construction] 2. Integration of Design and Construction: PPM worked with designer of record R&M to progress from concept to IFC to final construction 3. Role: Prime Contractor (CM/GC Contractor)	Preconstruction and construction services to demolish and replace the existing sheet pile bulkhead pier with a pier consisting of 36-in and 32-in steel pile. Site consists of exposed bedrock requiring piling to be drilled and socketed with 20-30- ft of embedment. Included upland civil earthworks. The site required significant environmental protections	Chris Willis - Lead Estimator / PD-B Director Chris Lundfelt - Superintendent Matt Rolf - Safety Director	Initial Contract Price - \$28,881,612 Final Contract Price - \$33,290,339	Contract Date for Completion - 08/2022 Actual Date of Completion - 07/2022	None (0)	Chris Chuhran 7400 S. Tucson Way Centennial, CO 80112 christopher. Chuhran. contractor@ usap.gov (253) 229-1289



KEY PERSONNEL RESUME SECTION 00 22 10 - ATTACHMENT 4

Name:	Aaron Atha	anas, P.E.					
Assignment and Role on this Project:	Mechanica	l Engineer					
Name of Firm:	Great North	hern Enginee	ering				
No. of Years: With this Firm	10	With other	r Firms:	12			
Education: Degree(s)/Year/Specialization	BSME/200	1/Mechanica	al/Process	Engineering			
Degree:	Mechanical Engineering						
Institution:	University	of Alaska Fa	airbanks				
Location:	Fairbanks,	Alaska					
Year:	2001						
Degree:	BSME						
Active Registration, if any:	Mechanica	l Engineer, N	ME11216				
	State	AK	No.	ME11216	Exp.	12/23	
	State LA No. 0045559 Exp. 09/23						

Instructions: Please fill out all fields in table and supply relevant project history (no more than 2 pages). Send a headshot as email attachment to <u>russell@trd-enterprises.com</u> along with completed form.

Brief Bio:

Aaron Athanas has over 20 years of experience in mechanical engineering, with roughly 10 of those years in the oil and gas sector working on downstream facilities including oil and gas processing, transfer, and storage, and 10 additional years in the Alaskan energy sector working the bulk storage and transfer of distillates, aromatics, natural gas, and LNG products. He has a wide range of experience with all of the relevant Alaskan codes, regulations, and requirements to provide fit for purpose designs that meet and exceed the owner's requirements whether they are a government or private entity. He specializes in bulk fuel storage, above and below ground pipelines, pump transfer and filtrations systems, marine loading/unloading systems. His knowledge of the arctic environment and how fuel storage and transfer systems interact with these environments is invaluable for project constructability and sustainability.

Relevant Experience:

Project Name: Port of Alaska Modernization Program, Petroleum and Cement Terminal (PCT), Phase 1

Project Role: Mechanical Designer of Record (DOR), Project Manager

Period of Performance: 2017-Present

Project Description: The PCT is the first phase of the Port of Alaska Modernization Program (PAMP) The PCT project includes a new PCT terminal with a new operations building, a new Hybrid POL Loading/Unloading Tower with 6 loading arms, integration of the cement unloading system, over 12,000 feet of new 12" diameter fuel transfer piping which has been integrated into the existing POA valve yard and piping system, and a completely upgraded electrical and controls system. This project budget was approximately 100M. This project has been designed and constructed while the POA has remained operational.

Project Name: Pertostar Fairbanks Rail and Truck Terminal

Project Role: Mechanical Designer of Record (DOR), Project Manager

Period of Performance: 2017-2019

Project Description: Aaron was the PM/DOR for PSI's green field effort to design and construct a new bulk fuel storage facility with both rail and truck loading and unloading facilities. The challenging and fast track design involved every aspect of the design, engineering, permitting, and construction support phases. The project was completed on time and within the budget and has since allowed GNE to be PSI's on call engineering support for many of their facilities. This project was approximately 40M. GNE has completed many other design projects since this project has been complete. This relationship continues to this day.

Project Name: Adak Island Fuel Skid, Pipelines, and Tank Farm Upgrades

Project Role: Mechanical Engineer/QC

Period of Performance: 2007-Present

Project Description: GNE has been contracted by the Aleut Native Cooperation (ANC) to complete multiple bulk fuel storage/fuel transfer system upgrades, replacements, and repairs on Adak Island. GNE has conducted multiple site visits and has helped ANC repair and maintain the aging infrastructure in this extremely harsh environment. Projects include new truck loading/unloading racks, new fuel transfer pipelines, new pump houses, leak detection systems, controls and alarm upgrades, cathodic protection systems, and bulk storage tank system inspections/repairs. Specific scopes of work are listed below:

GNE provided the design and construction support for a new truck loading rack and fueling system for the Adak Fuel Facility. The work included: inspection and assessment of the existing facilities, all engineering disciplines, a design basis, IFC drawings and specifications, fabricator selection, QA/QC oversite, construction support, and final installation inspection.

GNE provided the design for two new fueling pipelines (approximately 1-mile in length, each), connecting the cutand-cover tank farm to the pump house and new fueling skid at the Adak Fuel Facility. The work included: inspection and assessment of the existing pipelines and pump house, right-of-way selection, IFC drawings and specifications, crossing designs, cathodic protection, earthquake and expansion considerations, contractor evaluation and selection, QA/QC oversite, construction support, and final installation inspection. GNE provided Site Assessment Report to ANC for the existing fuel system and tank farm at the Adak Fuel Facility. The report included all findings and recommendations for upgrades and repairs. They provided a design for upgrades to leak detection system at the Adak Fuel Facility. Upgrades recommended included the tank farm, liners, vaults, and pipelines. Provided a design for upgrades and an expansion to the Adak Fuel Facility cathodic protection system. The work included testing and assessment of the existing system, a repair plan, and an IFC design for the new system.

GNE completed and stamped the Spill Prevention, Control, and Countermeasure (SPCC) Report for the Adak Fuel Facility. They also provided the design for a high-level alarm system for the Adak Fuel Facility.

Project Name: Fairbanks Natural Gas LNG

Project Role: DOR, Mechanical Engineer/PM

Period of Performance: 2017-2019

Project Description: The Interior Gas Utility (IGU) owns and operates the natural gas utility in Fairbanks, Alaska and surrounding communities. IGU began the design for expansion of their storage and vaporization facilities in October 2017, primarily around the installation of a new 125,000 BBL (5-million gallon) liquefied natural gas (LNG) storage tank in west Fairbanks. IGU bought Fairbanks Natural Gas LLC (FNG), the prior utility owner/operator, and now relies on FNG to be the day-to-day operator of the utility under IGU's ownership and direction. Design Alaska was the Design Manager and Designer of Record for the LNG storage and vaporization system expansion under contract initially with FNG and then transitioning to IGU early in the design. GNE provided process, mechanical engineering, instrumentation and controls design and construction support for both the LNG loading/unloading rack, the facility piping, and the balance of plant equipment. The total cost of the project was approximately \$50M.

Project Name: Delta Western Petroleum Term Contract

Project Role: Mechanical Engineer/PM

Period of Performance: 2017-Present

Since 2017, GNE performs on call engineering and inspection services for Delta Western's bulk fuel facilities. Detailed scope development, design documents, and construction engineering support at approximately (12) Delta Western bulk fuel storage, truck loading/unlading, and marine loading/unloading facilities across Alaska. Design scope has included all engineering disciplines for primarily bulk fuel storage tanks, and typically includes design requests for marine headers, vapor combustion systems, rail and truck loading racks, and pipelines. Locations include: Dutch Harbor, POA, Sitka, Ward Cove, Juneau, Dillingham, False Pass, Naknek, Ketchikan, Haines, Fairbanks, Deadhorse, and Yakutat.

KEY PERSONNEL RESUME SECTION 00 22 10 - ATTACHMENT 4

Name:	Torsten Mayrberger, P.E., Ph.D.
Assignment and Role on this Project:	Principal Geotechnical Engineer
Name of Firm:	PND Engineers, Inc.
No. of Years: With this Firm	11 With other Firms: 7
Education: Degree(s)/Year/Specialization	B.S., 1999, Civil Eng.; M.S., 2001, Civil Eng. – Geotechnical (Civil)
	and Rock Mechanics (Mining); PhD, 2010, Civil Eng Geotechnical
Degree:	B.S., Civil Engineering
Institution:	University of Alaska Anchorage
Location:	Anchorage, AK
Year:	1999
Degree:	M.S., Civil Engineering - Geotechnical (Civil) and Rock Mechanics
	(Mining)
Institution:	Michigan Technological University
Location:	Houghton, MI
Year:	2001
Degree:	Ph.D., Civil Engineering - Geotechnical
Institution:	Michigan Technological University
Location:	Houghton, MI
Year:	2010
Active Registration, if any:	Professional Civil Engineer
	State AK No. 14702 Exp. 2021
Institution: Location: Year: Degree: Institution: Location: Year: Degree: Institution: Location: Year: Active Registration, if any:	Anchorage, AK 1999 M.S., Civil Engineering – Geotechnical (Civil) and Rock Mechanics (Mining) Michigan Technological University Houghton, MI 2001 Ph.D., Civil Engineering – Geotechnical Michigan Technological University Houghton, MI 2001 Ph.D., Civil Engineering – Geotechnical Michigan Technological University Houghton, MI 2010 Professional Civil Engineer State AK No. 14702 Exp. 2021

Instructions: Please fill out all fields in table and supply relevant project history (no more than 2 pages). Send a headshot as email attachment to <u>russell@trd-enterprises.com</u> along with completed form.

Brief Bio:

PND Principal Engineer Torsten Mayrberger, P.E., Ph.D., has been working in Alaska for more than 35 years, providing him with extensive collective knowledge of the conditions and challenges present throughout the state. Mr. Mayrberger has 18 years of geotechnical engineering experience. His project experience involves large, remote arctic and marine geotechnical investigations, as well as deep foundation design in non-permafrost and permafrost soils, marine environments, and rock mass structures. Mr. Mayrberger taught drilling and blasting courses for civil works based on 10 years of experience in the drilling and blasting trade before becoming an engineer. He currently supervises PND's AASHTO/ASTM-accredited Soils-Materials Laboratory and arctic cold room facility. His specialties include advanced triaxial testing and in-situ instrumentation.

Relevant Experience:

Project Name: Sand Point Dock Replacement | Sand Point, AK

Project Role: Lead Geotechnical Engineer

Period of Performance: 2016-2019

Project Description: Mr. Mayrberger managed numerical analysis of the existing and expanded causeway to evaluate total and differential settlement and stability. The slope stability analysis was performed using a combination of Ensoft LPile, Rocscience Slide, and Settle3D. This project replaced the city's 35-year-old, steel pile-supported dock, which was used for receipt of shipment of conventional and containerized cargo, as well as for landing Alaska State Ferry passengers and vehicles.

Project Name: Kodiak Pier III Replacement | Kodiak, AK

Project Role: Principal Geotechnical Engineer

Period of Performance: 2012-2015

Project Description: Mr. Mayrberger was responsible for field investigation for the on- and offshore exploration of the Kodiak Pier III cargo-handling dock. PND conducted laboratory testing of recovered soil and rock samples and produced a geotechnical data report. Eight marine- and barge-supported boreholes were advanced to 100 feet below mudline. In addition to the geotechnical investigation, PND provided master planning services to review options; performed concept engineering; conducted metocean studies at the exposed site; managed and oversaw wave tank testing to examine replacement alternatives; performed detailed design; and provided construction administration and quality assurance support for the project. The replacement structure is a 330-foot-long pile-supported pier supporting a modern 100-foot gauge container crane. Soils at the site are a deep layer of very soft soils requiring piles to be socketed into bedrock. The structure was designed to accommodate large container-handling forklifts with 100-ton axle loads. The lateral resistance system uses an innovative sheet-pile system to drag lateral loads into the fill behind the dock structure. Dolphin structures extend the dock to more than 600 feet.

Project Name: Chignik Bulkhead Dock | Chignik, AK

Project Role: Lead Geotechnical Engineer

Period of Performance: 2012-2017

Project Description: Mr. Mayrberger managed the geotechnical investigation and materials testing for the new bulkhead dock. On- and offshore drilling provided information for dock design and usage of an upland stockpile. Results from previous nearby drill holes were compared and integrated into the design dataset to improve the understanding of the site. Materials were tested in PND's AASHTO/ASTM-accredited Soils-Materials Laboratory. The all-tide, deep-draft dock serves local and regional industry and provides berthing for Alaska Marine Highway System vessels.

Project Name: Seward Marine Center Mooring Dolphins | Seward, AK

Project Role: Principal Geotechnical Engineer

Period of Performance: 2011

Project Description: PND was responsible for the structural design of two four-pile breasting dolphins to modify the existing Seward Marine Center dock. The improvements were needed for mooring the new Alaska Region Research Vessel, Sikuliaq. The vessel is 261 feet long and one of the most advanced university research vessels; it is capable of breaking ice 2½-feet thick. PND also provided access to the dolphins from the dock, replaced the timber fenders along the existing dock, and added a dry fireline along the dock. Mr. Mayrberger advanced four offshore boreholes to 120 feet below mudline. Samples were collected every 5 feet and later tested at PND's soils lab for index properties and strength using lab triaxial testing.

Project Name: APL Terminal One Dock Repairs and Expansion Project | Dutch Harbor, AK

Project Role: Geotechnical Engineer

Period of Performance: 2017-2020; 2020-Current

Project Description: Mr. Mayrberger assisted with the geotechnical investigation to aid dock repairs, dredging work, and the dock modification and expansion at the American President Lines, Ltd. (APL) Terminal One Dock. The expansion will include dock upgrades and replace the existing pile-supported dock with a modern, high-capacity sheet-pile bulkhead dock extending south from the existing sheet-pile bulkhead. Concurrent with the dock and pad expansion, a material source will be developed in the hillside adjacent to the APL Terminal Yard.

Project Name: Unalaska Marine Center Expansion | Unalaska, AK

Project Role: Geotechnical Engineer

Period of Performance: 2014-2018

Project Description: Mr. Mayrberger assisted with a soils lab analysis following the geotechnical investigation for the replacement of the existing Unalaska Marine Center docks at Positions III and IV with a new high-capacity bulkhead dock facility with expanded container crane capabilities. Mr. Mayrberger also provided construction administration support. The project provides 610 feet of new dock face with a minimum water depth of approximately 45 feet and replaced two aging, pile-supported structures with a high-capacity bulkhead dock. The new facility incorporated a curved crane rail that allows a container crane to traverse from dock positions VII through IV around the curve to III, providing 1,350 linear feet of dock face now serviced by container cranes. PND designed a quarry on city-owned land to provide an optional source for shot rock material. The quarry was located in an industrial area along the steep shoreline cliffs of Dutch Harbor across the road from the UMC container terminal and adjacent to several bulk fuel storage tanks. The quarry plan and specifications were developed for a total face height of 180 feet with benches every 30 feet vertically. An emphasis was placed on monitoring and protecting the nearby infrastructure during blasting activities. Provisions were made for rock bolt stabilization dependent on conditions encountered during construction. The quarry plan included detailing the overburden at the top of the quarry for stabilization, rock wall stability analyses, a rock catchment ditch at the base, and drainage system. Identification of rock quality, jointing, and potential yield based on visual inspections rather than actual laboratory testing required familiarity with the geology of the area and typical characteristics of the rock in the nearby quarry. The quarry had to be designed to reduce the impact on nearby facilities during mining operations.

Project Name: Togiak Multipurpose Dock | Togiak, AK

Project Role: Lead Geotechnical Engineer

Period of Performance: 2013

Project Description: Mr. Mayrberger managed a soils investigation to evaluate subsurface conditions in support of the foundation design for the dock facility. He developed and supervised a drilling program, oversaw lab testing and analysis, and characterized subsurface conditions. While on site in Togiak, PND also visited the Togiak rock quarry site to assess shot rock for use as rip-rap and armor stone or road base course material. PND designed a robust, high-capacity, low-maintenance marine facility designed for the harsh environmental and ice conditions of the site to include a 30-foot-high sheet-pile dock, fender piles and bull rail, two pedestal cranes with foundations, improvements to the existing access road, and a concrete boat ramp. The ramp was designed for landing craft use and as a boat launch ramp for the local community. Armor rock was designed around all exposed edges to provide robust erosion protection in the exposed marine environment featuring up to 4-foot seas.

Lutak Dock Replacement

Request for Proposals ("RFP") Design-Builder









Pacific Pile & Marine, LP 4753 West 80th Ave Anchorage, AK 99502 Chris Willis P 206.331.3873 chrisw@pacificpile.com

Clip

Price Proposal Financial Submission - July 15, 2022

ATTACHMENT A TO REQUEST FOR PROPOSALS HAINES BOROUGH LUTAK DOCK REPLACEMENT PRICE PROPOSAL FORM AND INSTRUCTIONS

I. INSTRUCTIONS

A. Design-Builder's Phase 1 Lump Sum for Overhead and Profit

The Design-Builder's Phase 1 Lump Sum for Overhead and Profit will, if agreed upon by the Owner, be inserted in Section 6.2.1 of the Progressive Design-Build Agreement between Owner and Design-Builder and should be based on the Phase 1 Not to Exceed Amount proposed in Section B below as well as the Phase 1 Level of Effort proposed pursuant to Section VI.B.3.c of the RFP. The parties will negotiate the Phase 1 Level of Effort, the Lump Sum for Overhead and Profit, and the Phase 1 Not to Exceed Amount after award. For scoring purposes only, Proposers should assume that the Cost of the Work for the Project is \$20,000,000.00.

B. Phase 1 Not to Exceed Amount

The proposed Phase 1 Not to Exceed Amount will be inserted into Section 6.6.1.2 of the Agreement. The Phase 1 Not to Exceed Amount will not be scored. However, if accepted by the Owner after negotiations, shall become binding on the successful Finalist, subject to the terms and conditions of the Contract Documents.

- a. The Proposed Phase 1 Not to Exceed Amount should include all compensation to the Design-Builder during the Phase 1 set forth in the Contract Documents, including but not limited to Exhibit C of the Agreement and proposed in the Phase 1 Level of Effort described in the Management Proposal.
- b. The Owner reserves the right to reconcile the various proposals received and also reserves the right to seek best and final proposals for the scope and the cost of the Phase 1 Services and the Phase 1 Not to Exceed Amount; however, by submitting the Phase 1 Not to Exceed Amount, the Finalist warrants the following:
 - That the Phase 1 Level of Effort described in the Management Proposal is sufficient for the Design Build Team to perform the Work described for Phase 1 in the Contract Documents and provide the Owner with the Phase 1 deliverables as set forth in the revised Exhibit C proposed by the Finalist.
 - ii. That the Phase 1 Not to Exceed Amount is sufficient to perform the Work described in the Phase 1 Level of Effort in the Management Proposal.

C. Hourly Rates

Finalists will provide the hourly rates for Key Team Members. The Hourly Rates are not scored but will be incorporated into the Design-Build Agreement as Exhibit D. Separate rates shall be submitted for preconstruction and construction services should they differ.

D. Scoring of Price Proposal

The Design-Builder's Price Proposal shall be scored as follows:

The Finalist with the lowest Price Proposal will receive all fifteen points. The remaining Finalists will receive a proportionate share of the fifteen points, based on the proportion that the Price Proposal for their proposals exceeds the lowest Price Proposal. The points will be rounded to the next lowest whole number. No partial points will be awarded By way of example, if the second low Finalist proposes a

Haines Lutak Dock Project Price Proposal Form Instructions Price Proposal that is fourteen percent higher than the lowest Price Proposal, the second low Finalist shall receive 17 of the 20 allotted points. Fourteen percent of 20 is 2.8. 20 minus 2.8 equals 17.2. 17 is the next lowest whole number.

II. PRICE PROPOSAL FORM

Pacific Pile & Marine, LP

Finalist Name

Having carefully examined the Request for Proposal (RFP) for Design-Build Services for the Haines Borough, Alaska Lutak Dock Replacement Project, issued <u>June 17, 2022</u>, and Addenda numbers <u>1</u> through <u>2</u>, and the Agreement, the undersigned Design-Builder proposes the following Commercial Terms for the Project:

A. Design-Builder Lump Sum for Overhead and Profit that will is proposed to be inserted into Section 6.2.1 of the Agreement: (\$ 204,097.32 _____) Two Hundred Four Thousand Ninety-Seven Dollars & Thirty Two cents

B. Phase 1 Not To Exceed Amount (not scored)

 The proposed Phase 1 Not to Exceed Amount is

 One Million Nine Hundred Four Thousand Nine Hundred Eight dollars & Thirty two cents

 dollars (\$ 1,904,908.32

Phase 1 NTE is based upon a cost of the work of \$20 million as per instructions. If actual cost of the work changes significantly, PPM reserves the right negotiate a revised NTE number for Phase 1 Level of Effort.

C. Key Team Member Hourly Rates (not scored)

Name	Position	Hourly Rate	Hourly Rate
The Hourly Rates for Key	Team Members are as	fol Paresonstruction	Construction

*Please see attached Hourly Rates

PROPOSAL GUARANTEE

The undersigned hereby agrees that this Proposal may be accepted by Haines Borough anytime within ninety (90) calendar days immediately following the date indicated herein below, and the undersigned further agrees to submit a fully executed Agreement prior to the issuance of the Notice to Proceed that includes the Commercial Terms proposed in this Price Proposal Form.

PROPOSAL FROM:

Pacific Pile & Marine, LP

(Finalist Firm Name)

07 / 15 / 2022

:

(Authorized Representative Signature and Date)

Chris Willis, Executive Vice President

(Representative's Printed Name and Title)

CONE33893

(State of Alaska Contractor's License No.)

Key Staff	Job Class/Title	Preconstruction Rate/Hr	Construction Services Rate/Hr
(PND) John DeMuth, PE, SE	Senior Engineer VII	\$225	\$225
(PND) Dick Somerville, PE	Senior Engineer VII	\$225	\$225
(PND) Torsten Mayrberger, PE	Senior Engineer VI	\$210	\$210
(PND) Mark Sams, PE, SE	Senior Engineer VI	\$210	\$210
(PND) Tyler Bradshaw, PE	Senior Engineer V	\$190	\$190
(PND) Sean Sjostedt, PE	Senior Engineer V	\$190	\$190
(PND) Brenna Hughes, MS, CH	Environmental Scientist III	\$140	\$140
(PND) Ian Brown, PLS	Senior Land Surveyor III	\$135	\$135
(Respec) Ben Haight, PE	Senior Electrical Engineer	\$225	\$225
(GNE) Aaron Athanas, PE	Senior Mechanical Engineer	\$225	\$225
(PPM) Chris Willis	Chief Estimator	**\$0	*N/A
(PPM) Katie Laborde	Estimating Manager	\$55	*N/A
(PPM) Steve Grayson	Scheduler / Senior Estimator	\$100	*N/A
(PPM) Cameron Martin	Estimator	\$55	*N/A
(PPM) Chris Lundfelt	Superintendent	\$100	*N/A
(PPM) Stewart Willis	Project Manager	\$100	*N/A

*N/A – During the construction phase, PPM's Field overhead staff will be a cost of the work and will be included the cost estimate for Phase 2 Construction.

**We have decided not to charge our Project Director to keep the cost of phase 1 as low as possible.

07/14/2022 22-039	16:33 Lutak Dock NTE	[
*** Katie LaBo Biditem	Description	BID TOTALS Status - Rnd	Quantity	Units	Unit Price	Bid Total
10 20 30 40 50 60 70	35% Estimate + Review 65% Estimate + Review 95% Estimate + Review Weekly meetings. Design and construction workshops GMP Pricing and Negotiations PND Design/Permitting/ Geotechenical (cost = 20 M		1.000 1.000 1.000 26.000 1.000 1.000 1.000	LS LS LS WEEK LS LS LS	28,761.60 30,889.60 29,657.60 694.40 13,708.80 18,928.00 1,764,908.32	28,761.60 30,889.60 29,657.60 18,054.40 13,708.80 18,928.00 1,764,908.32
			Bid Total	>		\$1,904,908.32
		u B				
		r. V				
		r.				
		4				
		ir Fr				
		1				

.

		ESTIMATE	RECAP - BID Q	UANTITIES		
		DIRECT	INDIRECT	TOTAL	% OF TO	TAL
	Labor	102,600.00		102,600.00	6.0	32%
	Burden				0.0	00%
	Lab+Bur	102,600.00		102,600.00	6.03	32%
	Perm Matl				0.0	00%
	Const Exp	22,400.00		22,400.00	1.3	17%
	Equipment				0.0	00%
	Subs	1,575,811.00		1,575,811.00	92.6	51%
	Other				0.0	00%
	Total Costs:	1,700,811.00		1,700,811.00	100.0	00%
	% of Total	100.000%	0.000%	100.000%		
Escalation on:	Labor	Burden	Perm Matl	Const Matl	Со Едр	Rented Eqp
	0	0	0	0	0	0
	100.00%	100.00%	100.00%	100.00 %	100.00 %	100.00%
	Еq Ор Ехр	Sub	Hauling	Misc2	Misc3 To	tal Escalation
	0	0	0	0	0	0
	100.00 %	100.00 %	100.00%	100.00%	100.00 %	100.00%

The Summary Process was last run 07/14/2022 at 3:38 PM

Markup on Resource Costs	204,097.32	12.0000%	
MARKUP TOTALS ===>	204,097.32	12.0000%	
COST + MARKUP>	\$1,904,908.32 (On Takeoff Quantity)	(% of costs)	
There * ARE NOT * closing accounts for this bid.			
		-Effect on Bid-	
Rounding difference:			
Unbalancing difference:			
From Cut&Add Sheet-costs:		(on Bid Quantity)	
From Cut&Add Sheet-markup:		(on Bid Quantity)	
ass Through Adjustments:		None	
Net Adjustments (to the balanced bid):		[or desired bid]	
BALANCED BID TOTAL	\$1,904,908.32		
DESIRED BID (if specified)			
BID TOTAL (on bid quantities)	\$1,904,908.32		
BID COSTS (on bid quantities)	\$1,700,811.00		
MARKUP (on bid quantities)	\$204,097.32	12.000%	
EXPECTED JOB VALUE (on takeoff quantities):	\$1,904,908.32		

Pacific Pile & Mari 22-039 *** Katie LaBorde	ne Lutal	k Dock NTI	E			07/14/2022	16:32
EXPECTE EXPECTE	ED COSTS ED MARKUI	(on takeof P (on take	f quantities): off quantities):	\$	1,700,811.00 \$204,097.32	12.000%	
Adjust to I	Bid Quantitie	s =	Y				
			On Takeoff Quantities				
Labor Hrs. (incl burg	. (MH/MHS) den) 1(1,948 02,600	0 0	1,948 102,600			
Labor (DA	Y/DAYS)	0	0	0			
(incl burg	den)	0	0	0			
Labor (Oth (incl burg	terUnits) den <u>)</u>	0	0	0			
Labor Burg	len	0	0	0			
Spread Ind Spread Add	irects on: dons&Bond (on:	Labor Cost Total Cost	Sp	read Markup on:	Markup%	
Markup on:	Labor 12.00%	Burden 12.00%	PermMatl 12.00%	CM 12,00%	CoEqp 12.00%	RentedEqp 12.00%	
	EOE 12.00%	Sub 12.00%	Hauling 12,00%	Misc2 12.00%	Misc3 12=00%	w; •	<u>, , ¹⁸ , , , , ,</u>
Key Indicators							
Balanced Markup 204,097.32		 	Total Labor 102,600.00	=	Balanced Ma 198.93%	rkup/Total Labor	
ndirect Cost		1	Direct Cost	=	Indirect Cost	/Direct Cost	
).00		1	1,700,811.00	=	0.00%	-	
fotal Company Eqp).00		+ +	Balanced Markup By B 204,097.32	id Qty = =	Co Equip + F 204,097.32	lee	
Co Equin + Fee		1	Total Cost By Bid Oty	=	Co Equip + f	ee	
204,097.32		/	1,700,811.00	=	12.00%		
Virant Manhours		+	Indirect Manhours	~	Total Man U	oure	
,948.00		+ +	0.00		1,948.00	0018	
Co Equip + Fee		/	Bid Total	J.	Sales %		
10 1907 1 221		'		-	0.11		

*

Pacific Pile & Marine 22-039 *** Katie LaBorde	Lutak Dock NTE		07/14/2022	16:32			
ESTIMATE NO	DTES:						
Bid Date:	07/15/2022	Owner:					
Engr Firm:							
Estimator-In-Charge:		Desired Bid (if specified) =	0.00				
Notes:	UPDATED 8/12/2020						
	Rates valid 6/1/20 - 5/31/21						

	Source estimate used: C:\HEAVYBID\EST\PPM_JZ_2018						
	**********Estimate created on: 07/06/2022 by User#: 1 - Chris Willis						
	Source estimate used: L:\HEAVYBID\EST\PPM WA						
Last Summary on 07/1	4/2022 at 3:38 PM.	—					
T 1 1 0 1 1 0 1 1 1 0							

ļ

Last Spread on 07/14/2022 at 3:38 PM.
Pacific Pile & Marine 22-039 Lutak Dock NTE Katie LaBorde

ZCE

=> PPM Chief Estimator 1.00

16.00 MH

0.000

Direct Cost Report

Activity Resource	Desc	Pcs	Quantity Unit	U: Co	nit ost Labor	Perm Material	Constr Matl/Exp	Equip Sub- Ment Contract	Total
BID ITEM Description =	= 10 CLIER 35% Estimate + Review	NT#= 10	0	Unit =	LS Takeoff	'Quan:	1.000	Engr Quan:	1.000
10.01	Site Visit @ 35%			Quan: 1	.00 LS H1	·s/Shft;	8.00 Cal:	508 WC: NWC)
3AF 3HOTEL ZCF	Air Fare to Haines Hotel Costs	1.00 1.00	2.00 EA 4.00 DAY	1,000.0 200.0	00 00		2,000 800		2,000 800
ZSUP \$4,400.00	=> superitnendat 32.0000 MH/L	1.00 1.00 S	16.00 MH 16.00 MH 32.00 MH	100.0 [1600)0 1,600)] 1,600	I	2,800		1,600 4,400
10.02	35% Estimate			Quan: 1	.00 LS H1	s/Shft:	8.00 Cal: :	508 WC: NWC	94419
ZCE ZEM ZES ZSE ZSUP \$14,600.00	 PPM Chief Estimator PPM Estimating Mana PPM Estimator PPM Senior Estimator superitnendat 280.0000 MH/L3 	1.00 1.00 1.00 1.00 1.00 S	80.00 MH 40.00 MH 80.00 MH 40.00 MH 40.00 MH 280.00 MH	0.0 55.0 55.0 100.0 100.0 [14600	00 00 2,200 00 4,400 00 4,000 00 4,000 01 14,600				2,200 4,400 4,000 4,000 14,600
10.03	35% Review	637. SS38	大学之中心	Quan: 1	00 LS Hr	s/Shft;	8.00 Cal: :	508 WC: NWC	0.00
3AF 3HOTEL ZCE ZES ZSUP S6 680 00	Air Fare to Haines Hotel Costs —> PPM Chief Estimator —> PPM Estimator —> superitnendat 48,0000 MH/1.5	1.00 1.00 1.00 1.00 1.00	3.00 EA 6.00 DAY 16.00 MH 16.00 MH 16.00 MH 48.00 MH	1,000.00 200.00 55.00 100.00	00 00 00 880 00 1,600 11 2,480		3,000 1,200 4,200		3,000 1,200 880 1,600 6,680
	Totals: 10 -	35% Est	timate + Review		<u>``````</u> `````			<u></u>	<u> </u>
\$25,680.00 25,680.000	360.0000 MH/LS 1 LS		360,00 MH	[18680] 18,680 18,680.00		7,000 7,000.00		25,680 25,680.00
BID ITEM = Description =	= 20 CLIEN 65% Estimate + Review	/T#= 20	000	Unit = L	S Takeoff	Quan:	1.000	Engr Quan:	1.000
20.01	Site Visit @ 65%		an a	Quan: 1	00 LS Hr	s/Shft:	8.00 Cal: 5	08 WC: NWC) ()
3AF 3HOTEL ZSE	Air Fare to Haines Hotel Costs > PPM Senior Estimator	1.00 1.00 1.00	1.00 EA 2.00 DAY 16.00 MH	1,000.00 200.00 100.00	0 0 0 1,600		1,000 400		1,000 400 1,600
\$3,000.00	16.0000 MH/LS	3	16.00 MH	[1600] 1,600	01.64	1,400	Ad ANICE NAMO	3,000
20.02	05% Estimate	1.00	100.00 MII	Quan: 1	00 LS Ar	<u>s/5ntt:</u>	<u>8.00 Cal:</u> 2	UO WC: NYC	**Unreviewed
ZEM	=> PPM Estimating Mana	1.00	60.00 MH	55.00	0 3,300				3,300
ZES ZSE	=> PPM Estimator => PPM Senior Estimator	1.00	120.00 MH 40.00 MH	55.00 100.00	0 6,000 0 4,000				6,600 4,000
ZSUP \$17,900.00	==> superitnendat 380.0000 MH/LS	1.00	40.00 MH 380.00 MH	100.00 [17900	0 4,000] 17,900				4, 000 17,900
20.03	65% Review			Quan: 1,	00 LS Hr	s/Shft:	8.00 Cal: 5	08 WC: NWC	
3AF 3HOTEL	Air Fare to Haines Hotel Costs	1.00 1.00	3.00 EA 6.00 DAY	1,000.00 200.00	0		3,000 1,200		**Unreviewed 3,000 1,200

Pacific Pile & Marine 22-039 Lutak Dock NTE Katie LaBorde

Activity Resource	Desc	Pcs	Quantity Unit	Unit Cost	t Pe : Labor Mater	rın Constr ial Matl/Exp	Equip Sub- Ment Contract	- t Total
BID ITEM Description =	= 20 CL 65% Estimate + Review	.IENT# = 2	20	Unit= LS	Takeoff Quan	1.000	Enor Quan:	1.000
ZEQ	DDM Estimator	1.00	16.00 МЛ	6 mt 25		1.000	Liigi Quali,	1.000
ZES ZSUP \$6,680.00	=> PPM Estimator => superitnendat 48.0000 MI	1.00 1.00 H/LS	16.00 MH 16.00 MH 48.00 MH	100.000 [2480]	1,600 2,480	4,200		880 1,600 6,680
<pre>\$27,580.00 27,580.000</pre>	n Totals: 20 444.0000 MH/LS 1 LS	- 65% F	Estimate + Review 444.00 MH	[21980]	21,980 21,980.00	5,600 5,600.00		27,580 27,580.00
BID ITEM Description =	= 30 CL 95% Estimate + Review	JENT#= 3	30	Unit = LS	Takeoff Quan:	1.000	Engr Quan;	1.000
30.01	Site Visit @ 95%	1 K-1993 6406 2200 - 8003		Quan: 1.0	0 LS Hrs/Shft;	8.00 Cal: ;	508 WC; NWC	
3AF	Air Fare to Haines	1.00	1,00 EA	1,000.000		1,000		1,000
3HOTEL	Hotel Costs	1.00	2.00 DAY	200.000		400		400
ZSUP	=> superitnendat	1.00	16.00 MH	100.000	1,600	1 400		1,600
\$3,000.00	10,0000 MI	1/Lo	10.00 MH	[1000]	1,000	1,400		3,000
30.02	95% Estimate	yang ayang di		Quan: 1.0) LS (Hrs/Shft:	8.00 Cal: ;	508 WC: NWC	
ZCE	=> PPM Chief Estimate	or 1.00	120.00 MH	0.000				
ZEM	==> PPM Estimating Ma	ana 1.00	40.00 MH	55.000	2,200			2,200
ZES ZSF	==> PPM Estimator ==> PPM Senior Estimat	1.00 for 1.00	40.00 MH	100.000	4 000			4 000
ZSUP	=> superitnendat	1.00	40,00 MH	100.000	4,000			4,000
\$16,800.00		IALS		[16800]	16,800	<u> </u>	<u></u>	16,800
30.03	95% Review			Quan: 1.00) LS Hrs/Shft:	8.00 Cal: 5	08 WC: NWC	a Sate Sate Sate
3AF	Air Fare to Haines	1.00	3.00 EA	1,000,000		3,000		3.000
3HOTEL	Hotel Costs	1.00	6.00 DAY	200.000		1,200		1,200
ZCE	=> PPM Chief Estimate	or 1.00	16.00 MH	0.000				
ZES	=> PPM Estimator	1.00	16.00 MH	55.000	880			880
2SUP \$6,680.00	==> superimendat 48,0000 MH	1.00 I/LS	48.00 MH	[2480]	2,480	4,200		1,600 6,680
[tom	Totals: 30	050/ F	stimato ± Doviow	-				
\$26,480.000 26,480.000	424.0000 MH/LS 1 LS	- 7370 8	424.00 MH	[20880]	20,880 20,880.00	5,600 5,600.00		26,480 26,480.00
BID ITEM =	= 40 CL	IENT#= 4	0	I loit WEEV	Takaafi Quan	26,000	Enge Quan	26.000
Description -	weekiy meetings.				Takeon Quan:	20.000	Engr Quan.	20.000
(35.01	Weekly meetings		AND GLENCETING	Quan: 26.00	WEE Hrs/Shft;	8.00 Cal: 5	08 WC:NWC	Siney oc.
2 hrs per m	eeting		52 00 MIJ	0.000				
ZEM	=> PPM Estimating Ma	na 1.00	52.00 MH	55 000	2,860			2 860
ZES	==> PPM Estimator	1.00	52.00 MH	55.000	2,860			2,860
ZSE	=> PPM Senior Estimat	or 1.00	52.00 MH	100.000	5,200			5,200
ZSUP	=> superitnendat	1.00	52.00 MH	100.000	5,200			5,200
\$16,120.00	10.0000 MH	/WEEK	260.00 MH	[620]	16,120			16,120
> Item	Totals: 40	- Weekly	meetings.	-				

16:34

.

Pacific Pile & Marine 22-039 Lutak Dock NTE Katie LaBorde

Direct Cost Report

	Page 3
07/14/2022	16:34

:

Kesource	Desc	Pcs	Quantity U	nit	Unit Cost	Perm Labor Material	Constr Matl/Exp	Equip Sub- Ment Contract	Total
BID ITEM Description =	= 40 CLIF Weekly meetings.	ENT# = 4	10	Unit =	- WEEK	Takeoff Ouan:	26.000	Engr Ouan:	26.000
\$16.120.00	10.0000 MH/WEEK		260.00 M	IH	[620]	16.120		8- <	16.120
620.000	26 WEB	к 				620.00			620.00
BID ITEM	= 50 CLIE	ENT#= 5	50	Unit -	= TS	Takeoff Quan:	1 000	Engr Quan	1.000
Description -	Dosign and consultation we	лкопора		Ohn -		Takeon Quan,	1.000	Engr Quan.	1,000
37.10	Scheduling workshop		<u>n haanda kaana kaan Ana kaana k</u>	Quan	: 1,00) I.S Hrs/Shft;	8.00 Cal: 5	08 WC: NWC	
ZCE	==> PPM Chief Estimator	1.00	8.00 M	IH	0.000				
ZEM	==> PPM Estimating Man	a 1.00	8.00 M	H	55.000	440			440
ZSUP	=> superimendat	1.00	8.00 M	H	100.000	800			800 800
\$2,040.00	32.0000 MH/I	LS	32.00 M	Ή	[2040]	2,040			2,040
37.11	Permitting workshop			Quan	1.00	LS Hrs/Shft:	8.00 Cal: 5	08 WC; NWC	
ZCE	==> PPM Chief Estimator	1.00	8.00 M	Н	0,000				
ZEM	=> PPM Estimating Man	a 1.00	8.00 M	Н	55.000	440			440
ZSE	==> PPM Senior Estimator	r 1.00	8.00 M	Н	100.000	800			800
ZSUP	==> superitnendat	1.00	8.00 M	H	100.000	800			800
\$2.040.00	32 0000 MH/I	N		н	1 2040 1	2 040			2 0/0
\$2,040.00	32.0000 MH/I	د. 1970 (1970)	32.00 M	H	[2040]	2,040			2,040
\$2,040.00	32.0000 MH/l Bulkhead design worksh	_S ops	32.00 M	H Quan	[2040] : 1.00	2,040	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE	32.0000 MH/I Bulkhead design_worksh =>> PPM Chief Estimator	_S 0 PS 1.00	32.00 M	H Quan H	[2040] 	2,040 /LS_Hrs/Shft:	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM	32.0000 MH/I Bulkhead design_worksh	_S 0 ps 1.00 1_1.00	32.00 M 16.00 M 16.00 M	H Quan H H	[2040] 	2,040	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSLP	32.0000 MH/J Bulkhead design_worksh		32.00 M 16.00 M 16.00 M 16.00 M	H Quan H H H	[2040] 	2,040 LS Hrs/Shft: 880 1,600	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00	32.0000 MH/J Bulkhead design_worksh	1.00 1.00 1.00 1.00 1.00 .S	32,00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M	н Оцал Н Н Н Н Н	[2040] 0.000 55.000 100.000 100.000 [4080]	2,040 <u>LS</u> <u>Hrs/Shft:</u> 880 1,600 1,600 4,080	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM- ZSE ZSUP \$4,080.00 37.13	32.0000 MH/I Bulkhead design_worksh	S 	32,00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M	H Quan H H H H H M Ouan	[2040] 0.000 55.000 100.000 [4080] 1.00	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft:	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE	32.0000 MH/I Bulkhead design, worksh => PPM Chief Estimator => PPM Senior Estimator => superitnendat 64.0000 MH/I Dolphin workshop => PPM Chief Estimator	1.00 1.00 1.00 1.00 1.00 .S 1.00	32,00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M	H Quan H H H H M Ouan H	[2040] . 1.00 0.000 55.000 100.000 [4080] 1.00 0.000	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft:	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM	32.0000 MH/I Bulkhead design_worksh => PPM Chief Estimator => PPM Estimating Mana => PPM Senior Estimator => superitnendat 64.0000 MH/I Dolphin workshop => PPM Chief Estimator => PPM Estimating Mana	1.00 1.00 1.00 1.00 1.00 .5 1.00 1.00 1.00	32.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M	н Оцал Н Н Н Н Н Н Н Н Н Н	[2040] 	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE	32.0000 MH/I Bulkhead design_worksh PPM Chief Estimator PPM Senior Estimator PPM Senior Estimator superitnendat 64.0000 MH/I Dolphin workshop PPM Chief Estimator PPM Estimating Mana PPM Senior Estimator	1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00	32.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M 8.00 M	H Quan H H H H M Ouan H H H	[2040] . 1.00 0.000 55.000 100.000 [4080] . 1.00 0.000 55.000 100.000	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440 800	8.00 Cal: 5	08 WC: NWC	2,040 880
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP 2SE ZSUP	32.0000 MH/I Bulkhead design_worksh Second	1.00 1.00 1.00 5 1.00 1.00 5 1.00 1.00 1.00 1.00 1.00 1.00	32,00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M 8.00 M 8.00 M	H Quan H H H H H H H H H H	[2040] 0.000 55,000 100,000 100,000 [4080] 1.00 0.000 55,000 100,000 100,000 100,000	2,040 LS Hrs/Shft: 	8.00 Cal: 5	08 WC: NWC	2,040 <u>880</u>
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00	32.0000 MH/I Bulkhead design_worksh Second	1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 S	32,00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M	H Quan H H H H H H H H H H H H H	[2040] 	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440 800 800 2,040	8.00 Cal: 5	08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00 37.14	32.0000 MH/I Bulkhead design_worksh Second		32.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M	Н Оцал Н Н Н Н Н Н Н Н Н Н Ц Н Ц Ц Ц Ц	[2040] . 1.00 0.000 55.000 100.000 [4080] . 1.00 0.000 55.000 100.000 100.000 [2040] 1.00	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440 800 800 2,040 L Hrs/Shft:	8.00 Cal: 5 8.00 Cal: 5 8.00 Cal: 5	08 WC: NWC 08 WC: NWC	2,040 880
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE	32.0000 MH/I Bulkhead design_worksh => PPM Chief Estimator => PPM Estimating Mana => PPM Senior Estimator => superitnendat 64.0000 MH/I Dolphin workshop => PPM Chief Estimator => PPM Senior Estimator => PPM Senior Estimator => Superitnendat 32.0000 MH/I Revement and landing wo => PPM Chief Estimator	1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 S rkshop 1.00	32,00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M	H Quan H H H H H H H H H H H H H H H H H H H	[2040] 	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440 800 800 2,040 L Hrs/Shft:	8.00 Cal: 5	08 WC: NWC	2,040 <u>880</u> 1,600 1,600 4,080 440 800 800 2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM 2CE ZEM	32.0000 MH/I Bulkhead design_worksh => PPM Chief Estimator => PPM Estimating Mana => PPM Senior Estimator => superitnendat 64.0000 MH/I Dolphin workshop => PPM Chief Estimator => PPM Senior Estimator => superitnendat 32.0000 MH/I Revement and landing wo => PPM Chief Estimator => PPM Chief Estimator => PPM Chief Estimator => PPM Chief Estimator => PPM Chief Estimator	1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00	32.00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M	Н Оцал Н Н Н Н Н Н Н Н Н Н Н Н Н Н	[2040] 	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440 800 2,040 L Hrs/Shft: 440 200	8.00 Cal: 5	08 WC: NWC 08 WC: NWC	2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00	32.0000 MH/I Bulkhead design_worksh PPM Chief Estimator PPM Senior Estimator Superimendat 64.0000 MH/I Dolphin workshop PPM Chief Estimator PPM Senior Estimator PPM Senior Estimator PPM Senior Estimator PPM Chief Estimator PPM Chief Estimator PPM Chief Estimator PPM Chief Estimator PPM Senior Estimator PPM Senior Estimator PPM Senior Estimator PPM Senior Estimator PPM Senior Estimator PPM Senior Estimator	1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00	32.00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M 8.00 M	H Quan H H H H H H H H H H H H H H H H H H H	[2040] . 1.00 0.000 55.000 100.000 [4080] . 1.00 0.000 55.000 100.000 [2040] . 1.00 0.000 55.000 100.000 [2040] . 1.00 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.00000 0.00000 0.00000 0.00000 0.00000 0.00000 0.000000 0.00000000	2,040 LS Hrs/Shft:	8.00 Cal: 5	08 WC: NWC 08 WC: NWC	2,040 <u>880</u> 1,600 1,600 4,080 440 800 800 2,040 440 800 800 800
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00	32.0000 MH/I Bulkhead design_worksh > PPM Chief Estimator > PPM Senior Estimator > PPM Senior Estimator > superitnendat 64.0000 MH/I Dolphin workshop > PPM Chief Estimator > PPM Senior Estimator > superitnendat 32.0000 MH/I Revement and landing wo > PPM Chief Estimator > PPM Chief Estimator > PPM Chief Estimator > PPM Senior Estimator > PPM Senior Estimator > PPM Senior Estimator > Superitnendat > Superitnendat > Superitnendat > Superitnendat > Superitnendat > Superitnendat > Superitnendat	1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00	32,00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M	Н Quan Н Н Н Н Н Н Н Н Н Н Н Н Н	[2040] 	2,040 LS Hrs/Shft:	8.00 Cali 5	08 WC: NWC	2,040 <u>880</u> <u>1,600</u> <u>1,600</u> <u>4,080</u> <u>440</u> <u>800</u> <u>800</u> 2,040 <u>440</u> <u>800</u> <u>800</u> 2,040
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14	32.0000 MH/I Bulkhead design_worksh PPM Chief Estimator PPM Senior Estimator Superinendat 64.0000 MH/I Dolphin workshop. PPM Chief Estimator PPM Senior Estimator PPM Senior Estimator PPM Senior Estimator Superinendat 32.0000 MH/I Revement and landing wo PPM Senior Estimator PPM Senior Estimat	2,5 005 1,00 1,00 1,00 5 1,00 1,00 1,00 1,00 1,00 1,00 1,00 1,00 1,00 1,00 1,00 1,00 1,00	32.00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M	H Quan H H H H H H H H H H H H H H H H H H H	[2040] . 1.00 0.000 55.000 100.000 [4080] . 1.00 0.000 55.000 100.000 [2040] . 1.00 0.000 55.000 100.000 [2040] 	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440 800 2,040 L Hrs/Shft: 440 800 2,040	8.00 Cal: 5	08 WC: NWC 08 WC: NWC 08 WC: NWC	2,040 880
\$2,040.00 37.12 ZCE ZEM ZSE ZSUP \$4,080.00 37.13 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZEM ZSE ZSUP \$2,040.00 37.14 ZCE ZSE ZSUP \$2,040.00 37.14 ZCE ZSE ZSUP \$2,040.00 37.14 ZCE ZSE ZSUP \$2,040.00 37.14 ZCE ZSE ZSUP \$2,040.00 37.14 ZCE ZSE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 37.14 ZCE ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP \$2,040.00 20 ZSUP ZSUP ZSUP ZSUP ZSUP ZSUP ZSUP ZSUP ZSUP ZSUP	32.0000 MH/I Bulkhead design_worksh => PPM Chief Estimator => PPM Estimating Mana => PPM Senior Estimator => superitnendat 64.0000 MH/I Dolphin workshop. => PPM Chief Estimator => PPM Senior Estimator => superitnendat 32.0000 MH/I Revement and landing wo => PPM Chief Estimator => PPM Chief Estimator => PPM Chief Estimator => PPM Chief Estimator => PPM Senior Estimator => superitnendat 32.0000 MH/L n Totals: 50 - 192.0000 MH/LS	1.00 1.00 1.00 5 1.00 5 1.00	32.00 M 16.00 M 16.00 M 16.00 M 16.00 M 64.00 M 8.00 M	H Quan H H H H H H H H H H H H H H H H H H H	[2040] . 1.00 0.000 55.000 100.000 [4080] . 1.00 0.000 55.000 100.000 [2040] . 1.00 0.000 55.000 100.000 [2040] . 12240]	2,040 LS Hrs/Shft: 880 1,600 1,600 4,080 LS Hrs/Shft: 440 800 2,040 L Hrs/Shft: 440 800 2,040 12,240	8.00 Cali 5	08 WC: NWC	2,040 880

CLIENT#= 60 BID ITEM = 60 Description = GMP Pricing and Negotiations

Unit =

LS Takeoff Quan:

1.000 Engr Quan: 1,000

Pacific Pile & Marine 22-039 Lutak Dock NTE Katie LaBorde * . . .

· 2

Activity Resource	Desc	Pcs	Quantity Unit		Unit Cost	Perm Labor Material	Constr Matl/Exp	Equip Sub- Ment Contract	Total
BID ITEM Description =	= 60 CLl GMP Pricing and Negotia	ENT# = 6 tions	0	Unit =	LS	Takeoff Quan:	1.000	Engr Quan:	1.000
40.01	Travel to Haines fro GM	1P Negoti	itions	Quán:	1,00	LS Hrs/Shft:	8,00 Cal: 50	8 WC: NWC	
3AF 3HOTEL ZCE ZSE ZSUP S7 400 00	Air Fare to Haines Hotel Costs —> PPM Chief Estimato —> PPM Senior Estimat => superitnendat 48 0000 MH	1.00 1.00 r 1.00 or 1.00 1.00	3.00 EA 6.00 DAY 16.00 MH 16.00 MH 16.00 MH 48.00 MH	1,(2 1 1	000.000 200.000 0.000 100.000 100.000	1,600 1,600 3,200	3,000 1,200		3,000 1,200 1,600 1,600 7,400
40.02	GMP Final Pricing	ш	43.00 000	Ouan:	1.00	LS Hrs/Shft:	8.00 Cal: 50	8 WC: NWC	,,+00
ZCE ZEM ZES ZSE ZSUP \$7,300.00	 PPM Chief Estimato PPM Estimating Ma PPM Estimator PPM Senior Estimate superitnendat 140.0000 MH 	r 1.00 na 1.00 1.00 or 1.00 1.00 /LS	40.00 MH 20.00 MH 40.00 MH 20.00 MH 20.00 MH 140.00 MH] 1 [••••••	0.000 55.000 55.000 00.000 00.000 7300]	1,100 2,200 2,000 2,000 7,300	9 00 Cal+ 50	8 WG-NWO	1,100 2,200 2,000 2,000 7,300
ZCE ZEM \$2,200.00	PPM Chief Estimato PPM Estimating Mai 80.0000 MH	r 1.00 na 1.00 /LS	40.00 MH 40.00 MH 80.00 MH	[0.000 55.000 2200]	2,200 2,200			2,200 2,200
→ Item \$16,900.00 16,900.000	Totals: 60 268.0000 MH/LS 1 LS	- GMP P	ricing and Negotia 268.00 MH	tions []		12,700 2,700.00	4,2 00 4,2 00.00		16,900 16,900.00
BID ITEM = Description =	= 70 CLI PND Design/Permitting/ G	ENT# = 7(eotechcnic) al (c	Unit =	LS	Takeoff Quan:	1.000	Engr Quan:	1.000
(<u>50.01</u>	PND Design-35% to JFC	(NTE):	an a	Quan:	1.00	LS Hrs/Shft:	8.00 Cal: 50	8 WC: NWC) D
4PND	PND	1.00	1.00 LS	1,120,	820.000			1,120,820	1,120,820
50.02	PND- Environmental + J	Permitting	to IFC (Quan:	1.00	LS Hrs/Shft:	8.00 Cal: 50	8 WC: NWC	<u>(2004) (02</u>)
4PND	PND	1.00	1.00 LS	217,4	30.000			217,430	217,430
(<u>50.04</u>	PND Geotechnical / Geo	physical/]	opo and	Quan:	1.00	LS Hrs/Shft:	8.00 Cal: 50	8 WC: NWC)
4PND	PND	1.00	1.00 LS	237,5	61.000			237,561	237,561
Item \$1,575,811.00	Totals: 70 1 LS	- PND De	sign/Permitting/ G	eotechenical	(c			1,575,811 1,575,811.00	1,575,811 1,575,811.00
\$1,700,811.00	*** Report Totals **	**	1,948.00 MH			102,600	22,400	1,575,811	1,700,811

>>> indicates Non Additive Activity

-----Report Notes:-----

The estimate was prepared with TAKEOFF Quantities.

This report shows TAKEOFF Quantities with the resources.

Pacific Pile & Marine 22-039 Lutak Dock NTE Katie LaBorde

710 712

7days, 12 hrs a day

1

Activity Resource	Desc	Quanti Pcs	ty Unit	Unit Cost	Perm Labor Material	Constr Matl/Exp	Equip Sub- Ment Contract	Total
BID ITEM = Description =	= 70 PND Design/Pern	CLIENT# = 70 hitting/ Geotechenical (c	Unit =	LS	Takeoff Quan:	1.000	Engr Quan;	1.000
'Unreviewed' Ac	tivities are marked.							
Bid Date: 07/15/	22 Owner: Engine E	eering Firm: stimator-In-Charge:						
JOB NOTES UPDATED 8/1 Rates valid	2/2020 6/1/20 - 5/3 *Estimate cre	1/21 ated on: 05/02/2019	9 by User#: 0 -					
Source estim	mate used: C:	\HEAVYBID\EST\PPM_J	JZ_2018					
************ Source estin	*Estimate crea mate used: L:	ated on: 07/06/2022 \HEAVYBID\EST\PPM_W	2 by User#: 1 - MA	Chris Wi	illis			
* on units of MH [] in the Unit C In equipmen Calendar Co 508	l indicate average la cost Column = Labo nt resources, odes 5 days, 8 hrs	bor unit cost was used rathe r Unit Cost Without Labor I rent % and EOE % n	er than base rate. Burdens lot = 100% are b	represent	ed as XXX%YYY	where XX	X=Rent% and YY	Y=EOE %
510	5 dyas, 10 hr	s (Default Calendar)						
512	5 days a week	, 12 hr sper dy						
610	6 days a week	, 10 hrs per da						
710	7days, 10 hrs							

Alaska Business License # 914725

Alaska Department of Commerce, Community, and Economic Development

Division of Corporations, Business, and Professional Licensing PO Box 110806, Juneau, AK 99811-0806

This is to certify that

PACIFIC PILE & MARINE, L.P.

700 S RIVERSIDE DRIVE, SEATTLE, WA 98108

owned by

PACIFIC PILE & MARINE LIMITED PARTNERSHIP

is licensed by the department to conduct business for the period

March 10, 2022 to December 31, 2023 for the following line(s) of business:

23 - Construction



This license shall not be taken as permission to do business in the state without having complied with the other requirements of the laws of the State or of the United States.

This license must be posted in a conspicuous place at the business location. It is not transferable or assignable.

Julie Sande Commissioner PACIFIC PILE & MARINE, L.P. 700 S RIVERSIDE DRIVE SEATTLE, WA 98108



July 15, 2022

Haines Borough PO Box 1209 Haines, AK 99827

Re: Lutak Dock Replacement Progressive Design-Build Management Proposal

Dear Selection Committee Members,

Thank you for advancing the Western Marine Construction team to this phase of your selection process for the Lutak Dock Replacement project. We remain confident in our ability to fully support the Haines Borough in designing, permitting, and constructing a high-quality, cost-effective facility that serves local and regional interests for decades to come.

Our commitment to the Lutak Dock Replacement project will reflect the overarching philosophy of our firm: to cultivate trust and provide value at every step of the Progressive Design-Build (PDB) process. Relying heavily on insights gained during our work on the 2016 Haines Ferry Terminal Improvements (as well as our decades of other marine project experience), we will partner with the Haines Borough to develop a design and construct a facility the community is proud of.

Through implementation of our Public Involvement Plan (a draft of which we have included in the attached Management Proposal), our team will facilitate clear, continuous communication with user groups and other stakeholders to ensure the community understands not only what this project involves and accomplishes, but what it does not.

During our June 30 interactive interview, we asked what you see as the biggest risks and obstacles to this project's success. We heard your concerns, held a team meeting, and developed proactive strategies for addressing them, which we have detailed in this Management Proposal. If selected as your Design-Build Team, we will:

- ☑ Foster a collaborative, transparent partnership with the Haines Borough and other stakeholders to not only yield the best end product possible, but to avoid claims, contract disputes, and other disagreements.
- ☑ Maintain full functionality of dock facilities for its primary users (Alaska Marine Lines and Delta Western) during construction.
- \blacksquare Involve and inform the public to avoid misunderstanding of the project intent.
- ☑ Mitigate the cost and schedule impacts of structure/site unknowns.
- ☑ Evaluate how elements of this project intersect with other Borough goals and capture efficiencies where possible (for example, using the dock demolition disposal barge to remove junked cars or other local debris).

Western Marine Construction is adopting a partnership-based approach to this project, teaming with key technical service firms who bring specialized expertise and knowledge to the table. With its local office, Haines-based staff, and experience serving the Borough, our partner proHNS will be a valuable resource and fundamental to ensuring this project's success.

As requested in the RFP, the following is a list of all our proposed Design-Build Team Members, including contact information.

Haines Borough - Request for Proposals - Design Builder Lutak Dock Replacement Management Proposal

Kriss Hart - Project Executive/Contract Manager

Western Marine Construction 2775 Harbor Avenue SW Suite A Seattle, WA 98126 kriss@wmc2775.com

Patrick McHugh - Superintendent

Western Marine Construction 2775 Harbor Avenue SW Suite A Seattle, WA 98126 patrick@wmc2775.com

Ryan Bare - Environmental Manager

Rugged Coast Environmental 16200 Point Lena Loop Road Juneau, AK 99801 ruggedcoast.ev@gmail.com

Brad Ginn, PE - Marine Structures Design Lead

Art Anderson Associates 830 Pacific Avenue Bremerton, WA 98337 rginn@artanderson.com

Pat Gorman, PE - Electrical Design Lead

Gorman Engineers 10761 Horizon Drive Juneau, AK 99801 pgorman@gci.net

Julian Koerner, PE - Project Manager

Western Marine Construction 2775 Harbor Avenue SW Suite A Seattle, WA 98126 julian@wmc2775.com

Garret Gladsjo, PE - Design Manager

proHNS LLC 1945 Alex Holden Way #101 Juneau, AK 99801 garret@proHNS.com

Keith Mobley, PE, GE - Geotechnical Manager

Northern Geotechnical Engineering 11301 Olive Lane Anchorage, AK 99515 kmobley@nge-tft.com

Shane Hooten, PE - Fuel Systems Design Lead

Modern Mechanical 11001 Black Bear Road Juneau, AK 99801 shane@modern-mechanical.com

Kelly O'Neill, PLS - Surveyor

North 57 Land Surveying 8800 Glacier Hwy Suite 224 1/2 Juneau, AK 99801 north57landsurveying@yahoo.com

WMC and our team members have successfully worked together in various iterations. For example, on the Haines Ferry Terminal Improvements, WMC called on Northern Geotechnical Engineering to provide guidance when challenging pile driving conditions were encountered at the project site. With NGE's expertise, WMC was able to successfully install the piling without incurring significant damage to the piling or compiling significant cost overruns due to the unexpected driving conditions. As the Contractor on the Tenakee Ferry Terminal Improvements, WMC worked cooperatively with the proHNS Construction Administration and Inspection team to not only construct a quality facility, but to close the project out in record time.

Per the RFP, we have not included an Identification of Projects Table as we do not reference or cite any projects in this Management Proposal that were not listed in our previously submitted Statement of Qualifications.

Again, we appreciate being advanced to this stage in the selection process and you taking the time to review our Management Proposal. We strongly believe our team is the best fit for the Lutak Dock Replacement project and are confident we can successfully design, permit, and construct all three phases by December 2024 within the existing budget.

Sincerely,

kriss Hart

Kriss Hart, President Western Marine Construction

Table of Contents

1. Overall Management Approach	1
1a. Overall Management Approach to the Lutak Dock Replacement PDB Project	1
1b(i). Creating a Collaborative Environment and Exceeding Project Goal #2	3
1b(ii). Approach to Stakeholder Outreach and Incorporation of Input	4
2. Maximize Design within the Limited Budget	5
2a. Overall Approach to Exceeding Project Goal #1	5
2b. Strategies and Design Ideas for Exceeding Project Goal #1	5
2c. Challenges in Developing the Design and How to Address Them	7
2d. Communicating and Collaborating with Haines Borough and other Stakeholders	7
3. Project Controls, Cost Tracking, and GMP Development	8
3a. Strategies for Exceeding Project Goal #3	8
3b. Processes and Tools for Monitoring, Reporting, and Managing Cost	9
3b(i). Budget Control and Reporting Processes	9
3b(ii). Scope, Cost, and Schedule Baseline Development	9
3b(iii). Incorporating Input from Other Subcontractors	9
3b(iv). Primary Challenges in Establishing the GMP	9
3b(v). WMC's Differentiating Resources	9
3c. Phase I Level of Effort	10
3d. Deliverable Examples for Communication/Development of Cost and Schedule	10
4. Construction Management, Sequencing, and Scheduling	10
4a. Achieving Efficiencies in Scheduling and Construction Sequencing	10
4b. Achieving Performance Requirements and Optimizing Quality	10
4c. Exceeding Project Goal #5 and Maximizing Safety	11
4d. Construction Management, Sequencing, and Scheduling Challenges	11
4e. Construction Management, Sequencing, and Scheduling Tools	11
Attachment A: Draft Public Involvement Plan	

Attachment B: Example Decision and Design Alternatives Matrix

Attachment C: Phase I Level of Effort

Attachment D: Example CPM Schedule and Three-Week Look-Ahead Schedule Deliverables

Attachment E: Example Cost Estimating Deliverable

Attachment F: High-Level, Achievable Proposed Project Schedule

Attachment G: Construction Phasing - Part 1 and 2



1. Overall Management Approach

1a. Overall Management Approach to the Lutak Dock Replacement PDB Project

Collaboration and **value creation** are the guiding principles of our management approach to projects like the Lutak Dock Replacement. As evidenced by the success of our past projects including the Haines Ferry Terminal Improvements and the Tenakee Ferry Terminal Improvements, keeping these principles at the forefront of our decision-making earns the Owner's trust and ultimately yields a cost-effective facility that meets the needs of multiple users. It is worth noting that WMC's decades of project history boast a track record free of claims or litigation, a testament to the effectiveness of our implementation of these core philosophies.

A proactive and thoughtful approach to the design, permitting, construction, and operations of the Lutak Dock are essential to the success of this project. Building upon the conceptual work that has been done, we will collaborate with the Haines Borough, R&M Consultants (the Owner's Advisor), dock users (primarily Alaska Marine Lines and Delta Western), and the public at large to ensure the new Lutak Dock meets existing needs and is capable of meeting future ones.

All of our team members are tuned in to the details that will make this a successful project. At the Project Kickoff, we will make sure everyone is on the same page regarding expectations, deliverables, and milestones. This will set the stage for the entire project, allowing us to manage available resources, limit costly delays, and align our goals.

Setting the project up for success also involves anticipating its potential pitfalls and planning accordingly. Our approaches to dispute mitigation and risk management, detailed in the following pages, have been refined over the course of decades of experience and hundreds of projects. They have been integral to our management approach and maintaining a record of claim and litigation-free projects.

== Approach to Dispute Mitigation and Resolution ==

Building a challenging project such as Lutak Dock does not have to be confrontational, but potentially will be at some point due to limited budgets, differing site conditions that may be encountered, misunderstanding of project scope, or emotions that can arise as a result of the personal investment we make to seeing a project through to completion. However, we will manage confrontational situations and prevent them from becoming disputes (or at worst, a claim) by:

- Identifying the potential for conflict or dispute early. followed by immediate and open discussion amongst the parties regarding the confrontational subject matter. Too often parties will immediately go on the defensive when a confrontational matter arises, wasting valuable time that should be used for reaching resolution, and instead focusing on bolstering their respective positions on the matter. Instead, our team will tackle the issue head-on, laying out the risks, costs, and/or impacts of the subject matter to the Owner so that we can jointly focus on finding solutions.
- Recognizing, understanding, and respecting the position of the other party. For example, all parties will be concerned with cost, but for different reasons. An Owner is typically concerned about cost overruns and overall budget limitations, whereas a Contractor is concerned about cost control and lost profits. By recognizing these concerns, we can focus on finding middle ground that both parties can agree is fair.
- >> Prioritize finding solutions over assigning blame. Assigning blame to another party is the quickest way for a project challenge to move from confrontation to dispute. It puts the party on the defensive and turns their focus to counter arguments and blame assignment. This mindset wastes time and resources that can be much more productive if focused on finding solutions to the problem at hand.

= = Approach to Risk Management ==

The identification and management of project risks through open communication is a paramount portion of the collaborative work between our team and the Haines Borough. Our general approach to risk management involves three steps:

Step 1) Identify Potential Risk Factors and Sources: Sources of risk might include unanticipated Owner-directed scope or design changes, unforeseen existing site conditions, lack of specialized equipment, uncharacteristic weather conditions, global health pandemics, unachievable permit compliance, and more. For example, if a project were to require specialized, but not yet mobilized, equipment to complete out-of-scope work, we would identify potential schedule impacts as a risk.

Step 2) Analyze Risk Factors: After identifying the risk, we classify the risk as controllable (i.e. Owner-directed scope or design changes) or non-controllable (i.e. global health pandemics), and typically assign higher risk to a non-controllable factor. We then enumerate the potential impacts of that risk, such as change in cost, time, workmanship, and quality, or some combination of the four. Finally, we assign estimated quantities, durations, or reductions in design life to those potential impacts. We can then use this information to guide decision-making and response.

Step 3) Respond to the Risk: Following our analysis, our team will recommend a risk response to the Haines Borough. This might include: manage the risk, minimize the risk, share the risk, transfer the risk, or accept the risk as-is. Depending on the risk, the recommended response may require a change order, unit price reduction, liquidated damages withholding, or the removal/correction of non-conforming work or materials.

The project team will develop a running list of risks and work together to manage and mitigate the consequences associated with each risk through design, communication, coordination, and innovation. Below is a preliminary list of noted risks (several voiced by Selection Committee members during our June 30 interview) with corresponding potential mitigation measures.

Risk to Project Success	Mitigation Strategies
Public Perception and Misunderstanding of Goals	 Clear, consistent messaging from the Project Team Implement a robust Public Involvement Plan (PIP) to ensure public understanding of project scope and intent Initiate PIP as early as possible to foster community ownership of the project
Interruption of Facility Operations	 Work closely with Michael Ganey (AML) and Jack Eckhardt (Delta Western) to understand their respective needs and priorities Sequence construction to maintain partial facility footprint for continued essential operations
Unknown Geotech and Site Conditions	 Perform strategic geotechnical investigation at critical locations to avoid an overly conservative design Tap knowledge from our previous work within the Lutak Dock structure Prepare for variable soil quality within existing sheet pile cells
Permitting Delays	 Experienced, Juneau-based Environmental Manager with curated EndNote library of over 30,000 peer-reviewed scientific articles to draw from for drafting applications Ensure completeness of permit applications and reports on first submission Maintain regular communication with Agency partners to expedite the process

Lack of Collaboration and Teamwork	 Work to develop and earn mutual trust and respect Maintain open, honest lines of communication (formal and informal) Deliver regular, transparent, and realistic updates regarding public process, schedule, cost, and other project details
Instability in Management Team	 Western Marine Construction and subcontractor teams all led by long-standing per- sonnel, including firm founders Track record of sustained boots-on-the-ground, project-level involvement from com- pany principals
Schedule Delays	 Consistent, realistic schedule update deliverables using Primavera P6 Proactive regulatory agency engagement for permitting acquisition Prompt confirmation of design concepts to maintain project progression
Price Escalation and Market Conditions	 Identify material restrictions and requirements in funding sources Promptly confirm design concepts to maximize time windows for evaluating cost-effective supply options Communicate with vendors and purchase materials early to lock in price
Non-Compliance with Funding Sources	 Thoroughly review federal and state funding source requirements at project outset Establish productive working relationship with each funding source Point of Contact Review materials and contracts for compliance prior to purchase Maintain highly organized project files for smooth grant closeout
Budget Overruns	 Promptly develop preliminary estimates with risks identified and contingencies noted Design and budget for alternate scope items which may be added to the contract pending availability of funds once riskier portions of the project (ex. demolition) are complete Fast-track procurement of material with high cost volatility (ex. structural steel, sheet piling)

1b(i). Creating a Collaborative Environment and Exceeding Project Goal #2

Our approach to creating a collaborative environment for the duration of the project begins with establishment of a team-centric, partnering mindset for all design-build team members, Haines Borough representatives, and key stakeholders. From WMC President Kriss Hart to the field technician performing geotechnical investigation and data collection, our entire team will be working with the Borough to reach their goals for the Lutak Dock project. We will work to establish a cohesive Lutak Dock Project Team using strategies detailed in Section 1A, including honest communication, regular status meetings, clearly defined expectations from the outset, and deliverable requirements.

During design, WMC will manage the contract from their offices in Seattle and Juneau, while our engineering team members will work from their respective offices in Bremerton, Sitka, Juneau, Anchorage, and Haines. Most notably, we will lean heavily on the presence of proHNS in Haines to help coordinate field investigation and data collection activities, as well as public outreach and stakeholder engagement. Once notice to proceed with construction is issued, WMC will provide direct oversight and management of the project from field offices in Haines with the support of proHNS. While our team is spread throughout the State of Alaska and the greater Seattle area, technologies that allow file sharing and virtual team meetings will be employed extensively to ensure we are working as a single design-build team.

1b(ii). Approach to Stakeholder Outreach and Incorporation of Input

= = Public Involvement Plan ==

We have followed the local response to this project thus far, and virtually attended the April meeting where the Haines Borough Assembly approved conceptual plans for Phases I-III of the project. We understand there is some community confusion and misunderstanding about the project's scope and intentions – particularly surrounding Phase III – largely due to a presentation at a March Ports and Harbors Advisory Committee meeting.



proHNS Design Manager Garret Gladsjo, PE (right) listens to a Juneau resident's concerns about a CBJ road reconstruction project.

Having familiarized ourselves with the current situation and the concerns voiced thus far, we have developed a **Draft Public Involvement Plan (included as Attachment A)** to educate the public about the project's true scope and intent and ensure residents have a forum to offer their input. While Borough Code provides ample opportunity for public comment on design (the Planning Commission and Assembly will review plans at the 35%, 65%, and 95% stages), we believe a public process external to code-mandated meetings will help bring the community along on this process and quell some of the anxiety surrounding the project.

Our strategy is relatively straight-forward: 1) Engage the public early in the process, 2) Clarify the project's purpose, and 3) Open the door for all comments and make sure that the community understands what the project is and what it is not. A robust public involvement campaign will prevent further controversy to the greatest extent possible and foster community ownership of the project.

Upon project award, we will work with the Borough to develop a finalized Public Involvement Plan that includes particulars for meetings, notices, and other details.

A Note on Public Involvement Roles and Responsibilities

At the Project Kickoff meeting, we will discuss and define what role the Haines Borough would like the Design-Build Consultant to play in relation to the Owner's Advisor (R&M Consultants) when it comes to stakeholder involvement. The Owner's Advisor RFP also asked respondents to address stakeholder outreach, so it will be important to determine from the project outset who is taking the lead on outreach, who will play a more supporting role, how outreach tasks will be divided, etc. Our team is prepared to take on whatever level of involvement the Borough and R&M deem appropriate.

= = Method for Incorporation of Stakeholder Input ==

At the Project Kickoff meeting, we will define all stakeholders and their roles/tiered levels of involvement. For example, Delta Western and AML representatives would be classified as high-tier stakeholders with sustained involvement throughout the design process. Alaska Power & Telephone (AP&T) might be a mid-tier stakeholder with as-needed consultation, given their infrastructure is present at and services portions of the dock. A lower-tier stakeholder might be a tour operator who uses Lutak Road to bus visitors out to Chilkoot Lake; they may be slightly impacted by construction for short periods during equipment mobilization.

Over the course of the public meetings outlined in the Draft PIP and individual meetings with the tiered stakeholder groups, we will inevitably receive a deluge of comments. These comments will likely range from useful feedback about existing operations and future anticipated needs to vehement protests about the incorporation of any Phase III elements. It is unrealistic to assume the Design Team can incorporate every single suggested addition, deletion, or alteration; however, we have developed a useful method to organize comments, address their relative merit, and justify their incorporation or omission.

Our team will use a **Decision and Design Alternatives Study Matrix (example included as Attachment B)** to address ideas contributed throughout the stakeholder and public engagement process. Comments are organized into similar categories of concern, with descriptions of each item, potential solutions, advantages, disadvantages, rough order of magnitude costs, and a graphic, if applicable.

Along with the matrix, our team will draft a technical memo outlining design alternative recommendations and supporting justifications. In addition, an Area Map will be included to indicate the location within the project limits where each recommendation would be implemented. Items are categorized as "Recommended" "Conditionally Recommended" or "Not Recommended." The Haines Borough would make the final decision on which recommendations they would like to accept and incorporate, if any.

2. Maximize Design within the Limited Budget

2a. Overall Approach to Exceeding Project Goal #1

Our overall approach to designing and constructing a dock that maximizes program requirements within the limited budget mirrors our overarching company philosophy: **provide value at every step of the process**. Having designed and constructed numerous facilities involving the scoped components of the Lutak Dock Replacement, our team of professionals knows the means and methods for delivering the most service without sacrificing operational efficiency or blowing the budget.

For example, on the AK DOT&PF Tenakee Ferry Terminal Improvements (2020), we led a value-engineering change proposal (VECP) effort to redesign the staging dock, utility building, fueling system, and electrical systems providing cost savings, schedule savings, and a more functional facility. We also led the design and environmental permitting effort to proceed with the VECP concurrent with ongoing procurement and construction, eliminating possible project delays. Our team's efforts ultimately earned an Associated General Contractors of Alaska Excellence in Construction award for this project.



Our team has a long history of successful value engineering, including on AK DOT&PF's Tenakee Ferry Terminal Improvements (2020).

On AK DOT&PF's Haines Ferry Terminal Improvements, the State initially intended to salvage and transform one of the cells on the Borough's side of the structure; however, we pointed out that due to the Lutak Dock's condition, the facility would likely be replaced in the near future, at which point the cell would need to be removed anyway. Instead of salvaging the cell that would soon need to be demolished, WMC worked with AK DOT&PF to modify the design, armor the slope area, and save the State nearly \$1 million in construction costs.

From the time we receive Notice to Proceed to when we sign off on the last project closeout document, our team will be brainstorming and presenting ideas to save you money, time, and effort without compromising the integrity of the end product.

2b. Strategies and Design Ideas for Exceeding Project Goal #1

With more than half a century of experience successfully completing Southeast Alaska marine construction projects using various delivery methods, our Design-Build Team will use their knowledge from previous projects (including the Haines Ferry Terminal Improvements) and understanding of the existing facility to develop creative, efficient, and cost-saving design solutions. A combination of the following strategies and techniques (detailed on Page 6) will yield a robust structure that meets all identified needs without exceeding funding limits.

Haines Borough Request for Proposals - Design Builder Lutak Dock Replacement Management Proposal

Sequencing Work to Maximize Fill Reuse

Our team will develop a phasing plan to maximize the reuse of existing bulkhead backfill material during construction. Suitable existing material will be strategically stockpiled for use within the new bulkhead and uplands area, including the new boat ramp.

In general, it is anticipated that construction can begin from where the Haines Ferry Terminal project ended and proceed toward the AML facility, allowing for placement and stockpiling of fill materials without impacting existing operations. Fuel and barge operations can then shift to the newly constructed portions of the project, allowing for completion of the northwestern portion of the site. This will result in the least amount of imported and exported general fill, saving project funds while still maintaining operational continuity for barge and fuel service.



Using Various Drilling/Pile Driving Techniques

Our team is experienced with a vast array of pile driving and drilling techniques and equipment, including vibratory driving, impact driving, and down-the-hole hammer drilling for rock anchors, socketing, and drilled shafts. Detailed design analyses will be performed to verify the expected capacity requirements for each component of each structure allowing flexibility the maximum timely flexibility to adapt to conditions encountered in the field.

As various conditions (such as buried debris or variations in bedrock elevation) are identified in design development, our team will evaluate a variety of systems to achieve the required design load capacities. For example, we anticipate the presence of weak rock seams, and will include a sleeve to contain grout for all drilled tension anchors.



Capitalizing on Local Resources

We have established relationships with subcontractors and material suppliers who will provide the resources to construct the project in the most cost-effective manner. With rock supply, for example, with a variety of transportation options at our disposal, we will evaluate each scenario including WMC-owned barges, rock supplier furnished barging, WMC-chartered barge, or truck delivered as applicable from the source location within northern Southeast Alaska. -0-



Tapping Site-Specific Knowledge

Because of our experience demolishing and constructing new structures within the original footprint of the Lutak Dock (*shown above*), we know the many potential issues that exist in the removal of the existing dock. From being unable to completely extract existing sheet pile to unstable soils to buried debris and handling the large concrete cap structures, we will size equipment accordingly for each potential condition to be encountered. This will allow us to quickly adjust in the field minimizing downtime and the need for additional contingencies reducing the risk for unplanned cost overruns. For example, if unstable materials are encountered, placement of riprap slope protection will be prioritized to prevent slope erosion.

Strategically Positioning the New Structure

Our design team will focus on strategic placement of each new structure component to minimize potential conflicts with the existing structure. For example, dolphin structures will be placed so pilings can avoid the existing sheet pile alignments. Bulkhead piles will be configured so non-critical piles are located at any existing sheet pile cell interfaces to allow for field adjustments without impacting cost or the integrity of the structure.

Self-Performing All Construction Activities

With the in-house experience and resources to self-perform all of the primary construction aspects of the project, we can exercise maximum control of the project cost and schedule. This includes demolition, placement of slope protection, excavation, pile installation, and structure assembly.

By internally rebalancing resources, we can continue working productively even when issues like buried debris are encountered, whereas other contractors may be forced into incurring standby and remobilization costs of other potential subcontractors.

2c. Challenges in Developing the Design and How to Address Them

Our team has identified what we believe are the two biggest design challenges with the Lutak Dock Replacement project: unknowns with the existing structure and maximizing the facility for multi-purpose use with a vast array of stakeholders.

Challenge: Navigating Unknowns within the Existing Structure

From our experience working on Haines Ferry Terminal, we are well-aware of the wide-ranging challenges associated with the demolition of the deteriorated Lutak Dock structure. However, with this knowledge in-hand, our design-build team is the best equipped to manage, plan, and execute this work.

To address the challenges of the existing structure, our team will begin with a strategic approach to design. Design of the new structure will encompass adjusting alignments and structure configurations to minimize conflicts with any known aspects of the existing structure which may only allow for partial removal in spots. Components, such as dolphin structures, will also be designed with flexibility in mind, to allow for easier field realignment and reorientation of components if field conflicts are encountered.

We are prepared to effectively and quickly manage the unknowns within the existing structure and adapt to challenges as they are encountered. Our initial mobilization will include an array of equipment to ensure we have the necessary tools on hand to address issues and maintain progress. Our contingency plans will also allow us to quickly pivot and adjust structure locations to avoid obstructions when possible.

Finally, having encountered a variety of debris, intact structures, and deteriorated components during the past project, our team can accurately quantify the level of effort required to execute this work without excess contingencies that would place the project budget at risk.

Challenge: Maximizing Multi-Purpose Use of the Facility

We understand the Lutak Dock must accommodate a variety of existing functions, with the desire to incorporate additional functionality as the budget allows. Each function – from unloading bulk fuel barges and shipping bulk cargo to launching fishing boats – has unique operational and spatial requirements that need to be met within a limited footprint.

Prompt vetting of these priorities with the Haines Borough and other appropriate stakeholders will avoid impacts to the permitting schedule and design schedule. Immediately after contract award, we will open communication with the Borough's established Lutak Dock Project Group, which includes a convenient cross-section of user group representatives who are already up to speed on the project's status. We will discuss each desired function and its context in the facility as a whole to quickly develop the concepts required for environmental permitting.

2d. Communicating and Collaborating with the Haines Borough and other Stakeholders

At the Project Kickoff meeting, we will discuss and define what role the Haines Borough would like the Design-Build Consultant to play in relation to the Owner's Advisor (R&M Consultants) when it comes to stakeholder involvement. The RFP issued by the Haines Borough for an Owner's Advisor also asked the respondents to describe their approach regarding stakeholder engagement for this project (including but not limited to Delta Western, the Haines Borough Planning Commission, applicable Advisory Boards, the Borough Assembly, and the public), so it will be important to define who is responsible for what as soon as possible.

From our experience working with the Haines Borough, we are cognizant of the code-required reviews and meetings that design projects must navigate with the Assembly and Planning Commission. We know that the Planning Commission only

Haines Borough Request for Proposals - Design Builder Lutak Dock Replacement Management Proposal

meets once a month (the first Thursday) and the Assembly usually meets twice (the second and fourth Tuesdays, with only one meeting in November and December due to the holidays). These meetings need to be taken into account when planning contract approvals, design schedules, and other project elements that require Assembly or Planning Commission sign-off. We will build these dates into our initial schedule to avoid schedule delays or the need for emergency meetings.

We will request that all Owner Staff, their representatives, and key stakeholders provide their project input and review comments in writing, whether in the form of emails or plan sheet redlines. All Owner input and stakeholder will be evaluated for feasibility, both from a constructability and budgetary standpoint. If Owner and stakeholder input cannot be economically or physically incorporated into the project, our team will provide a written response to the input outlining our evaluation, findings, and any alternative recommendations.

Many of our other strategies for communicating and collaborating with the Haines Borough, stakeholders, and the general public are outlined in Section 1b of this Management Proposal and in our Draft Public Involvement Plan (included as Attachment A).

Success Example: Collaboration on the Tenakee Ferry Terminal Improvements

Our work on the Tenakee Ferry Terminal Improvements provides a prime example of how we communicated and collaborated with both the Owner (AK DOT&PF) and Tenakee residents to brainstorm a creative design solution that worked for everyone. When community members expressed concern that planned drilled rock anchors would compromise the town's famous hot springs, our team worked with community representatives and AK DOT&PF to devise an alternative design that replaced the concrete pier structure with a permanent fill dock that did not require anchor drilling.

The benefits to this solution were two-fold: 1) the hot springs were undamaged, much to the community's relief, and 2) the new structure not only remained within budget, it was large enough to accommodate a storage building and laydown area twice the size of the original plans.

3. Project Controls, Cost Tracking, and GMP Development

3a. Strategies for Exceeding Project Goal #3

Strategy 1: Quickly Identify Functional Requirements of the Facility

With the desire for the Lutak Dock to be a multi-purpose facility, it is critical for the Design-Build Team to quickly understand the functional and operational requirements versus the "wish list" aspects of the facility. This thorough understanding will allow the team to tailor its efforts accordingly as we initiate the design and permitting processes so that we can fast track the project. Furthermore, to obtain this information promptly, our team requests we schedule a design kickoff meeting as soon as possible with the Haines Borough team and key project stakeholders like AML and Delta Western.

Strategy 2: Mutually Agree Upon a Cost Breakdown with Contingencies

A key to providing transparent pricing for the Haines Borough's use is agreeing upon the format and level of detail associated with the cost breakdown. While some details will be negotiated and identified in the contract, the actual estimating layout for the various work activities will be agreed upon amongst the Design-Build Team and the Haines Borough to provide an efficient tool for review and discussion. Additionally, cost breakdown for project contingencies and optional scope items provide flexibility and transparency in decision making.

Strategy 3: Hold Regular Progress Meetings and Provide Schedule Updates

Schedule transparency is best accomplished through a combination of regular project schedule updates and progress meeting discussions. While an invaluable tool, CPM schedules do not always present the entire picture, nor are they easy for all team members and stakeholders to understand. Therefore, during progress meetings, a discussion will be held regarding the project schedule at both a micro-level in the near term and a macro-level as it relates to the overall project. Additionally, 3-week look-ahead schedules will be utilized with during the progress meeting to provide additional detail and insight to the team. As a unit, these tools shall provide the Owner with the desired schedule transparency for the project. See Attachment D for examples of what these schedule deliverables will look like.

3b. Processes and Tools for Monitoring, Reporting, and Managing Cost

3b(i). Budget Control and Reporting Processes

We will use a combination of reports from our ComputerEase accounting system and customizable spreadsheets to relay current cost data at the agreed upon intervals. Cost tracking will be broken down in a mutually agreeable Schedule of Values format as noted in the contract. For establishment of the GMP, customizable spreadsheets will be used to construct estimates and forecasts with the appropriate work breakdown structures requested and agreed upon with the Haines Borough. This will provide flexibility for formatting presentation data to the Haines Borough for evaluation. During Phase II, we will utilize a combination of tools similar to Phase I for tracking costs through project completion.

3b(iii). Incorporating Input from Other Subcontractors

Our design-build team has a long history of successfully working with specialty trade industry partners who we can quickly engage for this project if necessary. These team members will be asked to participate in stakeholder discussions as applicable to their specific scopes and to collaborate on the best approaches in meeting the needs of the stakeholders. Furthermore, these specialty trade subcontractors (such as electricians) will participate in regular design discussions and provide review comments as design development progresses.

3b(iv). WMC's Differentiating Resources

Our differentiating resources that will allow us to establish the GMP more accurately than our competitors is our collective years of experience and familiarity with the conditions at the project site. Our marine design and construction experience will enable us to develop cost-effective and constructable project needs which are definable, understandable, and mutually agreed upon. This agreement is essential to finalizing the GMP to proceed with Phase II, as well as management of the budget for the remainder of the project. Our familiarity with the site will also allow us to present the required level of effort and associated contingencies for various tasks in a logical manner, ensuring project partners can come to a consensus.

3b(ii). Scope, Cost, and Schedule Baseline Development

The development and evolution of scope, cost, and schedule are fundamental aspects to the preconstruction process on any project. In progressive design-build, these elements extend beyond the designer and encompass the construction team as well, requiring the establishment of solid lines of communication and protocols. Our protocols will include:

- Conduct Regular Design Review Meetings. The entire design-build team will participate in design review meetings to collaborate on and resolve identified comments associated with scope, cost, and schedule related to each design milestone.
- Provide Schedule and Estimate Narratives. Many projects require a schedule narrative outlining changes to a project schedule at each update interval. For the Lutak Dock Replacement, we will develop a narrative for each schedule and estimate update. The schedule narrative will outline progress to date, logic changes, and anticipated delays; the estimate narrative will portray major production and scope changes, differences in quantities, and adjustments to unit pricing.
- Utilize Tracking and Action Item Logs. These tools will provide a means of addressing and documenting input from each team members' perspective in regard to the scope, cost, and budget. They will also provide a record for tracking changes throughout the process.

3b(v). Primary Challenges in Establishing the GMP

The primary challenge in establishing the GMP for this project will be the identification and management of project contingencies. We will make sure to present a breakdown of the contingencies within each updated cost estimate. However, the level of effort to be carried within the GMP for specific tasks and the associated contingencies necessary for requirements above and beyond this level of effort will need to be negotiated and agreed upon.

3c. Phase 1 Level of Effort

Per the RFP, our team has developed a detailed breakdown outlining the tasks necessary to progress the design of the Lutak Dock Replacement through Phase I. See Attachment C for our Phase I Level of Effort breakdown.

3d. Deliverable Examples for Communication/Development of Cost and Schedule

Clear, consistent deliverables are the key to avoiding cost or schedule surprises. In the attachments, we have included examples deliverables we will rely on to communicate cost and schedule development though the design-build process.

<u>Attachment D: Example of CPM Schedule and Three-Week Look-Ahead Schedule Deliverables:</u> The CPM schedule and three-week look ahead schedules provide insight into the schedule status of the project. The CPM schedule tracks the overall project while the three-week look-ahead schedule provides a more detailed look at the upcoming pertinent activities. This information when regularly updated (monthly for the CPM schedule and weekly for the three-week lookahead) provides a means of tracking progress while presenting providing discussion points for coordination amongst the interested parties. This coordination will prevent unanticipated delays to the project.

<u>Attachment E: Example of Cost Estimating Deliverable</u>: Our team will develop and submit Engineer's Cost Estimates, along with backup quantity take-off calculations, with each 35%, 65%, and 95% project milestone. This information is useful to show changes in project quantities as the design develops and will be used to ensure the project as designed is aligned with the project budget. They also reflect variations in industry unit prices, methods employed for calculating quantities, and assumptions made in developing Engineer's unit price estimates. These documents will also provide an opportunity for our team to evaluate project costs and provide value engineering proposals throughout the entire design process.

4. Construction Management, Sequencing, and Scheduling

4a. Achieving Efficiencies in Scheduling and Construction Sequencing

Our Design-Build Team will implement a multi-faceted plan to optimize scheduling and construction sequencing for the project. First, the project will be planned and designed to perform multiple activities concurrently including the use of both landside and waterside operations. This will result in a decrease to the overall project schedule. Next, the team will work closely with stakeholders to understand both scheduling and operational requirements during construction. With the anticipation of constructing Phases I-III, this effort may be eased allowing the dock face portions of the project to be completed in two pieces.

As reflected in the **Attachment F High-Level, Achievable Proposed Project Schedule**, we envision breaking Project Phases II and III into two parts. Shown in the Lutak Dock Study drawings Sheet C4.0 (Phase III Site Plan), the project construction can be split between the two cargo barges depicted. Constructing the eastern portion first will allow for construction to progress reusing suitable fill on Phase I and progressing Phase II and III without interrupting freight and fuel barges which service the Haines community. Then, once the eastern portion is complete, the western portion of the Lutak Dock can be reconstructed with the eastern piece available for use by stakeholders. See **Attachment G: Construction Phasing - Part 1 and 2** for a visual depiction of this strategy.

4b. Achieving Performance Requirements and Optimizing Quality

Our team believes the best approach to ensuring a high-quality project begins with a commitment to quality assurance/ quality control at the onset of the design process through the completion of on-site construction. Prior to submission of deliverables for each project milestone, our team members will perform internal quality assurance reviews of documents prepared under their respective oversight to ensure conformance with project goals and design standards. Collectively, our team members will also perform peer reviews of each other's deliverables to identify and resolve any potential conflicts between disciplines. This holistic approach to quality assurance and control of design deliverables will ensure cohesion between the disciplines and overall project approach. Haines Borough Request for Proposals - Design Builder Lutak Dock Replacement Management Proposal

In conjunction with submission of the final design, our team will prepare and submit a Contractor Quality Control Plan outlining the testing and commissioning that will be performed. The Plan will include such details as material source requirements to be met, frequency of construction materials (concrete, aggregates, etc.) sampling and testing, the type of testing to be performing (including welding inspections), and the methods employed during testing. The Plan will outline commissioning of electrical and fuel systems, as well as provide the baseline requirements for the Operations & Maintenance Manual that will be provided upon project completion. Deliverables that will be provided during construction, such as material test reports and product submittals, and the timing of these deliverables to the Owner will be described in the Plan. Most importantly, we will seek the Owner's approval of the Contractor Quality Control Plan prior to beginning on-site construction, and once approved, will follow the Plan diligently until the completion of construction.

4c. Exceeding Project Goal #5 and Maximizing Safety

The goal of every WMC job is complete the work with zero injuries or incidents; we pride ourselves on our safety track record and strive to foster a safe environment on each and every project. Our Safety Program is built upon training, communication, and teamwork. Employees are equipped with the task-specific training and retrained regularly to ensure work is executed in a safe manner. Safety expectations are communicated during daily startup meetings, weekly safety meetings, and task preparatory meetings with Activity Hazard Analyses. We emphasize teamwork as an essential aspect of construction safety: every action has consequences, and not just for the individual responsible for the initiating action.

Western Marine proudly demonstrates an impressive safety record and has been repeatedly recognized by the United States Army Corps of Engineers for safety and achievement. This includes being nominated for the USACE Pacific Ocean Division Construction Management Award and awarded the USACE Alaska District Contractor of the Year Award multiple times. Our Safety Program is adaptable and applicable for implementing a safe working environment on any size project.

4d. Construction Management, Sequencing, and Scheduling Challenges

Scheduling and sequencing, quality, and safety will each present tests of the Design-Build Team ability to effectively manage and adapt to the site conditions.

- » Challenge #1 Sequencing and Scheduling with Stakeholder Operations. Our design-build team acknowledges there will be challenges working with and around stakeholders providing essential services to the Haines community. We will work with the Haines Borough and the stakeholders to develop a mutually agreeable plan with the understanding adjustments may be necessary as construction progresses.
- > Challenge #2 Quality. In addition to adhering to the testing and commissioning aspects of the quality control program for construction, WMC implements a USACE three-phase quality management approach. Through the preparatory, initial, and follow-up phases, quality requirements will be relayed to the team, reviewed during the execution of the work, and documented accordingly.
- > Challenge #3 Safety during Dock Demolition. From deteriorated structural components to unstable soils to large equipment operating in tight spaces, we understand the compounding potential for safety incidents to occur on this site. To avoid incidents, a hazard analysis will be performed, risks will be assessed, and mitigation measures will be defined. A detailed approach will be developed and clearly vetted with the crew performing the work to confirm there is an understanding of the risks and appropriate steps to take throughout the demolition process.

4e. Construction Management, Sequencing, and Scheduling Tools

Our primary tool is a defined Work Plan that addresses the work to be accomplished, the individual tasks required, the operational aspects that must be protected, and equipment/workforce requirements, all within a defined sequence. The regularly-updated Plan will have layout drawings and schedules; it also anticipates weather, tides, third-party users needs, etc.

Bar-chart look-ahead schedules will be used to further detail activities portrayed in the Primavera P6 CPM schedule. Project-specific tracking logs and spreadsheets will be customized to track project requirements including quality inspections and testing, submittals, RFIs, design review comments, certified payrolls, grant funding requirements, and progress payments.

Draft Public Involvement Plan for the Lutak Dock Replacement Project

The following is a DRAFT and example of what the Lutak Dock PIP might look like; the Haines Borough will have ultimate authority as to what tasks are included in the Final Public Involvement Plan, and who is responsible for what level of outreach.

Task 1 – Meet with Lutak Dock Project Group

- The Lutak Dock Project Group currently consists of:
 - o Mayor Doug Olerud and Public Facilities Director Ed Coffland
 - o Harbormaster Shawn Bell
 - Representatives from the Planning Commission and Ports and Harbors Advisory Committee
 - o Representatives from Alaska Marine Lines and Delta Western
 - o Two Haines residents
- Identify stakeholders (internal and external to the group) and their tiered levels of involvement in the design process. Ex. Who will be involved in weekly design progress meetings, who will only require monthly updates, etc.
- Look ahead to Assembly, Planning Commission, and Ports and Harbors meetings schedules to identify appropriate dates for updates outside of the code-required 35/65/95 review process (These groups can also request an update presentation at any time).
- Review Public Involvement Plan and make revisions as necessary.

Task 2 – Distribute Public Notice for Initial Public Meeting

- Occurs within one month of project kickoff meeting with Haines Borough.
- Posted on Project website, in the Chilkat Valley News, in KHNS PSAs, and on the KHNS Community Calendar.

Task 3 – Hold Initial Public Meeting

- Meeting should be held in person if possible (likely at the Chilkat Center, Assembly Chambers, or Library) with a virtual component, similar to how Haines Borough Assembly meetings are current conducted. Reasonable accommodations should be made to be inclusive of people with access issues and hearing/visual impairment.
- Meeting will begin with simple presentation of conceptual designs and schematics. Attendees will then be given the floor to ask questions and/or offer feedback.
- Attendees will also receive comment cards to fill out if they don't want to speak publicly and given information for submitting comments electronically, with a deadline for submission.
- Goal is to nail down the concerns and feedback from different groups. All comments will be documented and kept in the project record.

Task 4 – Design Team Comment Analysis and Recommendations

- Design Team holistically analyzes all comments to determine themes and identify primary needs, suggestions, concerns, and priorities.
- Design Team develops drafts Decision and Design Alternatives Study Matrix (see Attachment B for example) outlining potential solutions with corresponding advantages, disadvantages, and rough cost estimates.
- Design Team meets with Lutak Dock Project Group to discuss draft Matrix.

- Design Team composes Recommendation Memo categorizing Matrix solutions as Recommended, Conditionally Recommended, or Not Recommended, with justifications for each designation.
- Design Team meets with Haines Borough and Owner's Advisor to discuss Matrix and Memo. Borough issues written response identifying whether they concur with Design Team recommendations; Design Team adjusts design plans accordingly.

Task 5 – Meet with Lutak Dock Project Group and Owner's Advisor

- Present current plans.
- Give overview of how PIP has unfolded and discuss whether additional public outreach is necessary.

Attachment B: Example Decision and Design Alternative Study Matrix

Decision and Design Alternatives Study Matrix- Calhoun Avenue Reconstruction

Pages 3-5 removed for brevity

#	Categories	Method	Description	Advantages		Disadvantages	Cost Increase	Graphic
1	Traffic Calming Geometric Constraint Pedestrian Safety	Consistent and narrow traffic lanes	Reduced lane width between Capitol Avenue and W. 7 th Street to 9' lanes.	Provides consistency through the corridor. Narrowing traffic lanes is an identified method traffic calming. Reduced vehicle speeds and increased safety. Additional space for pedestrian facilities (exce locations of limited ROW/geometric constrain	d of ept at nt),	 Areas with limited ROW/geometric constraint force narrow sidewalks. Not compatible with the use of curb and gutter due to geometric constraint. 		
2	Traffic Calming Pedestrian Safety	Raising sidewalk elevation above road	Install standard 6" curb and gutter on Calhoun Ave, north of Goldbelt Avenue and south of W. 8 th Street.	Provides 6" high barrier between vehicles and pedestrians. Provides visual barrier that roadway is narrow	d	 Doesn't allow local neighbors to park on the lower existing sidewalk of roadway when off street parking is full. 		12 12 12 12 12 12 12 12 12 12
m.	Traffic Calming Pedestrian Safety	Raising sidewalk elevation above road	Raise sidewalks to 9" above road grade along entire Calhoun project corridor	Provides 9" high barrier between vehicles and pedestrians. Provides visual barrier that roadway is narrov	d wer	 Doesn't allow local neighbors to park on the lower existing sidewalk of roadway when off street parking is full. Non-standard curb height increases tripping hazard Vehicle strike could redirect it into oncoming traffic. 	\$\$	
	Traffic Calming Pedestrian Safety	Raised Crosswalks	Install raised crosswalks across Calhoun Avenue	Raised crosswalks are an identified method o traffic calming. Enhances crosswalk visibility. Makes crossing easier for pedestrians.	ıf	 Creates a maintenance issue for snow removal. Requires additional consideration to ensure roadway drainage. Reduces comfort of ride for vehicles and bicycles. Potential traffic risk for low clearance/long wheelbase vehicles. Increased construction and maintenance cost compared to standard crosswalks. Not standard for use in collector streets. 	55	
5	Traffic Calming Pedestrian Safety	Colored Crosswalks	Install colored sidewalks at Cope Park Rd, Capital Avenue, 825 Calhoun Avenue, W. 8 th Street and Governors House	 Colored traffic crossings are an identified me traffic calming. enhance crosswalk visibility. 	thod of	Reduced effectiveness in winter conditions. Increased construction cost compared to standard crosswalks.	\$5	

Page | 1

Decision and Design Alternatives Study Matrix- Calhoun Avenue Reconstruction

#	Categories	Method	Description	Advantages		Disadvantages	Cost Increase	Graphic
6	Traffic Calming Pedestrian Safety	Concrete Curb at Wall	Install concrete curb along existing stone retaining walls	 Visually narrows the road reducing traffic s Protects existing stone walls. Reduces wall maintenance. 	peed.	 Requires additional horizontal space which is limited in areas. 	\$\$	
	Traffic Calming	Traffic Striping	Maintain double yellow traffic striping through corridor, no centerline striping at intersections	 Painted traffic markings are an identified m traffic calming. Delineates lanes for vehicular safety. Enhance crosswalk visibility. Provides visuals for vehicles, pedestrians a bicyclists increasing safety. 	nethod of	Reduced effectiveness in winter conditions. Increased construction and maintenance cost.	s	
8	Traffic Calming	Stop Controlled Intersection	Three way stop controlled intersection at Calhoun Avenue and W. 8th Street	 Would require vehicles to stop at intersect slowing traffic and limiting speeds around existing blind corner. 	ion the	 Potential conflicts with sidewalk locations due to limited ROW. If perceived as unwarranted, may not be followed by vehicles. Not warranted by MUTCD. Unwarranted stop signs create liability for accidents. 	Installing a traffic control device contrary to industry standards exposes CBI to increased liability for any accidents or injury caused by the traffic control device.	STOP 3-WAY
9	Geometric Constraint	Easements/ROW acquisition	Use easements or acquire ROW at geometric constraint location at 230 W. 8th Street and 826 Calhoun to relocate a section of the existing stone wall	 Increased pedestrian safety. Increased vehicular safety. Allows for 5' wide side walks at choke point accommodation of 9' AASHTO travel lanes without curb and gutter. 	t with with or	Requires acquisition of private property. Significantly increased construction cost. Increase in project timeline. Requires all property owners to agree to ROW widening in order to be effective.	55555	
10	Geometric Constraint	Wall removal and replacement – Easements/ROW acquisition	Remove existing stone retaining wall below 825 Calhoun avenue and 214 W 8th Street and replace with new wall that is set back from road	Increased pedestrian safety. Increased vehicular safety.		Requires acquisition of private property. Increase design cost. Significantly increased construction cost. Increase in project timeline. Access required to 825 Calhoun Avenue. Construction may not be possible due to location of existing home foundations. Moving walls back far enough to achieve desired corridor width would likely require acquisition of homes above retaining walls.	\$\$\$\$55	

Page | 2

Decision and Design Alternatives Study Matrix- Calhoun Avenue Reconstruction

#	Categories	Method	Description	Advantages	Disadvantages	Cost Increase	Graphic
26	Pedestrian Safety	Flashing Traffic Signage	Flashing crosswalk/pedestrian crossing/road narrows signs	 Traffic signage is an identified method of traffic calming. Provides visuals for vehicles, pedestrians and bicyclists increasing safety. 	Potential conflicts with sidewalk locations due to limited ROW. Sing posts limit walk ability. Potential bedrock trenching for electrical components.	\$\$\$	

Assumptions/exclusions:

One-way option not included per direction from CBJ assembly.

Realignment of existing concrete retaining wall across from W. 8th Street (STA 17+00 to 19+00) not considered, cost exceeds project budget and construction may not be possible given existing homes above the wall.

Cost Increase Key:

- \$ <\$1,000
- \$\$ \$1,000 -\$10,000
- \$\$\$ \$10,000 \$50,000
- \$\$\$\$ \$50,000 \$100,000
- \$\$\$\$\$ >\$100,000

Recommendation Key:

Conditionally Recommended

Not Recommended

Resources:

FHWA Course on Bicycle and Pedestrian Transportation - Traffic Calming Portland Bicycle Plan - Bikeway Design - Best Practices NACTO - Urban Bikeway Design Guide - Shared Lane Markings NACTO - Urban Bikeway Design Guide - Bike Route Wayfinding Signage and Markings System Multi-way Stops – The Research Shows the MUTCD is Correct ALTA – Advisory Bike Lanes in North America

Attachment C: Level of Effort

Lutak Dock Replacement - Phase I Level of Effort

										Hours per
Task	Design-Build Team Task Descriptions	WMC	proHNS	NGE	RCE	AAA	MM	GE	N57	Task
Α	Project Delivery & Coordination	280	528	0	0	70	22	20	28	948
A.1	Contract/Project Management	100	192			48	4	4	12	360
A.2	Prepare and Submit Baseline Project Schedule	40				8	2	2		52
A.3	Coordination w/ Owner & Owner's Rep. (Includes Monthly Status Review Meetings)	60	144			14	6	6		230
A.4	Coordination w/ Key Stake Holders, e.g. Docks & Harbors, State of Alaska, etc.	40	120				4	4	16	184
A.5	Coordination w/ Facility Users, e.g. AML, Delta Western, Commercial Operators	40	72				6	4		122
В	Site Investigations and Data Collection	0	80	130	0	24	17	14	186	451
B.1	Research As-built Records, Master Plans, Similar Permits, & Historical Project Site Info		20	15		16	4	2	48	105
B.2	Review and Incorporate R&M Consultants Conceptual Design Data/Information		28	15		8	1		24	76
B.3	Site Visits by DB Team Members to Review Plans, Gather Data, & Obtain Field Notes		32				12	12	10	66
B.4	Site Surveying (Including Hydrographic) and Establishment of Uplands Survey Controls								104	104
B.5	Geotechnical Field Investigation & Reporting			100						100
С	Permitting	0	0	0	550	0	0	0	0	550
C.1	Prepare Draft Permit Documents				250					250
C.2	Prepare and Submit Final, 100% Permits on Behalf of Owner				200					200
C.3	Agency Coordination through Permit Acquisition				100					100
D	35% Draft Plans, Cost Estimate, Permitting, and Schedule	60	148	80	0	198	50	26	50	612
D.1	35% Civil Site Design for Phases 1, 2, and 3		88						50	138
D.2	35% Geotechnical Design for Phases 1, 2, and 3			80						80
D.3	35% Marine Structures Design for Phases 2 and 3. Including Approach Dock					198	2			200
D.4	35% Fuel System Design for Phase 2					150	24	2		200
D.5	35% Electrical Design for Phases 2 and 3						24	12	-	14
D.5	Prepare Preliminary Construction Phasing & Temporary Access Plan for Site		22				0	2		14
D.0	Propage and Submit 25% Construction Cost Estimate to Owner	50	32				6	2		42
0.7	Propare and Submit 35% Construction Cost Estimate to Owner	10	20				0	4		80
D.0	Propare and Submit 35% Progress Schedule to Owner	10	0				2	2		14
0.9	CERC Death Block Cast Schemate Description and Schedule	20	8	40		226	0	4	10	18
E 1	65% Civil Site Design for Phases 1, 2, and 2	30	232	40	0	220	00	32	10	210
E.2	65% Contachnical Design for Phases 1, 2, and 3		208	40					10	218
E.2	65% Marine Structures Decign for Phases 2 and 3 Including Approach Dock			40		220				40
E.3	65% Eval System Design for Phase 2, Including Temperature Access Plan					220	40	4		220
E.4	65% Floetsical Design for Design						48	4		52
E.5	Submit Draft Coastruction Design & Temporary Access Dian for Site to Owner		20				4	10		20
E.0	Submit brait construction Phasing & remporary Access Plan for Site to Owner	20	20				2	2		24
E./	Prepare and Submit 65% Construction Cost Estimate to Owner	20	10				0	4		46
E.8	Prepare and Submit 65% Progress Schedule to Owner	10					2	2		14
E.9	Prepare and Submit 65% Plans for Haines Borough Planning Commission Review	10	8			240	6	4		18
F	95% Draft Plans, Cost Estimate, Specifications, and Schedule	10	340	50	0	218	95	56	25	800
F.1	25% Civil Site Design for Phases 1, 2, and 3		108	22					25	193
F.Z	95% Marine Structurer Design for Dharer 2 and 2 Including American Design			32		210				32
F.3	95% Marine Structures Design for Phases 2 and 3, Including Approach Dock					218	20			218
F.4	95% Fuer System Design for Phases 2 and 2						30	4		34
F.5	55% Electrical Design for Phases 2 and 5		22				3	16		19
F.0	Prinanze construction Priasing & Temporary Access Plan		32				2	2		36
F./	Prepare and Submit Draft Construction QA/QC Plan to Owner		56	24			4	4		64
F.8	Prepare and Submit 95% Technical Specifications to Owner	10	/6	24			48	24		1/2
F.9	Prepare and Submit 95% Progress Schedule to Owner	10	8				2	2		22
F.10	Prepare and Submit 95% Plans for Haines Borough Planning Commission Review	405					6	4	distant in the second	10
G	Givir Development (Following Approval of 65% Design)	100	0	0	0	0	10	4	0	114
G.1	Prepare and Submit GMP Lost Estimate to Owner	100					10	4		114
G.2		I State States	Total and the second	Carlo and a second		Contraction of	in the second second	Contraction of the local division of the loc		0
	Totals for All Tasks	480	1348	306	550	736	262	152	299	4133

Design-Build Team Member

WMC	Western Marine Construction - Led by Kriss Hart, Project Executive and Julian Koerner, PE, Project Manager
proHNS	proHNS - Led by Garret Gladsjo, PE, Design Manager
NGE	Northern Geotechnical Engineering - Led by Keith Mobley, PE, GE, Geotechnical Manager
RCE	Rugged Coast Environmental - Led by Ryan Bare, Environmental Manager
AAA	Art Anderson Associates - Led by Brad Ginn, PE, Marine Structures Design Lead
MM	Modern Mechanical - Led by Shane Hooten, PE, Fuel System Design Lead

Attachment D: Example CPM Schedule and Three-Week Look-Ahead Schedule Deliverables

11.2	Activity Name	Original Start	Finish	1	and the second	Otr 4 2021	1	Name and Address	Otr 1 2022	21	10.10.20	Otr 2 2023		Versieller 1	Otr 3 2022	and the second second	Otra 202
	Adding Harris	Duration	1 11 1031 1	Seo	Oct	Nov	Dec	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct 0
Ediz Hook She	oreline Restoration	388d 25-Aug-21 08 AM	16-Sep-22 04 PM						100			may			, ing	16	Sep-22 04 PM
Milectones		388d 25-Aug-21 08 AM	16-Sep-22.04 PM		1	1						1				16	Sep-22 04 PM
Milestones	Notice of August	0d 25 Aug 21 08 AM	it sop in or the	Notice of	Award												1
M1000	Notice of Award	0d 02 Sep 21 08 AM		Notice	to Proceed							£				1	
M1070	PAK Mosting	0d 14 Sep 21 08 AM		- Las P	AKMeeting								1			1	
M1020	Commance Construction	0d 21 bil22 08 AM													Commence C	ostruction	
M1060	Complete Opsite Work	0d 2100122 00 AV	16-Sep-22 04 PM	-			1							P		Co	molete Onsite
Design	Complete Offsite Work	144d 14-Sep-21 08 AM	04-Feb-22 04 PM		_	_			04-Feb-	2 04 PM, D	Xesign						i i i i i i i i i i i i i i i i i i i
Design Desliminant F	Engineering/ Investigations	254 14 San 21 08 AM	01 New 21 04 PM		-	01-Nov-2	04 PM Pr	elminary Enc	ineering/ Inv	stigations							
Pretiminary E	Initial Site Superv	20d 14 Sep 21 08 AM	11.0ct 21 04 PM					1		ganaria							
PE1000	Proliminant Engineering (actualing Constal Eng)	200 14-Sep-21 08 AM	11-00-21 04 PM		L.10	itial Site Survey	·]		4								
PETOTO	Healthing Engineering (including Coastal Eng)	5d 14-Sep-21 08 AM	25-0CI-21 04 PM		1	Preliminary	Engineering	(Including Co	astal Eng)				1		1		
PE1020	Haz Mat Survey	50 26-Oct-21 08 AM	01-Nov-21 04 PM			Haz Mat	Survey	21-Dec-21 04	PM 35% Da	sion			1				
35% Design	258/ Design (instudios Constal Fact)	570 28-00-21 08 AM	21-Dec-21 04 PM		1		-	Decision	111, 00 10 00	agn			1		1		
Dibbo	35% Design (including Coastal Eng)	300 26-00-21 08 AM	06-Dec-21 04 PM	-	1		35% D	esign (includi	ng Coastal E	ng)	1		1				
D1010	35% Design Review	14d 07-Dec-21 08 AM	20-Dec-21 04 PM					5% Design R	aview		ļ						
D1020	35% Design Review Meeting	10 21-Dec-21 US AM	21-Dec-21 04 PM		1			35% Design F	Review Meetin		Darian						
Final Design	E I D I D I D I D I D I D I D I D I D I	450 224,960-21 08 AM	04-Feb-22 04 PM		1			1	044 60	2 04 FW, F	i ai Desigii		5				
FD1000	Final Design Development	20d 22-Dec-21 08 AM	20-Jan-22 04 PM					F	inal Design D	evelopment	ti i						
FD1010	Final Design Review	14d 21-Jan-22 08 AM	03-Feb-22 04 PM						Final De	sign Review	1	1					1
FD1020	Final Design Review Meeting	1d 04-Feb-22 08 AM	04-Feb-22 04 PM				.i	.i	+ Final De	sign Review	Meeting				AN WARNER	a anna	
Permitting		310d 14-Sep-21 08 AM	20-Jul-22 04 PM					1			1	1			20-001-22 04 P		ng.
P1000	Habitat/ Macrovegetation Survey	20d 14-Sep-21 08 AM	11-Oct-21 04 PM	-		abitat/ Macrove	octation Sur	Vev.									1
P1030	Develop JARPA/ BA Documentation (NAVFAC)	110d 14-Sep-21 08 AM	01-Jan-22 04 PM			Contraction of the local division of the loc	- Ingentering	Develop	JARPA BA D	ocumentatio	an (NAVFAC);					
P1070	Develop 401 WQ Certification Pre-Application (WMC)	110d 14-Sep-21 08 AM	01-Jan-22 04 PM	-				Develop	401 WQ Cen	fication Pre-	Application	(MMC)					
P1010	Develop Sampling and Analysis Plan (WMC)	45d 12-Oct-21 08 AM	25-Nov-21 04 PM		⊨		Devepp S	mpling and A	Analysis Plan	(WMC)		1	1				1
P1020	Approval of Sampling and Analysis Plan (DOE)	45d 26-Nov-21 08 AM	09-Jan-22 04 PM			-		Appro	val of Sampl	ing and Ana	lysis Plan (D	DØE)	-				1
P1040	Approval of JARPA/ Receipt of Permits	200d 02-Jan-22 08 AM	20-Jul-22 04 PM						and the second second	- minimilian		Contraction of the local division of the loc			Approval of J	ARPA/ Recei	p of Permits
P1080	Approval of 401 WQ Certification (DOE)	180d 02-Jan-22 08 AM	30-Jun-22 04 PM							No. of Concession, Name		-	-	- Approva	of 401 WQ 0	Certification (DOE)
P1090	Onsite Sampling and Analysis	20d 10-Jan-22 08 AM	29-Jan-22 04 PM						Onsite Sa	npling and 7	Analysis				.1		
Submittals	Manual Control of the	173d 26-Oct-21 08 AM	16-Apr-22 04 PM			Y I	1	1			16	-Apr-22 04 F	M, Submitta	l\$			
S1000	Accident Prevention Plan	30d 26-Oct-21 08 AM	06-Dec-21 04 PM			-	Accide	nt Prevention	Plan	•••••		·····	· • · · · · · · · · · · · · · · · · · ·	····	···•		•••••••••
S1010	Accident Prevention Plan Review	15d 07-Dec-21 08 AM	28-Dec-21 04 PM					Accident	revention Pla	n Review			1				
S1040	Quality Control Plan	45d 21-Dec-21 08 AM	03-Feb-22 04 PM			1			Quality	Control Plan	1						1
S1020	Enviromental Protection Plan	75d 02-Jan-22 08 AM	17-Mar-22 04 PM					-		En	viromental F	Protection Pla	an				
\$1050	Quality Control Plan Review	30d 04-Feb-22 08 AM	05-Mar-22 04 PM	- 1			1		i		Control Blor	Baulau					
51030	Environmental Protection Plan Review	30d 18.Mar 22.08 AM	16-Apr-22 04 PM							L. HVEDLY.	SQUUSI FIG	UTYEY GOY	Philadion Dia	Pour Pour	••••		<u>{</u>
Canatavatia	Environmental Protection Plan Neview	41d 21-14-22 08 AM	16-Sep-22 04 PM								1.10	TXIODIGLIGLI	JUIGMOIL FIG	LCC/ELVEW		16	6-Sep-22 04 P
Constructio	m					1					1		1				
C1000	Mobilization	5d 21-Jul-22 08 AM	27-Jui-22 04 PM	- 1		1					1	1			Mobilizatio	n	1
C1010	Demoition/ Fit Removal	150 28-JUI-22 08 AM	17-Aug-22 04 PM											1	De	ernolition/ Fil	Removal
C1020	Beach Nourishment/ Landscaping	150 18-Aug-22 08 AM	08-Sep-22 04 PM											· · · · · · · · · · · · · · · · · · ·		Beac	n Nourishmen
C1030	Demobilization	3d 14-Sep-22 08 AM	10-Sep-22 04 PM								1	1	1	1		13	-Sep-22 04 PM
Closeout	and the second	50 09-Sep-22 08 AM	13-Sep-22 04 PM										1				1
	Final Walkthrough and Punchlist	5d 09-Sep-22 08 AM	13-Sep-22 04 PM				1	1	1		3	(B)	4	1	3	Fin	al'Walkthroug

NOAA HOMEPORT

WMC				
	Project Start:	Wed,	6/15/2022	
	Display Week:	1		Jun 13, 2022 Jun 20, 2022 Jun 27, 2022 Jul 4, 2022 13 14 15 16 17 18 19 20 21 22 23 24 25 26 27 28 29 30 1 2 3 4 5 6 7 8 9 10
TASK ASSIGNED TO	PROGRESS	START	END	M T W T F S S M T W T F S S M T W T F S S M T W T F S S M T W T F S
Phase 1 Title				
Pile Driving	0%	6/15/22	2 6/21/22	
Barge	0%	6/20/22	2 6/22/22	
Set Bridge	0%	6/21/22	2 6/22/22	
Float Assembly	0%	6/23/22	2 6/ <mark>2</mark> 6/22	
Pile Driving (remaining)	0%	6/27/22	2 6/30/22	
Position Float & Secure	0%	6/30/22	2 7/1/22	
Camel Mods	0%	6/30/2	2 7/1/22	
Install Mooring Frames	0%	7/1/22	7/5/22	
Camel Installation	0%	7/2/22	7/6/22	
Ahtna utility work on Float	0%	6/30/2	2 7/7/22	

Attachment E: Example Cost Estimating Deliverable

Baseline Engineer's Estimate										
Project:	Chilkoot Loop Reta	ting Wall								
Owner:	wner: Haines Borough					N				
Date:	11/16/2021			N		4				
Prepared By:	E. Roemeling				HNIC					
Checked By:	G. Gladsjo			pro		C				
Pay Item	Pay Item Descri	ption	Pay Unit	Quantity	Unit Price	Amount				
1505.1	Mobilization		Lump Sum	All Req'd	\$4,000.00	\$4,000.00				
1550.1	Traffic Maintenance	Lump Sum	All Req'd	\$2,000.00	\$2,000.00					
1570.1	Erosion & Sedimen	Lump Sum	All Req'd	\$4,000.00	\$4,000.00					
2202.1	Unclassified Excavation		CY	152	\$20.00	\$3,040.00				
2702.1	Construction Surveying		Lump Sum	All Req'd	\$2,000.00	\$2,000.00				
2801.1	A.C. Pavement, Typ	e II-A, Class B	TON	10	\$340.00	\$3,400.00				
2806.1	Remove Existing As	phalt Surfacing	SY	63	\$10.00	\$630.00				
3201.1	Block Wall		SF	168	\$100.00	\$16,800.00				
3303.1	Concrete Sidewalk		SY	200	\$10.00	\$2,000.00				
3303.2	Rolled Curb		LF	40	\$100.00	\$4,000.00				
3304.1	Removal of Concrete Sidewalk		SY	200	\$10.00	\$2,000.00				
3304.2	Removal of Rolled 0	Curb	LF	40	\$10.00	\$400.00				
					Sub Total =	\$41,870.00				
					20% Cont. =	\$8,374.00				
					Total =	\$50,244.00				

	65% DR	FT Engineer's Estimate	
Project:	Chilkoot Loop Retating Wall		
Owner:	Haines Borough		
Date:	11/2/2021		
Prepared By:	E. Roemeling	pro HNS	
Checked By:	G. Gladsjo	pior II 4	
Pay Item	Pay Item Description	Pay Unit Quantity Uni	t Price Amount
1505.1	Mobilization	Lump Sum All Req'd \$4	,000.00 \$4,000.00
1550.1	Traffic Maintenance	Lump Sum All Req'd \$2	,000.00 \$2,000.00
1570.1	Erosion & Sediment Control	Lump Sum All Req'd \$4	,000.00 \$4,000.00
2202.1	Unclassified Excavation	CY 152	\$20.00 \$3,040.00
2702.1	Construction Surveying	Lump Sum All Req'd \$2	,000.00 \$2,000.00
2801.1	A.C. Pavement, Type II-A, Class B	TON 10 5	\$340.00 \$3,400.00
2806.1	Remove Existing Asphalt Surfacing	SY 63	\$10.00 \$630.00
3201.1	Precast Block MSE Retaining Wall	SF 320 5	\$125.00 \$40,000.00
3303.1	Concrete Sidewalk	SY 200	\$10.00 \$2,000.00
3303.2	Rolled Curb	LF 40 5	\$100.00 \$4,000.00
3304.1	Removal of Concrete Sidewalk	SY 200	\$10.00 \$2,000.00
3304.2	Removal of Rolled Curb	LF 40	\$10.00 \$400.00
		Sub To	tal = \$65,070.00
		10% Cc	ont. = \$6,507.00
		Total =	\$71.577.00

	95% Engi	neer's Es	timate			
Project:	Chilkoot Loop Retaining Wall Repairs - 437	599				
Owner:	Haines Borough			~		
Date:	2/18/2022			$\lambda \geq$		
Prepared By:	E. Roemeling			L	INIC	٩
Checked By:	G. Gladsjo			prol		
Pay Item	Pay Item Description		Pay Unit	Quantity	Unit Price	Amount
202.0002.000A	Removal of Pavement, Asphalt		SY	45	\$7.00	\$315.00
202.0003.0000	Removal of Sidewalk		SY	34	\$10.00	\$340.00
202.0009.0000	Removal of Curb and Gutter		LF	56	\$20.00	\$1,120.00
401.0001.002B	HMA Type II; Class B		TON	11	\$350.00	\$3,850.00
401.0004.5834	Asphalt Binder, Grade PG 58-34		TON	1	\$950.00	\$950.00
530.0000.0000	MSE Block Wall		SF	320	\$125.00	\$40,000.00
608.0001.0006	Concrete Sidewalk, 6 inches thick		SY	34	\$600.00	\$20,400.00
609.0002.0001	Curb and Gutter, Type 1		LF	56	\$225.00	\$12,600.00
640.0001.0000	Mobilization and Demobilization		Lump Sum	All Req'd	\$14,000.00	\$14,000.00
641.0001.0000	Erosion, Sediment and Pollution Control Ac	ministrati	Lump Sum	All Req'd	\$5,000.00	\$5,000.00
641.0003.0000	Temporary Erosion, Sediment and Pollution	n Control	Lump Sum	All Req'd	\$7,000.00	\$7,000.00
642.0001.0000	Construction Surveying		Lump Sum	All Req'd	\$3,000.00	\$3,000.00
643.0002.0000	Traffic Maintenance		Lump Sum	All Req'd	\$5,000.00	\$5,000.00
671.2005.0000	Stream Diversion and Dewatering		Lump Sum	All Req'd	\$20,000.00	\$20,000.00
					Sub Total =	\$133,575.00

		Final Engineer's E	stimate			
Project:	Chilkoot Loop Retai	ning Wall Repairs - 437599				
Owner:	Haines Borough			~		
Date:	ate: 3/11/2022			A		
Prepared By:	repared By: E. Roemeling			- H	INIS	4
Checked By:	G. Gladsjo			pro		
Pay Item	Pay Item Descrip	tion	Pay Unit	Quantity	Unit Price	Amount
202.0002.000A	Removal of Paveme	SY	45	\$7.00	\$315.00	
202.0003.0000	Removal of Sidewall	SY	34	\$10.00	\$340.00	
202.0009.0000	Removal of Curb and	LF	56	\$20.00	\$1,120.00	
401.0001.002B	HMA Type II; Class B	TON	10	\$600.00	\$6,000.00	
401.0004.5834	Asphalt Binder, Grad	le PG 58-34	TON	1	\$950.00	\$950.00
530.0000.0000	MSE Block Wall		SF	320	\$125.00	\$40,000.00
608.0001.0006	Concrete Sidewalk,	5 inches thick	SY	34	\$600.00	\$20,400.00
609.0002.0001	Curb and Gutter, Ty	pe 1	LF	56	\$225.00	\$12,600.00
640.0001.0000	Mobilization and De	Lump Sum	All Req'd	\$13,000.00	\$13,000.00	
641.0003.0000	Temporary Erosion,	Lump Sum	All Req'd	\$7,000.00	\$7,000.00	
642.0001.0000	Construction Survey	Lump Sum	All Req'd	\$3,000.00	\$3,000.00	
643.0002.0000	Traffic Maintenance		Lump Sum	All Req'd	\$5,000.00	\$5,000.00
671.2005.0000	Stream Diversion an	d Dewatering	Lump Sum	All Req'd	\$20,000.00	\$20,000.00
					Sub Total =	\$129,725.00

Attachment F: High-Level, Achievable Proposed Project Schedule

with ID	Artivity Name	Oppinal Start	Finish	2023	09-Jul-22 11 1
ivity it	Prove realize	Duration	- HASH		01
Haines Lutak	A Dock Replacement Proposal	795d 15-Jul-22 08 AM	19-Oct-24 04 PM		4 04 PM, Haines Lut
Traines Earca		TOSE AS IN 22 DEAM	10.0+24.04 PM	7 190d2	4 04 PM Milestones
Milestones	Prove set and a set of the set of		To success on the	Bronnelies Due	
MICOO	Proposais Due	0d 15-Jul-22 08 AM		no pose a care a construction of the second	
M1010	Review Approval of Phase 1 Design Contract at Haine's Borough As	00	23-Aug-22 04 PM*	Prenew Approvator Finale Locastin Connact a Fraine Solicoun Assertiary	
M1020	Present 35% Design to Haines Borough	Ud	13-OC-22 04 PMP	La Balanza Abrana Dorougi	
M1030	Review Approval of 35% Design by Haines Bolough	0d	28-Oct-22 04 PM*	and Design Dynamic and Congress and Congre	
MIUSU	Present 65% Design to Haines Borough	Ud	15-Dec-22 04 PM*	test of solessing of spinses conception	
M1060	Hevew Approval of 65% Design by Haines Bolough	Od	10-Jan-23 08 AM*	The review App March Bors Design by Harries Colory up to	
M1040	Review Approval of Final Design and Construction Fixed Price Conti	0d	24-Jan-23 04 PM*	The new opposite of the second s	
M1070	Present 95% Design to Haines Borougn	00	10-Mar-23 04 PM	Transition of the second se	
M1090	Review Approval of 95% Design by Haines Bolough	Ud	11-Apr-23 08 AM	and Sufficient Control of the Design of the Analysis of the Sufficient Suffic	
M1100	Submit Final Design Documents to Haines Borough	Ud	30-Apr-23 04 PM	Caultin Francisco Doganestation	
M1120	Commence Construction	0d 31-May-23 08 AM			
M1110	Receipt of IWWW Permits	Od	14-Jun-23 04 PM	Preception www.vermis	
M1135	Substantial Completion (Construction Phases 1-3)	0d	04-Oct-24 04 PM	Sugsame	Completion (Constru
M1140	Final Completion	b0	19-Oct-24 04 PM	i i i i i i i i i i i i i i i i i i i	notetion
Contractua	D. D. S. S. S. S. S. M. S. S. ONA S. S. DOL, M. S. D. DOL, M. S.	184d 15-Jul-22 08 AM	24-Jan-23 04 PM	v 24-sin-23 4 inv Conutrual	
C1000	Review of Proposals and Notice of Intent to Award	15d 15-Jul-22 08 AM	29-Jul-22 04 PM	Review of Proposats and Net Celof Intent to Award	
C1010	Phase 1 Design Contract Scoping and Negotiation	25d 30-Jul-22 08 AM	23-Aug-22 04 PM	Phase 1 Design Cantradit Scoping and Negotiation	
C1020	Finalize and Negotiate GMP Contract Amendment for Final Design a	20d 05-Jan-23 08 AM	24-Jan-23 04 PM	Finalize and Negotiate GMP Contract Amendment for Final Design and Construction	
Design		771d 08-Aug-22 08-AM	19-Oct-24-04 PM	i soit visation	4 04 PM Design
D1000	Site Survey Geotechnical Investigation, and 35% Design Developm	60d 08-Aug-22 08 AM	07-Oct-22 04 PM	See Sugar Contemporal Function in a 20% Design Development	
D1010	Finalize 35% Design	15d 14-Oct-22 08 AM	28-Oct-22 04 PM		
D1020	65% Design Development	45d 28-Oct-22 08 AM	15-Dec-22 04 PM	Trime discourse for the second	
D1030	Finalize 65% Design	151 16-Dec-22.08 AM	04-lan-23.04 PM	Pro Design percentantiti	
D1040	95% Design Development	60d 10-Jan-23 08 AM	10-Mar-23 04 PM	Triaze Down Cession	
01050	Epolice 05% Design	15d 11 Mar 23 08 AM	25.Mac 23 04 PM	and a processing of veropment	
01050	Develop Epol Design	20d 11-Apr 23-09 AM	20-Apr-23 04 PM		
01060	As Builte and Classes d	200 11-Apr-23 00 AM	30-Api-23 04 PM	Ceviliop Final Design Documents	
01070	As-Buits and Closeout	150 05-00-24 08 AM	19-00-24 04 PM	As-Builts	and Closeout
Permitting		SOLO DO-AUG-22 DO AVI	14-duine a ser mit		
P1000	EA and HA Development/ Cultural Resources Analysis and Consult	100d 08-Aug-22 08 AM	16-Nov-22 04 PM	EA and HA Developement/ Cultural Resources Analysis and Consultation	
P1010	USACE and NMFS Consultation	200d 18-Oct-22 08 AM	14-May-23 04 PM	USACE and NMFS Consultation	
P1020	IHA Approval, Permits Received, Consultations Completed, Final EA,	30d 15-May-23 08 AM	14-Jun-23 04 PM	IHA Approval, Permits Received, Consultations Completed, Final EA, and Decision Document submitted to MAP	AD
Constructi	ion is a second s	471d 31-May-23 OB AM	04-Oct-24 04 PM	U4_21:22	A PM, Construction
CON1000	Phase 1 Construction - Fill and Concrete Ramp	80d 31-May-23 08 AM	20-Aug-23 04 PM	Phase 1 Construction - Fill and Concrete Ramp	
CON1010	Phase 2 Construction - Bulkhead Removal & Dredging - Existing Cel	75d 15-Jun-23 08 AM	30-Aug-23 04 PM	Phage 2 Construction - Bukhéad Rémoval & Dredging - Existing Cells 7-12	
CON1020	Phase 3 Construction - Dolphin & Approach Dock Pile	20d 31-Aug-23 08 AM	20-Sep-23 04 PM	hase 3 Construction - Dolphin & Approach Dock Pae	
CON1030	Phase 2 Construction - Riprap Slope Protection (Part 1)	40d 06-Sep-23 08 AM	15-Oct-23 04 PM	Phase 2 Construction - Reirap Sippe Piotection (Part 1)	
CON1040	Phase 2 Construction - Dolphins 1-3	30d 21-Sep-23 08 AM	20-Oct-23 04 PM	Phase 2 Construction - Dotphins 1-3	
CON1050	2023-2024 Winter Shutdown	150d 21-Oct-23 08 AM	06-Apr-24 04 PM	2023-2024 Writer Shiutdown	
CON1060	Phase 2 Construction - New Bulkhead (Part 1)	40d 07-Apr-24 08 AM	16-May-24 04 PM	Phase 2 Construction - Kew Bukhead [F	art 1)
CON1070	Phase 3 Construction - Approach Dock Completion	10d 17-May-24 08 AM	26-May-24 04 PM	Phase 3 Construction - Approxich Doc	K Completion
CON1080	Phase 2 Construction - Utilities and Finishes (Part 1)	30d 27-May-24 08 AM	25-Jun-24 04 PM	Phase 2 Construction - Unities r	Ind Finishes (Part 1)
CON1090	Facility Switch Over	1d 26-Jun-24 08 AM	26-Jun-24 04 PM	🖵 Facility Switch Over	
CON1100	Phase 2 Construction - Bulkhead Removal & Dredging - Existing Cel	45d 27-Jun-24 08 AM	10-Aug-24 04 PM	Phase 2 Construction	- Bulkhead Removal
CON1110	Phase 2 Construction - Dolphins 4-5	25d 11-Aug-24 08 AM	04-Sep-24 04 PM	thase 2 Constitu	ction - Dolphins 4-5
CON1120	Phase 2 Construction - New Bulkhead (Part 2)	40d 11-Aug-24 08 AM	19-Sep-24 04 PM	Phase/2 Con/	truction - New Bulkh
CON1130	Phase 2 Construction - Utilities and Finishes (Part 2)	20d 05-Sep-24 08 AM	24-Sep-24 04 PM	Pilase 2 Cor	struction - Utilities ar
CON1140	Phase 2 Construction - Riprap Slope Protection (Part 2)	25d 10-Sep-24 08 AM	04-Oct-24 04 PM	+Phase 2 C	onstruction - Riprap
			- In a second		
ProjectB	laseline Bar Critical Remaining Work			Page 1 of 1	HESTERN MA
Actual W	fork Milestone				NY NY
Remainir	ng Work Summary				
					A Day of the local day

Attachment G: Construction Phasing Parts 1 and 2





II. PRICE PROPOSAL FORM

Western Marine Construction, Inc.

Finalist Name

Having carefully examined the Request for Proposal (RFP) for Design-Build Services for the Haines Borough, Alaska Lutak Dock Replacement Project, issued <u>June 17, 2022</u>, and Addenda numbers <u>1</u> through <u>2</u>, and the Agreement, the undersigned Design-Builder proposes the following Commercial Terms for the Project:

A. Design-Builder Lump Sum for Overhead and Profit that will is proposed to be inserted into Section 6.2.1 of the Agreement: Twenty thousand dollars (\$ 20,000)

B. Phase 1 Not To Exceed Amount (not scored)

The proposed Phase 1 Not to Exceed Amount is <u>\$ Six hundred eighty five thousand</u> dollars (<u>\$ 685,000</u>

C. Key Team Member Hourly Rates (not scored)

The Hourly Rates for Key Team Members are as follows:

Key Team Member Hourly Rates

Name	Pasition	Hourly Rate Preconstruction	Hourly Rate Construction
Kriss Hart	Project Executive	\$75	\$75
Julian Koerner, PE	Project Manager	\$75	\$75
Patrick McHugh	Superintendent	\$75	\$75
Garret Gladsjo, PE	Design Manager	\$156	\$156
Keith Mobley, PE	Geotechnical Manager	\$225	\$225
Ryan Bare	Environmental Manager	\$115	\$115
Brad Ginn, PE	Marine Structures Lead	\$185	\$185
Shane Hooten, PE	Fuel System Design Lead	\$200	\$200
Pat Gorman, PE	Electrical Design Lead	\$200	\$200
Kelly O'Neill, PLS	Survey Lead	\$150	\$150

PROPOSAL GUARANTEE

The undersigned hereby agrees that this Proposal may be accepted by Haines Borough anytime within ninety (90) calendar days immediately following the date indicated herein below, and the undersigned further agrees to submit a fully executed Agreement prior to the issuance of the Notice to Proceed that includes the Commercial Terms proposed in this Price Proposal Form.

PROPOSAL FROM:

Western Marine Construction, Inc.

(Finalist Firm Name)

7 / 15 / 2022

(Authorized Representative Signature and Date)

Kriss Hart, President (Representative's Printed Name and Title)

CONE15

(State of Alaska Contractor's License No.)
Alaska Business License # 14457

Alaska Department of Commerce, Community, and Economic Development

Division of Corporations, Business, and Professional Licensing PO Box 110806, Juneau, AK 99811-0806

This is to certify that

WESTERN MARINE CONSTRUCTION INC

2775 HARBOR AVE. S.W., SUITE A, SEATTLE, WA 98126-2138

owned by

WESTERN MARINE CONSTRUCTION INC

is licensed by the department to conduct business for the period

December 21, 2020 to December 31, 2022 for the following line(s) of business:

23 - Construction



This license shall not be taken as permission to do business in the state without having complied with the other requirements of the laws of the State or of the United States.

This license must be posted in a conspicuous place at the business location. It is not transferable or assignable.

Julie Anderson Commissioner



LUTAK Dock replacement

ADDRESS

8241 Dimond Hook Dr., Anchorage, AK 99507

PHONE

907-261-8960

PROGRESSIVE **DESIGN-BUILD** PROPOSAL

CONTACT

Jason Davis jdavis@turnagain.build



PAGE INTENTIONALLY BLANK

.UTAK DOCK RFP ••	
Table of Contents	
•	
1. Overall Management Approach	3
2. Maximize Design Within Limited Budget	5
3. Project Controls, Cost Tracking & GMP Developm	ent 11
4. Construction Management, Sequencing & Scheduli	ng 16

5. Identification of Projects 24

1. Overall Management Approach

Collaboration & Risk Mitigation

Collaboration and Risk Mitigation requires that all the stakeholders understand the objectives of the project, are informed of the unique challenges and difficulties of the project, and are committed to a mutually successful project outcome. The Turnagain team has worked at this facility in the past and understands the significance of providing the Haines Borough and its vendors a durable, high-design life freight facility that provides flexibility to service a number of diverse operations.

As Turnagain performed work at this facility in 2020, the team understands the risks, challenges, and coordination associated with constructing a new facility at the current Lutak dock location. Turnagain employs a group of highly skilled employees trained and experienced in the installation of pile-supported structures throughout the inconsistent bedrock conditions southeast Alaska is known for.

As an Alaskan-based entity, the Turnagain team is available for in-person, on-site communications with the Haines Borough and its stakeholders.

Partnering Methods

In 2016 Turnagain Marine submitted a proposal to construct the Gary Paxton Multi Use Dock under a design-build contract. The Owner established a maximum project cost of \$6.8m without additional contingency. In their response, Turnagain not only offered the most comprehensive, highest quality scope, but we also provided the lowest risk proposal to the Owner. The owner provided a geotechnical report but noted that the bedrock elevation was highly variable. The Owner had the contractor claim exposure if the anticipated bedrock elevation and the actual bedrock elevation were inconstant. Turnagain expressly accepted all risk for the differing site conditions and bedrock variability.

The Owner was skeptical that the contractor would take on such a significant project risk and follow through with their commitment if a major change occurred. After winning the project, Turnagain designed 3 foundation options: one for per plan bedrock elevation, one for shallower than anticipated bedrock elevation, and one for deeper than anticipated bedrock elevation. Through competent planning and preparation, Turnagain arrived on-site prepared to install any of the three foundation options.

At the first structure location, the bedrock was 100% deeper than expected. At the second structure location, the bedrock was 75% shallower than expected. Turnagain adapted their means and methods, installed the appropriate foundation option, and completed the project on time and for the pre-established lump sum amount without filing a request for additional compensation.

LUTAK DOCK RFP

Turnagain, views alternative delivery construction projects as opportunities to provide progressive owners with the highest value attainable. Fundamentally, value is attained only when each aspect of the project; scope, schedule, cost, safety, risk, and quality are genuinely optimized across all phases and elements of the project. Alternative delivery methods, including Design-Build, allow Turnagain to utilize its extensive marine infrastructure development knowledge and experience to participate as a true partner with the Lutak Dock Replacement stakeholder team—collaboratively delivering the best solutions to meet the program goals.

Starting with the end objective in mind, Turnagain will establish and has in fact already begun to establish a balance between project objectives and the completed project requirements that results in the greatest tangible value to Haines Borough and its vendors. The team will systematically break down the project focusing development efforts on elements that have the greatest influence on scope, schedule, cost, safety, risk, and quality. This methodology focuses efforts where they will yield the greatest and ultimately the most value possible to the Haines Borough. Due to the accelerated project schedule, Turnagain's previous work on this facility and on similar dock and terminal similar site locations provides an opportunity to shift more effort towards project optimization and collaboration with the Haines Borough.

Turnagain has thoroughly reviewed the RFP documents including the proposed contract form and is prepared to furnish a fully compliant scope of service and a highly effective progressive design-build team. The scope requirements and deliverables outlined in the RFP are acknowledged and will be delivered as required. In addition to the mandated review points (35%, 65%, etc.) and the specific deliverables they trigger, Turnagain desires to work closely with the overall project team to provide real-time collaboration to minimize re-work and backtracking from changes identified at the formal review points. Collaborative development of the milestone design sets will increase the efficiency and effectiveness of the preconstruction team efforts.

Compromise

Turnagain's portfolio demonstrates the drive to uphold our client's best interest and provide mutually beneficial solutions to challenges encountered through the project lifecycle. Although TMC's proposed design checks all the boxes, we are open to comments that will increase the value of the result since ensuring exceptional project delivery is in everyone's best interest, including the Haines Borough team, its stakeholders, and TMC. Turnagain is committed to thoroughly vetting all design and construction comments regardless of their source. In the event of an impassable disagreement, Turnagain will defer to our own 3rd party design QC and the Borough's independent consultants to collaborate on an objective solution. Despite all the challenges marine construction poses, Turnagain has a proven record of successful project completion and return clients.

Turnagain is committed to providing fair and transparent pricing throughout the project life cycle. Haines Borough will be provided access to Fonn Construction Management Software, HCSS software, and native files in addition to scheduled PDF reporting updates. Upon Haines Borough's acceptance, Turnagain will be able to place firm cost estimates early on in the project cycle, alleviating financial uncertainty.

LUTAK DOCK RFP

2. Maximize Design Within Limited Budget

A. Overall Approach

As a design-build firm, Turnagain performs all structural design, development, and construction Our engineering and in house. construction teams are completely all integrated, and design development is fully dissected and analyzed throughout the design and development process to ensure that design and construction meet the project budget and schedule. Turnagain is equipped to develop desian and plan construction simultaneously, locking down material specs for early procurement.



Beyond our ability to adapt the site-specific design elements, Turnagain is adept at managing progressive design-build to achieve maximum efficiency and value.



Turnagain is prepared to concurrently advance the permitting process and procurement activities if Haines Borough funding allows. Turnagain has been successful at shortening project durations and mitigating escalation risk using this method on previous Alaska design-build projects.



Turnagain has a fully onboarded team of subcontractors, sub-consultants, material suppliers, fabricators, and a design team ready to aid in the progressive design-build process.

In addition to Turnagain's management philosophy, the Turnagain team has developed a preliminary plan for this project that eliminates most of the demolition cost. In the process of determining the best value design solution, Turnagain thoroughly considered several structural retrofit options, multiple new pile-supported pier options utilizing different steel and precast concrete elements, and both cantilevered and cellular sheet pile bulkhead options. Turnagain's comprehensive alternatives analysis also considered the demolition effort, construction cost, timeline, and life cycle of many different concepts. After comparing all the design alternatives, Turnagain recommends replacing the existing structure with a pipe-pipe combi wall bulkhead with tiebacks to secure the best long-term solution. The new bulkhead will be driven three to four feet outside of the existing structure. The remaining structure will remain in place and be buried during new dock fill operations, eliminating nearly all demolition for this project. This recommended design will provide the Haines Borough with a world class multiuse facility for the next 50 plus years. The recommended facility will function well year-round as a freight/moorage dock. The proposed design also incorporates environmental sustainability elements, and provides in Turnagain's opinion, the best possible solution for meeting the goal of substantial completion by December of 2024 within or below budget.

B. Strategies & Design Ideas

I. IDEAS FOR CREATING SPACES THAT WILL HAVE FLEXIBLE USE OVER TIME

As most ports in Alaska service many different types of vendors and vessels, large and small, it is understood that the new Lutak facility must provide flexibility to service a number of different vessel types and sizes. Turnagain assists in creating flexible spaces through the following features:

FENDERING

Turnagain will reach out to the Borough for a list of vessels that will utilize the facility. After review of the vessels, Turnagain will develop a fendering system that caters to all vessels at different locations along the face of the dock. The properly designed and spaced fender system will encourage use of the dock by a number of different vessels and industries.

MOORING DEVICES

To accompany the well-designed fender system, Turnagain will provide mooring bollards and cleats that provide safe mooring capability for a wide range of mooring lines and vessels. Devices will be spaced according to vessel requirements provided during stakeholder review.



BARGE RAMP ACCESS

Utilizing a full 700-foot bulkhead dock approach, Turnagain will have the ability to develop a docking face that has multiple barge access points in the form of notches, similar to what is provided at the Alaska Marine Lines facility in Seattle, WA. Utilizing a full 700-foot dock face with multiple ramp locations the facility would be set up to accept multiple barges at any given time. In addition to the multiple ramp locations, sections of bullrail along the face of the dock will be made removable to allow for additional flexibility for vessel transfers. In both instances vendor and stakeholder input will assist in determining prime locations for ramp and removable bullrail locations.

FULL UTILITY PACKAGE

As the type and size of vessels that utilize the facility will vary greatly, Turnagain will provide a utility suite at multiple points along the face of the dock to accommodate multiple vessels at once. Turnagain will work with the Borough and its stakeholders to determine what utilities are required and in what locations.

II. INCORPORATING STAKEHOLDER INPUT INTO THE DESIGN

Prior to and during design the Turnagain team will reach out to the Haines Borough and its stakeholders for design coordination in efforts to provide a facility that fits a variety of vessels and industries. A design development kick off meeting will be scheduled after the project has been awarded. During this meeting Turnagain will welcome initial thoughts and ideas from the Borough and stakeholders to assist in design development. During design development up to 65% design submittal, Turnagain will hold weekly meetings with the Borough and stakeholders of their choosing. Meetings will focus overall design progress and on the salient features, such as fender system layout, barge ramp access locations, bollard spacing, etc., that effect the end stakeholders.

In addition to weekly meetings, the Borough will be provided a two-week review period after 35% and 65% design package submissions to provide comments on the design development, overall layout and facility features. At the completion of these review periods a meeting will be held to discuss the comments provided. Having all the major comments addressed during 35% and 65% review allows Turnagain to develop 95% and IFC drawings very efficiently.



III . INNOVATIVE CONSTRUCTABILITY SOLUTIONS THAT COULD REDUCE OVERALL BUDGET

As Turnagain has assessed the project it has developed a design and construction plan that allows flexibility in construction means and methods and also eliminates the majority of the demolition requirements of the previously supplied concepts. Turnagain will build the new pipe-pipe bulkhead wall offshore of the existing bulkhead structure, encapsulating the existing structure. After the new pipe-pipe bulkhead is installed, the existing structure will be buried as is and not require removal. Burying the existing bulkhead structure reduces overall project budget and significantly reduces unforeseen environmental impact.

In addition to eliminating demolition of the existing dock structure, Turnagain's proposed means and methods afford the project much flexibility during construction of the new facility. Building outside of the existing facility provides Turnagain the option to construct the new facility from the waterside via barge or from a landbased operation on the existing facility. Schedule constraints are not expected, but if necessary, Turnagain will be prepared to construct the facility from uplands and waterside.

The elimination of demolition of the new facility and the installation of the new facility outboard allows for much of the existing facility to remain open and operational during construction. Turnagain has had great success on previous projects at active freight and cruise facilities, working with municipalities and port departments to allow for active use of the existing facilities during construction. For instance, Turnagain constructed a new bulkhead structure for the City of Unalaska from 2017 to 2018, during this period the International Port of Dutch Harbor did not miss a port call. Turnagain will work with the Haines Borough and stakeholders to limit the amount of strain the construction places on freight and other services in and out of Haines.



LUTAK DOCK RFP

IV. TURNAGAIN LUMP SUM BULKHEAD OPTION



Turnagain has thoroughly reviewed the RFP and associated concepts and has developed a pipe-pipe bulkhead system that encapsulates the entire existing facility and Alaska Marine Lines barge ramp. The new facility will provide a longer dock face and an overall larger square footage than the existing facility. Turnagain will design and install this pipe-pipe bulkhead system for a Lump Sum Price of \$25,000,000.00. The longer dock face and added footprint provides the Haines Borough and its stakeholder added capacity and flexibility to utilize the community's resources to its full potential.

C. Challenges

One of the substantial challenges with developing the new design for this facility is providing the Haines Borough a facility design that provides the public and community the resource they need to provide stability and commerce for years to come. Turnagain understands this challenge and is developing a design that fits a number of vessels and industries very well rather than one vessel and industry perfectly. As discussed in the above section, Turnagain has a formal plan in place to ensure all community and stakeholder input is incorporated into the design of the new facility.

A second challenge that comes with any marine construction facility design and installation in Alaska is the variability in geotechnical conditions. During design and construction Turnagain will assume all geotechnical risk associated with the facility location. The Turnagain team has worked on projects that require shallow bedrock foundations throughout Southeast Alaska and employs a fleet of specially trained employees and equipment to successfully install piling in these conditions. The team has also previously installed socketed and drilled shaft piling at this location while constructing the new AML ramp and dolphins. Turnagain understands the conditions at this particular site far better than any other contractor.



An additional challenge that the Turnagain team has discovered when developing design concepts was the incorporation of the north end existing structures into the new facility. The north existing structures are relatively new, being constructed in Fall of 2020. After developing concepts Turnagain has identified ways to incorporate the existing structures into the new design. The main structure of concern on the north end of the facility is the three-pile berthing dolphin. The dolphin, constructed in 2020, was installed utilizing a drilled shaft foundation, followed by concreting the piles to just below the cap. The installation technique of this dolphin makes it extremely difficult to remove from its current position. The dolphin in its current position is useless to the new proposed facility that includes a berthing line that is shifted four feet offshore. The Turnagain design team has developed a concept that adds additional pile and framing to the existing dolphin that will push the tire fendering out to align with the new berthing face. Utilizing the existing dolphin as well as expanding the berthing face to the north affords the new facility approximately 190-feet of additional berthing space, 100-feet of bulkhead and 90-feet of open space to the berthing dolphin.

D. Communication & Collaboration

To ensure proper involvement by the Haines Borough and its stakeholders, Turnagain will utilize Fonn project management software for the implementation of this project. Fonn will be utilized during design, construction and closeout to provide efficient design document, submittal and RFI review. Different users will be provided different access restrictions depending on relevance to the project. Design documents, construction work plans, product data, etc. will be submitted to multiple parties for final review and buy off prior to implementation into the project.

Additionally, as discussed in previous sections of this RFP response, the project team will utilize weekly design meetings. These weekly meetings will be continued into the construction and closeout of the project. Communication and collaboration will continue throughout the construction phase to address any field conflicts or changes in design that may be required.

Subr	nittals						+ Create submittal	å import i
iD	status Any •	TITLE	PACKAGE	SECTION REFERENCE	SUBMITTER	SUBMIT BY Dette	APPROVER	APPROVE BY Disto
#64781	New	Test 1						
#60038	Now	elevators			nate@form.io			
#62453	Waiting for approval	Framing			eric@font.io		natei@form.io	Jan 8, 2021
#62450	Rejected	Elevator			eria@foran.ia		nate@fonn.io	Jan 8, 2021
#52686	Approved	1	±3.	elavator	nate@fonn.io		nate@tana.io	Oct 10, 2020
#52585	Approved	Framing	Framing	7	nate@fann.io		nate@fonn.lo	Oct 23, 2020

3. Project Controls, Cost Tracking and GMP Development

A. Three Strategies for Exceeding Project Goal 3

Turnagain excels at delivering quality facilities under fast-paced timelines while maintaining the budget. The team at Turnagain benefits from the efficiency gain of having an in-house design team, allowing Turnagain to be nimble during the design and costing efforts, and to adapt and grow the design with the owners and stakeholders in each project. Some key strategies that Turnagain employs are listed below.

TEarly and continuous collaboration between design and construction teams—The design and construction teams at Turnagain are all located in the same office. Both teams are involved in the entire process creating an environment where concepts can be proposed, vetted, and estimated in almost real time. With stakeholder involvement, this means that Turnagain can quickly turn around conceptual drawings and packages to facilitate scope definition. Turnagain has leveraged this collaborative structure successfully on past projects with tight schedules and budgets.

2 Front-loaded design efforts and early permit submittal—Once contracted and with a defined scope reached through stakeholder collaboration, Turnagain immediately invests significant effort into a robust early design package through 35% drawings. The in-house engineering team can deliver high-confidence design packages quickly to release long-lead procurement packages and maintain project schedule ahead of the finishing design touches. The team also frequently looks to shift fabrication to controlled environments where possible including for weldments and pre-cast concrete. Not only do these front-end efforts reduce on-site construction time, but they increase the quality of the final product. Early release packages are key to allowing for prefabrication efforts to be planned and executed within the project timeline.

Additionally, as soon as a concept is locked down and agreed upon with the stakeholders, Turnagain will release permit drawings and apply for the construction and IHA permits. Through close collaboration and history with local permitting agencies and subcontractors Turnagain has developed a process where we are able to submit permit applications early with conservative but realistic information, fast-tracking the project timeline.

Risk tolerance and sharing—Turnagain is willing to take on reasonable risk to keep a project moving forward. In the past, Turnagain took on the risk of uncertain geotechnical conditions on a project with a fixed budget. The team arrived to the project for construction prepared to address 3 different field conditions. In the end, bedrock was 100% deeper than planned in one of three footings, and 75% shallower than planned in one of three footings. Turnagain completed the project even with the varying site conditions within budget and at the previously agreed upon GMP.

Through a project lifecycle, there are going to be risks that can not be completely mitigated for a reasonable cost. Turnagain recognizes this and is willing to work with the owner to share and/or completely assume the risk where it makes sense operationally, and where Turnagain is best suited to do so.

LUTAK DOCK RFP

B. Monitoring, Reporting & Managing Costs

Turnagain leverages several supporting project management software systems for reporting and tracking progress and deliverables. These include Viewpoint Spectrum, Fonn, and Primavera P6.

Ι.

Turnagain will contractually commit to providing 4-each project concepts that can be constructed **within the available project funds**. Turnagain excels at Design-To-Budget project development. Too often, firms design projects that exceed the available funds jeopardizing the project schedule, wasting resources, and increasing design cost. For Turnagain, a project is only a success if it gets constructed and meets the Owner's expectations once in service.

Viewpoint Spectrum will be used to track all project related costs including subcontracts, Purchase Orders, and invoicing. Spectrum provides a real-time snapshot of the overall project finances. Periodic monthly reports will be made available to the owner, and more frequent reports can be pulled by request.

Π.

Communication is key to maintaining a successful and high-quality project delivered within budget and on schedule. Turnagain is practiced at delivering early fixed price projects. With the tools and experience the team has, we will present a baseline schedule, concepts, and pricing early in the collaboration process as a starting point for discussions. The presented information will be open for refinement, comment, and negotiation, but will also be ink ready. Turnagain is willing and capable of progressing this project at the pace desired by stakeholders.

These baseline exhibits will be set and maintained for the duration of the project. Fonn, will be our communications platform for information and documentation sharing with all project stakeholders and involved parties. Leveraging Fonn in conjunction with Spectrum, all costs, submittals, contracting, and other documentation will be made readily available.

III .

Turnagain has a full spectrum of design-build subcontractors that we have successfully worked with in the past and who are ready to take on sub scopes within the project. The estimating, construction, and structural-civil design professionals who will be working on this project are co-located in Anchorage Alaska, specifically for the purpose of effectively planning and executing Alaskan design-build projects. Sub-consultants for mechanical and electrical have been vetted by Turnagain on previous successful projects. The project team will be available to meet with the Haines Borough and other stakeholders virtually or in person at their Haines facility headquarters throughout all stages of the contract. Previous work has fostered valuable working relationships with these subcontractors and subconsultants. Their input is valued and acknowledged. Turnagain knows areas where we provide the best value and recognizes where to lean on outside expertise.



ν.

Turnagain has developed similar concepts to 65% design and has priced a similar project within the last 6 months. We know the market and have a current feel for costs. We keep tabs on and maintain good working relationships with our supply network. Through these connections Turnagain is often able to identify materials of opportunity or surplus materials and lock down prices that may not otherwise be achievable. With our previous experience and network, Turnagain remains confident that we can provide and perform to a GMP within the Borough's budget despite market conditions. Highlighted contributors to our success include:

- Extensive vendor network
- Recent experience pricing similar concepts
- Strong relationships with permitting agencies and subcontractors
- Past history of repeat clients speaks to our capacity to deliver best value and highest quality facilities

C. Phase 1 Level of Effort

Lutak Dock Replacement							
	Phase 1 Deliverables						
Exhibit C 1.03							
Item	Scope Item	Key Team Members	Hours				
A	Review analyze and validate the initial BOD	Josh Zellmer, PE	20				
		Chris Nielsen					
A	Review analyze and validate initial budget	Jason Davis	20				
A	Review analyze and validate initial schedule	Chris Nielsen	16				
		Chris Nielsen					
A	Review analyze and validate Commercial terms	Jason Davis	24				
		Josh Zellmer, PE					
A	Review analyze and validate other "Owner Provided Information"	Chris Nielsen	80				
		Chris Nielsen					
В	Perform Site investigations	Jason Davis	48				
В	Perform environmental assessments	Solstice AK	80				
В	Review regulatory and legal authority and restrictions	Solstice AK	80				
		Josh Zellmer, PE					
C	Collaborate with owner to develop new concepts	Jason Davis	80				
		Josh Zellmer, PE					
D	Study Sustainability objectives in BOD	Jason Davis	32				
		Josh Zellmer, PE					
E	Collaborate with owner on design and functionality to develop final BOD	Chris Nielsen	40				
		Josh Zellmer, PE					
F	Develop Collaboratively the Final BOD	Chris Nielsen	40				
F	Develop collaboratively final schedule	Chris Nielsen	40				
F	Develop collaboratively final GMP	Jason Davis	48				
G	Phase 1 Schedule	Chris Nielsen	32				
G	Prelim SOV	Chris Nielsen	16				
G	Prelim cost model	Chris Nielsen	16				
G	Subcontractor Procurement Procedure	Chris Nielsen	16				
G	Project Safety and Job Hazard Analysis	Chris Nielsen	40				
G	BIM Protocol	Josh Zellmer, PE	32				
G	Prelim Project Schedule	Chris Nielsen	32				
G	Monthly Schedule Updates	Chris Nielsen	16				
G	Monthly SOV updates	Chris Nielsen	16				
Н	GMP Proposal	Jason Davis	32				

D. Communication Deliverables

Turnagain will provide the following to aid in communications and transparency through the Design-Build process.

FONN

 Fonn access for real time status visibility and file share between all stakeholders and the Design-Build team.



P6 SCHEDULE PDF DOCUMENTS

 Updated schedules during development at two-week intervals through the Phase 1 process

SCHEDULE OF VALUE PDF DOCUMENTS

Updated SOV at two-week intervals through the Phase 1 process. Example below.

CONTRACTOR SCHEDULE OF VALUES

					TUR	ES - LUTAK DOG KAGAIN MARINE	K REPLACEMENT CONSTRUCTION		PAD	INVOICE DATE ESTIMATE NO		
CROK	OTAL TELE	A DESCRIPTION	(an	1234	UNITPHECK	FOE FOTAL	Ball Areas	CLARSING M	TODATE	IS COMPL	\$ VALUE 70	# PERMENING
1	1	MOBIL/ZATION	4	1.5	\$0.005	\$5.00	1 .	1 .	1 .	0.00%	1 .	5 .
2	7	PERMITS	1	1.3	30.00	90.00	1 1	1 7	1 1	0.00%	1 .	3 7
1	3	MARINE MANIMAL MONITORING & DUALITY CONTROL	1	6.0	\$0.00	\$0.00	\$ 1	1 .	1 1	0.00%	1 .	\$
4	4	TRESTLE & TRANSFER SPAN	1	1.5	\$0.00	\$0.00	5 .	1 .	\$ *	0.00%	3	5 .
4	5	INSTALL PLECTRICAL SYSTEM	11	1.5	\$45.66	\$8.65	1 4	1 .	1 .	6.00%	1 .	1 .
3	2	OFLAND WORK	1 3	11.0	\$0.00	\$6.00	\$.	1 .	3 .	0.09%	15 .	3 .
12	72	WATER CATHOENC SYSTEM	3	5.0	10.00	\$5.65	F	1 1	1 ×	0.00%		1 .
			207ALS			1 1	5 ,	1 .	1 .	8.59%	1	

MANAGEMENT OF CHANGE LOG

Updated and maintained in real time and uploaded to Fonn

Change Log									
Changed T B	ata 🔚 İnklator	T Description	T Same	Program Upd ata	Addon Regul	a T Ball In Court	Note a	Contingent	
	7/12/2022 John Smith	Example sharps order 1	G ale ded	First ded day to dia age ement	STATE TRANSPORT	ala internet	PARTY PROPERTY AND ADDRESS	P AC - TP ICANDING A STATE	
2	3/5/2022 John Smith	Example shange order 7	A 22-24/8	Finalized and information disseminated	1.5	10			
1	A/1/2022 Islan South	Example stange order 3	Printing	Rease with me quicted	Vies	John Sector	Contraction of the local division of the	AND A DESCRIPTION OF THE PARTY OF	

RISK MATRIX

- To be developed and reviewed with the Borough through Phase 1
- Maintained and referenced through project execution

Risk Register				Consequenc	е		
		Negligible	Minor	Moderate	Significant	Major	Risk Legend:
	Almost Certain						Low
poc	Likely						Medium
liho	Possible						High
Like	Unlikely						Very High
	Rare					Sound St.	Extreme

4. Construction Management, Sequencing & Scheduling

A. Plan

Turnagain has the capability of delivering this project on or ahead of schedule.

EFFICIENCIES IN SCHEDULING

Working the design simultaneously with the construction team and stakeholder input creates an environment where Turnagain maximizes productive planning for the project and eliminates disputes between the design and construction teams. Our design team continually communicates and refines design with the management team and stakeholders to optimize constructability and minimize project Turnagain excels at designing to a budget and providing early cost certainty. By identifying accurate construction cost early in the design phase, solutions and optimization can occur without impacting the schedule or the quality and scope of the development.

cost impact and duration. This eliminates the backtracking that often happens even under design-build contracts where the design and construction teams are not all under one roof. The result is a refined design that often is almost shop-drawing ready by the 65% drawing package or even sooner. An added benefit to this is the speed with which our team can develop the design and release work packages.

Turnagain plans to execute this project in the same manner. Upon award, we will focus effort to work towards a cohesive concept with the stakeholders and get permitting packages submitted. From there, we will detail the design with focus on long lead procurement packages that can be released individually. With the permitting in the works and long leads on order, Turnagain will shift gears to detailed construction planning and optimization of our on-site construction efforts. Turnagain has successfully implemented this process in delivering projects on-time or early for clients in the past.

EFFICIENCIES IN CONSTRUCTION SEQUENCING

Our team is currently in the planning stage of design and construction a 1400' long O-pile bulkhead structure. Many of the construction methods will carry over to the owner's concept for the Lutak dock replacement project. Together with the owner, Turnagain will develop the concept into a design that best satisfies all owner and functional goals of the project. Once defined, Turnagain will apply our past knowledge of varying construction techniques to select the most efficient means of construction with the best chance for success at the location. Turnagain is committed to maintaining schedule and budget, and regardless of whether the best method is the most economical or least economical, Turnagain will not let that dictate the direction we proceed and will not pass on unexpected cost to the owner once under contract at an agreed upon GMP.

	Optimizing CONSTRUCTABILITY - Our proven methodology
	Constructability and value integration are best
	achieved when our specific project development
	process is followed.
	UNDER STAND
	Do whatever it takes to understand the end users'
	needs inside and out. Know what is important to
	them, know their priorities, understand where they
	are flexible, and respect what operational conditions
	cannot be compromised.
	E VALU ATE
	E valuate all potential concepts to select the
	structural solution that best achieves the project
	objectives. Consider how environmental elements.
-	
	contractor capabilities, project risks, and lifecycle
	contractor capabilities, project risks, and lifecycle costs effect overall value.
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level characteristics of the primary elements
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level characteristics of the primary elements REFINE
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level characteristics of the primary elements REFINE After primary elements are selected engage our
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level characteristics of the primary elements REFINE After primary elements are selected engage our company resources and industry partners to
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level characteristics of the primary elements REFINE After primary elements are selected engage our company resources and industry partners to collaboratively optimize every detail and element that offeet value
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level characteristics of the primary elements REFINE After primary elements are selected engage our company resources and industry partners to collaboratively optimize every detail and element that effect value.
	contractor capabilities, project risks, and lifecycle costs effect overall value. DEVELOP Progress the design sufficiently to identify primary element materials, sizes and styles. Initiate outreach to vendors and fabricators to optimize the high-level characteristics of the primary elements REFINE After primary elements are selected engage our company resources and industry partners to collaboratively optimize every detail and element that effect value. OPTIMIZE

Haines Lutak				c	Classic	Schedule Layout											1	07-Jul-22	10:11
Activity ID	Activity Name	Original Start	Finish	Remaining	20)22		Ana Standard	New York	202	3		h-mark	日日書	aut i d	2	024		
		Duration	72 34863	Duration	Jun	Jul Aug Sep	Oct Nov Dec	Jan Feb Ma	r Apr May	y Jun	Jul Aug	Sep Oct	Nov De	ec Jan	Feb I	Mar Apr	May .	Jun Jul	Aug
💼 Lutak-Propo	sal Haines Lutak	533 30-Jun-22	15-Jul-24	533					1 1	1 1		1	1 1	1		1	1 1	T	15-Jul-2
두 Lutak-Prop	oosal.PD Preliminary Design Phase	91 01-Aug-22	05-Dec-22	91			05	Dec-22, Lutak-Pit	pposaLPD Pr	reliminary	Design Phas								
🖨 A1040	DEVELOP DRAFT 35% PLAN SETS WITH STAKEHOLDER INPUT	14 01-Aug-22	18-Aug-22	14		DEVE	OP DRAFT 35% P	AN SETS WITH	STAKEHOLD	ER INPL	T (4EA)						1 1	1	
🚍 🚍 A1050	BOROUGH REVIEW AND COMMENTS ON DRAFT 35% PLAN SE	7 19-Aug-22	29-Aug-22	7		во	OUGH REVIEW A	D COMMENTS	ON DRAFT 3	5% PLAN	SET	1		1			1 1	1	
🚍 🚍 A1060	ADDRESS COMMENTS AND SUBMIT 35% PLAN SETS (4EA)	7 30-Aug-22	07-Sep-22	7			DDRESS COMMEN	TS AND SUBMIT	35% PLAN S	SETS (4E	A)						J		
🚔 🥽 A1110	PROVIDE INITIAL CONSTRUCTION ESTIMATES 35%	7 30-Aug-22	07-Sep-22	7	1	P P	ROVIDE INITIAL CO	N STRUCTION E	STMATES 3	5%								1	
🖨 🤤 A1070	BOROUGH REVIEWS AND SELECTS 2-EA TO ADVANCE	21 08-Sep-22	06-Oct-22	21	1	- -	BOROUGH RE	VIEWS AND SEL	EÇTS 2-EA T	OADVA	ICE	1	1 1	1			1 1	1	
🖨 🧰 A1080	DEVELOP 65% PLAN SETS (2EA)	14 07-Oct-22	26-Oct-22	14			DEVELOP	55% PLAN SETS	(2EA)			1					1	1	
🚍 🚍 A1120	PROVIDE FIRM FIXED PRICING OF 65% CONCEPTS	14 07-Oct-22	26-Oct-22	14	1		PROVIDE	FIRM FIXED PRIC	CING OF 65%	6 CONCE	PTS	1	1 1		1		1 1	1	
A1090	BOROUGH REVIEWS AND COMMENTS ON 65% SETS	14 27-Oct-22	15-Nov-22	14	j		BORO	GH REVIEWS A	ND COMME	NTS ON	5% SETS						11		
🚍 🚍 A1100	COMMENTS ADDRESSED AND PREFERRED OPTION SELECTE	14 16-Nov-22	05-Dec-22	14	1			MMENTSADDR	ESSED AND	PREFER	RED OPTIO	SELECT	ED						
🚡 Lutak-Prop	oosal.Design Desigh Phase	35 06-Dec-22	23-Jan-23	35	1		-	7 23-Jan-23	Lutak-Propo	sal.Desig	n Desigh Ph	ase					1 1	1	
🖨 A1130	PREPARE 95% SUBMITTAL OF PREFERRED SOLUTION	14 06-Dec-22	23-Dec-22	14	1		+	PREPARE 95%	SUBMITTAL	OF PREF	ERRED SOL	UTION	1 1		1	1	1 1		
A1140	BOROUGH REVIEW AND COMMENT ON 95% SUBMITTAL	14 26-Dec-22	12-Jan-23	14	1		-	BOROUGH	REVIEWAN	DCOMM	ENT ON 95%	SUBMIT	AL				1 1	1	
🚍 A1150	ISSUED FOR CONSTRUCTION DOCUMENTS APPROVED	7 13-Jan-23	23-Jan-23	7	1			ISSUED P	OR CONST	RUCTION	DOCUMEN	TS APPRO	OVED	1			1 1		
🗧 🚡 Lutak-Prop	oosal.PP Procurment and Permitting	240 06-Dec-22	06-Nov-23	240	T								▼ 06 No	v-23, Luta	*-Propo	sal PP P	ncument	and Perm	itting
🖨 A1160	PERMIT ACQUISITION 404-SECTION 10-IHA	240 06-Dec-22	06-Nov-23	240	1		+		SUX DI	-	-		PERM	ITACOU	SITION	404-SEC	TION 10-	HA	1
🖬 🧰 A1170	FABRICATION AND MATERIAL PROCUREMENTEARLY WORK F	240 06-Dec-22	06-Nov-23	240			l 4			-	Sector Con		FABRI	ICATION .	AND M	ATERIAL	PROCUR	EMENTE	ARLYW
📲 Lutak-Prop	oosal.CP Construction Phase	180 07-Nov-23	15-Jul-24	180	1												1		15-Jul-2
🚍 A1180	ON SITE CONSTRUCTION	180 07-Nov-23	15-Jul-24	180	1							1	4	-	i i	-			ON SIT
Tutak-Prop	oosal.PC Pre-Construction	22 30-Jun-22	29-Jul-22	22	1	29-Júl-22	Lutak-Proposal.PC	Pre-Construction				1		1		1		1	1
🖨 A1190	PROJECT INTERACTIVE MEETING	0 30-Jun-22*		0	1 .	PROJECT NTE	RACTIVE MEETING	30-Jun-22*							1 1				1
A1000	SUBMIT RFP RESPONSE	0 15-Jul-22*		0	1	SUBMIT RE	RESPONSE 15-J	1-22*		1					1 1		1 1	1	1
A1020	PROJECTAWARD	5 25-Jul-22*	29-Jul-22	5	1	D PROJEC	AWARD												

Actual Level of Effort Remaining Work Milestone	Page 1 of 1	TASK filter: All Activities	© Oracle Corporation

6[

B. Performance Requirements & Optimization

Though the project goals are well-defined in the RFP, Turnagain will place initial focus on working with stakeholders and through them, the local community to clarify and incorporate any additional objectives and features into the design of the dock. Turnagain, at the request of the owner, will attend in-person meetings with the stakeholders and/or town hall style meetings with the local community to field and address any questions or comments that arise. Turnagain is confident that we can work towards a solution that achieves or exceeds all goals and which is also accepted by the community. These initial efforts are recognized as paramount to delivering a high quality and high value facility to not only the owner, but the community as well.

Meeting the project goals with design is only the first step. The project must be executed to the standards set by the both the stakeholders and Turnagain ourselves. On the Lutak Dock Replacement project, Turnagain's construction staff and design team will implement a system of checks and balances, audit practices, deficiency logs, management of change, and quality management processes that ensure the design is properly implemented in the field.

Through the design process, Turnagain will coordinate peer review for submitted packages and calculations. Peer review will include review of calculations, design, package compilations, constructability, risk, and cost impact. Findings will be documented and communicated with appropriate parties. Turnagain takes pride in having a history of successful lowest cost bids and project completion with zero contractor-initiated change orders. A major contributor to this successful record is the readily available design team, their attention to detail, and the teamwork of the entire Turnagain crew in delivering projects that meet or exceed Owner spec. With our design team in the home office, review during all project phases occurs naturally and expeditiously to create the highest value deliverable project.

During construction, although Turnagain maintains a dedicated quality manager to oversee and lead quality review efforts, our drive to provide top quality results is a culture within our crews. Turnagain fosters a "see it, own it" mentality that filters down to each and every crew member. Any member of the team, subcontractor, or direct hire is encouraged to ask questions and raise a question if something looks out of place. Turnagain recognizes that delivery of a high quality final product that meets all specifications is a whole team effort from the lead Project Manager, all the way through to the laborers performing the installation.



C. Maximizing Safety

Turnagain is committed to maintaining a safe work environment for employees. There is significant inherent risk to the work that Turnagain performs. However, with training, engineering controls, pre-work reviews, hazard identification workshops, incident reviews, and team communication, accidents can be prevented before they occur, and near misses can be learned from before they become an accident in the future.

Key principles in Turnagain's safety program include:

TRAINING

- Behavior and situational based on policy and relevant experience
- Effective initial hire foundational training
- Lessons learned from internal experience

CONTROLS

- Policy that is all inclusive based on EM-385, regional and federal standards
- Directives, memo's, instructions, and alerts used as immediate place holders for policy
- Enforced through clear and consistently enforced accountability standards

INCIDENT MANAGEMENT

- Immediate reporting of incidents to enable management of claims and obtain root cause as soon as possible
- Share, through effective messaging, lessons learned post incident
- Follow up with injured employees to ensure they understand the return-to-work process

SAFETY MANAGEMENT SYSTEM

Turnagain Marine has a developed comprehensive corporate safety program based on the OHSA18001-2007 Safety Systems which is compatible with quality and environmental standards to facilitate future integration of new hazards. This program currently has 21 policies and represents Turnagain Marine's full commitment to the health and safety of all employees and subcontractors. Our site-specific plan will address the specifics for each work activity as well as identify local support resources and facilities while being nested within the corporate safety policy. Training related to competent person level certifications is identified and planned at the supervisor level, based on the employees' title and risk profile. Below you will see the current list of safety policies of Turnagain Marine.

LUTAK DOCK RFP

S-01	INCIDENT REPORTING	S-12	PPE
S-02	SILICA	S-13	RESPIRATORY PROTECTION
S-03	SAFETY ACCOUNTABILITY	S-14	ERGONOMICS AND LIFTING
S-04	HAZCOM	S-15	OCCUPATIONAL ILLNESS
S-05	FALL PROTECTION	S-16	FORKLIFTS AND AERIAL LIFTS
S-06	TRENCHING AND SHORING	S-17	CRANES
S-07	LO/TO	S-18	SCAFFOLDING
S-08	TRANSPORTATION SAFETY	S-19	HOUSEKEEPING 5-S
S-09	S-09 INCLEMENT WEATHER		COVID-19
S-10	S-10 EMERGENCY SITUATIONS		RETURN TO WORK
S-11	CONFINED SPACE		

INNOVATIVE APPROACH

Our SSHO will closely monitor all subcontractor work taking place on the jobsite. The SSHO has the authority to rectify any safety related issues on the spot to include stopping work if necessary. All subcontractors will hold weekly internal safety meetings for definable features of work and provide evidence of these meetings to our managing staff. Additionally, all subcontractors working with Turnagain Marine will be evaluated using a comprehensive Safety Checklist. All checklist items requiring action will be corrected immediately. All subcontractor incidents are investigated in accordance with our site-specific safety plan and contract requirements. In addition to regular safety inspections conducted by the Site Safety and Health Officer, a third party Safety and Health Officer will conduct independent and random safety inspections and/or audits to discover, assess and correct unsafe working conditions or at-risk behaviors. The third party SSHO is not utilized for every Turnagain project, but findings and lessons learned from audits often apply to all jobsites and are incorporated into site specific policies.

D. Challenges

Often it is easy to get tunnel vision and chase a single concept without contemplating outside input. Turnagain's portfolio demonstrates the drive to uphold our client's best interest and provide mutually beneficial solutions to challenges encountered through the project lifecycle. Although TMC's proposed design checks all the boxes, we are open to comments that will increase the value of the result since ensuring exceptional project delivery is in all stakeholder's and Turnagain's best interest. Turnagain is committed to thoroughly vetting all design and construction comments regardless of their source. In the event of an impassable disagreement, Turnagain will defer to our own 3rd party design QC as well as any stakeholder independent consultants to collaborate on an objective solution.

Project Controls Challenges

Hesitance to report bad news

- When costs are increasing or schedule is slipping, nobody wants to deliver bad news
- Turnagain is committed to full transparency and early notification of changes
- Turnagain is willing to accept and share risk in certain circumstances agreed to in the performance contract

Delayed reporting

- When project controls are done through manual brute force efforts, controllers can get outpaced by the project which can lead to missed warning for potential impacts
- Turnagain will leverage software to facilitate forecasting, tracking, and document management through the project to stay ahead of any potential cost or schedule impacts

Compilation of data

- Project controllers must compile data from all sources of the project into a single, cohesive, and meaningful status
- Software utilization will streamline the data feeds to Turnagain management so that more effort can be spent interpreting data rather than simply formatting the data

Pricing often presents a sticking point in the project development process. Turnagain is committed to providing fair and transparent pricing throughout the project life cycle. The Haines Borough project team will be provided access to Fonn data sharing and project management platform along with P6 native files and scheduled PDF reporting updates. Further, Turnagain is confident in their proposed design to the point that our estimates will not lean on contingent sums to cover risk. Upon acceptance, Turnagain will be able to place firm cost estimates early in the project cycle, alleviating financial uncertainty.

Turnagain understands that the Haines Borough has never utilized Progressive-Design-Build project delivery method and looks forward to assisting in navigating the new challenges. Turnagain has successfully teamed up with Kodiak and Sitka to deliver their first PDB projects. All three projects were a great success for the community and led to additional design-build utilization.

E. Tools

Turnagain has already performed due diligence for this project based on the data made available in the RFP documentation. Turnagain will facilitate frequent open communication between all stakeholders, contractors, consultants, and subs involved in the successful completion of this dock replacement to deliver a final facility in alignment with all project goals.

Project Goals:

Design and construct a dock that maximizes the program requirements within the limited budget

Turnagain's ERP software, Spectrum along with our estimating software HCSS, will be utilized to first establish an accurate baseline budget, and then to track the spend progress and forecast any potential overruns prior to realizing them. Once under GMP contract, Turnagain will not pass on additional unforeseen cost to the owner unless previously agreed upon in the contract terms. With the financial tracking well in-hand Turnagain's primary efforts can be focused on refining the design and maximizing delivered value.

2 Execute a successful, collaborative PDB process to produce the envisioned process

Information will be shared through Fonn, keeping all stakeholders up to date on submittals, and project status. P6 will be utilized in conjunction to track the schedule and compare with financials to ensure that the project is on track. The open data sharing ensures that no time is lost to missed emails or delays in file sharing. With these tools and transparency, collaboration becomes an organic process. Milestones will be tracked, and comments vetted, addressed, and recorded. Turnagain will maintain updated versions to accurately project completion costs and dates.

3 Efficient pricing and scheduleSubmit sealed drawings to local authorities and assist with expediting the approval process.

Turnagain is prepared to work with the owner to develop a PDB process suitable for the scope of the Lutak dock replacement. However, Turnagain Marine is also prepared to present a firm fixed price lump sum contract for a replace-in-kind dock design which meets performance and budgetary goals. This approach would ensure the highest value and quickest delivery of the dock replacement.

Comply with legal requirements

Turnagain has developed a strong collaborative relationship with local permitting contractors. In 2020, Turnagain installed mooring and breasting dolphins for AML in the location where this work will be performed. We have also completed numerous marine construction projects in the state and are familiar with permitting and regulatory requirements local to the work site.

5 Design for safety

Turnagain will develop a site-specific safety plan for the work to be completed in Haines. All employees and subcontractors working on site will receive training and orientation on site relevant to construction. Additionally, utilizing prefabricated structures, we minimize laborer exposure to hazardous conditions during construction. These efforts also shorten the total on-site duration and allow for us to plan around scheduled operations to mitigate potential simultaneous construction and operation interference.

5. Identification of **Projects**

All relevant past projects referenced in Turnagain's Proposal were previously listed in the RFQ response. No additional projects have been referenced in this document.

ATTACHMENT A TO REQUEST FOR PROPOSALS HAINES BOROUGH LUTAK DOCK REPLACEMENT PRICE PROPOSAL FORM AND INSTRUCTIONS

I. INSTRUCTIONS

A. Design-Builder's Phase 1 Lump Sum for Overhead and Profit

The Design-Builder's Phase 1 Lump Sum for Overhead and Profit will, if agreed upon by the Owner, be inserted in Section 6.2.1 of the Progressive Design-Build Agreement between Owner and Design-Builder and should be based on the Phase 1 Not to Exceed Amount proposed in Section B below as well as the Phase 1 Level of Effort proposed pursuant to Section VI.B.3.c of the RFP. The parties will negotiate the Phase 1 Level of Effort, the Lump Sum for Overhead and Profit, and the Phase 1 Not to Exceed Amount after award.

B. Phase 1 Not to Exceed Amount

The proposed Phase 1 Not to Exceed Amount will be inserted into Section 6.6.1.2 of the Agreement. The Phase 1 Not to Exceed Amount will not be scored. However, if accepted by the Owner after negotiations, shall become binding on the successful Finalist, subject to the terms and conditions of the Contract Documents.

- a. The Proposed Phase 1 Not to Exceed Amount should include all compensation to the Design-Builder during the Phase 1 set forth in the Contract Documents, including but not limited to Exhibit C of the Agreement and proposed in the Phase 1 Level of Effort described in the Management Proposal.
- b. The Owner reserves the right to reconcile the various proposals received and also reserves the right to seek best and final proposals for the scope and the costof the Phase 1 Services and the Phase 1 Not to Exceed Amount; however, by submitting the Phase 1 Not to Exceed Amount, the Finalist warrants the following:
 - i. That the Phase 1 Level of Effort described in the Management Proposal is sufficient for the Design Build Team to perform the Work described for Phase 1 in the Contract Documents and provide the Owner with the Phase 1 deliverables as set forth in the revised Exhibit C proposed by the Finalist.
 - ii. That the Phase 1 Not to Exceed Amount is sufficient to perform the Work described in the Phase 1 Level of Effort in the Management Proposal.

C. Hourly Rates

Finalists will provide the hourly rates for Key Team Members. The Hourly Rates are not scored but will be incorporated into the Design-Build Agreement as Exhibit D. Separate rates shall be submitted for preconstruction and construction services should they differ.

D. Scoring of Price Proposal

The Design-Builder's Price Proposal shall be scored as follows:

The Finalist with the lowest Price Proposal will receive all fifteen points. The remaining Finalists will receive a proportionate share of the fifteen points, based on the proportion that the Price Proposal for their proposals exceeds the lowest Price Proposal. The points will be rounded to the next lowest whole number. No partial points will be awarded By way of example, if the second low Finalist proposes a Price Proposal that is fourteen percent higher than the lowest Price Proposal, the second low Finalist

Haines Lutak Dock Project Price Proposal Form Instructions

II. PRICE PROPOSAL FORM

Turnagain Marine Construction Corporation

Finalist Name

Having carefully examined the Request for Proposal (RFP) for Design-Build Services for the Haines Borough, Alaska Lutak Dock Replacement Project, issued <u>June 17th, 2022</u> and Addenda numbers <u>1</u> through <u>2</u>, and the Agreement, the undersigned Design-Builder proposes the following Commercial Terms for the Project:

A. Design-Builder Lump Sum for Overhead and Profit that will is proposed to be inserted into Section 6.2.1 of the Agreement: <u>Ten Thousand</u> dollars (\$ 10,000.00____)

B. Phase 1 Not To Exceed Amount (not scored)

The proposed Phase 1 Not to Exceed Amount is

<u>Three Hundred Thousand</u>
dollars (<u>\$300,000.00</u>)

C. Key Team Member Hourly Rates (not scored)

The Hourly Rates for Key Team Members are as follows:

Name	Position	Hourly Rate Preconstruction	Hourly Rate Construction	
Any	All	\$200.00	\$200.00	772 <u>146-07 146</u>
				4
ļ				
=				4

PROPOSAL GUARANTEE

The undersigned hereby agrees that this Proposal may be accepted by Haines Borough anytime within ninety (90) calendar days immediately following the date indicated herein below, and the undersigned further agrees to submit a fully executed Agreement prior to the issuance of the Notice to Proceed that includes the Commercial Terms proposed in this Price Proposal Form.

PROPOSAL FROM:

Turnagain Marine Construction Corporation

(Finalist Firm Name)

ason Davis

07 / 15 / 2022

(Authorized Representative Signature and Date)

Jason Davis, President

(Representative's Printed Name and Title)

CONE39620

(State of Alaska Contractor's License No.)

Alaska Business License # 1007444

Alaska Department of Commerce, Community, and Economic Development

Division of Corporations, Business, and Professional Licensing PO Box 110806, Juneau, AK 99811-0806

This is to certify that

TURNAGAIN MARINE CONSTRUCTION CORPORATION

8241 DIMOND HOOK DR UNIT A, ANCHORAGE, AK 99507

owned by

TURNAGAIN MARINE CONSTRUCTION CORPORATION

is licensed by the department to conduct business for the period

November 17, 2020 to December 31, 2022 for the following line(s) of business:

23 - Construction



This license shall not be taken as permission to do business in the state without having complied with the other requirements of the laws of the State or of the United States.

This license must be posted in a conspicuous place at the business location. It is not transferable or assignable.

Julie Anderson Commissioner Overall RFP Evaluation Summary Form Haines Lutak Dock Replacement

Scoring Criteria (Request for Proposals, Section IV.F)

Response to RFP		
	Overall Management Approach	30 points
	Maximize Design	15 points
	Project Controls, Cost Tracking and GMP Development	20 points
	Construction Management, Sequencing and Scheduling	15 points
Price Components		20 points
Total Points		100 points

Team Scores

	Overall Management Approach	Maximize Design	Project Controls	Constr. Mgmt	Technical Score	Price	
Pacific Pile	16	8	14	9	47	1	48
Turnagain Marine	25	15	18	12	70	20	90
Western Marine	20	8	10	13	51	10	61

Design-Build Team: Pacific Pile

Criteria	Strength	Weakness	Score
Overall Management Approach (max 30 points)	 The response is adequate They narrative demonstrates that they have depth and experience in the project type, and they have qualified field personnel and equipment. The narrative contained a good discussion regarding stakeholders. They have experience in Haines harbor. The discussion on collaboration was a positive. 	 In the meeting/interview the team talked at the Borough and didn't engage with the Borough. They seemed for more focused on themselves than the project. The approach was merely adequate and didn't really distinguish from the baseline requirements. People who were managing meeting were the executives, not the people in the field with whom the Borough would engage. Executive kept talking over the project manager. Teamwork appeared lacking. Previous experience with Borough had a lot of turn over with supervisory personnel (4 or 5 different people), and the project was not efficient. Further, the project ended in a claim. The previous project with Pacific Pile didn't demonstrate teamwork with owner. In the meeting, the team admitted that they didn't have experience with the public trying to stop the project. 	16
Maximize Design (max 15 pts)	 The team has experience. Pacific Pile has been involved with the Lutak Dock and evaluating the facility. Further, they have been following the progress of the project. They appear to have a good background knowledge and understanding of the project. 	 The proposal didn't consider anything other than what has already been suggested. They didn't suggest any innovation or any alternatives, contrary to the Project Goal of innovative design. The Port of Anchorage project is not a positive example. The team didn't discuss how design ideas would apply to project. They didn't demonstrate how the list of ideas would be applicable to this project. 	8

Design-Build Team: Pacific Pile

Total			47
Construction Management Sequencing and Scheduling (max 15 points)	 The narrative demonstrates that they are qualified. The discussion of tools and software was a positive. 	 The narrative contained a largely generic and boilierplate description of means and methods. There was no discussion of innovation. The discussion of challenges was more of a list and didn't provide much information regarding a solution or mitigation to the challenges The phasing discussion is problematic. 	9
Project Controls, Cost Tracking and GMP Development (max 20 pts)	 The discussion on strategies for estimating with frequent updates was good. 	 stakeholder input. The Borough's previous experience with PPM is that it is claims oriented. PND has tended to quickly shift away from responsibility when mistakes are made on projects. The company takes an aggressive posture in looking for extra costs. 	14
		 The Borough was uncomfortable with the suggestion of breaking up the permitting process. The discussion shows a lack of understanding of local concerns and 	

Design-Build Team: Western Marine

Criteria	Strength	Weakness	Score
Overall Management Approach (max 30 points)	 They have a great deal of experience with this dock area and dealing with some of the site conditions present. They presented a well organized team in the interview. They are very experienced. Garret Gadsjo is a strength; however, he is very busy and his capacity is low. The team has attended many public meetings, and they are well versed in stakeholder engagement. Knew specific individuals with stakeholders. The draft public involvement plan is exhaustive. There is a good discussion on risk management. They demonstrated high degree of confidence that they could perform the current design concept. 	 The design engineer didn't engage much at the meeting. They have many more subcontractors involved, which makes it more difficult to manage. In the meeting, indicated more of a committee approach, which may be inefficient Garret Gadsjo is very busy and may be overstretched. They discussed lean design and construction in the meeting, but they didn't discuss utilization in the proposal. 	20
Maximize Design (max 15 pts)	 The provided a detailed description of the previous ADOT work with details that other teams didn't emphasize, eg concrete pile cap, instability. They demonstrated a good understanding of the current design. The demonstrated a good understanding of code and permitting requirements. The project examples are helpful. 	 There was no innovation on design and didn't consider alternative design. What alternatives were provided were more based on construction means and methods. The Project Goals emphasize innovation in design. Their answer of permitting question was unclear regarding their understanding of current requirements. 	8
Project Controls, Cost Tracking and GMP Development (max 20 pts)	 The discussion on software was good. They understand the costs associated with where the rock for the project is sourced. The discussion regarding the importance of quality rock is a positive. 	 They didn't discuss lean tools such as target value design or design to budget. The cost estimates are at milestones rather than continuous estimating. They didn't demonstrate clear method to develop the GMP. Their example cost estimate is lump sum, not cost plus. 	10

Construction Management Sequencing and Scheduling (max 15 points)	 They had a good idea that they demonstrated in the meeting regarding demolition of the dock. They have done work at the dock, and this experience is very valuable to the project. They understand the challenges of the site and the current condition of the dock. 	 The construction phasing plan is not feasible because it doesn't accommodate users of the facility/account for current use; demonstrates lack of knowledge of how the dock is used. 	13	
Total			51	

Design-Build Team: Western Marine

Design-Build Team: Turnagain Marine

Criteria	Strength	Weakness	Score
Overall Management Approach (max 30 points)	 Management approach discussion regarding experience with public groups was a positive. Experience with Ward Cove was good. Their discussion during the meeting gave the Borough the confidence that the team could manage the stakeholder involvement. The company's management is focused around these types of design-build projects. They have demonstrated good success with many other communities. There is a lack of a learning curve for the project. The team is streamlined in their approach and very efficient with an integrated firm. In house engineers are focused only their work. This project will be focus for the company. They systematically break down the project focusing on individual elements to verify scope, schedule, and quality. They are willing to take on risk. Offered to take on differing site condition risk. The discussion regarding their in house engineers was good. Teaming with Solstice is a benefit. They have done a great deal of preparation into project and have a good sense of project The team demonstrates understanding of PDB. They asked good questions in the meeting. 	 The Borough was concerned that they are gambling on the project with taking on the differing site condition risk. Note: concern was addressed through subsequent questions. They don't have the depth of support personnel. The company might have difficulty juggling this project with others. It is unclear the design parameters in the lump sum. Note: clarified through subsequent questions. 	25
Maximize Design (max 15 pts)	• The new concept for the dock shows innovation, consistent with the Project Goals. The new approach eliminates "phase 3", which has been controversial within the community. This approach is a benefit to the community.	 Risk with the new concept; may not be supported by funding, community, etc. If the design pushes the dock face out, then the current berth needs modification. They 	15
Evaluation Summary Form Haines Lutak Dock Replacement

Design-Build Team:	Turnagain Marine
Design Duna ream.	runugun munic

Total		70	0
Construction Management Sequencing and Scheduling (max 15 points)	 The streamlined approach will facilitate scheduling. The specific discussion of project goals was a positive. They have an aggressive goal for scheduling. 	 This section was not as detailed as other sections. The new concept could delay the schedule to get approval. 	2
Project Controls, Cost Tracking and GMP Development (max 20 pts)	 The integrated team helps with efficient budgeting. The discussion on front loading the design and early permit submittal was a positive. The discussion on controlling escalation was a positive. They will provide estimated costs in real time, which is a significant strength. They will use design to budget techniques. They have 4 different project concepts. The software discussion was a positive. 	The 4 alternative concepts may not be efficient and may have detrimental effect with public involvement and schedule.	8
	 The discussion on locking down material specs for early procurement was a positive. They discussed concurrently advancing permitting and procurement. The Phase 1 deliverables discussion was good. 	 provided mitigating measure. Note: concern addressed during subsequent questions. Geotech is always a concern, and may be an issue, but the team is willing to take on the Geotech risk. There is a risk that they wouldn't be able to accomplish the method that proposing. Note: issue addressed during subsequent meeting. The narrative doesn't discuss impact of their design on the uplands. 	



PROGRESSIVE DESIGN-BUILD AGREEMENT BETWEEN OWNER AND DESIGN-BUILDER - WITH A GUARANTEED MAXIMUM PRICE

Note: This document contains differences from the DBIA Agreement form 530. Owner will provide a copy red-lined from the DBIA Agreement form upon written request.

Document No. 530 Second Edition 2010 © Design-Build Institute of America Washington, DC

TABLE OF CONTENTS

Article Name	Page
Article 1 Scope of Work	2
Article 2 Contract Documents	2
Article 3 Interpretation and Intent	2
Article 4 Ownership of Work Product	3
Article 5 Contract Time	4
Article 6 Contract Price	5
Article 7 Procedure for Payment	
Article 8 Termination for Convenience	
Article 9 Representatives of the Parties	
Article 10 Bonds and Insurance	
Article 11 Other Provisions	



Progressive Design-Build Agreement Between Owner and Design-Builder – with Cost Plus Fee and a Guaranteed Maximum Price

This document has important legal consequences. Consultation with an attorney is recommended with respect to its completion or modification.

This **AGREEMENT** is made as of the ______ day of ______ the year of 20 _____, by and between the following parties, for services in connection with the Project identified below:

OWNER:

Haines Borough, AK 103 Third Ave. Haines, AK 99827

DESIGN-BUILDER: (Name and address) Turnagain Marine Construction 8241 Dimond Hook Dr. Anchorage, AK 99507

907-261-8960

PROJECT:

Lutak Dock Replacement

In consideration of the mutual covenants and obligations contained herein, Owner and Design-Builder agree as set forth herein.

Article 1 Scope of Work

1.1 Design-Builder shall perform all design and construction services, and provide all material, equipment, tools and labor, necessary to complete the Work described in and reasonably inferable from the Contract Documents.

Article 2 Contract Documents

2.1 The Contract Documents are comprised of the following:

2.1.1 All written modifications, amendments, minor changes, and Change Orders to this Agreement issued in accordance with the *General Conditions of Progressive Design Contract Between Owner and Design-Builder* ("General Conditions of Contract");

2.1.2 The Phase 2 Amendment in accordance with Section 6.6.2 herein, provided such Amendment is executed between the parties;

2.1.3 This Agreement, including all exhibits but excluding the Phase 2 Amendment:

.1	Exhibit A:	Insurance Requirements;
----	------------	-------------------------

- .2 Exhibit B-1: Form of Performance Bond;
- .3 Exhibit B-2: Form of Payment Bond;
- .4 Exhibit C: Phase 1 and 2 Scope of Work;
- .5 Exhibit D: Owner's Program/Initial Basis of Design Documents;
- .6 Exhibit E: Design-Builder's Phase 1 Scope of Services and Hourly Rates;
- .7 Exhibit F-1: Phase 1 Change Order Form;
- .8 Exhibit F-2: Phase 2 Change Order Form;
- .9 Exhibit G: Form Phase 2 Amendment
- .10 Exhibit H: Required Federal Law Provisions
- .11 Exhibit I Proof of Insurance Form

2.1.4 The General Conditions of Progressive Design-Build Contract Between Owner and Design-Builder; and

2.1.5 Construction Documents prepared and approved in accordance with Section 2.4 of the General Conditions of Contract, provided the Phase 2 Amendment is executed between the parties.

2.1.6 Other documents as set forth in Exhibit C.

Article 3 Interpretation and Intent

3.1 Design-Builder, prior to execution of the Agreement, shall carefully review all the applicable Contract Documents, including the Owner's Program set forth in Exhibit D, for any conflicts or ambiguities. Design-Builder and Owner will discuss and resolve any identified conflicts or ambiguities prior to execution of the Agreement.

3.2 The Contract Documents are intended to be complementary and interpreted in harmony so as to avoid conflict, with words and phrases interpreted in a manner consistent with construction and design

industry standards. In the event inconsistencies, conflicts, or ambiguities between or among the Contract Documents are discovered after execution of the Agreement or after the parties' execution of the Phase 2 Amendment, Design-Builder and Owner shall attempt to resolve any ambiguity, conflict or inconsistency informally, recognizing that the Contract Documents shall take precedence in the order in which they are listed in Section 2.1 hereof.

3.3 Terms, words and phrases used in the Contract Documents, including this Agreement, shall have the meanings given them in the General Conditions of Contract.

3.4 If the Owner's Program contain design or prescriptive specifications, the Design-Builder shall be entitled to reasonably rely on the accuracy of the information represented in such design or prescriptive specifications and their compatibility with other information set forth in Owner's Program, including any performance specifications for the purposes of developing the Design-Builder's Phase 1 Scope of Services (Exhibit E), the Phase 1 Not to Exceed Amount and the Design-Builder's Lump Sum for Overhead and Profit. However, Design-Builder is required to perform an independent evaluation of such design or prescriptive specifications to verify the information provided by the Owner during Phase 1. Further, regardless of the inclusion of design or prescriptive specifications or criteria, Design-Builder shall remain responsible for meeting the performance requirements of the Project, including but not limited to the requirements that the Project meet the Basis of Design Documents as well as all applicable Legal Requirements.

3.5 The Contract Documents form the entire agreement between Owner and Design-Builder and by incorporation herein are as fully binding on the parties as if repeated herein. No oral representations or other agreements have been made by the parties except as specifically stated in the Contract Documents.

<u>Article 4</u> Ownership of Work Product

4.1 Work Product. All drawings, specifications and other documents and electronic data, including such documents identified in the General Conditions of Contract, furnished by Design-Builder to Owner under this Agreement ("Work Product") are deemed to be instruments of service and Design-Builder shall retain the ownership and property interests therein, including but not limited to any intellectual property rights, copyrights and/or patents, subject to the provisions set forth in Sections 4.2 through 4.5 below.

4.2 Owner's Limited License upon Project Completion and Payment in Full to Design-Builder. Upon Owner's payment in full for all Work performed under the Contract Documents, Design-Builder: (a) grants Owner a limited license to use the Work Product in connection with Owner's occupancy of the Project; and (b) transfers all ownership and property interests, including but not limited to any intellectual property rights, copyrights and/or patents, in that portion of the Work Product that consists of architectural and other design elements and specifications that are unique to the Project. The parties shall designate those portions of the Work Product for which ownership in the Work Product shall be transferred. Such grant and transfer are conditioned on Owner's express understanding that its alteration of the Work Product without the involvement of Design-Builder is at Owner's sole risk and without liability or legal exposure to Design-Builder or anyone working by or through Design-Builder, including Design Consultants of any tier (collectively the "Indemnified Parties"), and on the Owner's obligation to provide the indemnity set forth in Section 4.5 below.

4.3 Owner's Limited License upon Owner's Termination for Convenience or Design-Builder's Election to Terminate. If Owner terminates this Agreement for its convenience as set forth in Article 8 hereof, or if Design-Builder elects to terminate this Agreement in accordance with Section 11.4 of the General Conditions of Contract, Design-Builder shall, upon Owner's payment in full of the amounts due Design-Builder under the Contract Documents, grant Owner a limited license to use the Work Product to complete the Project and subsequently occupy the Project, and Owner shall thereafter have the same rights as set forth in Section 4.2 above, conditioned on the following:

4.3.1 Use of the Work Product is at Owner's sole risk without liability or legal exposure to any Indemnified Party, and on the Owner's obligation to provide the indemnity set forth in Section 4.5 below, and

4.3.2 Owner shall not be required to pay Design-Builder additional compensation for the right to use the Work Product to complete the Project and subsequently use the Work Product in accordance with Section 4.2 if Owner resumes the Project through its employees, agents, or third parties.

4.4 Owner's Limited License upon Design-Builder's Default. If this Agreement is terminated due to Design-Builder's default pursuant to Section 11.2 of the General Conditions of Contract, then Design-Builder grants Owner a limited license to use the Work Product to complete the Project and subsequently occupy the Project, and Owner shall thereafter have the same rights and obligations as set forth in Section 4.2 above. Notwithstanding the preceding sentence, if it is ultimately determined that Design-Builder was not in default, Owner shall be deemed to have terminated the Agreement for convenience, and Design-Builder shall be entitled to the rights and remedies set forth in Section 4.3 above.

4.5 Owner's Indemnification for Use of Work Product. If Owner is required to indemnify any Indemnified Parties based on the use or alteration of the Work Product under any of the circumstances identified in this Article 4, to the extent permitted by law Owner shall defend, indemnify, and hold harmless such Indemnified Parties from and against any and all claims, damages, liabilities, losses and expenses, including attorneys' fees, arising out of or resulting from the use or alteration of the Work Product.

Article 5 Contract Time

5.1 Date of Commencement. The Work shall commence within five (5) days of Design-Builder's receipt of Owner's Notice to Proceed ("Date of Commencement") unless the parties mutually agree otherwise in writing.

5.2 Substantial Completion and Final Completion.

5.2.1 Phase 1 shall be completed no later than ____2/15/2023_____ ("Phase 1 Completion Date"). The parties will establish a date for Substantial Completion of the entire Work ("Scheduled Substantial Completion Date") in the Phase 2 Amendment.

5.2.2 Interim milestones and/or Substantial Completion of identified portions of the Work ("Scheduled Interim Milestone Dates") shall be determined during Phase 1: (*Insert any interim milestones for portions of the Work with different scheduled dates for Substantial Completion*)

5.2.3 Final Completion of the Work or identified portions of the Work shall be achieved as expeditiously as reasonably practicable. Final Completion is the date when all Work is complete pursuant to the definition of Final Completion set forth in Section 1.2.14 of the General Conditions of Contract.

5.2.4 All of the dates set forth in this Article 5 (collectively the "Contract Time(s)") shall be subject to adjustment in accordance with the General Conditions of Contract.

5.3 Time is of the Essence. Owner and Design-Builder mutually agree that time is of the essence with respect to the dates and times set forth in the Contract Documents.

5.4 **Liquidated Damages.** Design-Builder and Owner recognize that timely completion of the Work is the essence of this Agreement and that the Owner will suffer financial loss if The Work is not delivered as

promised in accordance with the Agreement. These losses include employee overtime hours, additional wear and tear on Owner's alternate freight facility, losses of efficiency in shipment of goods to and from Haines with resultant increased costs of shipping, additional payments to consultants, lost opportunities for revenue from port fees and general public inconvenience. They also recognize that such losses multiply over time and that there is significant difficulty and expense in proving in a legal proceeding the actual loss suffered by Owner if The Work is not timely completed as promised. Accordingly, instead of requiring such proof, Design-Builder and Owner agree that as liquidated damages for failure to substantially complete the Work (but not as a penalty) within the time set for Substantial Completion by the contract terms including change orders Design-Builder shall pay Owner one thousand five hundred Dollars (\$1,500.00) for each day between the date set for Substantial Completion by the contract terms and the date Design-Builder Substantially Completes the Work. The Owner and Design-Builder may establish liquidated damages for other remedies during Phase 1 or as a Change Order or Amendment to the Agreement."

Article 6 Contract Price

6.1 Contract Price.

6.1.1 Subject to the provisions of the Contract Documents, the Owner shall pay Design Builder for each Phase of the Project in accordance with Section 6.6 of the Agreement. Design Builder's Compensation shall be subject to Phase 1 NTE and the GMP, as applicable, and Phase 1 NTE and the GMP, as applicable, shall be the maximum amount that the Design Builder may be compensated for the applicable Contract Phase. The maximum amount that the Design Builder may be compensated pursuant to this Agreement for any given phase shall also be referred to as the Contract Price ("Contract Price"). The elements of the Design Builder's Compensation, subject to the Contract Price are set forth herein. If the sum of the Design-Builder's Compensation is less than Phase 1 NTE and/or the GMP, the savings shall go to the Owner.

6.2 Design Builder's Lump Sum for Overhead and Profit

6.2.1 Design Builder's Phase 1 Lump Sum for Overhead and Profit shall be:

Ten thousand dollars (\$ 10,000.00).

- **6.2.2** If the Owner exercises its option to go forward with Phase 2 and the Parties enter into the Phase 2 Amendment, Owner and Design Builder shall negotiate a Phase 2 Lump Sum for Overhead and Profit.
- **6.2.3** The Lump Sum for Overhead and Profit will be earned and paid monthly on a percentage of completion basis and in accordance with the most recent Schedule of Values. If the Contract is terminated for any reason, the Design-Builder shall only be entitled to that portion of the Lump Sum for Overhead and Profit that represents the portion of Work completed in accordance with the Contract Documents.
- **6.2.4** The Lump Sum for Overhead and Profit for Phases 1 and 2 shall include the following items, which shall not be charged as either a Cost of the Work, or as part of the Lump Sum General Conditions Amount or any Allowance:
 - .1 All profit of the Design Builder for this Project;
 - .2 All regional and home office overhead expenses, including labor and materials, phone, facsimile, postage, internet service, and other incidental office expenses attributed to work on this Project; and
 - .3 All other direct and indirect costs incurred by the Design Builder that are not otherwise specifically identified in the Cost of the Work, the Lump Sum General Conditions Amount, the Design Builder's Contingency and/or any Allowance established by the Parties.

- **6.3 Cost of the Work.** The term Cost of the Work shall mean costs set forth in this Section that are reasonably and actually incurred by Design Builder in the proper performance of the Work. However, if the Owner exercises its option to enter into Phase 2 and the Parties enter into the Phase 2 Amendment, then the costs listed in Section 6.3.15 below as General Conditions Costs and included in the Lump Sum General Conditions Amount are excluded from the Cost of the Work. The term Cost of the Work shall include only the following:
 - **6.3.1** Direct labor costs of employees of Design Builder performing construction or design Work at the Site or, with Owner's agreement, at locations off the Site.
 - .1 The costs for those employees of Design Builder performing design or other services shall be calculated on the basis of prevailing market rates for design professionals performing such services or, if applicable, the Hourly Rates set forth Exhibit E.
 - .2 The costs for those employees of Design-Builder performing Work that is identified in Section 6.3.15 below shall, if applicable, be calculated on the basis of the Hourly Rates set forth in Exhibit E.
 - .3 Wages for any employees for whom there is not an established Hourly Rate shall be paid as follows: Basic wages and fringe benefits: The premium portion of overtime wages is not included unless pre-approved in writing by the Owner. The Design Builder shall provide to the Owner copies of payroll records, including certified payroll statements for itself and Subcontractors of any tier for the period upon the Owner's request. Direct labor costs include all costs directly associated with the employment of labor and include, but are not limited to, direct contributions for workers' compensation insurance and any other pension or insurance required by law or union agreements.
 - **6.3.2** Costs incurred by Design Builder for employee benefits, premiums, taxes, insurance, contributions and assessments required by law, collective bargaining agreements, or which are customarily paid by Design Builder, to the extent such costs are based on wages and salaries paid to employees of Design Builder covered under Section 6.3 hereof.
 - **6.3.3** Payments properly made by Design Builder to Subcontractors and Design Consultants for performance of portions of the Work, including any insurance and bond premiums incurred by Subcontractors and Design Consultants. The costs for those employees performing design services shall be calculated on the basis either the Hourly Rates set forth in Exhibit E or the prevailing market rates for design professionals performing such service. Contracts to Subcontractors and Design-Consultants that are paid on the basis of a Lump Sum must be approved in advance by the Owner, such approval shall not be unreasonably withheld.
 - **6.3.4** Costs, including transportation, inspection, testing, storage and handling, of materials, equipment and supplies incorporated or reasonably used in completing the Work. The material costs shall be based upon the net cost after all discounts or rebates, freight costs, express charges, or special delivery costs, when applicable. No lump sum costs will be allowed except when approved in writing in advance by the Owner. Discounts and rebates based on prompt payment need not be included, however, if the Design Builder offered but the Owner declined the opportunity to take advantage of such discount or rebate.
 - **6.3.5** Costs (less salvage value) of materials, supplies, temporary facilities, machinery, equipment and hand tools not customarily owned by the workers that are not fully consumed in the performance of the Work and which remain the property of Design Builder, including the costs of transporting, inspecting, testing, handling, installing, maintaining, dismantling and removing such items.
 - **6.3.6** Costs of removal of debris and waste from the Site.
 - **6.3.7** Rental charges and the costs of transportation, installation, minor repairs and replacements, dismantling and removal of temporary facilities, machinery, equipment and hand tools not customarily owned by the workers, which are provided by Design Builder at

the Site, whether rented from Design Builder or others, and incurred in the performance of the Work. The rental charge as established by the lower of the local prevailing rate published in the Rental Rates published by the Rental Rate Blue Book by Data Quest or the actual rate paid to an unrelated third party as evidenced by rental receipts. Rates and quantities of equipment rented that exceed the local fair market rental costs shall be subject to the Owner's prior written approval. Total rental charges for equipment or tools shall not exceed 75% of the fair market purchase value of the equipment or the tool. Actual, reasonable mobilization costs are permitted if the equipment is brought to the site solely for a change in the Work. The rental rates are the maximum rates allowable for equipment of modern design and in good working condition and include full compensation for furnishing all fuel, oil, lubrication, repairs, maintenance, and insurance. When rental rates payable do not include fuel, lubrication, maintenance, and servicing, as defined as operating costs in the Blue Book, such operating costs shall be reimbursed based on actual costs. Equipment not of modern design and/or not in good working condition will have lower rates. Hourly, weekly, and/or monthly rates, as appropriate, will be applied to yield the lowest total cost. The rate for equipment necessarily standing by for future use (and standing by for no longer than two (2) weeks') on the changed Work shall be 50% of the rate established above. The total cost of rental allowed shall not exceed the cost of purchasing the equipment outright. If equipment is required for which a rental rate is not established by The Rental Rate Blue Book, an agreed rental rate shall be established for the equipment, which rate and use must be approved by the Owner prior to performing the Work.

- 6.3.8 All fuel and utility costs incurred in the performance of the Work.
- 6.3.9 Sales, use or similar taxes, tariffs or duties incurred in the performance of the Work.
- **6.3.10** Costs for permits, royalties, licenses, tests and inspections incurred by Design Builder as a requirement of the Contract Documents.
- **6.3.11** Deposits which are lost, except to the extent caused by Design Builder's negligence or other fault.
- **6.3.12** Costs incurred in preventing damage, injury or loss in case of an emergency affecting the safety of persons and property.
- **6.3.13** Unit Prices established by the Parties.
- **6.3.14** Other costs reasonably and properly incurred in the performance of the Work to the extent approved in writing by Owner and not included in the Design Builder's Contingency, Design Builder's Fee, the Fixed Fee, or the Lump Sum General Conditions Amount.
- **6.3.15** General Conditions Costs. The following costs are reimbursable in Phase 1 as a Cost of the Work; however, if the Owner exercises its option to enter into Phase 2 and the parties enter into the Phase 2 Amendment, these costs shall be included in the Lump Sum General Conditions Amount set forth in Section 6.4.5 of the Agreement and shall not be included or reimbursable as part of the Cost of the Work in Phase 2.
 - .1 Wages or salaries of Design-Builder's supervisory and administrative personnel engaged in the performance of the Work and who are located at the Site or working off-Site to assist in the production or transportation of material and equipment necessary for the Work. Specifically, the following personnel are included in the Lump Sum General Conditions Amount:
 - a. Project Executive
 - b. Project Manger
 - c. Superintendent
 - d. Quality Control Manager
 - e. Project Engineer

- .2 Wages or salaries of Design Builder's personnel stationed at Design Builder's principal or branch offices, but only to the extent said personnel are approved in advance of the performance of the Work in writing by the Owner.
- .3 The reasonable portion of the cost of travel, accommodations and meals for Design-Builder's personnel necessarily and directly incurred in connection with the performance of the Work. As set forth below:

a. Meals and Incidental Expenses: Meals and incidental expenses will be limited to the Federal Per Diem rate for meals and incidentals established for the location where lodging is obtained. Federal Per Diem guidelines which includes the meal breakdown and Federal Per Diem rates for other locations can be found at <u>www.gsa.gov</u>.

b. Lodging: Lodging will be billed at cost, including applicable taxes, not to exceed the Federal Per Diem maximum lodging rate for the location where the work is being performed. The Owner may increase this limit in writing when circumstances require.

c. Travel: Air travel (at coach class or equivalent), airport shuttles, etc. billed at cost. Ground transportation by privately owned vehicle, if utilized, billed at the Internal Revenue Service mileage rate for privately owned vehicles in effect at the time of travel. Expenses for a rental car (including fuel), at cost, in the ratio of one mid-size class rental car for each three Contractor's personnel directly engaged in performance of the work at the prevailing rental rates then in effect. Rental car options such as refueling fees, GPS, collision & liability insurance, etc. will not be reimbursed by the Owner unless such options are approved in advance by the Owner's Representative. Appropriate insurance coverage should be included in the Contractor's insurance policies.

- .4 The reasonable costs and expenses incurred in establishing, operating and demobilizing the Site office, including the cost of facsimile transmissions, long-distance telephone calls, postage and express delivery charges, telephone service, photocopying and reasonable petty cash expenses.
- .5 Premiums for insurance and bonds required specifically by this Agreement or the performance of the Work by the Design Builder.
- .6 Accounting and data processing costs related to the Work.
- .7 Fees paid by the Design-Builder for the Notice of Work required by AS 36.05.045(a). The Design-Builder will remain responsible for the actual submittal of the Notice of Work to the Department of Labor.
- .8 General administrative costs not specifically listed in Sections 6.3.1 through 6.3.13 above, including but not limited to the following:
 - a. Shop Drawing Reproduction
 - b. Construction Schedule & Updates
 - c. Safety/Security
 - d. Field Office Set-up (mobilization/demobilization)
 - e. Office Supplies
 - f. Telephone System
 - g. Telephone Service Charge
 - h. Computer Network/System Set-up
 - i. Courier Service

- j. Postage (Fed-X, USPS)
- k. Furniture/Equipment
- I. Office Cleaning
- m. Project Superintendent Vehicle
- n. Computers
- o. Copy Machine
- p. Temporary Electric Hook-up/Removal
- q. Temporary Electric Material
- r. Project Signage
- s. Temporary Water Hook-up/Removal
- t. Drinking Water & Supplies
- u. Chemical Toilets
- v. O&M Manuals
- w. Project Record Documents
- x. Field Engineering/Layout Survey

6.4 Other Methods of Pricing

Within Phase 1 NTE or the GMP, the Parties may agree to the following methods of pricing:

6.4.1 Allowance Items and Allowance Values.

- .1 Any and all Allowance Items will be included in either Exhibit E or the Phase 2 Amendment and are included within any established NTE and the GMP, as applicable. The description of the Allowance Item shall include the scope of the Allowance Item and the estimated cost of the Allowance Item, (the "Allowance Value") and any assumptions regarding the Allowance Item. Design-Builder shall obtain written approval from the Owner for any Allowance Items for subcontractors.
- .2 The establishment of Allowance Items and Allowance Values by the Design Builder and the Owner are a representation that the Design Builder and Owner have worked together to review the Allowance Items and Allowance Values based on information then available to determine that the Allowance Values constitute reasonable estimates for the Allowance Items. Design Builder and Owner will continue working closely together to develop Construction Documents consistent with the Allowance Values. Nothing herein is intended in any way to constitute a guarantee by Design Builder that the Allowance Item in question can be performed for the Allowance Value.
- .3 No work shall be performed on any Allowance Item without Design Builder first obtaining in writing advanced authorization to proceed from Owner. Owner agrees that if Design Builder is not provided written authorization to proceed on an Allowance Item by the date set forth in the Project schedule, due to no fault of Design Builder, Design Builder may be entitled to an adjustment of the Contract Time(s) and the applicable Contract Price.
- .4 The Allowance Value for an Allowance Item includes the direct cost of labor, materials, equipment, transportation, taxes and insurance associated with the applicable Allowance Item. All other costs, including design fees, Design Builder's overall project management and general conditions costs, overhead and fee, are not included in the Allowance Value and are deemed to be included in the applicable Contract Price, and are not subject to adjustment, regardless of the actual amount of the Allowance Item.
- .5 Whenever the actual costs for an Allowance Item is more than or less than the stated Allowance Value, the applicable Contract Price shall be adjusted accordingly by

Change Order, subject to Section 6.4.1 above; however, Design Builder must provide written notice of the difference between the actual cost and the Allowance Value pursuant to Section 10.1 of the General Conditions. The amount of the Change Order shall reflect the difference between actual costs incurred by Design Builder for the particular Allowance Item and the Allowance Value.

6.4.2 Not To Exceed Sums

- .1 The Owner and Design Builder may establish Not to Exceed ("NTE") Sums for specific scopes of the Work. Any such NTE Sum will be negotiated between the Owner and Design Builder. The NTE Sum agreed upon by the Parties shall be incorporated into the Agreement via Amendment or a Change Order, and the Parties shall include the following information:
 - a. A specific description of the Scope of the Work that is subject to the NTE Sum;
 - b. An updated Schedule of Values that incorporates the NTE Sum; and
 - c. Any milestone dates associated with the scope of the Work associated with the NTE Sum.
- .2 For each scope of work for which a NTE Sum has been established, the Design Builder shall be reimbursed as set forth herein; however, Design Builder's Compensation for the scope of work in the NTE shall not exceed the NTE Sum without a written Change Order.
- .3 Design Builder must identify all costs that are subject to any applicable NTE in the Payment Application, and Design Builder may not also submit such costs under any other line item in the Payment Application.
- .4 NTE Sums may only be modified by written Change Order or Contract Amendment pursuant to the General Conditions.

6.4.3 Lump Sums

- .1 The Owner and Design Builder may establish Lump Sums for specific scopes of the Work. Any such Lump Sum will be negotiated between the Owner and Design Builder. Lump Sums agreed upon by the Parties shall be incorporated into the Agreement via Amendment or a Change Order, and the Parties shall include the following information:
 - a. A specific description of the Scope of the Work that is subject to the Lump Sum;
 - b. All line items that are identified as a Cost of the Work in Section 6.3 of the Agreement that are included in the Lump Sum;
 - c. An updated Schedule of Values that incorporates the Lump Sum; and
 - d. Any milestone dates associated with the scope of the Work associated with the Lump Sum.
- .2 For each scope of work for which a Lump Sum has been established, the Design Builder shall be compensated pursuant to the Schedule of Values based on the percentage complete of the Scope of the Work subject to the Lump Sum.
- .3 If any line item that is identified as a Cost of the Work in Section 6.3 of the Agreement is subsequently included in any Lump Sum, Design Builder shall not thereafter request reimbursement for those line items as a Cost of the Work. Design-Builder may, however, request reimbursement through the Design-Builder's Contingency set forth in Section 6.4.4.1.b of the Agreement.

.4 Lump Sums may only be modified via written Change Order or Contract Amendment pursuant to the General Conditions

6.4.4 Design Builder's Contingency

- .1 The Parties shall establish, as part of any NTE and the GMP, the following Contingencies, which are available for Design Builder's exclusive use for the below described unanticipated costs it has incurred that are not a Cost of the Work and not the basis for a Change Order under the Contract Documents (collectively "Contingency Items"). Contingency Items include the following costs, which are subject to written approval by the Owner. The Owner may, in its discretion, approve other costs that may be reimbursed under the Contingency; however, in no case shall the Design-Builder be entitled to use the Contingency for payment of Liquidated Damages that it may be assessed pursuant to this Agreement.
 - (a) Cost of the Work Contingency. The Cost of the Work Contingency is reimbursed as a Cost of the Work. The Cost of the Work Contingency is available to the Design-Builder for the following items:
 - (i) Trade buy-out differentials;
 - (ii) Escalation of materials; and
 - (iii) Other direct Costs of the Work that are not included in the Design-Builder's Contingency, but only with the prior written consent of the Owner.
 - (b) Design-Builder's Contingency. The Design-Builder's Contingency is available to the Design-Builder for items that are not excluded by Section 6.5 hereof and include but are not limited to the following items:
 - (i) Overtime or acceleration;
 - (ii) Costs incurred by Design-Builder in repairing or correcting defective, damaged or nonconforming Work (excluding any warranty or corrective Work performed after Substantial Completion), provided that such Work was beyond the reasonable control of Design-Builder, or caused by the ordinary mistakes or inadvertence, and not the negligence, of Design-Builder or those working by or through Design-Builder. If the costs associated with such Work are recoverable from insurance, Subcontractors or Design Consultants, Design-Builder shall exercise its best efforts to obtain recovery from the appropriate source and provide a credit to Owner if recovery is obtained;
 - (iii) Legal costs, court costs and costs of mediation and arbitration reasonably arising from Design-Builder's performance of the Work, provided such costs do not arise from disputes between Owner and Design-Builder;
 - (iv) Subcontractor or other tier defaults to the extent not compensated by any surety or bond; or
 - (v) Costs that are in excess of an NTE Sum or Lump Sum.
- .2 The Design Builder shall be reimbursed for Contingency Items in the same manner as set forth in Section 6.3 of the Agreement; however, Design Builder's Compensation for Contingency Items shall not cumulatively exceed the amount set forth as the Design Builder's Contingency in the applicable NTE or GMP without a written Change Order.
- .3 Prior to the final accounting, the Contingency is not available to Owner for any reason, including, but not limited to changes in scope or any other item which

would enable Design Builder to increase an NTE or GMP under the Contract Documents.

- .4 Design Builder shall provide Owner notice of all anticipated charges against the Contingency and shall provide Owner as part of the monthly status report required by the General Conditions of Contract an accounting of the Contingency, including all reasonably foreseen uses or potential uses of the Contingency in the upcoming three (3) months. Design Builder agrees that with respect to any expenditure from the Contingency relating to a Subcontractor default or an event for which insurance or bond may provide reimbursement, Design Builder will in good faith exercise reasonable steps to obtain performance from the Subcontractor and/or recovery from any surety or insurance company. Design Builder agrees that if Design Builder is subsequently reimbursed for said costs, then said recovery will be credited back to the Contingency.
- .6 At the conclusion of the Project, all savings from any Contingency shall go to the Owner.

6.4.5 Lump Sum General Conditions Amount

- .1 If the Owner exercises its option to enter into Phase 2, and Parties enter into the Phase 2 Amendment, the Parties shall establish a Lump Sum amount for the General Conditions Costs ("Lump Sum General Conditions Amount") that are set forth in Section 6.3.15 of the Agreement.
- .2 If the Owner exercises its option to enter into Phase 2 and Parties enter into the Phase 2 Amendment, the costs identified in Section 6.3.15 of the Agreement shall not be included in the Cost of the Work, and the Design Builder's sole compensation for the costs set forth in Section 6.3.15 shall be through the Lump Sum General Conditions Amount. Design Builder shall not be entitled to be compensated for the identified Lump Sum General Conditions Amount as part of the Cost of the Work.
- .3 The Owner shall have the right to examine the back up documentation establishing the Lump Sum General Conditions Costs, including but not limited to all estimates, proposals, contracts and other financial documentation on a transparent basis.
- 4. The Lump Sum General Conditions Amount shall only be modified if the Design-Builder is entitled to compensation for a delay pursuant to Section 8.2 of the General Conditions. Any modification to the Lump Sum General Conditions Amount shall be calculated as follows:
 - a. The Design Builder shall be entitled to receive a liquidated daily rate for extended General Conditions Costs ("Design-Builder's Delay Rate") for each day that the Contract Time is extended pursuant to Section 8.2 of the General Conditions.
 - i. The Design-Builder's Delay Rate shall be calculated by dividing the Lump Sum General Conditions Amount by the number of days in the Contract Time for Phase 2.
 - ii. Then, the Design-Builder's Delay Rate is multiplied by the number of days that the Contract Time is extended for Design-Builder's Delay, subject to a determination of entitlement pursuant to Article 8 of the General Conditions.
 - iii. The result from the Design-Builder's Delay Rate multiplied by the number of days is the Extended General Conditions Costs which shall be added to the Lump Sum General Conditions Amount by Change Order and paid to the Design Builder pursuant to the Schedule of Values, subject to a determination of entitlement pursuant to Article 8 of the General Conditions.

- b. The Design-Builder's Delay Rate shall not apply to delays occurring after Substantial Completion is achieved.
- c. The Parties agree that determining the Design Builder's damages for delay would be extremely difficult or impracticable to determine and that the Design-Builder's Delay Rate, as calculated in this Section 6.4.5.4, is a reasonable estimate of and reasonable Sum for such damages; therefore, the Design-Builder's Delay Rate shall be payable to the Design Builder as liquidated damages and not as a penalty.

6.5 Non-Reimbursable Costs.

- **6.5.1** The following shall not be deemed as costs of the Work:
 - .1 Compensation for Design-Builder's personnel stationed at Design-Builder's principal or branch offices, except as expressly provided herein.
 - .2 Overhead and general expenses, except as provided for in Section 6.3 hereof.
 - .3 The cost of Design-Builder's capital used in the performance of the Work.
 - .4 Costs that would cause the GMP, the Design Builder's Contingency, or any other NTE or Lump Sum Amount, as adjusted in accordance with the Contract Documents, to be exceeded.

6.6 **Project Phases**.

6.6.1 Phase 1 – Validation and GMP Development

- .1 **Scope of Work for Phase 1.** Phase 1 shall commence upon a written Notice to Proceed from the Owner and shall end on Phase 1 Completion Date as set forth below. The services to be provided by the Design Builder during Phase 1 are set forth in Exhibit C to the Agreement "Phases 1 and 2 Scope of Work" and Exhibit E to the Agreement "the Design-Builder's Phase 1 Scope of Services".
- .2 **Phase 1 Not to Exceed Amount.** Design Builder guarantees that during Phase 1, Design Builder's Compensation shall not exceed Phase 1 Not to Exceed Amount ("Phase 1 NTE") of <u>Three Hundred Thousand</u> Dollars (\$300,000.00). Design Builder agrees that it will be responsible for paying all costs of completing Phase 1 Work which exceed Phase 1 NTE and shall not seek reimbursement from the Owner for any costs that exceed Phase 1 NTE, as adjusted in accordance with the Contract Documents including by written Change Order.
- .3 Phase 1 Completion Date. Phase 1 Completion Date is _per 5.2.1_
- .4 **Design Builder's Phase 1 Compensation.** Design Builder's compensation for Work performed in Phase 1 shall consist of the following:
 - a. The Cost of the Work as set forth in Section 6.3 of the Agreement for Work performed in Phase 1. The Cost of the Work includes the following:
 - i. the Cost of the Work Contingency set forth in Section 6.4.4.1.a; and
 - ii. any Not to Exceed or Lump Sum Amount established as part of the Cost of the Work;
 - b. The Design-Builder's Lump Sum for Overhead and Profit;
 - c. Any Allowances established by the Parties; and
 - d. The Design-Builder's Contingency set forth in Section 6.4.4.1.b.

.5 **Phase 2 Proposal.** At the conclusion of Phase 1, the Design Builder will submit a Phase 2 Proposal pursuant to the requirements set forth in Exhibit C. Unless the Parties agree otherwise, the Phase 2 Proposal shall include the deliverables set forth in Exhibit C.

.6 **Owner's Option to Enter Into Phase 2**

- a. After submission of the Phase 2 Proposal, Design Builder and Owner shall meet to discuss and review the Phase 2 Proposal. The Owner shall make its best efforts to provide such comments within thirty (30) days of the Owner's receipt of the Phase 2 Proposal, unless the Owner provides notification that it requires additional time for review. If Owner has any comments regarding the Phase 2 Proposal or finds any inconsistencies or inaccuracies in the information presented, it shall give written notice to Design Builder of such comments or findings in a reasonably prompt manner. If appropriate, Design Builder shall, upon receipt of Owner's notice, make appropriate adjustments to the Phase 2 Proposal. To assist in the Owner's review of the Phase 2 Proposal, the Design Builder shall, upon the Owner's Request, provide all information, including but not limited to all data, reports, cost analysis, pricing, designs and specifications on which the Design Builder relied or used as a basis for the Phase 2 Proposal within thirty (30) days of receipt of the revised Phase 2 Proposal.
- b. The Owner, at its sole discretion, may exercise its option to enter into Phase 2 of the Agreement.
 - i. If the Owner accepts the Phase 2 Proposal, the parties shall enter into the Phase 2 Amendment. The total compensation paid to Design Builder for this Project shall not exceed the GMP, as amended pursuant to this Contract.
 - ii. The Owner may suggest modifications to the Phase 2 Proposal, whereupon, if such modifications are accepted in writing by Design Builder, the Phase 2 Proposal shall be deemed accepted and the Parties shall proceed in accordance with subsection i above.
- c. If Owner decides not to exercise its option to enter into Phase 2 and/or rejects the Phase 2 Proposal or fails to notify Design Builder in writing on or before the date specified in the Phase 2 Proposal that it has exercised its option to enter into Phase 2, the Phase 2 Proposal shall be deemed withdrawn and of no effect. In such event, Owner and Design Builder shall meet and confer as to how the Project will proceed, with Owner having the following options:
 - i. Owner may authorize Design Builder to continue to proceed with the Work on the basis of reimbursement as provided in Section 6.6.1.4 hereof; however, Design Builder may not exceed any NTE or Lump Sum that may be established between the Parties; or
 - ii. Owner may elect not to exercise its option to enter into Phase 2. In such case, the Design-Build Agreement shall be terminated, and Design-Builder shall be compensated for the amount incurred pursuant to Section 6.6.1.4 above, as supported by Design-Builder's Payment Applications and subject to Phase 1 Not to Exceed Amount. The compensation set forth herein shall be the Design-Builder's sole compensation for the Project if the Owner elects not to exercise its option to enter into Phase 2, and the Design Builder hereby agrees that it will not seek any other compensation, remedy or damages of any kind whatsoever if the Owner elects not to exercise its option to enter into Phase 2.
- d. The Design Builder shall not perform any Work after the submission of the Phase 2 Proposal unless the Owner exercises its option to enter into Phase 2 and has approved and signed the Phase 2 Proposal unless the Design Builder obtains the Owner's prior, written consent to perform such Work and only to the extent that such Work is expressly described in writing in such written consent.

e. If the Design Builder performs Work after the submission of the Phase 2 Proposal but before the Parties enter into the Phase 2 Amendment, Design Builder shall be compensated pursuant to Section 6.6.1.4 of the Agreement; however, in no case shall the Design Builder be entitled to be paid in excess of the Phase 2 NTE, as amended by the Parties.

6.6.2 Phase 2, Post GMP Period.

- .1 **Commencement and Scope of Work.** Phase 2 shall commence when the Owner exercises its option to enter into Phase 2 and both Parties sign the Phase 2 Amendment. The Phase 2 Amendment shall be in the form set forth in Exhibit G, unless the parties agree otherwise. Phase 2 is the final phase of the Contract. The scope of Work for Phase 2 will be developed during Phase 2 and set forth in the Phase 2 Amendment, but it will, at a minimum, include the services set forth in Exhibit C, including but not limited to the following:
 - a. Completion of the design services and the development of Construction Documents for the Project,
 - b. Performance and completion of construction Work, start-up, testing and commissioning and closeout of the Project in accordance with the requirements of the Contract Documents; and
 - c. Any ongoing contractual obligations after Final Completion, such as guarantees, warranty services, and/or obligations to provide insurance and indemnity to the Owner.
- .2 **Guaranteed Maximum Price.** The GMP has been established in this Agreement and shall not be changed except through the Phase 2 Amendment or a written Change Order. Design Builder agrees that it will be responsible for paying all costs of completing the Phase 2 Work which exceed the GMP, as adjusted in accordance with the Contract Documents. Execution of the Phase 2 Amendment constitutes Design Builder's representation and agreement to the following:
 - a. The Project is adequately defined, that the Basis of Design Documents are sufficiently defined to provide an accurate GMP;
 - b. The Project is sufficiently clear and understandable for the Design Builder to perform the Work in accordance with the Contract Documents for an amount that will not exceed the GMP and within the Project Schedule; and
 - c. If the Work cannot be completed for the agreed GMP, any additional costs shall be the responsibility of the Design Builder, and Design Builder hereby assumes liability for such costs without reimbursement by the Owner.
- .3 **Project Schedule.** The Substantial and Final Completion Dates will be set forth in the Phase 2 Amendment. By entering into the Phase 2 Amendment, the Design-Builder makes the following representations:
 - a. The Project Schedule is sufficient time to complete the Project in accordance with the Phase 2 Amendment and the Contract Documents.
 - b. If the Design-Builder fails to achieve Substantial Completion by the date set forth in the Phase 2 Amendment, the Design-Builder will pay liquidated damages in the amount set forth in Section 5.4 of the Agreement and the Phase 2 Amendment as agreed compensation to the Owner for the cost of delay and not as a penalty.
- .4 Design Builder's Compensation. Design Builder shall be compensated for Phase 2 for the following costs up to the established GMP. At the Owner's option, the Contract Price may be converted into a Lump Sum Amount, in which case, Design-Builder shall be compensated pursuant to Section 6.4.3. Any costs incurred in excess of the GMP or Contract Price shall be the responsibility of the Design Builder.
 - a. The Cost of the Work as set forth in Section 6.3 of the Agreement for Phase 2 Work, excluding the costs identified in Section 6.3.15 of the Agreement as Design-Builder's

through the Lump Sum General Conditions Amount. The Cost of the Work also includes the following:

- i. the Cost of the Work Contingency pursuant to Section 6.4.4.1.a; and
- ii. any Not to Exceed Amount established as part of the Cost of the Work;
- b. The Design-Builder's Phase 2 Lump Sum for Overhead and Profit, calculated pursuant to Section 6.2 of the Agreement;
- c. The Lump Sum General Conditions Amount allocated to Phase 2 Work, which shall be calculated pursuant to Section 6.4.5 of the Agreement;
- d. Any additional Lump Sum Amounts established by the Parties in the Phase 2 Amendment;
- e. Any Allowances established by the Parties in the Phase 2 Amendment;
- f. The Design-Builder's Contingency pursuant to section 6.4.4.1.b of the Agreement; and
- f. Any Incentive Payments established by the Parties in the Phase 2 Amendment.

6.6.3 Savings and Incentives.

The parties may establish incentive payments to the Design-Builder during Phase 1. If the parties establish incentive payments during Phase 1, the Design-Builder shall provide the agreed upon incentive plan as part of the Phase 2 Proposal, and the incentive plan shall be incorporated into the Contract Documents in the Phase 2 Amendment.

Article 7 Procedure for Payment

7.1 **Progress Payments.**

7.1.1 Design-Builder shall submit to Owner on the twenty fifth (25th) day of each month, beginning with the first month after the Date of Commencement, Design-Builder's Application for Payment in accordance with Article 6 of the General Conditions of Contract.

7.1.2 Owner shall make payment within thirty (30) days after Owner's receipt of each properly submitted and accurate Application for Payment in accordance with Article 6 of the General Conditions of Contract, but in each case less the total of payments previously made, and less amounts properly withheld under Sections 6.3 and 6.4 of the General Conditions of Contract.

7.1.3 The amount of Design-Builder's Lump Sum for Overhead and Profit to be included in Design-Builder's monthly Application for Payment and paid by Owner shall be proportional to the percentage of the Work completed, less payments previously made on account of Design-Builder's Lump Sum for Overhead and Profit.

7.2 Retainage on Progress Payments.

7.2.1 The Owner will withhold retainage in the amount of five percent (5%) of the Contract Price, and Owner shall release such retainage pursuant to state law. Interest will be paid on retainage pursuant to AS 36.90.250.

7.3 Final Payment. Design-Builder shall submit its Final Application for Payment to Owner in accordance with Section 6.8 of the General Conditions of Contract. Owner shall make payment on Design-Builder's properly submitted and accurate Final Application for Payment pursuant and subject to all applicable laws and regulations, provided that Design-Builder has satisfied the requirements for final payment set forth in Section 6.8.1 of the General Conditions of Contract.

7.4 Interest. Payments due and unpaid by Owner to Design-Builder, whether progress payments or final payment, shall bear interest commencing five (5) days after payment is due at the rate of ten and one-half_percent (10.5%) per year until paid.

Record Keeping and Finance Controls. Design-Builder acknowledges that this Agreement is to 7.5 be administered on an "open book" and transparent arrangement relative to all costs on the Project. Design-Builder shall keep full and detailed accounts and exercise such controls as may be necessary for proper financial management, using accounting and control systems in accordance with generally accepted accounting principles and as may be provided in the Contract Documents. During the performance of the Work and for a period of six (6) years after Final Payment, Owner, Owner's accountants, and any State or Federal agency with jurisdiction shall be afforded access to, and the right to audit from time-to-time, upon reasonable notice, Design-Builder's records, books, correspondence, receipts, subcontracts, purchase orders, vouchers, memoranda and other data relating to the Work, all of which Design-Builder shall preserve for a period of six (6) years after Final Payment. Such inspection shall take place at Design-Builder's offices during normal business hours unless another location and time is agreed to by the parties. Any multipliers or markups agreed to by the Owner and Design-Builder as part of this Agreement are only subject to audit to confirm that such multiplier or markup has been charged in accordance with this Agreement, with the composition of such multiplier or markup not being subject to audit. Notwithstanding the foregoing, prior to agreeing to a multiplier or markup, the Owner shall have the right to review the underlying costs of any multiplier or mark up. The audit may be performed by employees of Owner or a representative of Owner. Design-Builder, and its Subcontractors, shall provide adequate facilities acceptable to Owner, for the audit during normal business hours. Design-Builder, and all Subcontractors, shall make a good faith effort to cooperate with Owner's auditors. All records shall be maintained for a period of six (6) years after final payment under this Contract.

Article 8 Termination for Convenience

8.1 Upon ten (10) days' written notice to Design-Builder or if the Owner decides to not exercise its option to enter into Phase 2, Owner may, for its convenience and without cause, elect to terminate all or a portion of this Agreement. In such event, Owner shall pay Design-Builder for the following:

8.1.1 All Work executed and for proven loss, cost or expense in connection with the Work;

8.1.2 The reasonable costs and expenses attributable to such termination, including demobilization costs and amounts due in settlement of terminated contracts with Subcontractors and Design Consultants; and

8.1.3 The fair and reasonable sums for overhead and profit on the sum of items 8.1.1 and 8.1.2 above based on Design-Builder's Lump Sum for Overhead and Profit.

8.2 If Owner terminates this Agreement pursuant to Section 8.1 above and proceeds to design and construct the Project through its employees, agents or third parties, Owner's rights to use the Work Product shall be as set forth in Section 6.3 hereof. Such rights may not be transferred or assigned to others without Design-Builder's express written consent and such third parties' agreement to the terms of Article 4.

<u>Article 9</u> Representatives of the Parties

9.1 Owner's Representatives.

9.1.1 Owner designates the individual listed below as its Senior Representative ("Owner Senior Representative"), which individual has the authority and responsibility for avoiding and resolving disputes under Section 10.2.3 of the General Conditions of Contract: (*Identify individual's name, title, address and telephone numbers*)

9.1.2 Owner designates the individual listed below as its Owner's Representative, which individual has the authority and responsibility set forth in Section 3.4 of the General Conditions of Contract: (*Identify individual's name, title, address and telephone numbers*)

9.2 Design-Builder's Representatives.

9.2.1 Design-Builder designates the individual listed below as its Senior Representative ("Design-Builder's Senior Representative"), which individual has the authority and responsibility for avoiding and resolving disputes under Section 10.2.3 of the General Conditions of Contract: (*Identify individual's name, title, address and telephone numbers*)

9.2.2 Design-Builder designates the individual listed below as its Design-Builder's Representative, which individual has the authority and responsibility set forth in Section 2.1.1 of the General Conditions of Contract: (*Identify individual's name, title, address and telephone numbers*)

Article 10 Bonds and Insurance

10.1 Insurance. Design-Builder and Owner shall procure the insurance coverages set forth in the Insurance Exhibit attached hereto and in accordance with Article 5 of the General Conditions of Contract.

10.2 Bonds and Other Performance Security. Upon execution of this Agreement, Design-Builder shall provide a performance and a labor and material bond, pursuant to AS 36.25.010, equal to one hundred percent (100%) of Phase 1 NTE in the form set forth as Exhibit B. Upon execution of the Phase 2 Amendment, Design-Builder shall provide a performance and labor and material bond, pursuant to AS 36.25.010, equal to one hundred percent (100%) of the GMP set forth in the Phase 2 Amendment in the form set forth as Exhibit B.

Article 11 Other Provisions

11.1 Other provisions, if any, are as follows: (*Insert any additional provisions*)

11.2 Wages.

11.2.1 Design-Builder shall pay all mechanics, laborers, or field surveyors employed on the Project a minimum of the prevailing wage as determined by the Alaska Department of Labor and published in the Department of Labor Pamphlet titled ."Laborer's & Mechanics' Minimum Rates of Pay" that was in effect within ten (10) days of the date bids are submitted, Pamphlet 400, Issue 44, issued April 2022.

11.2.2 Design-Builder and all subcontractors of the Design-Builder shall pay all employees unconditionally and not less than once a week. Wages may not be less than those stated in the advertised specifications, regardless of the contractual relationship between the Design-Builder or subcontractors and laborers, mechanics, or field surveyors. The scale of wages to be paid shall be posted by the Design-Builder in a prominent and easily accessible place at the site of the work.

11.2.3 Owner shall withhold so much of the accrued payments as is necessary to pay to laborers, mechanics, or field surveyors employed by the Design-Builder or subcontractors the difference between:

- (A) the rates of wages required by the contract to be paid laborers, mechanics, or field surveyors on the work; and
- (B) the rates of wages in fact received by laborers, mechanics, or field surveyors.

11.3 Business Registration Requirement.

11.3.1 Design-Builder represents and warrants that it and all of its subconsultants, subcontractors and suppliers are properly licensed to perform the work for which they are

contracted and have all applicable business licenses, including but not limited to any licenses or registrations required by the State of Alaska.

11.4 Contractor's Registration Requirement.

11.4.1 Design-Builder represents and warrants that it and all of its subconsultants, subcontractors and suppliers performing construction work are properly licensed pursuant to state law.

11.5 Federal Requirements

11.5.1 Design-Builder shall comply with the Federal Requirements set forth in Exhibit H.

In executing this Agreement, Owner and Design-Builder each individually represents that it has the necessary financial resources to fulfill its obligations under this Agreement, and each has the necessary corporate approvals to execute this Agreement, and perform the services described herein.

OWNER:

DESIGN-BUILDER:

Haines Borough	Turnagain Marine Construction
(Signature)	(Signature)
(Printed Name)	(Printed Name)
(Title)	(Title)
Date:	Date:

Caution: An original DBIA document has this caution printed in blue. This is a printable copy and an original assures that changes will not be obscured as may occur when documents are reproduced.



GENERAL CONDITIONS OF PROGRESSIVE DESIGN-BUILD CONTRACT BETWEEN OWNER AND DESIGN-BUILDER

Note: This document contains differences from the DBIA Agreement form 535. Owner will provide a copy red-lined from the DBIA Agreement form upon written request.

Document No. 535 (modified) Second Edition, 2010 © Design-Build Institute of America Washington, DC

Article 1

General

1.1. Mutual Obligations

1.1.1 *Owner and Design-Builder* commit at all times to cooperate fully with each other and proceed on the basis of trust and good faith, to permit each party to realize the benefits afforded under the Contract Documents.

1.1.2 Integrated Delivery: The Parties wish to fully embrace the principles of collaboration and integrated delivery in the performance of the Work of this Project. Integrated delivery emphasizes a cooperative approach to problem solving involving all key parties to the Project: the Owner, Design-Builder, Designer and principal Subcontractors (electrical, mechanical and others as the Design-Builder and the Owner jointly agree are appropriate). Toward that end, the Parties agree to employ the following techniques to maximize efficiency and minimize waste:

- .1 Create a culture of open and honest communication throughout the course of the Project;
- .2 Resolve disputes at the lowest possible level;
- .3 Integrate the design and construction team (including key specialty contractors and trade partners) as early as possible into the design process;
- .4 Utilize lean construction methods efficiently and effectively;
- .5 Establish a collaborative environment where all parties have the opportunity to contribute their best efforts for the benefit of the Project as a whole rather than to the benefit of individual parties; and
- .6 Establish business terms that allow for equitable shared risk and reward for the parties who are members of the Design-Build Team.

1.2. Basic Definitions

1.2.1. Agreement refers to the executed contract between Owner and Design-Builder under a ;modified DBIA Document No. 530, Progressive Design-Build Agreement Between Owner and Design-Builder – with Cost Plus Fee and a Guaranteed Maximum Price (2010 Edition), as amended.

1.2.2. Allowance Item is a scope of work for a designated portion of the Project that the parties agree to manage pursuant to Section 6.4.1 of the Agreement.

1.2.3. Basis of Design Documents are those documents set forth in the Phase 2 Amendment that establish the Scope of Work for Phase 2 of the Project.

1.2.4. *Commercial Terms* are any terms that establish the Contract Price or Design-Builder's Compensation, including but not limited to the GMP, any Not to Exceed amount, any Lump Sum, any Allowance, or the Design-Builder's Contingency. The term "Commercial Terms" also includes any terms that establish the Contract Time, including but not limited to the Project Schedule, Substantial Completion, and Final Completion.

1.2.5. Construction Documents are the documents, consisting of Drawings and Specifications, to be prepared or assembled by the Design-Builder consistent with the Basis of Design Documents unless a deviation from the Basis of Design Documents (as applicable) is specifically set forth in a Change Order executed by both the Owner and Design-Builder, as part of the design review process contemplated by Section 2.4 of these General Conditions of Contract.

1.2.6. *Day* or *Days* shall mean calendar days unless otherwise specifically noted in the Contract Documents.

1.2.7. *Design-Build Team* is comprised of the Design-Builder, the Design Consultant, and key Subcontractors identified by the Design-Builder.

1.2.8. *Design-Builder's Lump Sum for Overhead and Profit* is the amount established pursuant to in Section 6.2 of the Agreement.

1.2.9. *Design-Builder's Delay Rate* means the daily delay rate set forth in Section 6.4.5.4 of the Agreement if the Design-Builder is entitled to delay pursuant to Section 8.2 of the General Conditions.

1.2.10. *Design Consultant* is a qualified, design professional licensed in the State of Alaska who is not an employee of Design-Builder, but is retained by Design-Builder, or employed or retained by anyone under contract with Design-Builder, to furnish design services required under the Contract Documents. A Design Sub-Consultant is a qualified, licensed design professional who is not an employee of the Design Consultant but is retained by the Design Consultant or employed or retained by anyone under contract to Design Consultant, to furnish design services required under the Contract Documents.

1.2.11. *Design Log* is a log of Reliable Design Decisions agreed upon by the parties. The Design Log supplements the Owner's Program and the Basis of Design Documents, as applicable.

1.2.12. *Design Submission* means any and all documents, shop drawings, electronic information, including computer programs and computer generated materials, data, plans, drawings, sketches, illustrations, specifications, descriptions, models and other information developed, prepared, furnished, delivered or required to be delivered by, or for, the Design-Builder: (1) to the Owner under the Contract Documents; or (2) developed or prepared by or for the Design-Builder specifically to discharge its duties under the Contract Documents.

1.2.13. *Final Basis of Design Documents* are the documents agreed upon in the Phase 2 Amendment by the Owner and Design-Builder at the conclusion of the Phase 1 that comprise the performance and other requirements of the Project.

1.2.14. *Final Completion* is the date on which all Work is complete in accordance with the Contract Documents, including but not limited to, any items identified in the punch list prepared under Section 6.7.1 and the submission of all documents set forth in Section 6.8.1.

1.2.15. *Force Majeure Events* are those events that are beyond the control of both Design-Builder and Owner, including the events of war, floods, labor disputes, earthquakes, epidemics, adverse weather conditions not reasonably anticipated, and other acts of God.

1.2.16. *General Conditions Costs* are the costs set forth in Section 6.3.15 of the Agreement that are included in the Lump Sum General Conditions Amount pursuant to Section 6.4.5 of the General Conditions if the parties enter into Phase 2 of the Contract.

1.2.17. General Conditions of Contract refer to this General Conditions of Progressive Design-Build Contract Between Owner and Design-Builder.

1.2.18. *Hazardous Conditions* are any materials, wastes, substances and chemicals deemed to be hazardous under applicable Legal Requirements, or the handling, storage, remediation, or disposal of which are regulated by applicable Legal Requirements.

1.2.19. *Legal Requirements* are all applicable federal, state and local laws, codes, ordinances, rules, regulations, orders and decrees of any government or quasi-government entity having jurisdiction over the Project or Site, the practices involved in the Project or Site, or any Work.

1.2.20. *Lump Sum Fee* is the lump sum amount established pursuant to Section 6.2 of the Agreement, provided the parties enter into the Phase 2 Amendment.

1.2.21. Original GMP or Original Guaranteed Maximum Price means the Guaranteed Maximum Price that is set forth in the original Phase 2 Amendment entered into by the parties.

1.2.22. *Phase 2 Amendment* is the amendment to the Agreement entered into by the parties at the conclusion of Phase 1 that establishes the Basis of Design Documents, the GMP, the Project Schedule and other terms agreed to by the parties.

1.2.23. *Phase 2 Proposal* means that proposal developed by Design-Builder in accordance with Section 6.6 of the Agreement and Exhibit C.

1.2.24. *Project Schedule or Schedule* is the schedule provided by the Design-Builder pursuant to Section 2.1.3 of the General Conditions.

1.2.25. *Owner's Program* are developed by or for Owner to describe Owner's program requirements and objectives for the Project, including use, space, price, time, site and expandability requirements, as well as submittal requirements and other requirements governing Design-Builder's performance of the Work. Owner's Program may include conceptual documents, design criteria, design performance specifications, design specifications, and other Project-specific technical materials and requirements.

1.2.26. *Reliable Design Decision* is a decision, development, or election that refines the Basis of Design Documents, that is approved by the Owner and that is set forth in the Design Log. A Reliable Design Decision cannot change the Owner's Program or the Basis of Design Documents but shall instead constitute a further development or refinement of the design for the Project with which all subsequent Design Submissions, design submissions and Construction Documents shall be consistent.

1.2.27. Site is the land or premises on which the Project is located.

1.2.28. *Subcontractor* is any person or entity retained by Design-Builder as an independent contractor to perform a portion of the Work and shall include Design Consultants, materialmen, and suppliers.

1.2.29. *Sub-Subcontractor* is any person or entity retained by a Subcontractor as an independent contractor to perform any portion of a Subcontractor's Work and shall include but not be limited to, design consultants, design sub-consultants, design-build subcontractors, materialmen, and suppliers.

1.2.30. Substantial Completion or Substantially Complete means the date on which the Work, or an Interim Milestone Date, is sufficiently complete in accordance with the Contract Documents so that Owner can occupy and use the Project or a portion thereof for its intended purposes without compromising the building operation (including materially increasing operating expenses) or the user's ability to reasonably use all parts of the Project.

1.2.31. *Trend* is an issue identified in the Trend Log.

1.2.32. *Trend Log* is a log of issues that have been identified by the Design-Builder or the Owner during the design process that may cause a change to the Owner's Program or the Basis of Design Documents, as applicable and/or any Commercial Term and is further described in Section 2.4.1.6 of the General Conditions.

1.2.33. *Work* shall mean the services, design and construction to be completed by the Design-Builder under the terms of this Contract. Work specifically includes the furnishing of all services, labor, materials, equipment, and all incidentals necessary to the successful completion of the services, design and construction, whether expressly required by or reasonably inferable from the Contract Documents, whether they are temporary or permanent, and whether they are incorporated into the finished Work or not. Work also includes all other obligations imposed on the Design-Builder by the Contract. The Work is sometimes generally referred to as the "Project."

Article 2

Design-Builder's Services and Responsibilities

2.1 General Services.

2.1.1 Design-Builder's Representative shall be reasonably available to Owner and shall have the necessary expertise and experience required to supervise the Work. Design-Builder's Representative shall communicate regularly with Owner and shall be vested with the authority to act on behalf of Design-Builder. Design-Builder's Representative may be replaced only with the mutual agreement of Owner and Design-Builder.

2.1.2 Design-Builder shall provide Owner with a monthly status report detailing the progress of

the Work, including (i) whether the Work is proceeding according to schedule, (ii) whether discrepancies, conflicts, or ambiguities exist in the Contract Documents that require resolution, (iii) whether health and safety issues exist in connection with the Work; (iv) status of the contingency account to the extent provided for in the Agreement; and (v) other items that require resolution so as not to jeopardize Design-Builder's ability to complete the Work for the Contract Price and within the Contract Time(s). Status reports shall be submitted with the Design-Builder's draft Payment Applications as a pre-requisite to payment.

2.1.3 Design-Builder shall prepare and submit, pursuant to Exhibit C, schedules for the execution of the Work for Owner's review and response. The Project Schedule shall indicate the dates for the start and completion of the various stages of Work, including the dates when Owner information and approvals are required to enable Design-Builder to achieve the Contract Time(s). The schedule shall be revised as required by conditions and progress of the Work and as set forth in Exhibit C, but such revisions shall not relieve Design-Builder of its obligations to complete the Work within the Contract Time(s), as such dates may be adjusted in accordance with the Contract Documents. Owner's review of, and response to, the schedule shall not be construed as relieving Design-Builder of its obligations, sequences and techniques for executing the Work.

2.1.4 The parties will meet pursuant to the requirements in Exhibit C, and in any event, within seven (7) days after execution of the Agreement to discuss issues affecting the administration of the Work and to implement the necessary procedures, including those relating to submittals and payment, to facilitate the ability of the parties to perform their obligations under the Contract Documents.

2.1.5 The Design-Build Team, which at a minimum shall consist of the Design-Builder's Representative and a representative from the lead designer and lead constructor, shall meet with the Owner at least on a weekly basis and shall provide to the Owner a written update regarding the status of the Project, including but not limited to the following information: any updates to the Project Schedule, status of any changes or potential changes to the Initial and/or Final Basis of Design Documents or the Project Schedule, progress of the design, and any issues that may have a material effect on the Project. The Design-Build Team shall issue meeting minutes within three days of meeting.

2.2 Design Professional Services.

2.2.1 Design-Builder shall, consistent with applicable state licensing laws, provide through qualified, licensed design professionals employed by Design-Builder, or procured from qualified, independent licensed Design Consultants, the necessary design services, including architectural, engineering and other design professional services, for the preparation of the required drawings, specifications and other design submittals to permit Design-Builder to complete the Work consistent with the Contract Documents. Nothing in the Contract Documents is intended or deemed to create any legal or contractual relationship between Owner and any Design Consultant.

2.2.2 Prior to the date that Design Consultants and/or Design Subconsultants perform Work on the Project, Design-Builder shall provide to Owner a list of all Design Consultants and Design Sub-Consultants who will perform material portions of the Work. "Material portions of the Work" shall, at a minimum, include the civil, landscape, architectural, structural, mechanical, electrical (including low voltage) and plumbing design. Owner may reasonably object to Design-Builder's selection of any Design Consultant or Design Subconsultant, provided that the Contract Price and/or Contract Time(s) shall be adjusted to the extent that Owner's decisions impacts Design-Builder's cost and/or time of performance. Design-Builder shall not substitute a listed Design Consultant or Sub-Consultant without obtaining Owner's prior written consent, such consent shall not be unreasonably withheld. The Contract Documents shall not be construed to create a contractual relationship of any kind between Owner and any Design Consultant or Sub-Consultant of any tier. Selection of Design Consultants and Design Sub-Consultants that have not been identified in the Design-Builder's Proposal shall be in accordance with Section 2.8 of the General Conditions.

2.3 Standard of Care.

2.3.1 The standard of care for all professional services performed to execute the Work shall be

the care and skill ordinarily used by members of the applicable profession practicing under similar conditions at the same time within the State of Alaska. The Design-Builder shall also perform the design and construction so that the Work meets or exceeds the performance requirements set forth in the Owner's Program.

2.4 Design Development Services.

2.4.1 Design-Builder and Owner shall, consistent with any applicable provision of the Contract Documents, agree upon any interim and final Design Submissions that Owner may wish to review.

.1 Design Submissions shall be consistent with the Owner Project Requirements as well as the Basis of Design Documents, as the Basis of Design Documents may have been changed or supplemented through the design process set forth in this Section 2.4, including but not limited to changes recorded in the Design Log and through Change Orders. By submitting Design Submissions, the Design Builder represents to the Owner that the Work depicted and otherwise shown, contained or reflected in Design Submissions can be constructed in compliance with the then current Commercial Terms. Notwithstanding the above, Design Builder may propose Design Submissions that may alter either the Basis of Design Documents, or the Commercial Terms; however, Design Builder must provide notice thereof in accordance with Article 10 of the General Conditions and obtain a Change Order before such proposed Design Submissions are incorporated into the Construction Documents. Alternatively, if the Owner agrees in writing, the proposed Design Submission may be included in the Trend Log pursuant to 2.4.1.7 of the General Conditions.

.2 Unless the parties agree in writing otherwise, the Design-Builder shall provide the Milestone Design Submissions set forth in Contract Documents. On or about the time of the scheduled submission of the Milestone Design Submissions set forth in the Contract Documents, Design Builder and Owner shall meet and confer about the Milestone Design Submissions, with Design Builder identifying during such meetings, among other things, the evolution of the design and any changes to the Owner's Program, the Basis of Design Documents, or, if applicable, previously submitted Design Submissions.

.3 The Owner shall review and comment on Design Submissions, providing any comments and/or concerns about the Design Submissions. The Owner shall provide all comments on the Design Submissions within the time provided by the Contract Documents. The Design Builder shall revise the Design Submissions (and any other deliverables) in response to the Owner's comments and incorporate said responses into the next submission of Design Submissions.

.4 If incorporation of the Owner's comments result in a design that is inconsistent with or otherwise give rise to a change in the Owner's Program, the Basis of Design Documents, or the applicable Commercial Terms, the Design Builder shall provide notice thereof in accordance with Articles 9 and 10 of the General Conditions. Changes to the Basis of Design Documents or the Commercial Terms, including those that are deemed minor changes, shall be processed in accordance with Article 9 of the General Conditions. Alternatively, if the Owner agrees in writing, the proposed Design Submission may be included in the Trend Log pursuant to Section 2.4.1.7 of the General Conditions.

.5 The Design Builder shall provide an updated cost model for the Project periodically as set forth Exhibit C. The cost model will be based on a detailed labor and material cost estimate for the GMP and the other Commercial Terms as required in Contract Documents. The cost model will be supplemented pursuant to Contract Documents.

.6 <u>Design Log</u>. A Design Log, including a full listing of Reliable Design Decisions and all changes to the Basis of Design Documents, will be maintained by the Design Builder and provided to the Owner for review.

- a. The Design Log shall be updated after every Design Review Meeting, and in any case, on a weekly basis.
- b. The purpose of the Design Log is to record design decisions that are consistent with the Owner's Program, the Commercial Terms, and the Basis of Design

Documents, as applicable. Both parties must agree to include a Reliable Design Decision in the Design Log. If a Reliable Design Decision will cause a change in the Basis of Design Decisions, or any of the other Commercial Terms, such changes must be processed pursuant to Articles 9 and 10 of the General Conditions.

c. Once a Reliable Design Decision is incorporated into the Design Log, it shall be binding on the Design Builder as if set forth in the Owner's Program and/or the Basis of Design Documents, as applicable.

.7 <u>Trend Log</u>. If the Design-Builder does not know the extent to which a Design Submissions or a Design Submission will alter a Commercial Term, the Design-Builder shall request in writing for the Owner to agree to identify the Trend in the Trend Log.

a. The request to include a Trend in the Trend Log must include the following information:

- i. Identification of the portion of the Design Submissions or Design Submission for which the costs are uncertain and may cause any Commercial Term to be exceeded;
- ii. The estimated change in the applicable Commercial Term; and
- iii. Potential impacts or changes to the Owner's Program or Basis of Design Documents as a result of the Trend.

b. The Design-Builder must obtain the Owner's consent to include the Trend in the Trend Log. The Design-Builder will track the Trend on the Trend Log, and the Trend Log shall be updated with the most recent information on a weekly basis.

c. The Parties will work collaboratively to resolve Trends in the Trend Log as quickly as possible. When a Trend in the Log is resolved, and the resolution changes the Basis of Design Documents and/or any other Commercial Term, the resolution shall be memorialized in a Change Order.

2.4.2 Design-Builder shall submit to Owner Construction Documents setting forth in detail drawings and specifications describing the requirements for construction of the Work. The Construction Documents shall be consistent with the latest set of interim design submissions, as such submissions may have been modified in a design review meeting and recorded as set forth above. The parties shall have a design review meeting to discuss, and Owner shall review and approve, the Construction Documents in accordance with the procedures set forth in Section 2.4.1 above. Design-Builder shall proceed with construction in accordance with the approved Construction Documents and shall submit one set of approved Construction Documents to Owner prior to commencement of construction.

.1 The Construction Documents shall provide information customarily necessary in documents for projects of similar size, complexity, and quality, including its phasing and subcontracting mode. The Construction Documents shall include all information required by the building trades to complete the construction of the Project, other than such details customarily developed by others during construction. To the extent not prohibited by the Contract Documents or Applicable Code Requirements, and subject to written approval by the Owner, Design Builder may prepare Construction Documents for approved Construction Packages for a portion of the Work to permit construction to proceed on that portion of the Work prior to completion of the Construction Documents for the entire Work.

.2 It is acknowledged by the parties hereto that inherent in a design build project, the production and review of Construction Documents may be a continuing process with portions thereof completed at different times. The Design Builder will limit the Construction Packages for Owner's review to a reasonable number, unless approved in writing by the Owner. Contract Schedule shall indicate the times for the Owner to review the completion of each such portion of the Construction Documents and a reasonable time for review of same.

2.4.3 Owner's review and approval of Design Submissions, meeting minutes, the Design Log, the Trend Log, and the Construction Documents is for the purpose of mutually establishing a conformed set of Contract Documents compatible with the requirements of the Work. Neither Owner's review nor approval of any Design Submissions, meeting minutes, the Design Log, the Trend Log and Construction Documents shall be deemed to transfer any design liability from Design-Builder to Owner, and Owner's review shall not be deemed an approval or waiver by the Owner of any deviation from, or of the Design Builder's failure to comply with, any provision or requirement of the Contract Documents, unless such deviation or failure has been expressly identified as such in writing in the documents submitted by the Design Builder and approved by the Owner. Design-Builder shall provide Owner with sufficient time in the Project Schedule to review and approve the design submissions, such time period shall not be less than ten business days.

2.4.4 To the extent not prohibited by the Contract Documents or Legal Requirements and with the Owner's written permission, Design-Builder may prepare interim Design Submissions and Construction Documents for a portion of the Work to permit construction to proceed on that portion of the Work prior to completion of the Construction Documents for the entire Work.

2.5 Legal Requirements.

2.5.1 Design-Builder shall perform the Work in accordance with all Legal Requirements and shall provide all notices applicable to the Work as required by the Legal Requirements.

2.5.2 The Commercial Terms shall be adjusted to compensate Design-Builder for the effects of any changes in the Legal Requirements enacted after the date the parties agree upon the Commercial Term. Such effects may include, without limitation, revisions Design-Builder is required to make to the Construction Documents because of changes in Legal Requirements.

2.6 Government Approvals and Permits.

2.6.1 Except as identified in an Owner's Permit List attached as an exhibit to the Agreement, Design-Builder shall obtain and pay for all necessary permits, approvals, licenses, government charges and inspection fees required for the prosecution of the Work by any government or quasi-government entity having jurisdiction over the Project.

2.6.2 Design-Builder shall provide reasonable assistance to Owner in obtaining those permits, approvals and licenses that are Owner's responsibility.

2.7 Design-Builder's Construction Services.

2.7.1 Unless otherwise provided in the Contract Documents to be the responsibility of Owner or a separate contractor, Design-Builder shall provide through itself or Subcontractors the necessary supervision, labor, inspection, testing, start-up, material, equipment, machinery, temporary utilities and other temporary facilities to permit Design-Builder to complete construction of the Project consistent with the Contract Documents.

2.7.2 Design-Builder shall perform all construction activities efficiently and with the requisite expertise, skill and competence to satisfy the requirements of the Contract Documents. Design-Builder shall at all times exercise complete and exclusive control over the means, methods, sequences and techniques of construction.

2.8 Subcontracts

2.8.1 Design-Builder shall employ only Subcontractors who are duly licensed and qualified to perform the Work consistent with the Contract Documents. Prior to the date that Subcontractors perform Work on the Project, Design-Builder shall identify in writing to Owner all Subcontractors. To the extent that the Design-Builder has not selected a Subcontractor prior to performing the Work, Design-Builder shall provide Owner in writing a list of any subsequently added Subcontractors prior to their performing Work on the Project. Owner may reasonably object to Design-Builder's selection of any Subcontractor, provided that the Contract Price and/or Contract Time(s) shall be adjusted to the extent that Owner's decision impacts Design-Builder's cost and/or time of performance. Design-Builder may not substitute listed Subcontractors identified in the Design-Builder's Proposal or previously approved by Owner without Owner's prior consent; such consent shall not be unreasonably withheld. The Contract Documents shall not be construed to create a contractual

relationship of any kind between Owner and any Subcontractor of any tier.

2.8.2 Design-Builder shall submit a Subcontracting Procurement Procedure during Phase 1 as required in Exhibit C, subject to the approval of the Owner. After approval by the Owner, Design-Builder may only modify the Subcontracting Plan upon obtaining written approval from the Owner. Design-Builder may not award any Subcontract on the basis of a lump sum price without obtaining prior written permission from the Owner, such permission shall not be unreasonably withheld.

2.8.3 All subcontracted Work associated with the performance of the construction shall be awarded by Design-Builder in accordance with a Subcontractor Procurement Procedure established during Phase 1. Unless otherwise agreed in writing by the Parties, the best value selection process shall contain mutually acceptable evaluation Requirements for the proposal and selection process that is clear and consistent and, when applicable, includes both qualifications and price. The Subcontractor Procurement Procedure shall comply with the following requirements:

.1 Design-Builder shall identify the scope of subcontracted Work ("Subcontract Package") and shall identify at least three pre-qualified Subcontractors for each Subcontract Package. The Owner may reject any pre-qualified Subcontractor for good cause.

.2 Design-Builder shall select from the pre-qualified Subcontractors for the Subcontract Package, unless Design-Builder obtains prior, written approval from the Owner.

.3 If Design-Builder cannot reasonably identify three pre-qualified Subcontractors, then it shall inform the Owner in writing as to the reason for the inability to identify the prequalified Subcontractors, and Design-Builder shall not proceed with the selection of a Subcontractor without obtaining prior, written approval from the Owner.

.4 Design-Builder shall select Subcontractors on the basis of the Best Value for the Project. If the Best Value is not the lowest price, Design-Builder shall obtain written approval of the Subcontractor selection from the Owner, such approval shall not be unreasonably withheld.

2.8.4 Design-Builder must obtain prior, written approval from the Owner for the Design-Builder to self-perform construction Work.

.1 For each scope of Work for which Design-Builder proposes self-performance, Design-Builder must submit to the Owner a proposal that contains the following minimum information as well as any other information reasonably requested by the Owner:

- a. A detailed description of the scope of Work;
- b. A detailed explanation of the effect of the self-performed construction Work on the Project, including but not limited to cost savings, benefits to the Project, and risks to the Project; and
- c. An explanation of i) how the self-performed construction Work will be priced (i.e. Lump Sum, Not to Exceed, etc.), and ii) how the reasonableness of the costs for the self-performed construction Work will be verified.

.2 Design-Builder will provide the Owner with an estimate of the costs for all selfperformed construction Work on an open book basis. In calculating the costs for selfperformed construction Work, whether such costs are proposed on the basis of a Cost of the Work or a Lump Sum, the following shall apply:

- a. The costs for self-performed construction Work shall not include costs that are also included in the Lump Sum General Conditions Amount.
- b. Notwithstanding the above, Design-Builder may include in the costs for selfperformed construction Work additional general conditions costs that are directly associated with the self-performed construction Work that Design-Builder would not have incurred but for the self-performed construction Work.

2.8.5 Design-Builder assumes responsibility to Owner for the proper performance of the Work of Subcontractors and any acts and omissions in connection with such performance. Nothing in the Contract Documents is intended or deemed to create any legal or contractual relationship between Owner and any Subcontractor or Sub-Subcontractor, including but not limited to any third-party beneficiary rights.

2.8.6 Design-Builder shall coordinate the activities of all Subcontractors. If Owner performs other work on the Project or at the Site with separate contractors under Owner's control, Design-Builder agrees to reasonably cooperate and coordinate its activities with those of such separate contractors so that the Project can be completed in an orderly and coordinated manner without unreasonable disruption.

2.8.7 Design-Builder shall keep the Site reasonably free from debris, trash and construction wastes to permit Design-Builder to perform its construction services efficiently, safely and without interfering with the use of adjacent land areas. Upon Substantial Completion of the Work, or a portion of the Work, Design-Builder shall remove all debris, trash, construction wastes, materials, equipment, machinery and tools arising from the Work or applicable portions thereof to permit Owner to occupy the Project or a portion of the Project for its intended use.

2.9 Design-Builder's Responsibility for Project Safety.

2.9.1 Design-Builder recognizes the importance of performing the Work in a safe manner so as to prevent damage, injury or loss to (i) all individuals at the Site, whether working or visiting, (ii) the Work, including materials and equipment incorporated into the Work or stored on-Site or off-Site, and (iii) all other property at the Site or adjacent thereto. Design-Builder assumes responsibility for implementing and monitoring all safety precautions and programs related to the performance of the Work. Design-Builder shall, prior to commencing construction, designate a Safety Representative with the necessary qualifications and experience to supervise the implementation and monitoring of all safety precautions and programs related to the Work. Unless otherwise required by the Contract Documents, Design-Builder's Safety Representative shall be an individual stationed at the Site who may have responsibilities on the Project in addition to safety. The Safety Representative shall make routine daily inspections of the Site and shall hold weekly safety meetings with Design-Builder's personnel, Subcontractors and others as applicable.

2.9.2 Design-Builder and Subcontractors shall comply with all Legal Requirements relating to safety, as well as any Owner-specific safety requirements set forth in the Contract Documents, provided that such Owner-specific requirements do not violate any applicable Legal Requirement. Design-Builder will immediately report in writing any safety-related injury, loss, damage or accident arising from the Work to Owner's Representative and, to the extent mandated by Legal Requirements, to all government or quasi-government authorities having jurisdiction over safety-related matters involving the Project or the Work.

2.9.3 Design-Builder's responsibility for safety under this Section 2.9 is not intended in any way to relieve Subcontractors and Sub-Subcontractors of their own contractual and legal obligations and responsibility for (i) complying with all Legal Requirements, including those related to health and safety matters, and (ii) taking all necessary measures to implement and monitor all safety precautions and programs to guard against injuries, losses, damages or accidents resulting from their performance of the Work.

2.10 Design-Builder's Warranty.

2.10.1 Design-Builder warrants to Owner that the construction, including all materials and equipment furnished as part of the construction, shall be new unless otherwise specified in the Contract Documents, of good quality, in conformance with the Contract Documents and free of defects in materials and workmanship. Design-Builder's warranty obligation excludes defects caused by abuse, alterations, or failure to maintain the Work in a commercially reasonable manner. Nothing in this warranty is intended to limit any manufacturer's warranty which provides Owner with greater warranty rights than set forth in this Section 2.10 or the Contract Documents. Design-Builder will provide Owner with all manufacturers' warranties upon Substantial Completion.

2.11 Correction of Defective Work.

2.11.1 Design-Builder agrees to correct any Work that is found to not be in conformance with the Contract Documents, including but not limited to that part of the Work subject to Section 2.10 hereof, within a period of one year from the date of Substantial Completion of the Work or any portion of the Work, or within such longer period to the extent required by any specific warranty included in the Contract Documents.

2.11.2 Design-Builder shall, within seven (7) days of receipt of written notice from Owner that the Work is not in conformance with the Contract Documents, take meaningful steps to commence correction of such nonconforming Work, including the correction, removal or replacement of the nonconforming Work and any damage caused to other parts of the Work affected by the nonconforming Work. If Design-Builder fails to commence the necessary steps within such seven (7) day period, Owner, in addition to any other remedies provided under the Contract Documents, may provide Design-Builder with written notice that Owner will commence correction of such nonconforming Work with its own forces. If Owner does perform such corrective Work, Design-Builder shall be responsible for all reasonable costs incurred by Owner in performing such correction. If the nonconforming Work creates an emergency requiring an immediate response, the seven (7) day period identified herein shall be deemed inapplicable.

2.11.3 The one-year period referenced in Section 2.11.1 above applies only to Design-Builder's obligation to correct nonconforming Work and is not intended to constitute a period of limitations for any other rights or remedies Owner may have regarding Design-Builder's other obligations under the Contract Documents.

Article 3

Owner's Services and Responsibilities

3.1 Duty to Cooperate.

3.1.1 Owner shall, throughout the performance of the Work, cooperate with Design-Builder and perform its responsibilities, obligations and services in a timely manner to facilitate Design-Builder's timely and efficient performance of the Work and so as not to delay or interfere with Design-Builder's performance of its obligations under the Contract Documents.

3.1.2 Owner shall provide timely reviews and approvals of interim design submissions and Construction Documents consistent with the turnaround times set forth in Design-Builder's schedule.

3.1.3 Owner shall give Design-Builder timely notice of any Work that Owner notices to be defective or not in compliance with the Contract Documents.

3.2 Furnishing of Services and Information.

3.2.1 Owner has provided information in the Owner's Program ("Owner Provided Information"). The Owner Provided Information contains design or prescriptive specifications, and the Design-Builder shall be entitled to reasonably rely on the accuracy of the information represented in such design or prescriptive specifications and their compatibility with other information set forth in Owner Provided Information, including any performance specifications, but only for the purposes of developing the Design-Builder's Phase 1 Scope of Services (Exhibit E), the Phase 1 Not to Exceed Amount and the Design-Builder's Lump Sum for Overhead and Profit. Notwithstanding the above, Design-Builder is required to perform an independent evaluation of the Owner Provided Information during Phase 1 as set forth in Exhibit C to the Agreement and may not rely on the Owner Provided Information for the purposes of performing the Work. Provided Design-Builder complies with other requirements set forth in the Contract Documents regarding entitlement to adjustment of Commercial Terms, such as but not limited to those regarding notice of claims to the Owner and identification of differing site conditions, Design-Builder may be entitled to an adjustment in Phase 1 Scope of Services, Phase 1 Not to Exceed Amount and/or the Design-Builder's Lump Sum for Overhead and Profit, but only to the extent Design-Builder's cost and/or time of performance have been adversely impacted by such inaccurate design or prescriptive specifications in the Owner Provided Information that is discovered in Phase 1.

3.2.2 Owner is responsible for securing and executing all necessary agreements with adjacent land or property owners that are necessary to enable Design-Builder to perform the Work. Owner is further responsible for all costs, including attorneys' fees, incurred in securing these necessary agreements.

3.3 Financial Information.

3.3.1 If Design-Builder has a reasonable belief that Owner will not have sufficient funds to complete the Project, at Design-Builder's written request, Owner shall promptly furnish reasonable evidence satisfactory to Design-Builder that Owner has adequate funds available and committed to fulfill all of Owner's contractual obligations under the Contract Documents. If Owner fails to furnish such financial information in a timely manner, Design-Builder may stop Work under Section 11.3 hereof or exercise any other right permitted under the Contract Documents.

3.3.2 Design-Builder shall cooperate with the reasonable requirements of Owner's lenders or other financial sources. Notwithstanding the preceding sentence, after execution of the Agreement Design-Builder shall have no obligation to execute for Owner or Owner's lenders or other financial sources any documents or agreements that require Design-Builder to assume obligations or responsibilities greater than those existing obligations Design-Builder has under the Contract Documents.

3.4 Owner's Representative.

3.4.1 Owner's Representative shall be responsible for providing Owner-supplied information and approvals in a timely manner to permit Design-Builder to fulfill its obligations under the Contract Documents. Owner's Representative shall also provide Design-Builder with prompt notice if it observes any failure on the part of Design-Builder to fulfill its contractual obligations, including any errors, omissions or defects in the performance of the Work. Owner's Representative shall communicate regularly with Design-Builder and shall be vested with the authority to act on behalf of Owner.

3.5 Government Approvals and Permits.

3.5.1 Owner shall obtain and pay for all necessary permits, approvals, licenses, government charges and inspection fees set forth in Section 2.6.1.

3.5.2 Owner shall provide reasonable assistance to Design-Builder in obtaining those permits, approvals and licenses that are Design-Builder's responsibility.

3.6 Owner's Separate Contractors.

3.6.1 Owner is responsible for all work performed on the Project or at the Site by separate contractors under Owner's control. Owner shall contractually require its separate contractors to cooperate with and coordinate their activities so as not to interfere with, Design-Builder in order to enable Design-Builder to timely complete the Work consistent with the Contract Documents.

Article 4

Hazardous Conditions and Differing Site Conditions

4.1 Hazardous Conditions.

4.1.1 Unless otherwise expressly provided in the Contract Documents to be part of the Work, Design-Builder is not responsible for any Hazardous Conditions encountered at the Site that could have been reasonably discovered during the Phase 1. Unless working with such Hazardous Condition is part of the scope of the Work, upon encountering any Hazardous Conditions, Design-Builder will stop Work immediately in the affected area and duly notify Owner and, if required by Legal Requirements, all government or quasi-government entities with jurisdiction over the Project or Site.

4.1.2 Upon receiving notice of the presence of suspected Hazardous Conditions that are not set forth as part of the Work or that could not have been reasonably discovered during the Phase 1, Owner shall take the necessary measures required to ensure that the Hazardous Conditions are remediated or rendered harmless. Such necessary measures shall include Owner retaining qualified independent experts to (i) ascertain whether Hazardous Conditions have actually been encountered, and, if they have been encountered, (ii) prescribe the remedial measures that Owner must take either to remove the Hazardous Conditions or render the Hazardous Conditions harmless.

4.1.3 Design-Builder shall be obligated to resume Work at the affected area of the Project only after Owner's expert provides it with written certification that (i) the Hazardous Conditions have been removed or rendered harmless and (ii) all necessary approvals have been obtained from all government and quasi-government entities having jurisdiction over the Project or Site.

4.1.4 Unless expressly provided in the Contract Documents to be part of the Work, Design-Builder will be entitled, in accordance with these General Conditions of Contract, to an adjustment in its Contract Price and/or Contract Time(s) to the extent Design-Builder's cost and/or time of performance have been adversely impacted by the presence of Hazardous Conditions.

4.1.5 To the fullest extent permitted by law, Owner shall indemnify, defend and hold harmless Design-Builder, Design Consultants, Subcontractors, anyone employed directly or indirectly by any of them, and their officers, directors, employees and agents, from and against any and all claims, losses, damages, liabilities and expenses, including reasonable attorneys' fees and expenses, arising out of or resulting from the presence, removal or remediation of Hazardous Conditions at the Site pursuant to this Section.

4.1.6 Notwithstanding the preceding provisions of this Section 4.1, Owner is not responsible for Hazardous Conditions introduced to the Site by Design-Builder, Subcontractors or anyone for whose acts they may be liable. To the fullest extent permitted by law, Design-Builder shall indemnify, defend and hold harmless Owner and Owner's officers, directors, employees and agents from and against all claims, losses, damages, liabilities and expenses, including attorneys' fees and expenses, arising out of or resulting from those Hazardous Conditions introduced to the Site by Design-Builder, Subcontractors or anyone for whose acts they may be liable.

4.1.7 With respect to Hazardous Conditions that are part of the Work, Design-Builder agrees to comply with all applicable regulatory authorities, including but not limited to any statute, regulation or regulatory agency regarding such Hazardous Conditions. Design-Builder agrees to work cooperatively with Owner and regulatory agencies with jurisdiction over the Project to properly handle, dispose of, and/or remediate any Hazardous Conditions.

4.2 Differing Site Conditions.

4.2.1 Concealed or latent physical conditions or subsurface conditions at the Site that (i) materially differ from the conditions indicated in Exhibit D or (ii) are of an unusual nature, differing materially from the conditions ordinarily encountered and generally recognized as inherent in the Work are collectively referred to herein as "Differing Site Conditions." If Design-Builder encounters a Differing Site Condition, Design-Builder will be entitled to an adjustment in the applicable Commercial Term to the extent Design-Builder's cost and/or time of performance are adversely impacted by the Differing Site Condition.

4.2.2 Pursuant to Exhibit C, Design-Builder is required to submit a Differing Site Conditions Report at the conclusion of Phase 1 with the Phase 2 Proposal. Notwithstanding the above, provided the parties sign the Phase 2 Amendment, Design-Builder shall not be entitled to a Change Order for Differing Site Conditions pursuant to Section 4.2.1 above if the Differing Site Condition could have been discovered, with reasonable diligence, during Phase 1 and was not included in the Phase 2 Proposal, including all information required in Exhibit C.

4.2.3 Upon encountering a Differing Site Condition, Design-Builder shall provide prompt written notice to Owner of such condition, which notice shall not be later than fourteen (14) days after such condition has been encountered. Design-Builder shall, to the extent reasonably possible, provide such notice before the Differing Site Condition has been substantially disturbed or altered. Design-Builder and Owner shall work together cooperatively to determine the appropriate course of action

Article 5

Insurance and Bonds

5.1 Design-Builder's Insurance Requirements.

5.1.1 Design-Builder is responsible for procuring and maintaining the insurance for the coverage amounts all as set forth in the Insurance Exhibit to the Agreement. Coverage shall be secured from insurance companies authorized to do business in Alaska , and with a minimum rating set forth in the Agreement.

5.1.2 Design-Builder's insurance shall specifically delete any design-build or similar exclusions that could compromise coverages because of the design-build delivery of the Project.

5.1.3 Upon signing and returning the signed Agreement to the Owner, and in any event, prior to performing any Work under this Agreement, Design-Builder shall provide Owner with certificates and a Proof of Insurance in the form attached as Exhibit I evidencing that (i) all insurance obligations required by the Contract Documents are in full force and in effect and will remain in effect for the duration required by the Contract Documents and (ii) no insurance coverage will be canceled, renewal refused, or materially changed unless at least thirty (30) days prior written notice is given to Owner. If any of the foregoing insurance coverages are required to remain in force after final payment are reasonably available, an additional certificate and Proof of Insurance evidencing continuation of such coverage shall be submitted with the Final Application for Payment. If any information concerning reduction of coverage is not furnished by the insurer, it shall be furnished by the Design-Builder with reasonable promptness according to the Design-Builder's information and belief.

5.2 Owner's Liability Insurance.

5.2.1 Owner will maintain its usual insurance applicable to the Project.

5.3 Design-Builder's Property (Builder's Risk) Insurance.

5.3.1 Design-Builder shall procure and maintain from insurance companies authorized to do business in Alaska builder's risk insurance upon the entire Project to the full insurable value of the Project, including professional fees, overtime premiums, and all other expenses incurred to replace or repair the insured property. The builder's risk insurance obtained by Design-Builder shall be the broadest coverage commercially available and shall include as additional insureds the interests of Owner, Design-Builder, Design Consultants and Subcontractors of any tier. Such insurance shall include but not be limited to the perils of fire and extended coverage, theft, vandalism, malicious mischief, collapse, flood, earthquake, debris removal, testing and start-up of building systems, and reasonable compensation for architect's and contractor's services and expenses as a result of a loss, and other perils or causes of loss as called for in the Contract Documents. The builder's risk insurance shall include physical loss or damage to the Work, including materials and equipment in transit, at the Site or at another location as may be indicated in Design-Builder's Application for Payment and approved by Owner. The Design-Builder is responsible for the payment of any deductibles under the insurance required by this Section 5.3.1.

5.3.2 Prior to Design-Builder commencing any Work, Design-Builder shall provide Owner with certificates and a Proof of Insurance in the form attached as Exhibit I evidencing that (i) all Design-Builder's property insurance obligations required by the Contract Documents are in full force and in effect and will remain in effect until Design-Builder has completed all of the Work and has received final acceptance from Owner and (ii) no insurance coverage will be canceled, renewal refused, or materially changed unless at least thirty (30) days prior written notice is given to Owner. Design-Builder shall notify Owner within ten (10) days of receipt of any notice of cancellation or non-renewal sent by the insurance company. Design-Builder's property insurance shall not lapse or be canceled if Owner occupies a portion of the Work pursuant to Section 6.7.2 hereof. Design-Builder shall provide Owner with the necessary endorsements from the insurance company prior
to occupying a portion of the Work.

5.3.3 Any loss covered under Design-Builder's property insurance shall be adjusted with Owner and Design-Builder and made payable to both of them as trustees for the insureds as their interests may appear, subject to any applicable mortgage clause. All insurance proceeds received as a result of any loss will be placed in a separate account and distributed in accordance with such agreement as the interested parties may reach. Any disagreement concerning the distribution of any proceeds will be resolved in accordance with Article 10 hereof.

5.3.4 Owner and Design-Builder waive against each other and each of their Subcontractors, Design Consultants, Subcontractors, agents and employees of each of them, all damages covered by property insurance provided herein, except such rights as they may have to the proceeds of such insurance. Design-Builder and Owner shall, where appropriate, require similar waivers of subrogation from Owner's separate contractors, Design Consultants and Subcontractors and shall require each of them to include similar waivers in their contracts. These waivers of subrogation shall not contain any restriction or limitation that will impair the full and complete extent of its applicability to any person or entity unless agreed to in writing prior to the execution of this Agreement.

Article 6

Payment

6.1 Schedule of Values.

6.1.1 Design-Builder shall submit for Owner's review and approval a schedule of values for all of the Work pursuant to Exhibit C. The Schedule of Values will (i) subdivide the Work into its respective parts, (ii) include values for all items comprising the Work and (iii) serve as the basis for monthly progress payments made to Design-Builder throughout the Work. Design-Builder will furnish, as part of the Schedule of Values, adequate and reliable cost justification and documentation so as to provide both Owner and Design Builder a transparent understanding of the cost data estimates and bids that comprise the initial baseline Schedule of Values as well as any updates thereto.

6.1.2 The Owner will timely review and approve the schedule of values so as not to delay the submission of the Design-Builder's first application for payment. The Owner and Design-Builder shall timely resolve any differences so as not to delay the Design-Builder's submission of its first application for payment.

6.2 Monthly Progress Payments.

6.2.1 On or before the date established in the Agreement, Design-Builder shall submit for Owner's review and approval its Application for Payment requesting payment for all Work performed as of the date of the Application for Payment. The Application for Payment shall be accompanied by all supporting documentation required by the Contract Documents and/or established at the meeting required by Section 2.1.4 hereof. When Design-Builder submits its monthly Application for Payment, it shall include, in addition to other requirements a waiver and release of claims and mechanic's liens. Payments will not be considered due and payable by Owner unless these forms are properly completed and timely received by Owner.

6.2.2 Reconciliation. At the time it submits an Application for Payment, Design-Builder shall analyze and reconcile, to the satisfaction of Owner, the actual progress of the Work with the Project Schedule.

6.2.3 Stored Materials. If authorized by Owner, the Application for Payment may include request for payment for material delivered to the Site and suitably stored, or for completed preparatory work. Payment may similarly be requested for material stored off Site, provided Design-Builder complies with or furnishes satisfactory evidence of the following:

- .1 The material will be placed in a warehouse that is structurally sound, dry, lighted and suitable for the materials to be stored;
- .2 The warehouse is located within a 10-mile radius of the Project. Other locations may

be utilized, if approved in writing, by Owner;

- .3 Only materials for the Project are stored within the warehouse (or a secure portion of a warehouse set aside for the Project);
- .4 Design-Builder furnishes Owner a certificate of insurance and Proof of Insurance (Exhibit I) extending Design-Builder's insurance coverage for damage, fire, and theft to cover the full value of all materials stored, or in transit;
- .5 The warehouse (or secure portion thereof) is continuously under lock and key, and only Design-Builder's authorized personnel shall have access;
- .6 Owner shall at all times have the right of access in company of Design-Builder;
- .7 Design-Builder and its surety assume total responsibility for the stored materials; and
- .8 Design-Builder furnishes to Owner certified lists of materials stored, bills of lading, invoices, and other information as may be required, and shall also furnish notice to Owner when materials are moved from storage to the Site.

6.2.4 All discounts offered by Subcontractor, Sub-Subcontractors and suppliers to Design-Builder for early payment shall accrue one hundred percent to Design-Builder to the extent Design-Builder advances payment. Unless Owner advances payment to Design-Builder specifically to receive the discount, Design-Builder may include in its Application for Payment the full undiscounted cost of the item for which payment is sought.

6.2.5 By submitting the Application for Payment Design-Builder (a) represents that the Work described herein has been performed consistent with the Contract Documents and has progressed to the point indicated in the Application for Payment; (b) certifies that all Subcontractors have been paid, less earned retainage in accordance with the Agreement, as their interests appeared in the last preceding Application for Payment, if payment for the application has been paid to the Design-Builder more than 10-days prior to the current application; and (c) recertifies that Design-Builder's prior certifications are true and correct, to the best of Design-Builder's knowledge, as of the date of the Application for Payment., and that title to all Work will pass to Owner free and clear of all claims, liens, encumbrances, and security interests upon the incorporation of the Work into the Project, or upon Design-Builder's receipt of payment, whichever occurs earlier.

6.3 Payments.

6.3.1 Payment. Owner shall make progress payments, in such amounts as Owner determines are properly due, within thirty (30) days after receipt of the Initial invoice or a properly executed Application for Payment. Owner shall notify Design-Builder if an Application for Payment does not comply with the requirements of the Contract Documents or if payment will be withheld.

6.3.2 Retainage. Owner shall retain five percent (5%) of the amount of each progress payment due under an Application for Payment and receipt of all documents required by Governmental Rule or the Contract Documents, including, at Owner's request, consent of surety to release of the retainage. At Owner's option, it may decide to stop collecting retainage after the project reaches 50% completion.

6.3.3 Title to Work Covered by Progress Payments. Title to all Work and materials covered by a progress payment shall pass to Owner at the time of such payment free and clear of all liens, claims, security interests, and encumbrances. Passage of title shall not, however, relieve Design-Builder from any of its duties and responsibilities for the Work or materials, or waive any rights of Owner to insist on full compliance by Design-Builder with the Contract Documents.

6.4 Owner's Right to Withhold Payment and Offset

6.4.1 Withholding of Payment. Owner may withhold or, on account of subsequently discovered evidence, nullify the whole or part of any payment to such extent as may be necessary to protect Owner from loss or damage for reasons including:

- .1 Work not in accordance with the Contract Documents;
- .2 Reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Price;

- .3 Work by Owner to correct defective Work or to complete the Work;
- .4 Design-Builder's failure to perform in accordance with the Contract Documents; and
- .5 Costs, claims, or liability that are the result of Design-Builder's failure to perform in accordance with the Contract Documents, including Liquidated Damages.

6.4.2 Owner's Offset Rights. If, at the time any payment by Owner is due under this Article 6, Design-Builder is liable to Owner for any amounts in accordance with the provisions of the Contract Documents (including Liquidated Damages), Owner may deduct the outstanding amount of such claims against Design-Builder from the amount payable to Design-Builder.

6.4.3 Payment Disputes. If Design-Builder disputes Owner's determination of payments due hereunder, or disputes any offsets or withholding by Owner, Design-Builder shall have the right to submit the dispute for resolution in accordance with Article 10. Pending resolution of any such dispute, Design-Builder shall continue its performance of the Work in accordance with the Contract Documents. Amounts determined by such resolution process to have been properly due shall be payable by Owner within thirty (30) days after (a) the effective date of the Parties' negotiated settlement or (b) absent such settlement, the arbitration award issued pursuant to Section 10.3.2.

6.5 Right to Stop Work and Interest.

6.5.1 If Owner fails to pay timely Design-Builder any undisputed amount that becomes due, Design-Builder, in addition to all other remedies provided in the Contract Documents, may stop Work pursuant to Section 11.3 hereof, provided Design-Builder gives Owner five business days' written notice of its intent to stop work and an opportunity to cure the late payment. All payments due and unpaid shall bear interest at the rate set forth in the Agreement.

6.6 Design-Builder's Payment Obligations.

6.6.1 Design-Builder will pay Design Consultants and Subcontractors, in accordance with its contractual obligations to such parties and Alaska state law, all the amounts Design-Builder has received from Owner on account of their work. Design-Builder will impose similar requirements on Design Consultants and Subcontractors to pay those parties with whom they have contracted. Design-Builder will indemnify and defend Owner against any claims for payment and mechanic's liens as set forth in Section 7.3 hereof.

6.7 Substantial Completion.

6.7.1 Design-Builder shall notify Owner when it believes the Work, or to the extent permitted in the Contract Documents, a portion of the Work, is Substantially Complete. Within five (5) days of Owner's receipt of Design-Builder's notice, Owner and Design-Builder will jointly inspect such Work to verify that it is Substantially Complete in accordance with the requirements of the Contract Documents. If such Work is Substantially Complete, Owner shall prepare and issue a Certificate of Substantial Completion that will set forth (i) the date of Substantial Completion of the Work or portion thereof, (ii) the remaining items of Work that have to be completed before final payment, (iii) provisions (to the extent not already provided in the Contract Documents) establishing Owner's and Design-Builder's responsibility for the Project's security, maintenance, utilities and insurance pending final payment, and (iv) an acknowledgment that warranties commence to run on the date of Substantial Completion, except as may otherwise be noted in the Certificate of Substantial Completion.

6.7.2 Owner, at its option, may use a portion of the Work which has been determined to be Substantially Complete, provided, however, that (i) a Certificate of Substantial Completion has been issued for the portion of Work addressing the items set forth in Section 6.7.1 above, (ii) Design-Builder and Owner have obtained the consent of their sureties and insurers, and to the extent applicable, the appropriate government authorities having jurisdiction over the Project, and (iii) Owner and Design-Builder agree that Owner's use or occupancy will not interfere with Design-Builder's completion of the remaining Work.

6.8 Final Payment.

6.8.1 Application for Final Payment. Once Owner has issued a Certificate of Final Acceptance, Design-Builder shall be entitled to submit an Application for Final Payment, which

application will include the following information:

- .1 an affidavit that all payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which Owner might in any way be responsible have been paid or otherwise satisfied and that there are no claims, obligations, or liens outstanding or unsatisfied for labor, services, Equipment and Material, taxes, or other items performed, furnished or incurred for or in connection with the Work which will in any way affect Owner's interests;
- .2 a written notice of any outstanding disputes or claims between Design-Builder and any of its Subcontractors, including the amounts and other details thereof;
- .3 a general release executed by Design-Builder waiving, upon receipt of final payment by Design-Builder, all claims, except those claims pending in accordance with Article 10;
- .4 consent of Design-Builder's surety to final payment;
- .5 certificates of insurance and Proof of Insurance (Exhibit I) confirming that required coverages will remain in effect and will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to Owner, consistent with the requirements of the Contract Documents;
- .6 a written statement that Design-Builder knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents;
- .7 Owner's receipt of verification from the State of Alaska Department of Labor and Workforce Development that:
 - (i) Design-Builder has complied with AS 36.05.045(a) and
 - (ii) the Department is not conducting an investigation and
 - (iii) the Department has not issued a notice of violation of AS 36.05 to Design-Builder or any subcontractor..

6.8.2 Payment. Within thirty (30) days after receipt of an acceptable Application for Final Payment, Owner shall pay to Design-Builder the unpaid balance of the Contract Price (less any Retainage per Article 6), reduced by any amounts owed by Design-Builder to Owner pursuant to this Contract which have not been paid by Design-Builder. Retainage funds shall be released pursuant to state law.

6.8.3 Upon making final payment, Owner waives all claims against Design-Builder except claims relating to (i) Design-Builder's failure to satisfy its payment obligations, if such failure affects Owner's interests, (ii) Design-Builder's failure to complete the Work consistent with the Contract Documents, including defects appearing after Substantial Completion and (iii) the terms of any special warranties required by the Contract Documents.

6.8.4 Deficiencies in the Work discovered after Substantial Completion, whether or not such deficiencies would have been included on the Punch List if discovered earlier, shall be deemed warranty Work. Such deficiencies shall be corrected by Design-Builder under Sections 2.10 and 2.11 herein and shall not be a reason to withhold final payment from Design-Builder, provided, however, that Owner shall be entitled to withhold from the Final Payment the reasonable value of completion of such deficient work until such work is completed.

Article 7

Indemnification

7.1 Patent and Copyright Infringement.

7.1.1 Design-Builder shall defend any action or proceeding brought against Owner based on any claim that the Work, or any part thereof, or the operation or use of the Work or any part thereof,

constitutes infringement of any United States patent or copyright, now or hereafter issued. Owner shall give prompt written notice to Design-Builder of any such action or proceeding and will reasonably provide authority, information and assistance in the defense of same. Design-Builder shall indemnify and hold harmless Owner from and against all damages and costs, including but not limited to attorneys' fees and expenses awarded against Owner or Design-Builder in any such action or proceeding. Design-Builder agrees to keep Owner informed of all developments in the defense of such actions.

7.1.2 If Owner is enjoined from the operation or use of the Work, or any part thereof, as the result of any patent or copyright suit, claim, or proceeding, Design-Builder shall at its sole expense take reasonable steps to procure the right to operate or use the Work. If Design-Builder cannot so procure such right within a reasonable time, Design-Builder shall promptly, at Design-Builder's option and at Design-Builder's expense, (i) modify the Work so as to avoid infringement of any such patent or copyright or (ii) replace said Work with Work that does not infringe or violate any such patent or copyright.

7.1.3 Sections 7.1.1 and 7.1.2 above shall not be applicable to any suit, claim or proceeding based on infringement or violation of a patent or copyright (i) relating solely to a particular process or product of a particular manufacturer specified by Owner and not offered or recommended by Design-Builder to Owner or (ii) arising from modifications to the Work by Owner or its agents after acceptance of the Work. If the suit, claim or proceeding is based upon events set forth in the preceding sentence, Owner shall defend, indemnify and hold harmless Design-Builder to the same extent Design-Builder is obligated to defend, indemnify and hold harmless Owner in Section 7.1.1 above.

7.1.4 The obligations set forth in this Section 7.1 shall constitute the sole agreement between the parties relating to liability for infringement of violation of any patent or copyright.

7.2 Tax Claim Indemnification.

7.2.1 If, in accordance with Owner's direction, an exemption for all or part of the Work is claimed for taxes, Owner shall indemnify, defend and hold harmless Design-Builder from and against any liability, penalty, interest, fine, tax assessment, attorneys' fees or other expenses or costs incurred by Design-Builder as a result of any action taken by Design-Builder in accordance with Owner's directive. Owner shall furnish Design-Builder with any applicable tax exemption certificates necessary to obtain such exemption, upon which Design-Builder may rely.

7.3 Payment Claim Indemnification.

7.3.1 Provided that Owner is not in breach of its contractual obligation to make payments to Design-Builder for the Work, Design-Builder shall indemnify, defend and hold harmless Owner from any claims or mechanic's liens brought against Owner or against the Project as a result of the failure of Design-Builder, or those for whose acts it is responsible, to pay for any services, materials, labor, equipment, taxes or other items or obligations furnished or incurred for or in connection with the Work. Within three (3) days of receiving written notice from Owner that such a claim or mechanic's lien has been filed, Design-Builder shall commence to take the steps necessary to discharge said claim or lien, including, if necessary, the furnishing of a mechanic's lien bond. If Design-Builder fails to do so, Owner will have the right to discharge the claim or lien and hold Design-Builder liable for costs and expenses incurred, including attorneys' fees.

7.4 Design-Builder's General Indemnification.

7.4.1 Except as set forth in Section 7.4.2 below, Design-Builder, to the fullest extent permitted by law, shall indemnify, hold harmless and defend Owner, its Consultants, and their respective, its officers, directors, and employees (collectively "Indemnitees") from and against non-party claims, losses, damages, liabilities, including attorneys' fees and expenses, for bodily injury, sickness or death, and property damage or destruction (other than to the Work itself) but only to the extent resulting from the negligent acts or omissions of Design-Builder, Design Consultants, Subcontractors, anyone employed directly or indirectly by any of them or anyone for whose acts any of them may be liable. Design-Builder's duty to indemnify shall not apply to liability for damages arising out of Design-Builder's services or out of bodily injury to persons or damage to property that are (a) caused by or resulting from the sole negligence of Indemnitee or (b) caused by or resulting

from the concurrent negligence of (i) Indemnitee, its agents or employees and (ii) Design-Builder, its agents or employees, with such liability limited only to the extent of the negligence of Design-Builder, it's agents or employees.

7.4.2 For indemnity obligations that arise from professional errors and omissions, Design-Builder, to the fullest extent permitted by law, shall indemnify Owner, its officers, directors, and employees from and against claims, losses, damages, liabilities, including attorneys' fees and expenses, for non-party bodily injury, sickness, or death and non-party property damage or destruction (other than to the Work itself) but only to the extent resulting from the negligent acts or omissions of Design-Builder, Design Consultants, Subcontractors, anyone employed directly or indirectly by any of them or anyone for whose acts any of them may be liable.

7.4.3 If an employee of Design-Builder, Design Consultants, Subcontractors, anyone employed directly or indirectly by any of them or anyone for whose acts any of them may be liable has a claim against Owner, its officers, directors, employees, or agents, Design-Builder's indemnity obligation set forth in Sections 7.4.1 and 7.4.2 above shall not be limited by any limitation on the amount of damages, compensation or benefits payable by or for Design-Builder, Design Consultants, Subcontractors, or other entity under any employee benefit acts, including workers' compensation or disability acts. Solely for the purposes of the indemnification obligations under this Agreement, Design Builder specifically and expressly waives any immunity that may be granted it under the worker's compensation laws under the Alaska Workers' Compensation Act, provided that such waiver shall be expressly limited to Design-Builder's indemnification obligations herein and shall not be intended as a benefit to any third party. Further, the indemnification obligation under this Agreement shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable to or for any third party under workers compensation acts, disability benefits acts, or other employee benefits acts.

7.4.3 THE PARTIES ACKNOWLEDGE THAT THE INDEMNIFICATION OBLIGATIONS IN THIS AGREEMENT AND THE WAIVER OF IMMUNITY UNDER THE ALASKA WORKERS' COMPENSATION ACT WERE MUTUALLY NEGOTIATED.

OWNER'S INITIALS: (____) DESIGN-BUILDER'S INITIALS: (____)

7.4.4 The Owner shall not be responsible or be held liable for any damage to person or property consequent upon the use, misuse or failure of any crane, hoist, rigging, blocking, scaffolding or other equipment used by the Design-Builder or any of its Subcontractors, even though the said crane, hoist, rigging, blocking, scaffolding, or other equipment be furnished or loaned to the Design-Builder by the Owner. The acceptance and/or use of any such crane, hoist, rigging, blocking, scaffolding or other equipment by the Design-Builder or its Subcontractors shall be construed to mean that the Design-Builder accepts all responsibility for any claims for damages whatsoever resulting from the use, misuse or failure of such apparatus whether such damages by its own employees or property or to the employees or property of other contractors, the Owner, or otherwise.

7.5 Lower Tier Contractors Indemnification Obligations

7.5.1 Design-Builder shall include in its contracts with all lower tier contractors, including but not limited to its Design Consultant, Subconsultants, and Subcontractors, the indemnification obligations set forth in this Agreement and the General Conditions and shall include Owner as an Indemnitee for all such indemnification provisions.

7.6 Limited Recourse.

7.6.1 None of the obligations set forth in this Agreement (on behalf of any Party) constitute personal obligations of any natural persons who are the officers, shareholders, members, partners, employees, or agents of any Party unless the natural person is expressly identified as a contracting party. All Parties to this Agreement shall not seek recourse against any natural person described

herein. This provision, however, shall not protect such natural persons from liability for willful misconduct, illegal acts or intentional violation of any duty of corporate loyalty.

Article 8

Time

8.1 Obligation to Achieve the Contract Times.

8.1.1 Design-Builder agrees that it will commence performance of the Work and achieve the Contract Time(s) in accordance with Article 5 of the Agreement and any Amendment to the Agreement.

8.2 Delays to the Work.

8.2.1 If Design-Builder is delayed on the critical path in the performance of the Work due to acts, omissions, conditions, events, or circumstances beyond its control and due to no fault of its own or those for whom Design-Builder is responsible, the Contract Time(s) for performance shall be reasonably extended by Change Order. By way of example, events that will entitle Design-Builder to an extension of the Contract Time(s) include acts or omissions of Owner or anyone under Owner's control (including separate contractors), changes in the Work, Differing Site Conditions, Hazardous Conditions, and Force Majeure Events.

8.2.2 In addition to Design-Builder's right to a time extension for those events set forth in Section 8.2.1 above, Design-Builder shall also be entitled to the Design-Builder's Delay Rate set forth in Section 6.4.5.4 of the Agreement, provided, however, that the Contract Price shall not be adjusted for Force Majeure Events unless otherwise provided in the Agreement.

Article 9

Changes to the Contract Price and Time

9.1 Change Orders.

9.1.1 A Change Order is a written instrument issued after execution of the Agreement signed by Owner and Design-Builder, stating their agreement upon all of the following:

- .1 The scope of the change in the Work;
- .2 The amount of the adjustment to the Contract Price or any Commercial Term; and
- .3 The extent of the adjustment to the Contract Time(s) or any Commercial Term.

9.1.2 All changes in the Work authorized by applicable Change Order shall be performed under the applicable conditions of the Contract Documents. Owner and Design-Builder shall negotiate in good faith and as expeditiously as possible the appropriate adjustments for such changes.

9.1.3 If Owner requests a proposal for a change in the Work from Design-Builder and subsequently elects not to proceed with the change, a Change Order shall be issued to reimburse Design-Builder for reasonable costs incurred for estimating services, design services and services involved in the preparation of proposed revisions to the Contract Documents.

9.1.4 Owner may make changes in the Project, including but not limited to adding and/or removing Work from the Project. In such case, Design-Builder shall adjust the remaining Work to meet Owner's Project changes as reasonably possible within the applicable Commercial Term. At Owner's sole discretion, it may remove Work from the Project rather than increase the applicable Commercial Term to equitably adjust for claims by Design-Builder or increased costs on the Project.

9.2 Work Change Directives.

9.2.1 A Work Change Directive is a written order prepared and signed by Owner directing a

change in the Work prior to agreement on an adjustment in the Contract Price and/or the Contract Time(s).

9.2.2 Owner and Design-Builder shall negotiate in good faith and as expeditiously as possible the appropriate adjustments for the Work Change Directive. Upon reaching an agreement, the parties shall prepare and execute an appropriate Change Order reflecting the terms of the agreement.

9.3 Minor Changes in the Work.

9.3.1 Minor changes in the Work do not involve an adjustment in the Contract Price and/or Contract Time(s) and do not materially and adversely affect the Work, including the design, quality, performance and workmanship required by the Contract Documents. Design-Builder may make minor changes in the Work consistent with the intent of the Contract Documents, provided, however, that Design-Builder shall promptly inform Owner, in writing, of any such changes and record such changes on the documents maintained by Design-Builder.

9.4 Contract Price Adjustments.

9.4.1 The increase or decrease in Contract Price resulting from a change in the Work shall be determined by one or more of the following methods:

.1 Unit prices set forth in the Agreement or as subsequently agreed to between the parties;

.2 A mutually accepted lump sum, properly itemized and supported by sufficient substantiating data to permit evaluation by Owner; or

.3 As set forth in Section 9.4.3 below.

9.4.2 If unit prices are set forth in the Contract Documents or are subsequently agreed to by the parties, but application of such unit prices will cause substantial inequity to Owner or Design-Builder because of differences in the character or quantity of such unit items as originally contemplated, such unit prices shall be equitably adjusted.

9.4.3 Pricing Components for Changed Work. The value of any Changed Work that is compensable, of any disputed Work Change Directive and of any other increase or decrease in the Contract Price, including a Claim, shall be limited to the following costs to the extent that the Design-Builder demonstrates that the costs are both reasonable, actually incurred, not otherwise disallowed (collectively "Changed Work"), Changed Work shall be subject to any Not to Exceed Amount agreed upon by the Parties.

- .1 For Changed Work that is priced on the basis of the Cost of the Work, Design Builder shall be compensated up to a Not to Exceed Sum for the following:
 - a. The Cost of Changed Work, which shall be determined in the same way as the Cost of the Work set forth in Section 6.3 of the Agreement;
 - b. Any Allowance pursuant to Section 6.4.1 of the Agreement; and
 - c. Design Builder's Contingency pursuant to Section 6.4.4 of the Agreement.
- .2 For Extra Work that is priced on a Lump Sum basis, Design Builder shall be compensated pursuant to Section 6.4.3 of the Agreement.
- .3 If the parties have entered into the Phase 2 Amendment, the Cost of Extra Work shall not include any items included in the Lump Sum General Conditions Amount pursuant to Section 6.4.5 of the Agreement or the General Conditions Costs set forth in Section 6.3.15 of the Agreement.
- .4 Design-Builder shall be entitled to include an appropriate amount for Overhead and Profit, not to exceed 10% in the compensation for Changed Work. The fee for Subcontractor's Changed Work shall be computed as follows:
 - a. Design-Builder shall receive eight percent (8%) of fixed-price costs or six percent (6%) of the time-and-material costs owed directly to a Subcontractor for materials

supplied and/or Work properly performed by that Subcontractor or owed directly to a Design Consultant for services it properly performs.

b. Each Subcontractor of any tier shall receive eight percent (8%) of fixed-price costs or six percent (6%) of the time-and-material costs owed directly to a lower-tier Subcontractor for materials supplied and/or Work properly performed by that Subcontractor.

9.4.4 If Owner and Design-Builder disagree upon whether Design-Builder is entitled to be paid for any services required by Owner, or if there are any other disagreements over the scope of Work or proposed changes to the Work, Owner and Design-Builder shall resolve the disagreement pursuant to Article 10 hereof. As part of the negotiation process, Design-Builder shall furnish Owner with a good faith estimate of the costs to perform the disputed services in accordance with Owner's interpretations. If the parties are unable to agree and Owner expects Design-Builder to perform the services in accordance with Owner's interpretations, Design-Builder shall proceed to perform the disputed services, conditioned upon Owner issuing a written order to Design-Builder (i) directing Design-Builder to proceed and (ii) specifying Owner's interpretation of the services that are to be performed. If this occurs, Design-Builder shall be entitled to submit in its Applications for Payment an amount equal to fifty percent (50%) of its reasonable estimated direct cost to perform the services, and Owner agrees to pay such amounts, with the express understanding that (i) such payment by Owner does not prejudice Owner's right to argue that it has no responsibility to pay for such services and (ii) receipt of such payment by Design-Builder does not prejudice Design-Builder's right to seek full payment of the disputed services if Owner's order is deemed to be a change to the Work.

9.5 Emergencies.

9.5.1 In any emergency affecting the safety of persons and/or property, Design-Builder shall act, at its discretion, to prevent threatened damage, injury or loss. Any change in the Contract Price and/or Contract Time(s) on account of emergency work shall be determined as provided in this Article 9.

Article 10

Contract Adjustments and Disputes

10.1 Requests for Contract Adjustments and Relief.

10.1.1 If either Design-Builder or Owner believes that it is entitled to relief against the other for any event arising out of or related to the Work or Project, such party shall provide written notice to the other party of the basis for its claim for relief. Such notice shall, if possible, be made prior to incurring any cost or expense and in accordance with any specific notice requirements contained in applicable sections of these General Conditions of Contract. In the absence of any specific notice requirement, written notice shall be given within a reasonable time, not to exceed twenty-one (21) days, after the occurrence giving rise to the claim for relief or after the claiming party reasonably should have recognized the event or condition giving rise to the request, whichever is later. Such written notice shall be separate from the Design Log or Trend Log maintained by the Design-Builder, unless the parties specifically agree to allow the Trend Log to operate as such written notice of claims. The Design-Builder shall provide more complete information with respect to the claim within fourteen (14) days of the initial notice, the more complete information shall include sufficient information to advise the other party of the circumstances giving rise to the claim for relief, the specific contractual adjustment or relief requested and the basis of such request. The failure to provide timely written notice of any claim shall operate as a waiver of such claim, but only to the extent that the failure to provide timely written notice prejudices the position of the non-claiming party.

10.2 Dispute Avoidance and Resolution.

10.2.1 The parties are fully committed to working with each other throughout the Project and agree to communicate regularly with each other at all times so as to avoid or minimize disputes or

disagreements. If disputes or disagreements do arise, Design-Builder and Owner each commit to resolving such disputes or disagreements in an amicable, professional and expeditious manner so as to avoid unnecessary losses, delays and disruptions to the Work.

10.2.2 Design-Builder and Owner will first attempt to resolve disputes or disagreements at the field level through discussions between Design-Builder's Representative and Owner's Representative which shall conclude within fourteen (14) days of the written notice provided for in Section 10.1.1 unless the Owner and Design-Builder mutually agree otherwise.

10.2.3 If a dispute or disagreement cannot be resolved through Design-Builder's Representative and Owner's Representative, Design-Builder's Senior Representative and Owner's Senior Representative, upon the request of either party, shall meet as soon as conveniently possible, but in no case later than thirty (30) days after such a request is made, to attempt to resolve such dispute or disagreement. Five (5) days prior to any meetings between the Senior Representatives, the parties will exchange relevant information that will assist the parties in resolving their dispute or disagreement.

10.2.4 If after meeting the Senior Representatives determine that the dispute or disagreement cannot be resolved on terms satisfactory to both parties, the parties shall submit within thirty (30) days of the conclusion of the meeting of Senior Representatives the dispute or disagreement to non-binding mediation. The mediation shall be conducted by a mutually agreeable impartial mediator, or if the parties cannot so agree, a mediator designated by the American Arbitration Association ("AAA") pursuant to its Construction Industry Mediation Rules. The mediation will be governed by and conducted pursuant to a mediation agreement negotiated by the parties or, if the parties cannot so agree, by procedures established by the mediator. Unless otherwise mutually agreed by the Owner and Design-Builder and consistent with the mediator's schedule, the mediation shall commence within ninety (90) days of the submission of the dispute to mediation. Good faith mediation is a condition precedent to proceeding with arbitration or other binding dispute resolution procedure. Representatives of the parties with authority to resolve the dispute shall be present at any mediation.

10.3 Dispute Resolution.

10.3.1 Any claims, disputes or controversies between the parties arising out of or relating to the Agreement, or the breach thereof, which have not been resolved in accordance with the procedures set forth in Section 10.2 above, shall be decided by the Superior Court of Alaska First Judicial District, unless the parties mutually agree otherwise.

10.3.2 The prevailing party in any final, binding dispute proceeding upon which the parties may agree, shall be entitled to recover from the other party reasonable attorneys' fees and expenses incurred by the prevailing party. The prevailing party, if any, shall be determined by the applicable binding dispute tribunal.

10.4 Duty to Continue Performance.

10.4.1 Unless provided to the contrary in the Contract Documents, Design-Builder shall continue to perform the Work and Owner shall continue to satisfy its payment obligations for undisputed amounts to Design-Builder as well as any further amounts pursuant to Section 9.4.4, pending the final resolution of any dispute or disagreement between Design-Builder and Owner.

10.5 CONSEQUENTIAL DAMAGES.

10.5.1 NOTWITHSTANDING ANYTHING HEREIN TO THE CONTRARY (EXCEPT AS SET FORTH IN SECTION 10.5.2 BELOW), NEITHER DESIGN-BUILDER NOR OWNER SHALL BE LIABLE TO THE OTHER FOR ANY CONSEQUENTIAL LOSSES OR DAMAGES, WHETHER ARISING IN CONTRACT, WARRANTY, TORT (INCLUDING NEGLIGENCE), STRICT LIABILITY OR OTHERWISE, INCLUDING BUT NOT LIMITED TO LOSSES OF USE, PROFITS, BUSINESS, REPUTATION OR FINANCING.

10.5.2 The consequential damages limitation set forth in Section 10.5.1 above is not intended to affect the payment of liquidated damages or lost early completion bonus, if any, set forth in Article 5 of the Agreement, which both parties recognize has been established, in part, to reimburse Owner or reward Design-Builder for some damages that might otherwise be deemed to be consequential.

10.5.3 The consequential damages limitation set forth in Section 10.5.1 above is not intended to affect the ability of any party to recover consequential damages that are covered by insurance.

Article 11

Stop Work and Termination for Cause

11.1 Owner's Right to Stop Work.

11.1.1 Owner may, without cause and for its convenience, order Design-Builder in writing to stop and suspend the Work. Such suspension shall not exceed sixty (60) consecutive days or aggregate more than ninety (90) days during the duration of the Project.

11.1.2 Design-Builder is entitled to seek an adjustment of the Contract Price and/or Contract Time(s) if its cost or time to perform the Work has been adversely impacted by any suspension of stoppage of the Work by Owner.

11.2 Owner's Right to Perform and Terminate for Cause.

11.2.1 If Design-Builder persistently fails to (i) provide a sufficient number of skilled workers, (ii) supply the materials required by the Contract Documents, (iii) comply with applicable Legal Requirements, (iv) timely pay, without cause, Design Consultants or Subcontractors, (v) prosecute the Work with promptness and diligence to ensure that the Work is completed by the Contract Time(s), as such times may be adjusted, or (vi) perform material obligations under the Contract Documents, then Owner, in addition to any other rights and remedies provided in the Contract Documents or by law, shall have the rights set forth in Sections 11.2.2 and 11.2.3 below.

11.2.2 Upon the occurrence of an event set forth in Section 11.2.1 above, Owner may provide written notice to Design-Builder that it intends to terminate the Agreement unless the problem cited is cured, or commenced to be cured, within seven (7) days of Design-Builder's receipt of such notice. If Design-Builder fails to cure, or reasonably commence to cure, such problem, then Owner may give a second written notice to Design-Builder of its intent to terminate within an additional seven (7) day period. If Design-Builder, within such second seven (7) day period, fails to cure, or reasonably commence to cure, such problem, then Owner may declare the Agreement terminated for default by providing written notice to Design-Builder of such declaration.

11.2.3 Upon declaring the Agreement terminated pursuant to Section 11.2.2 above, Owner may enter upon the premises and take possession, for the purpose of completing the Work, of all materials, equipment, scaffolds, tools, appliances and other items thereon, which have been purchased or provided for the performance of the Work, all of which Design-Builder hereby transfers, assigns and sets over to Owner for such purpose, and to employ any person or persons to complete the Work and provide all of the required labor, services, materials, equipment and other items. In the event of such termination, Design-Builder shall not be entitled to receive any further payments under the Contract Documents until the Work shall be finally completed in accordance with the Contract Documents. Design-Builder will only be entitled to be paid for Work performed prior to its default. If Owner's cost and expense of completing the Work exceeds the unpaid balance of any Commercial Term, then Design-Builder shall be obligated to pay the difference to Owner. Such costs and expense, including attorneys' fees and expenses, incurred by Owner in connection with the reprocurement and defense of claims arising from Design-Builder's default, subject to the waiver of consequential damages set forth in Section 10.5 hereof.

11.2.4 If Owner improperly terminates the Agreement for cause, the termination for cause will be converted to a termination for convenience in accordance with the provisions of Article 8 of the Agreement.

11.3 Design-Builder's Right to Stop Work.

11.3.1 Design-Builder may, in addition to any other rights afforded under the Contract Documents or at law, stop the Work for the following reasons:

.1 Owner's failure to provide financial assurances as required under Section 3.3 hereof; or

.2 Owner's failure to pay amounts properly due under Design-Builder's Application for Payment.

11.3.2 Should any of the events set forth in Section 11.3.1 above occur, Design-Builder has the right to provide Owner with written notice that Design-Builder will stop the Work unless said event is cured within seven (7) days from Owner's receipt of Design-Builder's notice. Design-Builder shall not stop work unless it provides such written notice and the Owner has failed to cure the reason for default within the seven (7) day cure period. If Owner does not cure the problem within such seven (7) day period, Design-Builder may stop the Work. In such case, Design-Builder shall be entitled to make a claim for adjustment to the Contract Price and Contract Time(s) to the extent it has been adversely impacted by such stoppage.

11.4 Design-Builder's Right to Terminate for Cause.

11.4.1 Design-Builder, in addition to any other rights and remedies provided in the Contract Documents or by law, may terminate the Agreement for cause for the following reasons:

.1 The Work has been stopped for sixty (60) consecutive days, or more than ninety (90) days during the duration of the Project, because of court order, any government authority having jurisdiction over the Work, or orders by Owner under Section 11.1.1 hereof, provided that such stoppages are not due to the acts or omissions of Design-Builder or anyone for whose acts Design-Builder may be responsible.

.2 Owner's failure to provide Design-Builder with any information, permits or approvals that are Owner's responsibility under the Contract Documents which result in the Work being stopped for sixty (60) consecutive days, or more than ninety (90) days during the duration of the Project, even though Owner has not ordered Design-Builder in writing to stop and suspend the Work pursuant to Section 11.1.1 hereof.

.3 Owner's failure to cure the problems set forth in Section 11.3.1 above after Design-Builder has stopped the Work.

11.4.2 Upon the occurrence of an event set forth in Section 11.4.1 above, Design-Builder may provide written notice to Owner that it intends to terminate the Agreement unless the problem cited is cured, or commenced to be cured, within seven (7) days of Owner's receipt of such notice. If Owner fails to cure, or reasonably commence to cure, such problem, then Design-Builder may give a second written notice to Owner of its intent to terminate within an additional seven (7) day period. If Owner, within such second seven (7) day period, fails to cure, or reasonably commence to cure, such problem, then Design-Builder may declare the Agreement terminated for default by providing written notice to Owner of such declaration. In such case, Design-Builder shall be entitled to recover in the same manner as if Owner had terminated the Agreement for its convenience under Article 8 of the Agreement.

11.5 Bankruptcy of Owner or Design-Builder.

11.5.1 If either Owner or Design-Builder institutes or has instituted against it a case under the United States Bankruptcy Code (such party being referred to as the "Bankrupt Party"), such event may impair or frustrate the Bankrupt Party's ability to perform its obligations under the Contract Documents. Accordingly, should such event occur:

.1 The Bankrupt Party, its trustee or other successor, shall furnish, upon request of the non-Bankrupt Party, adequate assurance of the ability of the Bankrupt Party to perform all future material obligations under the Contract Documents, which assurances shall be provided within ten (10) days after receiving notice of the request; and

.2 The Bankrupt Party shall file an appropriate action within the bankruptcy court to seek assumption or rejection of the Agreement within sixty (60) days of the institution of the bankruptcy filing and shall diligently prosecute such action.

If the Bankrupt Party fails to comply with its foregoing obligations, the non-Bankrupt Party shall be entitled to request the bankruptcy court to reject the Agreement, declare the Agreement terminated and pursue any other recourse available to the non-Bankrupt Party under this Article 11.

11.5.2 The rights and remedies under Section 11.5.1 above shall not be deemed to limit the

ability of the non-Bankrupt Party to seek any other rights and remedies provided by the Contract Documents or by law, including its ability to seek relief from any automatic stays under the United States Bankruptcy Code or the right of Design-Builder to stop Work under any applicable provision of these General Conditions of Contract.

Article 12

Electronic Data

12.1 Electronic Data.

12.1.1 The parties recognize that Contract Documents, including drawings, specifications and three-dimensional modeling (such as Building Information Models) and other Work Product may be transmitted among Owner, Design-Builder and others in electronic media as an alternative to paper hard copies (collectively "Electronic Data").

12.2 Transmission of Electronic Data.

12.2.1 Owner and Design-Builder shall agree upon the software and the format for the transmission of Electronic Data. Each party shall be responsible for securing the legal rights to access the agreed-upon format, including, if necessary, obtaining appropriately licensed copies of the applicable software or electronic program to display, interpret and/or generate the Electronic Data.

12.2.2 Neither party makes any representations or warranties to the other with respect to the functionality of the software or computer program associated with the electronic transmission of Work Product. Unless specifically set forth in the Agreement, ownership of the Electronic Data does not include ownership of the software or computer program with which it is associated, transmitted, generated or interpreted.

12.2.3 By transmitting Work Product in electronic form, the transmitting party does not transfer or assign its rights in the Work Product. The rights in the Electronic Data shall be as set forth in Article 4 of the Agreement. Under no circumstances shall the transfer of ownership of Electronic Data be deemed to be a sale by the transmitting party of tangible goods.

12.3 Electronic Data Protocol.

12.3.1 The parties acknowledge that Electronic Data may be altered or corrupted, intentionally or otherwise, due to occurrences beyond their reasonable control or knowledge, including but not limited to compatibility issues with user software, manipulation by the recipient, errors in transcription or transmission, machine error, environmental factors, and operator error. Consequently, the parties understand that there is some level of increased risk in the use of Electronic Data for the communication of design and construction information and, in consideration of this, agree, and shall require their independent contractors, Subcontractors and Design Consultants to agree, to the following protocols, terms and conditions set forth in this Section 12.3.

12.3.2 Electronic Data will be transmitted in the format agreed upon in Section 12.2.1 above, including file conventions and document properties, unless prior arrangements are made in advance in writing.

12.3.3 The Electronic Data represents the information at a particular point in time and is subject to change. Therefore, the parties shall agree upon protocols for notification by the author to the recipient of any changes which may thereafter be made to the Electronic Data, which protocol shall also address the duty, if any, to update such information, data or other information contained in the electronic media if such information changes prior to Final Completion of the Project.

12.3.4 The transmitting party specifically disclaims all warranties, expressed or implied, including, but not limited to, implied warranties of merchantability and fitness for a particular purpose, with respect to the media transmitting the Electronic Data. However, transmission of the Electronic Data via electronic means shall not invalidate or negate any duties pursuant to the applicable standard of care with respect to the creation of the Electronic Data, unless such data is materially changed

or altered after it is transmitted to the receiving party, and the transmitting party did not participate in such change or alteration.

Article 13

Miscellaneous

13.1 Confidential Information.

13.1.1 Confidential Information is defined as information which is determined by the transmitting party to be of a confidential or proprietary nature and: (i) the transmitting party identifies as either confidential or proprietary; (ii) the transmitting party takes steps to maintain the confidential or proprietary nature of the information; and (iii) the document is not otherwise available in or considered to be in the public domain. The receiving party agrees to maintain the confidentiality of the Confidential Information and agrees to use the Confidential Information solely in connection with the Project.

13.2 Assignment.

13.2.1 Neither Design-Builder nor Owner shall, without the written consent of the other assign, transfer or sublet any portion or part of the Work or the obligations required by the Contract Documents.

13.3 Successorship.

13.3.1 Design-Builder and Owner intend that the provisions of the Contract Documents are binding upon the parties, their employees, agents, heirs, successors and assigns.

13.4 Governing Law.

13.4.1 The Agreement and all Contract Documents shall be governed by the laws of Alaska, without giving effect to its conflict of law principles.

13.5 Severability.

13.5.1 If any provision or any part of a provision of the Contract Documents shall be finally determined to be superseded, invalid, illegal, or otherwise unenforceable pursuant to any applicable Legal Requirements, such determination shall not impair or otherwise affect the validity, legality, or enforceability of the remaining provision or parts of the provision of the Contract Documents, which shall remain in full force and effect as if the unenforceable provision or part were deleted.

13.6 No Waiver.

13.6.1 The failure of either Design-Builder or Owner to insist, in any one or more instances, on the performance of any of the obligations required by the other under the Contract Documents shall not be construed as a waiver or relinquishment of such obligation or right with respect to future performance.

13.7 Headings.

13.7.1 The headings used in these General Conditions of Contract, or any other Contract Document, are for ease of reference only and shall not in any way be construed to limit or alter the meaning of any provision.

13.8 Notice.

13.8.1 Whenever the Contract Documents require that notice be provided to the other party, notice will be deemed to have been validly given (i) if delivered in person to the individual intended to receive such notice, (ii) four (4) days after being sent by registered or certified mail, postage prepaid to the address indicated in the Agreement, (iii) if transmitted by facsimile, by the time stated in a machine generated confirmation that notice was received at the facsimile number of the intended recipient, or (iv) by electronic mail, by the time frame stated in the email generated confirmation that notice was received at the facsimile number of the intended recipient.

13.9 Amendments.

13.9.1 The Contract Documents may not be changed, altered, or amended in any way except in writing signed by a duly authorized representative of each party.

EXHIBIT A DESIGN-BUILDER'S INSURANCE REQUIREMENTS HAINES BOROUGH LUTAK DOCK REPLACEMENT

1.1 Insurance Types and Limits.

1.1.1 Design-Builder shall purchase and maintain insurance of the types, with limits of liability, containingsuch endorsements and subject to such terms and conditions as follows, as well as Article 5 of DBIA Document No. 535, *Standard Form of General Conditions of Contract Between Owner and Design-Builder* (2010 Edition):

Type of Insurance [Insert Rating of Carrier]	Minimum LimitsRequired Per Claim/Occurrence	Minimum Limits Required Aggregate Policy Limits	Maximum Deductible
1. Worker's Compensation	Statutory Limits	Statutory Limits	-
2. Employer's Liability (Bodily Injury by Accident)			
a. By Disease	\$ 2,000,000	n/a	
b. Each Accident	\$ 2,000,000	n/a	
c. Each Employee	\$ 2,000,000	n/a	
3. Commercial General Liability			
 a. Bodily Injury/Property Damage per occurrence limit 	\$ 2,000,000	n/a	
b. Bodily Injury/Property Damage aggregate limit	n/a	\$ 4,000,000	
c. Products/Completed Operation aggregatelimit	n/a	\$ 4,000,000	Commercially
d. Personal and Advertising Injury aggregatelimit	n/a	\$ 4,000,000	deductibles (maximum of
e. Medical Expense limit (any one person)	\$ 5,000	n/a	\$50,000). All
4. Contractor's Protective Liability (if applicable)	Separate coverage or included in item #6		be paid by the design-
5. Commercial Automobile Liability	\$ 2,000,000 CSL	n/a	builder.
6. Professional Errors and Omissions pursuant to Section 1.1.3 (A) and 1.1.3 (B) below (per claim/aggregate) providing coverage for services performed by the named insured and any person or entity for whom the named insured is responsible	\$ 2,000,000	\$ 2,000,000	
7. Contractor's Pollution Liability including coverage for microbial matter (if applicable)	\$n/a	n/a	
8. Umbrella Excess Liability Insurance	\$10,000,000	10,000,000	
9. Builder's risk insurance provided pursuant to Article 5 of the General Conditions	\$ An amount equal to the full insurable value of the completed project on a replacement cost basis		

1.1.2 The insurance required by this Section 1.1.1 shall be written for not less than limits of liability specified in the table above or required by law, whichever coverage is greater. Coverages, whether writtenon an occurrence or claims-made basis, shall be maintained without interruption from date of commencement of the Work until date of Final Payment.

1.1.3 PROFESSIONAL LIABILITY INSURANCE.

1.1.3(A) Professional Liability Insurance To Be Provided By Design Consultant. Such policies must provide coverage for the scope of professional services to be provided by or on behalf of the Design Consultant.

The requirement for professional liability coverage on this Project shall be the standard form practice policy provided by Design Consultant.

Design-Builder shall provide Owner with prior written notice of any cancellation or nonrenewal of the Design Consultant's practice policy and shall include in the Design Consultant Agreement a provision requiring the Design Consultant to give the Design-Builder 30 Days written notice of any cancellation or non-renewal.

1.1.3(A).1 The only permissible exclusion, limitation or restriction with respect to construction means, methods and techniques is one that applies to the implementation of such construction means, methods, techniques, sequences, or procedures by the Design Consultant or any person or entity providing design or other professional services as its Sub-Consultant. This exclusion is permissible only if such entities are not performing any construction activities. Notwithstanding the above, a Design Consultant's professional liability policy also cannot contain any restriction, limitation or exclusion pertaining to the design of construction means, methods, techniques, sequences or procedures.

1.1.3(A).2 Any exclusion, limitation or restriction related to Products or Product Design must be modified so as to provide coverage for goods or products installed.

1.1.3(A).3 Faulty Work exclusion, limitation or restriction can only be applicable to the work self-performed by the Design Consultant.

1.1.3(A).4 The policy must provide coverage for damages resulting from delays, including delays in project completion and cost overruns that result from the rendering or failure to render professional services.

1.1.3(A).5 If any portion of the design or other professional service is to be performed by any person or entity other than Design Consultant then it is the responsibility of Design Consultant to ensure that such person or entity provide Design-Builder and Design Consultant with evidence of insurance to comport with this Exhibit.

1.1.3(A).6 Waiver of subrogation is to be provided in favor of Design-Builder and its officers, directors and employees, and (if commercially available) Owner and its officers, directors and employees.

1.1.3(B) Professional Liability Insurance To Be Provided By Design-Builder. Such policies must provide coverage for the scope of professional services to be provided by or on behalf of the Design-Builder.

The requirement for professional liability coverage on this Project shall be the standard form practice policy provided by Design-Builder.

Design-Builder shall provide Owner with prior written notice of any cancellation or nonrenewal of the Design-Builder's practice policy. **1.1.3(B).1** The Design-Builder's policy cannot contain any restriction, limitation or exclusion pertaining to construction means, methods, techniques, sequences or procedures except that the professional liability policy can exclude, limit or restrict coverage for claims, but only to the same extent that such coverage is provided by the Design-Builder's valid and collectible commercial general liability and umbrella/excess liability policies. Notwithstanding the above, a Design-Builder's professional liability policy also cannot contain any restriction, limitation or exclusion pertaining to the design of construction means, methods, techniques, sequences, or procedures.

1.1.3(B).2 Any exclusion, limitation or restriction related to Products or Product Design must be modified so as to provide coverage for goods or products installed.

1.1.3(B).3 Faulty Work exclusion, limitation or restriction can only be applicable to the work self-performed by the Design-Builder.

1.1.3(B).4 The policy must provide coverage for damages resulting from delays, including delays in project completion, and cost overruns that result from the rendering or failure to render professional services.

1.1.3(B).5 If any portion of the design or other professional service is to be performed by any person or entity other than Design-Builder then it is the responsibility of Design-Builder to ensure that person or entity provide Design-Builder with evidence of insurance to comport with this Exhibit.

1.1.3(B).6 Waiver of subrogation is to be provided in favor of Design-Builder and Owner (if commercially available) and their respective officers, directors and employees.

1.1.4 Any coverage required to be maintained after Final Payment shall be identified below: General Liability, including completed operations coverage

Worker's Compensation

Professional Liability, including Contractor's Protective Liability, if applicable. Umbrella Coverage

Such coverage shall remain in place for six (6) years after Substantial Completion.

2.1 Coverage Parameters and Endorsements.

2.1.1 Commercial General Liability Insurance shall be written on an occurrence basis, utilizing standard ISO unmodified coverage form CG 00 01 or its equivalent. Endorsements excluding, restricting, or limiting coverage may be acceptable under certain circumstances provided the same are agreed upon by Owner and Design-Builder.

2.1.1.1 Acceptable professional liability exclusions to the Design-Builder's commercial general liability insurance are limited to ISO endorsements CG 2280 or CG 2279 or their equivalent.

- **2.1.2** General Liability, Automobile Liability, Worker's Compensation/Employers Liability and UmbrellaExcess Liability policies shall each include the following endorsements:
 - **2.1.2.1** Unintentional Errors and Omissions Endorsement
 - **2.1.2.2** Notice of Occurrence Endorsement
 - 2.1.2.3 Knowledge of Occurrence Endorsement
- **2.1.3** Commercial Automobile Liability coverage shall be provided by standard ISO Commercial Automobile or Truckers Policy covering all Owned, Non-Owned and Hired Vehicles.
- **2.1.4** Umbrella/Excess Liability must schedule Commercial General Liability, Automobile/Truckers Liability and Employers Liability as underlying policies. The Umbrella/Excess Liability policies shall bewritten in accordance with the scheduled underlying policies and must be as broad as the underlying policies.

- **2.1.5** Contractors Pollution Liability shall either be written on an occurrence or claims-made basis. Ifwritten on a claims-made basis, the policy must comport to Section 4.1.5.
 - **2.1.5.1** The policy is to provide coverage for off-site transportation by all applicable modes of conveyance. When required, coverage is also to be provided for claims involving materials removed from the site and brought to off-site disposal, treatment and storage facilities.
 - **2.1.5.2** Any restriction, limitation, or exclusion related to Naturally Occurring Substances must be modified so as not to apply to microbial matter and the release of such Naturally Occurring Substances as a result of the performance of Operations.

3.1 Additional Insureds

- **3.1.1** Owner and Owner's officers, directors and employees shall be included as an additional insured ongeneral liability, umbrella/excess and automobile liability policies of insurance of the Design-Builder and its Subcontractors and Design Consultants at any tier. If required, as set forth above, Owner shall also be included as an additional insured on the Design-Builder's Contractor's Pollution Liability policy of insurance. No person shall be named as an additional insured on any professional liability policy or worker's compensation. Any coverage granted to an additional insured shall be primary and that coverage independently carried by an additional insured shall not contribute. Design-Builder shall furnish to Owner a copy of all Certificates of Insurance showing the Owner as additional insured as set forth above as well as Proof of Insurance in the form attached as Exhibit I. Design-Builder shall require Subcontractors and Design Consultants of any tier to furnish such certificates and Proof of Insurance, and upon request of the same will furnish them to the Owner.
- **3.1.2** Each of the policies designated in section 3.1 is to provide a waiver of subrogation in favor of those persons or entities included as additional insureds. A waiver of subrogation is also to be provided tosuch entities under Worker's Compensation/Employer's Liability policies.
- **3.1.3** Additional Insured coverage provided under the Commercial General Liability/Umbrella/Excessand, if applicable, Design-Builder's Contractor's Pollution Liability policies, shall cover both the premises/operations and completed operations hazards.

4.1 Terms and Effective Dates.

- **4.1.1** If the General Liability coverages are provided by a Commercial General Liability Policy on a claims-made basis, the policy date or Retroactive Date shall predate the Agreement. The termination dateof the policy or applicable extended reporting period shall be no earlier than the termination date of coverages required to be maintained after Final Payment is made.
- **4.1.2** If the Contractor's Pollution Policy is made on a claims-made basis, the policy date or RetroactiveDate shall predate the Agreement. The termination date of the policy or applicable extended reporting period shall be no earlier than the termination date of coverages required to be maintained after Final Payment is made.
- **4.1.3** Professional Liability coverage shall be retroactive to the date that professional services first commenced.
- **4.1.4** All Claims-Made Policies must: (a) permit reporting of circumstances that could give rise to a claim; and (b) provide coverage for post-expiration claims resulting from such circumstances.
 - **4.1.5** Any coverage required to be maintained after Final Payment shall be identified below: General Liability, including completed operations coverage

Worker's Compensation

Professional Liability, including Contractor's Protective Liability, if applicable. Umbrella Coverage

Such coverage shall remain in place for six (6) years after Substantial Completion.

EXHIBIT B1 PERFORMANCE BOND FORM HAINES BOROUGH LUTAK DOCK RESTORATION



PERFORMANCE BOND FOR DESIGN-BUILD PROJECTS

This bond form has been endorsed by The National Association of Surety Bond Producers and The Surety & Fidelity Association of America

DESIGN-BUILDER/PRINCIPAL: (Name and address)	SURETY: (Name and contact information)
OWNER/OBLIGEE:	PROJECT:
(Name and address)	(Name and location)
103 Third Ave	Lutek Deek Destavation
Haines. AK 99827	Lutak Dock Restoration
,	
DESIGN-BUILD AGREEMENT:	BOND DATE:
	(Not earlier than date of Design-Build Agreement)
Amount:	

MODIFICATIONS TO THIS BOND:

(List modifications to this Bond below. If none, write "None")

BOND TERMS AND CONDITIONS

1 Binding Effect. The Design-Builder and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors and assigns to the Owner for the performance of the Design-Build Agreement, which is incorporated herein by reference.

2 Intent of Bond. If the Design-Builder performs its obligations under the Design-Build Agreement, then the Surety's obligations under this Bond are null and void, except to participate in meetings as provided in Section 5.

3 Waiver of Notice. The Surety hereby waives notice of changes to the Design-Build Agreement, including changes within the general scope, or of time or price, or to related subcontracts or purchase orders.

4 Owner's Obligations. If there is no default in Owner's obligations under the Design-Build Agreement, then the Surety's obligation under this Bond shall arise after the following steps have been taken by Owner, as a condition precedent to a Bond claim:

4.1 The Owner has first provided written notice to the Design-Builder and Surety at the addresses listed on page 1 of this Bond, that Owner is considering declaring the Design-Builder in default and has requested and attempted to arrange a meeting with the Design-Builder and Surety, to be held not later than fourteen (14) days after receipt of Owner's notice, to discuss methods of performing the Design-Builder's obligations under the Design-Build Agreement. If the Owner, Design-Builder and Surety agree, the Design-Builder shall be allowed a reasonable time to perform its obligations under the Design-Build Agreement, but such an agreement shall not waive the Owner's right, if any, subsequently to declare the Design-Builder in default;

4.2 The Owner declares the Design-Builder to be in default, terminates the Design-Build Agreement and notifies the Surety in writing; and

4.3 The Owner has agreed to pay the balance remaining under the Design-Build Agreement (i.e., the total amount payable by the Owner to the Design-Builder thereunder less amounts properly paid by the Owner to the Design-Builder, the "Contract Balance") to:

.1 The Surety, in accordance with the terms of the Design-Build Agreement; or

.2 Another design-builder selected pursuant to Section 5.3 to perform the remaining obligations under the Design-Build Agreement.

5 Surety's Obligations. When Owner has satisfied the conditions of Section 4, the Surety shall promptly take one of the following actions, at the Surety's expense:

5.1 Arrange for the Design-Builder to perform and complete the remaining obligations under the Design-Build Agreement, with consent of Owner;

5.2 Undertake to perform and complete the remaining obligations under the Design-Build Agreement itself, through its agents or through independent contractors;

5.3 Obtain bids or negotiated proposals from qualified design-builders acceptable to Owner for a contract for performance and completion of the Design-Build Agreement, arrange for a contract to be prepared for execution by Owner and a design-builder selected with Owner's concurrence, to be secured by performance and payment bonds equivalent to those for the Design-Build Agreement, issued by a qualified surety. The Surety shall: a. make available as Work progresses sufficient funds to pay the cost of completion of the Design-Build Agreement; and, b. pay to Owner the amount of damages as described in Section 7;

5.4 Waive its right to complete the Work under Sections 5.2 or 5.3, and reimburse the Owner the amount of its reasonable costs to complete the Work; or

5.5 Deny liability, in whole or in part, and notify the Owner in writing, citing reasons therefor.

6 Owner's Rights. If the Surety does not proceed as provided in Section 5 with reasonable promptness, the Surety shall be deemed to be in default on this Bond seven (7) days after receipt of an additional written notice from the Owner to the Surety demanding that the Surety perform its obligations under this Bond and stating that the Owner shall be entitled to enforce any remedy available to the Owner. If the Surety proceeds as provided in Section 5.4, and the Owner refuses the payment, or the Surety has denied liability, in whole or in part, under Section 5.5, the Owner shall be entitled without further notice to enforce any remedy available to it.

7 Damages Covered. In any event, the Surety's obligations to the Owner, and the Owner's obligations to the Surety, shall not be greater than those of the Owner and Design-Builder to each other, respectively, under the Design-Build Agreement. Subject to commitment by Owner to payment of the Contract Balance, the Surety is obligated without duplication for:

7.1 The responsibilities of Design-Builder for correction of defective Work and completion of the Project;

7.2 Additional legal, design professional and delay costs resulting from Design-Builder's default, and resulting from the actions or failure to act of Surety under Paragraph 5; and

7.3 Liquidated damages, or if no liquidated damages are specified in the Design-Build Agreement, actual damages caused by delayed performance or non-performance of Design-Builder.

8 Bond Liability. The Surety shall not be liable to the Owner or others for obligations of the Design-Builder that are unrelated to the Design-Build Agreement, and the Contract Balance shall not be reduced or set off on account of any such unrelated obligations.

9 Beneficiaries. No right of action shall accrue on this Bond to any person or entity other than Owner or its heirs, executors, administrators, or successors, unless some other party is named in this Bond as a dual obligee.

10 Dispute Resolution. All disputes related to this Bond shall be instituted in any court of competent jurisdiction in the location in which the Project is located and shall be commenced within six (6) years after: a. the Owner declares the Design-Builder in default under Section 4.2; or, b. Substantial Completion of the Project, whichever occurs first. If the provisions of this Section 10 are prohibited by law, the minimum period of limitation available to sureties in the jurisdiction in which the Project is located shall be applicable.

10.1 In the event of bankruptcy of the Design-Builder, the Surety agrees that the Design-Builder is not a necessary or indispensable party to any legal action by Owner against Surety to enforce the Surety's obligations under this Bond.

11 Notice. Unless otherwise noted below, written notice under this Bond to Surety, Owner or Design-Builder shall be mailed or delivered electronically or by hard mail to the contact information shown on page 1.

(List any alternate contact information below for notice to the Surety of any claim on this Bond. If none, then use the contact information on page 1)

For Claims on this Bond: (check appropriate box) Use the contact information shown on page 1; (fill in Surety claims administrator contact information below)

12 Statutory Compliance. If this Bond has been furnished to comply with a statutory requirement in the location where the Project is located, then any provision herein that conflicts with a statutory requirement shall be deemed deleted and replaced by provisions conforming to such statutory requirement. The intent is that this Bond shall be construed as a statutory bond conforming to the applicable statutes.

13 Warranty Obligation. The Surety's obligations to the Owner for warranties of the Design-Builder shall be the same as those required of the Design-Builder under the Design-Build Agreement, subject to the time limitation in Section 10. Unless otherwise stated below, the Surety's obligation for such warranties excludes: a) products, materials or equipment covered by a manufacturer's separate warranty; and b) warranty claims by the Owner first noticed to Surety in writing more than one year after the effective date of such warranty as specified under the Design-Build Agreement. (*List below any exceptions to the above limitations on Surety's warranty obligation, if any*)

14 Authorization. The Surety represents that it is admitted to act as an authorized corporate surety in the state in which the Project is located. Surety and Design-Builder, intending to be legally bound hereby, subject to the terms set out above, do each cause this Performance Bond to be duly executed on its behalf by its authorized officer, agent, or representative.

DESIGN-BUILDER (AS PRINCIPAL) Company:	SURETY Company:
Signature:	Signature:
Name and Title:	Name and Title:
	Corporate Seal
	(Attach Power of Attorney)

(Space is provided below for signatures of additional parties, if required.)

Attest:

Signature and Title

EXHIBIT B2 PAYMENT BOND FORM HAINES BOROUGH LUTAK DOCK RESTORATION



PAYMENT BOND FOR DESIGN-BUILD PROJECTS

This bond form has been endorsed by The National Association of Surety Bond Producers and The Surety & Fidelity Association of America

DESIGN-BUILDER/PRINCIPAL:	SURETY:
(Name and address)	(Name and address)
OWNER/OBLIGEE: (Name and address) Haines Borough 103 Third Ave. Haines, AK 99827	PROJECT: (Name and location) Lutak Dock Restoration
DESIGN-BUILD AGREEMENT:	BOND DATE:
Dated:	(Not earlier than date of Design-Build Agreement)
Amount:	BOND AMOUNT:

MODIFICATIONS TO THIS BOND:

(List modifications to this Bond below. If none, write "None")

BOND TERMS AND CONDITIONS

1 Binding Effect. The Design-Builder and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors and assigns to pay for labor, services, materials and equipment furnished by Claimants for use in the performance of the Design-Build Agreement, which is incorporated herein by reference.

2 Intent of Bond. If the Design-Builder promptly makes payment of all sums for all labor, services, materials, and equipment furnished for use in the performance of the Design-Build Agreement, then the Surety's obligations under this Bond are null and void. Otherwise the Surety's obligations shall remain in full force and effect.

3 Notice of Claim. Every Claimant who has not been paid in full before the expiration of a period of ninety (90) days after such Claimant provided or performed the last of the work, services or labor, or furnished the last of the materials or equipment for which said claim is made, may have a right of action on this Bond.

3.1 Claimants shall provide written notice to the Surety and send a copy, or notice thereof, to Owner and Design-Builder, stating that a claim is being made under this Bond and, with substantial accuracy, the amount of the claim, and the last date such work, services or labor were performed, or the last materials or equipment were furnished in furtherance of the Design-Build Agreement.

3.2 If Claimant does not have a direct contract with Design-Builder, the notice shall identify the person or entity with whom Claimant contracted and who has not made payment to Claimant.

4 Surety's Obligations. When a Claimant has satisfied the conditions of Section 3, the Surety shall promptly take the following actions at the Surety's expense:

4.1 Send an answer to that Claimant, with a copy to the Owner and Design-Builder, within sixty (60) days after receipt of the claim, stating the amounts that are undisputed and the basis for challenging any disputed portions or amounts.

4.2 Pay or arrange for payment of any undisputed amounts.

5 Bond Liability. If the Surety fails to discharge its obligations under Sections 4.1 or 4.2, the Surety shall indemnify the Claimant for the reasonable attorney's fees the Claimant incurs thereafter to successfully recover any sums found to be due and owing to the Claimant. If Claimant does not recover the entire amount claimed in its notice under Section 3, then such attorney's fees shall be reduced in proportion to the amount actually recovered.

5.1 The Surety shall not be liable to the Owner, Claimants or others for obligations of the Design-Builder that are unrelated to the Design-Build Agreement, and the Contract Balance shall not be reduced or set off on account of any such unrelated obligations.

6 Waiver of Notice. The Surety hereby waives notice of changes to the Design-Build Agreement, including changes within the general scope, or of time or price, or to related subcontracts or purchase orders.

7 Dispute Resolution. No suit or action shall be commenced by a Claimant under this Bond other than in a court of competent jurisdiction in the State in which the Project is located. Such suit or action must be filed within one (1) year from the date of final settlement of the Design-Build Agreement .If the provisions of this Section 7 are prohibited by law, the minimum period of limitation available to sureties in the jurisdiction in which the Project is located shall be applicable.

7.1 In the event of bankruptcy of the Design-Builder, the Surety agrees that the Design-Builder is not a necessary or indispensable party to any legal action by any party against the Surety to enforce the Surety's obligations under this Bond.

8 Statutory Compliance. If this Bond has been furnished to comply with a statutory requirement in the location where the Project is located, then any provision herein that conflicts with a statutory requirement shall be deemed deleted and replaced by provisions conforming to such statutory requirement. The intent is that this Bond shall be construed as a statutory bond conforming to the applicable statutes.

9 Copy To Be Furnished. Upon written request of any person or entity appearing to be a potential Claimant on this Bond, Design-Builder shall promptly furnish a copy of this Bond or shall permit a copy to be made.

10 Claimant Defined. A Claimant is any individual or entity having a direct contract with the Design-Builder or having a contract with a subcontractor that has a direct contract with the Design-Builder to furnish services, labor, materials or equipment for use in the performance of the Design-Build Agreement.

10.1 A Claimant may include amounts owed by the Design-Builder for design and other professional services furnished or performed by Claimant regardless of whether such services might form the basis for a mechanic's lien under applicable State law.

11 Notice. Unless otherwise noted below, written notice under this Bond to Surety, Owner or Design-Builder shall be mailed or delivered electronically or by hard mail to the contact information shown on page 1.

(List any alternate contact information below for notice to the Surety of any claim on this Bond. If none, then use the contact information on page 1)

For Claims on this Bond: (check appropriate box) □ Use the contact information shown on page 1; or □ Use the following alternate contact information: (fill in Surety claims administrator contact information below)

12 Subcontractor Bonds. If this Bond is issued for an agreement between the Design-Builder and a subcontractor, the term Design-Builder in this Bond shall be deemed to be the bonded subcontractor and the term Owner shall be deemed to be Design-Builder.

13 Authorization. The Surety represents that it is admitted to act as an authorized corporate surety in the state in which the Project is located. Surety and Design-Builder, intending to be legally bound hereby, subject to the terms set out above, do each cause this Payment Bond to be duly executed on its behalf by its authorized officer, agent, or representative.

DESIGN-BUILDER (AS PRINCIPAL) Company:	SURETY Company:
Signature:	Signature:
Name and Title:	Name and Title:
	Corporate Seal
	(Attach Power of Attorney)

(Space is provided below for signatures of additional parties, if required.)

Attest:

Signature and Title

EXHIBIT C PHASE 1 AND 2 SCOPE OF WORK HAINES BOROUGH, AK LUTAK DOCK REPLACEMENT

PART 1 PHASE 1 PROGRAM VALIDATION PERIOD SCOPE OF SERVICES

- 1.01 SUMMARY OF WORK
 - A. This Section sets forth the Scope of Work, the Deliverables, and the execution activities for Phase 1.

1.02 OWNER'S PROJECT GOALS

The Owner has established the following Project Goals for the Project. The Parties agree to work in good faith to meet and/or exceed the Project Goals:

- A. **Design and Construct a Dock that Maximizes the Program Requirements within the Limited Budget.** The Design-Build Team will leverage the efficiencies of the progressive design-build process through innovative and lean design and construction techniques that provide an efficient and effective design with the most scope and programming within the Owner's established budget. The design will also optimize efficiency of operations and reduce long term maintenance.
- B. **Execute a successful, collaborative Progressive Design-Build (PDB) Process to produce the envisioned project:** The Design-Build team will develop and utilize a collaborative relationship between the Owner, its stakeholders, and the Design-Build Team to exceed the Project Goals within the Owner's budget and schedule and demonstrating exemplary design and project management. The Design-Build Team will work with existing port users to minimize disruptions and to ensure the facially will function at a high level of service and efficiency when complete.
- C. **Efficient Pricing and Schedule**. The Design-Build Team will provide transparent pricing and scheduling that allows the Owner to track design and construction concurrently as well as fast track design and construction to maximize the Owner's budget within the Project Schedule.
- D. **Comply with Legal Requirements.** The Design-Build Team will understand and comply with all applicable State and Federal Legal Requirements.
- E. **Design for Safety**. The Design-Build Team will create a design that enhances the safety of the project. The design and construction process will reduce re-work and interference with operations with a goal of no recordable incidents.
- 1.03 PHASE 1 SCOPE
 - A. Design Builder shall review, analyze, and validate the Initial Basis of Design Documents, the project budget, the Project Schedule, the Commercial Terms and any other information provided by the Owner, collectively referred to as "Owner Provided Information".

- B. Design Builder shall conduct such site investigations, environmental assessments, review of regulatory and legal authority and restrictions, and assess other information as reasonably necessary to verify and validate the Owner Provided Information.
- C. Design Builder shall review, analyze and validate the concepts for the Project elements as shown in the Initial Basis of Design Documents. In addition, Design Builder shall work collaboratively with the Owner and the Stakeholders to examine whether new concepts will better maximize the Owner's Project Goals, and if approved by the Owner, further develop such new concepts and incorporate them into the Project.
- D. Not used.
- E. Design Builder shall engage and work collaboratively with the Owner and the Project Stakeholders to obtain input regarding the Project design and functionality, as well as other major Project elements and to develop the Final Basis of Design Documents.
- F. Design Builder shall engage and work collaboratively with the Owner and the Project Stakeholders to progress the design to a sufficient state to develop the Final Basis of Design Documents, the Project Schedule, and the Guaranteed Maximum Price within the project budget. The timing of the GMP Proposal and the percentage complete of the designs and specifications will be jointly determined by the Owner and the Design-Builder.
- G. Design Builder shall provide the Deliverables during Phase 1 as set forth in Sections 2.01.A. and B. herein. Deliverables shall be provided in a format acceptable to the Owner.
- H. At the conclusion of Phase 1, Design Builder shall prepare a GMP Proposal, including any modifications and/or clarifications to the Initial Basis of Design Documents as set forth in Section 2.01.B herein.
- 1.04 VALIDATION OF INFORMATION.
 - A. During Phase 1, Design Builder shall perform such assessments, reviews and investigations of the Owner Provided Information, as determined by Design Builder to be reasonably necessary to validate the Owner Provided Information as well as investigate any other information required to accomplish the Project, including but not limited to the information below. Additional reviews, assessments and investigations of Owner Provided Information shall include, if reasonably necessary, the following:
 - 1. Verification that the As-Built drawings (if applicable) and other architectural and engineering drawings, plans and specifications are correct,

- 2. Constructability, including proposed methods of construction, of the proposed structures in the Initial Basis of Design Documents,
- 3. Verification of the architectural, engineering and other assumptions and calculations (if any) in any Owner Provided Information,
- 4. Examination and verification of actual site conditions as set forth below,
- 5. Verification of any surveys,
- 6. Review and assessment of all applicable legal and regulatory rules and restrictions on the Project, including consultation with permit authorities regarding their requirements,
- 7. Verification and validation of assumptions regarding the establishment of the Commercial Terms, including but not limited to the GMP, the Project Schedule, and the Initial and Final Basis of Design Documents.
- B. Design Builder shall perform site investigations as necessary for Design Builder to verify the Owner Provided Information and to validate the Commercial Terms and the Initial Basis of Design Documents. Design Builder shall visit the Site and examine thoroughly and understand the nature and extent of the Work, site, locality, actual conditions, as-built conditions, and all local conditions and federal, state, and local laws and regulations that in any manner may affect cost, time, progress, performance or furnishing of the Work or which relate to any aspect of the design and the means, methods, techniques, sequences or procedures of construction to be employed by Design Builder and safety precautions and programs incident thereto. Such additional investigations shall be conducted to sufficiently identify or characterize utility locations (underground and overhead), site conditions, contaminated materials, and observable or concealed conditions in the existing facilities, including but not limited to the following:
 - 1. Undertake surveys, investigations and analysis to provide necessary data and information for project design including sufficient information to evaluate design alternatives.
 - 2. Complete a comprehensive archaeological site survey and conduct a literature and data search to determine potentially archaeologically significant sites and conditions.
 - 3. Perform Geotechnical soils sampling, testing, and analysis as necessary data and information for Project design At a minimum, test for contamination in areas to be excavated.
 - 4. Subsurface investigation work, including the disturbance of existing vegetation, cannot proceed until all required permits have been obtained.

- C. Design Builder will conduct or obtain and understand all such examinations, investigations, explorations, tests, reports and studies, in addition to or to supplement those referred to above, that pertain to the subsurface conditions, as-built conditions, underground facilities and all other physical conditions at or contiguous to the Site or otherwise that may affect the cost, time, progress, performance or furnishing of Work, as Design Builder considers necessary for the performance or furnishing of Work for the Commercial Terms and in accordance with the Initial Basis of Design Documents as well as other terms and conditions of the Contract Documents, and no additional examinations, investigations, explorations, tests, reports, studies or similar information or data are or will be required from the Owner by Design Builder for such purposes.
- D. All reports or analyses generated by Design Builder's research, testing, inspections, and investigations, including but not limited to geotechnical evaluations and hazardous materials studies, archaeological site surveys, hazardous materials investigations, etc., shall be provided to the Owner promptly, within seven (7) business days, after such reports are analyzed and generated.
- E. Design Builder shall be responsible for ensuring that its design documents and construction work accurately conforms to, and interfaces with, the existing conditions and shall not request a change or claim for unforeseen or concealed conditions except as provided under the provisions of the Contract Documents.
- F. The Design Builder shall work with the Owner to determine if additional examinations, investigations, explorations, tests, reports, studies or similar may be required after partial or complete demolition of the existing stations. This work shall be completed by the Design Builder and included in the GMP.

1.05 DEVELOPMENT OF FINAL BASIS OF DESIGN DOCUMENTS

- A. Design Builder shall manage the design process in a collaborative, efficient, transparent and coordinated manner and conduct design workshops as required by the Contract Documents. The Final Basis of Design Documents will establish the scope of the Work and provide the basis for the GMP. The Final Basis of Design Documents must be consistent with the Initial Basis of Design Documents, unless the Owner has consented to modify its requirements in writing through a Change Order, Field Directive, or other written means allowed by the Contract Documents.
- B. Design Builder shall provide for an orderly and timely approval process by the Owner and third parties, document review comments from the Owner and third parties, and take appropriate action.
- C. The Owner will review and comment on the Design Submissions in a timely fashion. The Design-Builder will allow adequate time for the Owner to review the Design Submissions, which shall not be less than 10 business days.

- D. Design Builder shall submit a written response to the Owner's design review comments, describing the action taken for each comment. Design Builder shall, in a timely fashion, bring to the attention of the Owner areas where new technologies, such as BIM or Design-Build processes, may require modifications to these requirements.
- E. By submitting Design Submissions, Design Builder represents to the Owner that the Design Submissions may be designed and constructed for the then current Commercial Terms and in accordance with the Initial Basis of Design Documents, the Design Log, and any changes made thereto. Notwithstanding the above, Design Builder may propose Designs, Plans or other Submissions that may alter a Commercial Term or the Initial Basis of Design Documents; however, with any such Design Submissions, Design Builder must provide notice pursuant to Article 10 of the General Conditions.

1.06 DEVELOPMENT OF GMP PRICING

- A. The forecasting and development of accurate project cost estimates throughout each phase of the Project is vital to the Owner's financial management strategy. The Owner relies on the Design Builder to provide and validate current and detailed cost estimates and forecasts that will be incorporated into the overall cost controls for the Owner.
- B. Throughout the Project, Design Builder will update estimates and forecasts and provide data to the Owner to reflect real time information. Design Builder will provide all pricing, estimates and other data used to develop the Commercial Terms on an open and transparent basis. The project controls system used by the Design Builder shall be acceptable to the Owner and will be capable of being broken down and reported in a number of different work breakdown structures, including but not limited to organizing the financial data by cost element codes, subcontracts, vendors, Construction Document packages, etc.
- C. The Design Builder will coordinate the development of the GMP pricing with the development of the Final Basis of Design Documents as well as the Project Schedule so that the Owner may obtain an accurate understanding of the GMP. The GMP set forth it the Agreement shall not be exceeded without a written Change Order.
- 1.07 DEVELOPMENT OF PROJECT SCHEDULE
 - A. The forecasting and development of the Project Schedule, including but not limited to the project phasing and Schedule of Values, is a vital element of the Design Builder's ability to deliver this Project in a timely fashion. The Owner will rely on the Design Builder's scheduling information to coordinate with its Stakeholders, schedule activities in and around the Project, and manage the dock facilities.
 - B. Design Builder shall provide the Owner with frequent updates to the project schedule in a format acceptable to the Owner.

PART 2 PHASE 1 DELIVERABLES

2.01 SUBMITTALS

- A. *Submittals After Phase 1 Notice to Proceed:* Design Builder shall provide the following Submittals within 10 days after the Notice to Proceed with Phase 1, unless otherwise noted in Phase 1 Schedule.
 - 1. Phase 1 Schedule pursuant to Section 2.02.A.
 - 2. Preliminary Schedule of Values for the GMP pursuant to Section 2.04.A.1
 - 3. Preliminary Cost Model pursuant to Section 2.04.B.5
 - 4. Subcontractor Procurement Procedure pursuant to Section 2.05.A
 - 5. Project Safety and Job Hazard Analysis pursuant to Section 2.06.A.
- B. *Submittals During Phase 1:* Design Builder shall provide the following submittals during Phase 1.
 - 1. Within 2 weeks of the Notice to Proceed for Phase 1:
 - a. Preliminary Project Schedule pursuant to Section 2.02.B
 - 2. On a monthly basis:
 - a. Updates to the Phase 1 Schedule, Schedule of Values, Project Schedule pursuant to Section 2.02. A and B
 - b. Design Submissions Packages pursuant to Section 2.03.A.
 - c. Preliminary estimating information pursuant to Section 2.04.A and updates to the Cost Model Pursuant to Section 2.04.B.
- C. GMP Proposal: At the conclusion of Phase 1, Design Builder shall provide a GMP Proposal that includes the following Deliverables.
 - 1. GMP Pricing and Verification of GMP pursuant to Section 2.04.C.
 - 2. Final Basis of Design Documents pursuant to Section 2.03.B.
 - 3. Project Schedule pursuant to Section 2.02.D.
 - 4. Schedule of Values and Cost Model pursuant to Section 2.04.B.
 - 5. A list of the assumptions and clarifications made by the Design-Builder in preparation of the GMP Proposal.
 - 6. Project Safety and Job Site Hazard Analysis pursuant to Section 2.06.B.
 - 7. Project Phasing/Staging Analysis pursuant to 2.07.
 - 8. Permitting Strategy Plan pursuant to Section 2.08.
 - 9. QA/QC Plans pursuant to 2.9.

- 10. Contract Close-Out Plan pursuant to Section 2.10.
- 11. Differing Site Conditions Report pursuant to Section 2.11.
- 2.02 SCHEDULES
 - A. Phase 1 Schedule. By the date set forth in Section 2.01A herein, Design Builder shall provide a Phase 1 Schedule.
 - 1. Phase 1 Schedule shall show the activities of the Owner and Design Builder necessary to meet Phase 1 requirements.
 - 2. Phase 1 Schedule shall be updated periodically with the level of detail for each schedule update reflecting the information then available.
 - 3. If an update to Phase 1 Schedule indicates that a previously approved milestone will not be met, Design Builder shall submit a corrective action plan and recovery schedule to the Owner pursuant to the Contract Documents.
 - B. Preliminary Project Schedule. By the date set forth in Section 2.01.B, Design Builder shall submit a Preliminary Project Schedule that reflects Design Builder's sequence of design, procurement and construction activities including the interrelationships of the Demolition and Construction Packages.
 - 1. The Preliminary Schedule shall show the activities of the Owner and Design Builder necessary to meet the Project completion requirements.
 - 2. The Preliminary Schedule shall be updated periodically monthly with the level of detail for each schedule update reflecting the information then available.
 - 3. If an update to the Preliminary Schedule indicates that a previously approved milestone will not be met, Design Builder shall submit a corrective action plan and recovery schedule to the Owner pursuant to the Contract Documents.
 - C. Design Builder shall meet with the Owner to review the Preliminary Schedule and updates. In the event that the Owner has any comments relative to the Preliminary Schedule or Schedule Updates or finds any inconsistencies or inaccuracies in the information presented, it shall give prompt written notice of such comments or findings to Design Builder, who shall make appropriate adjustments to the Preliminary Schedule, its basis, or both. The parties will work collaboratively to make adjustments in the Final Basis of Design Document, the Project Schedule, or GMP to fit within the Owner's objectives.
 - D. With the GMP Proposal, Design Builder shall provide a Project Schedule that will incorporate the Preliminary Schedule developed collaboratively during Phase 1 along with any updates to the schedule.

- E. All schedules must be in the format of a Critical Path Method (CPM) Resource loaded schedule as set forth below.
- F. Critical Path Method (CPM) Resource loaded schedule
 - 1. The CPM Schedule will contain the following
 - a. All tasks required to complete the scope of work for the project.
 - b. Durations for all tasks in the project schedule.
 - c. Logical ties and sequence of work for every task in the schedule.
 - d. Resources for project hours and major material quantities for site construction.
 - 2. Project Schedule shall be detailed and organized according to pre-defined Design-Builder's WBS that is developed in the Scope Management Plan. The project schedule will include all activities and relationships identified in the Design-Builder's Scope of Work Narrative. Each major area of work within Design-Builder's scope shall be represented by activities in the schedule.
 - 3. Design-Builder shall prepare a detailed resource loaded CPM Project Schedule in accordance with this specification. The schedule shall be submitted to the Owner for their review. The detailed schedule shall reflect, at a minimum, engineering, procurement, construction, fabrication, and delivery activities for each piece of procured equipment, key drawing release dates by discipline, and logic and interrelationships between activities so that a logical progression of the work is depicted. Project Milestones shall also be included in schedule.
 - 4. Design-Builder and subcontractors shall meet with the Owner to review and approve the detailed CPM baseline Project Schedule.
 - 5. Once the detailed project schedule has been approved by the Owner, Design-Builder will establish a baseline schedule. Thereafter Design-Builder shall advise the Owner of any proposed Critical Path Schedule changes and promptly provide the Owner with any revisions thereto and recovery plans as required to meet the contractual dates.
 - 6. Schedule Validity and Content
 - a. Prepare schedules in a format acceptable to the Owner.
 - b. Contain Work Breakdown Structure coding matching deliverables and work packages.
 - c. Schedule will reflect all deliverables and tasks mention in the Scope of Work narrative.

- d. Schedules shall be coded for grouping by engineering, procurement, construction, and commissioning
- e. Project schedule activities that Design-Builder is responsible for performing shall be resource loaded with engineering and procurement activities. Construction activities shall be resourced loaded 60 days prior to site mobilization.
- f. Resource loading for project hours and major material quantities for site construction.
- g. Engineering, procurement and construction activities shall be included, such that Project staffing requirements can be determined or verified with schedule. The original resource-loaded construction schedule shall form basis for progress reporting, and payment.
- h. Critical path for Design-Builder's schedule activities.
- 2.03 DESIGN DOCUMENTS
 - A. Preliminary Design Submissions. As Design Builder develops the Final Basis of Design Documents, Design Builder shall collaborate with the Owner to submit and review the Preliminary Design Submissions that will be incorporated into the Final Basis of Design Documents. The Preliminary Design Submissions will be submitted pursuant to the Schedule provided by the Design-Builder and approved by the Owner.
 - 1. Design Builder shall coordinate with the Owner to determine the schedule for submission of preliminary Design Submissions Packages to review collaboratively with the Owner. Design Builder shall schedule the review of the Design Submissions Packages such that the review of each package submitted is of reasonable scope for prompt and thorough review by the Owner.
 - 2. The parties will work collaboratively to make adjustments in the Design Submissions and in the proposed Final Basis of Design Documents to fit within the Owner's Project Goals.
 - B. The Owner and Design Builder shall work collaboratively to develop the Final Basis of Design Documents provided as part of the GMP Proposal. The Final Basis of Design Documents submitted with the GMP Proposal shall include at a minimum the following documents and set forth the assumptions and clarifications on which the GMP and Project Schedule are based.
 - 1. Project Manual, which shall set forth both the general objectives for the Owner, as well as specific uses for each of the project elements set forth in the Initial Basis of Design Documents.
 - 2. Unless the parties agree in writing otherwise, Design Builder will, in addition to periodic design submissions, provide the following Milestone Design Deliverables to the Owner for submission to the
Haines Planning Commission and approval by the Borough:

- a. 35% Design
- b. 65% Design
- c. 95% Design
- d. Construction Documents
- 3. The Milestone Deliverables shall include major building elements and components, such as curtain walls, and finishes and shall include, but not be limited to the following.
 - a. Plan and profile drawings
 - b. Structural renderings
 - c. Structural details
 - d. Bill of materials
 - e. Material specifications
 - f. Permitting and environmental compliance narrative
 - g. Geotechnical letter report
- 4. Design-Builder must have written approval from the Owner to proceed with the project after submission of each of the Milestone Design Deliverables set forth above.
- 5. Design Builder shall schedule the review of the Construction Packages such that the review of each package submitted is of reasonable scope for prompt and thorough review by the Owner.
- 6. Design Builder shall highlight any material differences and developments between the Initial Basis of Design Documents, any Design Submissions, and the Final Basis of Design Documents as the Final Basis of Design Documents are being developed.
- 7. In the event that the Owner has any comments relative to the Design Submissions or finds any inconsistencies from the Initial Basis of Design Documents or discovers inaccuracies in the Design Submissions, the Owner shall give prompt written notice of such comments or findings to Design Builder, who shall make appropriate adjustments to the proposed Final Basis of Design Documents.
- 8. The parties will work collaboratively to make adjustments in the Design Submissions and in the proposed Final Basis of Design Documents to fit within the Owner's Project Goals as well as the GMP.

- 9. Performance Specifications, which shall set forth the specific requirements for the project and identification of each major system, including but not limited to the following:
 - a. Live load, seismic, and vessel mooring requirements of the dock structure.
 - b. Geotechnical report

- 2.04
- 2.05 GMP PRICING
 - A. Preliminary Schedule of Values
 - 1. Preliminary Schedule of Values. Within the date set forth in Section 2.01.A, Design Builder shall submit a preliminary Schedule of Values for the Project in such a form and supported by such data to substantiate its accuracy in reflecting the breakdown for administrative and payment purposes as the Owner may require. The Schedule of Values shall be further organized to conform to the Construction Specifications Institute (CSI) standard format for divisions and sections.
 - 2. With the submission of Design Submissions Packages, Design Builder shall provide preliminary estimates of costs associated with the Design Submissions in a format acceptable to the Owner that will be incorporated into the GMP.
 - 3. The preliminary estimates shall be provided on a bi-weekly basis and shall be updated with new information as Design Builder develops and finalizes the GMP.
 - B. Schedule of Values and Cost Model
 - 1. Schedule of Values. On the schedule established in Section 2.01.B and with the GMP Proposal, Design Builder shall provide an updated Schedule of Values for the Work with actual start and/or finish dates and percentages complete. Updates shall compare the planned progress from baseline schedule with actual progress from the current schedule. The Schedule of Values shall be in conformance with the requirements below and in such a form and supported by such data to substantiate its accuracy in reflecting the breakdown for administrative and payment purposes as the Owner may reasonably require. The Schedule of Values shall be further organized to conform to the Construction Specifications Institute (CSI) standard format for divisions and sections.
 - 2. The sum of all values listed in schedule shall equal the project budget and, when established, the GMP.
 - 3. Schedule of Values Form and Content
 - a. Schedule of Values will be in a form acceptable to the Owner.

- b. Title of Project and location.
- c. Project number.
- d. Name and Address of Design-Builder.
- e. Date of submission.
- f. Schedule of Values columns will contain at a minimum the following information
 - (1) Line Item # Corresponding back to the WBS and CPM Schedule
 - (2) Line Item Description
 - (3) Budgetary Cost
 - (4) Current Period % Complete
 - (5) Current Period Cost
 - (6) Job to Date (JTD) % Complete
 - (7) JTD Cost
 - (8) Variance Column Representing Budgetary Cost Minus JTD Cost
- 4. Cost Model. Within the time frame set forth in Section 2.01.A.3, Design Builder shall provide a Cost Model, for the Owner's review and acceptance.
 - a. The Cost Model shall, at a minimum, provide the following information:
 - (1) List for all Design and Construction Packages, organized by CSI;
 - (2) Estimated base bid amounts for all Construction Packages;
 - (3) Construction Package Allowances.
 - b. Design Builder shall utilize a project controls management system (PCMS) that will be reviewed for acceptance to the Owner.
 - c. Estimates and forecasts within the Cost Model will need to have the capability to be broken down and reported on in many different formats. These formats may include organizing the estimate by different projects, project funding types, Owner cost element codes, contracts, vendors, Construction Package Sets, Construction Packages, etc. Design-Builder shall collaborate with the Owner to determine the appropriate Work Breakdown Structure that will be used for the development of the Cost Model and all Project cost estimates.

- d. In developing its Construction Package Plan, Design-Builder shall coordinate with the Owner to determine a packaging strategy deemed advantageous to all parties. The agreed-upon packaging strategy will be incorporated into the Cost Model and Project schedule.
- e. On the schedule set forth in Section 2.01.B.2 and with the GMP Proposal, Design-Builder shall update estimates and forecasts as data becomes available to reflect real time information. The Owner will rely on this real-time information for accuracy of overall Owner cost forecasts across all Owner projects.
- 5. Work Breakdown Structure (WBS)
 - a. The Work Breakdown Structure (WBS) is a task-oriented division of work necessary to engineer, procure, and construct the Project. It categorizes successively smaller tasks, in order to achieve scope, schedule, and budget control at the most practical level.
 - b. Design-Builder will develop a WBS structure at the completion of Phase 1. Design-Builder will work with the Owner to develop a mutual compatible WBS system to satisfy the intent of the project. The WBS structure will represent the Design-Builder's entire scope for the project, broken down into manageable deliverables or work packages.
- 6. Scope of Work Narrative
 - a. Design-Builder will develop, from the Work Breakdown Structure, a Scope of Work Narrative for the project before Phase 2 is approved. This document will provide a description of the work to be done for each WBS work package. This document will identify the Design-Builder's general understanding of the project, as well as, provide a description of the work that will be done, and deliverables that will be produced for work packages in the WBS. A narrative for each work package will include, but is not limited to the following:
 - b. Narrative of work to be performed
 - c. List of major deliverables
- C. GMP.
 - 1. With the GMP Proposal, Design Builder shall prepare and submit the GMP Pricing to the Owner, in a format acceptable to the Owner, reflecting Design Builder's total cost for the Project on an open book basis. The GMP in the GMP Proposal shall include:
 - a. Design Builder's Lump Sum Fee as defined in Section
 6.2.3 of the Agreement.

- b. The Cost of the Work as defined in Section 6.3 of the Agreement
- c. The Lump Sum General Conditions Costs as defined in Section 6.4.5 of the Agreement.
- d. If applicable, any Allowance established by the Parties pursuant to Section 6.4.1 of the Agreement
- e. Design Builder's Contingencies established pursuant to Section 6.4.4 of the Agreement.
- 2. In support of the proposed GMP, Design Builder shall provide:
 - a. A list of Not to Exceed Amounts and the information required in Section 6.4.2 of the Agreement
 - b. A list of Lump Sums and the information required in Section 6.4.3 of the Agreement
 - c. A list of the assumptions and clarifications made by Design Builder in the preparation of the GMP to supplement the information contained in the Final Basis of Design Documents.
 - d. All material changes from the Initial Basis of Design Documents and Design Builder's Proposal and the costs associated with such changes.
- D. Design Builder shall meet with the Owner to review the proposed GMP. In the event that the Owner has any comments relative to the proposed GMP or finds any inconsistencies or inaccuracies in the information presented, it shall give prompt written notice of such comments or findings to Design Builder, who shall make appropriate adjustments to the proposed GMP, its basis, or both. The parties will work collaboratively to make adjustments in the Final Basis of Design Documents, Project Schedule, or GMP to meet the Owner's objectives.

2.06 SUBCONTRACTOR PROCUREMENT PROCEDURE

- A. By the date set forth in Section 2.01.A herein, Design Builder shall submit for approval the Subcontractor Procurement Procedure as required in Section 2.8 of the General Conditions.
- 2.07 PROJECT SAFETY AND JOB SITE HAZARD ANALYSIS
 - A. By the date set forth in Section 2.01A herein, Design Builder shall submit a Project Safety and Job Site Hazard Analysis for the activities associated with Phase 1.
 - B. With GMP Proposal, Design Builder shall submit a Project Safety Plan with Job Site Hazard Analyses addressing all phases of the project after Phase 1.
 - C. No field investigation or construction activities will be authorized without acceptance of safety plans as required for the Work.

2.08 PROJECT PHASING /STAGING ANALYSIS

A. With the GMP Proposal, Design Builder shall provide a Project Phasing/Staging analysis for all Demolition and Construction Packages that includes detailed plans for the phasing of the following elements of the Project, including but not limited to all modifications and all other construction activities including the staging of construction materials and facilities.

2.09 PERMITTING STRATEGY PLAN:

- A. With the GMP Proposal, Design Builder shall provide a Permitting Strategy Plan detailing the process for obtaining the building and site development permits for various phases of the project.
- B. During Phase 1, Design Builder must meet with the applicable building officials and develop processes and time lines for plan check approvals.
- C. Design Builder shall coordinate with all authorities with jurisdiction over the Project for the approval of environmental mitigation measures.
- 2.10 QA/QC PLANS
 - A. Prepare a Quality Management Plan (QMP) in accordance with the Contract requirements and submit it with the GMP Proposal.
 - B. Design Quality Management Plan.
 - 1. Design Quality Management Plan (DQMP): shall be developed in accordance with the requirements outlined in the Contract.
 - 2. Design Quality Audits: Design Quality Assurance Manager shall audit all design packages for compliance with the requirements outlined in the DQMP.
 - 3. Independent Technical Reviews: The Design Quality Assurance Manager will appoint appropriate technical staff to conduct Independent Technical Reviews of each design package. These reviews will occur concurrently with the Inter- Disciplinary Reviews and Constructability Reviews.
 - C. Construction Quality Management Plan.
 - 1. Construction Quality Management Plan (CQMP): shall be developed in accordance with the requirements outlined in the Contract.

2.11 CONTRACT CLOSEOUT PLAN

A. With the GMP Proposal, Design Builder shall provide a Project Closeout Plan that integrates all aspects of project closeout proactively over the life of the project. The Closeout Plan will be a living document that will grow and expand as the design and construction progress. The Project Closeout Plan should include, but not be limited to mechanisms and procedures for:

- 1. Closeout provisions included in subcontract procurement documents
- 2. Phased completions and early subcontract closeouts
- 3. Commissioning
- 4. Warranties
- 5. Training
- 6. O&M Documentation
- 7. Record Documents
- 8. Cost Reconciliations
- 9. Permit and Regulatory Requirements
- 2.12 DIFFERING SITE CONDITIONS REPORT
 - A. With the GMP Proposal, Design Builder shall provide a report of all Differing Site Conditions as defined in Section 4.2 of the General Conditions of the Contract that are discovered during Phase 1.
 - B. The Differing Site Conditions Report shall include the following information for each of the identified Differing Site Conditions identified in the Report.
 - 1. The location of the Differing Site Condition;
 - 2. A description of the Differing Site Condition that explains why it qualifies as a Differing Site Condition pursuant to Section 4.2 of the General Conditions;
 - 3. The date the Differing Site Condition was discovered;
 - 4. The impact of the Differing Site Condition on the Initial Basis of Design Documents, the Final Basis of Design Documents, and/or any Commercial Term, as applicable.

Phase 2 Scope of Services

PART 3 PHASE 2

Unless the parties agree otherwise in writing, this Section sets forth the Scope of Work, the Deliverables, and the execution activities for Phase 2.

- 3.01 PHASE 2 SCOPE
 - A. Design Builder shall complete the design and construction services as set forth in the GMP Amendment.
 - B. Design Builder shall provide the deliverables set forth in this Attachment during the course of Phase 2. Deliverables shall be provided in a format acceptable to the Owner and consistent with the requirements for Phase 1.
- 3.02 COMPLETION OF DESIGN
 - A. Design Builder shall provide for an orderly and timely approval process by the Owner and third parties, document review comments from the Owner and third parties, and take appropriate action.
 - B. The Owner will review and comment on the Construction Documents and other Design Submissions in a timely fashion.
 - C. Design Builder shall submit a written response to the Owner's design review comments, describing the action taken for each comment. Design Builder shall, in a timely fashion, bring to the attention of the Owner areas where new technologies, such as BIM or Design-Build processes, may require modifications to these requirements.
 - D. By submitting Design Submissions, including but not limited to the Construction Documents, Design Builder represents to the Owner that the Construction Documents may be constructed for the then current Commercial Terms and in accordance with the Initial Basis of Design Documents and the Final Basis of Design Documents. Notwithstanding the above, Design Builder may propose Designs, Plans or other Submissions that may alter a Commercial Term or the Initial Basis of Design Documents; however, with any such Design Submissions, Design Builder must provide notice pursuant to Article 10 of the General Conditions. The Construction Documents must be consistent with the Final Basis of Design Documents, approved Design Submissions and the Design Log, unless the Owner has consented to modify its Requirements in writing through a Change Order, Field Directive, or other written means allowed by the Contract Documents.

3.03 SCHEDULE OF VALUES AND COST MODEL

A. The forecasting and development of accurate project cost estimates throughout each phase of the Project is vital to the Owner's financial management strategy. The Owner relies on the Design Builder to provide

and validate current and detailed cost estimates and forecasts that will be incorporated into the overall cost controls for the Owner.

- B. Unless modified by the parties in writing, on the schedule set forth in Section 2.01.B, Design Builder will continue to update estimates and forecasts in the format required above and provide data to the Owner to reflect real time information. Design Builder will provide all pricing, estimates and other data used to develop the Commercial Terms on an open and transparent basis.
- C. The Schedule of Values and Cost Model must be consistent with the GMP Amendment and the format required above, unless the parties have agreed on a Change to the terms set forth in the GMP Amendment pursuant to Article 10 of the General Conditions.
- 3.04 PROJECT SCHEDULE
 - A. The forecasting and development of the project schedule, including but not limited to the project phasing and Schedule of Values, is a vital element of the Design Builder's ability to deliver this Project in a timely fashion. The Owner will rely on the Design Builder's scheduling information to coordinate with its Stakeholders, schedule activities in and around the Project, and manage its dock facilities.
 - B. Design Builder shall provide the Owner with updates to the project schedule on the schedule set forth in Section 2.01.B and in the format required above for a scheduled completion within the GMP established in the GMP Amendment.
- 3.05 CONSTRUCTION SERVICES
 - A. Design-Builder shall provide Construction Services and complete the construction of the Project pursuant to the Contract Documents.
- 3.06 COMMISSIONING, TESTING AND CLOSEOUT
 - A. Design Builder shall provide commissioning, testing, and closeout of the Project pursuant to the Contract Documents.

PHASE 2 DELIVERABLES

- 3.07 DELIVERABLES
 - A. Design Builder shall provide the following Milestone Design Deliverable pursuant to the Project Schedule:
 - 1. 100% Construction Documents for review and approval by the Owner.
 - 2. Design Builder shall not proceed with the project after submission of the 100% Construction Documents until it receives the Owner's written approval.
 - B. Design Builder shall provide such other deliverables as set forth in the Contract Documents to successfully complete the Project.

EXHIBIT D OWNER'S PROGRAM HAINES BOROUGH LUTAK DOCK REPLACEMENT

The Owner's Program consists of the following :

The Owner's intent is to restore the existing Lutak Dock face and slightly increase the foot print of the Lutak Dock area. The Owner desires the contractor to design a facility that maximizes the uses, expands the current footprint, and restores the facility as originally designed. The design and parameters for the Lutak Dock Restoration Project are that the design will rebuild the existing Lutak Dock to include a new O Pile retaining wall, leaving existing cells in place, and tied back to the existing fill. This will minimize the environmental impacts and safety concerns related to the removal of the existing cells. Electrical will provide adequate security/safety lighting and camera coverage for the entire marine cargo facility while supporting the power needed for the current dock office. The Owner does require the installation of a fire hydrant from the existing water system at the facility. The Contractor will provide all the milestones on their "Basis of Design. G101" Concept Drawing with one adjustment: Under Corrosion Protection System, Anode System, the Anodes will be installed at the time of construction. (G101 Basis of Design is attached to this document).

Contractor will ensure current users are able to maintain operations during construction and re-establish operations post construction as they are currently configured. Contractor must consult with and gain approval from current users prior to any plans that may interrupt or change their current operations.

This will be a shift from the three phased concept originally submitted with the RAISE Grant application in that there will be no filling of uplands as noted for the planned Phase 1, nor the Phase 3 pass/pass. The contractor will be responsible for all the required NEPA, USACE, EHP and Section 106 permits and will work with the owner to provide all the documentation needed by MARAD to complete these processes. If required by the Haines Borough, contractor is prepared to shift back to the original conceptual designs for the the Lutak Dock.

BASIS OF DESIGN

DESIGN LIFE = 50 YEARS

LOADING REQUIREMENTS:

UNIFORM LIVE LOAD = 2,000 PSF

DESIGN VEHICLE (UNRESTRICTED) = MANITOWOC 4000W

DESIGN VEHICLE (RESTRICTED) = LHM 420 MOBILE HARBOR CRANE USE OF CRANE MATS REQUIRED FOR MOBILE HARBOR CRANE

DESIGN VESSEL CARGO = 60,000 TON MAX DISPLACEMENT 110 FT X 630 FT

DESIGN VESSEL CRUISE SHIP = 75,000 TON MAX DISPLACMENT 155 FT X 1050 FT

MOORING BOLLARDS = 150 TONS

BERTHING FENDERS = 1435 KIP-FT / 534 KIP

BULKHEAD PILE FOUNDATIONS:

ALL BULKHEAD PILES WILL BE KEYED INTO BEDROCK

CORROSION PROTECTION SYSTEM:

COATING: ONE COAT INORGANIC ZINC PRIMER AND TWO COATS OF COAL TAR EPOXY (16 MILS TOTAL), OR AN EQUIVALENT CORROSION PROTECTION SYSTEM

ANODE SYSTEM TO BE INSTALLED BY OWNER AFTER 10 YEAR OF DOCK OPERATIONS

TOTAL ALLOWABLE STEEL WASTAGE IN ZONE OF HIGH ATTACK (TIDAL ZONE) = 2 MILS PER YEAR

SEISMIC PERFORMANCE REQUIREMENTS:

ASCE 7-16 RISK CATEGORY = IV (ESSENTIAL FACILITY)

ASCE 61-14 DESIGN CLASSIFICATION = HIGH

OPERATING LEVEL EARTHQUAKE (OLE) PERFOMANCE LEVEL = MINIMAL DAMAGE 72 YEAR RETURN PERIOD PEAK GROUND ACCELERATION = 0.072g

CONTINGENCY LEVEL EARTHQUAKE (CLE) PERFORMANCE LEVEL = REPAIRABLE DAMAGE 475 YEAR RETURN PERIOD PEAK GROUND ACCELERATION = 0.200g

DESIGN EARTHQUAKE (DE) PERFORMANCE LEVEL = LIFE SAFETY PROTECTION 2250 YEAR RETURN PERIOD PEAK GROUND ACCELERATION = 0.494g

LIQUEFACTION MITIGATION

CURRENT BULKHEAD SOILS ANALYSIS SHOWS THAT LIQUEFACTION OF SOILS WITHIN THE BULKHEAD WILL NOT OCCUR DURING THE OPERATING LEVEL EARTHQUAKE

GROUND IMPROVEMENT WILL BE PROVIDED BASED ON GEOTECHNICAL DESIGN REQUIREMENTS SO THAT LIQUEFACTION WITHIN THE BULKHEAD WILL NOT OCCUR DURING A CONTINGENCY LEVEL EARTHQUAKE

THE BULKHEAD IS DESIGNED TO MEET OLE, CLE AND DE PERFORMANCE LEVELS ASSUMING ALL SOILS IN FRONT OF THE BULKHEAD ARE FULLY LIQUEFIED TO BEDROCK DEPTH DURING THESE SEISMIC EVENTS

Turnagain Marine Construction 8241 DIMOND HOOK DR, UNIT A ANCHORAGE, AK 99507 PHONE: (907) 261-8960 PROJECT NUMBER: 20-003 CONCEPT DRAWINGS
LUTAK DOCK REPLACEMENT HAINES, ALASKA
REV# DESCRIPTION DATE
JULY 25, 2022 SHEET TITLE: GENERAL NOTES SHEET NUMBER: G101

EXHIBIT E DESIGN-BUILDER'S PHASE 1 SCOPE OF SERVICES AND HOURLY RATES HAINES BOROUGH LUTAK DOCK REPLACEMENT

Design-Builder's Phase 1 Scope of Services is set forth in Exhibit C.

Design-Builder's hourly rates are as follows:

Name	Position	Hourly Rate Preconstruction	Hourly Rate Construction
All	All	\$200	\$200

EXHIBIT F1 FORM PHASE 1 CHANGE ORDER HAINES BOROUGH LUTAK DOCK REPLACEMENT

Phase 1 Contract Change Order

hase i bontract onange order		
Design-Builder		
	Design-Builder	

Project Name: Lutak Dock Replacement

Change Order#: _____

COR#	Description of Change	Change in Contract Time	Change in Phase 1 NTE

Original Phase 1 Not to Exceed Amount	\$
Net change by previous authorized Change Orders	\$
Total net [check one] [Increase / [Decrease in the Phase 1 NTE by this	\$
Change Order	
New Phase 1 NTE including this Change Order	\$

The new Contract Time including this Change Order is:

Phase 1 Completion Date	
Interim Milestone Dates:	

The changes in the Phase 1 Not to Exceed Amount and Contract Time identified in this Change Order include all costs and time extensions associated with performing the changes set forth herein.

Haines Borough	Design-Builder
By:	By:
Signature	Signature
Printed Name	Printed Name
Title:	Title:
Date:	Date:

EXHIBIT F2 FORM PHASE 2 CHANGE ORDER HAINES BOROUGH LUTAK DOCK REPLACEMENT

Phase 2 Contract Change Order

Design-Builder	Haines Borough

Project Name: Lutak Dock Replacement

Change Order#: _____

COR#	Description of Change	Change in Contract Time	Change in GMP

Original Guaranteed Maximum Price	\$
Net change by previous authorized Change Orders	\$
Total net [check one] [Increase / [Decrease in the GMP by this Change	\$
Order	
New Guaranteed Maximum Price including this Change Order	\$

The new Contract Time including this Change Order is:

New Scheduled Substantial Completion Date	
New Scheduled Interim Milestone Dates:	
New Final Completion Date	

The changes in the GMP and Contract Time identified in this Change Order include all costs and time extensions associated with performing the changes set forth herein.

Haines Borough	Design-Builder
By:	By:
Signature	Signature
Printed Name	Printed Name
Title:	Title:
Date:	Date:

EXHIBIT G FORM OF PHASE 2 AMENDMENT HAINES BOROUGH LUTAK DOCK REPLACEMENT

Unless the parties agree in writing otherwise, the Phase 2 Amendment shall be in a substantially similar form as follow:

1. Pursuant to Section 6.6.1 of the Agreement, this Phase 2 Amendment incorporates the following terms into the Agreement. To the extent any terms set forth in this Phase 2 Amendment conflict with the Agreement, the terms in this Phase 2 Amendment shall govern.

2. The Design-Builder has submitted to Owner the Phase 2 Proposal pursuant to Section 6.6.1.9 of the Agreement.

3. The Owner has reviewed the Phase 2 Proposal, the parties have reconciled the Owner's Comments pursuant to Section 6.6.1.7 of the Agreement, and the Owner has accepted the Phase 2 Proposal as reconciled. The conformed, reconciled Phase 2 Proposal is attached to this Phase 2 Amendment at Exhibit A and is incorporated as if fully set forth herein.

4. The Owner has decided to exercise its option to enter into Phase 2 of the Agreement pursuant to Section 6.6.1.9.b of the Agreement.

5. Consistent with the Phase 2 Proposal, the parties hereby establish the following Commercial Terms:

Guaranteed Maximum Price	\$
Cost of the Work	\$
Design-Builder's Lump Sum for Overhead and	\$
Profit	
Lump Sum General Conditions Costs	\$
Cost of the Work Contingency (Section 6.4.4.1.a)	\$
Design-Builder's Contingency (Section 6.4.4.1.b)	\$
Substantial Completion Date	
Final Completion Date	

6. Other Commercial Terms are set forth pursuant to the following Exhibits:

- a. Allowances as set forth in Section 6.4.1 of the Agreement are set forth and described in Exhibit B to the Phase 2 Amendment.
- b. Not to Exceed Sums as set forth in Section 6.4.2 of the Agreement are set forth and described in Exhibit C to the Phase 2 Amendment.
- c. Lump Sums as set forth in Section 6.4.3 of the Agreement (with the exception of the Design-Builder's Lump Sum for Overhead and Profit set forth above) are set forth and described in Exhibit D to the Phase 2 Amendment.
- d. Contingencies as set forth in Section 6.4.4 of the Agreement are set forth above and described in Exhibit E to the Phase 2 Amendment.
- e. Design-Builder's Lump Sum General Conditions Costs as set forth in Section 6.4.5 of the Agreement are set forth above and described in Exhibit F to the Phase 2 Amendment.
- f. Unit Prices and Hourly Rates as set forth in Section 6.5.6 of the Agreement are described in Exhibit G to the Phase 2 Amendment.

g. Liquidated Damages as provided in Section 5.4 – 5.6 of the Agreement are \$1,500 per calendar day.

7. Pursuant to Section 10.2 of the Agreement, Design-Builder shall provide a Payment and Performance Bond pursuant to Alaska Statutes Title 36 Chapter 25 equal to one hundred percent (100%) of the amount of the Guaranteed Maximum Price set forth above.

In executing this Amendment, Owner and Design-Builder each individually represents that it has the necessary financial resources to fulfill its obligations under this Amendment, and each has the necessary corporate approvals to execute this Amendment, and perform the services described herein.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed on the date set forth below.

HAINES BOROUGH	DESIGN-BUILDER		
Ву			
	Name of Design-Builder		
Its	Ву		
	Its		
Date:	Date:		
	DESIGN-BUILDER'S ADDRESS AND PHONE:		

EXHIBIT H HAINES BOROUGH LUTAK DOCK REPLACEMENT FEDERAL CONTRACT REQUIREMENTS

Pursuant to the requirements in the funding for the Project, the following provisions are incorporated into the Design-Build Agreement. In this Exhibit, the term "Contractor" shall also mean "Design-Builder".

1.1 Contracting with Small and Minority Businesses, Women's Business Enterprises, and Labor Surplus Firms.

1.1.1 The Design-Builder must take all necessary affirmative steps to assure that minority businesses, women's business enterprises, and labor surplus area firms are used with possible.

1.1.2 Affirmative steps must include:

.1 Placing qualified small and minority businesses and women's business enterprises on solicitation lists;

.2 Assuring that small and minority businesses, and women's business enterprises are solicited whenever they are potential sources;

.3 Dividing total requirements, when economically feasible, into smaller tasks or quantities to permit maximum participation by small and minority businesses, and women's business enterprises;

.4 Establishing delivery schedules, where the requirement permits, which encourage participation by small and minority businesses, and women's business enterprises; and

.5 Using the services and assistance, as appropriate, of such organizations as the Small Business Administration and the Minority Business Development Agency of the Department of Commerce.

1.2 Preference for Goods, Products, and Materials Produced in the US.

1.2.1 To the extent consistent with law and to the greatest extent practicable, Design-Builder shall prefer the purchase, acquisition, or use of goods, products, or materials (including but not limited to iron, aluminum, steel, cement, and other manufactured products) that are produced in the United States.

1.2.2 For the purposes of this section:

.1 "Produced in the United States" means, for iron and steel products, that all manufacturing processes, from the initial melting stage through the application of coatings, occurred in the United States;

.2 "Manufactured products" means the items and construction materials composed in whole or in part of non-ferrous metals such a aluminum; plastics and polymer-based products such as polyvinyl chloride pipe; aggregates such as concrete; glass, including optical fiber; and lumber.

1.3 Procurement of Recovered Materials

1.3.1 Design-Builder must comply with section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act. The requirements of Section 6002

include procuring only items designated in guidelines of the Environmental Protection Agency (EPA) at 40 CFR part 247 that contain the highest percentage of recovered materials practicable, consistent with maintaining a satisfactory level of competition, where the purchase price of the item exceeds \$10,000 or the value of the quantity acquired during the preceding fiscal year exceeded \$10,000; procuring solid waste management services in a manner that maximizes energy and resource recovery; and establishing an affirmative procurement program for procurement of recovered materials identified in the EPA guidelines.

1.4 Equal Employment Opportunity

During the performance of this contract, the Design-Builder agrees as follows:

(1) The Design-Builder will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. The Design-Builder will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following:

Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The Design-Builder agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.

(2) The Design-Builder will, in all solicitations or advertisements for employees placed by or on behalf of the Design-Builder, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.

(3) The Design-Builder will not discharge or in any other manner discriminate against any employee or applicant for employment because such employee or applicant has inquired about, discussed, or disclosed the compensation of the employee or applicant or another employee or applicant. This provision shall not apply to instances in which an employee who has access to the compensation information of other employees or applicants as a part of such employee's essential job functions discloses the compensation of such other employees or applicants to individuals who do not otherwise have access to such information, unless such disclosure is in response to a formal complaint or charge, in furtherance of an investigation, proceeding, hearing, or action, including an investigation conducted by the employer, or is consistent with the Design-Builder's legal duty to furnish information.

(4) The Design-Builder will send to each labor union or representative of workers with which it has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of the Design-Builder's commitments under this section, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.

(5) The Design-Builder will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.

(6) The Design-Builder will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to his books, records, and accounts by the administering

agency and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.

(7) In the event of the Design-Builder's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations, or orders, this contract may be canceled, terminated, or suspended in whole or in part and the Design-Builder may be declared ineligible for further Government contracts or federally assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.

(8) The Design-Builder will include the portion of the sentence immediately preceding paragraph (1) and the provisions of paragraphs (1) through (8) in every subcontract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The Design-Builder will take such action with respect to any subcontract or purchase order as the administering agency may direct as a means of enforcing such provisions, including sanctions for noncompliance:

Provided, however, that in the event the Design-Builder becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the administering agency, the Design-Builder may request the United States to enter into such litigation to protect the interests of the United States.

1.5 Davis-Bacon Act/Wage and Hour Provisions

1.5.1 Minimum wages.

(i) All laborers and mechanics employed or working upon the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor which is attached hereto and made a part hereof, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics. Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of paragraph (a)(1)(iv) of this section; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in § 5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein: Provided. That the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under paragraph (a)(1)(ii) of this section) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

(ii)

(A) The contracting officer shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The contracting officer shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:

(1) The work to be performed by the classification requested is not performed by a classification in the wage determination; and

(2) The classification is utilized in the area by the construction industry; and

(3) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.

(B) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the contracting officer agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the contracting officer to the Administrator of the Wage and Hour Division, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

(C) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the contracting officer do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the contracting officer shall refer the questions, including the views of all interested parties and the recommendation of the contracting officer, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

(D) The wage rate (including fringe benefits where appropriate) determined pursuant to <u>paragraphs (a)(1)(ii) (B)</u> or (C) of this section, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.

(iii) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.

(iv) If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, *Provided*, That the Secretary of Labor has found, upon the written request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.

(2) *Withholding.* The Haines Borough shall upon its own action or upon written request of an authorized representative of the Department of Labor

withhold or cause to be withheld from the contractor under this contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), all or part of the wages required by the contract, the (Agency) may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.

(3) Payrolls and basic records.

(i) Payrolls and basic records relating thereto shall be maintained by the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work (or under the United States Housing Act of 1937, or under the Housing Act of 1949, in the construction or development of the project). Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected, and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

(ii)

(A) The contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to the applicant, sponsor, or owner, as the case may be, for transmission to the DOT/MARAD. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on weekly transmittals. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from the Wage and Hour Division Web site at <u>http://www.dol.gov/esa/whd/forms/wh347instr.htm</u> or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to the to the applicant, sponsor, or owner, as the case may be, for transmission to the DOT/MARAD, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing

wage requirements. It is not a violation of this section for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the sponsoring government agency (or the applicant, sponsor, or owner).

(B) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:

(1) That the payroll for the payroll period contains the information required to be provided under § 5.5 (a)(3)(ii) of Regulations, <u>29 CFR part 5</u>, the appropriate information is being maintained under § 5.5 (a)(3)(i) of Regulations, <u>29 CFR part 5</u>, and that such information is correct and complete;

(2) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in Regulations, <u>29 CFR part 3</u>;

(3) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.

(C) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH-347 shall satisfy the requirement for submission of the "Statement of Compliance" required by paragraph (a)(3)(ii)(B) of this section.

(D) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under section 1001 of title 18 and section 231 of title 31 of the United States Code.

(iii) The contractor or subcontractor shall make the records required under <u>paragraph (a)(3)(i)</u> of this section available for inspection, copying, or transcription by authorized representatives of the (write the name of the agency) or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the Federal agency may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to <u>29 CFR 5.12</u>.

(4) Apprentices and trainees -

(i) *Apprentices.* Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services) to be

eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

(ii) Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program, the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

(iii) *Equal employment opportunity.* The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and <u>29 CFR part 30</u>.

(5) **Compliance with Copeland Act requirements.** The contractor shall comply with the requirements of <u>29 CFR part 3</u>, which are incorporated by reference in this contract.

(6) **Subcontracts.** The contractor or subcontractor shall insert in any subcontracts the clauses contained in $\underline{29 \text{ CFR 5.5(a)(1)}}$ through (10) and such other clauses as the (write in the name of the Federal agency) may by appropriate instructions require, and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in $\underline{29 \text{ CFR 5.5(a)(1)}}$ the contract of the subcontract or lower tier subcontractor with all the contract clauses in $\underline{29 \text{ CFR 5.5(a)}}$.

(7) **Contract termination: debarment.** A breach of the contract clauses in <u>29 CFR 5.5</u> may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in <u>29 CFR 5.12</u>.

(8) **Compliance with Davis-Bacon and Related Act requirements.** All rulings and interpretations of the Davis-Bacon and Related Acts contained in <u>29 CFR parts 1</u>, <u>3</u>, and <u>5</u> are herein incorporated by reference in this contract.

(9) **Disputes concerning labor standards.** Disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in <u>29 CFR parts</u> <u>5</u>, <u>6</u>, and <u>7</u>. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and the contracting agency, the U.S. Department of Labor, or the employees or their representatives.

(10) Certification of eligibility.

(i) By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or <u>29 CFR 5.12(a)(1)</u>.

(ii) No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).

(iii) The penalty for making false statements is prescribed in the U.S. Criminal Code, <u>18 U.S.C.</u> <u>1001</u>.

(b) **Contract Work Hours and Safety Standards Act.** The Agency Head shall cause or require the contracting officer to insert the following clauses set forth in <u>paragraphs (b)(1), (2), (3)</u>, and (4) of this section in full in any contract in an amount in excess of \$100,000 and subject to the overtime provisions of the Contract Work Hours and Safety Standards Act. These clauses shall be inserted in addition to the clauses required by § 5.5(a) or § 4.6 of part 4 of this title. As used in this paragraph, the terms *laborers* and *mechanics* include watchmen and guards.

(1) **Overtime requirements.** No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek.

(2) **Violation; liability for unpaid wages; liquidated damages.** In the event of any violation of the clause set forth in <u>paragraph (b)(1)</u> of this section the contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a

territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in <u>paragraph (b)(1)</u> of this section, in the sum of \$29 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in <u>paragraph (b)(1)</u> of this section.

(3) *Withholding for unpaid wages and liquidated damages.* The (write in the name of the Federal agency or the loan or grant recipient) shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in paragraph (b)(2) of this section.

(4) **Subcontracts.** The contractor or subcontractor shall insert in any subcontracts the clauses set forth in <u>paragraph (b)(1)</u> through (4) of this section and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in <u>paragraphs</u> (b)(1) through (4) of this section.

(c) The contractor or subcontractor shall maintain payrolls and basic payroll records during the course of the work and shall preserve them for a period of three years from the completion of the contract for all laborers and mechanics, including guards and watchmen, working on the contract. Such records shall contain the name and address of each such employee, social security number, correct classifications, hourly rates of wages paid, daily and weekly number of hours worked, deductions made, and actual wages paid. The records to be maintained under this paragraph shall be made available by the contractor or subcontractor for inspection, copying, or transcription by authorized representatives of Haines Borough and the Department of Labor, and the contractor or subcontractor will permit such representatives to interview employees during working hours on the job.

1.6 Contract Work Hours and Safety Standards Act

Design-Builder shall comply with 40 U.S.C. 3702 and 3702, as supplement by the Department of Labor regulations 29 CFR Part 5. Design-Builder must compute the wages of every mechanic and laborer on the basis of a standard work week of 40 hours. Work in excess of the standard work week is permissible provided that the work is compensated at a rate of not less than one and a half times the basic rate of pay for all hours worked in excess of 40 hours in the work week. No laborer or mechanic must be required to work in surroundings or under working conditions which are unsanitary, hazardous or dangerous. These requirements do not apply to the purchases of supplies or materials or articles ordinarily available on the open market, or contracts for transportation or transmission of intelligence.

1.7 Clean Air Act and the Federal Water Pollution Control Act

Design-Builder must comply with all applicable standards, orders, or regulations issued pursuant to the Clean Air Act (42 U.S.C. 7401-7671q) and the Federal Water Pollution Control Act as amended (33 U.S.C. 1251-1387). Violations will be reported to the Federal awarding agency and the Regional Office of the Environmental Protection Agency.

1.8 Debarment and Suspension

Design-Builder warrants that it has not been debarred or suspended from work for the United States Government or listed on the governmentwide exclusions in the System for award Management or otherwise ineligible for award pursuant to United States Executive Order 12549.

1.9 Byrd Anti-Lobbying Amendment

Design-Builder must not use the funds from this Contract to pay any person or organization for influencing or attempting to influence an officer or employee of any agency, a member of Congress, officer or employee of Congress, or an employee of a member of Congress in connection with obtaining any Federal contract, grant or any other award covered by 31 U.S.C. 1352. Design-Builder must submit a certification to Owner verifying this section.

1.10 Prohibition on certain telecommunications and video surveillance services or equipment

Design-Builder may not use the funds obtained pursuant to this Contract to:

(1) Procure or obtain;

(2) Extend or renew a contract to procure or obtain; or

(3) Enter into a contract (or extend or renew a contract) to procure or obtain equipment, services, or systems that uses covered telecommunications equipment or services as a substantial or essential component of any system, or as critical technology as part of any system. As described in <u>Public Law 115-232</u>, section 889, covered telecommunications equipment is telecommunications equipment produced by Huawei Technologies Company or ZTE Corporation (or any subsidiary or affiliate of such entities).

(i) For the purpose of public safety, security of government facilities, physical security surveillance of critical infrastructure, and other national security purposes, video surveillance and telecommunications equipment produced by Hytera Communications Corporation, Hangzhou Hikvision Digital Technology Company, or Dahua Technology Company (or any subsidiary or affiliate of such entities).

(ii) Telecommunications or video surveillance services provided by such entities or using such equipment.

(iii) Telecommunications or video surveillance equipment or services produced or provided by an entity that the Secretary of Defense, in consultation with the Director of the National Intelligence or the Director of the Federal Bureau of Investigation, reasonably believes to be an entity owned or controlled by, or otherwise connected to, the government of a covered foreign country.

(b) In implementing the prohibition under <u>Public Law 115-232</u>, section 889, subsection (f), paragraph (1), heads of executive agencies administering loan, grant, or subsidy programs shall prioritize available funding and technical support to assist affected businesses, institutions and organizations as is reasonably necessary for those affected entities to transition from covered communications equipment and services, to procure replacement equipment and services, and to ensure that communications service to users and customers is sustained.

EXHIBIT I FORM OF AFFIDAVIT OF INSURANCE COVERAGE HAINES BOROUGH LUTAK DOCK REPLACEMENT

I, _____, on behalf of _____ ("Proposer") being first duly sworn on oath, depose and state that:

- 1. I have reviewed the Design Builder's Insurance Requirements for the Haines Borough Lutak Dock Replacement Project as set forth in Exhibit A of the Contract Documents and Article 5 of the General Conditions.
- 2. Proposer has in place all insurance coverages with all terms required by the Contract Documents.
- 3. In particular, The insurance coverages referenced in the certificate of insurance have (or do not have) the following terms:
 - a. The Design Consultant's -Professional's liability policy does not contain any restriction, limitation, or exclusion pertaining to the design of construction means, methods, techniques, sequences or procedures.
 - Any faulty work exclusion, restriction or limitation of coverage in the Design Consultant's -Professional's liability policy related to Products or Product Design has been drafted or modified so as to provide coverage for goods or products installed.
 - c. Any exclusion, limitation, or restriction with respect to construction means, methods and techniques in the Design Consultant's -Professional's liability policy is one that applies to the implementation of such construction means, methods, techniques, sequences, or procedures by the Design Consultant or any person or entity providing design or other professional services as its Sub- Consultant and applies only if such entities are not performing any construction activities.
 - d. Any Faulty Work exclusion, limitation, or restriction in the Design Consultant's -Professional's liability policy is only applicable to the work self-performed by the Design Consultant.
 - e. The Design Consultant's -Professional's liability policy provides coverage for damages resulting from delays, including delays in project completion and cost overruns that result from the rendering or failure to render professional services.
 - f. The Design-Builder's professional liability policy's exclusion pertaining to construction means, methods, techniques, sequences or procedures only excludes, limits or restricts coverage for claims, to the same extent that such coverage is provided by the Design-Builder's valid and collectible commercial general liability and umbrella/excess liability policies.

- g. The Design-Builder's professional liability policy does not contain any restriction, limitation or exclusion pertaining to the design of construction means, methods, techniques, sequences, or procedures.
- h. Any faulty work exclusion, restriction or limitation of coverage in the Design Builder's Professional's liability policy related to Products or Product Design has been drafted or modified so as to provide coverage for goods or products installed.
- Any Faulty Work exclusion, limitation, or restriction in the Design Builder's Professional's liability policy is only applicable to the work self-performed by the Design Builder.
- j. The Design Builder's Professional's liability policy provides coverage for damages resulting from delays, including delays in project completion and cost overruns that result from the rendering or failure to render professional services.
- k. Professional liability exclusions in the Design-Builder's commercial general liability insurance have been limited to ISO endorsements CG 2280 or CG 2279 or their equivalent.
- I. The pollution liability insurance coverage provides coverage for off-site transportation by all applicable modes of conveyance.
- m. Any restriction, limitation, or exclusion related to Naturally Occurring Substances in the pollution liability insurance coverage has been modified so as not to apply to microbial matter and the release of such Naturally Occurring Substances as a result of the performance of Operations
- n. Any coverage provided on a claims made policy: (i) permits reporting of circumstances that could give rise to a claim; and (ii) provides coverage for post-expiration claims resulting from such circumstances.

SIGNATURE OF AFFIANT

SUBSCRIBED AND SWORN TO before me this _____ day of _____, 2022.

Notary Public in and for the State of _____

My Commission expires: _____

Attached hereto are:

ACORD (or equivalent) Certificates of Insurance with description of additional insureds and waiver of subrogation as applicable/required.





Assembly Agenda Bill

ASKA AS

Agenda Bill No.: 22-1214 Assembly Meeting Date: 09/13/22

Business Item Description:	Attachments:
Subject:	1. Resolution 22-09-997
Authorize the Appointment of Election Officials for	
October 4, 2022 Regular Election and set wages	
Originator: Percurah Clork	
Originating Department:	
Administration	
Date Submitted:	
9/6/22	
Full Title/Motion:	
Motion: Adopt Resolution 22-09-997	
Administrative Recommendation:	
This resolution is required by HBC 11.40.010.	
Figure 1	
Fiscal Impact:	at Declarate d
Expenditure Required Amou	nt Budgeted Appropriation Required
\$ TBD \$ 1,80	3 \$ none anticipated
Comprohonsivo Plan Consistonov Povi	
Comp Plan Policy Nos :	
Comp Plan Policy Nos.:	Consistent: 🛛 Yes 🔲 No
Comp Plan Policy Nos.:	Consistent: XYes No
Comp Plan Policy Nos.:	Consistent: 🛛 Yes 🔲 No
Comp Plan Policy Nos.: Summary Statement:	Consistent: XYes No
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the	Consistent: Yes No
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election of	Consistent: Yes No
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election y determined by assembly resolution. Election Clerks sh	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief)
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election y determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections.
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election y determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election of determined by assembly resolution. Election Clerks sho The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state increased the state increased the state increased the state increased the state increased the state increased the state increased the state increased the state increased the state	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour).
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election y determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state increased This recolution complian with the order requirements	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour).
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election of determined by assembly resolution. Election Clerks shi The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state increased This resolution complies with the code requirements.	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour).
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election of determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state increased This resolution complies with the code requirements.	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour).
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election y determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state incr This resolution complies with the code requirements. Referral:	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour).
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election of determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state increased This resolution complies with the code requirements. Referral: Sent to:	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour).
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election of determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state increased This resolution complies with the code requirements. Referral: Sent to: Recommendation: Refer to:	Consistent: Yes No a borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour). Date: Meeting Date:
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election y determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state incr This resolution complies with the code requirements. Referral: Sent to: Recommendation: Refer to:	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour). Date: Date:
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election of determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state increased This resolution complies with the code requirements. Referral: Sent to: Recommendation: Refer to: Assembly Action:	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour). Date: Date:
Comp Plan Policy Nos.: Summary Statement: HBC 11.40.010 provides that prior to each election the shall appoint election workers for each polling place, a may appoint a replacement. Additionally, the election v determined by assembly resolution. Election Clerks sh The FY23 budget provides for specific hourly rates and and it is anticipated the budget will be sufficient for the election officials has increased this year (the state incr This resolution complies with the code requirements. Referral: Sent to: Recommendation: Refer to: Morkshop Date(s):	Consistent: Yes No e borough clerk, with assembly approval through resolution, nd if a worker is unable to serve on Election Day, the clerk workers shall be compensated for their services as is all be scheduled for up to three at any given time (for relief). d estimates the number of hours based on previous elections, October 4, 2022 Election even though the hourly wage for eased its hourly wage to \$20 per hour). Date: Date: Public Hearing Date(s):

HAINES BOROUGH, ALASKA RESOLUTION No. 22-09-997 Draft

A Resolution of the Haines Borough Assembly authorizing the Election Supervisor to appoint Election Officials for the regular election to be held October 4, 2022, and establishing the wages.

WHEREAS, HBC 11.40.010 provides that prior to each election, the borough clerk, with assembly approval through resolution, shall appoint election workers for each polling place, and if a worker is unable to serve on election day, the clerk may appoint a replacement; and

WHEREAS, the election workers shall be paid such compensation for their services as is determined by resolution of the assembly; and

WHEREAS, the FY22 budget provides for specific hourly rates and estimates the number of hours based on previous elections, and it is anticipated the budget will be sufficient for the October 4, 2022 Regular Election,

NOW, THEREFORE, BE IT RESOLVED by the Haines Borough Assembly:

Section 1. The Borough Clerk, serving as the Election Supervisor, shall appoint Jeanette Heinrich as Election Chair (and Election Judge) of Precinct 3-330 – Haines #1 and Holly Thomas as Election Chair (and Election Judge) of Precinct 3-335 – Haines #2; and

Section 2. The Borough Clerk, serving as the Election Supervisor, shall further appoint at least two more judges of election for each polling place, as follows:

Diann Ahrens for Precinct 3-330 Marian Carlson for Precinct 3-330 Connie Staska for Precinct 3-330 Heather Rogers for Precinct 3-335 Brian Staurseth for Precinct 3-335

and may appoint up to three election clerks at any polling place where they are needed to conduct an orderly election and to relieve the election judges of undue hardship, as follows:

Melissa Aronson and Judy Ewald for Precinct 3-330 Donna Catotti and Rhonda Hinson for Precinct 3-330 Tom Heywood and Bill Broste for Precinct 3-330 Kim Phillips for Precinct 3-335 Hannah Clark for Precinct 3-335 Richard Gunick for Precinct 3-335

Section 3. The election chair/judges shall be paid \$16.50 per hour, and the election judges and clerks shall be paid \$15.00 per hour.

Adopted by a duly-constituted quorum of the Haines Borough Assembly on this 13th day of September, 2022.

Douglas Olerud, Mayor

ATTEST:

Alekka Fullerton, CMC, Borough Clerk



11A5

Assembly Agenda Bill

Agenda Bill No.: 22-1215

Assembly Meeting Date: 09/13/2022

Business Item Description:	Attachments:
Subject: Authorize Change Order # 2 for additional	1. Resolution 22-09-998
pipe on the Young Road Repair (\$128,173.00)	2. Proposal from SE Road Builders
Originator:	3. Change Order #2
Contracts and Grants Administrator	
Originating Department:	
Public Facilities	
Date Submitted:	
9/6/2022	

Full Title/Motion:

Motion: Adopt Resolution 22-09-998

Administrative Recommendation:

This resolution is recommended by the Director of Public Facilities and the WWTP Superintendent

Fiscal Impact:					
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets		
\$ 128,173.00	\$ See below	\$ 0	Reduced maintenance costs		

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Objective 2B, Pages 56-57

Consistent: Yes

□No

Summary Statement:

In December 2020 the Haines Borough experienced a major winter weather event that resulted in flooding, debris flows, and landslides throughout the borough. On June 14, 2022 the Assembly voted to award the construction contract to Glacier Construction, dba South East Road Builders for Young Road Repairs. During construction it was observed the existing piping was in worse condition than originally anticipated. Within the flow line of the existing CMP, the metal pipe walls have eroded and are leaking. With the existing sidewalk and curbing removed for construction, it is the optimal time to replace this failed existing storm drain piping. Therefore, the Engineer recommends replacing 839 linear feet of existing 18" CMP with new 18" CPP. The Haines Borough has an active grant DR 4585 AK, with the Federal Emergency Management Agency (FEMA) to reimburse the costs for repairs to Borough infrastructure and roads affected by the storm and the funds are sufficient to cover the costs of repair. Recommend approval of Change Order #2 for an amount not to exceed \$128,173.00.

Referral:

Referred to: Recommendation: Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/22

Public Hearing Date(s): Postponed to Date:

A Resolution of the Haines Borough Assembly authorizing the Borough Manager to execute a change order with Glacier Construction, Inc dba Southeast Road Builders, for additional Young Road repairs for an amount not-to-exceed \$128,173.00.

WHEREAS, in December 2020 the Haines Borough experienced a major winter weather event that resulted in flooding, debris flows, and landslides throughout the borough; and

WHEREAS, Young Road experienced extensive damage in the weather event, damage to the design, function, and capacity of the road; and

WHEREAS, Young Road can benefit from hazard mitigation improvements to the overall drainage issues related to the disaster damages; and

WHEREAS, on June 14, 2022, the Haines Borough Assembly awarded a construction contract to Glacier Construction, Inc. dba Southeast Road Builders for the repairs; and

WHEREAS, the Haines Borough has included mitigation measures to replace water and sewer lines that were compromised or exposed during the December 2020 storm which have been approved by FEMA and included in the original costs to repair Young Road; and

WHEREAS, the existing corrugated metal pipe (CMP) has deteriorated beyond the expected condition (the metal pipe walls have eroded and are leaking) and must be replaced; and

WHEREAS, with the existing sidewalk and curbing removed for construction, now is the optimal time to replace the failed existing storm drain piping; and

WHEREAS, recommends replacing 839 linear feet of existing 18" CMP with new 18" CPP (corrugated plastic pipe); and

WHEREAS, the Borough has an active grant DR 4585 AK, with the Federal Emergency Management Agency (FEMA) and CIP fund for other work in the Young Road area to reimburse the costs for repairs to the Borough infrastructure and roads affected by the storm,

NOW, THEREFORE, BE IT RESOLVED that the Haines Borough Assembly authorizes the Borough Manager to execute a change order with Glacier Construction, Inc. dba Southeast Road Builders for an amount not to exceed \$128,173.00 to replace 839 linear feet of existing 18"CMP with new 18" CPP.

Haines Borough Resolution No. 22-09-998 Page 2 of 2

Adopted by a duly-constituted quorum of the Haines Borough Assembly on this 13th day of September, 2022.

Douglas Olerud, Mayor

Attest:

Alekka Fullerton, CMC, Borough Clerk



HC60 Box 4800 Haines, Ak 99827 PHONE: 907-766-2833 FAX: 907-766-2832

CONTRACTOR'S LICENSE NO.: 134313 BUSINESS LICENSE NO.: 1086571

QUOTATION FOR: ProHns Attn: Garret Gladsjo @_garret@prohns.com Ph: (907) 780-4004

OWNER:

Haines Borough Attn: Mr. Ed Coffland, P.E. DATE: 9/2/2022 Project: Young Road Repairs - DEC 2020 Storm Event P/W # 00024(435785 Re: RFP Dated Wednesday 8/31/2022

ITEM	DESCRIPTION	QUANTITY	UNIT	UNIT PRICE	AMOUNT
202.03.00	Removal of Sidewalk	9.0	SY	\$16.00	\$144.00
202.04.00	Removal of Culvert Pipe	839.0	LF	\$12.00	\$10,068.00
202.09.00	Removal of Curb and Gutter	20.0	LF	\$9.00	\$180.00
603.21.18	Corrugated Polyethylene Pipe - 18"	839.0	LF	\$134.00	\$112,426.00
608.01.06	Conc. Sidewalk 6" Thick	9	SY	\$345.00	\$3,105.00
609.02.01	Curb and Gutter, Type I	20.0	LF	\$112.50	\$2,250.00
	TOTAL AMOUNT OF QUOTATION				\$12 8,173.00
	Estimated time to complete: - 8 days (Assumed Contract Extension Time is approved) Exiting Inventory of 18" Pipe is approved for use (new)				
Existing Base Bid Contract Pay Items Apply Grouting of Pipe/Field Inlets - Incidental - No Charge Traffic Control - Incidental - no charge provided existing signaling devices are allowed/approved/operational (if flaggers are necessary add'l charges apply) Mobilization: Project Superintendent & support Vehicle (incidental-no charge) Project Engineer & Support Vehicle (incidental-no charge) Corrugated Polyethylene Pipe: Cost to replace existing inventory (risk of material price increase- no charge)					
EXCLUSION	S: (If necessary, reserve the option to be compense Cold Weather Concrete Cold Weather Paving Extension applies No accommodation for lighting and work lights due to This quote may be withdrawn if not signed and r	sated on T&M o extended tim returned withi) e n 30 davs .		

• TERMS: Full payment due within 30 days of invoice, unless defined otherwise via subcontact A FINANCE CHARGE OF .88% per month (which is an annual percentage rate of 10.5%) will be imposed on any portion of your account not paid within 30 days after billing date.

• A service charge of 2.5% will be imposed on all credit card payments

If this proposal meets your approval, please complete and return. Your signature on one copy returned to us will make this a legal contract for the performance of the above work and your acceptance of the terms listed.

Company:

Phone No.

Authorized Signer:

Billing Address:

(printed name)

SIGNATURE:

HAINES BOROUGH CHANGE ORDER

ORDER NO. 2

Page <u>1</u> of <u>6</u>

DATE: September 2, 2022 PROJECT NAME: Young Road Repairs PW00024(435785) CONTRACT AGREEMENT DATE: June 22, 2022 OWNER: Haines Borough CONTRACTOR: Glacier Construction Inc, dba Southeast Road Builders **The following changes are hereby made to the CONTRACT DOCUMENTS:** Original CONTRACT AGREEMENT AMOUNT: \$2,256,879.50 DAYS/DATE: October 31, 2022 Current CONTRACT AGREEMENT as adjusted by previous CHANGE ORDER(S): \$2,310,529.50 This CHANGE ORDER will increase the CONTRACT AGREEMENT AMOUNT by: \$128,173.00 This CHANGE ORDER will increase the CONTRACT COMPLETION TIME. The new CONTRACT AGREEMENT AMOUNT including this CHANGE ORDER will be: \$2,438,702.50. The new CONTRACT COMPLETION TIME including this CHANGE ORDER will be: \$2,438,702.50.

DESCRIPTION OF CHANGES	INCREASE IN CONTRACT AMOUNT (\$)	(DECREASE) IN CONTRACT AMOUNT (\$)	CONTRACT TIME EXTENSION (DAYS)
During construction of water services installed under the existing 18" diameter corrugated metal pipe (CMP) storm drain system running along Project Left, it was observed the existing piping was in worse condition than originally anticipated. Within the flow line of the existing CMP, the metal pipe walls have eroded and are leaking. This type of failure can lead to the washout of pipe bedding, which in turn will cause culvert/pipe collapse beneath the roadway. With the existing sidewalk and curbing removed for construction, it is the optimal time to replace this failed existing storm drain piping. Therefore, the Engineer recommends replacing 839 linear feet of existing 18" CMP with new 18" CPP as shown in the attached Plan Sheets 13-16 and described below:			+8 Days
Adjust Quantity for Pay Item 202.0003.0000 Removal of Sidewalk: <i>Increase quantity from 90 SY to 99 SY at bid unit price of \$16.00/SY.</i>	+\$144.00		
Adjust Quantity for Pay Item 202.0004.0000 Removal of Culvert Pipe: Increase quantity from 479 LF to 1,318 LF at bid unit price of \$12.00/LF.	+\$10,068.00		
Adjust Quantity for Pay Item 202.0009.0000 Removal of Curb and Gutter: <i>Increase quantity from 358 LF to 378</i> <i>LF at bid unit price of \$9.00/LF.</i>	+\$180.00		
Adjust Quantity for Pay Item 603.0021.0018 Corrugated Polyethylene Pipe 18 Inch: <i>Increase quantity from 1,308</i> <i>LF to 2,147 LF at bid unit price of \$134.00/LF.</i>	+\$112,426.00		
--	---------------	----------	--------
Adjust Quantity for Pay Item 608.0001.0006 Concrete Sidewalk, 6 Inches Thick: <i>Increase quantity from 142 SY</i> <i>to 151 SY at bid unit price of \$345.00/SY.</i>	+\$3,105.00		
Adjust Quantity for Pay Item 609.0002.0001 Curb and Gutter, Type 1: <i>Increase quantity from 1,276 LF to 1,296</i> <i>LF at bid unit price of \$112.50/LF.</i>	+\$2,250.00		
TOTALS	+\$128,173.00	(\$0.00)	8 Days
NET CHANGE CONTRACT AMOUNT INCREASE OR (DECREASE)	+\$128,173.00		

Signatures Required:

The undersigned Contractor approves the foregoing Change Order as to the changes, if any, in the contract price specified for each item, including any and all supervision costs and other miscellaneous costs relating to the change in work, and as to the extension of time allowed, if any, for completion of the entire work on account of said Change Order. The Contractor agrees to furnish all labor and materials and perform all other necessary work, inclusive of that directly or indirectly related to the approved time extension, required to complete the Change Order items. This document will become a supplement of the contract and all provisions will apply hereto. It is understood that the Change Order shall be effective when all signatures are in place.

Owner:	Date:
Contractor:	Date:
State or Federal Agency, if applicable:	Date:











11A6

Assembly Agenda Bill

Agenda Bill No.: 22-1216

Assembly Meeting Date: 09/13/2022

Business I tem Description:	Attachments:
Subject: Joining AML in expressing concern over	1. Resolution 22-09-999
Constitutional Convention	2. AML's Resolution #2022-03
Originator:	
Assembly	
Originating Department:	
Administration	
Date Submitted:	
9/2/22	

Full Title/Motion:

Motion: Adopt Resolution 22-09-999

Administrative Recommendation:

This resolution is recommended by Assembly Member Gabe Thomas, the Mayor and the Manager

Fiscal Impact:			
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets
\$0	\$0	\$ 0	

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Consistent: Yes

No

Summary Statement:

On November 17, 2021, the Alaska Municipal League (AML) passed Resolution #2022-03 expressing concern over the impact of a Constitutional Convention. Over the last year, many municipalities, including the City and Borough of Juneau, have adopted the same Resolution. On August 24, 2022, Mayor Olerud described AML's concern for the conveying of a Constitutional Convention. On September 2, Assembly Member Gabe Thomas requested that the Assembly take a formal stand and join AML in expressing concern over the convening of a Constitutional Convention.

Referral:

Referred to: Recommendation: **Referral Date:**

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/22

Public Hearing Date(s):

Postponed to Date:

HAINES BOROUGH, ALASKA RESOLUTION No. 22-09-999 Draft

A Resolution Of The Haines Borough Assembly joining the Alaska Municipal League in expressing concern in the consideration of a Constitutional Convention, and taking an active role in educating the public and policy makers about its potential impact.

WHEREAS, the Alaska Municipal League's mission is to strengthen local governments, and AML's principles include to support the Alaska Constitution's mandate "to provide for maximum local self-government;" and

WHEREAS, Alaska's Constitution contains numerous provisions that provide for and enhance maximum local self-governments; and

WHEREAS, Art. XIII, sec. 3 of the Alaska Constitution provides that "if during any tenyear period a constitutional convention has not been held, the lieutenant governor shall place on the ballot for the next general election the question: "Shall there be a Constitutional Convention?..."; and

WHEREAS, Alaska voters overwhelmingly rejected the call for a constitutional convention in 1972 and in successive elections held in 1982, 1992, 2002, and 2012; and

WHEREAS, Alaska voters will be asked to vote on whether to convene a constitutional convention at the next general election on November 2, 2022; and

WHEREAS, the Alaska Constitution's system has been flexible, allowing cities and boroughs to merge when appropriate, and allowing for classes of municipalities, including home rule municipalities; and

WHEREAS, the Alaska Constitution commits the State to support public education, public health, public welfare, and for the University system; and

WHEREAS, the Alaska Constitution provides a statewide court system that relieves local governments of the responsibility for operating county courts, and allows for local prosecutions and defenses when and if a local government chooses to support them; and

WHEREAS, during the more than 60 years since Statehood the Alaska Municipal League has supported Alaska local governments in thoughtfully choosing among the options that the Alaska Constitution has offered them, and in making sure that there is revenue sharing from the State that will support the local governments that people choose to have; and

WHEREAS, where local government is concerned, the Alaska Constitution as written provides a necessary and empowering legal framework within which to exert local control and respond to resident priorities; and

WHEREAS, if voters were to authorize a convention, convention delegates elected at a subsequent election, would have plenary power to change any and all aspects of the constitution for submission to the voters; and

WHEREAS, there have been numerous proposals advanced over the years to shift public funding for education from the state to local governments, to alter public retirement programs, modify the selection of judges, alter the taxing power of government and other similar proposals that could fundamentally harm Alaska's local governments.

Haines Borough Resolution No. 22-09-999 Page 2 of 2

NOW, THEREFORE, BE IT RESOLVED that the Haines Borough Assembly and the Alaska Municipal League believe that a Constitutional Convention is unnecessary at this point in time and may negatively impact local governments; and

BE IT FURTHER RESOLVED that the Haines Borough Assembly and Alaska Municipal League express concern about any campaign for calling a Constitutional Convention, to which municipalities would educate voters of the value of the Constitution as currently written and the numerous options available to address perceived or real needs; and

BE IT FURTHER RESOLVED that the Haines Borough Assembly and the Alaska Municipal League encourage all local governments to work toward correcting any misinformation or misperception as may exist or that develops as part of the election cycle ahead.

Adopted by a duly-constituted quorum of the Haines Borough Assembly on this 13th day of September, 2022.

Douglas Olerud, Mayor

ATTEST:

Alekka Fullerton, CMC, Borough Clerk

Resolution of the Alaska Municipal League

Resolution #2022-03

A Resolution expressing concern in the consideration of a Constitutional Convention, and taking anactive role in educating the public and policy makers about its potential impact

WHEREAS, the Alaska Municipal League's mission is to strengthen local governments, and AML's principles include to support the Alaska Constitution's mandate "to provide for maximum local self-government;" and

WHEREAS, Alaska's Constitution contains numerous provisions that provide for and enhance maximumlocal self-governments; and

WHEREAS, Art.XIII, sec. 3 of the Alaska Constitution provides that "if during any ten-year period a constitutional convention has not been held, the lieutenant governor shall place on the ballot for the next general election the question: "Shall there be a Constitutional Convention?" and

WHEREAS, Alaska voters overwhelmingly rejected the call for a constitutional convention in 1972 and insuccessive elections held in 1982, 1992, 2002, and 2012; and

WHEREAS, Alaska voters will be asked to vote on whether to convene a constitutional convention at the next general election on November 2, 2022; and

WHEREAS, the Alaska Constitution's system has been flexible, allowing cities and boroughs to mergewhen appropriate, and allowing for classes of municipalities, including home rule municipalities; and

WHEREAS, the Alaska Constitution commits the State to support public education, public health, publicwelfare, and for the University system; and

WHEREAS, the Alaska Constitution provides a statewide court system that relieves local governments of the responsibility for operating county courts, and allows for local prosecutions and defenses when and if a local government chooses to support them; and

WHEREAS, during the more than 60 years since Statehood the Alaska Municipal League has supported Alaska local governments in thoughtfully choosing among the options that the Alaska Constitution has offered them, and in making sure that there is revenue sharing from the State that will support the local governments that people choose to have; and

WHEREAS, where local government is concerned, the Alaska Constitution as written provides a necessary and empowering legal framework within which to exert local control and respond to resident priorities; and

WHEREAS, if voters were to authorize a convention, convention delegates elected at a subsequent election, would have plenary power to change any and all aspects of the constitution for submission to the voters; and

WHEREAS, there have been numerous proposals advanced over the years to shift public funding for education from the state to local governments, to alter public retirement programs, modify the selection of judges, alter the taxing power of government and other similar proposals that could fundamentally harm Alaska's local governments.

NOW THEREFORE BE IT RESOLVED that the Alaska Municipal League believes that a ConstitutionalConvention is unnecessary at this point in time and may negatively impact local governments; and

BE IT FURTHER RESOLVED that the Alaska Municipal League expresses concern about any campaign for calling a Constitutional Convention, to which the League would enjoin in opposition by educating voters of the value of the Constitution as currently written and the numerous options available to address perceived or real needs; and

BE IT FURTHER RESOLVED that AML encourages all local governments to work toward correcting anymisinformation or misperception as may exist or that develops as part of the election cycle ahead.

PASSED AND APPROVED BY THE ALASKA MUNICIPAL LEAGUE ON THE $17^{\rm TH}$ DAY OF NOVEMBER 2021

Signed: Vete Petersen

Attest:

President, Alaska Municipal League

Executive Director, Alaska Municipal League



11B1

Assembly Agenda Bill

Agenda Bill No.: 22-1213

Assembly Meeting Date: 09/13/22

Business I tem Description:	Attachments:
Subject: Reimbursement of School Bond Debt	1. Ordinance 22-09-628
shortfall from FY20 and FY21	2. Certified Excerpt 3. Clerk's Memo from August 17, 2022 Meeting
Originator:	
Assembly	
Originating Department:	
Administration	
Date Submitted:	
08/24/22	

Full Title/Motion:

Motion: Introduce Ordinance 22-09-628 and schedule it for its first public hearing 09/27/22 and a second public hearing 10/11/22.

Administrative Recommendation:

This ordinance is recommended by the Clerk.

Fiscal Impact:			
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets
\$	\$	\$	Increases in Dock Revenue

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Consistent:	Yes	
0011010101111		

■Yes □No

Summary Statement:

At its August 24, 2022 meeting, the Borough Assembly directed the Clerk to bring an Ordinance to redeposit permanent funds the voter's allowed to be used to pay the State's deficiency in school bond debt. An additional motion was made to reimburse the CIP funds used and to designate funds to be used for a projected State school bond debt in FY24.

The Clerk has made two changes to the direction, as follows:

1. The motion was to repay FY17 and FY20 bond debt as allowed by the voters, when in fact the voters approved FY20 and FY21 bond debt reimbursement.

2. The second motion called for a Resolution when an Ordinance (allowing more public comment) was felt to be appropriate, particularly since it can all be in one Ordinance since it is the same subject matter (school bond debt reimbursement).

Referral: Referred to:

Recommendation:

Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/22

Public Hearing Date(s): 9/27/22 and 10/11/22 Postponed to Date:

HAINES BOROUGH ORDINANCE No. 22-09-628



An ordinance of the Haines Borough reimbursing funds withdrawn to pay FY20 and FY21 school bond debt underfunded by the State of Alaska.

WHEREAS, in the FY20 fiscal year, the Haines Borough experienced a loss of \$464,053 of school debt reimbursement funding from the State of Alaska when the governor vetoed 50% of budgeted funding to municipalities from the State of Alaska; and

WHEREAS, in the FY21 fiscal year, the Haines Borough also experienced a loss of 70% school debt reimbursement from the State of Alaska which amounted to \$896,298; and

WHEREAS, the Haines Borough diverted Capital Improvement Funds (50) to offset \$324,075 of the loss of debt reimbursement from the State; and

WHEREAS, the Haines Borough voters authorized a distribution of \$450,000 from the permanent fund (97) to offset some of the loss of debt reimbursement from the State; and

WHEREAS, the State of Alaska has reimbursed the Haines Borough for the losses of school bond debt reimbursement sustained in FY17, FY20 and FY21 and the Borough has received \$1,548,663 for the repayment; and

WHEREAS, the State of Alaska has authorized additional reimbursement to the Haines Borough for the FY22 loss in school bond debt reimbursement in the additional amount of \$523,941; and

WHEREAS, the Legislative Finance's overview of Governor Dunleavy's 10-year plan states that "beginning in FY24, School Debt Reimbursement is funded at 50%" which will result in an approximate \$448,717 loss to the Haines Borough; and

WHEREAS, the Haines Borough would like to plan for the State's projected underfunding of its school bond obligation in FY24; and

WHEREAS, Haines Borough Code allows the Assembly to deposit funds into the Permanent Fund if it does so by ordinance per HBC 3.24.030(D),

NOW, THEREFORE, BE IT ENACTED BY THE HAINES BOROUGH ASSEMBLY:

Section 1. <u>Classification</u>. This ordinance is a non-code ordinance.

Section 2. <u>Severability</u>. If any provision of this ordinance or any application thereof to any person or circumstance is held to be invalid, the remainder of this ordinance and the application to other persons or circumstances shall not be affected thereby.

Section 3. <u>Effective Date</u>. This ordinance shall become effective immediately upon adoption.

Section 4. <u>Appropriation</u>. This appropriation is hereby authorized as part of a budget for the fiscal year July 1, 2022 through June 30, 2023.

Section 3. <u>Purpose</u>. It is hereby determined to be for a public purpose and in the public interest of the Haines Borough to:

Haines Borough Ordinance No. 22-09-628 Page 2 of 2

- Redeposit the amount of \$450,000 into the Permanent Fund (Fund 97) per HBC
 3.24.030(D) which was withdrawn by the voters on October 6, 2020
- (2) Increase the FY23 Capital Improvement Project Account (Fund 50) in the amount of \$324,075 which was used from CIP fund balance in FY21.
- (3) To designate the amount of \$448,717 of the Areawide General Fund (Fund 01) to pay the State of Alaska's anticipated FY24 school bond debt reimbursement shortfall.

ADOPTED BY A DULY CONSTITUTED QUORUM OF THE HAINES BOROUGH ASSEMBLY THIS 11th DAY OF OCTOBER, 2022.

Attest:

Douglas Olerud, Mayor

Alekka Fullerton, CMC, Borough Clerk

Date of Introduction:09/13/22Date of First Hearing:09/27/22Date of Second Hearing:10/11/22

Haines Borough Borough Assembly Regular Meeting August 24, 2022 EXCERPT OF MINUTES

Present: Mayor **OLERUD**, Assembly Members Gabe **THOMAS**, Paul **ROGERS**, Jerry **LAPP**, Cheryl D. **STICKLER**, Tyler **HULING** and Debra **SCHNABEL**.

Staff Present: Annette **KREITZER**/Borough Manager, Alekka **FULLERTON**/Borough Clerk, Steven **AUCH**/Tourism Director.

<u>Visitors Present</u>: Sharon **RESNICK**, Sabrina **STICKLER**, Zach **WENTZEL**, Margarette **JONES**, Karen **GARCIA**, Lucy **SILBAUGH/CVN**, Ted **POPLEY**, Julie **KORSMEYER**, Evelyna **VIGNOLA** and others.

10. UNFINISHED BUSINESS

A. Redeposit into Permanent Fund of Voter Approved Withdrawl of Funds used to pay part of the the FY20 and FY21 School Bond Debt

<u>Motion</u>: STICKLER moved to "Direct the Clerk to draft a Resolution to repay the permanent fund for the deficiency in the School Bond Debt Reimbursement from FY17 and FY20,"

Primary Amendment: **SCHNABEL** moved to amend the word "Resolution" with the word "Ordinance", and the motion carried unanimously. And the motion carried unanimously.

Motion: THOMAS moved to "Direct the Clerk to draft a Resolution to set aside \$448,717.50 anticipated deficiency for the FY24 School Bond Debt Reimbursement and reimbursement of \$324,075 of CIP funds previously used for School Bond Debt Reimbursement," and

Motion to Divide Question: SCHNABEL moved to divide the question and the motion carried unanimously.

Divided question:

"Direct the Clerk to draft a Resolution to set aside \$448,717.50 anticipated deficiency for the FY24 School Bond Debt Reimbursement," and the motion carried unanimously.

"Moved to reimburse \$324,075 of CIP funds previously used for School Bond Debt Reimbursement," and the motion carried unanimously.

I certify that the above is a true and complete excerpt of agenda item 10 from the August 9, 2022 borough assembly meeting.



legandu

Alekka Fullerton, CMC, Borough Clerk Haines Borough, Alaska

Memo from the Borough Clerk



Date:August 17, 2022To:Haines Borough AssemblyFrom:Alekka FullertonRe:School Bond Debt Reimbursement

In FY17, the State of Alaska underfunded its School Debt obligation in the amount of \$188,312. For that year, the Borough shifted Areawide property tax from the general fund (01) to the debt service fund (76) in order to cover the shortfall. The underfunded amount was, therefore, taken from the General Fund. The Haines Borough has received reimbursement for the \$188,312 shortfall and recognition of the funds is reflected in Ordinance 22-07-624.

In FY20, the State of Alaska again underfunded its School Debt obligation (by 50%) in the amount of \$464,053. That amount was again spread out and paid out of the Areawide General Fund. The Haines Borough has received reimbursement for the \$464,053 shortfall and recognition of the funds is reflected in Ordinance 22-07-624.

In FY21, The State of Alaska defunded the entire amount of its School Debt obligation. This resulted in a shortfall of \$896,298. The Assembly went to the voters for permission as follows:

Shall the Haines Borough withdraw \$450,000 of the Permanent Fund principal to reimburse the Areawide General Fund for a portion of the FY20 and FY21 school bond debt payment?

The proposition passed and the \$450,000 was taken from the Permanent Fund. The remaining shortfall balance was taken from the CIP Fund in the amount of \$324,075.00 and the Areawide General Fund contributed \$122,223. The Haines Borough has just received the FY21 reimbursement in the amount of \$896,298 for the FY21 School Debt obligation shortfall and has requested that Ordinance 22-07-624 be amended to reflect the additional revenue.

In FY22, the State of Alaska again underfunded its School Debt obligation (by 50%) and the \$523,941 shortfall was taken from the Areawide General Fund. Reimbursement to the Haines Borough for the \$523,941 for FY22 has been authorized by the State of Alaska but the funds have not yet been received by the Haines Borough.

At the August 9, 2022 assembly meeting, Assembly Member Stickler moved to:

Motion: STICKLER moved to "Direct the Clerk to draft a Resolution to repay the permanent fund for the deficiency in the School Bond Debt Reimbursement from FY17 and FY20," and the motion was postponed until this meeting.

Of the \$2,072,604 reimbursements to the Haines Borough (some yet to be received), \$22,386 was retroactively paid directly to employees who did not get a wage increase as a direct result of the underfunding of the State's School Debt obligation.

The remaining \$2,050,218 has yet to be appropriated. The motion on the floor is to return \$450,000 of this amount to the Permanent Fund.

The Manager supports reimbursing the Permanent Fund as well as reimbursing the CIP Fund for the \$324,075 taken from that fund for the FY21 School Debt shortfall. The Manager also suggests the designation of \$448,717.50 of the funds for FY24 since the Legislative Finance's overview of Governor Dunleavy's 10-year plan states that "beginning in FY24, School Debt Reimbursement is funded at 50%". If all of the Manager's suggestions are taken, the Assembly will still have \$827,425.50 to appropriate.

Attached is information regarding debt service from each of the subject years.



11B2

Assembly Agenda Bill

Agenda Bill No.: 22-1212

Assembly Meeting Date: 09/13/22

Business Item Description:	Attachments:
Subject: Amend Harbor Fee Schedule	1. Ordinance 22-09-629 2. PHAC Recommendation 3. Proposed Changes to Harbor Fees
Originator:	
Harbormaster	
Originating Department:	
Ports and Harbors	
Date Submitted:	
08/25/22	

Full Title/Motion:

Motion: Introduce Ordinance 22-09-629 and schedule it for its first public hearing 09/27/22 and a second public hearing 10/11/22.

Administrative Recommendation:

This ordinance is recommended by the harbormaster.

Fiscal Impact:			
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets
\$	\$	\$	Increases in Dock Revenue

Comprehensive Plan Consistency Review:Comp Plan Goals/Objectives:
4.5.5 Borough Enterprise Funds; Page 53Consistent: Yes No

Summary Statement:

See attached memo from the Harbormaster.

This is recommended by the Port and Harbor Advisory Committee.

Referral: Referred to:

Recommendation:

Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/22

Public Hearing Date(s): 9/27/22 and 10/11/22 Postponed to Date:

An Ordinance of the Haines Borough Adopting a Revised Fee Schedule for Harbor Facility Usage.

WHEREAS, the Haines Borough owns and operates the Portage Cove Harbor and the Letnikof Cove Harbor; and

WHEREAS, the Borough Assembly establishes rates, fees, and penalties for Haines Borough harbor facilities; and

WHEREAS, Haines Borough Code, section 2.12.020(B), requires fines and penalties be provided by ordinance,

NOW THEREFORE BE IT ENACTED, by the Haines Borough Assembly, that the following Harbor Fee Schedule is hereby adopted by ordinance.

Section 1. <u>Classification</u>. This ordinance is not of a general and permanent nature and shall not become a part of the Haines Borough Code of Ordinances.

Section 2. <u>Severability</u>. If any provision of this ordinance or any application thereof to any person or circumstance is held to be invalid, the remainder of this ordinance and the application to other persons or circumstances shall not be affected thereby.

Section 2. <u>Effective Date</u>. This ordinance shall become effective immediately upon adoption.

Section 3. <u>Purpose</u>. To adopt, via ordinance, the following revised Harbor Fee Schedule:

Haines Borough – Harbor Fee Schedule

(See HBC Title 16 for regulations and more information)

Portage Cove Small Boat Harbor

Transient Boat Moorage Fees

Short-term (less than two weeks)	\$.65
Long-term (two weeks and longer)	\$6.50

Winter Rates – From October 1 through March 31, transient moorage vessels qualify for winter rates: one-half the amount that the vessel would ordinarily be charged. To qualify vessel must be up-to-date on all Port and Harbor Fees and pay for the winter moorage rate in advance.

* Transient moorage which must be invoiced will be charged at twice the normal rate (HBC 16.16.050).

Boat Launch Ramp Use Fees

	\$60.00 first trailer per year
Annual Launch Ramp Sticker –	\$30.00 each for second, third, & fourth trailers
Discounted Rate Does Not Apply For	per year (same owner) with proof of
Commercial Use	ownership
	\$60.00 each additional trailer per year
Daily Launch Ramp Use Fee	\$15.00 per trailer per day

Live-Aboard Fees for the Portage Cove Small Boat Harbor

1 to 14 days	No charge other than the regular moorage rate
Over 14 days	\$75.00 \$80.00 per month in addition to the
(prohibited October 15 th to April 1 st)	regular moorage rate

Haines Borough Ordinance No. 22-09-629 Page 2 of 6

Portage Cove Small Boat Harbor (Continued)

Boat Houses, Floats, Scows, Barges, Pile Drivers, and Dredges

These vessels are allowed only on a temporary basis at the harbormaster's discretion and, when allowed, are assessed a fee twice the rate of other vessels. Note: Living aboard is not allowed on these vessels and watercraft.

Electrical Connection and Service Rates

Connection Costs (permanent stalls or open moorage) <i>Paid to the local utility</i>	Actual cost of connection (boat owner makes the arrangements with the local utility and pays them directly)
Service Fee (all vessels utilizing electrical service) Paid to the borough	\$10.00 per day
Service Fee (utilizing 240 service)	\$20.00 per day
Annual Service Maintenance fee (if Electrical service is activated)	\$15.00 annual

Harbor Crane Use

Annual Crane Use Permit (not available for transient vessels)	\$50.00 <mark>\$60.00</mark> per year	
Transient Vessels	\$20.00 per hour	

Miscellaneous Fees

Grid Use	\$15.00 min or \$.50 <u>\$0.68 per ft per tide cycle</u>
Harbor Staff Labor (regularly scheduled duty hours)	\$40.00 per hour (one hour minimum)
Harbor Staff Labor (After regular hours)	\$80.00 per hour (one hour minimum)
Tie-up Fee	\$20.00 per time plus cost of line (Harbor staff time not included)
Fee to Move a Boat	\$50.00 per time
Waitlist Fee	\$25.00 (first year) and \$10.00 per year thereafter
Seniority Transfer list Fee	\$10.00 per year
Pump Fee	\$20.00 per hour
Portable Electric Generator Fee	\$20.00 per day
Upland Storage Fee	\$.20 <u>\$0.40</u> per square foot per month
Vessel Storage Fee (current annual stall renters-with all fees paid in advance)	\$.10 per square foot per month
Boat Stand Fee	\$2.50 per month or portion of a month

Hazardous Waste Disposal

Motor Oil	Included in moorage fees
Hydraulic Oil	Included in moorage fees
Diesel	Included in moorage fees
<mark>Gasoline</mark>	Included in moorage fees
Antifreeze	Included in moorage fees
Oil filters and Oiled Rags	Included in moorage fees
Paint	Included in moorage fees
Hazardous Materials Not Listed From	Included in moorage fees
Vessel	

Haines Borough Ordinance No. 22-09-629 Page 3 of 6

Portage Cove Small Boat Harbor (Continued)

Permanent Open Moorage Vessels

Long-term vessels with no permanent stall that meet the conditions outlined in HBC 16.16.105(A) shall qualify for rates equal to the regular moorage rate for their size vessel (see #14 - *Annual Moorage Rates*). If the conditions are not met, transient rates apply (see #1 – *Transient Boat Moorage Fees*).

Permanent stall renters that have a second commercial vessel and meet the conditions outlined in HBC 16.16.105(A) shall qualify for rates equal to the regular moorage rate for their size vessel (see #14 - *Annual Moorage Rates*) provided the permanent stall fee is paid. If the conditions are not met, transient rates apply (see #1 – *Transient Boat Moorage Fees*), and from April 1st through September 30th these vessels shall be charged the long-term transient moorage rate.

Commercial Vessel Per-Passenger Fee

Any commercial passenger vessel utilizing harbor facilities shall pay a per-passenger fee for all embarking and disembarking passengers. Transient harbor moorage rates apply to any vessel moored for over 3 hours.

Per-Passenger Fee Loading/unloading	2020	2021	2022	2023	2024	2025	2026	2027
5% Annual Increase	\$.50	\$.53	\$.55	<mark>\$.58</mark>	<mark>\$0.61</mark>	<mark>\$0.64</mark>	<mark>\$0.67</mark>	<mark>\$0.70</mark>

Permanent stall renters that operate a commercial passenger vessel with a max capacity of 50 passengers and meet the conditions outlined in HBC 16.16.105(A) shall be exempt from paying the Per Passenger Fee.

Annual Moorage Rates

Length	10/1/19 to 9/30/20	10/1/20 t o 9/30/21	10/1/21 to 9/30/22	<u>10/1/22 to</u> <u>9/30/23</u>	<u>10/1/23 to</u> 9/30/24	<u>10/1/24 to</u> <u>9/30/25</u>
(feet)	(\$26/ft up to 40 ft; \$32/ft over 40 ft)	(\$27/ft up to 40 ft; \$33/ft over 40 ft)	(\$28/ft up to 40 ft; \$34/ft over 40 ft)	<u>(\$28.84/ft</u> up to 40 ft; <u>\$35.02/ft</u> over 40 ft)	<u>(\$29.71/ft up to</u> <u>40 ft; \$36.07/ft</u> <u>over 40 ft)</u>	<u>(\$30.60/ft up to</u> <u>40 ft; \$37.15/ft</u> <u>over 40 ft)</u>
up to 25	\$650	\$675	\$700	<mark>\$721.00</mark>	<u>\$ 742.75</u>	<u>\$ 765.00</u>
26	\$676	\$702	\$728	<mark>\$749.84</mark>	<mark>\$ 772.46</mark>	<mark>\$ 795.60</mark>
27	\$702	\$702	\$756	<mark>\$778.68</mark>	<mark>\$ 802.17</mark>	<mark>\$ 826.20</mark>
28	\$728	\$756	\$784	<mark>\$807.52</mark>	<mark>\$ 831.88</mark>	<mark>\$ 856.80</mark>
29	\$754	\$783	\$812	<mark>\$836.36</mark>	<mark>\$ 861.59</mark>	<mark>\$ 887.40</mark>
30	\$780	\$810	\$840	<mark>\$865.20</mark>	<mark>\$ 891.30</mark>	<mark>\$ 918.00</mark>
31	\$806	\$837	\$868	<mark>\$894.04</mark>	<mark>\$ 921.01</mark>	<mark>\$ 948.60</mark>
32	\$832	\$864	\$896	<mark>\$922.88</mark>	<mark>\$ 950.72</mark>	<mark>\$ 979.20</mark>
33	\$858	\$891	\$924	<mark>\$951.72</mark>	<mark>\$ 980.43</mark>	<u>\$ 1,009.80</u>
34	\$884	\$918	\$952	<mark>\$980.56</mark>	<mark>\$ 1,010.14</mark>	<mark>\$ 1,040.40</mark>
35	\$910	\$945	\$980	<mark>\$1,009.40</mark>	<mark>\$ 1,039.85</mark>	<u>\$ 1,071.00</u>

Haines Borough Ordinance No. 22-09-629 Page 4 of 6

Length (feet)	10/1/19 to 9/30/20	10/1/20 to 9/30/21	10/1/21 to 9/30/22	<u>10/1/22 to</u> 9/30/23	<u>10/1/23 to</u> 9/30/24	<u>10/1/24 to</u> <u>9/30/25</u>
	(\$26/ft up to 40 ft; \$32/ft over	(\$27/ft up to 40 ft; \$33/ft over 40 ft)	(\$28/ft up to 40 ft; \$34/ft over 40 ft)	(\$28.84/ft up to 40 ft; \$35.02/ft over 40 ft)	<u>(\$29.71/ft up to</u> <u>40 ft; \$36.07/ft</u> <u>over 40 ft)</u>	<u>(\$30.60/ft up to</u> <u>40 ft; \$37.15/ft</u> over 40 ft)
37	\$962	\$999	\$1,036	\$1,067.08	\$ 1,099.27	\$ 1,132.20
38	\$988	\$1,026	\$1,064	<mark>\$1,095.92</mark>	<mark>\$ 1,128.98</mark>	\$ 1,162.80
39	\$1,014	\$1,053	\$1,092	<mark>\$1,124.76</mark>	<mark>\$ 1,158.69</mark>	\$ 1,193.40
40	\$1,040	\$1,080	\$1,120	<mark>\$1,153.60</mark>	<mark>\$ 1,188.40</mark>	\$ 1,224.00
41	\$1,312	\$1,353	\$1,394	<mark>\$1,435.82</mark>	<mark>\$ 1,478.87</mark>	\$ 1,523.15
42	\$1,344	\$1,386	\$1,428	<mark>\$1,470.84</mark>	\$ 1,514.94	\$ 1,560.30
43	\$1,376	\$1,419	\$1,462	<mark>\$1,505.86</mark>	\$ 1,551.01	<u>\$ 1,597.45</u>
44	\$1,408	\$1,452	\$1,496	<mark>\$1,540.88</mark>	<mark>\$ 1,587.08</mark>	<u>\$ 1,634.60</u>
45	\$1,440	\$1,485	\$1,530	<u>\$1,575.90</u>	<u>\$ 1,623.15</u>	<u>\$ 1,671.75</u>
46	\$1,472	\$1,518	\$1,564	<mark>\$1,610.92</mark>	<u>\$ 1,659.22</u>	<u>\$ 1,708.90</u>
47	\$1,504	\$1,551	\$1,598	<mark>\$1,645.94</mark>	\$ 1,695.29	<u>\$ 1,746.05</u>
48	\$1,536	\$1,584	\$1,632	<mark>\$1,680.96</mark>	<u>\$ 1,731.36</u>	<u>\$ 1,783.20</u>
49	\$1,568	\$1,617	\$1,666	<mark>\$1,715.98</mark>	<mark>\$ 1,767.43</mark>	<u>\$ 1,820.35</u>
50	\$1,600	\$1,650	\$1,700	<mark>\$1,751.00</mark>	<mark>\$ 1,803.50</mark>	<u>\$ 1,857.50</u>
51	\$1,632	\$1,683	\$1,734	<mark>\$1,786.02</mark>	<mark>\$ 1,839.57</mark>	<u>\$ 1,894.65</u>
52	\$1,664	\$1,716	\$1,768	<mark>\$1,821.04</mark>	<mark>\$ 1,875.64</mark>	<u>\$ 1,931.80</u>
53	\$1,696	\$1,749	\$1,802	<mark>\$1,856.06</mark>	<mark>\$ 1,911.71</mark>	\$ 1,968.95
54	\$1,728	\$1,782	\$1,836	<mark>\$1,891.08</mark>	<mark>\$ 1,947.78</mark>	<u>\$ 2,006.10</u>
55	\$1,760	\$1,815	\$1,870	<mark>\$1,926.10</mark>	<mark>\$ 1,983.85</mark>	<u>\$ 2,043.25</u>
56	\$1,792	\$1,848	\$1,904	<mark>\$1,961.12</mark>	<mark>\$ 2,019.92</mark>	<u>\$ 2,080.40</u>
57	\$1,825	\$1,881	\$1,938	<mark>\$1,996.14</mark>	<mark>\$ 2,055.99</mark>	<u>\$ 2,117.55</u>
58	\$1,856	\$1,914	\$1,972	<mark>\$2,031.16</mark>	<mark>\$ 2,092.06</mark>	<u>\$ 2,154.70</u>
59	\$1,888	\$1,947	\$2,006	<mark>\$2,066.18</mark>	<mark>\$ 2,128.13</mark>	<u>\$ 2,191.85</u>
60	\$1,920	\$1,980	\$2,040	<u>\$2,101.20</u>	<u>\$ 2,164.20</u>	<u>\$ 2,229.00</u>
61	\$1,952	\$2,013	\$2,074	<mark>\$2,136.22</mark>	<mark>\$ 2,200.27</mark>	<u>\$ 2,266.15</u>
62	\$1,984	\$2,046	\$2,108	<mark>\$2,171.24</mark>	<mark>\$ 2,236.34</mark>	<u>\$ 2,303.30</u>
63	\$2,016	\$2,079	\$2,142	<mark>\$2,206.26</mark>	<u>\$ 2,272.41</u>	<u>\$ 2,340.45</u>
64	\$2,048	\$2,112	\$2,176	<mark>\$2,241.28</mark>	<mark>\$ 2,308.48</mark>	<u>\$ 2,377.60</u>
65	\$2,080	\$2,145	\$2,210	<mark>\$2,276.30</mark>	<mark>\$ 2,344.55</mark>	<u>\$ 2,414.75</u>
66	\$2,112	\$2,178	\$2,244	<u>\$2,311.32</u>	<mark>\$ 2,380.62</mark>	<u>\$ 2,451.90</u>
67	\$2,144	\$2,211	\$2,278	<mark>\$2,346.34</mark>	<mark>\$ 2,416.69</mark>	<u>\$ 2,489.05</u>
68	\$2,244	\$2,244	\$2,312	<mark>\$2,381.36</mark>	<mark>\$ 2,452.76</mark>	<u>\$ 2,526.20</u>
69	\$2,208	\$2,277	\$2,346	<u>\$2,416.38</u>	<u>\$ 2,488.83</u>	<u>\$ 2,563.35</u>
70	\$2,224	\$2,310	\$2,380	<mark>\$2,451.40</mark>	<mark>\$ 2,524.90</mark>	<u>\$ 2,600.50</u>
71	\$2,272	\$2,343	\$2,414	<mark>\$2,486.42</mark>	<mark>\$ 2,560.97</mark>	<u>\$ 2,637.65</u>
72	\$2,304	\$2,376	\$2,448	<u>\$2,521.44</u>	<u>\$ 2,597.04</u>	<u>\$ 2,674.80</u>
73	\$2,336	\$2,40 9	\$2,482	<mark>\$2,556.46</mark>	\$ 2,633.11	<u>\$ 2,711.95</u>

Haines Borough Ordinance No. 22-09-629 Page 5 of 6

Length (feet)	10/1/19 to 9/30/20	10/1/20 to 9/30/21	10/1/21 to 9/30/22	<u>10/1/22 to</u> 9/30/23	<u>10/1/23 to</u> 9/30/24	<u>10/1/24 to</u> <u>9/30/25</u>
	(\$26/ft up to 40 ft; \$32/ft over 40 ft)	(\$27/ft up to 40 ft; \$33/ft over 40 ft)	(\$28/ft up to 40 ft; \$34/ft over 40 ft)	<u>(\$28.84/ft</u> up to 40 ft; <u>\$35.02/ft</u> over 40 ft)	<u>(\$29.71/ft up to</u> 40 ft; \$36.07/ft over 40 ft)	<u>(\$30.60/ft up to</u> <u>40 ft; \$37.15/ft</u> <u>over 40 ft)</u>
74	\$2,368	\$2,442	\$2,516	<mark>\$2,591.48</mark>	<mark>\$ 2,669.18</mark>	<mark>\$ 2,749.10</mark>
75	\$2,400	\$2,475	\$2,550	<mark>\$2,626.50</mark>	<u>\$ 2,705.25</u>	<u>\$ 2,786.25</u>
76	\$2,432	\$2,508	\$2,584	<mark>\$2,661.52</mark>	<u>\$ 2,741.32</u>	<u>\$ 2,823.40</u>
77	\$2,464	\$3,541	\$2,618	<mark>\$2,696.54</mark>	<mark>\$ 2,777.39</mark>	<u>\$ 2,860.55</u>
78	\$2,496	\$2,574	\$2,652	<mark>\$2,731.56</mark>	<mark>\$ 2,813.46</mark>	<mark>\$ 2,897.70</mark>
79	\$2,528	\$2,607	\$2,686	<mark>\$2,766.58</mark>	<u>\$ 2,849.53</u>	<u>\$ 2,934.85</u>
80	\$2,560	\$2,640	\$2,720	<mark>\$2,801.60</mark>	<mark>\$ 2,885.60</mark>	<u>\$ 2,972.00</u>
81	\$2,592	\$2,673	\$2,754	<mark>\$2,836.62</mark>	<mark>\$ 2,921.67</mark>	<mark>\$ 3,009.15</mark>
82	\$2,624	\$2,706	\$2,788	<mark>\$2,871.64</mark>	<u>\$ 2,957.74</u>	<u>\$ 3,046.30</u>
83	\$2,656	\$2,739	\$2,822	<mark>\$2,906.66</mark>	<mark>\$ 2,993.81</mark>	<mark>\$ 3,083.45</mark>
84	\$2,688	\$2,772	\$2,856	<mark>\$2,941.68</mark>	<mark>\$ 3,029.88</mark>	<mark>\$ 3,120.60</mark>
85	\$2,720	\$2,805	\$2,890	<u>\$2,976.70</u>	<u>\$ 3,065.95</u>	<u>\$ 3,157.75</u>
86	\$2,752	\$2,838	\$2,924	<mark>\$3,011.72</mark>	<u>\$ 3,102.02</u>	<mark>\$ 3,194.90</mark>
87	\$2,784	\$2,871	\$2,958	<mark>\$3,046.74</mark>	<mark>\$ 3,138.09</mark>	<u>\$ 3,232.05</u>
88	\$2,816	\$2,904	\$2,992	<u>\$3,081.76</u>	<u>\$ 3,174.16</u>	<u>\$ 3,269.20</u>
89	\$2,848	\$2,937	\$3,026	<mark>\$3,116.78</mark>	<u>\$ 3,210.23</u>	<u>\$ 3,306.35</u>
90	\$2,880	\$2,970	\$3,060	<mark>\$3,151.80</mark>	<mark>\$ 3,246.30</mark>	<mark>\$ 3,343.50</mark>
91	\$2,912	\$3,003	\$3,094	<mark>\$3,186.82</mark>	<u>\$ 3,283.37</u>	<u>\$ 3,380.65</u>
92	\$2,944	\$3,036	\$3,128	<mark>\$3,221.84</mark>	<mark>\$ 3,318.44</mark>	<u>\$ 3,417.80</u>
93	\$2,976	\$3,069	\$3,162	<mark>\$3,256.86</mark>	<mark>\$ 3,354.51</mark>	<mark>\$ 3,454.95</mark>
94	\$3,008	\$3,102	\$3,196	<mark>\$3,291.88</mark>	<u>\$ 3,390.58</u>	<u>\$ 3,492.10</u>
95	\$3,040	\$3,135	\$3,230	<mark>\$3,326.90</mark>	<u>\$ 3,426.65</u>	<u>\$ 3,529.25</u>
96	\$3,072	\$3,168	\$3,264	<mark>\$3,361.92</mark>	<mark>\$ 3,462.72</mark>	<mark>\$ 3,566.40</mark>
97	\$3,104	\$3,201	\$3,298	<mark>\$3,396.94</mark>	<mark>\$ 3,498.79</mark>	<u>\$ 3,603.55</u>
98	\$3,136	\$3,23 4	\$3,332	<mark>\$3,431.96</mark>	<mark>\$ 3,534.86</mark>	<mark>\$ 3,640.70</mark>
99	\$3,168	\$3,267	\$3,366	<mark>\$3,466.98</mark>	\$ 3,570.93	\$ 3,677.85

Letnikof Cove Harbor

Letnikof Cove Annual Moorage Rates* April- September

Length of Vessel	Annual Fee
Up to 25 feet	\$350.00
26 to 40 feet	\$475.00
41 feet and over	\$600.00

Haines Borough Ordinance No. 22-09-629 Page 6 of 6

Letnikof Cove Harbor (continued)

* Permit holder would receive a moorage sticker after paying the required fee in advance. They would have unlimited access only to the Letnikof Harbor and moorage would still only be available during the summer months. Moorage at Letnikof would remain on a "first come, first served" basis.

Special Joint-Use Permit Fees

(only for boat harbor stall renters who are paid current)

Length of Vessel	Annual Fee
Up to 40 feet	\$100.00 <mark>\$103.00</mark>
41 to 60 feet	\$125.00 <mark>\$128.75</mark>
61 to 80 feet	\$150.00 <mark>\$154.50</mark>
Over 80 feet	\$200.00 <mark>\$206.00</mark>

Live-Aboard Fees for Letnikof Cove Harbor

1 to 14 days	No charge other than the regular moorage rate
Over 14 days	\$ 35.00 <u>\$ 40.00</u> per month in addition to the
(prohibited October 15 th to April 1 st)	regular moorage rate

Letnikof Transient Boat Moorage Fees

Short-term (less than two weeks)	\$.65
Long-term (two weeks and longer)	\$6.50

ADOPTED BY A DULY CONSTITUTED QUORUM OF THE HAINES BOROUGH ASSEMBLY THIS _____ DAY OF OCTOBER, 2022.

Douglas Olerud, Mayor

ATTEST:

Alekka Fullerton, CMC, Borough Clerk



Haines Borough BOROUGH ASSEMBLY ACTION REQUEST

- DATE: Aug 25, 2022
- TO: Borough Assembly
- FROM: Port and Harbor Advisory Committee

ACTION:

Motion: Lapham moved, Hughes seconded, PHAC recommends the Harbor Fee Schedule changes as attached; Motion passed w/ no one opposed.

RATIONALE:

We are striving to keep increases consistent and easy. We feel that a percent increase is fair and less of an accounting difficulty.

BOARD REQUEST:

For the Haines Borough Assembly to approve the changes made to the Harbor Fee Schedule, as presented.

SUBMITTED BY: Diana Lapham, Chair Port and Harbor Advisory Committee

Haines Borough – Harbor Fee Schedule

(See HBC Title 16 for regulations and more information)

Portage Cove Small Boat Harbor

Transient Boat Moorage Fees 5.	" annual increase
Short-term (less than two weeks) 💋 . 68	\$-65 per lineal foot per day*
Long-term (two weeks and longer) $\frac{36.33}{5}$	\$6.50 per lineal foot per month*

Winter Rates – From October 1 through March 31, transient moorage vessels qualify for winter rates: one-half the amount that the vessel would ordinarily be charged. To qualify vessel must be up-to-date on all Port and Harbor Fees and pay for the winter moorage rate in advance.

* Transient moorage which must be invoiced will be charged at twice the normal rate (HBC 16.16.050).

Boat Launch Ramp Use Fees

Annual Launch Ramp Sticker – Discounted Rate Does Not Apply For Commercial Use	\$60.00 first trailer per year \$30.00 each for second, third, & fourth trailers per year (same owner) with proof of ownership \$60.00 each additional trailer per year
Daily Launch Ramp Use Fee	\$15.00 per trailer per day

Live-Aboard Fees for the Portage Cove Small Boat Harbor

1 to 14 days	No charge other than the regular moorage rate
Over 14 days 3 80 (prohibited October 15 th to April 1 st)	\$75.00 per month in addition to the regular moorage rate

Boat Houses, Floats, Scows, Barges, Pile Drivers, and Dredges

These vessels are allowed only on a temporary basis at the harbormaster's discretion and, when allowed, are assessed a fee twice the rate of other vessels. Note: Living aboard is not allowed on these vessels and watercraft.

Electrical Connection and Service Rates

Connection Costs (permanent stalls or open moorage) Paid to the local utility	Actual cost of connection (boat owner makes the arrangements with the local utility and pays them directly)
Service Fee (all vessels utilizing electrical service) Paid to the borough	\$10.00 per day
Service Fee (utilizing 240 service)	\$20.00 per day
Annual Service Maintenance fee (if Electrical service is activated)	\$15.00 annual

Harbor Crane Use

Annual Crane Use Permit (not available for transient vessels)	\$ 60	\$50.00 per year
Transient Vessels		\$20.00 per hour

Portage Cove Small Boat Harbor (Con't)

Miscellaneous Fees

iscellaneous Fees	Match Daily Medrage Rak
Grid Use 🛛 😽 , 💪 🎖	\$15.00 min or \$ 50 per ft per tide cycle
Harbor Staff Labor (regularly scheduled duty hours)	\$40.00 per hour (one hour minimum)
Harbor Staff Labor (After regular hours)	\$80.00 per hour (one hour minimum)
Tie-up Fee	\$20.00 per time plus cost of line (Harbor staff time not included)
Fee to Move a Boat	\$50.00 per time
Waitlist Fee	\$25.00 (first year) and \$10.00 per year thereafter
Seniority Transfer list Fee	\$10.00 per year
Pump Fee	\$20.00 per hour
Portable Electric Generator Fee	\$20.00 per day
Upland Storage Fee 🦉 40	\$ 20 per square foot per month
Vessel Storage Fee (current-annual stall renters-with all fees paid in advance)	\$.10 per square foot per month
Boat Stand Fee	\$2.50 per month or portion of a month

Hazardous Waste Disposal

Motor Oil	Included in moorage fees
Hydraulic Oil	Included in moorage fees
Diesel	Included in moorage fees
Gasoline	Included in moorage fees
Antifreeze	Included in moorage fees
Oil filters and Oiled Rags	Included in moorage fees
Paint	Included in moorage fees
Hazardous Materials Not Listed From Vessel	Included in moorage fees

Permanent Open Moorage Vessels

Long-term vessels with no permanent stall that meet the conditions outlined in HBC 16.16.105(A) shall qualify for rates equal to the regular moorage rate for their size vessel (see #14 - Annual Moorage Rates). If the conditions are not met, transient rates apply (see #1 - Transient Boat Moorage Fees).

Permanent stall renters that have a second commercial vessel and meet the conditions outlined in HBC 16.16.105(A) shall qualify for rates equal to the regular moorage rate for their size vessel (see #14 - Annual Moorage Rates) provided the permanent stall fee is paid. If the conditions are not met, transient rates apply (see #1 - Transient BoatMoorage Fees), and from April 1st through September 30th these vessels shall be charged the long-term transient moorage rate.

Commercial Vessel Per-Passenger Fee

Any commercial passenger vessel utilizing harbor facilities shall pay a per-passenger fee for all embarking and disembarking passengers. Transient harbor moorage rates apply to any vessel moored for over 2 hours

vessel moored for over 3 nours.	K	eep 5%	annual 11	ninese
Per-Passenger Fee	2020	2021	2022	2023
Loading/Unloading				
5% Annual Increase	\$.50	\$.53	\$.55	\$.58

Permanent stall renters that operate a commercial passenger vessel with a max capacity of 50 passengers and meet the conditions outlined in HBC 16.16.105(A) shall be exempt from paying the Per Passenger Fee.

Portage Cove Small Boat Harbor (Con't)

Annual Moorage Rates

				21	Accest	The store
Length	10/1/19 to 9/30/20	10/1/20 to 9/30/21	10/1/21 to 9/30/22	34.	HAR well	LACREASE
(feet)	(\$26/ft up to 40 ft; \$32/ft over 40 ft)	(\$27/ft up to 40 ft; \$33/ft over 40 ft)	(\$28/ft up to 40 ft; \$34/ft over 40 ft)	135.0.	2	
up to 25	\$650	\$675	\$ 721 \$200			
26	\$676	\$702	\$728			
27	\$702	\$729	\$756			
28	\$728	\$756	\$784			
29	\$754	\$783	\$812			
30	\$780	\$810	\$840			
31	\$806	\$837	\$868			
32	\$832	\$864	\$896	-		
33	\$858	\$891	\$924			
34	\$884	\$918	\$952			
35	\$910	\$945	\$980			
36	\$936	\$972	\$1,008			
37	\$962	\$999	\$1,036	-		
38	\$988	\$1026	\$1,064			
39	\$1014	\$1053	\$1,092			
40	\$1040	\$1080	\$1,120			
41	\$1312	\$1353	\$ 1,435,84\$1,394			
42	\$1344	\$1386	\$1,428	-		
43	\$1376	\$1419	\$1,462			
44	\$1408	\$1452	\$1,496			
45	\$1440	\$1485	\$1,530	-		
46	\$1472	\$1518	\$1,564	-		
47	\$1504	\$1551	\$1,598	-		
48	\$1536	\$1584	\$1,632			
49	\$1568	\$1617	\$1,666	-		
50	\$1600	\$1650	\$1,700	-		
51	\$1632	\$1683	\$1,734	-		
52	\$1664	\$1716	\$1,768	-		
53	\$1696	\$1749	\$1,802	-		
54	\$1728	\$1782	\$1,836			
55	\$1760	\$1815	\$1,870			
56	\$1792	\$1848	\$1,904	-		
57	\$1825	\$1881	\$1,938	-		
58	\$1856	\$1914	\$1,972	-		
59	\$1888	\$1947	\$2,006	-		
60	\$1920	\$1980	\$2,040			
61	\$1952	\$2013	\$2,074			

Longth	10/1/19 to 9/30/20	10/1/20 to 9/30/21	10/1/21 to 9/30/22
(feet)	(\$26/ft up to 40 ft;	(\$27/ft up to 40 ft; \$33/ft over 40 ft)	(\$28/ft up to 40 ft; \$34/ft over 40 ft)
	452,110 over 10 10)		
62	\$1984	\$2046	\$2,108
63	\$2016	\$2079	\$2,142
64	\$2048	\$2112	\$2,176
65	\$2080	\$2145	\$2,210
66	\$2112	\$2178	\$2,244
67	\$2144	\$2211	\$2,278
68	\$2244	\$2244	\$2,312
69	\$2208	\$2277	\$2,346
70	\$2224	\$2310	\$2,380
71	\$2272	\$2343	\$2,414
72	\$2304	\$2376	\$2,448
73	\$2336	\$2409	\$2,482
74	\$2368	\$2442	\$2,516
75	\$2400	\$2475	\$2,550
76	\$2432	\$2508	\$2,584
77	\$2464	\$3541	\$2,618
78	\$2496	\$2574	\$2,652
79	\$2528	\$2607	\$2,686
80	\$2560	\$2640	\$2,720
81	\$2592	\$2673	\$2,754
82	\$2624	\$2706	\$2,788
83	\$2656	\$2739	\$2,822
84	\$2688	\$2772	\$2,856
85	\$2720	\$2805	\$2,890
86	\$2752	\$2838	\$2,924
87	\$2784	\$2871	\$2,958
88	\$2816	\$2904	\$2,992
89	\$2848	\$2937	\$3,026
90	\$2880	\$2970	\$3,060
91	\$2912	\$3003	\$3,094
92	\$2944	\$3036	\$3,128
93	\$2976	\$3069	\$3,162
94	\$3008	\$3102	\$3,196
95	\$3040	\$3135	\$3,230
96	\$3072	\$3168	\$3,264
97	\$3104	\$3201	\$3,298
98	\$3136	\$3234	\$3,332
99	\$3168	\$3267	\$3,366

Letnikof Cove Harbor

Letnikof Cove Annual Moorage Rates* April- September

.p	5% annual increase
Length of Vessel	Annual Fee
Up to 25 feet	\$350.00 \$ 367.50
26 to 40 feet	\$475.00 \$ 498.75
41 feet and over	\$600.00 \$ 630.00

* Permit holder would receive a moorage sticker after paying the required fee in advance. They would have unlimited access only to the Letnikof Harbor and moorage would still only be available during the summer months. Moorage at Letnikof would remain on a "first come, first served" basis.

Special Joint-Use Permit Fees

(only for boat harbor stall renters who are paid current)

Length of Vessel	Annual Fee
Up to 40 feet	\$100.00 🖪 /03.00
41 to 60 feet	\$125.00 \$ 128.75
61 to 80 feet	\$150.00 # 154, 50
Over 80 feet	\$200.00 \$ 206.00

Live-Aboard Fees for Letnikof Cove Harbor

1 to 14 days	No charge other than the regular moorage rate
Over 14 days	\$35.00 per month in addition to the regular
(prombiled October 15** to April 1**)	moorage rate

Ļ	etnikof Transient Boat Moorage Fees	Some as Portage Cove.	Harbor
	Short-term (less than two weeks) 🏻 💋 . 68	\$-65 per lineal foot per day	
l	Long-term (two weeks and longer) 🔏 6.83	\$6.50 per lineal foot per month	



11B3

Assembly Agenda Bill

Agenda Bill No.: 22-1194

Assembly Meeting Date: 09/13/22

Business Item Description:	Attachments:
Subject: Amending Intent of Mud Bay Planning/Zoning	1. Ordinance 22-09-630
District	
Originator:	
GASC	
Originating Department:	
Assembly	
Date Submitted: 08/02/22	

Full Title/Motion:

Motion: Introduce Ordinance 22-09-630 and scheduled it for its first public hearing 9-27-22 and its second public hearing 10-11-22.

Administrative Recommendation:

Fiscal Impact:			
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets
\$	\$	\$ 0	Reduced maintenance costs

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Consistent:	Yes	
001131310111		

Yes 🗆 No

Summary Statement:

This Ordinance is supported by the Government Affairs and Services Committee. The Haines Borough Code currently includes Intent language for the Mud Bay Planning/Zoning District which is confusing. While "Intent" provisions are not enforceable, it is confusing to have a code section that implies that a community of people within the political unit of the Haines Borough have the right to self-determination.

Referral:

Referred to: Recommendation: Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 09/1322

Public Hearing Date(s): 9/13/22 and 9/27/22 Postponed to Date:

Draft

An Ordinance of the Haines Borough Assembly amending Haines Borough Code Title 18 Intent of Mud Bay Planning/Zoning District.

BE IT ENACTED BY THE HAINES BOROUGH ASSEMBLY:

Section 1. <u>Classification</u>. This ordinance is of a general and permanent nature and the adopted amendment shall become a part of the Haines Borough Code.

Section 2. <u>Severability</u>. If any provision of this ordinance or any application thereof to any person or circumstance is held to be invalid, the remainder of this ordinance and the application to other persons or circumstances shall not be affected thereby.

Section 3. <u>Effective Date</u>. This ordinance is effective upon adoption.

Section 4. <u>Amendment of Section 18.70.030(B)(1).</u> Section 18.70.030(B)(1) of the Haines Borough Code is hereby amended, as follows:

NOTE: **Bolded**/<u>UNDERLINED</u> ITEMS ARE TO BE ADDED STRIKETHROUGH</u> ITEMS ARE DELETED

18.70.030 Zoning Districts – Zones

...

B Mud Bay Planning/Zoning District

1. Intent. The intent of this district is to preserve the existing-zoning provisions of the former land use service area No. 1 (Mud Bay) <u>as a rural residential zone</u> and for the people of the Mud Bay community to preserve their lifestyle, community scale, self-sufficiency, self-determination, and the basic rights of health, safety and welfare. This is in accordance with the goals and objectives of the former Mud Bay land use service area board which were developed from the Haines Borough comprehensive plan.

ADOPTED BY A DULY CONSTITUTED QUORUM OF THE HAINES BOROUGH ASSEMBLY THIS _____ DAY OF ______, 2022.

ATTEST:

Douglas Olerud, Mayor

Alekka Fullerton, CMC, Borough Clerk

11**C**1



HAINES BOROUGH, ALASKA P.O. BOX 1209, HAINES, ALASKA 99827 Annette Kreitzer, Borough Manager 907.766.6404 akreitzer@haines.ak.us

MEMO

TO:	Haines Borough Assembly
	Mayor Doug Olerud

DATE: September 8, 2022

RE: Building Resilient Infrastructure & Communities (BRIC)

Twenty-one months following the December 2020 weather event that heavily impacted Borough residents and property, the options for residents affected by the slides are clearer. Only the Borough can apply for compensatory relief on behalf of the qualifying residents. Residents, on their own, cannot apply for BRIC or Hazard Mitigation Grant Program (HMGP) funding for home buy-outs, for instance.

The Department of Natural Resources held a public meeting August 10, 2022 at the Chilkat Center to explain options available for residents to apply for under the Grants of Land after a Natural Disaster. Those options are included in this memo.

As we learned in the October 13, 2021, Committee of the Whole meeting (BRIC/HGMP Overview), any assistance for residents under the BRIC program required the Borough to have an updated Hazard Mitigation Plan (HMP). Unfortunately, delays at the federal level in awarding the grant, which included the Haines Borough HMP update, resulted in the Borough not being in a position to apply for BRIC earlier than now.

Much of the assistance from the State and Federal governments rests on whether the landslides are considered a continuing imminent risk or whether the focus shifts to mitigation of any future slide. The receipt of the Landslide Technologies report was designed to help answer this question. Additionally, the amount of funds available to the State for the HMGP program is contingent upon previous disasters. The 2020 Disaster DR 4585 AK has helped to significantly increase the funding currently available to the State for the HMGP Program, making this funding stream more plausible for projects in Haines. The State has approved the HMP, and the Federal Emergency Management Agency's (FEMA) 45-day review period ends September 25. We expect release of the Plan at that time. The Assembly could still move forward with a BRIC and/or HGMP application, if it chooses to do so.

Any disaster relief programs, whether at the federal or state level, are limited by their enabling statutes. We cannot shape these programs into something they are not. As we have learned, these programs have severe limitations.

Keeping options open for residents has been the Borough's main approach to the disaster response. Disaster recovery for local governments is somewhat less complicated than recovery for private landowners. As Sitka, Skagway and other municipalities moving through disaster recovery have found, the process is methodical and slow. Funds can only be used for specific items such as mapping, something not available to residents.

As residents received Individual Assistance from the State following the disaster, the Borough was constrained by not wanting to affect that assistance. For instance, the sign stating that Beach Road was closed to through traffic stayed up until all Individual Assistance payments had been exhausted. Federal programs like BRIC and HMGP require that there be an imminent threat. The best way to establish that was to allow the monitoring under the Landslide Technologies contract. Waiting for the monitoring data to better inform the Borough and residents of options available to them has been frustrating for all concerned.

REQUIREMENT	BRIC	HMGP	LAND SWAP/DNR
Primary	Must be	Must be	Primary residence,
Residence	primary	primary	business, or
	residence	residence	commercial
Structure on	Structure	Structure	Structure must
Property	must have	must have	have been present
	been present	been present	before slide
	before slide	before slide	
Structure	Damaged or	Damaged or	Does not require
Structure Damage	Damaged or intact	Damaged or intact	Does not require the structure to
Structure Damage	Damaged or intact properties	Damaged or intact properties	Does not require the structure to have been
Structure Damage	Damaged or intact properties qualify	Damaged or intact properties qualify	Does not require the structure to have been damaged, only the
Structure Damage	Damaged or intact properties qualify	Damaged or intact properties qualify	Does not require the structure to have been damaged, only the land.
Structure Damage Financial	Damaged or intact properties qualify Mortgage	Damaged or intact properties qualify Mortgage	Does not require the structure to have been damaged, only the land. Must be no
Structure Damage Financial Requirement	Damaged or intact properties qualify Mortgage must be	Damaged or intact properties qualify Mortgage must be	Does not require the structure to have been damaged, only the land. Must be no mortgage or lien on

WHAT ARE THE OPTIONS?

Deconstruction Cost	paid through buy-out; no liens on property; property taxes must be current Based on cost benefit analysis	paid through buy-out; no liens on property; property taxes must be current Based on cost benefit analysis (CBA)	if the land is quitclaimed to the State. No Deconstruction cost
	isn't cost effective, likely won't qualify for BRIC	cost effective, likely won't qualify for HMGP	
Rebuilding	If the Borough were to buy- out the property, no rebuilding would be allowed per HBC18.60.010 Q(2,3) – Based on cost benefit analysis – if CBA isn't cost effective, likely won't qualify for BRIC. The Borough would be responsible for perpetual green/open space.	If the Borough were to buy- out the property, no rebuilding would be allowed per HBC18.60.010 Q(2,3) – Based on cost benefit analysis – if CBA isn't cost effective, likely won't qualify for HMGP. Borough would be responsible for perpetual green/open space.	Land must be unusable and will be conveyed to the State or Borough
COST SHARE	Borough responsibility: 25% of total cost	No cost share for Borough (State covers cost)	No cost share for Borough

BRIC and HMGP (404)

Both programs are administered under the aegis of the Federal Emergency Management Agency (FEMA). HMGP has a more limited pool of money, although it has been increased to \$1,000,000 for FY23. We would be competing against other Alaskan communities which have also had disasters (*a 500-1,000-year rain event/Girdwood area, Massive snowfall in Delta & Greely, Copper River, Fairbanks North Star Borough, Nenana, and the Denali Borough; 90 mph+ winds in the Mat-Su Borough flip planes and tip semis, knocking power out to some 20,000 residents during frigid temperatures, etc.*). BRIC is a nationally competitive grant, with a larger pool of funds. We may still be at a competitive disadvantage owing to the number of disasters in the Lower 48 recently. According to our contacts within the State Division of Homeland Security-Emergency Management, Alaska doesn't score as well on federal projects due to a lack of enforceable of building codes.

HOW WOULD A BUY-OUT PROGRAM BE ADMINISTERED?

The Borough can apply for a Scoping and Planning Capacity and Capability Building grant through HMGP. This would enable us to develop a "buy-out" program which could be submitted under BRIC or under the State HMGP.

A buy-out program is what it sounds like – government purchasing a home that is within a risk area – that will not be rebuilt. The program is voluntary, no one can be forced to participate in a buy-out program.

Under BRIC, the Borough would need to match 25% of the cost of deconstructing the homes. The real cost of this program is that the homes must be dismantled piece by piece. None of the deconstructed elements may be re-used, sold or traded – they are disposed of according to FEMA requirements. The property then becomes a green space, in perpetuity. It is off tax rolls and not available for future development or other use besides recreational. With HMGP, the State pays the cost share and there isn't a cost to the Borough. Since the program is administered by the State, it can be a shorter wait time for residents (3-4 years vs up to 8 years through BRIC).

What is the potential financial exposure to the Borough and taxpayers?

A licensed appraiser determines the value of the home up to the fair market value PRIOR to the disaster. BRIC requires the Borough to match 25% of whatever the market value is of the homes. FEMA must engineer the deconstruction of the home (adding to cost). Using a benefit/cost analysis, FEMA will determine if it costs more to deconstruct the home than the home is worth, and may decline to offer the buy-out under either BRIC or HMGP.

POTENTIAL NEXT STEPS

The Assembly could direct the Borough Manager to gauge affected property owner's interest in either BRIC or HMGP. This could be accomplished by sending letters setting out the above criteria to property owners affected by the December 2020 Weather Event and asking if there is interest in pursuing either approach.

Once the Assembly knows how many homeowners are interested in either of these approaches, it can better determine whether the Borough should or could bear the 25% cost share under BRIC or pursue progress under HMGP. BRIC must be pursued within 5 years of the disaster, and HMGP must be pursued within 7 years of the disaster.

To pursue progress under HMGP, we could apply for funding for a Capabilityand Capacity-Building ("C&CB") grant under the State HMGP. This would allow us to scope out a potential buy-out project and be prepared to apply for HMGP funding to carry out the project.

<u>Proposed Motion</u>: Direct the Manager to gauge the interest in BRIC, HMGP or State Land Swap of property owners with qualifying property from the December 2020 Weather Event.


11C2

Assembly Agenda Bill

Agenda Bill No.: 22-1217

Assembly Meeting Date: 09/13/2022

Business Item Description:	Attachments:
Subject: Non-Profit Grant Funding	 Non-Profit Funding Application Haines Friends of Recycling (HFR) application Lynn Canal Broadcasting (KHNS) application
Originator:	4. Southeast Alaska Independent Living (SAIL) application
Assembly	5. Haines Senior Center (Southeast Senior Services/CCS)
Originating Department:	application
Government Affairs and Services Committee	
Date Submitted:	
9/6/22	

Full Title/Motion:

GASC Proposed Motion: All requests (HFR, SAIL, KHNS, Senior Center) be funded and the balance of funds (\$2,810) be available for application as overage for capital projects.

Administrative Recommendation:

Fiscal Impact:						
Expenditure Required	Amount Budgeted	Appropriation Required	Projected Impact to Future Operating Budgets			
\$0	\$0	\$ 0				

Comprehensive Plan Consistency Review:

Comp Plan Goals/Objectives:

Consistent: Yes

s □No

Summary Statement:

On September 6, 2022, the GASC committee met and reviewed all of the FY23 Non-profit Grant funding applications. They found that all applications were in order and fully funding all applications was within the \$38,000 appropriated by the Assembly for this purpose. GASC, therefore, recommended fully funding:

*Requested \$7,190 to Haines Friends of Recycling for community scrap metal recycling program

*Requested \$5,000 to KHNS for a generator

*Requested \$8,000 to SAIL for operational support

*Requested \$15,000 to Haines Senior Center for support of senior lunches

Further, the GASC committee recommended the balance of the appropriated funds (\$2,810) be available for application as overage for capital projects.

Referral:

Referred to: Recommendation: Referral Date:

Meeting Date:

Assembly Action:

Meeting Date(s): 9/13/22

Public Hearing Date(s): Postponed to Date:

Haines Borough



GRANTS FOR NON-PROFIT ORGANIZATIONS

APPLICATION MATERIALS FOR FISCAL YEAR 2023

Complete grant applications to be returned to the Borough Clerk's Office

<u>Grant Deadline</u> – September 1, 2022 5:00 PM

For further information, please contact Alekka Fullerton, Borough Clerk Haines Borough Box 1209 (907) 766-6402 <u>afullerton@haines.ak.us</u>

GRANTS FOR NONPROFIT ORGANIZATIONS FROM THE HAINES BOROUGH

PROCEDURES

Applications are due in the Haines Borough Clerk's office by 5 PM, Thursday, September 1, 2022.

Please submit the following:

- Summary Sheet;
- Elements A through I from Table A;
- "Required Documentation" as described in Item I from Table A; Please supply the required documentation as a separate, clearly labeled attachment so that staff may easily ascertain that each required document is supplied.
- Final Report of FY22 grant monies received, if any.

Table A outlines the rating schedule for grant applications from non-profits within the Haines Borough. Please limit your response to each element to the number of pages specified with the exception of "Required Documentation."

Refer to Table B for an explanation of "Grant Category" and for total available amounts of grant funds.

All applications will be reviewed for qualification by Borough staff and scored by the Government and Services Committee before being submitted to the Borough Assembly for funding approval.

Scheduled date for Assembly approval: September 13, 2022.

SUMMARY SHEET

Name of Organization:	
Name of Contact Person:	
Phone:	Email:
Mailing address:	
Grant Category:	Fund 01- General Fund- \$38,000 total funds to be granted
Dollars Requested:	
Brief Description of the Pu	rpose of the Grant:
I,	, do hereby certify that all information provided for this
grant application is accurat	e and complete to the best of my knowledge.
Name:	
Title:	Date:



Date Scored:

FY23 HAINES BOROUGH Non-Profit Grants Table A.

Organization:

Please submit each category on its own page for scoring purposes.

Category	Description	Maximum Page Length	Total Points Possible	
A. Detailed Plan for Funds	 Describe what will be done with the funds: Who will oversee the project? Who will be served? When will the service be provided? Explain reasoning for grant category. Outcome - describe the tangible community benefit. Outcome - how will you measure success? 	1	10	
B. Statement of Need	 Explain how the proposed need you has been documented: Does the funding from this request help leverage other funds for the project? How does the request align with the goals and objectivesin the Haines Borough Comprehensive Plan? (See https://www.hainesalaska.gov/cp2025) 	1	10	
C. Organizational Capacity	 Describe the organization's history and finance capacity: Board qualifications & involvement(s); List of grants received or pending for prior andcurrent years; % of revenue from the Borough; Other financing secured/committed. 	1	10	
D. Contribution(s) to Economic Development	 Explain how funds will be used to: Stimulate the Haines economy; Address Haines Borough Comprehensive Plan Goals and Objectives. 	1	5	
E. Health & Safety	 Describe how the funds will provide health and/or safety education or services: Will funds provide essential services for children, elderly, disabled, or other populations in Haines? If yes, please explain. How does the request address Haines Borough Comprehensive Plan Goals and Objective? 	1	5	

F. Broad Community Support	 Provide evidence of broad community support: Number of members/beneficiaries served in previous year - in the last 3 years; Petitions/letters of support; Other evidence. 	1	5	
G. Budget	 Provide a rationale for the budget request: Is the project budget appropriate to meet the need? How does the need justify the expenditure? 	1	5	
H. Timely Submission	Application is complete and on time.	N/A	N/A	
I. Required Documentation	 Copy of IRS 990 form and audit, if applicable. Copy of non-profit documentation (IRS). Current State of Alaska and Haines Borough business license. Minutes of meeting where the governing body approved grant request. Organization structure including names of the board of directors and the executive director if applicable. If funded in FY22, written report of use of funds. Profit and Loss for previous year. Detailed budget for current year, including fundsfor this project. 	N/A	N/A	
Total Score (50 possible)			

Applications receiving an average score of 30 points or fewer are eliminated from funding.

Table B.

Grant Category and Description	Amount Budgeted for Appropriations from Assembly for FY23
General Fund (Fund 01): Description: Revenue for this fund will support programs and projects that aim to enhance the quality of life for community members in response to an action plan described in the Haines Borough Comprehensive Plan.	\$ 106,000
<u>Allocations already made</u> : Becky's Place SEAKFair Alaska Avalanche Association	<24,000> <20,000> <24,000>
Remaining Balance for Non-Profits to be awarded on a competitive basis	\$38,000

Description of Grant Application Procedures, Requirements & Restrictions

- Complete applications must be received in the Municipal Clerk's office by the due date listed on the cover of this application.
- All complete applications received timely shall be reviewed and scored by the Government and Services Committee prior to being submitted to the Borough Assembly for approval.
- In order to be considered, applicants must be current on all Haines Borough liabilities including sales taxes, property taxes, utility payments, lease payments, loan payments, promissory notes, etc.
- Grant monies shall not be used to pay for any tax debt (federal, state, or municipal).
- Grant monies not awarded or used during the fiscal year will rollover as nonprofit grant funds for the next fiscal year. Any rollover monies will be included in the total grant funds appropriated for the next fiscal year in any of the grant categories.

If your organization received funding in FY22, make sure you have submitted your final report regarding expenditures of the FY22 grant monies either prior to, or along with your FY23 Grant Application. Failure to timely submit reports may result in the nonprofit organization being ineligible in the next fiscal year for grant appropriations.



SUMMARY SHEET

AUG 2 6 2022

Name of Organization: Haines Friends of Recycling

Name of Contact Person: Melissa Aronson

Phone: 766-2185 Email: aronson@aptalaska.net

Mailing address: PO Box 822, Haines

Grant category: fund 01 - general fund- \$38,000 total funds to be granted

Dollars Requested: \$7,190 to fill two 40 foot vans with scrap metal

Brief description of the Purpose of the Grant: This grant proposal is to provide for FREE scrap metal recycling for all households in the Borough for up to 100 pounds of scrap metal. The two scrap metal collections will be held in the summer of 2023.

I, Melissa Aronson, do hereby certify that all information provided for this grant application is accurate and complete to the best of my knowledge.

Name: Melissa Aronson

Medina fromm

Title: Chairperson, Haines Friends of Recycling

Date: August 25, 2022

HAINES BOROUGH

A. Detailed Plans for Funds

This project to collect and ship two 40-foot vans of scrap metal will be overseen by the Board, staff and volunteers of Haines Friends of Recycling. Each van holds approximately 14,000 pounds of scrap metal.

Every household in the Haines Borough will be able to participate by bringing up to 100 pounds of scrap metal for free to the Recycling Center on our collection days.

The scrap metal collection will be held on two Saturdays, one in early summer and one in late summer. The collection will be from 10 AM to 2 PM. We are seeking funding for the cost of preparing the scrap for shipping and for the cost of shipping to a metal recycling company in the Pacific Northwest. We are asking for funding for two 40-foot vans at \$3,595 each for a total of \$7,190. One van will be shipped in early summer and one in late summer for a total of approximately 28,000 pounds of scrap metal.

Haines Friends of Recycling is a 501(c)(3) that provides service to the community to reduce waste and promote recycling.

Scrap metal is one of the major litter problems in the Borough. Disposing of it at the landfill is expensive (27 cents a pound) and many people choose to just illegally dump metal rather than take it to the landfill. The tangible benefit of this grant is to provide every household in the Borough the option of bringing up to 100 pounds of scrap metal to the Haines Friends of Recycling Center for FREE. The outcome will be to reduce the amount of scrap metal litter in the Borough.

We will measure success by the number of households that participate in the FREE scrap metal recycling and by the total number of pounds of scrap metal that we send out for recycling.

B. Statement of Need

Haines Friends of Recycling (HFR) sends out between 200,00 and 300,000 pounds of recycling every year in six vans provided by Alaska Marine Lines for free. We do have to pay the "spotting charge" that moves the vans from the Port of Seattle to the recycling company in the Seattle area. Generally we send out vans of: (1) electronics, (2) plastic, (3) aluminum and steel cans, (4) cardboard and other fiber, (5) scrap metal, and (6) one other van of one of the above or fish nets or high value metal. We would like to serve the community further by sending out an additional two vans of scrap metal but need to be able to cover our costs of loading the van, shipping, and the spotting charge.

Scrap metal is one of the major sources of litter in the Borough. This grant will help get rid of some of the existing litter and prevent further litter by providing FREE scrap metal recycling to every household in the Borough.

This grant will align with the Borough ordinance 8.08.020 (Littering) as well as the Haines Borough Comprehensive Plan. The Comprehensive Plan in section 2 talks about the value of wildlife. Reducing litter, including metals, that can cause injuries to wildlife is important in their protection. Section 3 talks about the quality of life. Reducing the amount of litter is an important component in improving quality of life by protecting the outdoors and natural beauty (3.1.1). Reducing litter is also an important component of providing a beautiful area for our tourists. Providing lowincome household with ways to dispose of their scrap metal waste for free is an important component of a socially just quality of life. The scrap metal collection and loading will be done by our staff person and a large group of volunteers; as noted in 3.1.3, volunteerism is a hallmark of the Haines experience. Section 3.1.4 speaks to our economic challenges; providing FREE scrap metal recycling to all households will be one way to assist with economic challenges. The description of Haines Friends of Recycling (9.4.1) need to be updated in the Comp Plan, but does speak to our role in providing recycling services to the Borough. This grant will enhance our abilities to continue to serve the Chilkat Valley by reducing waste and promoting recycling.

C. Organizational Capacity

Haines Friends of Recycling, a 501(c)(3) established in 1999 is a highly successful non-profit. Our 9 person board is highly qualified and very active in our success. Each board member has a specialty area, including organizational management, facility maintenance and improvement, volunteer coordination, fish net recycling and marine debris cleanup, electronic recycling, social media etc. In addition to our very active board, we have 125 active volunteers that assist with recycling and waste reduction at major events (e.g. the Fishermen's BarBQ, SEAK Fair, July 4th, electronics recycling, scrap metal recycling, Earth Day, etc.). Our annual report is available on our website: hainesrecycle.org.

We currently have over 330 families, businesses, and organizations that are paying members of HFR. Many others use our recycling services and may leave donations but don't choose to be members.

We receive funding every year from the Chilkat Valley Community Foundation, most recently for banners for our recycling stations at major events and the weekly "Waste Reduction Tips" in the Chilkat Valley News. We are in the process of preparing a budget request for 2023 to the CVCF. Our most recent grant from the Borough for \$4,000 was in 2020 for fish net recycling that happened in the summer of 2021.

Our annual budget for 2022 is for \$57,000; we anticipate a budget of \$60,000 for 2023. Our funding is sustainable over time. Approximately one-third of our income comes from the sale of recyclable materials, one-third from memberships and donations, and one-third from grants and major gifts. The \$7,190 requested in this grant request would be approximately 12.6 percent of our budget for 2023.

D. Contributions to Economic Development

This grant will stimulate the Haines economy by addressing socioeconomic needs through making scrap metal recycling FREE to all households in the Borough. This will particularly benefit those households of limited means who might otherwise be tempted to illegally just dump their scrap metal in the woods or along rivers.

The grant addresses two areas of the Goals and Objectives of the Comprehensive Plan.

Goal 1 calls for "sustaining the quality of life that Haines residents rate as excellent based on outdoors and natural beauty". This grant will reduce the amount of ugly illegal littering of scrap metal because of the two FREE scrap metal recycling events, one in early summer and one in late summer.

Objective 15G: "Provide, or ensure provision of, solid and hazardous waste disposal in a manner that does not adversely impact air, land, and water quality. Promote a Reduce-Reuse-Recycle approach and efforts." This grant is totally in line with this objective and will provide for the shipping of two forty foot vans of scrap metal to a recycling company in the Pacific Northwest. Each scrap metal van holds roughy 14,000 pounds of recyclable material.

E. Health and Safety

Imagine walking along an ocean or river beach or hiking in the woods and stepping (or falling) on a sharp piece of metal and getting cut. Or your dog getting cut. Or a child. Because of the amount of dumped scrap metal in the Borough, that is a possible scenario. This grant will provide for FREE scrap metal recycling to Borough residents, with the result of less inappropriately discarded scrap metal. Cleaning up or preventing litter is a health and safety issue.

It is also a social justice issue. By making this scrap metal collection FREE with Borough support, people of limited means can send their scrap metal to be recycled at no cost to them.

As noted in "D" above, the grant addresses two areas of the Goals and Objectives of the Comprehensive Plan.

Goal 1 calls for "sustaining the quality of life that Haines residents rate as excellent based on outdoors and natural beauty". This grant will reduce the amount of ugly illegal littering of scrap metal because of the two FREE scrap metal recycling events, one in early summer and one in late summer.

Objective 15G: "Provide, or ensure provision of, solid and hazardous waste disposal in a manner that does not adversely impact air, land, and water quality. Promote a Reduce-Reuse-Recycle approach and efforts." This grant is totally in line with this objective and will provide for the shipping of two forty foot vans of scrap metal to a recycling company in the Pacific Northwest. Each scrap metal van holds roughy 14,000 pounds of recyclable material.

F. Broad Community Support

Haines Friends of Recycling serves everyone in the Borough that wants to bring us their recycling. At no cost. We have collection bins for aluminum cans, steel cans, #1 and #2 plastics, white paper, shredded paper, and newsprint. The collection bins are outside and open to the public 24/7/365. Other recyclable materials can be brought to us when staff is available (Thursdays and Fridays 9 to noon, Saturdays 10-2) for a modest fee. A full list of what we take for recycling is available on our website: hainesrecycle.org.

Because the collection bins are always available, we don't have an accurate count of how many people we serve. But we know it is more than the 327 + families, businesses, and organizations that are paying members.

Additionally we serve the entire community and tourists through our recycling and composting zero waste efforts at major events such as the SEAK Fair, the Fishermen's BarBQ, July 4th celebration and more.

We truly are a non-profit that serves everyone in the Haines Borough.

G. Budget

Amount requested:

For early summer van:		
Shipping of 40 foot van t	o Seattle:	\$2,300.
Spotting charge: SEA po	rt to recycling company	700.
Staff time (15 hours) to lo	bad van @\$30/hour	450.*
Ad in Chilkat Valley News	6	145
2	total:	\$3,595.
For late summer van:		
Shipping of 40 foot van t	o Seattle:	\$2.300.
Spotting charge: SFA po	rt to recycling company	700.
Staff time (15 hours) to lo	ad van @\$30/hour	450.*
Ad in Chilkat Valley News	8	145
	total:	\$3,595.
		+-,
	Total requested:	\$7190.
	Total requested:	\$7190.
In-kind:	Total requested:	<u>\$7190.</u>
In-kind: Preparing refrigerators/fr	Total requested:	\$7190.
In-kind: Preparing refrigerators/fr Volunteer hours on collec	Total requested: eezers to hold scrap	\$2,000** \$1,200***
In-kind: Preparing refrigerators/fre Volunteer hours on collec Twenty hours volunteer ti	Total requested: eezers to hold scrap ction days ime preparation time	\$2,000** \$1,200*** \$ 400
In-kind: Preparing refrigerators/fr Volunteer hours on collec Twenty hours volunteer ti Lunch and refreshments	Total requested: eezers to hold scrap ction days ime preparation time for volunteers, 2 days	\$2,000** \$1,200*** \$ 400 \$ 400
In-kind: Preparing refrigerators/fre Volunteer hours on collect Twenty hours volunteer to Lunch and refreshments Pocket magnets for volun	Total requested: eezers to hold scrap ction days ime preparation time for volunteers, 2 days nteers	\$7190. \$2,000** \$1,200*** \$ 400 \$ 400 \$ 60
In-kind: Preparing refrigerators/fre Volunteer hours on collect Twenty hours volunteer to Lunch and refreshments Pocket magnets for volun Social media posting volu	Total requested: eezers to hold scrap ction days ime preparation time for volunteers, 2 days nteers unteer	\$7190. \$2,000** \$1,200*** \$ 400 \$ 400 \$ 60 \$ 40 \$ 40
In-kind: Preparing refrigerators/fre Volunteer hours on collect Twenty hours volunteer to Lunch and refreshments Pocket magnets for volun Social media posting volu Flyer development and p	Total requested: eezers to hold scrap ction days ime preparation time for volunteers, 2 days nteers unteer printing	\$7190. \$2,000** \$1,200*** \$ 400 \$ 400 \$ 60 \$ 40 \$ 100
In-kind: Preparing refrigerators/fre Volunteer hours on collec Twenty hours volunteer to Lunch and refreshments Pocket magnets for volun Social media posting volu Flyer development and p	Total requested: eezers to hold scrap ction days ime preparation time for volunteers, 2 days nteers unteer printing	\$7190. \$2,000** \$1,200*** \$ 400 \$ 400 \$ 60 \$ 40 \$ 100 \$ 400 \$ 100

 * Includes overtime and payroll taxes
 ** We used refrigerators and freezers with the refrigerant evacuated to hold the scrap metal.

*** 15 volunteers for 4 hours both days @ \$20/hour **** Does not include rental of Recycling Center or other auxiliary costs

I. Required Documentation and notes

1. Copy of IRS 990 from FY 2021. Please note we are not subject to audit. Our fiscal year runs from January 1 through December 31. Please let us know if you need attachments beyond the enclosed cover sheet.

2. Copy of non-profit documentation included.

3. Current State of Alaska and Haines Borough business license included.

4. Minutes of August 2022 HFR Board meeting authorizing this grant request. See Item 5 J "Borough Grant Proposal" (page 4) which is highlighted in yellow.

5. Organizational structure. This is the names of our Board and other key volunteers as approved at our October 2021 annual meeting. Our next annual meeting is scheduled for September 17, 2022 and should only result in a few minor updates.

6. If funded in FY 22, written report of use of funds. The last grant we received from the Borough was for \$4,000 for fish net recycling, received in July 2021. A report of the activities is attached, as prepared by Molly Sturdevant for our Board in November 2021. Also attached is the budget summary for the grant.

7. Profit and loss statement for 2021 included.

8. Detailed budget for 2022. Please note that when we approved our 2022 budget in February 2022 we did not anticipate that the Borough would have grants for non-profits in 2022 and so did not budget for that. Should we be awarded this grant, the actual scope of work will occur in the summer of 2023.



KHNS Radio • Haines • Skagway • Klukwan

August 29, 2022

SUMMARY SHEET

Name of Organization: Lynn Canal Broadcasting, KHNS FM

Name of Contact Person: Kay Clements

Phone: 907-766-2020 x3

Email: gm@khns.org

Mailing address: KHNS POBox 1109, Haines, AK 99827

Grant Category: Fund 01 – General Fund - \$38,000 total funds to be granted

Dollars Requested: \$5,000

Brief Description of the Purpose of the Grant: The KHNS studios rely on emergency generator backups both at our transmitter site on FAA Road and at the Chilkat Center Studios. We are requesting funding to help replace our 20+ year old generator with a newer, more robust model to ensure that the radio station has electricity during outages, remaining on air able to broadcast emergency information in any event from short to extended outages.

, do hereby certify that all information provided for this grant applications is accurate and complete to the best of my knowledge.

Name: Kay Clements

Title: General Manager

Date: August 18th, 2022

FY 23 Haines Borough Non Profit Grants -Table A

A. Detailed Plan for Funds – Describe what will be done with the funds

KHNS is requesting funds to help with the replacement and installation of a new generator to back up the studio and offices of KHNS at the Chilkat Center, serving the community of Haines.

KHNS (a local 501c3) is the public broadcast station and part of the comprehensive emergency plan for Haines. Our studios, located in the Chilkat Center, are counted on to broadcast 24/7, especially in the event of an emergency. In order to fulfill this mandate, KHNS has two generators in Haines, one at our location on FAA road which houses our transmitter and one at the Chilkat Center which keeps the offices and studio running.

Both are essential to maintain programming but it is the generator at the Chilkat Center which is the focus of this grant. The current generator is over 20 years old and while continuing to function, has been less and less reliable over the last couple of years. We have put time, energy and funds into maintenance, recently building a small shelter to keep the elements at bay but this generator has outlived its service, can no longer be counted on and will need to be replaced.

In a power outage, whether as a result of a disaster or a simple outage, getting information to the public can make the difference for all concerned. As evidenced during the weather event and subsequent landslide of 2020, we work closely with our local fire & police departments, borough staff and social services to make sure accurate and timely information is disbursed. In the event of an ongoing power outage, it is critical for KHNS and the community to rely on the propane-fueled generator which is why we are requesting funding to help with this critical piece of equipment.

The KHNS manager, Kay Clements will be overseeing and disbursing funds. Marley Horner is our on-site engineer and will be participating in the installation and well as working with local technicians hired to assist.

When funds are awarded we will place the order and install as soon after delivery as possible, our guess is early fall or mid-winter, depending on supply chains and transport. Meanwhile, we will continue to have generator coverage with the help of local mechanics.

The tangible community benefits are explained above but to reiterate, as part of the Haines Borough Emergency Plan, KHNS needs to have a reliable generator to ensure the uninterrupted flow of information no matter the circumstances. Success is measured by our ability to be functional and continue broadcasting, delivering accurate and information as it is relayed to us as well as the critical information which often follow emergency and/weather events. There are a number of needs in, during and following a special event and having a reliable broadcast source that can reach the people in the simplest most dependable way possible is our mission. B. Statement of Need – Explain how the proposed need you have has been documented. Documented evidence of need is recorded in increased parts and maintenance costs over the past couple of years. We believe, as a result of having to replace different systems that we will only be spending more time and expense on fixing after-the-fact problems. The wear and tear of weather is another ingredient and what encouraged us to build a shelter at the site in preparation for a new generator.

As a small staff, we especially need to rely on our equipment systems, making sure they are well maintained and reliable. As mentioned above, we have used due diligence with maintenance but wear and tear trumps maintenance and now is when we need to replace.

Does the funding from this request help leverage other funds for the project? We are applying to a special radio-related FEMA grant to find matching funds so that we will able to replace both of our generators at the same time, creating a baseline of maintenance history. Funds from the Borough will be matched three to one to allow comprehensive generator replacement both at the transmitter site and at the Chilkat Center.

How does the request align with the goals and objectives in the Haines Borough Comprehensive Plan?

In reading through the HBCP, most of the goals and objectives revolve around getting information out to the community and then receiving input on the various initiatives and goals. KHNS daily delivers information on meetings and events for a variety of subjects both sponsored by the Borough or suggested for review by the Borough as part of the Plan. We do our best to relay what is in progress and/or under discussion. We host panels that can focus in on one specific issue or promote discussion of several projects and their ramifications for community living. We initiated making assembly meetings more accessible by providing a stream through our website. Under the heading of "less formal outreach and engagement efforts to encourage a diversity of public contact, such as Open Houses or Town Meetings.." we have the Mayor on air every other week to preview the assembly meeting agenda and talk about upcoming issues of interest. In the event of an emergency, we welcome all relevant officials on air to provide timely information.

C. Organizational Capacity – Describe the organization's history and finance capacity

Lynn Canal Broadcasting was granted non-profit status in 1978 and received their FCC license to begin broadcasting as a community radio station in October of 1980. Initial funding was a combination of Federal, State, listener and regional grant funding. Over the years, with the ups and downs of government funding, our budget has adjusted accordingly. Currently, we are operating with an annual budget of around \$400,000. This very spare budget is supported by listener and business sponsorship, a federal grant, various local and regional grants and the odd t-shirt sale.

During Covid, we were fortunate to access some state and federal grants that have given us a small cushion for navigating the next couple of years and we are using those funds to upgrade and replace equipment and systems that are no longer being supported by parts/tech or are simply old and worn out.

Board qualifications and involvements – we have a board with eight members, 5 in Haines and 3 from Skagway. Many of them are retired and have lived in and contributed to the community for years, others are newer and bring a fresh perspective. Criteria for being eligible for the KHNS board is that you have been a member in good standing for a least a year, have the community interests at heart and you cannot be married or partnered with a staff member.

List of grants received or pending for prior and current years - See Attached

% of revenue from this Borough - Haines Borough has a contract with KHNS to provide studio and office space in the Chilkat Center for the arts in exchange for certain facility management work. The in-kind value of the exchange is broken down annually by the Borough.

Other financing secured/committed – specifically for this generator project, KHNS is submitting a grant proposal to FEMA for the radio-specific funds that have been allocated by the federal government to help with emergency broadcast infrastructure. Total cost of replacing our two generators in Haines estimated to be \$15,000 factoring in shipping, supply chain and cost of unit. Deadline for this grant is Sept 30th so we hope to combine funding to complete the generator phase of the project.

D. Contributions to Economic Development – Explain how funds will be used to stimulate the Haines economy

When fully staffed, KHNS employs (year-round) 4-5 full time staff and 4 part-time staff all of whom are a vital part of the Haines economy and rely on KHNS to pay their salaries. Five own homes in the area and pay property taxes, the others pay rent so that their landowners can pay their property taxes. We all support local utility companies, shop for groceries, support other non-profits, spend money at the shops and keep our money in the local bank. We provide gainful employment that contributes to a vital economy.

Our request for funds to help replace our generator is part of the greater and arguably one of the most important missions of KHNS, to be able to serve in times of crisis. Were KHNS to be unable to serve it might precede a lack of faith in the station, leading to less income, laying off of staff and eventually, in an extreme case, closure of the station.

Address Haines Borough Comprehensive Plan Goals and Objectives

"... 31% of residents surveyed reported being employed part-time or seasonally and 3% were unemployed and looking for work. Of these two groups, the majority say they would work more if they could.."

KHNS, as noted above, provides solid and challenging jobs with a good wage/salary and benefits.

E. Health and Safety – Describe how the funds will provide health and/or safety education or services;

The most essential service the radio can provide is in the event of an emergency. The first step in ensuring that the radio station stays on the air, ready to broadcast vital information is for the generator to be fully functional and reliable. Our current generator is aging and needs to be retired so that we can confidently assume reliable generator service to the station when any kind of outage occurs, extreme or otherwise.

Will Funds provide essential services for children, elderly, disabled or other populations?

During the course of programming service, daily events, weather and possible or real road closures, transportation updates are addressed by live hosts. During an outage or extreme event when we would expect the generator to kick in, essential services such as shelter, food, water, transport and messages to others in the community would be broadcast to listeners – but only if we have a reliable back up system.

How does the request address Haines Borough Comprehensive Plan Goals and Objectives?

KHNS is one of the key ways to keep up on local issues (see excerpt from BCP).

"Over 90% of residents feel at least somewhat informed about public issues. Those who said they feel either somewhat or not informed were asked which information sources would be best for them to learn more about local public issues. The top preferences for ways to learn about local issues are through the newspaper (43%), **radio (24%)** and the Borough website."

KHNS works closely with paper and work closely on many issues but in the case of an immediate crisis, radio is the key source to receive current, in the moment directives. We must be able to rely on our back up system to keep us on the air to deliver any critical information from the Borough EOC and other public services as laid out in the Comprehensive Plan.

F. Broad Community Support – Provide evidence of broad community support. Number of members/beneficiaries served in previous year and last 3 years

KHNS has been a broadcast station since 1980 serving Haines, Klukwan and Skagway. Our support comes from listeners and businesses across the board. We have two annual fundraisers, the main one in early summer and we added a second one in the fall specifically to support our news department. The combination of the two provides 25% or our annual \$400,000 income. For a small area with a fluctuating year-round population, we consider this dramatic evidence of the role we serve in our communities. As a public station, many more listeners tune in than support financially. Some listeners are 24/7, others tune in specifically for the local news or extreme event updates. In addition, we enjoy the local support of dozens of businesses who provide another \$45, 000 to \$65,000 in funding (depending on the year). We can only count the listeners who contribute financially but those account for about 700/year. If we consider the upper Lynn Canal, our signal area, has a general population of 3000, that is about 23% of the population.

FY 20 – Membership: \$106,343
FY 21 – Membership: \$130,354
FY 22- Membership: \$112,107
Bus Sponsorship: \$64,407

G. Budget – Provide a rational for the budget request. How does the need justify the expenditure During the Haines extreme weather event that led to the landslide in Dec. 2020, KHNS sustained major damage to one of the towers at our transmitter site which required expensive repairs. The sustained outages also affected our equipment leading to replacement of more equipment.

We received emergency federal grant funding to subsidize the replacement and installation of equipment, disable and remove the tower and paid to bring our engineer to Haines to oversee the work. At the time, in 2020, the engineer suggested that new generators would be a smart investment in the next couple of years to further stabilize our back up power system.

KHNS, because we often have other funding options, does not generally try and tap into the non-profit pool of funds, preferring to leave it to smaller non-profits to use. This year we have experienced a high volume of engineering needs, including replacement of our two generators. We have some matching funds and would like to take this opportunity this year to apply for community funds as the project itself will serve the entire community and not just those who are members or donate. Our experience is that when there is vital information needed locally, many more people will turn to KHNS for information as we work closely with our other services to get the word out.

Is the project budget appropriate to meet the need -

The total project has a budget of \$15,000 of which we are asking for \$5000. We have researched prices and this should be more than adequate to purchase 2 new generators and have them transported and installed.

H. Timely Submission – Application is complete and on time

I. Required Documentation – in attached folder

- IRS 990 Form and Audit
- Non-profit documentation
- Current State of Alaska and Haines Borough business license if required
- Minutes of meeting where governing body approved grant request
- Organization structure including Board Directors and Executive Director
- Profit and Loss for previous year
- Current year budget including funds for this project
- Not funded in FY22



SEP 1 2022 :

SUMMARY SHEET

HAINES BOROUGH CLERK'S OFFICE

Name of Organization:	Southeast Alaska Independent Living				
Name of Contact Person:	Sierra Jimenez, Haines Program Director				
Phone: 888-487-1015 Email: sjimenez@sailinc.org					
Mailing address: PO Box	Box 183, Haines, AK 99827				
Grant Category:	Fund 01- General Fund- \$38,000 total funds to be granted				
Dollars Requested: \$8000)				
Brief Description of the Purp	pose of the Grant:				
Haines Program. SAIL is an Aging	g and Disability Resource Center as well as an Independent Living Center. We provide				
a myriad of essential services for	our community's most vulnerable residents.				
4					
I, Juna June grant application is accurate	, do hereby certify that all information provided for this and complete to the best of my knowledge.				
Name: Sierra Jimenez					
Title: Haines Program Direc	ctorDate:08/29/22				
	RECEIVED				

SEP 1 2022

HAINES BOROUGH CLERK'S OFFICE ⊰pm A. Plan for Funds: Southeast Alaska Independent Living (SAIL) is pleased to submit this proposal for \$8,000 for general operating support for the Southeast Alaska Independent Living (SAIL) Haines Program for FY23.

Who will oversee the project? The SAIL Haines office employs 3.5 FTEs. Sierra Jimenez has been employed by SAIL for the past 22 years. She has been the Program Director in Haines since opening our doors in 2007. She is responsible for all aspects of the Haines SAIL office including program offerings, staff supervision, budget and grant management.

What will be done, who will be served and when: As an Aging and Disability Resource Center, SAIL serves all seniors over the age of 60. We also serve people of all ages and disabilities including people with physical, cognitive, neurological, mental health, and other disabilities. Our mission is simple and summarizes our services: SAIL "Inspiring Personal Independence." SAIL does not duplicate services provided by other agencies, instead we collaborate with existing agencies and attempt to pick up what 'falls through the cracks'. Our leadership role in the Haines Disaster Long Term Recovery is an example of SAIL's ability to collaborate and meet community needs. Our work with individuals at risk for homelessness is another example of SAIL's ability to respond to unmet community needs. We maintain a year round office located at 431 Union Street.

While SAIL's targeted population is seniors and people with disabilities, we also serve the greater community with our extensive Durable Medical Equipment Loan Closet which contains an array of wheelchairs, walkers, scooters, grab bars, and assistive devices and equipment such as magnifiers, and sound amplifiers. These items are for short and long term loans for anyone in our community who is in need. In addition to our Loan Closet, we provide Information and Referral service regarding any aging or disability related question. On average we respond to 5-10 requests per week for information from the local business community, local government, potential tourists, and community members.

Grant Category: SAIL is applying for funds from the General Fund (Fund 01).

Outcome: This fiscal year, the SAIL Haines Program goal is to inspire personal independence by assisting a minimum of 100 Haines and Klukwan residents to become or remain as independent as possible and, in doing so, increase their quality of life. The benefits of the SAIL program are widely felt in the community - from having elders continue to live independently and safely in their own homes with lifelines and assistive devices, to young people entering the workforce with adaptive technology accommodating their disabilities. The SAIL program not only touches the recipient, but also their family and the entire community. Residents with disabilities and local seniors will increase or maintain independence, increase self-esteem, and become or remain active, productive and involved citizens. Community attitudes and perceptions about people with disabilities also change as people increase their independence, become successfully employed and fully integrate into the community.

Measuring Success: Both formal and informal evaluation methods will be used to measure and report the program's success and provide guidance for program improvement. Informal feedback will be encouraged with consumers and their families, volunteers, and staff through one-on-one settings and email. SAIL's website also has a mechanism for anonymous feedback. Formal evaluation is two-pronged: statistical analysis and a consumer-satisfaction survey. SAIL will also document and analyze participation of collaborative partners/volunteers, consumers served, and services provided.

B. Statement of Need

Documentation of Need: According to the State of Alaska Governor's Council on Disabilities and Special Education, "it is generally accepted that 1 in 7 people have a disability". Using these figures, roughly 350 people in the Chilkat Valley experience a disability. According to *Healthy Alaskans*, published by the Alaska Health and Social Services Department, people with disabilities report significantly lower levels of social participation and general satisfaction with their own life compared to people without disabilities. Alaskans with disabilities have identified a number of reasons for this discrepancy, including: 1) needs for physical accessibility and assistive technologies 2) a general lack of public awareness about the abilities, needs, rights and responsibilities of people who experience disabilities and 3) limited incomes, which further prevent their full participation in community life. The SAIL Haines Program directly works with individuals to address all of these discrepancies. The Haines Borough Comprehensive Plan states, *"The Haines Borough has more senior citizens per capita than any other Borough or community in Alaska."* As a state-recognized Aging and Disability Resource Center, SAIL is working hard to meet the needs of this large population in our community and provide the services necessary for seniors to safely 'age in place.'

Ability to Leverage Funding: One of SAIL's greatest strengths is our ability to collaborate and leverage funding and in-kind resources from other entities and sources to benefit clients/consumers. Last fiscal year the Haines SAIL Program helped local seniors and people with disabilities by leveraging resources from an impressive variety of sources, including: Alaska Department of Vocational Rehabilitation, Alaska Mental Health Trust Authority, Developmental Disabilities Mini Grants, Traumatic and Acquired Brain Injury (TABI) grant, Chilkat Valley Community Foundation, Haines Ministerial Association, SEARHC's Healing Hand Foundation, Team Rubicon, Alaska Center for the Blind and Visually Impaired and ATLA (Assistive Technology Library of Alaska).

Alignment with the Haines Borough Comprehensive Plan:

Objective 3L and 17D: Sustain, promote, and selectively expand services and amenities for retirees. SAIL's HomeMAP (Home Modifications for Aging in Place) includes a detailed home assessment, in which a SAIL staff member produces a detailed report outlining possible solutions for safely 'aging in place.' SAIL maintains a loan closet of assistive technology (magnifiers, large button telephones, hearing amplification devices, etc.) as well as durable medical equipment (wheelchairs, walkers, grab bars, toilet risers, etc.). These items are available on loan, free of charge, to anybody in need in the community.

Goal 3: Achieve a strong, diversified local economy that provides employment and income for all citizens that desire to work while protecting the health of the environment and quality of life.

SAIL provides a wide array of vocational services for individuals with disabilities who want to enter the workforce for the first time or return to work after becoming disabled. SAIL also works with individuals with disabilities to become self-employed by assisting in the development of business plans which strengthens entrepreneurial activity in Haines. During FY22, the SAIL Haines Program worked with 17 individuals to gain or maintain successful employment in the community. In 2016, SAIL started a new program called PreETS (Pre-Employment and Transition Services) in partnership with the Department of Labor, Division of Vocational Rehabilitation and the Haines Borough School District. The PreETS program is designed to provide employment related skills training and work experience for at-risk youth ages 14-18 so that they are more prepared to join the workforce after they graduate. We are exceptionally proud of our popular Summer Work Program where we have placed over 30 students in summer internships in local businesses and organizations since starting the program. As testament to the success of the SAIL PreETS Program, many of these individuals remain successfully employed in our local businesses and organizations.

C. Organizational Capacity

Our Track Record: SAIL has maintained an impeccable record for grant management and service delivery since our inception in 1992. Initially SAIL served Haines and the surrounding areas through the Rural Outreach Program (itinerant services), but due to the high demand for our services in the community, SAIL opened an office in Haines in the fall of 2007. The SAIL Haines Program has continued to grow in leaps and bounds every year since. The State Independent Living Council conducts a consumer satisfaction survey every year. The results consistently show overwhelming enthusiasm for the program. The agency and staff have also received local, state, and national recognition for excellence. The Alaska Legislature honored SAIL with a proclamation for providing "people with disabilities the chance to work for self-determination, equal opportunities and self-respect." SAIL has demonstrated success for the last thirty years, managing dozens of grants every year—every one, without exception, completed successfully.

Board Qualifications and Involvement: SAIL is fortunate to have an active and engaged Board of Directors. SAIL is proud to report that 100% of our Board of Directors make financial contributions to the agency annually. One of the things that differentiates SAIL from other social service providers is that ours is truly a consumer run organization. SAIL's bylaws mandate that 51% of SAIL's staff and Board of Directors are people with disabilities themselves. The philosophy being that people with disabilities are their own best advocates and serve as models of success to their peers. Please see Attachment # 5 for a complete Board of Directors roster and affiliations.

List of grants for the current and prior year:

The following list of grants is for the Haines SAIL program. For a full list of agency wide funding sources see attachment # 7 FY23 Budget and FY22 Profit and Loss Statement

Current Year FY23: US Administration of Community Living (secured), State of Alaska – Aging and Disability Resource Center (secured), State of Alaska- Developmental Disability Resource Connection (secured), State of Alaska-Independent Living Grant (secured), State of Alaska Community Initiative Matching Grant (secured), Haines Borough Non-Profit grant (pending), Chilkat Valley Community Foundation (1 secured, 1 pending), State of Alaska, DOL Div. of Vocational Rehabilitation Contract (secured).

Prior Year FY22: US Administration of Community Living Department, State of Alaska –DDRC, State of Alaska- Older Blind Alaskans, State of Alaska- Independent Living Grant, State of Alaska- Aging and Disability Resource Center, State of Alaska Traumatic and Acquired Brain Injury Grant, Chilkat Valley Community Foundation

This request of \$8,000 is 1.8% of the SAIL Haines Program FY23 budget.

D. Contribution to Economic Development

SAIL's **Employment Services** directly contribute to economic development in Haines. People with disabilities have twice the rate of unemployment compared to people without disabilities. This is not because people with disabilities do not want to work. SAIL is addressing this discrepancy by working with individuals who want to be employed, by providing employment skills training, job search assistance and on the job support (job coaching). SAIL also works with local businesses by providing disability awareness training and education about tax and other incentives for hiring people with disabilities.

Self- Employment: In addition to employer/employee support, SAIL also helps entrepreneurial-minded individuals start their own business by assisting to develop business plans and connecting them to resources such as local mentors, micro-grants and loans. Objective 3F of the Haines Borough Comprehensive Plan states: *"strengthen entrepreneurial activity and businesses. Provide entrepreneurs with the resources that will enable them to develop and expand their businesses."* Self-employment is often a good solution for people with disabilities because they can design a business based on their strengths and abilities and can offer better flexibility to accommodate their disability.

Pre Employment Training Services (PreETS) include employment related skills training classes and work experiences in the community for at-risk youth ages 14-18 so that they are more prepared to join the workforce after they complete high school.

Leveraging Entitlements: One of our most requested services at SAIL is helping people apply for and navigate the complicated Social Security and Alaska Public Assistance system. When an individual receives financial and medical benefits through these programs, they are more likely to succeed, become involved members of the community and ultimately stay in Haines contributing to the economy. There is a common myth that if you are receiving Social Security or Public Assistance benefits, you cannot work or you will lose your benefits. This is simply not true. SAIL works to help people fully understand their entitlements and assist people returning to work, or entering the workforce for the first time, without fear of losing their benefits.

As described in Section B (Leveraged Funds), the SAIL Haines Program has a proven success rate of leveraging outside dollars to help individuals with disabilities with unmet needs. As a direct result, these community members are able to contribute more to our local economy.

Alignment with the Haines Borough Comprehensive Plan

Goal 3- Achieve a strong and diversified local economy that provides employment and income for all citizens that desire to work. All of SAIL's employment initiatives directly address Goal 3.

E. Health and Safety

Essential Services for Youth, Elderly, People with Disabilities:

Haines Borough funding awarded to the SAIL Haines Program will provide health and safety services for seniors and people of all ages who experience a disability. Over 95% of SAIL's clients/consumers fall into the low or very low income bracket. SAIL works closely with the Haines School District, the Haines Borough Police Department, The Haines Senior Center, Haines Assisted Living, SEARHC Behavioral Health, SEARHC Medical Clinic, REACH, Cornerstone Home Health, The Salvation Army, Chilkoot Indian Association, Body IQ, Four Ravens Occupational Therapy, Southeast Occupational Therapy and other organizations to make sure that we are meeting the basic needs of vulnerable Haines residents. As mentioned in Section B (Statement of Need), SAIL and the Haines Borough School District have identified numerous 'at-risk' High School students that we are working together to enroll in our Pre-Employment Classes in the school and our Summer Work Program in FY23.

One example of how SAIL provides essential health and safety services is through our mini grant program. In FY22, the SAIL Haines Program wrote 52 mini grants for vulnerable residents. This effort leveraged a total of \$36,731 for meeting the basic health and safety needs of individuals living in Haines. Goods and services purchased with these grant funds included emergency food, housing, utility assistance, winter clothing, assistance with medical travel, and specialized equipment for medical treatments not covered by other sources.

Alignment with the Haines Borough Comprehensive Plan

This request supports Goal 17 of the Comprehensive Plan (Provide or support community services that enable residential living, economic opportunity, and add to quality of life. Give special attention to services that support families and seniors) by providing essential services for youth, seniors and people with disabilities.

F. Broad Community Support

Evidence of Broad Community Support: Since opening our doors in Haines in 2007, the SAIL program has been met with enthusiasm and widespread community support. Over the past 5 years, 127 households have donated a total of \$87,000 to support the Haines SAIL Program, a clear demonstration of their support and trust in our program. In addition to cash donations, numerous residents and businesses donate in-kind gifts to SAIL valued at over \$20,000.

Number of Beneficiaries: Last fiscal year (FY22), the SAIL Haines Program served 103 seniors and individuals with disabilities. In addition to these individuals who have developed personal goals and a 'plan of action' with SAIL, we have reached an additional 91 community members who borrowed assistive technology and assistive devices through our Loan Closet. One of SAIL's core services is Information and Referral (I&R). Last year our office provided Information and Referral services to 168 community members.

Letters of Support: please see the attached Letters of Support from partner agencies in the community as well as a letter from a consumer (client) detailing the quality and quantity of services they have received from the SAIL Haines Program.

Please see attachment # 9 Letters of Support from:

Dusty Trails Manager Marnie Hartman, DPT/Body IQ Southeast Senior Services (Haines Senior Center) Team Rubicon Chun Williamson

Haines Borough Proposal - Fiscal Year 2023 Haines Program Budget								
			This		Oth	er	Bu	dget
Expenditures	FTE		Requ	est	Fun	ds	FY	2023
A. 100 Personnel Services								
Haines staff		3.5	\$	4,000.00	\$	195,738.00	\$	199,738.00
Fringe Benefits/Payroll taxes			\$	1,360.00	\$	65,446.00	\$	66,806.00
TOTAL Personnel Services			\$	5,360.00	\$	261,184.00	\$	266,544.00
B 200 Travel								
Staff travel			\$	-	\$	20,000.00	\$	20,000.00
Auto gas			\$		\$	400.00	\$	400.00
TOTAL Travel			\$		\$	20,400.00	\$	20,400.00
C. 300 Facility				120				
Office rental			\$	1,000.00	\$	19,000.00	\$	20,000.00
Postage			\$	16.00	\$	584.00	\$	600.00
Telephone			\$	300.00	\$	4,740.00	\$	5,040.00
TOTAL Facility			\$	1,316.00	\$	24,324.00	\$	25,640.00
D. 400 Supplies								
Office supplies			\$	90.00	\$	1,510.00	\$	1,600.00
Program supplies			\$	-	\$	65,000.00	\$	65,000.00
Fundraising Supplie			\$	200	\$	2,000.00	\$	2,000.00
TOTAL Supplies			\$	90.00	\$	68,510.00	\$	68,600.00
E. 500 Equipment								
Vehicle			\$	-	\$	2,000.00	5	2,000.00
TOTAL Equipment			\$	27	\$	2,000.00	\$	2,000.00
F. 600 Other Operating Expe	nse							
Agency liability insurance			\$	85	\$	5,000.00	\$	5,000.00
Advertising			\$	1 <u>-</u>	\$	1,000.00	\$	1,000.00
Education & Training			\$	-	\$	500.00	\$	500.00
Indirect Cost Rate			\$	1,234.00	\$	54,766.00	\$	56,000.00
TOTAL Other Expense			\$	1,234.00	\$	61,266.00	\$	62,500.00
TOTAL PROJECT COSTS			\$	8,000.00	\$	437,684.00	\$	445,684.00

SOUTHEAST ALASKA INDEPENDENT LIVING, INC.

Budget Narrative: Due to the nature of this type of work, salaries and wages are the majority of the SAIL Haines Program expenses. Please see Attachment # 8 for a detailed Program Budget, which includes funding for this proposal. The Indirect Cost Rate (ICR) is our federally approved administrative cost. This includes: Insurance, dues, and audit costs.

I. Attachments:

- 1) Copy of IRS 990 form and audit
- 2) Copy of non-profit documentation (IRS 501c3 determination letter)
- 3) Current State of Alaska and Haines Borough business license
- 4) Minutes of the meeting where the Board of Directors approved Request
- 5) Board of Director Roster and Affiliations
- 6) NA- SAIL did not receive Borough Funds in FY22
- 7) Profit and Loss for FY22 and FY23 Agency Budget
- 8) Detailed Program Budget for FY23
- 9) Letters of Support



Southeast Senior Services

A Division of Catholic Community Service, Inc.



Helping elders in Southeast Alaska stay healthy, safe and independent

August 31, 2022

Ms. Alekka Fullerton, Borough Clerk Haines Borough P.O. Box 1209 Haines, AK 99827

Dear Ms. Fullerton:

Thank you for the opportunity to apply for grant funds from the Haines Borough to help support the Haines Senior Center Programs. On behalf of the Haines Senior Center and the seniors and citizens with disabilities who benefit from its services, I respectfully submit this application requesting **\$15,000** for the Haines Senior Program.

Our programs provide hot and nutritious lunches for seniors who come to the center, home delivered meals for seniors who are home-bound and door-to-door transportation for seniors and citizens with disabilities. The continued support from the Haines Borough is essential to the program and will help make it possible for us to continue to provide high quality, responsive services to Haines' citizens during the upcoming fiscal year (July 1, 2022 - June 30, 2023).

On behalf of the senior citizens of Haines, thank you again for your generous support of the Haines Senior Center and thank you for your consideration of this request. Please contact me at 463-6154 or email me at <u>marianne.mills@ccsjuneau.org</u> if you have questions or suggestions regarding this application or our services.

Sincerely,

Mauannemille

Marianne Mills Program Director



1803 Glacier Highway, Juneau, AK 99801 • Telephone: (907) 463-6177 • Fax: 888-756-6664 Find us on <u>www.facebook.com/CatholicCommunityServiceInc</u> • www.ccsak.org

Haines Borough



GRANTS FOR NON-PROFIT ORGANIZATIONS

APPLICATION MATERIALS FOR FISCAL YEAR 2023

Complete grant applications to be returned to the Borough Clerk's Office

<u>Grant Deadline</u> – September 1, 2022 5:00 PM

For further information, please contact Alekka Fullerton, Borough Clerk Haines Borough Box 1209 (907) 766-6402 <u>afullerton@haines.ak.us</u>
SUMMARY SHEET

Name of Organization: Haines	s Senior Center, Southeast Senior Services/CCS
Name of Contact Person: Maria	nne Mills
Phone: 907-463-6154 Email: marianne.mills@ccsjuneau.org	
Mailing address: 1803 Glacier H	lighway, Juneau, Alaska 99801
Grant Category: Fund 0	1- General Fund- \$38,000 total funds to be granted
Dollars Requested: \$15,000	
Brief Description of the Purpose of t	he Grant: This grant will help provide hot, nutritious congregate
and home delivered lunches to H	laines seniors. The amount of \$15,000 is needed to fund
the meal services which have grown	n dramatically since the pandemic. The senior meal program offers
a daily safety check as well as ge	ood nutrition to help promote health, safety, and quality of life.
I, Wally Lulie	do hereby certify that all information provided for this upuete to the best of my knowledge.
Name: Erin Walker-Tolles	
Title: Executive Director	Date:

A. Detailed Plan for Funds

Describe what will be done with the funds: The \$15,000 in funding will allow the Haines Senior Center to maintain senior nutrition and transportation services 52 weeks a year in light of inflation and the significant increase in meals currently being provided.

- Who will oversee the project? The Haines Senior Center Site Manager Cari O'Daniel will oversee the meal and transportation program which includes the Cook, Assistant Cook, and two Drivers. Her supervisor, NTS Program Administrator Sarah Lansing, will provide support and training.
- Who will be served? Seniors age 60 and over will be served a hot and nutritious meal during lunchtime at the senior center or through our "Meals on Wheels" home delivered meal service. Transportation will be provided to and from the senior center for lunch and other places seniors need to access. Transportation to persons with disabilities of all ages is also provided by the Haines Senior Center.
- When will the service be provided? The services will be provided throughout FY23 (July 1, 2022 June 30, 2023), beginning as soon as funding is available.
- **Explain reasoning for grant category.** General Fund (Community Chest) is the grant category that best fits because the funds will be used to serve senior citizens throughout the Haines Borough.
- Outcome describe the tangible community benefit.
 - Local seniors will maintain or increase their health, safety, and quality of life.
 - The families of elderly seniors will have more support in caring for their beloved elders if the elders receive meal and ride services through the entire fiscal year.
 - Haines seniors will be able to remain in their own homes and in the community of Haines longer.
 - Senior will spend their resources year-round on a broad range of local goods and services which will strengthen the local economy.
 - The senior center will also support the local economy as it operates the meal and rides services, using the funds to cover the expenses of local staff, facility and supplies (primarily groceries and fuel).

Outcome: how will you measure success?

Each year Southeast Senior Services (SESS) mails a survey to all seniors who participate in the meal or ride programs. The feedback received from this survey is used to better our program for the seniors. The Site Manager also receives verbal comments from participants and written comments from the suggestion box available at the center.

We also measure our success through monthly, quarterly, and annual analyses of how many clients receive services and the number of rides, congregate and home delivered meals provided, inspiring our staff to increase the active participation of local seniors in our services.

B. Statement of Need

Explain how the proposed need has been documented.

According to the 2021 Alaska census population estimates, the Haines Borough had 855 seniors 60 and over out of a total population for the Borough of 2,614, This means about 33% of the Haines Borough population is age 60 or over. According to our records, Haines seniors received 14,505 nutritious lunches in Fiscal Year 22, compared to 5,643 lunches served in FY20 which represents a substantial increase in need at a time when the cost of food is higher than ever.

• Does the funding from this request help leverage other funds for the project?

Yes, the funding from this request helps leverage other funds for this project. Other funding is received from Title III (through the State of Alaska) and VI (through Chilkoot Indian Association) of the federal Older Americans Act, a purchase of transportation services grant from Alaska Community Transit through the Haines Borough, meal and ride contributions from seniors, Medicaid reimbursements for meals/rides, and fund raising.

How does the request align with the goals and objectives in the Haines Borough Comprehensive Plan? (See <u>https://www.hainesalaska.gov/cp2025</u>

Goal 1. Quality of Life

At the Haines Senior Center, seniors enjoy a sense of community, meet long-time and new friends, as well as enjoy nutritious meals and a variety of activities. The personalized transportation service keeps them connected to the Haines community and all that it has to offer.

Goal 3. Economic Development

Objective 3L specifically states: Sustain, promote, and selectively expand services and amenities for retirees and seniors. We collaborate with other service agencies in Haines to minimize duplication of effort and maximize services available to local seniors. We also support the local purchase of food and fuel. By helping Haines seniors access community businesses and remain in the community, we bring to the local economy their stable, year-round spending.

Goal 4. Transportation and Strategy 5 (expansion of senior transportation services)

For years, Southeast Senior Services has worked with others in the community to expand transportation options for seniors and people with disabilities. The Chilkoot Indian Association, the Haines Borough, REACH, and SAIL have been among our partners. We maximize coordination with these agencies.

Objective 4F cites the support of our transportation service through the annual update of the coordinated transportation plan to enable us to continue to receive grants from the Alaska Department of Transportation. Alaska Community Transit grants offer the means by which we replace our wheelchair lift-equipped vehicles. The Haines Senior Center program has the only lift- equipped vehicles available for public use in Haines. We provide seniors with transportation four days a week through our Older Americans Act funding sources. A Purchase of Services grant from the Borough/Alaska Community Transit allows us to provide after-hour and weekend rides to both seniors and individuals with disabilities of any age.

Goal 17. Community Services

Objective 17D specifically addresses Sustaining, promoting and selectively expanding services and amenities for retirees. Seniors who frequent the Senior Center currently receive physical activity through weekly exercise programs as well as playing fitness games on the Wii. The Site Manager regularly invites speakers on topics of interest to the seniors, including nutrition education and health maintenance topics. She also manages marketing of services available at the Center, through regular radio spots and newspaper ads.

C. Organizational Capacity

Describe the organization's history and finance capacity:

SESS is the largest provider of senior services in Southeast Alaska, serving at least 2,500 unduplicated individuals annually. The SESS mission and vision "to promote the health, independence and quality of life of seniors in Southeast Alaska" has guided its development of services. In 1974, SESS opened senior centers in the Angoon, Juneau, Ketchikan and Sitka, providing nutrition and transportation services to the seniors in these communities. Between 1974 and 1983, SESS developed and expanded services for seniors into a network of 14 senior centers. Services began in Haines in 1976.

SESS continually strives to maximize resource and funding capabilities. We utilize collaborative partnerships with senior organizations, municipalities, tribal organizations, and other service providers to deliver quality, effective, and responsive services. SESS has steadily widened its focus to include special programs designed to help older adults remain in their homes and communities, despite multiple challenges or disabilities.

CCS has a proven track record of initiating and managing effective programs. CCS employs over 130 individuals, has approximately 100 volunteers, and operates with an annual budget of over \$7 million from over 40 funding sources. In all of our granting and contractual relationships, we have clearly demonstrated our ability to meet project timelines, work cooperatively with their staff, meet budget requirements, and provide quality services. We have historically maintained standards as demonstrated in past quality assurance reviews by the State and by maintenance of standards for licensing and certification.

• Board qualifications & involvement(s).

The CCS Board of Directors assumes full legal authority and fiduciary responsibility for the overall organization and operation of its programs. The Board annually approves all project budgets and plans of service. All financial management of these programs is in compliance with the Accounting Policies and Procedures of CCS, according to Generally Accepted Accounting Principles (GAAP). CCS is audited annually by an independent CPA firm, in accordance with *Government Auditing Standards* and the *State of Alaska Audit Guide and Compliance Supplement for State Single Audits*.

The CCS Board of Directors is comprised of well-qualified, dedicated community members who fully support the mission of the CCS: *Strengthening families and individuals in Southeast Alaska, with dignity, care, and compassion.* Haines resident, Vince Hansen has been a board member since February of 2015. Other board members represent Juneau, Ketchikan and Klawock.

• List of grants received or pending for prior and current years.

- 1. Older Americans Act, Title III Nutrition, Transportation, and Support Services grant from the State of Alaska
- 2. Older Americans Act, Title VI, Nutrition and Supportive Services grant from the U.S. Administration for Community Living through the Chilkoot Indian Association
- 3. Older Americans Act, Title VI, Caregiver Support Services grant from the U.S. Administration for Community Living through the Chilkoot Indian Association
- 4. Vehicle Replacement grants from Alaska Community Transit (2007, 2011, 2015)
- 5. Mobility Manager grants from Alaska Community Transit (2011, 2012, 2013)
- 6. Purchase of Services and Operating Assistance grants from Alaska Community Transit

• % of total revenue from the Borough.

When our Board of Directors approved the FY23 budget at the end of June 2022, we anticipated 21% of the revenue coming from the Haines Borough in the form of \$60,000 from the Purchase of (transportation) Services grant. We had budgeted \$55,000 from the State of Alaska Division of

Senior & Disabilities Services which was added by the Alaska State Legislature this session. However, the Governor vetoed this anticipated increase in our State grant, so we find ourselves needing more local support. Both budgets have been included in this application. The FY23 budget after veto shows our \$15,000 Haines grant request, which would result in 26% total revenue from the Borough.

• Other additional financing secured/committed.

The FY23 operating budget for the Haines Senior Center is \$285,610. Other additional financing secured/committed includes grants from the Alaska Division of Senior and Disabilities Services NSIP (6,180, Title III (\$90,000), the Chilkoot Indian Association Title VI program (\$77,890), Title VI NSIP(\$560), and in-kind revenue (\$17,500). \$10,000 is **anticipated** from Medicaid Waiver meal and ride service reimbursement and \$13,000 is anticipated from client contributions. In addition, local fund raising in the amount of \$10,000 was budgeted. The State of Alaska is in the process of issuing us at least one American Rescue Plan Act grant in the coming weeks.

D. Contribution(s) to Economic Development

Explain how funds will be used to:

• Stimulate the Haines economy.

The FY23 Haines Senior Center budget is 285,610. Most of these funds will remain in the Haines community through the purchase of food, gas, and other program supplies, rent, and employee wages and benefits.

Enabling seniors to remain in Haines is vital to the city's economy. Senior citizens spend their resources, whether Social Security, Medicare health care dollars, Permanent Fund Dividends, Senior Benefits, Public Assistance, or retirement income, on a broad range of local goods and services. Their stable, year-round spending has an economic multiplier impact which results in job creation and income generation that expands the size of the economy. Support services also make it possible for family caregivers to continue to work and contribute to the local economy by arranging care for their older loved ones. This program effectively makes it possible for Haines' oldest citizens to access federal, state and private dollars and inject them into the Haines economy. Haines' senior population is one of the fastest growing in the nation, and is a stable, year-round industry, strengthening the local economy.

Address Haines Borough Comprehensive Plan Goals and Objectives

Goal 1. Quality of Life

The funds will support nutrition services, socialization, access to community services (through transportation) which translate into the health, safety, and quality of life of Haines seniors.

Goal 3. Economic Development

As addressed at the top of this page, the funds will support local purchase as well as helping Haines seniors to remain in Haines and contribute to the local community through their year-round spending.

Goal 4. Transportation

As we recover from the pandemic, these funds will help us support the expansion of public transportation for seniors and people with disabilities.

Goal 17. Community Services

The Haines Borough will sustain, promote, and selectively expand services and amenities for retirees and seniors through its additional support of the Haines Senior Center. Now that local seniors are coming back to the senior center after the pandemic, we will be able to serve as the community focal point for activities, information and resources.

E. Health & Safety

Describe how the funds will provide health and/or safety education or services:

• Will funds provide essential services for children, elderly, disabled, or other populations in Haines? If yes, please explain.

The funds we are requesting will allow us to serve congregate and home delivered meals to elders as well as provide wheelchair accessible door to door transportation. All meals include a third of the daily recommended allowance of Vitamin A and C, a protein source, vegetables and fruits, a carbohydrate source such as rice, potatoes, or bread, milk, tea and coffee and an occasional dessert. A sample hot lunch menu could include 3 ounces of baked salmon, fruit slaw, mixed vegetables, a biscuit, 8 ounces of milk, and a dessert. There is no charge for a meal; however seniors are encouraged to make a free will donation towards the cost of the meal.

Meals are currently available Monday through Thursday. Many of the senior center meal participants are frail or have age-related disabilities. This lunch is their main meal of the day. These nutritious meals help seniors stay healthy as long as possible and therefore, remain in their own homes and communities longer.

Coming to the center allows the seniors the opportunity to socialize with other members of their community and receive information about programs available to them. Seniors who are unable to leave their homes may receive home-delivered meals. The meal delivery driver is often the only person the elder will see that day, offering a friendly smile and a safety check in addition to the nutritious meals.

Door to door, wheelchair transportation is also available. The rides provided by the Haines Senior Center allow Haines seniors and citizens with disabilities to access community activities and resources, serving as a virtual "lifeline" to encourage activity and prevent isolation

In FY22, 114 seniors received 14,505 home delivered meals and 49 seniors and citizens with disabilities received 2,336 rides. The senior center recently opened for congregate meals and activities after the pandemic.

How does the request address Haines Borough Comprehensive Plan Goals and Objectives?

Southeast Senior Services strives to promote the health, independence and quality of life of seniors through the development of community services. The grant funding will help to maintain and increase the quantity and array of services available through the Haines Senior Center. In turn, this will address Comprehensive Plan goals of Quality of life, Transportation, and Community Services. As Haines seniors age in place with the support of our services, maintain their activity in the community and access local businesses, they will also support the local economy.

F. Broad Community Support

Provide evidence of broad community support for the organization:

Cari O'Daniel, Site Manager, stays in close contact with service agencies such as Cornerstone Home Care, Haines Assisted Living, and SAIL to ensure their clients have access to our services. These agencies have sent countless support letters over the years, participate in the annual Coordinated Transportation Plan meeting, and attend public events held at the center.

The Chilkoot Indian Association has partnered with SESS since 2013 to secure Title VI Older Americans Act funds to serve Native elders and their family caregivers. These funds also support the Haines Senior Center's meal and ride programs as well as provide a Family Caregiver Support Advocate for family caregivers.

The support of the Haines Borough has been tremendous over the years, offering us the facility from which to operate the meal and ride program. Borough staff has assisted us in accessing and making use of Legislative Capital grants for facility improvements and the Borough has offered to serve as the lead agency in the development of the Haines Coordinated Transportation Plan. In recent years, the Borough has worked with us to access Alaska Community Transit funds toward the goal of increasing transportation services for senior citizens as well as citizens with disabilities who live in Haines.

- Number of members/beneficiaries served in previous year in the last 3 years;
 - 1. FY 20: 120 (July 1, 2019 June 30, 2020)
 - 2. FY 21: 79 (July 1, 2020 June 30,2021)
 - 3. FY 22: 114 (July 1, 2021 June 30, 2022)
- Petitions/letters of support: See attached
- Other evidence

Project Description

This project will ensure that safe, accessible, and reliable transportation will continue to be available to senior citizens and people with disabilities throughout the Haines Borough. The Community Care-A-Van Transportation Services are operated by Catholic Community Service through its Haines Senior Center transportation system. The Haines Senior Center will provide lift-equipped vehicles and drivers trained in proper passenger assistance to offer these citizens access to community activities, resources, and services. The Haines Borough will offer its support of the Services through a Purchase of Services grant from Alaska Community Transit/DOT & PF which will pay for 80% of the expenses related to the project.

G. Budget

Provide a rationale for the budget request:

Is the project budget appropriate to meet the need?

The amount of \$15,000 is being requested to continue to provide meal and ride services year-round for Haines senior citizens. Since the pandemic, the number of meals we have been serving has nearly tripled. The cost of food and fuel continues to rise, so we need to balance this with the knowledge there are other deserving organizations in Haines that need support. Between anticipated ARPA funds coming from the State of Alaska, this grant from the Haines Borough, and local fundraising, we are hopeful the funds will be sufficient to meet the need.

• How does the need justify the expenditure?

The funds will support the health and safety of local seniors in addition to moving the Borough toward the goals outlined in its Comprehensive Plan. There are no non-essential expenses proposed in the grant request. The funds are needed to help pay for nutritious meals and personalized transportation.

Vulnerable seniors in particular will benefit from the senior center meal services. When a senior or one of their guests comes to the center for a delicious hot lunch, it may appear that the only expense involved is for the food being served. But in actuality, the food is a small portion of all expenses which includes staff wages, rent, home delivered meal containers, household supplies, training and reporting requirements for compliance with grants and Medicaid.

The same holds true for transportation. One could assume that the only expenses for rides are gas and driver wages. But as with meals, there are many other costs not apparent to the general public such as reporting, background checks, pre-hire and random drug/alcohol testing, driver training, and insurance. Our vehicles require preventative maintenance and, with the growing age of our buses, frequent repairs are required. This can cost over \$1,000 per repair.

The Haines Senior Center staff works hard to be budget conscious and also works hard to fund raise to help balance their budget. The Chilkoot Indian Association has partnered with SESS to acquire Title VI funds, and the meal and ride program participants donate what they can each year. The senior center is located in a building owned by the Haines Borough. It is a community building that is shared by many. It seems appropriate that the Haines Borough would want to take this opportunity to support the Haines Senior Center.

We thank you for your consideration.

H. Timely Submission

Application is complete and on time.

All sections of this application were thoroughly addressed and all attachments are included.

This application was submitted to Alekka Fullerton on Thursday, September 1, 2022 before the 5:00 pm deadline.

I. Required Documentation

X 1. Copy of IRS 990 form and audit, if applicable.

X 2. Copy of non-profit documentation (IRS).

X 3. Current State of Alaska and Haines Borough business license.

 \underline{X} 4. Minutes of meeting where the governing body approved request.

 \underline{X} 5. Organization structure including names of the board of directors and the executive director if applicable.

X_6. If funded in FY22, written report of use of funds. We did not apply for these funds in FY22.

X 7. Profit and Loss for previous year.

X 8. Detailed budget for current year, including funds for this project.







DIVISION OF MINING, LAND & WATER Southeast Regional Land Office

> P. O. Box 111020 Juneau, Alaska 99811-1020 Main: 907.465.3400 TTY: 711 or 800-770-8973 Fax: 907.500.9011

ALASKA DEPARTMENT OF NATURAL RESOURCES NOTICE OF PRELIMINARY DECISION LEASE AMENDMENT ADL 108979 HAINES HUTS

In accordance with AS 38.05.035, the Division of Mining, Land and Water (DMLW) has issued a preliminary decision to amend a 10-year public and charitable lease to Haines Huts ("Applicant") for 2.6 acres of state-owned land within the Haines State Forest, along the southern flank of Mt. Tukgahgo, Haines, Alaska. The applicant plans to construct a 16-foot by 20-foot public use backcountry cabin and a 4-foot by 4-foot outhouse for the Haines community. The applicant was issued an Entry Authorization for the requested activity in 2019. In 2021, DMLW received an application requesting to amend the location of the proposed lease to reduce impacts to wildlife, which is the subject of this decision. All other aspects of the applicant's requested activity remain the same.

Public notice of this preliminary written decision will be posted on the Alaska Public Notice System website for 30 calendar days. The public may provide written comments to DMLW during that time. Additionally, notice of this preliminary decision will be given to adjacent landowners. The Haines Post Office will also be requested to post the notice of this preliminary decision.

The public is invited to comment on this preliminary decision. Written comments must be received by DNR no later than 5:00 p.m. October 3rd, 2022. All comments must be submitted to: DNR, Division of Mining, Land and Water, Southeastern Regional Office by mail at 400 Willoughby Avenue, P.O. Box 111020, Juneau, AK 99811-1020, by fax at 907-465-3886, or by electronic mail to megan.hillgartner@alaska.gov in order to ensure consideration. Please include your mailing address and telephone contact. In order to establish appeal rights regarding this decision, you are required by law to meaningfully participate in the decision process by commenting on the decision, in writing, prior to the comment deadline. Following the deadline, all timely written comments will be considered, and DNR may modify this decision based on public comments received.

The names of commenters and comments received will be part of the public record.

If DNR determines that public comments in response to this notice indicate the need for significant changes to the decision, additional public notice will be given. If no significant changes are required, the preliminary decision, after any necessary minor changes, will be issued as a final decision. A copy of the final decision, along with instructions on filing an appeal, will be sent to all persons who comment on the preliminary decision. Persons who do not submit written comments during the comment period will have no legal right to appeal the final decision.

The Division of Mining, Land and Water reserves the right to waive technical defects in this publication.



STATE OF ALASKA DEPARTMENT OF NATURAL RESOURCES DIVISION OF MINING, LAND AND WATER

Southeast Regional Land Office

Preliminary Decision ADL 108979 Haines Huts and Trails Lease Amendment AS 38.05.810

Executive Summary

This Preliminary Decision (PD) is the State's preliminary best interest finding regarding a proposed disposal of interest in state land. Haines Huts and Trails applied for a lease amendment during the term of their originally issued Entry Authorization to change their leasehold location. The State intends to authorize this lease amendment, contingent on comments received during public notice and the best interest finding contained herein. The public is invited to comment on this PD. The deadline for commenting is 5:00 pm October 3rd, 2022. Please see the Public Notice section of this decision for requirements related to submitting comments for consideration. Only the applicant and those who comment have the right to appeal the Final Finding and Decision (FFD).



Preliminary Decision, Lease Amendment ADL 108979 Page 1 of 12

Requested Action

Haines Huts and Trails ("applicant") was issued an Entry Authorization on December 9th, 2019, to lease approximately 2,000 square feet of land for the purposes of constructing a 16-foot by 20-foot backcountry cabin and a 4-foot by 4-foot outhouse on the south flanks of Mt. Tukgahgo, near Haines, Alaska. The purpose of the lease is to establish and operate a public use backcountry cabin and outhouse facility for the Haines community. On December 2nd, 2021, the applicant submitted an application requesting to amend their lease to change the location of the lease site to an area further down Mt. Tukgahgo. The reason for the location change is because the original lease site was located in sensitive goat habitat. The applicant selected the new location to reduce potential impacts to wildlife. The applicant requested a 55-year lease. A development diagram is included as Attachment 1.

Proposed Improvements:

- (1) 16-foot by 20-foot timber backcountry hut containing approximately 320 square feet.
- (1) 4-foot by 4-foot outhouse containing approximately 16 square feet.

The total area requested by the applicant is approximately 2.6 acres.

Proposed Action

The Division of Mining, Land & Water (DMLW) proposes to issue a 10-year public and charitable lease to the applicant for the construction, operation, and maintenance of the cabin and outhouse facilities, as outlined in the applicant's lease development plan (Attachment 1).

Scope of Decision

The scope of this decision is limited to the determination of whether it is in the State's best interest to issue a 10-year public and charitable lease to the applicant. The administrative review for this authorization is defined by AS 38.05.035(e)(1)-(2) and is limited to (1) reasonably foreseeable, significant effects of the uses to be authorized; (2) applicable statutes and regulations; (3) facts pertaining to the land or resources; and (4) issues that are material to the determination.

Authority

This lease application is adjudicated pursuant to AS 38.05.035(b)(1) and AS 38.05.035(e) Powers and Duties of the Director, AS 38.05.810 Public and Charitable Use and AS 38.05.945 Notice. The authority to execute the PD, Final Finding and Decision (FFD), and the lease has been delegated to the Regional Managers of DMLW.

Administrative Record

The administrative record for the proposed action consists of the Constitution of the State of Alaska, the Alaska Land Act as amended, applicable statutes and regulations referenced here-in, the Haines State Forest Management Plan (HSFMP, 2002) and other classification references described herein, and the casefile for the application serialized by DNR as ADL 108979.

Location Information

Geographic Location:

The proposed site is located at an elevation of 3.200 feet on the south flanks of Mt. Tukgahgo, approximately 8 miles from downtown Haines.

Property Description:

The proposed lease is located within Section 11, Township 30 South, Range 58 East, Copper River Meridian.

Other Land Information:

Municipality: Haines Regional Corporation: Sealaska Corporation Village Corporation: N/A Federally Recognized Tribe: Chilkat Indian Village and Chilkoot Indian Association Approximate Leasehold Corners Coordinate Location: o 59°17'6.60"N, 135°36'20.53"W

- o 59°17'8.08"N, 135°36'21.09"W
- o 59°17'7.09"N, 135°36'3.82"W
- o 59°17'5.96"N, 135°36'4.27"W

Title

The State of Alaska received tentative approval for the lands from the United States Department of the Interior in 1980 under General Grant GS-1540. This tentative approval was recorded in 2012 under document number 2012-0000278-0.

Title is subject to valid existing rights, including reservations, easements, and exceptions in the U.S. Patent or other State or federal conveyance, and in acts authorizing the issue thereof, easements, rights-of-way, covenants, conditions, reservations, notes on the plat, and restrictions of record, if any.

Adjacent Landowners

The site is located entirely within the Haines State Forest and is surrounded by state-owned land.

Third Party Interests

Non-identified.

Planning and Classification

The proposed lease is consistent with the 2002 HSFMP. Located in Management Unit 8 Subunit 8d, Resource Management Land is the land classification (HSFMP, 3-64).

Management Intent: "Subunit 8d will be managed for its scenic, recreational, wildlife habitat, and potential mineral values under a land classification of Resource Management to represent the diversity of resources. There are approximately 224 acres of operable commercial timber in this

Preliminary Decision, Lease Amendment ADL 108979 Page 3 of 12

unit. Because of the remote, mountainous location of this subunit, development is not anticipated or considered appropriate. If authorizations are granted by the Department, the general standards of Chapter 2 apply except for the specific requirements of a special use designation prohibiting motorized uses in certain parts of the subunit," (HSFMP, 3-65).

Chapter 2 of the HSFMP establishes the forest-wide management policies the Department must consider. "Recreation" is the most pertinent and applicable section relevant to this the applicant's requested use of state land:

Chapter 2 Guidelines – Recreation; Public Use Cabins and Trails: "A system of public use cabins and trails should be established in the State Forest, subject to the availability of funding. Public use cabins and trails may be constructed where analysis indicates a need, and where the state, federal, or local government, or local non-profit organizations are able to construct and maintain the facilities. The Division of Forestry, Division of Parks and Outdoor Recreation and Department of Fish and Game shall closely coordinate the establishment of any public use cabin site or trail [...] Public use cabins will be sited to avoid conflicts with existing or planned resource uses such as timber or mineral development," (HSFMP, 2-25).

While development in this subunit was not considered appropriate in 2002 when the HSFMP was created, the demand for public use cabins was clearly foreseen and anticipated. The level of disturbance to the landscape and ecosystem from the construction of the cabin and outhouse proposed in the application is minimal: the cabin and outhouse comprise a small square footage and the facilities are designed to be dismantled in one weekend. Additionally, all cabin users are required to adhere to leave no trace practices.

Special Use Designation ADL 106929 (Appendix B), "prohibits the use of motorized vehicles in the area of Mt. Ripinski north to Seven Mile Saddle [...] this designation also affects [...] subunit 8d," (HSFMP, 3-52).

With the appropriate stipulations in place, the proposed lease is consistent with the classification and management intent of the Haines State Forest Management Plan.

Traditional Use Finding

This finding is not required since the proposed leasehold is located within an organized borough (AS 38.05.830). The Borough will be notified of this Preliminary Decision.

Access

The 2019 Preliminary Decision for ADL 108979 noted issues with legal access. While there are several trailheads that may be used to connect to a trail system providing physical access to the

Preliminary Decision, Lease Amendment ADL 108979 Page 4 of 12 requested lease area, the trailheads did not provide legal access because they either began on private land or lands not managed by DMLW/DOF (i.e., Mental Health Trust and University of Alaska lands). The original PD stated that it was the applicant's responsibility to establish legal access to their lease site (via one of the existing trailheads) during the Entry Authorization (EA) period by working with pertinent stakeholders/landowners for permission or authorization.

The applicant has since clarified in this amendment request that the primary route to the leasehold will be via 7 Mile Saddle Trail. This trail bisects land owned by the University and Mental Health Trust (MHT) near Mile 7 of the Haines Highway. On May 17th, 2022, the applicant obtained a letter of non-objection and support for the proposed lease from the University of Alaska¹. The applicant also supplied DMLW with proof of a public, non-exclusive trail use agreement between the Haines Borough and MHT for use of 7 Mile Saddle Trail².

As noted above, the site is also subject to Special Land Use Designation, ADL 109269. This designation prohibits motorized access within certain portions of Subunit 8d.

Reservation of Mineral Estate

In accordance with section 6(i) of the Alaska Statehood Act and AS 38.05.125, the state, in this decision, reserves unto itself the mineral estate, including oil and gas, and the rights expressed in the reservation clause of the statute, that being the right to reasonable access to the surface for purposes of exploring for, developing and producing the reserved mineral resources. Exploration and development, if any, which could occur, would be consistent with AS 38.05.130 and other applicable statutes and regulations.

Mineral Orders

The proposed leasehold does not fall within the areas delineated in an Administrative Mineral Closing Order (MCO). Neither an MCO nor a leasehold location order is necessary or appropriate for this lease.

Hazardous Materials and Potential Contaminants

Hazardous materials, specifically a 250-gallon propane tank for cooking and heat will be stored within the proposed leasehold. Stipulations will be included in the lease to ensure proper handling and storage. The use and storage of all hazardous substances must be done in accordance with existing federal, state, and local laws. Debris (such as soil) contaminated with used motor oil, solvents, or other chemicals may be classified as a hazardous substance and must be removed from the leasehold and disposed of in accordance with state and federal law.

¹ Letter from Adrienne Stolpe, Director of UA Land Management. May 17th, 2022.

² Revocable License for Land Use, MHT 9100994 (Mental Health Trust and Haines Borough)

The applicant will also store human waste in a 55-gallon drum within the outhouse. The drums will be removed annually via helicopter and disposed of according to state and local regulations at the Haines Borough sewer treatment facility.

Agency Review

An agency review was conducted on March 4th, 2022. The deadline for agency comments was April 4th, 2022.

The following agencies were included in the review:

- ADF&G Wildlife Cons.
- ADF&G Habitat Division
- DEC Commissioners Office
- DEC Water Plan Review Section Manager
- DNR DMLW OHA Review & Compliance
- DOT&PF Statewide ROW Chief
- DNR Division of Forestry
- DNR Division of Parks and Outdoor Recreation

Agency Review Comment and Response:

<u>ADF&G Division of Wildlife Conservation Comment</u>: ADF&G does not object to the issuance of the authorization but notes that while the new cabin location makes an attempt to avoid sensitive goat habitat, ADF&G models still predict that the site has a high probability for selection by mountain goats. This location is also near modeled high quality brown bear denning habitat.

To minimize potential impacts to wildlife, ADF&G provided the following recommendations:

- Limit helicopter use to the period of July 16th October 31st to avoid the potential for disturbing denning brown bears and mountain goats.
- Helicopter should maintain a minimum distance of 1,500 meters from observed mountain goats.
- Food, fuel, waste, or other potential bear/animal attractants should be properly stored. Any conflicts with wildlife should be reported to the Douglas office of ADF&G at (907) 465-4265 or the Alaska Wildlife Troopers if an emergency occurs.

Lastly, ADF&G notes that the youth mountain goat hunting area (hunting season September 15th – November 15th, unless closed by emergency order) includes the Takshanuk Ridge from 10-mile to the town of Haines, and the primary access used by

most hunters is the 7 Mile Saddle Trail. Noise from construction activities may displace or alter goat behavior which may cause conflict with hunting user groups.

<u>DMLW Response</u>: The result of the special use designation (ADL 106929) is to preclude the use of personal and commercial motorized vehicles within the designated Mt. Ripinski non-motorized area, as these uses would be otherwise generally allowed on state-owned land. 11 AAC 96.020(D)-(E) outlines what types of vehicles are generally allowed on state land, while 11 AAC 96.020(F) addresses what aircraft and watercraft uses are generally allowed. Thus, a clear distinction is drawn in regulation between vehicles and other modes of transportation, and the Mt. Ripinski non-motorized special use designation only applies to motorized vehicles. However, under 11 AAC 96.025(3)(c), activities that are generally allowed must be conducted in a manner that minimizes disturbance to fish and wildlife resources.

While helicopter use is not a part of this authorization because it is generally allowed and not restricted by the special use designation, the applicant will be advised of ADF&G's recommendations on how to minimize disturbance to wildlife. Additionally, a stipulation will be incorporated in the lease to address the recommendations regarding attractants to wildlife.

To avoid impacts to wildlife and reduce the potential for conflict with hunting user groups, DMLW will require that the applicant avoid construction between September 15th and November 15th.

<u>DNR Division of Forestry Comment</u>: DOF stated that while the application does not discuss removal of vegetation, the applicant indicated in an email exchange with DOF that vegetation removal could consist of a 26-foot by 30-foot area and 10-foot by 10-foot area for the human waste facility. DOF notes that removing the subalpine vegetation for the development of the cabin could promote adverse impacts to the surrounding vegetation. DOF noted concerns that removing vegetation could lead visitors to continue cutting the vegetation back for fuel once they found that cutting has already occurred. DOF would prefer that the applicant select a specific site that does not require the removal of subalpine forest or shrubs for development.

DOF indicated that the management plan supports dispersed recreation in this subunit and that allowing cabin construction is contrary to the plan guidelines. If authorized, DOF requests that development impacts to wildlife, habitat, and recreational users are as minimal as possible.

<u>DMLW Response</u>: In response to DOF's comment regarding brush and vegetation clearing, DMLW will include the following stipulation in the lease agreement:

Site Disturbance:

Preliminary Decision, Lease Amendment ADL 108979 Page 7 of 12

- a) Site disturbance shall be kept to a minimum to protect local habitats. All activities at the site shall be conducted in a manner that will minimize the disturbance of soil and vegetation and changes in the character of natural drainage systems. Any ground disturbances that may occur shall be contoured to blend with the natural topography to protect human and wildlife health and safety. Particular attention must be paid to preventing pollution and siltation of any waterways and to preventing disturbances to fish and wildlife populations and habitats.
- b) Brush clearing is allowed only to the extent necessary to maintain the present development. The Lessee may use dead timber that is down. The Lessee shall not cut standing timber within the leased area unless specifically authorized by DNR's Division of Forestry. The removal of vegetation shall be kept to a minimum and areas requiring disturbance should be allowed to revegetate as soon as possible after disturbance. To the extent possible, associated vegetation should be left intact to enhance stability, control erosion and enhance scenic qualities.

As noted in the *Management Intent* section above, development within the subunit was not considered in 2002 when the HSFMP was created. However, the HSFMP clearly acknowledged and anticipated that the demand for public use cabins would grow in the following years. This demonstrated in the HSFMP Chapter 2 Area Guidelines for Recreation, which specifically notes, "a system of public use cabins and trails should be established in the State Forest, subject to the availability of funding. Public use cabins and trails may be constructed where analysis indicates a need, and where the state, federal, or local government, or local non-profit organizations are able to construct and maintain the facilities," (HSFMP, 2-25). The applicant's proposal to construct a cabin and outhouse is consistent with the management plan.

This amendment request is a result of the applicant voluntarily relocating after previously receiving approval for a lease in 2019 in an effort to reduce potential impacts to wildlife and habitat. The impacts to state lands from the proposed activities are de minimis in nature, consist of a small square foot area, and are designed to be dismantled in one weekend.

Background

DMLW received the original application for lease on February 25th, 2019. The applicant was issued an EA on December 9th, 2019. In November of 2021, DMLW received an application requesting to amend the location of the proposed lease to reduce impacts to wildlife, as the previously authorized lease area was located directly within sensitive mountain goat and brown bear denning habitat. The amendment proposes to move the location of the lease area to the

3,200 foot elevation level (still within the southern flank of Mt. Tukgahgo), which is the subject of this preliminary decision.

Discussion

Haines Huts is a nonprofit corporation seeking to establish several public use cabins in the Haines area. Much of the uplands in Haines are owned and managed by the State.

The mission of Haines Huts is to establish and operate an area hut system that provides reliable backcountry cabins for residents and visitors, with a purpose of providing a place that fosters wilderness education for all ages, a jumping off point for backcountry adventures, and a safe place for emergency situations.

DMLW proposes to issue Haines Huts a 10-year public and charitable lease under AS 38.05.810.

The lease shall be subject to the standard DMLW Lease Agreement, Special Stipulations and the terms and conditions set forth therein (Attachment 2).

Development Plan

The Development Plan (DP) attached to this decision (Attachment 1) and dated November 7th, 2021, is under consideration by DMLW. Should the proposed lease be granted, it is anticipated that the DP will need to be updated throughout the life of the lease as activities and/or infrastructure are added or subtracted. All updates must be approved, in writing, by DMLW before any construction, deconstruction, replacement of infrastructure, or change in activity will be authorized. DMLW reserves the right to require additional agency review and/or public notice for changes that are deemed by DMLW to be beyond the scope of this decision.

Performance Guaranty

In accordance with AS 38.05.035, AS 38.05.860, and 11 AAC 96.060(a) Performance Guaranty, the applicant will be required to submit performance guaranties for the lease to incentivize performance of the conditions of the lease. This provides a mechanism for the state to ensure that the lessee shares any financial costs associated with noncompliance of the lease agreement for site cleanup, restoration, and any associated costs after termination or expiration of the leases. The following bonds will be required:

\$6,570 Performance Bond:

Performance guaranties provide a means to pay for corrective action if the lessee fails to comply with the lease requirements. In accordance with AS 38.05.035(a)(4), the applicant will be required to submit a performance guaranty. The amount of the performance guaranty is based on the scope and the nature of the activity and the potential cost of restoring the site. Performance guaranties are subject to periodic adjustments during the term of the authorization to address increases or decreases in the costs of rectifying problems and rehabilitating state land due to inflation, changes in the level or nature of development, or other appropriate factors.

The 2019 Final Finding and Decision for ADL 108979 dictated that the applicant submit a Performance Guaranty in the amount of \$6,570. The lease amendment contemplated in this Preliminary Decision makes no changes to the 2019 Performance Guaranty requirement.

Insurance

Consistent with AS 38.05.035(a) to protect the State from liability associated with the use of the site, the applicant shall provide and maintain a comprehensive general liability insurance policy with the State of Alaska named as an additional insured party per the stipulations of the Lease Agreement. The applicant shall secure or purchase at its own expense and maintain in force at all times during the term of this lease, liability coverage and limits consistent with what is professionally recommended as adequate to protect the applicant and the State, its officers, agents and employees from the liability exposures of all the insured's operations on state land. The insurance requirement may be adjusted periodically.

Survey

In accordance with AS 38.04.045, this lease does not require a survey. However, the State of Alaska reserves the right to require one in the future, should the need arise due to changes in statutes or increased use of the area. The applicant will be required to submit a scaled diagram with GPS coordinate points for all leasehold corners. Remittance of an approved diagram will be required before the lease may be issued. DMLW reserves the right per AS 38.05.035(a) to require a survey in the future, should conditions arise that warrant a survey. If a survey is required, the applicant will be responsible for the costs of the survey.

Compensation and Appraisal

AS 38.05.810(b-d): DNR is setting the annual fee for the lease at \$0.00, because of the youth encampment fee exemption. Under AS 38.05.097(a) a nonprofit organization using state land leased by it under AS 38.05.810 for a youth encampment or similar recreational purpose is exempt from lease rental payment on that land.

In 2019, Haines Huts submitted a written request for this exemption stating:

"Haines Huts is a nonprofit organization that provides backcountry cabins for the purpose of providing a place for youth and young adults to participate in outdoor recreation living. Our cabins provide an opportunity for young people to foster personal and social development while experiencing the outdoors. This cabin will be used yearround for youth and young adult outdoor programs and for this reason should qualify for the youth encampment exemption."

Given that 11 AAC 58.825 calls to liberally interpret the applicable statutes and regulations in determining the qualifications for a youth encampment fee exemption, DMLW determined that the applicant qualifies for the exemption.

Preliminary Decision, Lease Amendment ADL 108979 Page 10 of 12

Haines Huts and Trails

Subleases

Subleasing is permissible through AS 38.05.095, if the proposed leased is approved. As this lease is authorized under AS 38.05.810(b-d) for public and charitable use, the lessee may only sublease to another qualified entity under AS 38.05.810(b-d). No commercial use of the leasehold is allowed under AS 38.05.810(b-d). All potential subleases must first be approved in writing by DMLW. As required by 38.05.095(b), a nonprofit organization that is exempted from paying rent on state land under AS 38.05.810 may not sublease. DMLW may conduct further agency review and/or public notice before making a determination on the appropriateness of the proposed sublease. The sublease fee will not be less than 25% of the annual fee paid to the lessee by the sublessee.

Reclamation

In accordance with AS 38.05.090, the leasehold must be restored to a "good and marketable condition" as determined by DMLW within 120 days after termination of the lease.

Public Notice

Pursuant to AS 38.05.945, this PD will be noticed for a 30-day public comment period, starting on September 1st, 2022. In addition, the post office(s) located near the proposed leasehold will be requested to post the notice pursuant to AS 38.05.945(b)(3)(C). The notice will also be posted on the State of Alaska Online Public Notice website pursuant to AS 38.05.945(b)(3)(B) located at: <u>https://aws.state.ak.us/OnlinePublicNotices/Default.aspx</u>. In accordance with AS 38.05.946, a municipality or a corporation entitled to receive notice under AS 38.05.945(c) may hold a hearing within 30 days after the receipt of the notice.

Comments

The public is invited to comment on this PD. All comments received during the public comment period will be considered in the FFD. If public comments result in significant changes to the PD, additional public notice may be given. A copy of the FFD, along with instructions on filing an appeal, will be sent to all persons who comment on the PD. To be eligible to appeal, a person affected by the FFD must provide written comments during the public comment period per AS 38.05.035(i).

Written comments about this project must be received in this office no later than 5:00 PM on October 3rd, 2022 to be considered.

To submit comments please choose one of the following methods:

Mail: Department of Natural Resources Division of Mining, Land and Water Southeast Region Office ATTN: *Megan Hillgartner* 400 Willoughby Avenue, 4th Floor

Preliminary Decision, Lease Amendment ADL 108979 Page 11 of 12

	PO Box 111020
	Juneau, AK 99811-1020
Email:	megan.hillgartner@alaska.gov
Fax:	(907) 500 - 9011

Ouestions about the lease portion of this project can be directed to Megan Hillgartner at (907) 465-3401.

Signature page follows:

Adjudicator Recommendation

Based on the information provided by the applicant and other agencies, as well as review of planning documents, statutes, and regulations, I recommend approving a 10-year lease to the applicant authorizing the applicant to construct a backcountry public use cabin and outhouse on the southern flanks of Mt. Tukgahgo. I recommend proceeding to public notice for the purpose of providing the members of the public and those entities identified in AS 38.05.945 an opportunity to review and submit comments.

Megan Hillgartner	9/1/2022	
Megan Hillgartner, Natural Resources Manager	Date	

Division of Forestry Concurrence

Greg Palmieri, Forester, Haines

Preliminary Decision

It is the determination of the Division of Mining, Land & Water that it may be in the State's best interest to issue a 10-year lease to the applicant, as described above. This preliminary decision shall now proceed to public notice.

Benjamin White, Southeast Regional Manager

September 1, 2022 Date

Date

ATTACHMENTS:

Attachment 1. Development Plan Attachment 2. Standard Lease Agreement and Special Stipulations

Preliminary Decision, Lease Amendment ADL 108979 Page 12 of 12

Haines Huts and Trails

Haines Hutst Land Lease Development Plan 11, 28, 2021



Introduction

Haines, unlike most communities in Southeast Alaska, does not have a federal or state agency establishing and operating backcountry cabins for public use. Haines Huts, a 501c3 federal nonprofit, was established to fill this gap. The mission of Haines Huts and Trails is to foster wilderness experiences in some of our region's most inspiring locations by developing and maintaining public use backcountry cabins and trails. The purpose of these backcountry cabins is to offer a place to foster wilderness education opportunities for all ages, a jumping-off point for backcountry adventures and a safety net for emergency situations.

Starting in 2012 Haines Huts have been meeting with Haines State Forest users and the Haines State Forest office staff to identify and prioritize potential locations for the first backcountry hut established and operated by Haines Huts.

The development of backcountry cabins is supported in the Haines Borough Comprehensive Plan and the Haines State Forest Five-Year Forest Management Schedule for calendar years 2018-2022. The Tukga-Hut location is specifically mentioned in the 2018-22 Management Schedule.

The Tukga-Hut will provide residents and visitors the opportunity to experience the Haines State Forest in a way that supports and enhances the primary purposes of the Haines State Forest Resource Management Area, outlined in the Haines State Forest Management Plan.

This Development Plan is accompanied by the Department of Natural Resources' Division of Mining Land and Water Application for Public and Charitable Use Purchase of State Land Lease Application, Agreement to Bear costs, Applicant

Haines Huts and Trails: Land Lease Development Plan- November, 2021

Environmental Risk Questionnaire, and kmz file,

Legal description

SEC 11, T30S, R58E, Copper River Meridian

See *tukgahut_lease_21.kmz* in attached file.

Corners of the proposed lease location: 59°17′6.60″N 135°36′20.53"W 59°17′8.08"N 135°36′21.09"W 59°17′7.09"N 135°36′3.82"W 59°17′5.96"N 135°36′4.27"W

Terrain/ground cover

The proposed hut will be approximately 8 miles north of downtown Haines, Alaska, in the Takshanuk Mountain Range, on the south flanks of Mount Tukgahgo. It will be at 3,200ft, where Chilly Ridge climbs to the north above 7-mile saddle (See images on next page). The Takshanuk Mountain Range lies between the Chilkoot and Chilkat Rivers in northern Lynn Canal. The area is a geologically young and dynamic landscape that includes a mix of perennial snowfields, small glaciers, alpine meadows and rugged broken terrain that descends to a rocky, tidewater coastline (White and Barten 2008). The existing ground cover is patches of mountain Hemlock and dwarf conifers giving way to a mosaic of lush subalpine meadows and then a patchwork of alpine heather tundra and bare rock. 8 months of the year the ground is covered with snow.

The proposed changes to the terrain/ground cover for this project are minimal. The foundation for the 320 square foot cabin will be 2 rows of 3-4 footings.

Haines Huts and Trails: Land Lease Development Plan- November, 2021



View of proposed hut location from the North, looking southeast



View of proposed hut location from the South, looking northwest

Haines Huts and Trails: Land Lease Development Plan- November, 2021

Access

This project will not restrict public access.

In the summer, the most established access to the project location is via hiking from the "7 Mile Saddle Trailhead" at 7-Mile on the Haines Highway. The existing trail ascends to '7-Mile Saddle' then splits south to Peak 3920 and north up to Chilly Ridge. The Hut location is ~3 miles and 3200' from the trailhead. Materials and supplies for construction will be slung in by helicopter and deposited without landing. Alternative public access routes on the West side exist into 10-mile bowl and from the northeast down the east ridge of Mt Tukgahgo to the 'Glory Hole Road' and down from '7 Mile Saddle' to the Lutak Road.



Existing Mt Ripinski trail infrastructure and proposed Hut approach. Base image from Division of Forestry, Haines, 2008

Buildings and other structures

Construction will be limited to one 16'x20' Hut & one 4'x4' outhouse. The floor will be built with traditional wooden joists. The structure will be locally milled and crafted timber. The Hut will be semi-permanent, but will be built with reclamation in mind. The entire hut could be dismantled in one weekend. The hut will be pre-assembled in town, then broken down and slung up to the building site in modular sections.

Power source

There will be a small propane generator to support the weather monitoring, web cameras & data link. The hut will not have traditional 110v power.

Waste types, waste sources, and disposal methods

Pack it in, Pack it out. With no running water, human waste collected in 55-gallon drums via outhouse is the only anticipated waste that will be generated on site. Full drums will be removed annually via helicopter and disposed of according to state and local regulations at the Haines Borough sewer treatment facility. Cabin users will agree to a "leave no trace" ethic in the rental contract. If household garbage is left in cabin, Haines Huts will penalize users according to a rental contract and manage waste accordingly. Cameras are part of building design and will be used for enforcement

The hut location is just over a quarter mile from the nearest surface water stream.

Hazardous substances

Propane gas will be used in the cabin for cooking, heat & limited power infrastructure. A 250-gallon propane tank will be secured along the back gable end of the hut. The risk to environment has been greatly reduced by choosing to use propane gas instead of diesel or other liquid petroleum fuel. See SDS (propane_sds.pdf) in attached file.

Water supply

There will be no running water, and therefore no waste water. Users will bring their own drinking water. The closest water supply is a perennial stream just over a quarter mile North of the cabin. When snow is covering the ground, users may melt snow for drinking water.

Parking areas and storage areas

N/A

Number of people using the site

The cabin will have beds for 8 people. There will be one supervisor who is responsible for managing the site. The Haines Huts business model anticipates that the site will be used 100 days per year with a total of at least 250 user days per year

Maintenance and operations

The cabin will be maintained and operated by Haines Huts. Similar to many backcountry cabin systems operated around the world, Haines Huts long-term maintenance requirements for this type of cabin will include an annual helicopter service, which will fly maintenance materials & propane fuel in and fly human waste out. The cabin will have several web-linked cameras both outside and inside the hut. The exterior cameras will post real time condition images to the Haines Huts website and the Haines Huts manager will check the internal cameras regularly. The cameras will help ensure if and when damage to the property does occur the cabin manager is able to access the property and address damages.

Closure/reclamation plan

Haines Huts has secured a reclamation bond to cover the cost of deconstruction. The cabin has been designed to be dismantled quickly. Insulated panel construction allows for quick disassembly and removal by helicopter.

Sketch or blue line portion of the development plan

Foundation for the proposed hut will be concrete footings.







View from West



Plan view, ground floor



Plan view, sleeping loft & emergency egress